### UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

	FORM 1	0-Q
(Mark (	One)	
$\boxtimes$ Q	QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15	5(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the quarterly period ended OR	1 September 30, 2023
□ T	RANSITION REPORT PURSUANT TO SECTION 13 OR 15	5(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the transition period for Commission File Number:	romto 001-39757
	Velo3D,	Inc.
	(Exact name of registrant as s	pecified in its charter)
	Delaware	98-1556965
(State or ot	her jurisdiction of incorporation or organization)	(I.R.S. Employer Identification No.)
4	511 Division Street, Campbell, CA	95008

(408) 610-3915

(Zip Code)

(Address of Principal Executive Offices)

Registrant's telephone number, including area code

Securities registered pursua	ant to Section 12(b) of the Act:		
	Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common stock, par value S	\$0.00001 per share	VLD	New York Stock Exchange
Warrants to purchase one s \$11.50 per share	share of common stock, each at an exercise price of	VLD WS	New York Stock Exchange
such shorter period that the re Indicate by check mark whetl	egistrant was required to file such reports), and (2) has been s	subject to such filing requirements ve Data File required to be submitt	ed pursuant to Rule 405 of Regulation S-T (§ 232.405 of this
Indicate by check mark whetl		iler, a non-accelerated filer, a small	er reporting company, or an emerging growth company. See the
Large accelerated filer			Accelerated filer $\square$
Non-accelerated filer	$\boxtimes$		Smaller reporting company ⊠
			Emerging growth company ⊠
	any, indicate by check mark if the registrant has elected not to Section 13(a) of the Exchange Act. $\Box$	o use the extended transition period	for complying with any new or revised financial accounting
Indicate by check mark whetl	her the registrant is a shell company (as defined in Rule 12b-	-2 of the Act). Yes □ No ⊠	
As of November 16, 2023,	the registrant had 200,793,531 shares of common stock,	, \$0.00001 per share outstanding	

### TABLE OF CONTENTS

		Page
	Part I. Financial Information	
Item 1.	Financial Statements (unaudited)	4
	Condensed Consolidated Balance Sheets as of September 30, 2023 and December 31, 2022 (unaudited)	4
	Condensed Consolidated Statements of Operations and Comprehensive Loss for the Three and Nine Months Ended September 30, 2023 and 2022 (unaudited)	5
	Condensed Consolidated Statements of Cash Flows for the Nine Months Ended September 30, 2023 and 2022 (unaudited)	7
	Condensed Consolidated Statements of Stockholders' Equity for the Three and Nine Months Ended September 30, 2023 and 2022 (unaudited)	9
	Notes to Condensed Consolidated Financial Statements (unaudited)	10
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	36
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	56
Item 4.	Controls and Procedures	56
	Part II. Other Information	
Item 1.	Legal Proceedings	59
Item 1A.	Risk Factors	60
Item 2.	Unregistered Sales of Equity Securities, Use of Proceeds and Issuer Purchases of Equity Securities	64
Item 3.	Defaults Upon Senior Securities	64
Item 4.	Mine Safety Disclosures	64
Item 5.	Other Information	64
Item 6.	Exhibits	65
Signatures		66

### **Explanatory Note**

Unless otherwise stated in this Quarterly Report or the context otherwise requires, references to:

- "Legacy Velo3D" refer to Velo3D, Inc., a Delaware corporation, prior to the closing of the Merger;
- "Merger" refer to the merger contemplated by that certain Business Combination Agreement, dated as of March 22, 2021, by and among JAWS Spitfire Acquisition Corporation, a Cayman Islands exempted company ("JAWS Spitfire"), Legacy Velo3D and Spitfire Merger Sub, Inc., a Delaware corporation ("Merger Sub"), as amended by Amendment No. 1 to the Business Combination Agreement, dated as of July 20, 2021 (the "Business Combination Agreement"), whereby Merger Sub merged with and into Legacy Velo3D, with Legacy Velo3D surviving the merger as a wholly-owned subsidiary of the Company, on September 29, 2021;
- "Velo3D" refer to Velo3D, Inc., a Delaware corporation (f/k/a JAWS Spitfire Acquisition Corporation, prior to its domestication), and its consolidated subsidiaries following the closing of the Merger;
- "we," "us," and "our" or the "Company" refer to Velo3D following the closing of the Merger and to Legacy Velo3D prior to the closing of the Merger; and
- "2022 Form 10-K" refer to our Annual Report on Form 10-K for the year-ended December 31, 2022, filed with the Securities and Exchange Commission (the SEC") on March 20, 2023.

"Velo", "Velo3D", "Sapphire" and "Intelligent Fusion" are registered trademarks of Velo3D, Inc; and "Without Compromise", "Flow" and "Assure" are trademarks of Velo3D, Inc.

#### PART I. FINANCIAL INFORMATION

#### Forward-looking Statements

Certain statements in this Quarterly Report may constitute "forward-looking statements" for purposes of the federal securities laws. Our forward-looking statements include, but are not limited to, statements regarding our or our management team's expectations, hopes, beliefs, intentions or strategies regarding the future. In addition, any statements that refer to projections, forecasts or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. The words "anticipate," "believe," "can," "contemplate," "continue," "could," "estimate," "expect," "forecast," "intend," "may," "might," "plan," "possible," "potential," "predict," "project," "seek," "should," "target," "will," "would" and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking. Forward-looking statements in this Quarterly Report may include, for example, statements about:

- · our market opportunity;
- the ability to maintain the listing of our common stock and our public warrants on the New York Stock Exchange (the "NYSE"), and the potential liquidity and trading
  of such securities:
- our ability to execute our business plan, which may be affected by, among other things, competition and our ability to grow and manage growth profitably, maintain relationships with customers and retain our key employees;
- · changes in applicable laws or regulations;
- · the inability to develop and maintain effective internal control over financial reporting;
- our ability to service and comply with our indebtedness, including the existing event of default underour senior secured convertible notes due 2026 and our negotiations with the holders of the notes regarding a proposed amendment to the notes;
- our ability to raise financing in the future;
- · our success in retaining or recruiting, or changes required in, our officers, key employees or directors;
- the period over which we anticipate our existing cash and cash equivalents will be sufficient to fund our operating expenses and capital expenditure requirement and our ability to continue as a going concern;
- the potential for our business development efforts to maximize the potential value of our portfolio;
- · regulatory developments in the United States and foreign countries;
- · the impact of laws and regulations;
- · our expectations regarding our strategic realignment and related initiatives;
- · our estimates regarding expenses, future revenue, capital requirements and needs for additional financing;
- our financial performance;
- macroeconomic conditions, including economic downturns or recessions, inflation, interest rate fluctuations, supply chain shortages and any lingering effects of the COVID-19 pandemic on the foregoing; and
- other factors detailed under the section entitled "Risk Factors".

The forward-looking statements contained in this Quarterly Report are based on current expectations and beliefs concerning future developments and their potential effects on us. There can be no assurance that future developments affecting us will be those that we have anticipated. These forward-looking statements involve a number of risks, uncertainties (some of which are beyond our control) or other assumptions that may cause actual results or performance to be materially different from those expressed or implied by these forward-looking statements. These risks and uncertainties include, but are not limited to, those factors described under the section entitled "Risk Factors". Should one or more of these risks or uncertainties materialize, or should any of our assumptions prove incorrect, actual results may vary in material respects from those projected in these forward-looking statements. Some of these risks and uncertainties may in the future be amplified by the impact of other macroeconomic factors and there may be additional risks that we consider immaterial or which are unknown. It is not possible to predict or identify all such risks. We do not undertake any obligation to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

### **Item 1. Financial Statements**

## Velo3D, Inc. CONDENSED CONSOLIDATED BALANCE SHEETS (Unaudited)

(In thousands, except share and per share data)

(in thousands, except snare and per snare data)				
	Se	ptember 30,	Decemb	,
		2023	202	22
Assets				
Current assets:				
Cash and cash equivalents	\$	58,131	\$	31,983
Short-term investments		13,468		48,214
Accounts receivable, net		12,597		9,185
Inventories		81,159		71,202
Contract assets		15,901		6,805
Prepaid expenses and other current assets		2,515		5,533
Total current assets		183,771		172,922
Property and equipment, net		17,430		19,812
Equipment on lease, net		7,020		9,070
Other assets		21,782	1	23,310
Total assets	\$	230,003	\$	225,114
Liabilities and Stockholders' Equity	<u> </u>			
Current liabilities:				
Accounts payable	\$	13,135	\$	12,207
Accrued expenses and other current liabilities		11,215		15,877
Debt – current portion		52,211		2,775
Debt derivatives		17,538		_
Contract liabilities		4,847		15,194
Total current liabilities		98,946		46,053
Long-term debt – less current portion		_		5,422
Contingent earnout liabilities		14,414		17,414
Warrant liabilities		2,883		2,745
Other noncurrent liabilities		10,805		12,634
Total liabilities		127,048		84,268
Commitments and contingencies (Note 13)				
Stockholders' equity:				
Common stock, \$0.00001 par value - 500,000,000 shares authorized at September 30, 2023 and December 31, 2022, 200,314,984 and 187,561,368 shares issued and outstanding as of September 30, 2023 and December 31, 2022, respectively		2		2
Additional paid-in capital		399,847		361,528
Accumulated other comprehensive loss		(252)		(837)
Accumulated deficit		(296,642)		(219,847)
Total stockholders' equity		102,955		140,846
	\$	230,003	\$	225,114
Total liabilities and stockholders' equity	φ	430,003	Ψ	443,114

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

# Velo3D, Inc. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (Unaudited) (In thousands, except share and per share data)

	7	Three Months Ended September 30,				Nine Months Ended September 30,				
		2023		2022		2023		2022		
Revenue										
3D Printer	\$		\$	16,537	\$	69,193	\$	44,336		
Recurring payment		531		1,183		1,141		3,042		
Support services		1,849		1,395		5,422		3,599		
Total Revenue		23,808		19,115		75,756		50,977		
Cost of revenue										
3D Printer		20,273		16,574		61,975		42,686		
Recurring payment		111		656		893		2,059		
Support services		1,936		2,006		5,495		5,106		
Total cost of revenue		22,320		19,236		68,363		49,851		
Gross profit (loss)		1,488		(121)		7,393		1,126		
Operating expenses										
Research and development		9,819		12,558		32,820		38,438		
Selling and marketing		5,772		5,632		18,054		17,864		
General and administrative		11,118		9,642		31,569		27,191		
Total operating expenses		26,709		27,832		82,443		83,493		
Loss from operations		(25,221)		(27,953)		(75,050)		(82,367)		
Interest expense		(1,107)		(129)		(1,671)		(362)		
Gain (loss) on fair value of warrants		1,587		(6,612)		(138)		11,039		
Gain (loss) on fair value of contingent earnout liabilities		10,810		(40,885)		3,000		58,110		
Loss on fair value of debt derivatives		(3,648)				(3,648)		´—		
Loss on debt extinguishment		(253)		_		(253)		_		
Other income, net		436		384		965		993		
Loss before provision for income taxes		(17,396)		(75,195)		(76,795)		(12,587)		
Provision for income taxes		_		_		_		_		
Net loss	\$	(17,396)	\$	(75,195)	\$	(76,795)	\$	(12,587)		
Net loss per share:										
Basic	\$	(0.09)	\$	(0.41)	\$	(0.40)	\$	(0.07)		
Diluted	\$	(0.09)	\$	(0.41)	\$	(0.40)	\$	(0.07)		
Shares used in computing net loss per share:										
Basic		197,833,109		185,560,177		193,816,804		184,454,371		
Diluted		197,833,109		185,560,177		193,816,804		184,454,371		
Net loss	\$	(17,396)	\$	(75,195)	\$	(76,795)	\$	(12,587)		

Net unrealized holding gain (loss) on available-for-sale investments	 149	(178)	585	(1,121)
Total comprehensive loss	\$ (17,247)	\$ (75,373)	\$ (76,210)	\$ (13,708)

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

# Velo3D, Inc. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (In thousands)

Net loss         \$ (76,795)         \$ (12,887)           Adjustments to reconcile net loss to net eash used in operating activities         4,516         3,328           Depreciation and amortization         4,516         3,328           Stock-based compensation         19,486         15,090           (Gain) loss on fair value of contingent earnout liabilities         3,648         1,100           Loss on fair value of debt derivatives         3,648         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         (1,867)         (1,876)         (1,876)         (1,876)         (2,887)         (3,818)         (1,876)         (1,876)         (2,887)         (3,816)         (3,816)         (3,816)         (3,816)         (3,816)         (3,816)         (3,817)		I	Nine months ended Septembe			
Net loss         \$ (76,795)         \$ (12,887)           Adjustments to reconcile net loss to net eash used in operating activities         4,516         3,328           Depreciation and amortization         4,516         3,328           Stock-based compensation         19,486         15,090           (Gain) loss on fair value of contingent earnout liabilities         3,648         1,100           Loss on fair value of debt derivatives         3,648         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         —           Loss on fair value of debt derivatives         3,548         (1,867)         (1,876)         (1,876)         (1,876)         (2,887)         (3,818)         (1,876)         (1,876)         (2,887)         (3,816)         (3,816)         (3,816)         (3,816)         (3,816)         (3,816)         (3,817)			2023	2022		
Aginemats to reconcile net loss to net eash used in operating activities   4,516   3,328	Cash flows from operating activities					
Depreciation and amortization         4,516         3,328           Stock-based compensation         19,486         15,090           (Gain) loss on fair value of varrants         3138         (11,039)           Gain on fair value of cottingent earnout liabilities         3,048         9-1           Loss on fair value of devidervatives         3,648         9-1           Loss on debt extinguishment         233         9-1           Changes in assets and liabilities         (3,412)         (8,144)           Inventories         (3,312)         (4,807)           Contract assets         (9,096)         (2,096)           Opposite expenses and other current assets         3,503         7,342           Other assets         1,503         1,970           Accounts payable         (1,556)         1,177           Accured expenses and other liabilities         (1,556)         1,177           Accured and in operating activities         (8,109)         (2,245)           Other assets         (1,0347)         3,788           Other assets and other liabilities         (1,037)         3,789           Other assets and other juilities         (1,037)         3,789           Other assets and other juilities         (1,037)         3,789	Net loss	\$	(76,795) \$	(12,587)		
Stock-based compensation         19,486         15,090           (Gain) loss on fair value of warrants         138         (11,309)           Gain on fair value of contingent earnout liabilities         3,040         (8,8110)           Loss on fair value of clebt derivatives         253         —           Loss on debt extinguishment         253         —           Changes in assets and liabilities         —         (3,412)         (8,144)           Accounts receivable         (3,739)         (14,807)         (1,807)           Contract assets         (9,096)         (2,096)         (2,096)           Prepaid expenses and other current assets         3,503         7,342           Other assets         1,503         1,970           Accounts payable         (1,556)         1,177           Accorned expenses and other liabilities         (1,322)         (1,127)           Other assets         (10,347)         3,789           Other noncurrent liabilities         (1,037)         (3,289)           Other noncurrent liabilities         (1,047)         (1,829)           Other noncurrent liabilities         (1,047)         (2,228)           Other noncurrent liabilities         (1,022)         (2,141)           Purchase of property and equipmen	Adjustments to reconcile net loss to net cash used in operating activities					
Gain jo ns in value of contingent earmout liabilities         3,000         (58,110)           Loss on fair value of contingent earmout liabilities         3,648         —           Loss on debt extinguishment         253         —           Changes in assets and liabilities         —         Changes in assets and liabilities           Accounts receivable         (3,412)         (8,144)           Inventories         (3,909)         (2,096)           Prepaid expenses and other current assets         9,996         (2,096)           Prepaid expenses and other current assets         1,503         1,742           Accounts payable         (1,556)         1,177           Accrued expenses and other liabilities         (10,347)         3,789           Other assets         (10,347)         3,789           Other assets in operating activities         (81,109)         12,125           Net cash used in operating activities         (81,109)         12,214           Production of equipment for lease to customers         (2,965)         4,174           Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         4,174           Purchase of property and equipment activities         31,055         (74,507)	Depreciation and amortization		4,516	3,328		
Gain on fair value of contingent eamout liabilities         3,000         (58,110)           Loss on fair value of cloth derivatives         3,648         —           Loss on debt extinguishment         253         —           Changes in assets and liabilities         —         (8,144)           Inventories         (3,739)         (41,807)           Contract assets         (9,906)         (2,906)           Contract assets         3,503         7,342           Other assets         1,503         1,973           Accounts payable         (1,556)         1,177           Accrued expenses and other liabilities         (1,356)         1,177           Accrued expenses and other liabilities         (1,0347)         3,789           Other assets         (1,010)         1,480           Contract liabilities         (1,0347)         3,789           Other noncurrent liabilities         (1,0347)         3,789           Net cash used in operating activities         (81,109)         (92,154)           Total property and equipment         (1,072)         (1,2228)           Purchase of property and equipment for lease to customers         (81,109)         (92,154)           Proceeds from maturity of available-for-sale investments         —         (87,655) </td <td>Stock-based compensation</td> <td></td> <td>19,486</td> <td>15,090</td>	Stock-based compensation		19,486	15,090		
Loss on fair value of debt derivatives         3,648         —           Loss on debt extinguishment         253         —           Changes in assets and liabilities         (3,412)         (8,144)           Accounts receivable         (3,739)         (14,807)           Contract assets         (9,096)         (2,096)           Prepaid expenses and other current assets         3,503         7,342           Other assets         1,503         1,970           Accounts payable         (1,556)         1,177           Accounts payable         (10,347)         3,83           Other assets         (10,147)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (10,228)         (1,215)           Net cash profine investing activities         (1,022)         (1,212)           Proceash from investing activities         (2,065)         (4,174)           Production of equipment for lease to customers         (2,065)         (4,174)           Production of property and equipment for lease to customers         (2,065)         (4,255)	(Gain) loss on fair value of warrants		138	(11,039)		
Loss on debt extinguishment         253         —           Changes in assets and liabilities         3,412         (8,144)           Inventories         (3,739)         (41,807)           Contract saces         (9,96)         (2,096)           Prepaid expenses and other current assets         3,503         7,342           Other assets         1,503         1,970           Accounts payable         (1,656)         1,177           Account clabilities         (10,347)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (1,02)         (1,225)           Net cash used in operating activities         (81,109)         (2,154)           Cash Hows from investing activities         (10,02)         (12,228)           Proceds from prepry and equipment         (1,072)         (12,228)           Proceds of property and equipment for lease to customers         (2,965)         (4,174)           Purchases of property and equipment for lease to customers         (2,955)         (4,174)           Proceds from maturity of available-for-sale investments         3,502         29,550           Net cash provided by (used in) investin	Gain on fair value of contingent earnout liabilities		(3,000)	(58,110)		
Changes in assets and liabilities         3,412         (8,144)           Accounts receivable         (3,739)         (41,867)           Inventories         (9,096)         (2,096)           Contract assets         (9,096)         (2,096)           Prepaid expenses and other current assets         3,503         7,342           Other assets         1,503         1,977           Accounts payable         (1,556)         1,177           Accrued expenses and other liabilities         (10,347)         3,898           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other cash used in operating activities         (1,02)         (1,215)           Net cash used in operating activities         (1,02)         (1,228)           Production of equipment of lease to customers         (2,96)         (4,174)           Production of equipment of lease to customers         (2,96)         (4,174)           Proceds from maturity of available-for-sale investments         2         (8,055)           Proceds from fount primare, act of issuance costs         3,05         (7,550)           Act ash provided by (used in junvesting activities         5,50         2           Proceeds from conve	Loss on fair value of debt derivatives		3,648	_		
Accounts receivable         (3,412)         (8,144)           Inventories         (3,739)         (41,807)           Contract asets         (9,096)         (2,096)           Prepaid expenses and other current assets         3,503         7,342           Other assets         1,503         1,970           Accounts payable         (1,556)         1,177           Accrued expenses and other liabilities         (10,347)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (81,109)         (2,215)           Net cash used in operating activities         (81,109)         (2,215)           Purchase of property and equipment         (1,072)         (1,2228)           Production of equipment for lease to customers         (2,965)         (4,174)           Purchases of available-for-sale investments         2,965)         (4,174)           Proceach from maturity of available-for-sale investments         31,055         (74,507)           Proceeds from from furtier, and vailable-for-sale investments         31,055         (74,507)           Proceeds from from furtier, and vailable-for-sale investments         31,055         (74,507)	Loss on debt extinguishment		253	_		
Inventories	Changes in assets and liabilities					
Contract assets         (9,096)         (2,096)           Prepaid expenses and other current assets         3,503         7,342           Other assets         1,503         1,970           Accounts payable         (1,556)         1,177           Accounts payable         (1,556)         1,177           Accounts payable         (1,0347)         3,789           Other noncurrent liabilities         (1,0347)         3,789           Other noncurrent liabilities         (1,0347)         3,789           Other noncurrent liabilities         (1,032)         (1,215)           Net cash used in operating activities         8(1,09)         (2,154)           Cash flows from investing activities         1,072         (12,228)           Production of equipment for lease to customers         (1,072)         (12,228)           Proceads for investing activities         - (87,655)         (4,174)           Proceads from maturity of available-for-sale investments         - (87,655)         (74,507)           Res (500) In provided by (used in) investing activities         - (87,655)         (74,507)           Res (500) In financing activities         - (87,655)         (74,507)           Res (500) In financing activities         - (8,664)         (8,899)           Proceeds from	Accounts receivable		(3,412)	(8,144)		
Prepaid expenses and other current assets         3,503         7,342           Other assets         1,503         1,970           Accounts payable         (1,556)         1,177           Accrued expenses and other liabilities         (10,347)         3,789           Other noncurrent liabilities         (10,347)         3,789           Other noncurrent liabilities         (81,109)         (92,154)           Net cash used in operating activities         (81,109)         (92,154)           Net cash from investing activities         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         (4,174)           Production of equipment for lease to customers         (2,965)         (4,174)           Production of equipment for lease to customers         (2,965)         (4,174)           Proceds from maturity of available-for-sale investments         31,055         (7,505)           Net cash provided by (used in) investing activities         31,055         (7,507)           Cash flows from financing activities         31,055         (7,507)           Cash flows from finance, net of issuance costs         —         6,664           Repayment of loans in connection with loan refinance         —         6,8089           Proceeds from ATM offering, net of issuance	Inventories		(3,739)	(41,807)		
Other assets         1,503         1,970           Accounts payable         (1,556)         1,177           Accrued expenses and other liabilities         (1,382)         10,148           Contract liabilities         (10,347)         3,789           Other noncurrent liabilities         (18,109)         (92,154)           Net cash used in operating activities         (81,009)         (92,154)           Cash flows from investing activities           Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         (4,174)           Purchases of available-for-sale investments         2,965)         (4,174)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         —         6,664           Repayment of loans refinance, net of issuance costs         —         6,664           Repayment of loans in connection with loan refinance costs         —         6,664           Repayment of loans in connection with loan refinance costs         65,736         —           Proceeds from Convertible note, net of issuance costs         18,423	Contract assets		(9,096)	(2,096)		
Accounts payable         (1,556)         1,177           Accrued expenses and other liabilities         (4,382)         10,148           Contract liabilities         (10,347)         3,789           Other noncurrent liabilities         (1,829)         (1,215)           Net cash used in operating activities         (81,09)         (92,154)           Purchase of property and equipment         (1,072)         (1,2228)           Purchases of available-for-sale investments         (2,965)         (4,174)           Purchases of available-for-sale investments         -         (87,655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         -         6,664           Repayment of loans in connection with loan refinance         -         6,664           Repayment of loans in connection with loan refinance         -         8,089           Proceeds from ATM offering, net of issuance costs         65,736         -           Proceeds from equipment loans         1,600         -           Proceeds from equipment loans         1,600         -           Repayment of requipment loans         6,950         (355	Prepaid expenses and other current assets		3,503	7,342		
Accrued expenses and other liabilities         (4,382)         10,148           Contract liabilities         (10,347)         3,789           Other noncurrent liabilities         (1,829)         (1,215)           Net cash used in operating activities         (81,109)         (92,154)           Cash flows from investing activities           Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         (4,174)           Purchase of available-for-sale investments         -         (87,655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         -         6,664           Repayment of loans in connection with loan refinance         -         6,664           Repayment of loans in connection with loan refinance         -         6,808           Proceeds from convertible note, net of issuance costs         65,736         -           Proceeds from equipment loans         1,600         -           Proceeds from equipment loans         (6,956)         (355)           Repayment of requipment loans         (6,956)	Other assets		1,503	1,970		
Contract liabilities         (10,347)         3,789           Other noncurrent liabilities         (1,829)         (1,215)           Net cash used in operating activities         (81,109)         (92,154)           Cash flows from investing activities         """>""">"""           Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (1,072)         (12,228)           Purchases of available-for-sale investments         2,056         (4,174)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         —         6,664           Repayment of loans in connection with loan refinance         —         6,664           Repayment of loans in connection with loan refinance         —         6,808           Proceeds from convertible note, net of issuance costs         —         6,604           Repayment of loans in connection with loan refinance         —         6,808           Proceeds from equipment loans         18,423         —           Proceeds from revolver facility         14,000         —           Repayment of equipment loans         (6,95	Accounts payable		(1,556)	1,177		
Contract liabilities         (10,347)         3,789           Other noncurrent liabilities         (1,829)         (1,215)           Net cash used in operating activities         (81,009)         (2,9154)           Cash flows from investing activities           Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         (4,174)           Purchase of available-for-sale investments         —         (87,655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         —         6,664           Repayment of loans in connection with loan refinance costs         —         6,664           Repayment of loans in connection with loan refinance         —         6,664           Repayment of sum accounter flow, net of issuance costs         —         6,664           Proceeds from convertible note, net of issuance costs         —         6,664           Proceeds from ATM offering, net of issuance costs         —         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (1,000)         —           Repayment of equipment loans         (6,956)	Accrued expenses and other liabilities		(4,382)	10,148		
Net cash used in operating activities         (81,109)         (92,154)           Cash flows from investing activities         8           Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         (4,174)           Purchases of available-for-sale investments         –         (87,655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         –         6,664           Repayment of loans in connection with loan refinance costs         –         6,664           Repayment of loans in connection with loan refinance costs         –         6,889           Proceeds from convertible note, net of issuance costs         –         6,889           Proceeds from revolver facility         14,000         –           Proceeds from revolver facility         1,600         –           Repayment of revolver facility         (17,000)         –           Repayment of equipment loans         (6,956)         (355)           Issue of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activ			(10,347)	3,789		
Cash flows from investing activities           Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         (4,174)           Purchases of available-for-sale investments         —         (8,7655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         —         (8,087)           Cash flows from financing activities         —         (8,089)           Proceeds from loan refinance, net of issuance costs         —         (8,089)           Proceeds from convertible note, net of issuance costs         65,736         —           Proceeds from ATM offering, net of issuance costs         18,423         —           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of exchange rate changes on cash and cash equivalent	Other noncurrent liabilities		(1,829)	(1,215)		
Cash flows from investing activities           Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         (4,174)           Purchases of available-for-sale investments         —         (8,7655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         —         (8,087)           Cash flows from financing activities         —         (8,089)           Proceeds from loan refinance, net of issuance costs         —         (8,089)           Proceeds from convertible note, net of issuance costs         65,736         —           Proceeds from ATM offering, net of issuance costs         18,423         —           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of exchange rate changes on cash and cash equivalent	Net cash used in operating activities		(81,109)	(92,154)		
Purchase of property and equipment         (1,072)         (12,228)           Production of equipment for lease to customers         (2,965)         (4,174)           Purchases of available-for-sale investments         —         (87,655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         —         6,664           Net cash provided by (used in) investing activities         —         6,664           Repayment of loans in connection with loan refinance         —         6,664           Repayment of loans in connection with loan refinance         —         6,8989           Proceeds from convertible note, net of issuance costs         —         6,664           Repayment of flering, net of issuance costs         —         6,664           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of						
Production of equipment for lease to customers         (2,965)         (4,174)           Purchases of available-for-sale investments         — (87,655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         — 6,664           Repayment of loans in connection with loan refinance         — (8,089)           Proceeds from loan refinance, net of issuance costs         — (8,089)           Proceeds from ATM offering, net of issuance costs         — (8,089)           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of exchange rate changes on cash and cash equivalents         26,148         (167,255)           Net change in cash and cash equivalents         26,148         (167,255)           Cash and cash equivalents and restricted cash at beginning of period <td>g .</td> <td></td> <td>(1,072)</td> <td>(12,228)</td>	g .		(1,072)	(12,228)		
Purchases of available-for-sale investments         —         (87,655)           Proceeds from maturity of available-for-sale investments         35,092         29,550           Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         —         6,664           Proceeds from loan refinance, net of issuance costs         —         6,664           Repayment of loans in connection with loan refinance         —         (8,089)           Proceeds from convertible note, net of issuance costs         65,736         —           Proceeds from equipment loans and TM offering, net of issuance costs         18,423         —           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (337)           Effect of exchange rate changes on cash and cash equivalents         (11)         (57)           Net cash grovided by (used in) financing activities         (26,148         (167,255)<			(2,965)	(4,174)		
Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         —         6,664           Proceeds from loan refinance, net of issuance costs         —         6,664           Repayment of loans in connection with loan refinance         —         (8,089)           Proceeds from convertible note, net of issuance costs         65,736         —           Proceeds from ATM offering, net of issuance costs         18,423         —           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of exchange rate changes on cash and cash equivalents         (11)         (57)           Net change in cash and cash equivalents         26,148         (167,255)           Cash and cash equivalents and restricted cash at beginning of period         32,783         208,402	Purchases of available-for-sale investments			(87,655)		
Net cash provided by (used in) investing activities         31,055         (74,507)           Cash flows from financing activities         —         6,664           Proceeds from loan refinance, net of issuance costs         —         6,664           Repayment of loans in connection with loan refinance         —         (8,089)           Proceeds from convertible note, net of issuance costs         65,736         —           Proceeds from ATM offering, net of issuance costs         18,423         —           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of exchange rate changes on cash and cash equivalents         (11)         (57)           Net change in cash and cash equivalents         26,148         (167,255)           Cash and cash equivalents and restricted cash at beginning of period         32,783         208,402	Proceeds from maturity of available-for-sale investments		35,092	29,550		
Cash flows from financing activities           Proceeds from loan refinance, net of issuance costs         —         6,664           Repayment of loans in connection with loan refinance         —         (8,089)           Proceeds from convertible note, net of issuance costs         65,736         —           Proceeds from ATM offering, net of issuance costs         18,423         —           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of exchange rate changes on cash and cash equivalents         (11)         (57)           Net change in cash and cash equivalents         26,148         (167,255)           Cash and cash equivalents and restricted cash at beginning of period         32,783         208,402	•			(74 507)		
Proceeds from loan refinance, net of issuance costs         —         6,664           Repayment of loans in connection with loan refinance         —         (8,089)           Proceeds from convertible note, net of issuance costs         65,736         —           Proceeds from ATM offering, net of issuance costs         18,423         —           Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of exchange rate changes on cash and cash equivalents         (11)         (57)           Net change in cash and cash equivalents         26,148         (167,255)           Cash and cash equivalents and restricted cash at beginning of period         32,783         208,402			51,000	(7.,507)		
Repayment of loans in connection with loan refinance       —       (8,089)         Proceeds from convertible note, net of issuance costs       65,736       —         Proceeds from ATM offering, net of issuance costs       18,423       —         Proceeds from revolver facility       14,000       —         Proceeds from equipment loans       1,600       —         Repayment of revolver facility       (17,000)       —         Repayment of equipment loans       (6,956)       (355)         Issuance of common stock upon exercise of stock options       410       1,243         Net cash provided by (used in) financing activities       76,213       (537)         Effect of exchange rate changes on cash and cash equivalents       (11)       (57)         Net change in cash and cash equivalents       26,148       (167,255)         Cash and cash equivalents and restricted cash at beginning of period       32,783       208,402	<u> </u>		_	6.664		
Proceeds from ATM offering, net of issuance costs18,423—Proceeds from revolver facility14,000—Proceeds from equipment loans1,600—Repayment of revolver facility(17,000)—Repayment of equipment loans(6,956)(355)Issuance of common stock upon exercise of stock options4101,243Net cash provided by (used in) financing activities76,213(537)Effect of exchange rate changes on cash and cash equivalents(11)(57)Net change in cash and cash equivalents26,148(167,255)Cash and cash equivalents and restricted cash at beginning of period32,783208,402			_	(8,089)		
Proceeds from revolver facility         14,000         —           Proceeds from equipment loans         1,600         —           Repayment of revolver facility         (17,000)         —           Repayment of equipment loans         (6,956)         (355)           Issuance of common stock upon exercise of stock options         410         1,243           Net cash provided by (used in) financing activities         76,213         (537)           Effect of exchange rate changes on cash and cash equivalents         (11)         (57)           Net change in cash and cash equivalents         26,148         (167,255)           Cash and cash equivalents and restricted cash at beginning of period         32,783         208,402	Proceeds from convertible note, net of issuance costs		65,736			
Proceeds from equipment loans 1,600 — Repayment of revolver facility (17,000) — Repayment of equipment loans (6,956) (355) Issuance of common stock upon exercise of stock options 410 1,243 Net cash provided by (used in) financing activities 76,213 (537) Effect of exchange rate changes on cash and cash equivalents (11) (57) Net change in cash and cash equivalents 26,148 (167,255) Cash and cash equivalents and restricted cash at beginning of period 32,783 208,402	Proceeds from ATM offering, net of issuance costs			_		
Repayment of revolver facility(17,000)—Repayment of equipment loans(6,956)(355)Issuance of common stock upon exercise of stock options4101,243Net cash provided by (used in) financing activities76,213(537)Effect of exchange rate changes on cash and cash equivalents(11)(57)Net change in cash and cash equivalents26,148(167,255)Cash and cash equivalents and restricted cash at beginning of period32,783208,402	Proceeds from revolver facility		14,000	_		
Repayment of equipment loans(6,956)(355)Issuance of common stock upon exercise of stock options4101,243Net cash provided by (used in) financing activities76,213(537)Effect of exchange rate changes on cash and cash equivalents(11)(57)Net change in cash and cash equivalents26,148(167,255)Cash and cash equivalents and restricted cash at beginning of period32,783208,402	Proceeds from equipment loans		1,600	_		
Issuance of common stock upon exercise of stock options4101,243Net cash provided by (used in) financing activities76,213(537)Effect of exchange rate changes on cash and cash equivalents(11)(57)Net change in cash and cash equivalents26,148(167,255)Cash and cash equivalents and restricted cash at beginning of period32,783208,402	Repayment of revolver facility		(17,000)	_		
Net cash provided by (used in) financing activities76,213(537)Effect of exchange rate changes on cash and cash equivalents(11)(57)Net change in cash and cash equivalents26,148(167,255)Cash and cash equivalents and restricted cash at beginning of period32,783208,402	Repayment of equipment loans		(6,956)	(355)		
Effect of exchange rate changes on cash and cash equivalents(11)(57)Net change in cash and cash equivalents26,148(167,255)Cash and cash equivalents and restricted cash at beginning of period32,783208,402	Issuance of common stock upon exercise of stock options		410	1,243		
Effect of exchange rate changes on cash and cash equivalents(11)(57)Net change in cash and cash equivalents26,148(167,255)Cash and cash equivalents and restricted cash at beginning of period32,783208,402	Net cash provided by (used in) financing activities		76,213	(537)		
Net change in cash and cash equivalents26,148(167,255)Cash and cash equivalents and restricted cash at beginning of period32,783208,402			(11)	(57)		
Cash and cash equivalents and restricted cash at beginning of period 32,783 208,402				(167,255)		
	·			. , ,		
	Cash and cash equivalents and restricted cash at end of period	\$	58,931 \$	41,147		

Supplemental disclosure of cash flow information		
Cash paid for interest	\$ 1,123 \$	253
Supplemental disclosure of non-cash information		
Unpaid liabilities related to property and equipment	27	_
Equipment for lease to customers returned to inventory	4,198	2,619
Issuance of common stock warrants in connection with refinancing	_	170

The following table provides a reconciliation of cash, cash equivalents, and restricted cash shown on the condensed consolidated statements of cash flows:

		Septem	ıber 30,	
		2023		2022
	·	(In tho	usands)	
Cash and cash equivalents	\$	58,131	\$	40,347
Restricted cash (Other assets)		800		800
Total cash and cash equivalents and restricted cash	\$	58,931	\$	41,147

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

### $\label{eq:Velo3D} Velo3D, Inc.$ CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (Unaudited) (In thousands, except share data)

	Common Stock		Add	Accumulated Other Comprehensive Income						otal Stockholders'	
	Shares	A	mount	I	n Capital		(Loss)	Accumulated Deficit			Equity
Balance as of June 30, 2022	184,909,608	\$	2	\$	350,797	\$	(957)	\$	(167,259)	\$	182,583
Issuance of common stock upon exercise of stock options and release of restricted stock units	1,248,369		_		673		_	-	_		673
Stock-based compensation	254,841		_		5,157		_		_		5,157
Issuance of common stock warrants in connection with financing	_		_		(170)		_		_		(170)
Net loss	_		_		_		_		(75,195)		(75,195)
Other comprehensive loss	_		_		_		(178)		_		(178)
Balance as of September 30, 2022	186,412,818	\$	2	\$	356,457	\$	(1,135)	\$	(242,454)	\$	112,870
								_		_	
Balance as of June 30, 2023	196,737,320	\$	2	\$	390,240	\$	(401)	\$	(279,246)	\$	110,595
Issuance of common stock upon exercise of stock options	106,334				60						60
Issuance of common stock upon release of RSUs	1,479,555		_		_		_		_		_
Stock-based compensation	_		_		6,716		_		_		6,716
Issuance of common stock in connection with At-the-Market offering, net of issuance costs	1,991,775		_		2,831		_		_		2,831
Net loss	_		_				_		(17,396)		(17,396)
Other comprehensive income	_		_		_		149		` _		149
Balance as of September 30, 2023	200,314,984	\$	2	\$	399,847	\$	(252)	\$	(296,642)	\$	102,955
		_				=					

	Common Stock		_ Additional Paid-			Accumulated Other Comprehensive Income			т	otal Stockholders'	
	Shares	A	Amount	nt In Capital			(Loss)		cumulated Deficit	•	Equity
Balance as of December 31, 2021	183,232,494	\$	2	\$	340,294	\$	(14)	\$	(229,867)	\$	110,415
Issuance of common stock upon exercise of stock options and release of restricted stock units	2,795,000		_		1,243				_		1,243
Stock-based compensation	385,324		_		15,090		_		_		15,090
Issuance of common stock warrants in connection with financing	_		_		(170)		_		_		(170)
Net income	_		_		_		_		(12,587)		(12,587)
Other comprehensive loss	_		_		_		(1,121)		_		(1,121)
Balance as of September 30, 2022	186,412,818	\$	2	\$	356,457	\$	(1,135)	\$	(242,454)	\$	112,870
			:			_		_	-		
Balance as of December 31, 2022	187,561,368	\$	2	\$	361,528	\$	(837)	\$	(219,847)	\$	140,846
Issuance of common stock upon exercise of stock options	1,401,769		_		410		_		_		410
Issuance of common stock upon release of RSUs	3,119,282		_		_		_		_		_
Stock-based compensation	_		_		19,486		_		_		19,486
Issuance of common stock in connection with At-the-Market offering, net of issuance costs	8,232,565		_		18,423		_		_		18,423
Net loss	_		_		_		_		(76,795)		(76,795)
Other comprehensive income	_		_		_		585		` _		585
Balance as of September 30, 2023	200,314,984	\$	2	\$	399,847	\$	(252)	\$	(296,642)	\$	102,955

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

### Note 1. Description of Business and Basis of Presentation

Velo3D, Inc., a Delaware corporation ("Velo3D"), formerly known as JAWS Spitfire Acquisition Corporation ("JAWS Spitfire"), produces metal additive three dimensional printers ("3D Printers") which enable the production of components for space rockets, jet engines, fuel delivery systems and other high value metal parts, which it sells or leases to customers for use in their businesses. The Company also provides support services ("Support Services") for an incremental fee.

Velo3D's subsidiaries are Velo3D US, Inc., formerly known as Velo3D, Inc. ("Legacy Velo3D"), which was founded in June 2014 as a Delaware corporation headquartered in Campbell, California, Velo3D B.V., which was founded in September 2021 in the Netherlands, and Velo3D GmBH, which was founded in June 2022 in Germany. The first commercially developed 3D Printer was delivered in the fourth quarter of 2018.

Unless otherwise stated herein or unless the context otherwise requires, references in these notes to the "Company" refer to (i) Legacy Velo3D prior to the consummation of the Merger (as defined in "Explanatory Note—Certain Defined Terms"); and (ii) Velo3D and its consolidated subsidiaries following the consummation of the Merger.

### Basis of Presentation

The unaudited condensed consolidated financial statements include the accounts of the Company and its subsidiaries and have been prepared in accordance with accounting principles generally accepted in the United States of America ("U.S. GAAP") and the requirements of the U.S. Securities and Exchange Commission (the "SEC") for interim financial reporting. Intercompany balances and transactions have been eliminated in consolidation. As permitted under those rules, certain footnotes or other financial information that are normally required by U.S. GAAP can be condensed or omitted. Accordingly, these unaudited condensed consolidated financial statements should be read in conjunction with the audited consolidated financial statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2022 (the "2022 Form 10-K") and the related notes, which provide a more complete discussion of the Company's accounting policies and certain other information. The condensed consolidated balance sheet as of December 31, 2022 has been derived from the audited financial statements of the Company. These unaudited condensed consolidated financial statements have been prepared on the same basis as the Company's annual financial statements and, in the opinion of management, reflect all adjustments, consisting only of normal recurring adjustments, which are necessary for the fair statement of the Company's financial information. These interim results are not necessarily indicative of the results to be expected for the fiscal year ending December 31, 2023, or for any other interim period or for any other future year.

### Going Concern, Financial Condition and Liquidity and Capital Resources

The condensed consolidated financial statements are unaudited and have been prepared on the basis of continuity of operations, the realization of assets and satisfaction of liabilities in the ordinary course of business. Since inception, the Company has not achieved profitable operations or generated positive cash flows from operations. The Company's operating plan may change as a result of many factors currently unknown and there can be no assurance that the current operating plan will be achieved in the time frame anticipated by the Company, and while the Company expects it will need to engage in additional financings to fund its operations in the near-term, it may need to seek such additional funds sooner than planned. If adequate funds are not available to the Company on a timely basis, it may be required to delay, limit, reduce, or terminate certain commercial efforts, or pursue merger or acquisition strategies, all of which could adversely affect the holdings or the rights of the Company's stockholders. The Company has incurred losses from operations and negative cash flows from operations in every year since inception and expects this to continue for the foreseeable future. As of September 30, 2023, the Company had an accumulated deficit of \$296.6 million.

On February 6, 2023, the Company entered into a sales agreement (the "ATM Sales Agreement") with Needham & Company, LLC ("Needham"), as agent, pursuant to which the Company may offer and sell, from time to time through Needham, shares of its common stock, par value \$0.00001 per share. See Note 16. At-the-Market Offering, for further information

In addition, on August 10, 2023, the Company entered into a securities purchase agreement (the "Securities Purchase Agreement") with certain affiliated institutional investors (collectively, the "Investor") pursuant to which the Company agreed to issue and sell, in a registered public offering by the Company directly to the Investor (the "Offering"), up to \$105 million aggregate principal amount of the Company's senior secured convertible notes (the "Notes"). On August 14, 2023, the Company issued \$70 million aggregate principal amount of Notes (the "Initial Notes") to the Investor for approximately \$66 million in net proceeds, and used approximately \$22 million of the net proceeds to repay in full indebtedness outstanding under its third amended and restated loan and security agreement, as amended (the "Loan Agreement"). The Notes contain customary affirmative and negative covenants (including covenants that limit the Company's ability to incur debt, make investments, transfer assets, engage in certain transactions with affiliates and merge with other companies, in each case, other than those permitted by the Notes) and events of default. Furthermore, the Company will be required to maintain a minimum of \$30 million of unrestricted cash and cash equivalents and to maintain minimum levels of quarterly revenue through the quarter ended June 30, 2026 specified in the Notes. See Note 9. *Long-Term Debt*, for further information.

As noted above, the Notes require the Company to maintain minimum levels of quarterly revenue through the quarter ended June 30, 2026. As of November 20, 2023, the issuance date of the unaudited condensed consolidated financial statements, the Company was not in compliance with the minimum revenue covenant for the quarter ended September 30, 2023 and, as a result, the Investor has the right to declare the Notes due and payable in cash in an amount equal to the Event of Default Acceleration Amount (as defined in the Notes). In addition, the Company has experienced less revenue growth than expected due to the impact of delayed shipments during the three and nine months ended September 30, 2023. Further, due to the impact of fourth quarter customer order delays and the Company's bookings to date, the Company expects additional contraction of revenue growth in the near term.

The Company has been negotiating a proposed amendment to the Notes with the Investor, which the Company anticipates completing in November 2023; however, the Company may not be able to obtain a waiver or an amendment on favorable terms or at all. If a waiver or amendment is not agreed to by the Investor, the Company does not have sufficient capital to satisfy the outstanding principal and interest due, and the Investor may foreclose upon the Company's assets securing the Notes. Accordingly, as the Company was not in compliance with the minimum quarterly revenue covenant and the Notes are now callable by the Investor, the Company believes that there is substantial doubt about its ability to continue as a going concern for the twelve-month period following the issuance date of the unaudited condensed consolidated financial statements.

The Company is undertaking expense reduction and cash savings initiatives as part of a company wide restructuring and strategic realignment plan to help conserve working capital. The expense reduction and cash saving initiatives include streamlining facilities, managing working capital, reducing capital expenditures, and reducing overall selling, general and administrative expenses.

Even if the Company is able to agree an amendment with the Investor, the Company expects that it will need to engage in additional financings to fund its operations in the near-term as well as to respond to business challenges and opportunities. Accordingly, subject to the Company's compliance with the covenants in the Notes, the Company expects it will need to engage in additional equity or debt financings to secure additional funds.

### Note 2. Summary of Significant Accounting Policies

For a detailed discussion about the Company's significant accounting policies and for further information on significant accounting updates adopted in the prior year, see Note 2, *Summary of Significant Accounting Policies*, to the audited consolidated financial statements in the 2022 Form 10-K. During the nine months ended September 30, 2023, there were no significant updates to the Company's significant accounting policies other than as described below.

### At-the-Market Offering

On February 6, 2023, the Company entered into the ATM Sales Agreement with Needham, as agent, pursuant to which the Company may offer and sell, from time to time through Needham, shares of its common stock pursuant to its universal shelf registration statement (the "Shelf Registration Statement"), which the Company filed with the SEC on November 14, 2022. For the nine-months ended September 30, 2023, the Company has sold \$18.4 million of shares, net of issuance costs of \$1.2 million. See Note 16. Atthe-Market Offering, for further information.

### Fair Value of Debt Derivatives

Debt derivative liabilities, related to the Notes, are recorded at fair value on a recurring basis in the condensed consolidated balance sheets, and are categorized in accordance with the fair value hierarchy based upon the level of judgment associated with the inputs used to measure their fair values. Changes to the fair value of the debt derivative liabilities occurring from subsequent remeasurement are recorded as gains or losses in the condensed consolidated statements of operations and comprehensive income (loss). See Note 10. Equity Instruments, for further information.

### Convertible Notes

As of September 30, 2023, we carried the Notes at the initially allocated liability value less unamortized debt discount and issuance costs on our condensed consolidated balance sheet. The unamortized debt discount and issuance costs accrete to the carrying amount of the Notes over the term of the agreement using the effective interest rate method and are included in "Interest expense" in the condensed consolidated statements of operations and comprehensive income (loss). See Note 9. *Long-Term Debt*, for further information

### Recently Issued Accounting Pronouncements

In June 2016, the FASB issued ASU 2016-13, "Financial Instruments — Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments ("Topic 326")", and has since released various amendments including ASU No. 2019-04. The guidance modifies the measurement of expected credit losses on certain financial instruments. This guidance is effective for the Company for the fiscal year beginning after December 15, 2022, and interim periods within those fiscal years. Early adoption is permitted. The Company adopted the new guidance in the first quarter of 2023. The effect on the consolidated financial statements and related disclosures was not material.

### Note 3. Basic and Diluted Net Loss per Share

The following table sets forth the computation of the Company's basic and diluted net loss per share attributable to common stockholders:

		Three Mor Septen				Nine Mon Septem		
		2023		2022		2023		2022
	(In t	housands, except sl	hare	and per share data)	(In	thousands, except sh	are	and per share data)
Numerator:								
Net loss	\$	(17,396)	\$	(75,195)	\$	(76,795)	\$	(12,587)
Denominator:	-							
Weighted average shares outstanding-basic and diluted		197,833,109		185,560,177		193,816,804		184,454,371
Net loss per share-basic and diluted	\$	(0.09)	\$	(0.41)	\$	(0.40)	\$	(0.07)

The following potentially dilutive shares of common stock equivalents "on an as-converted basis" were excluded from the computation of diluted net loss per share for the periods presented because including them would have had an antidilutive effect:

	Three Months I September 3		Nine Months Ended September 30,			
	2023	2022	2023	2022		
Common stock options	15,136,666	17,731,063	15,136,666	17,731,063		
Common stock warrants	13,145,000	13,145,000	13,145,000	13,145,000		
Restricted stock units	11,491,036	6,198,472	11,491,036	6,198,472		
Debt derivatives	45,132,171	_	45,132,171	_		
Total potentially dilutive common share equivalents	84,904,873	37,074,535	84,904,873	37,074,535		

Total potentially dilutive common share equivalents for the three and nine months ended September 30, 2023 and 2022 exclude£1,265,936 and 21,758,148, respectively, shares related to the earnout liability as these shares are contingently issuable upon meeting certain triggering events.

### Note 4. Fair Value Measurements

The Company's assets and liabilities that were measured at fair value on a recurring basis were as follows:

	Fair Value Measured as of September 30, 2023								
	Level 1			Level 2		Level 3		Total	
				(In tho	usand	s)			
Assets									
Money market funds (i)	\$	52,174	\$	_	\$	_	\$	52,174	
U.S. Treasury securities (ii)		6,952		_		_		6,952	
Corporate bonds (ii)				6,516		<u> </u>		6,516	
Total financial assets	\$	59,126	\$	6,516	\$		\$	65,642	
Liabilities									
Common stock warrant liabilities (Public) (iii)	\$	1,850	\$	_	\$	_	\$	1,850	
Common stock warrant liabilities (Private Placement) (iii)		_		_		934		934	
Common stock warrant liabilities (2022 Private Warrant) (iii)		_		_		99		99	
Contingent earnout liabilities		_		_		14,414		14,414	
Debt derivatives						17,538		17,538	
Total financial liabilities	\$	1,850	\$	_	\$	32,985	\$	34,835	

	Fair Value Measured as of December 31, 2022							
	Level 1			Level 2		Level 3		Total
				(In tho	usand	s)		
Assets								
Money market funds (i)	\$	31,728	\$	_	\$	_	\$	31,728
U.S. Treasury securities (ii)		24,701		_		_		24,701
Corporate bonds (ii)		_		23,513		_		23,513
Total financial assets	\$	56,429	\$	23,513	\$	_	\$	79,942
Liabilities								
Common stock warrant liabilities (Public) (iii)	\$	1,748	\$	_	\$	_	\$	1,748
Common stock warrant liabilities (Private Placement) (iii)		_		_		888		888
Common stock warrant liabilities (2022 Private Warrant) (iii)		_		_		109		109
Contingent earnout liabilities		_		_		17,414		17,414
Total financial liabilities	\$	1,748	\$	_	\$	18,411	\$	20,159

- (i) Included in cash and cash equivalents on the condensed consolidated balance sheets.
- (ii) Included in short-term investments on the condensed consolidated balance sheets.
- (iii) Included in warrant liabilities on the condensed consolidated balance sheets.

For more information regarding the Public Warrants, Private Placement Warrants, the 2022 Private Warrant, Contingent earnout liabilities, and Debt derivative, see Note 10, Equity Instruments.

The aggregate fair value of the Company's money market funds approximated amortized cost and, as such, there were no unrealized gains or losses on money market funds as of September 30, 2023 and December 31, 2022. Realized gains and losses, net of tax, were not material for any of the periods presented.

The following table presents a summary of the changes in the fair value of the Company's Level 3 financial instruments:

	Decem	per 31, 2022	Change in fair value	June 30, 2023	Change in fair value	Sep	tember 30, 2023
				(In thousands)			
Private placement warrant liabilities	\$	888	\$ 600	\$ 1,488	\$ (554)	\$	934
2022 Private Warrant		109	31	140	(41)		99
Contingent earnout liabilities		17,414	7,810	25,224	(10,810)		14,414
Debt derivatives		_	_	_	17,538		17,538
	\$	18,411	\$ 8,441	\$ 26,852	\$ 6,133	\$	32,985
	Deceml	per 31, 2021	Change in fair value	June 30, 2022	Change in fair value	Sep	tember 30, 2022
	Deceml	per 31, 2021	Change in fair value	June 30, 2022 (In thousands)	Change in fair value	Sep	tember 30, 2022
	Decemb	per 31, 2021	Change in fair value		Change in fair value	Sep	tember 30, 2022
Private placement warrant liabilities	December 1	7,387	9		9		3,605
Private placement warrant liabilities 2022 Private Warrant			9	(In thousands)	9		,
*			9	(In thousands)	\$ 2,225		3,605
2022 Private Warrant		7,387	\$ (6,007)	(In thousands)  1,380	\$ 2,225 245		3,605 245

The fair value of the Private Placement Warrant liability, the 2022 Private Warrant, Contingent earnout liability, and Debt derivative are based on significant unobservable inputs, which represent Level 3 measurements within the fair value hierarchy.

In determining the fair value of the Private Placement Warrant liability the Company used the Monte Carlo Simulation Model that assumes optimal exercise of the Company's redemption option at the earliest possible date (see Note 10, Equity Instruments).

In determining the fair value of the 2022 Private Warrant, the Company used the Black-Scholes option pricing model to estimate the fair value using unobservable inputs including the expected term, expected volatility, risk-free interest rate and dividend yield (see Note 10, *Equity Instruments*).

In determining the fair value of the Contingent earnout liability, the Company used the Monte Carlo simulation valuation model using a distribution of potential outcomes on a weekly basis over the applicable earnout period using the most reliable information available (see Note 10, *Equity Instruments*).

In determining the fair value of the Debt derivatives, the Company used the Monte Carlo simulation valuation model using a distribution of potential outcomes on a weekly basis over the applicable Debt derivative period using the most reliable information available (see Note 10, *Equity Instruments*).

### Note 5. Investments

### Available-for-sale Investments

The following table summarizes the Company's available-for-sale ("AFS") investments. These are classified as "Short-term investments" on the condensed consolidated balance sheets.

	<b>September 30, 2023</b>							
			Gro	oss Unrealized				
	Amortized Cost	Gross Unreali	zed Gain	Loss	Fair Value			
			(In thousands)					
U.S. Treasury securities	\$ 7,03	9 \$	— \$	(87)	\$ 6,952			
Corporate bonds	6,68	4	_	(168)	6,516			
Total available-for-sale investments	\$ 13,72	3 \$	— \$	(255)	\$ 13,468			

		December	31, 2022					
	Gross Unrealized							
	Amortized Cost	Gross Unrealized Gain	Loss	Fair Value				
		(In tho	usands)					
U.S. Treasury securities	\$ 25,124	\$ —	\$ (423)	\$ 24,701				
Corporate bonds	23,927	_	(414)	23,513				
Total available-for-sale investments	\$ 49,051	\$	\$ (837)	\$ 48,214				

The following table presents the breakdown of the AFS investments in an unrealized loss position as of September 30, 2023 and December 31, 2022, respectively.

	<b>September 30, 2023</b>				<b>December 31, 2022</b>			
	Fair Value		oss Unrealized Loss		Fair Value	Gro	ss Unrealized Loss	
			(In tho	usar	nds)			
U.S. Treasury securities								
Less than 12 months	\$ 6,952	\$	(87)	\$	16,702	\$	(365)	
12 months or longer	_		_		7,999		(58)	
Total	\$ 6,952	\$	(87)	\$	24,701	\$	(423)	
Corporate bonds								
Less than 12 months	\$ 6,516	\$	(168)	\$	18,951	\$	(387)	
12 months or longer	_		_		1,478		(27)	
Total	\$ 6,516	\$	(168)	\$	20,429	\$	(414)	

The Company does not believe these AFS investments to be other-than-temporarily impaired as of September 30, 2023 and December 31, 2022.

There were no material realized gains or losses on AFS investments during the three and nine months ended September 30, 2023 and September 30, 2022.

All remaining contractual maturities of AFS investments held at September 30, 2023 are as follows:

	Less than	months		Greater than 12 months				
	Fair value Gross unrealized losses				Fair value		Gross unrealized losses	
			(In the	ousa	nds)			
Corporate bonds	\$ 6,516	\$	(168)	\$	-	_	\$	
U.S. Treasury securities	6,952		(87)		-	_	_	
Total	\$ 13,468	\$	(255)	\$	-	Ξ	\$	

Actual maturities may differ from the contractual maturities because the Company may sell these investments prior to their contractual maturities.

### **Note 6. Balance Sheet Components**

### Accounts Receivable, Net

Accounts receivable, net consisted of the following:

		September 30, 2023		December 31, 2022
		ls)		
Trade receivables	\$	12,697	\$	9,639
Less: Allowances for credit losses		(100)		(454)
Total	\$	12,597	\$	9,185

### Inventories

Inventories consisted of the following:

	September 30,	December 31,
	2023	2022
	(In	thousands)
Raw materials	\$ 69,7	01 \$ 58,585
Work-in-progress	10,3	00 12,617
Finished goods	1,1	58 —
Total	\$ 81,1	59 \$ 71,202

### **Prepaid Expenses and Other Current Assets**

Prepaid expenses and other current assets consisted of the following:

	Sep	tember 30,	De	cember 31,
		2023		2022
		(In tho	usands)	
Prepaid insurance and other	\$	2,030	\$	3,316
Vendor prepayments		485		2,217
Total	\$	2,515	\$	5,533

### Property and Equipment, Net

Property and equipment, net consisted of the following:

	Sep	tember 30,	December 31,
		2023	2022
	<u></u>	(In thousa	ands)
Computers and software	\$	2,490 \$	2,222
Lab equipment and other equipment		7,995	7,379
Furniture and fixtures		246	181
Leasehold improvements		16,060	16,273
Total property, plant and equipment		26,791	26,055
Less accumulated depreciation and amortization		(9,361)	(6,243)
Property, plant and equipment, net	\$	17,430 \$	19,812

Depreciation expense for the three months ended September 30, 2023 and 2022 was \$1.5 million and \$0.9 million, respectively. Depreciation expense for the nine months ended September 30, 2023 and 2022 was \$3.8 million and \$2.1 million, respectively.

The manufacturing facility operating lease at Campbell (McGlincy) was terminated on March 31, 2023, and is no longer in use. There were no significant asset retirement obligations. The Company wrote-off \$0.6 million in fully-depreciated leasehold improvements related to the Campbell (McGlincy) lease during the nine months ended September 30, 2023.

### Other Assets

Other assets consisted of the following:

	Se	ptember 30,	Decemb	ber 31,
		2023	202	22
		(In thou	isands)	
Right of use assets	\$	11,600	\$	13,545
Net investments in sales-type lease		5,747		6,554
Non-current prepaid expenses and other assets		4,435		3,211
Total Other assets	\$	21,782	\$	23,310

### **Accrued Expenses and Other Current Liabilities**

Accrued expenses and other current liabilities consisted of the following:

	Se	September 30,		cember 31,
		2023		2022
		(In tho	usands)	
Accrued expenses	\$	5,727	\$	8,602
Accrued salaries and benefits		3,171		4,830
Lease liability – current portion		2,317		2,445
Total Accrued expenses and other current liabilities	\$	11,215	\$	15,877

### Other Noncurrent Liabilities

Other noncurrent liabilities consisted of the following:

	Sep	September 30, 2023		ecember 31, 2022
		(In tho	usands)	
Lease liabilities – noncurrent portion	\$	10,550	\$	12,206
Other noncurrent liabilities		255		428
Total other noncurrent liabilities	\$	10,805	\$	12,634

Please refer to Note 10, Equity Instruments, for further details of the contingent earnout liability and warrant liabilities.

### Note 7. Equipment on Lease, Net

The equipment leased to customers had a cost basis of \$7.4 million and accumulated depreciation of \$0.4 million as of September 30, 2023. The equipment leased to customers had a cost basis of \$10.6 million and accumulated depreciation of \$1.5 million as of December 31, 2022.

The total depreciation expense was \$0.1 million and \$0.3 million included in cost of revenue for the three months ended September 30, 2023 and 2022, respectively. The total depreciation expense was \$0.7 million and \$1.3 million included in cost of revenue for the nine months ended September 30, 2023 and 2022, respectively.

Lease payments from customers consisted of the following:

	Three Months Er	Nine Months Ended September 30,			
	2023	2023 2022		23	2022
	(In the		(In thousands)		
Equipment on lease payments	\$ 531	\$ 923	\$	1,141	\$ 2,637
Equipment on lease variable payments	_	260		_	405
Total lease payments	\$ 531	\$ 1,183	\$	1,141	\$ 3,042

The Company entered into debt secured by certain leased equipment to customers. See Note 9.Long-term Debt, for a description of these financing arrangements.

### Note 8. Leases

The Company leases its office and manufacturing facilities under four non-cancellable operating leases, including options to extend, which expire in 2024 to 2032. The agreements include a provision for renewal at the then prevailing market rate for terms specified in each lease.

As noted above in Note 6, *Balance Sheet Components*, the manufacturing facility operating lease at Campbell (McGlincy) was terminated on March 31, 2023, and is no longer in use. The Company's right-of-use assets and lease liabilities related to McGlincy were amortized in full over the life of the lease.

Total right-of-use ("ROU") assets and lease liabilities are as follows:

	Se	September 30, 2023		ecember 31, 2022
		(In tho	usands)	
Right-of-use assets:				
Net book value (Other assets)	\$	11,600	\$	13,545
Operating lease liabilities:	· · · · · · · · · · · · · · · · · · ·			
Current (Accrued expense and other current liabilities)	\$	2,201	\$	2,410
Noncurrent (Other noncurrent liabilities)		10,399		12,201
		12,600		14,611
Financing lease liabilities:				
Current (Accrued expense and other current liabilities)	\$	43	\$	35
Noncurrent (Other noncurrent liabilities)		56		5
	\$	99	\$	40
Total lease liabilities	\$	12,699	\$	14,651

There were no impairments recorded related to these assets as of September 30, 2023 and December 31, 2022.

Information about lease-related balances were as follows:

	Three Months Ended September 30,			Nine Months Ended September 30,				
		2023		2022		2023		2022
	(Ir	n thousands, except	years	and percentages)		In thousands, except y	ears	and percentages)
Operating lease expense	\$	746	\$	721	\$	2,256	\$	2,156
Financing lease expense		18		9		36		27
Short-term lease expense		78		95		250		262
Total lease expense	\$	842	\$	825	\$	2,542	\$	2,445
Cash paid for leases	\$	710	\$	689	\$	2,116	\$	1,653
Weighted – average remaining lease term – operating leases (years)		3.4		4.3		3.4		4.3
Weighted – average discount rate – operating leases		8.8%		8.6%		8.8%		8.6%

Maturity of operating lease liabilities as of September 30, 2023 are as follows:

	(	(In thousands)
Remainder of 2023	\$	711
2024		2,860
2025		2,397
2026		2,402
2027		2,400
Thereafter	<u></u> .	11,270
Total operating lease payments	\$	22,040
Less portion representing imputed interest		(9,440)
Total operating lease liabilities	\$	12,600
Less current portion		2,201
Long-term portion	\$	10,399

### Note 9. Long-Term Debt

Long-term debt consisted of the following:

	 September 30, 2023	December 31, 2022
	(In thous	ands)
Revolving credit line	\$ _ \$	3,000
Equipment loan	_	5,356
Convertible notes	56,110	_
Deferred financing costs	(3,899)	(159)
Total	\$ 52,211	8,197
Debt – current portion	52,211	2,775
Long-term debt – less current portion	\$ 	5,422

As of September 30, 2023, the Company's debt arrangements comprised the Initial Notes (as defined below) issued to the Investor pursuant to the Securities Purchase Agreement and the Company's indenture dated as of August 14, 2023 (the "Base Indenture") between the Company and U.S. Bank Trust Company, National Association, a national banking association, as trustee (the "Trustee"), as supplemented by a first supplemental indenture dated as of August 14, 2023, between the Company and the Trustee (the "Supplemental Indenture" and, together with the Base Indenture, the "Indenture"). See *Senior Secured Convertible Notes* section below for a full description of these debt arrangements. The Notes contain customary affirmative and negative covenants (including covenants that limit the Company's ability to incur debt, make investments, transfer assets, engage in certain transactions with affiliates and merge with other companies, in each case, other than those permitted by the Notes) and events of default. Furthermore, the Company will be required to maintain a minimum of \$30 million of unrestricted cash and cash equivalents and to maintain minimum levels of quarterly revenue through the quarter ended June 30, 2026 specified in the Notes. As described below, as of the issuance date of the unaudited condensed consolidated financial statements, the Company was not in compliance with the minimum revenue covenant for the quarter ended September 30, 2023.

On August 14, 2023, the Company used approximately \$22 million of the net proceeds from the offering of the Initial Notes to repay all outstanding obligations under the Loan Agreement originally entered into with Silicon Valley Bank ("SVB"), which included the Company's prior revolving credit line and equipment loans. With the payoff of the debt, the Loan Agreement was terminated and is no longer available to the Company. The Company recorded a loss on debt extinguishment of \$0.3 million related to the termination of the Loan Agreement.

The Company amortizes deferred financing costs over the life of the applicable indebtedness. As of September 30, 2023 and December 31, 2022, the remaining unamortized balance of deferred financing costs was \$3.9 million and \$0.2 million, respectively, and was included in Debt — current portion on the balance sheets.

Senior Secured Convertible Notes — On August 10, 2023, the Company entered into the Securities Purchase Agreement with the Investor pursuant to which the Company agreed to issue and sell in an offering up to \$105 million aggregate principal amount of Notes.

On August 14, 2023 (the "Initial Closing Date"), the Company issued \$70 million aggregate principal amount of the Notes (the "Initial Notes") to the Investor. In addition, the Company has granted the Investor the right to purchase up to an additional \$35 million aggregate principal amount of the Notes (the "Additional Notes") so long as the notice to exercise such option is provided no later than the first anniversary of the Initial Closing Date.

The Notes bear interest at 6.00% per annum, payable quarterly in cash on January 1, April 1, July 1 and October 1 of each year, commencing on October 1, 2023, and will mature on August 1, 2026. Beginning on January 1, 2024, the Investor may elect to require the Company to repay principal on the Notes quarterly pursuant to the terms of the Notes at a repayment price (the "Repayment Price") equal to 115% of the Notes principal balance repaid plus accrued interest. The repayments are calculated at rate of 12.5% of 115% of the principal balance and will reduce the principal balance of the Notes by the amount repaid divided by a rate of 1.15. The end of term maturity balance is the principal balance of the Note multiplied by 115%. The effective interest rate was 39.9% for the three and nine months ended September 30, 2023.

The Notes include terms that provide the Investor seniority over other unsecured obligations in any settlement negotiations in the event of liquidation. Additionally, the Notes contain redemption features in the event of default or a fundamental change in control that would make the Notes immediately callable at a predetermined rate as described in the Notes. The redemption features are settled in cash. As of September 30, 2023, the Company has included the effect of an event of default in the valuation of the Notes and related debt derivatives. As of September 30, 2023, the Company has not included the effect of a fundamental change in control in the valuation of the Notes and related debt derivatives, as the Company believes the likelihood of this occurring is remote. The Company will continue to monitor the likelihood of these events in future reporting periods.

The Notes are convertible based on a conversion rate of 644.7453 shares of the Company's common stock per \$1,000 principal amount of Notes as of September 30, 2023 (equivalent to a conversion price of approximately \$1.55 per share of the Company's common stock).

The Company incurred deferred financing costs of \$4.1 million related to the Initial Closing, which were capitalized upon issuance and are being accreted over the term of the indebtedness using the effective interest rate method and are included in "Interest expense" in the accompanying Condensed Consolidated Statements of Operations and Comprehensive Income (Loss).

Additionally, the Company is accreting discounts of \$28.6 million and capitalizing to the carrying value of the Notes over the term of the indebtedness using the effective interest rate method. The \$28.6 million discount includes \$13.9 million related to its debt derivatives, \$10.5 million related to the Repayment Price, \$0.1 million related to lender fees paid, and \$4.1 million in third-party fees. For further information on the valuation of the debt derivative discount, see Note 10 Equity Instruments below.

As of the issuance date of the unaudited condensed consolidated financial statements, the Company was not in compliance with the Notes' minimum revenue covenant for the quarter ended September 30, 2023 and, as a result, the Investor has the right to declare the Notes due and payable in cash in an amount equal to the Event of Default Acceleration Amount (as defined in the Notes). The Company has been negotiating a proposed amendment to the Notes with the Investor, which the Company anticipates completing in November 2023; however, the Company may not be able to obtain a waiver or an amendment on favorable terms or at all. If a waiver or amendment is not agreed to by the Investor, the Company does not have sufficient capital to satisfy the outstanding principal and interest due.

**Revolving Credit Line**—For the three months ended September 30, 2023, the Company did not draw from the revolving credit facility and for the nine months ended September 30, 2023, the Company drew \$14.0 million on the revolving credit facility. As of September 30, 2023, the Company had no outstanding balance or remaining revolving credit line availability due to the repayment and termination of the Loan Agreement in connection with the issuance of the Initial Notes.

Interest on the balance of the revolving credit line was payable monthly at an annual rate of the greater of (1) the Wall Street Journal Prime Rate plus 2.5% and (2) 5.0% when the Company's Adjusted Quick Ratio ("AQR") was at least 1.50 to 1.0, and at an annual rate of the greater of (1) the Wall Street Journal Prime Rate plus 2.75% and (2) 5.50% when the Company did not maintain such AQR. The effective interest rate was 12.0% and 5.8% for the three months ended September 30, 2023 and 2022, respectively. The effective interest rate was 15.8% and 5.2% for the nine months ended September 30, 2023 and 2022, respectively. Interest expense for the three months ended September 30, 2023 includes an additional \$0.3 million for final payments made during the extinguishment of the SVB loans.

*Equipment Loan* — As of September 30, 2023, the Company had no outstanding balance or remaining equipment loan availability due to the repayment and termination of the Loan Agreement in connection with the issuance of the Initial Notes.

The effective interest rate was 18.1% and 5.5% for the three months ended September 30, 2023 and 2022, respectively. The effective interest rate was 12.6% and 4.1% for the nine months ended September 30, 2023 and 2022, respectively. Interest expense for the three months ended September 30, 2023 includes an additional \$0.1 million for final payments made under the Loan Agreement.

The future minimum aggregate payments for the above borrowings are equal to the quarterly payments made using the Repayment Price, beginning are as follows as of September 30, 2023:

	_	(In thousands)	
2023	\$		_
2024			40,250
2025			40,250
	\$		80,500

#### **Note 10. Equity Instruments**

#### Common stock

The holders of common stock are entitled to one vote for each share held of record on all matters submitted to a vote of the stockholders but are not entitled to cumulative voting rights, are entitled to receive ratably such dividends as may be declared by the Company's Board of Directors out of funds legally available therefor subject to preferences that may be applicable to any shares of redeemable convertible preferred stock currently outstanding or issued in the future, are entitled to share ratably in all assets remaining after payment of liabilities and the liquidation preference of any then outstanding redeemable convertible preferred stock in the event of the Company's liquidation, dissolution, or winding up, have no preemptive rights and no right to convert their common stock into any other securities, and have no redemption or sinking fund provisions applicable to the common stock.

### Common Stock Reserved for Future Issuance

Shares of common stock reserved for future issuance on an "as if converted" basis were as follows:

	September 30, 2023	December 31, 2022
	(share dat	ia)
Common stock warrants	13,145,000	13,145,000
Common shares related to debt derivative	45,132,171	_
Shares available for future grant under 2021 Equity Incentive Plan	30,298,210	20,861,294
Reserved for At-the-Market offering	11,876,135	_
Reserved for employee stock purchase plan	7,371,214	5,495,601
Total shares of common stock reserved	107,822,730	39,501,895

In February 2023, the Company entered into the ATM Sales Agreement with Needham, as agent, pursuant to which the Company may offer and sell, from time to time through Needham, up to \$40.0 million shares of its common stock pursuant to the Shelf Registration Statement, and in connection therewith, the Company reserved20,000,000 shares of common stock for issuance under the ATM Sales Agreement. In March 2023, pursuant to the evergreen provisions of the Company's 2021 Equity Incentive Plan (the "2021 EIP"), the Company registered an additional 9,378,068 shares of common stock for issuance under the 2021 EIP and1,875,613 shares of common stock for issuance under the 2021 ESPP. In August 2023, the Company issued the Initial Notes, which are convertible into 644.7453 shares of common stock per \$1000 in principal amount for a total of up to 45,132,171 shares of common stock. However, the Notes provide that the number of shares of common stock issuable upon conversion or otherwise pursuant to the Notes may not exceed 39,349,491 shares in the aggregate until the Company's stockholders approve the issuance of such shares in accordance with NYSE listing rules.

The shares available for future grant under the 2021 EIP are net of any un-exercised stock options (vested and unvested) and unvested restricted stock units ("RSUs") outstanding that may convert to common stock in the future upon exercise or vesting as of September 30, 2023 and December 31, 2022.

### Common Stock Warrant Liabilities

Warrants for common stock of 13,145,000 were exercisable 1-to-1 as of September 30, 2023 and December 31, 2022. The warrants on common stock are liability classified and recorded at fair value on the issue date with periodic remeasurement. Warrants for shares of common stock consist of 8,625,000 publicly-traded warrants (the "Public Warrants"), 4,450,000 private placement warrants (the "Private Placement Warrants") and a warrant to purchase 70,000 shares of common stock (the "2022 Private Warrant"), as summarized in the following table:

	December 31, 2022 and September 30, 2023						
	Issue Date	Expiration Date	Number of Warrants	Exercise Price per warrant	Fair Value on Issue Date per warrant		
Private placement warrants - Common Stock	12/02/2020	09/29/2026	4,450,000	\$11.50	\$2.00		
2022 Private Warrant – Common Stock	07/25/2022	07/24/2034	70,000	\$2.56	\$2.43		
Public warrants – Common Stock	12/02/2020	09/29/2026	8,625,000	\$11.50	\$3.30		
			13,145,000				

### Warrant Liabilities - Fair Value

The issuance of the Private Placement Warrant and Public Warrant liabilities were accounted for upon the reverse recapitalization. See Note 3, Reverse Recapitalization, in the audited consolidated financial statements included in the 2022 Form 10-K. The 2022 Private Warrant was issued in connection with the Company's entry into the joinder and fourth loan modification with SVB. See Note 10, Long-Term Debt, in the consolidated financial statements included in the 2022 Form 10-K. The liability for warrants on common stock carried at fair value was as follows:

	Fair Value on December 31, 2022		Gain on fair value of warrants		Fair Value on June 30, 2023		s on fair value of warrants	Fair Value on September 30, 2023
					(In thousands)			
Private placement warrants - Common Stock	\$ 88	8 5	\$ 600	\$	1,488	\$	(554)	\$ 934
2022 Private Warrant - Common Stock	10	9	31		140		(41)	99
Public warrants - Common Stock	1,74	8	1,094		2,842		(992)	1,850
	\$ 2,74	5	\$ 1,725	\$	4,470	\$	(1,587)	\$ 2,883

	Fair Value on I December 31, 2021		s on fair value of warrants	Fair Value on June 30, 2022		Gain on fair value of warrants			Fair Value on September 30, 2022
					(In thousands)				
Private placement warrants - Common Stock	\$ 7,387	\$	(6,007)	\$	1,380	\$	2,225	\$	3,605
2022 Private Warrant - Common Stock	_		_		_		245		245
Public warrants - Common Stock	14,318		(11,645)		2,673		4,313		6,986
	\$ 21,705	\$	(17,652)	\$	4,053	\$	6,783	\$	10,836

The liabilities associated with the Private Placement Warrants and 2022 Private Warrant were subject to remeasurement at each balance sheet date using the Level 3 fair value inputs and the Public Warrants were subject to remeasurement at each balance sheet date using Level 1 fair value inputs for the three and nine months ended September 30, 2023 and September 30, 2022.

Each Private Placement Warrant is exercisable to purchase one share of common stock at a price of \$1.50 per share. Subject to certain exceptions, the Private Placement Warrants have terms and provisions that are identical to those of the Public Warrants. The 2022 Private Warrant is exercisable to purchase one share of common stock at a price of \$2.56 per share and allows cashless exercise in whole or part. The Public Warrants may only be exercised for a whole number of shares. The Public Warrants became exercisable on December 7, 2021.

Private Placement Warrant - Fair Value Assumptions

The fair value assumptions used in the Monte Carlo simulation model for the recurring valuation of the private placement common stock warrant liability were as follows:

	As of Sep	tember 30, 2023	As of December 31, 2022		
Current stock price	\$	1.56	\$ 1.79		
Expected volatility		84.0 %	68.0 %		
Risk-free interest rate		4.8 %	4.1 %		
Dividend rate		— %	— %		
Expected Term (years)		3.00	3.75		

Expected volatility: The volatility is determined iteratively, such that the concluded value of the Public Warrant is equal to the traded price.

Risk-free interest rate: The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of grant for zero-coupon U.S. Treasury notes with maturities corresponding to the expected term of the common stock warrants.

Expected dividend yield: The expected dividend rate is zero as the Company currently has no history or expectation of declaring dividends on its common stock.

Expected term: The expected term represents the period that the warrants are expected to be outstanding and is determined using the simplified method, which deems the term to be the average of the time to vesting and the contractual life of the common stock warrants.

### 2022 Private Warrant - Fair Value Assumptions

The fair value assumptions used in the Black-Scholes simulation model for the recurring valuation of the 2022 Private Warrant liability were as follows:

	As of So	eptember 30, 2023	As of	<b>December 31, 2022</b>
Current stock price	\$	1.56	\$	1.79
Expected volatility		101.9 %		86.9 %
Risk-free interest rate		4.6 %		3.9 %
Dividend rate		— %		— %
Expected Term (years)		10.82		11.57

Expected volatility: The expected volatility was derived from the implied volatility of the Company's publicly traded common stock as of September 30, 2023.

Risk-free interest rate: The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of grant for zero-coupon U.S. Treasury notes with maturities corresponding to the expected term of the common stock warrants.

Expected dividend yield: The expected dividend rate is zero as the Company currently has no history or expectation of declaring dividends on its common stock.

Expected term: The expected term represents the period that the warrant is expected to be outstanding and is determined using the simplified method, which deems the term to be the average of the time to vesting and the contractual life of the warrant.

### **Contingent Earnout Liabilities**

The contingent earnout liability is for Earnout Shares (as defined below) for pre-closing Legacy Velo3D equity holders (as defined in the Business Combination Agreement as holders of Legacy Velo3D shares, Legacy Velo3D warrants, Legacy Velo3D convertible notes and Legacy Velo3D options immediately prior to the closing date) ("Eligible Legacy Velo3D Equityholders"). During the time period between September 29, 2021 (the "Closing Date") and the five-year anniversary of the Closing Date, Eligible Legacy Velo3D Equityholders may receive up to 21,758,148 shares of additional common stock (the "Earnout Shares"), which is based on two tranches or 10,879,074 per tranche. The Earnout Shares issuable to holders of employee stock options are accounted as stock-based compensation expense as they are subject to forfeiture based on the satisfaction of certain employment conditions. See Note 11, Equity Incentive Plans & Stock Based Compensation, for further discussion.

The rollforward for the contingent earnout liabilities for the three months ended September 30, 2023 and 2022, was as follows:

	Fair Value on December 31, 2022	Gain on fair value of warrants	Fair Value on June 30, 2023	Loss on fair value of warrants	Fair Value on September 30, 2023
			(In thousands)		
Contingent Earnout Liabilities	\$ 17,414	\$ 7,810	\$ 25,224	\$ (10,810)	\$ 14,414
	\$ 17,414	\$ 7,810	\$ 25,224	\$ (10,810)	\$ 14,414

	Fair Value on ecember 31, 2021		Loss on fair value of warrants		Fair Value on June 30, 2022		in on fair value of warrants	Fair Value on September 30, 2022		
				(	(In thousands)					
Contingent Earnout Liabilities	\$ 111,487	\$	(98,994)	\$	12,493	\$	40,884	\$	53,377	
	\$ 111,487	\$	(98,994)	\$	12,493	\$	40,884	\$	53,377	

Fair Value Assumptions - Contingent Earnout Liabilities

Assumptions used in the fair value of the contingent earnout liabilities are described below.

	As of September 30, 2023	As of December 31, 2022
Current stock price	\$1.56	\$1.79
Expected volatility	84.0%	89.9%
Risk-free interest rate	4.8%	4.1%
Dividend yield	<del></del> %	—%
Expected Term (years)	3.00	3.75

Expected volatility: The expected volatility was derived from the implied volatility of the Company's publicly traded common stock as of September 30, 2023.

Risk-free interest rate: The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of grant for zero-coupon U.S. Treasury notes with maturities corresponding to the expected term of the Earnout Shares.

Expected dividend yield: The expected dividend rate is zero as the Company currently has no history or expectation of declaring dividends on its common stock.

Expected term: The expected term represents the period that the Company's stock-based awards are expected to be outstanding and is determined using the simplified method, which deems the term to be the average of the time to vesting and the contractual life of the Earnout Shares.

#### **Debt Derivatives**

In August 2023, the Company issued the Initial Notes, which contain embedded debt derivative liabilities that convert principal into shares of the Company's common stock (as described below) in order to receive other favorable financing terms. The embedded debt derivative liabilities are bifurcated from the Notes and presented separately in the condensed consolidated balance sheets. The debt derivative liabilities are recorded at fair value on the issuance date with periodic remeasurement. Changes to the fair value of the debt derivative liabilities are recorded as gains or losses in the condensed consolidated statements of operations and comprehensive income (loss).

The Initial Notes are convertible into 644.7453 shares of common stock per \$1,000 in principal for a total of up to 45,132,171 shares of common stock as of September 30, 2023. The debt derivative liability is related to Convertible Shares (as defined below) for the Initial Notes. See Note 9. Long-Term Debt, for further information. As of August 14, 2023, the Notes entitled the Investor to receive 475.1722 shares of the Company's common stock per \$1,000 principal amount of Notes (equivalent to an initial conversion price of approximately \$2.10 per share of the Company's common stock), or 33,262,054 shares of additional common stock (the "Convertible Shares") in respect of the Initial Notes. On August 17, 2023, the conversion rate applicable to the Notes was adjusted to 644.7453 shares of the Company's common stock per \$1,000 principal amount of Notes (equivalent to a conversion price of approximately \$1.55 per share of the Company's common stock), or 45,132,171 of Convertible Shares in respect of the Initial Notes. However, the Notes provide that the number of Convertible Shares issuable upon conversion or otherwise pursuant to the Notes may not exceed 39,349,491 shares in the aggregate until the Company's stockholders approve the issuance of such shares in accordance with NYSE listing rules. The derivative can be settled in cash, shares, or a combination of cash and shares.

In addition to standard antidilution provisions, the Notes include terms that could change the timing of the conversion. The Notes include a "Forced Conversion Trigger" that allows the Company to elect to convert the entire principal amount of the Notes into a number of shares at the applicable conversion rate on the date of conversion if the last reported sale price of the Company's common stock exceeds 175% of the then applicable conversion price on at least twenty volume-weighted average price ("VWAP") trading days in any thirty consecutive VWAP trading day period beginning after the issuance date of the Notes and ending on the date upon which the Company notifies the Investor of such conversion.

The rollforward for the debt derivative liabilities for the three months ended September 30, 2023, was as follows:

	Fair Value on August 14, 2023 (Issuance Date)	Loss on fair value of debt derivatives	Fair Value on September 30, 2023
		(In thousands)	
Debt derivatives	\$ 13,890	\$ (3,648)	\$ 17,538
	\$ 13,890	\$ (3,648)	\$ 17,538

Fair Value Assumptions - Debt Derivative Liabilities

Assumptions used in the fair value of the debt derivative liabilities are described below.

	As of September 30, 2023
Current stock price	\$1.56
Expected volatility	65.8%
Risk-free interest rate	4.8%
Dividend yield	<del></del> %
Expected Term (years)	2.84

Expected volatility: The expected volatility was derived from the median volatility derived from the Company's guideline public companies as of September 30, 2023.

Risk-free interest rate: The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of grant for zero-coupon U.S. Treasury notes with maturities corresponding to the expected term of the Earnout Shares.

Expected dividend yield: The expected dividend rate is zero as the Company currently has no history or expectation of declaring dividends on its common stock.

Expected term: The expected term represents the period that the Notes are expected to be outstanding and is determined using the simplified method, which deems the term to be the average of the time to conversion and the contractual life of the Notes.

### Note 11. Equity Incentive Plans and Stock-Based Compensation

As of September 30, 2023, the Company had a remaining allocated reserve of 30,298,210 shares of its common stock for issuance under its 2021 Equity Incentive Plan (the "2021 EIP"), which provides for the granting of stock options, restricted stock units ("RSUs") and stock appreciation rights to employees, directors, and consultants of the Company. As of September 30, 2023, the Company had an allocated reserve of 7,371,214 shares of its common

stock for issuance under its 2021 Employee Stock Purchase Plan ("2021 ESPP"). As of September 30, 2023, the Company had not begun any offering periods for the 2021 ESPP.

### Stock options

Activity under the 2021 EIP is set forth below:

	Options	Weighted-Average Exercise Price	Weighted-Average Remaining Contractual Term in years
	(In thousands)	(Per share data)	(Years)
Outstanding as of December 31, 2021	21,191	\$ 0.58	8.2
Granted	_	\$ _	
Exercised	(2,795)	\$ 0.42	
Forfeited or expired	(665)	\$ 2.49	
Outstanding as of September 30, 2022	17,731	\$ 0.54	7.5
Options vested and expected to vest as of September 30, 2022	17,731	\$ 0.54	
Vested and exercisable as of September 30, 2022	10,145	\$ 0.65	
Outstanding as of December 31, 2022	16,960	\$ 0.54	7.3
Granted	_	\$ _	
Exercised	(1,400)	\$ 0.30	
Forfeited or expired	(424)	\$ 0.62	
Outstanding as of September 30, 2023	15,136	\$ 0.56	6.7
Options vested and expected to vest as of September 30, 2023	15,136	\$ 0.56	
Vested and exercisable as of September 30, 2023	12,327	\$ 0.64	

The aggregate intrinsic value of options outstanding was \$18.4 million and \$24.4 million, respectively, as of September 30, 2023 and December 31, 2022. Intrinsic value of options exercised for the nine months ended September 30, 2023 and 2022 was \$3.8 million and \$10.5 million, respectively. The total grant date fair value of options vested was \$0.6 million and \$1.5 million for the nine months ended September 30, 2023 and 2022, respectively.

As of September 30, 2023, total unrecognized compensation cost related to options was \$0.4 million related to 2.8 million unvested options and is expected to be recognized over a weighted-average period of 1.1 years.

For the nine months ended September 30, 2023, there wereno options granted.

### Restricted Stock Units

The fair value of RSUs under the Company's 2021 EIP is estimated using the value of the Company's common stock on the date of grant.

The following table summarizes outstanding and expected to vest RSUs as of September 30, 2023 and 2022 and their activity during the nine months ended September 30, 2023 and 2022:

	Number of Shares		Weighted-Average rant Date Fair Value	A	Aggregate Intrinsic Value
	(In thousands)	(Per share data)			(In thousands)
Balance as of December 31, 2021	4,041	\$	7.26	\$	31,563
Granted	2,912		3.81		11,500
Released	(385)		5.52		1,991
Cancelled	(370)		6.42		1,503
Balance as of September 30, 2022	6,198	\$	5.80	\$	24,422
Expected to vest as of September 30, 2022	6,198	\$	5.80	\$	24,422
Balance as of December 31, 2022	9,623	\$	4.47	\$	17,225
Granted	6,987		2.08		14,562
Released	(3,138)		3.39		5,862
Cancelled	(1,981)		3.97		4,454
Balance as of September 30, 2023	11,491	\$	3.40	\$	17,926
Expected to vest as of September 30, 2023	11,491	\$	3.40	\$	17,926

The aggregate intrinsic value of outstanding RSUs is calculated based on the closing price of the Company's common stock as of the date outstanding. As of September 30, 2023, there was \$33.2 million of unrecognized compensation cost related to 11.5 million unvested RSUs, which is expected to be recognized over a weighted average period of approximately 2.5 years. As of September 30, 2022, there was \$30.3 million of unrecognized compensation cost related to 6.2 million unvested RSUs, which is expected to be recognized over a weighted average period of approximately 3.1 years.

### Earnout Shares—Employees

The Earnout Shares issuable to holders of employee stock options are accounted as stock-based compensation expense as they are subject to forfeiture based on the satisfaction of certain employment conditions. The estimated fair values of the Earnout Shares associated with vested stock options are recognized as an expense and determined by the Monte Carlo simulation valuation model using a distribution of potential outcomes on a monthly basis over the five-year earnout period. The portion of the Earnout Shares associated with unvested stock options are recognized as an expense and considers the vesting continuing employment requirements.

### Stock-based Compensation Expense

The following sets forth the total stock-based compensation expense by type of award included in the statements of operations:

	Three Months Ended September 30,				Nine Months Ended September 30,				
	2023			2022		2023		2022	
				(In the	usands)				
Restricted stock units	\$	4,917	\$	2,831	\$	13,720	\$		7,787
Stock options		88		415		956			1,473
Earnout shares-employees		1,711		1,911		4,810			5,830
	\$	6,716	\$	5,157	\$	19,486	\$		15,090

The following sets forth the total stock-based compensation expense for the stock options and RSUs included in the statements of operations:

	Three Months Ended September 30,				Nine Months Ended September 30,			
		2023		2022		2023		2022
				(In tho	usands	s)		
Research and development	\$	3,115	\$	2,407	\$	9,104	\$	7,485
Selling and marketing		1,637		1,144		4,718		3,399
General and administrative		1,964		1,606		5,664		4,206
	\$	6,716	\$	5,157	\$	19,486	\$	15,090

### Note 12. Income Taxes

The income tax provision is calculated for an interim period by distinguishing between elements recognized in the income tax provision through applying an estimated annual effective tax rate (the "ETR") to a measure of year-to-date operating results referred to as "ordinary income (or loss)," and discretely recognizing specific events referred to as "discrete items" as they occur. The income tax provision or benefit for each interim period is the difference between the year-to-date amount for the current period and the year-to date amount for the period prior. Under ASC 740-270-30-36, entities subject to income taxes in multiple jurisdictions should apply one overall ETR instead of separate ETRs for each jurisdiction when calculating the interim-period income tax or benefit related to ordinary income (or loss) for the year-to-date interim period, except in certain circumstances. The Company's effective tax rates for the three and nine months ended September 30, 2023 and 2022 differ from the federal statutory rate principally as a result of valuation allowances expected to be applied to net operating loss carry-forwards which will not meet the threshold for recognition as deferred tax assets.

### Note 13. Commitments and Contingencies

The Company may be involved in various lawsuits, claims, and proceedings, including intellectual property, commercial, securities, and employment matters that arise in the normal course of business. The Company accrues a liability when management believes information available prior to the issuance of the condensed consolidated financial statements indicates it is probable a loss has been incurred as of the date of the condensed consolidated financial statements and the amount of loss can be reasonably estimated. The Company adjusts its accruals to reflect the impact of negotiations, settlements, rulings, advice of legal counsel, and other information and events pertaining to a particular case. Legal costs are expensed as incurred. As of September 30, 2023 and December 31, 2022, the Company is not aware of any litigation, claim or assessment in which the outcome, individually or in the aggregate, would have a material adverse effect on its financial positions, results of operations, cash flows or future earnings.

The Company's purchase commitments, per terms and conditions with suppliers and vendors, are cancellable in whole or in part prior to shipment. Non-cancellable purchase commitments (purchase orders) of \$30.8 million for parts and assemblies are due upon receipt and will primarily be delivered in the remainder of 2023 and the first half of 2024. If inventory is shipped, the Company will accrue a liability under accrued expenses. The Company has no other commitment and contingencies, except for the operating leases. See Note 8, *Leases*, for further discussion.

### Note 14. Employee Defined-Contribution Plans

The Company has a defined-contribution plan intended to qualify under Section 401 of the Internal Revenue Code (the "401(k) Plan"). The Company contracted with a third-party provider to act as a custodian and trustee, and to process and maintain the records of participant data. Substantially all of the expenses incurred for administering the 401(k) Plan are paid by the Company. Accrued salaries and benefits included accruals related to the 401(k) plans the Company offers to its employees. In order to qualify for these plans, employees must meet the minimum age requirement (21 years) and begin participating on their entry date which is the first paycheck date in the month following the month of eligibility described above. Employee and employer contributions are immediately 100% fully vested. The plans offer employer contributions of 3.0% of an employee's eligible compensation following safe-harbor rules. The Company's contribution to the 401(k) plan was \$0.3 million and \$0.3 million for the three months ended September 30, 2023 and 2022, respectively, and \$1.0 million and \$0.8 million for the nine months ended September 30, 2023 and 2022, respectively. The Company has paid all matching contributions as of September 30, 2023.

### Note 15. Revenue

### **Customer Concentration**

The customer concentration for balances greater than 10% of revenues and 10% of accounts receivables, net, respectively, are presented below:

		Total Re	Accounts Receivable, Net									
	Three Months Ended	Three Months Ended September 30,		September 30,	September 30,	December 31,						
	2023	2022	2023	2022	2023	2022						
		(as a percentage)										
Customer 1	14.2 %	—%	<10 %	<b>—</b> %	— %	— %						
Customer 2	22.8 %	— %	<10 %	<b>—</b> %	— %	— %						
Customer 3	14.1 %	—%	<10 %	<b>—</b> %	<10 %	— %						
Customer 4	13.3 %	— %	<10 %	— %	— %	— %						
Customer 5	12.4 %	—%	<10 %	<b>—</b> %	18.0 %	— %						
Customer 6	<10 %	36.7 %	<10 %	38.6 %	13.0 %	<10 %						
Customer 7	<10 %	<10 %	<10 %	11.5 %	<10 %	— %						
Customer 8	<10 %	17.1 %	<10 %	<10 %	<10 %	— %						
Customer 9	<10 %	12.5 %	<10 %	<10 %	<10 %	— %						

### Revenue by Geographic Area

The Company currently sells its products in the geographic regions as follows:

	Three Months Ended September 30,				Nine Months Ended September 30,				
	2023		2022		2023		2022		
		(In thousands)							
Americas	\$	23,407	\$	18,933	\$	66,620	\$	50,592	
Europe		329		37		8,927		111	
Other		72		145		209		274	
Total	\$	23,808	\$	19,115	\$	75,756	\$	50,977	

#### Contract Assets and Liabilities

There was \$0.6 million and \$2.3 million of revenue recognized during the three and nine months ended September 30, 2023, respectively, included in contract liabilities as of December 31, 2022. The amount of revenue recognized during the three and nine months ended September 30, 2022 included in contract liabilities as of December 31, 2021 was \$0.9 million and \$2.5 million, respectively. The change in contract assets reflects the difference in timing between the Company's satisfaction of remaining performance obligations and the Company's contractual right to bill its customers. The Company had no material asset impairment charges related to contract assets in the periods presented.

### Variable Consideration

The Company estimates its variable consideration on a quarterly basis based on the latest data available, and adjust the transaction price accordingly by recording an adjustment to net revenue and contract assets. The Company has recognized the estimate of variable consideration to the extent that it is probable that a significant reversal will not occur as a result from a change in estimation. Revenue related to variable consideration represented \$3.9 million and \$3.1 million of the Company's revenue during the three months ended September 30, 2023 and 2022, respectively, and \$3.9 million and \$4.0 million for the nine months ended September 30, 2023 and 2022, respectively.

### Note 16. At-the-Market Offering

On February 6, 2023, the Company entered into the ATM Sales Agreement with Needham, as agent, pursuant to which the Company may offer and sell, from time to time through Needham, shares of its common stock. As of the three and nine months ended September 30, 2023, the Company has sold \$2.8 million and \$18.4 million of shares, respectively, net of issuance costs of \$0.3 million and \$1.2 million, respectively.

The offer and sale of the shares of common stock will be made pursuant to the Shelf Registration Statement and a related prospectus supplement and accompanying prospectus. Pursuant to the prospectus supplement and the ATM Sales Agreement, the Company may offer and sell up to \$40 million of shares of common stock pursuant to its Shelf Registration Statement. Sales of shares, if any, under the prospectus supplement and the accompanying prospectus may be made by any method permitted that is deemed to be an "at the market offering" as defined in Rule 415(a)(4) promulgated under the Securities Act.

The Company will pay Needham commissions for its services in acting as agent in the sale of the shares pursuant to the ATM Sales Agreement. Needham will be entitled to compensation at a fixed commission rate equal to 3.0% of the aggregate gross proceeds from each sale of the shares pursuant to the ATM Sales Agreement. The Company has agreed to provide Needham with customary indemnification and contribution rights, including for liabilities under the Securities Act. The Company also will reimburse Needham for certain specified expenses in connection with entering into the ATM Sales Agreement. The Sales Agreement contains customary representations and warranties and conditions to the placements of the shares pursuant thereto.

# Velo3D, Inc. Notes to Condensed Consolidated Financial Statements (Unaudited)

## Note 17. Subsequent Events

# Reduction in Force & Facilities Expense

On October 9, 2023, the Company commenced a reduction in force plan to streamline its business operations, reduce costs and create further operating efficiencies, which affected 62 employees, representing approximately 21% of the Company's workforce. In connection with the reduction in force, the Company currently estimates it will incur approximately \$1.0 million to \$2.0 million of costs, consisting primarily of personnel expenses such as salaries and wages, one-time severance payments, and other benefits. The majority of the cash payments related to these expenses will be paid out during the fourth quarter of 2023. The reduction in force was completed by the end of October 2023. Additionally, the Company has begun the process of closing down several of its facilities. The carrying value of the right of use asset and leasehold improvements of these facilities is \$0.7 million and \$0.5 million, respectively, and the period of amortization would be accelerated to the end of term.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis provides information which our management believes is relevant to an assessment and understanding of our results of operations and financial condition. This discussion should be read in conjunction with our Annual Report on Form 10-K for the year ended December 31, 2022 and our unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report. This discussion contains forward-looking statements based upon our current expectations, estimates and projections, and involves numerous risks and uncertainties. Actual results may differ materially from those contained in any forward-looking statements due to, among other considerations, the matters discussed in the sections titled "Risk Factors" and "Forward-looking Statements" herein.

#### Overview

We seek to fulfill the promise of additive manufacturing ("AM"), also referred to as three-dimensional printing ("3D printing"), to deliver breakthroughs in performance, cost, and lead time in the production of high value metal parts.

We produce a fully integrated hardware and software solution based on our proprietary laser powder bed fusion ("L-PBF") technology, which greatly reduces and often eliminates the need for support structures. Our technology enables the production of highly complex, mission-critical parts that existing AM solutions cannot produce without the need for redesign or additional assembly.

Our Sapphire family of systems (collectively referred to as the "3D Printers") give our customers who are in space, aviation, defense, automotive, energy and industrial markets the freedom to design and produce metal parts with complex internal features and geometries that had previously been considered impossible for AM. We believe our technology is years ahead of competitors.

Our technology is novel compared to other AM technologies based on its ability to deliver high-value metal parts that have complex internal channels, structures, and geometries. This affords a wide breadth of design freedom for creating new metal parts and it enables replication of existing parts without the need to redesign the part to be manufacturable with AM. Because of these features, we believe our technology and product capabilities are highly valued by our customers. Our customers are primarily original equipment manufacturers ("OEMs") and contract manufacturers ("CMs") who look to AM to solve issues with traditional metal parts manufacturing technologies. Traditional manufacturing technologies rely on processes, including casting, stamping and forging, that typically require high volumes to drive competitive costs and have long lead times for production. Our customers look to AM solutions to produce assemblies that are lighter, stronger, and more reliable than those manufactured with traditional technologies. Our customers also expect AM solutions to drive lower costs for low-volume parts and substantially shorter lead times. However, many of our customers have found that legacy AM technologies failed to produce the required designs for the high-value metal parts and assemblies that our customers wanted to produce with AM. As a result, other AM solutions often require that parts be redesigned so that they can be produced and frequently incur performance losses for high-value applications.

In contrast, our technology can deliver complex high value metal parts with design advantages, lower costs and faster lead times associated with AM, and generally avoids the need to redesign the parts. As a result, our customers have increasingly adopted our technology into their design and production processes. We believe our value is reflected in our sales patterns, as most customers purchase a single machine to validate our technology and purchase additional systems over time as they embed our technology in their product roadmap and manufacturing infrastructure. We consider this approach a "land and expand" strategy, oriented around a demonstration of our value proposition followed by increasing penetration with key customers.

#### Recent Trends and Strategic Realignment

We have experienced less revenue growth than expected due to the impact of delayed shipments during the nine months ended September 30, 2023. Further, due to the impact of fourth quarter customer order delays and our bookings to date, we expect additional contraction of revenue growth in the near term.

As a result, we now believe that our focus on revenue growth has come at the expense of our cash flow and profitability and our commitment to the highest level of customer service. In October 2023, we made a strategic decision to realign our operations to pivot from emphasizing revenue growth to optimizing our free cash flow, maximizing customer success, reducing expenditures, and improving our operational efficiency. We are undertaking expense reduction and cash savings initiatives as part of a company-wide restructuring and strategic realignment plan to help conserve working capital. The expense reduction and cash saving initiatives include streamlining facilities, managing working capital, reducing capital expenditures, and reducing overall selling, general and administrative expenses. See Note 17, Subsequent Events, in the unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report. In addition, we have also implemented new go-to-market and service strategies to rebuild our bookings and backlog pipeline.

Certain of the delays in our fourth quarter customer orders in part relate to transitional disruption resulting from the implementation of these operational realignment initiatives and sales and service strategies. In addition, we believe the uncertainty regarding our financial situation as a result of the negotiations with the holders of our senior secured convertible notes (the "Notes") may have impacted, and may continue to impact our bookings in the near term. See "—Liquidity and Capital Resources" below.

#### **Key Financial and Operational Metrics**

We believe that our performance and future success depend on many factors that present significant opportunities for us but also pose risks and challenges, including those discussed below and in the section of this Quarterly Report titled "Risk Factors."

We regularly evaluate several metrics, including the metrics presented in the table below, to measure our performance, identify trends affecting our business, prepare financial projections, make strategic decisions, and establish performance goals for compensation and we periodically review and revise these metrics to reflect changes in our business.

	As of and for the Three 30,		Months Ende	ed September	As of and for the Nine Months Ended Septe 30,			
	2	023		2022	:	2023		2022
Revenue (\$ in millions)	\$	24	\$	19	\$	76	\$	51
Bookings (\$ in millions)	\$	11	\$	27	\$	47	\$	63
Backlog (\$ in millions)	\$	6	\$	66	\$	6	\$	66

Bookings (\$ in millions): Bookings (\$ in millions) are defined as a confirmed order for a 3D printer system in contracted dollars.

Backlog (\$ in millions): Backlog (\$ in millions) is defined as the unfulfilled 3D printer systems to be delivered to customers in contracted dollars as of period end.

Customer Concentration

Our operating results for the foreseeable future will continue to depend on sales to a small group of customers. For the three months ended September 30, 2023 and 2022, sales to the top three customers accounted for 51.1% and 66.3%, respectively. Of the top three customers for the three months ended September 30, 2023, all customers were different from the top three customers for the comparable period in 2022.

For the nine months ended September 30, 2023 and 2022, sales to the top three customers accounted for 24.7% and 57.4%, respectively. Of the top three customers for the three months ended September 30, 2023 all customers were different from the top three customers for the comparable period in 2022.

While our objective is to diversify our customer base, we could continue to be susceptible to risks associated with customer concentration. See *Risk Factors—Risks* Related to Our Business—Risks Related to Our Financial Position and Need for Additional Capital—We expect to rely on a limited number of customers for a significant portion of our near-term revenue" in our Annual Report on Form 10-K for the year ended December 31, 2022 and see Note 2\$\mummary\$ of Significant Accounting Policies—Concentration of Credit Risk and Other Risks and Uncertainties, in the audited consolidated financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2022.

#### Continued Investment and Innovation

We continue to be a customer-focused company working to develop innovative solutions to address customers' needs and focus on our customers to identify the most impactful areas for research and development as we seek to further improve the capabilities of our AM solutions. We believe this process has contributed significantly to our development of the most advanced metal AM systems in the world. We believe that continued investments in our products are important to our future growth and, as a result, we will invest in enhancing our portfolio of AM solutions through certain research and development projects based on customer demand.

## **Impact of World Events**

In May 2023, the World Health Organization announced that the COVID-19 pandemic no longer qualifies as a global emergency. Nevertheless, we may continue to face various risks and uncertainties related to the effect of the COVID-19 pandemic. For example, we may be impacted by various supply chain constraints, travel restrictions that limit our ability to sell, install and service our additive manufacturing systems, government-enforced workplace restrictions and/or other future pandemic related mandates, which may adversely impact our operations.

General economic and political conditions such as recessions, interest rates, fuel prices, actual or perceived instability in the U.S. or global banking systems, inflation, foreign currency fluctuations, international tariffs, social, political and economic risks and acts of war or terrorism (including, for example, the war in Israel and the war in Ukraine and the economic sanctions related to it), have added uncertainty in timing of customer orders and supply chain constraints.

During 2022, supply chain challenges increased our material and shipping costs, resulted in shipping delays and impacted our gross margins. We implemented several supply chain and manufacturing improvements in response and intend to continue to focus on driving further operational improvements during 2023 as well as company-wide initiatives to reduce operating costs, as described above under "—Recent Trends and Strategic Realignment."

## Climate Change

Material pending or existing climate change-related legislation, regulations, and international accords could have an adverse effect on our business, financial condition, and results of operations, including: (1) material past and/or future capital expenditures for climate-related projects, (2) material indirect consequences of climate-related regulation or business trends, such as the following: decreased/increased demand for goods or services that produce significant greenhouse gas emissions or are related to carbon-based energy sources; increased competition to develop innovative new products that result in lower emissions; increased demand for generation and transmission of energy from alternative energy sources; and any anticipated reputational risks resulting from operations or products that produce material greenhouse gas emissions and (3) material increased compliance costs related to climate change. In addition, extreme weather and other natural disasters may become more intense or more frequent, which may disrupt our operations or the operations of our suppliers and customers.

#### Components of Results of Operations

Revenue

Our revenue is primarily derived from our AM fully integrated hardware and software solution based on our proprietary L-PBF technology. Sapphire, Sapphire 1MZ, Sapphire XC and Sapphire XC 1MZ metal AM printer using our L-PBF technology and Assure quality validation software (collectively referred to as the "3D Printer"). Contracts for 3D Printers also include post-sale customer support services ("Support Services"), except for our distributor partners, which are qualified to perform support services.

We sell our fully integrated hardware and software AM solutions through two types of transaction models: a 3D Printer sale transaction and a recurring payment transaction ("Recurring Payment"). Support services are included with a 3D Printer sale transaction and a recurring payment transaction. For 3D Printer sale transactions where the support service period has expired, customers have purchased extended support service contracts.

**3D Printer sale transactions** - can be divided into two categories: structured fixed purchase price and sale and utilization (variable consideration) fee model. In the sale and utilization fee model, customers pay a partial amount upfront to acquire the system, which is less than the full purchase price. This amount is then complemented by an hourly usage fee for the duration of the system's life. The variable payments are recognized when there is an event that determines the amount of variable consideration to be paid. Sales with variable consideration represented \$3.9 million and \$3.1 million of our revenue during the three months ended September 30, 2023 and 2022, respectively. Sales with variable consideration represented \$3.9 million and \$4.0 million of our revenue during the nine months ended September 30, 2023 and 2022, respectively. For more information, see "—Critical Accounting Policies and Significant Estimates—Revenue — Variable Consideration" in Part II, Item 7 "Management's Discussion and Analysis of Financial Condition and Results of Operations" of our Annual Report on Form 10-K for the year ended December 31, 2022.

The timeframe from order to completion of the site acceptance test usually occurs over three to six months. As we scale our production, we expect to reduce this timeframe. Contract consideration allocated to the 3D Printer is recognized at a point in time, which occurs upon transfer of control to the customer at shipment.

The initial sales of 3D Printers and Support Services are included in one contract and are invoiced together. Contract consideration is allocated between the two performance obligations based on relative fair value. This allocation involves judgement and is periodically updated as new relevant information becomes available.

**Recurring payment transactions -** are our leased 3D Printer transactions. We define our Recurring Payment transactions as operating leases. Under the leased 3D Printer transaction, the customer typically pays an amount for a lease which entitles the customer to a base number of hours of usage. For usage above that level, the customer typically pays an hourly usage fee. Most of our leases have a 12-month term, though in some instances the lease term is longer.

The Recurring Payment transactions, structured as operating leases, were 2.2% and 6.2% of revenue for the three months ended September 30, 2023 and 2022, respectively. The Recurring Payment transactions were 1.5% and 6.0% of revenue for the nine months ended September 30, 2023 and 2022, respectively. Under this model, the customer typically pays a base rent and variable payments based on usage in excess of a defined threshold. Most of our leases have a 12-month term, though in certain cases the lease term is longer.

Support Services - are included with most 3D Printer sale transactions and Recurring Payment transactions. Support services consist of field service engineering, phone and email support, preventative maintenance, and limited on and off-site consulting support. A subsequent Extended Support Agreement is available for renewal after the initial contract period based on the then-fair value of the service, which is paid for separately. Support Service revenue is recognized over the contract period beginning with customer performance test acceptance.

Other revenue included under 3D Printer sales includes parts and consumables, such as filters, powder or build plates, that are sold to customers and recognized upon transfer of control to the customer at shipment. Other revenue was not material for the three months ended September 30, 2023 and 2022, respectively.

#### Cost of Revenue

Our cost of revenue includes the "Cost of 3D Printers," "Cost of Recurring Payment" and "Cost of Support Services."

Cost of 3D Printers includes the manufacturing cost of our components and subassemblies purchased from vendors for the assembly, as well as raw materials and assemblies, shipping costs and other directly associated costs. Cost of 3D Printers also includes allocated overhead costs from headcount-related costs, such as salaries, stock-based compensation, depreciation of manufacturing related equipment and facilities, and information technology costs.

Cost of Recurring Payment includes depreciation of the leased equipment over the useful life of five years less the residual value, and an allocated portion of Cost of Support Services.

Cost of Support Services includes the cost of spare or replacement parts for preventive maintenance, installation costs, headcount-related costs such as salaries, stock-based compensation, depreciation of manufacturing related equipment and facilities, and information technology costs. The headcount-related costs are directly associated with the engineers dedicated to remote and on-site support, training, travel costs and other services costs.

#### Gross Profit and Gross Margin

Our gross profit is revenue less cost of revenue and our gross margin is gross profit as a percentage of revenue. The gross profit and gross margin for our products are varied and are expected to continue to vary from period to period due to the mix of products sold through either a 3D Printer sale transaction or a Recurring Payment transaction, new product introductions and efforts to optimize our operational costs. Other factors affecting our gross profit include changes to our material costs, assembly costs that are themselves dependent upon improvements to yield, and any increase in assembly overhead to support a greater number of 3D Printers sold and markets served.

#### Research and Development Expenses

Our research and development expenses represent costs incurred to support activities that advance the development of innovative AM technologies, new product platforms and consumables, as well as activities that enhance the capabilities of our existing product platforms. Our research and development expenses consist primarily of salaries and related personnel costs for individuals working in our research and development departments, including stock-based compensation, prototypes, design expenses, information technology costs and software license amortization, consulting and contractor costs, and an allocated portion of overhead costs, including depreciation of property and equipment used in research and development activities.

#### Selling and Marketing Expenses

Sales and marketing expenses consist primarily of salaries and related personnel costs for individuals working in our sales and marketing departments, including stock-based compensation, costs related to trade shows and events, advertising, marketing promotions, travel costs and an allocated portion of overhead costs, including information technology costs and costs for customer validation.

## General and Administrative Expenses

General and administrative expenses consist primarily of salaries and related personnel costs for individuals associated with our executive, administrative, finance, legal, information technology and human resources functions, including stock-based compensation, professional fees for legal, audit and compliance, accounting and consulting services, general corporate costs, facilities, rent, information technology costs, insurance, bad debt expenses and an

allocated portion of overhead costs, including equipment and depreciation and other general and administrative expenses.

Interest Expense

Interest expense primarily consists of interest incurred under our outstanding debt and finance leases.

Gain (Loss) on Fair Value of Warrants

Gain (loss) on valuation of warrant liabilities relates to the changes in the fair value of warrant liabilities, including liabilities related to the public warrants and private placement warrants, which are subject to remeasurement at each balance sheet date.

Gain (Loss) on Fair value of Contingent Earnout Liabilities

Gain (loss) on valuation of contingent earnout liabilities relates to the changes in fair value of the contingent earnout liabilities in connection with the earnout shares, which are subject to remeasurement at each balance sheet date. See Note 10, Equity Instruments—Contingent Earnout Liabilities, in the unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report.

Gain (loss) on Fair value of Debt Derivatives

Gain (loss) on valuation of debt derivatives relates to the non-cash fair value change of the debt derivatives driven by the change in stock price. See Note 10Equity Instruments, in the unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report.

Loss on Debt Extinguishment

Loss on debt extinguishment relates to the loss recorded in accordance with the termination of our third amended and restated loan and security agreement, as amended (the "Loan Agreement"), originally entered into with Silicon Valley Bank ("SVB"). See Note 9, Long—Term Debt, in the unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report.

Other Income (Expense), Net

Other income (expense), net includes interest earned on our bank sweep account, gains and losses on disposals of fixed assets and other miscellaneous income/expenses.

Income Taxes

No provision for federal and state income taxes was recorded for any periods presented due to projected losses, and we maintained a full valuation allowance on the deferred tax assets as of September 30, 2023 and September 30, 2022.

We will continue to review our conclusions about the appropriate amount of the valuation allowance on a quarterly basis. If we were to generate profits, the U.S. valuation allowance position could be reversed in the foreseeable future. We expect a benefit to be recorded in the period the valuation allowance reversal is recorded and a higher effective tax rate in periods following the valuation allowance reversal.

# **Results of Operations**

# Comparison of the Three Months Ended September 30, 2023 and 2022:

The following table summarizes our historical results of operations for the periods presented:

	Three Months En	ded September 30,		
	2023	2022	Change	%
		(In thousands, ex	ccept for percentages)	
Revenue				
3D Printer	\$ 21,428	\$ 16,537	\$ 4,891	29.6 %
Recurring payment	531	1,183	(652)	(55.1)%
Support services	1,849	1,395	454	32.5 %
Total Revenue	23,808	19,115	4,693	24.6 %
Cost of revenue				
3D Printer	20,273	16,574	3,699	22.3 %
Recurring payment	111	656	(545)	(83.1)%
Support services	1,936	2,006	(70)	(3.5)%
Total cost of revenue	22,320	19,236	3,084	16.0 %
Gross profit	1,488	(121)	1,609	(1329.8)%
Operating expenses				
Research and development	9,819	12,558	(2,739)	(21.8)%
Selling and marketing	5,772	5,632	140	2.5 %
General and administrative	11,118	9,642	1,476	15.3 %
Total operating expenses	26,709	27,832	(1,123)	(4.0)%
Loss from operations	(25,221)	(27,953)	2,732	(9.8)%
Interest expense	(1,107)	(129)	(978)	758.1 %
Gain on fair value of warrants	1,587	(6,612)	8,199	(124.0)%
Gain on fair value of contingent earnout liabilities	10,810	(40,885)	51,695	(126.4)%
Loss on fair value of debt derivatives	(3,648)	_	(3,648)	100.0 %
Loss on debt extinguishment	(253)	_	(253)	100.0 %
Other income, net	436	384	52	13.5 %
Income (loss) before provision for income taxes	(17,396)	(75,195)	57,799	(76.9)%
Provision for income taxes	_		_	<b>—</b> %
Net income (loss)	\$ (17,396)	\$ (75,195)	\$ 57,799	(76.9)%

#### Revenue

The following table presents the revenue disaggregated by products and service type, as well as the percentage of total revenue.

		1	Three Months En	ded	September 3	80,		
		20	023		20	)22	Change	%
	-		(	In th	nousands, exc	ept for percentages	)	
3D Printer sales	\$	21,428	90.0 %	\$	16,537	86.5 %	\$ 4,891	29.6 %
Recurring payment		531	2.2 %		1,183	6.2 %	(652)	(55.1)%
Support services		1,849	7.8 %		1,395	7.3 %	454	32.5 %
Total Revenue	\$	23,808	100.0 %	\$	19,115	100.0 %	\$ 4,693	24.6 %

Total revenue for the three months ended September 30, 2023 and 2022 was \$23.8 million and \$19.1 million, respectively, an increase of \$4.7 million, or 24.6%.

3D Printer sales were \$21.4 million and \$16.5 million for the three months ended September 30, 2023 and 2022, respectively, an increase of \$4.9 million. The improvement in revenue was due to a change in our product mix toward more Sapphire XC systems compared to Sapphire systems and fewer launch customer and early reservation shipments at discounted prices, resulting in an increase in the average selling price. The 3D Printer sales included parts and consumables revenue.

Recurring Payment, structured as an operating lease, was \$0.5 million and \$1.2 million for the three months ended September 30, 2023 and 2022, respectively. The decrease was primarily attributable to a decrease in the number of 3D Printer systems in service generating Recurring Payment revenue as of September 30, 2023, compared to the number of 3D Printer systems in service as of September 30, 2022.

Our Support Service revenue was \$1.8 million and \$1.4 million for the three months ended September 30, 2023 and 2022, respectively. The increase was primarily attributable to the increase in the number of 3D Printer systems in service as of September 30, 2023, compared to the number of 3D Printers in service as of September 30, 2022.

As discussed above, due to the impact of fourth quarter customer order delays and our bookings to date, we expect additional contraction of revenue growth in the near term, as we focus on the implementation of our operational realignment initiatives and go-to-market and service strategies, and because of the uncertainty regarding our financial situation as a result of the negotiations with the holders of the Notes. In the medium term, we expect the demand for the Sapphire family of systems, primarily the Sapphire XC to drive future revenue growth. We also expect our Recurring Payment and Support Service revenue to increase as the number of systems we have in the field increases. As of September 30, 2023, our backlog for firm orders was \$6 million for 3D Printers. If we are able to resolve our negotiations with the holders of the Notes, we expect our new sales strategy, especially in the defense and aerospace industries, will begin to yield positive results in the coming year 2024.

In connection with our realignment strategy, we are taking a strategic approach to our marketing and selling efforts to provide an improved customer experience, which we expect will enhance our efforts to introduce our technology to new customers and expand our existing customer network to increase demand.

#### Cost of Revenue

The following table presents the Cost of Revenue disaggregated by product and service type, as well as the percentage of total cost of revenue.

	 T	hree Months End	led S	September 3	30,			
	20	123		20	022	(	Change	%
		(Iı	n the	ousands, exce	pt for percentages)	1		
Cost of Revenue								
Cost of 3D Printers	\$ 20,273	90.8 %	\$	16,574	86.2 %	\$	3,699	22.3 %
Cost of Recurring Payment	111	0.5 %		656	3.4 %		(545)	(83.1)%
Cost of Support Services	1,936	8.7 %		2,006	10.4 %		(70)	(3.5)%
Total Cost of Revenue	\$ 22,320	100.0 %	\$	19,236	100.0 %	\$	3,084	16.0 %

Total cost of revenue for the three months ended September 30, 2023 and 2022 was \$22.3 million and \$19.2 million, respectively, an increase of \$3.1 million, or 16.0%.

Cost of 3D Printers was \$20.3 million and \$16.6 million for the three months ended September 30, 2023 and 2022, respectively. The increase of \$3.7 million was due to a change in our product mix towards more Sapphire XC systems, which have a higher total per system cost compared to the Sapphire, for the three months ended September 30, 2023 compared to 2022.

Cost of Recurring Payment was \$0.1 million and \$0.7 million for the three months ended September 30, 2023 and 2022, respectively. This decrease of \$0.5 million was due to a decrease in the number of 3D Printer systems in service incurring costs related to Recurring Payment revenue as of September 30, 2023, compared to September 30, 2022.

Cost of Support Services was \$1.9 million and \$2.0 million for the three months ended September 30, 2023 and 2022, respectively. Cost of Support Services decreased by \$0.1 million, although there were more 3D Printers in service as of September 30, 2023, compared to September 30, 2022, Cost of Support Services decreased on a per unit basis. Field service engineering support costs related to the ramp-up of the Sapphire XC, Sapphire 1MZ and Sapphire XC 1MZ systems in the field have been offset by the benefits of preventative maintenance we performed and enhancements we made to system reliability performance. We expect Cost of Support Services to continue to decrease on a per unit basis as the Sapphire XC, Sapphire 1MZ and Sapphire XC 1MZ system performance improves. We expect our Cost of Support Services will increase with the delivery of more 3D Printer systems to customers.

Cost of revenue as a percentage of revenue was 93.8% and 100.6% for the three months ended September 30, 2023 and 2022, respectively. The decrease in the cost of revenue as a percentage of revenue was primarily due to a per system reduction in material costs for our Sapphire family of systems and improved manufacturing efficiency offset by the change in our product mix towards more Sapphire XC systems compared to Sapphire systems.

We continue to focus on reducing our material costs through improved purchasing and inventory planning, accelerating production cycle times and improving efficiencies on the production floor, to lower our cost of revenue. We expect our cost of revenue as a percentage of revenue to continue to improve as we address material costs and manufacturing efficiency, and as our production volume increases and system reliability performance improves.

# Gross Profit and Gross Margin

Total gross profit (loss) was \$1.5 million and \$(0.1) million for the three months ended September 30, 2023 and 2022, respectively. As a percentage of revenue, the gross margin was 6.3% and (0.6)% for the three months ended September 30, 2023 and 2022, respectively. The higher gross profit for the three months ended September 30, 2023 was primarily attributable to higher average selling prices for our Sapphire family of systems, and improved

manufacturing efficiencies, during the three months ended September 30, 2023, as compared to the three months ended September 30, 2022.

Our gross profit and gross margin are influenced by a number of factors, including:

- Product mix of Sapphire, and Sapphire XC systems;
- · Average selling prices for our systems;
- · Trends in materials and shipping costs;
- Production volumes that may impact factory overhead absorption;
- · System reliability performance; and
- Impact of product mix changes, including new product introductions, and other factors, on our Cost of Support Services

During the remainder of 2023, we expect our gross margins will remain relatively flat while we continue to focus on improving our average selling prices, reductions in material costs, and improvements to manufacturing efficiency.

Research and Development Expenses

Research and development expenses were \$9.8 million and \$12.6 million for the three months ended September 30, 2023 and 2022, respectively, a decrease of \$2.8 million. The decrease in research and development expenses in 2023 was driven by a \$2.9 million decrease in purchased materials, a \$0.4 million decrease in headcount, salaries and employee-related expenses, and a decrease of \$0.2 million in other operating expenses, offset by an increase of \$0.7 million in stock-based compensation.

We expect research and development costs to decrease for the remainder of 2023 as our Sapphire family of systems mature. We will invest in enhancing our portfolio of AM solutions through certain research and development projects based on customer demand.

Selling and Marketing Expenses

Selling and marketing expenses were \$5.8 million and \$5.6 million for the three months ended September 30, 2023 and 2022, respectively, an increase of \$0.2 million. The increase was attributable to an increase of \$0.5 million in stock-based compensation, a \$0.4 million increase in headcount, salaries and employee-related expenses, and a \$0.1 million increase in general marketing initiatives, offset by an \$0.8 million decrease in European marketing costs.

We expect selling and marketing expenses to decrease for the remainder of 2023 as we implement our realignment strategy. During the remainder of 2023, we intend to focus on certain markets that show strong attendance at additive manufacturing conferences to build product awareness.

General and Administrative Expenses

General and administrative expenses were \$11.1 million and \$9.6 million for the three months ended September 30, 2023 and 2022, respectively, an increase of \$1.5 million. The increase was attributable to a \$0.9 million increase in headcount, salaries and employee-related expenses, a \$0.5 million increase in public company-related expenses in advisory, legal and accounting fees and insurance, and a \$0.3 million increase in stock-based compensation, offset by a \$0.2 million decrease in facilities expenses.

We expect general and administrative expenses to increase for the remainder of 2023 due to one-time charges related to our reduction in force of approximately \$1.0 million to \$2.0 million and charges to consolidate our facilities of approximately \$2.0 million. We continue to focus on our company-wide initiatives to

reduce operating costs for the remainder of 2023 as we implement our realignment strategy by reducing our general and administrative expenses through reducing outside consultants, managing effective facility spending and negotiating with vendors for improved pricing.

Interest Expense

Interest expense was \$1.1 million and \$0.1 million for the three months ended September 30, 2023 and 2022, respectively.

We expect our interest expense will increase as a result of our recent issuance of the Notes (for further information, see "—Liquidity and Capital Resources" and Note 9, Long-Term Debt, in the unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report).

Gain (loss) on Fair Value of Warrants

The change in fair value of warrants resulted in a gain of \$1.6 million and a loss of \$6.6 million for the three months ended September 30, 2023 and 2022, respectively, and was related to the non-cash fair value change of the warrant liabilities driven by the change in our stock price.

Gain (loss) on Fair value of Contingent Earnout Liabilities

The change in fair value of the contingent earnout liability was a gain of \$10.8 million and a loss of \$40.9 million for the three months ended September 30, 2023 and 2022, respectively, and was related to the non-cash fair value change of the contingent earnout liabilities driven by the change in our stock price.

Loss on Fair value of Debt Derivatives

The change in fair value of the debt derivatives was a loss of \$3.6 million for the three months ended September 30, 2023, and was related to the non-cash fair value change of the debt derivatives driven by the change in our stock price.

Loss on Debt Extinguishment

The loss on debt extinguishment was \$0.3 million for the three months ended September 30, 2023, and was related to the termination of the Loan Agreement with Silicon Valley Bank.

Other Income, Net

Other income, net was \$0.4 million and \$0.4 million for the three months ended September 30, 2023 and 2022, respectively.

Income Taxes

No provision for federal and state income taxes was recorded for both the three months ended September 30, 2023 and 2022 due to projected losses, and we maintained a full valuation allowance on the deferred tax assets as of September 30, 2023 and December 31, 2022.

We will continue to review our conclusions about the appropriate amount of the valuation allowance on a quarterly basis. If we were to generate profits in 2023 and beyond, the U.S. valuation allowance position could be reversed in the foreseeable future. We expect a benefit to be recorded in the period the valuation allowance reversal is recorded and a higher effective tax rate in periods following the valuation allowance reversal.

# **Results of Operations**

# Comparison of the Nine Months Ended September 30, 2023 and 2022:

The following table summarizes our historical results of operations for the periods presented:

	N	ine Months End	ded September 30,		
		2023	2022	Change	%
			(In thousands, e	xcept for percentages)	
Revenue					
3D Printer	\$	69,193	\$ 44,336	\$ 24,857	56.1 %
Recurring payment		1,141	3,042	(1,901)	(62.5)%
Support services		5,422	3,599	1,823	50.7 %
Total Revenue		75,756	50,977	24,779	48.6 %
Cost of revenue					
3D Printer		61,975	42,686	19,289	45.2 %
Recurring payment		893	2,059	(1,166)	(56.6)%
Support services		5,495	5,106	389	7.6 %
Total cost of revenue		68,363	49,851	18,512	37.1 %
Gross profit		7,393	1,126	6,267	556.6 %
Operating expenses					
Research and development		32,820	38,438	(5,618)	(14.6)%
Selling and marketing		18,054	17,864	190	1.1 %
General and administrative		31,569	27,191	4,378	16.1 %
Total operating expenses		82,443	83,493	(1,050)	(1.3)%
Loss from operations		(75,050)	(82,367)	7,317	(8.9)%
Interest expense		(1,671)	(362)	(1,309)	361.6 %
Gain (loss) on fair value of warrants		(138)	11,039	(11,177)	(101.3)%
Gain (loss) on fair value of contingent earnout liabilities		3,000	58,110	(55,110)	(94.8)%
Loss on fair value of debt derivatives		(3,648)	_	(3,648)	100.0 %
Loss on debt extinguishment		(253)	_	(253)	100.0 %
Other income, net		965	993	(28)	(2.8)%
Income (loss) before provision for income taxes		(76,795)	(12,587)	(64,208)	510.1 %
Provision for income taxes			` <u> </u>	` <u> </u>	<u> </u>
Net income (loss)	\$	(76,795)	\$ (12,587)	\$ (64,208)	510.1 %

#### Revenue

The following table presents the revenue disaggregated by products and service type, as well as the percentage of total revenue.

			Nine Months En	ded S	September 30	0,		
		20	023		20	)22	Change	%
	_		(	(In th	nousands, exce	ept for percentages	s)	
3D Printer sales	\$	69,193	91.3 %	\$	44,336	87.0 %	\$ 24,857	56.1 %
Recurring payment		1,141	1.5 %		3,042	6.0 %	(1,901)	(62.5)%
Support services		5,422	7.2 %		3,599	7.0 %	1,823	50.7 %
Total Revenue	\$	75,756	100.0 %	\$	50,977	100.0 %	\$ 24,779	48.6 %

Total revenue for the nine months ended September 30, 2023 and 2022 was \$75.8 million and \$51.0 million, respectively, an increase of \$24.8 million, or 48.6%.

3D Printer sales were \$69.2 million and \$44.3 million for the nine months ended September 30, 2023 and 2022, respectively, an increase of \$24.9 million, or 56.1%. The improvement in revenue was due to more systems sold, a mix of higher production volumes, and change in our product mix toward more Sapphire XC systems compared to Sapphire systems, and fewer launch customer and early reservation shipments at discounted prices resulting in an increase in the average selling price. The 3D Printer sales included parts and consumables revenue.

Recurring Payment, structured as an operating lease, was \$1.1 million and \$3.0 million for the nine months ended September 30, 2023 and 2022, respectively. The decrease was primarily attributable to a decrease in the number of 3D Printer systems in service generating Recurring Payment revenue as of September 30, 2023, compared to the number of 3D Printer systems in service as of September 30, 2022. In addition, one system that was originally executed through a Recurring Payment transaction was subsequently purchased through a 3D Printer sale transaction, and two systems executed through a Recurring Payment transaction were returned in the nine months ended September 30, 2023.

Our Support Service revenue was \$5.4 million and \$3.6 million for the nine months ended September 30, 2023 and 2022, respectively. The increase was primarily attributable to a significant increase in the number of 3D Printer systems in service as of September 30, 2023, compared to the number of 3D Printers in service as of September 30, 2022.

## Cost of Revenue

The following table presents the Cost of Revenue disaggregated by product and service type, as well as the percentage of total cost of revenue.

	Nine Months Ended September 30,							
	2	023		20	022	(	Change	%
		(	In t	housands, exc	ept for percentages	5)		
Cost of Revenue								
Cost of 3D Printers	\$ 61,975	90.7 %	\$	42,686	85.6 %	\$	19,289	45.2 %
Cost of Recurring Payment	893	1.3 %		2,059	4.1 %		(1,166)	(56.6)%
Cost of Support Services	5,495	8.0 %		5,106	10.3 %		389	7.6 %
Total Cost of Revenue	\$ 68,363	100.0 %	\$	49,851	100.0 %	\$	18,512	37.1 %

Total cost of revenue for the nine months ended September 30, 2023 and 2022 was \$68.4 million and \$49.9 million, respectively, an increase of \$18.5 million, or 37.1%.

Cost of 3D Printers was \$62.0 million and \$42.7 million for the nine months ended September 30, 2023 and 2022, respectively. The increase of \$19.3 million was due to an increase in the number of 3D Printers sold in the nine months ended September 30, 2023, compared to the number of 3D Printers sold in the nine months ended September 30, 2022. In addition, the change in our product mix towards more Sapphire XC systems, which have a higher total per system cost compared to the Sapphire, has increased the cost of 3D Printers for the nine months ended September 30, 2023 compared to 2022.

Cost of Recurring Payment was \$0.9 million and \$2.1 million for the nine months ended September 30, 2023 and 2022, respectively. This decrease of \$1.2 million was due to a decrease in the number of 3D Printer systems in service incurring costs related to Recurring Payment revenue as of September 30, 2023, compared to September 30, 2022.

Cost of Support Services was \$5.5 million and \$5.1 million for the nine months ended September 30, 2023 and 2022, respectively. This increase of \$0.4 million in Cost of Support Services was primarily attributable to the costs for preventative maintenance, costs incurred to enhance system reliability performance, and field service engineering labor costs due to more 3D Printers in service as of September 30, 2023, compared to September 30, 2022. However, field service engineering support costs related to the ramp of the Sapphire XC, Sapphire 1MZ and Sapphire XC 1MZ systems in the field, have been offset by the benefits of preventative maintenance we performed and enhancements we made to system reliability performance.

Cost of revenue as a percentage of revenue was 90.2% and 97.8% for the nine months ended September 30, 2023 and 2022, respectively. The decrease in the cost of revenue as a percentage was primarily due to a per system reduction in material costs for our Sapphire family of systems and improved manufacturing efficiency offset by the change in our product mix towards more Sapphire XC systems compared to Sapphire systems.

## Gross Profit and Gross Margin

Total gross profit was \$7.4 million and \$1.1 million for the nine months ended September 30, 2023 and 2022, respectively. As a percentage of revenue, the gross margin was 9.8% and 2.2% for the nine months ended September 30, 2023 and 2022, respectively. The higher gross profit for the nine months ended September 30, 2023 was primarily attributable to higher average selling prices for our Sapphire family of systems, and improved manufacturing efficiencies, during the nine months ended September 30, 2023, as compared to the nine months ended September 30, 2022.

## Research and Development Expenses

Research and development expenses were \$32.8 million and \$38.4 million for the nine months ended September 30, 2023 and 2022, respectively, a decrease of \$5.6 million. The decrease in research and development expenses in 2023 was driven by a \$5.8 million decrease in purchased materials, and a \$1.5 million decrease in other operating expenses, offset by an increase of \$0.1 million for additional headcount, salaries and employee-related expenses, and a \$1.6 million increase in stock-based compensation.

## Selling and Marketing Expenses

Selling and marketing expenses were \$18.1 million and \$17.9 million for the nine months ended September 30, 2023 and 2022, respectively, an increase of \$0.2 million. The increase was attributable to a \$1.3 million increase in stock-based compensation, a \$0.6 million increase in marketing initiatives, trade shows and branding expenses, and a \$0.4 million increase in additional headcount, salaries and employee-related expenses, offset by a \$2.1 million decrease in European marketing costs.

## General and Administrative Expenses

General and administrative expenses were \$31.6 million and \$27.2 million for the nine months ended September 30, 2023 and 2022, respectively, an increase of \$4.4 million. The increase was attributable to a \$1.3

million increase for additional headcount, salaries and employee-related benefits, a \$0.2 million increase in facilities expenses, a \$1.5 million increase in stock-based compensation, and a \$1.4 million increase in public company-related expenses in advisory, legal and accounting fees and insurance.

Interest Expense

Interest expense was \$1.7 million and \$0.4 million for the nine months ended September 30, 2023 and 2022, respectively.

Gain (loss) on Fair Value of Warrants

The change in fair value of warrants resulted in a loss of \$0.1 million and a gain of \$11.0 million for the nine months ended September 30, 2023 and 2022, respectively, and was related to the non-cash fair value change of the warrant liabilities driven by the change in our stock price.

Gain (loss) on Fair value of Contingent Earnout Liabilities

The change in fair value of the contingent earnout liability was a gain of \$3.0 million and \$58.1 million for the nine months ended September 30, 2023 and 2022, respectively, and was related to the non-cash fair value change of the contingent earnout liabilities driven by the change in our stock price.

Loss on Fair value of Debt Derivatives

The change in fair value of the debt derivatives was a loss of \$3.6 million for the nine months ended September 30, 2023, and was related to the non-cash fair value change of the debt derivatives driven by the change in our stock price.

Loss on Debt Extinguishment

The loss on debt extinguishment was \$0.3 million for the nine months ended September 30, 2023, and was related to the termination of the Loan Agreement with Silicon Valley Bank.

Other Income, Net

Other income, net was \$1.0 million and \$1.0 million for the nine months ended September 30, 2023 and 2022, respectively.

Income Taxes

No provision for federal and state income taxes was recorded for both the nine months ended September 30, 2023 and 2022 due to projected losses, and we maintained a full valuation allowance on the deferred tax assets as of September 30, 2023 and December 31, 2022.

## Liquidity and Capital Resources

As of September 30, 2023, we had raised net proceeds of \$447.9 million, comprised of approximately \$66 million from the offering of \$70.0 million aggregate principal amount of Notes (the "Initial Notes"), which closed on August 10, 2023, \$18.4 million from the ATM Offering (described below), \$278.3 million from the Merger and related private placement of shares of our common stock (the "PIPE Financing"), which closed on September 29, 2021, and \$150.0 million from the issuance of redeemable convertible preferred stock (series A to series D), third-party financing and convertible notes. We have incurred net losses of \$76.8 million and \$12.6 million for the nine months ended September 30, 2023 and 2022, respectively. As of September 30, 2023 and December 31, 2022, we had \$71.6 million and \$80.2 million in cash, cash equivalents and short-term investments, respectively, and an accumulated deficit of \$296.6 million and \$219.8 million, respectively. Our business requires substantial amounts

of cash for operating activities, including salaries and wages paid to our employees, component and sub-assembly purchases, general and administrative expenses, and others.

Our purchase commitments per terms and conditions with suppliers and vendors are cancellable in whole or in part prior to shipment. Non-cancellable purchase commitments (purchase orders) of \$30.8 million for parts and assemblies are due upon receipt and will primarily be delivered in the remainder of 2023 and first half of 2024. If inventory is shipped, we will accrue a liability under accrued expenses. We have no other commitment and contingencies, except for the operating leases. See Note 8, *Leases*, in the unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report for further discussion.

As further described in Note 9, Long-Term Debt in the unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report, the Notes require us to maintain minimum levels of quarterly revenue through the quarter ended June 30, 2026. As of the filing date of this Quarterly Report, we were not in compliance with the minimum revenue covenant for the quarter ended September 30, 2023 and, as a result, the holders of the Notes have the right to declare the Notes due and payable in cash in an amount equal to the Event of Default Acceleration Amount (as defined in the Notes). In addition, as discussed above, we have experienced less revenue growth than expected due to the impact of delayed shipments during the nine months ended September 30, 2023. Further, due to the impact of fourth quarter customer order delays and our bookings to date, we expect additional contraction of revenue growth in the near term.

We have been negotiating a proposed amendment to the Notes with the holders thereof, which we anticipate completing in November 2023; however, we may not be able to obtain a waiver or an amendment on favorable terms or at all. If a waiver or amendment is not agreed to by the holders of the Notes, we do not have sufficient capital to satisfy the outstanding principal and interest due, and the holders may foreclose on our assets securing the Notes. Accordingly, as the Company was not in compliance with the minimum quarterly revenue covenant and the Notes are now callable by the Investor, we believe that there is substantial doubt about our ability to continue as a going concern for the twelve-month period following the filing date of this Quarterly Report.

Even if we are able to agree an amendment with the holders of the Notes, we expect that we will need to engage in additional financings to fund our operations in the near-term as well as to respond to business challenges and opportunities, including the need to repay the Notes, provide working capital, develop new features or enhance our products, expand our manufacturing capacity, improve our operating infrastructure, or acquire complementary businesses and technologies. Accordingly, subject to our compliance with the covenants in the Notes, we expect we will need to engage in equity or debt financings to secure additional funds, including seeking additional capital from public or private offerings of our equity or debt securities, electing to repay, restructure or refinance our existing indebtedness, or electing to borrow additional amounts under new credit lines or from other sources. We may also seek to raise additional capital, including from offerings of our equity or debt securities, on an opportunistic basis when we believe there are suitable opportunities for doing so. For example, on November 14, 2022, we filed with the SEC a shelf registration statement (the "Shelf Registration Statement") that was subsequently declared effective on November 21, 2022 and permits us to sell from time-to-time additional shares of our common stock or other securities in one or more offerings in amounts, at prices and on the terms that we will determine at the time of offering for aggregate gross sale proceeds of up to \$300.0 million, of which we may offer and sell up to \$40.0 million shares of our common stock from time to time pursuant to an "at-the-market" offering (the "ATM Offering") sales agreement (the "ATM Sales Agreement") we entered into in January 2023 with Needham & Company, LLC ("Needham"), as agent, subject to the terms and conditions described in the ATM Sales Agreement and SEC rules and regulations. Furthermore, the Notes require us, on or before December 31, 2023, to establish a new "at-the-market" offer

However, our recent and projected financial results, our existing event of default under the Notes and the related conditions that raise substantial doubt about our ability to continue as a going concern, and general concerns among potential investors and creditors about our financial well-being, may make taking such actions on commercially reasonable terms especially difficult.

Our ability to meet our cash requirements depends on, among other things, our operating performance, competitive and industry developments, and financial market conditions, all of which are significantly affected by business, financial, economic, political, and other factors, many of which we may not be able to control or influence. To the extent that our actual operating results or other developments differ from our expectations, our liquidity could be adversely affected.

As described above, we are undertaking expense reduction and cash savings initiatives as part of a company-wide restructuring and strategic realignment plan to help conserve working capital. The expense reduction and cash saving initiatives include streamlining facilities, managing working capital, reducing capital expenditures, and reducing overall selling, general and administrative expenses.

#### Debt Facilities

As of September 30, 2023, our debt arrangements comprised the Initial Notes issued to certain affiliated institutional investors pursuant to a securities purchase agreement dated August 10, 2023 (the "Securities Purchase Agreement") and our indenture dated as of August 14, 2023 (the "Base Indenture") between us and U.S. Bank Trust Company, National Association, a national banking association, as trustee (the "Trustee"), as supplemented by a first supplemental indenture dated as of August 14, 2023, between us and the Trustee (the "Supplemental Indenture" and, together with the Base Indenture, the "Indenture").

As described above, on August 14, 2023, we used approximately \$22 million of the net proceeds from the offering of the Initial Notes to repay all outstanding obligations under our third amended and restated loan and security agreement, as amended, with SVB. In connection with the repayment of the debt, our third amended and restated loan and security agreement, as amended, was terminated and is no longer available to us. Following the repayment and termination, as of November 14, 2023, we have \$70.0 million of Notes outstanding. We do not hedge our exposure to changes in interest rates. A 10% change in interest rates would not have a material impact on annualized interest expense.

For more information, see Note 9, Long-Term Debt, in the notes to the condensed consolidated financial statements in this Quarterly Report.

#### **Cash Flow Summary**

The following table summarizes our cash flows for the nine months ended September 30, 2023 and 2022:

	 Nine Months End September 30,	ed	
	 2023	2022	Change
	(In thousands)		
Net cash (used in) operating activities	\$ (81,109) \$	(92,154) \$	11,045
Net cash (used in) provided by investing activities	\$ 31,055 \$	(74,507) \$	105,562
Net cash (used in) provided by financing activities	\$ 76,213 \$	(537) \$	76,750

# Operating Activities

Net cash used in operating activities for the nine months ended September 30, 2023 was \$81.1 million, consisting primarily of a net loss of \$76.8 million, non-cash loss of \$25.0 million described below, and a decrease in net operating assets of \$29.4 million. The decrease in net operating assets was comprised of a decrease from accounts payable of \$1.6 million, a decrease from contract liabilities of \$10.3 million, a decrease from other net operating assets of \$9.4 million, a decrease from accounts receivable of \$3.4 million due to timing of customer payments, a decrease in accrued expenses and other current liabilities of \$4.4 million, and a decrease from inventories of \$3.7 million for Sapphire XC, Sapphire 1MZ and Sapphire XC 1MZ system production, offset by an increase from prepaid expenses of \$3.5 million related to insurance and vendor prepayments. The noncash loss of

\$25.0 million consisted of the loss on fair value of debt derivatives of \$3.6 million, the loss on debt extinguishment of \$0.3 million, the loss on fair value of warrants of \$0.1 million, depreciation and amortization of \$4.5 million and stock-based compensation expense of \$19.5 million, partially offset by the gain on fair value of contingent earnout liabilities of \$3.0 million.

Net cash used in operating activities for the nine months ended September 30, 2022 was \$92.2 million, consisting primarily of a net loss of \$12.6 million, noncash income of \$50.7 million described below, and a decrease in net operating assets of \$28.8 million. The decrease in net operating assets was primarily comprised of a decrease from inventories of \$41.8 million for Sapphire and Sapphire XC system production, a decrease from other net operating assets of \$0.2 million, and a decrease in accounts receivable of \$8.1 million due to timing of customer payments, partially offset by an increase of prepaid expenses of \$7.3 million related to insurance and vendor prepayments, an increase in accrued expenses and other current liabilities of \$10.1 million, and an increase from contract liabilities of \$3.8 million. The noncash income of \$50.7 million primarily consisted of the change in fair value of warrants of \$11.0 million and the gain in fair value of contingent earnout liabilities of \$58.1 million, partially offset by depreciation and amortization and stock-based compensation expense.

We expect our cash used in operating activities to decrease, driven by working capital requirements stabilizing related to efforts for expense reduction and cash savings initiatives as part of the company wide restructuring and strategic realignment plan for the remainder of 2023, which may be offset by one-time severance payments and other benefits and one-time cost related to our facilities consolidation.

## Investing Activities

Net cash provided by investing activities during the nine months ended September 30, 2023 was \$31.1 million, consisting of property and equipment purchases of \$1.1 million and production of equipment for lease to customers of \$3.0 million, offset by maturities of available-for-sale investment securities of \$35.1 million.

Net cash used in investing activities during the nine months ended September 30, 2022 was \$74.5 million, consisting of property and equipment purchases of \$12.2 million, production of equipment for lease to customers of \$4.2 million and purchases of available-for-sale investments of \$87.7 million, partially offset by maturities of available-for-sale investment securities of \$29.6 million.

We expect our capital expenditures, to decrease in the remainder of 2023 as we intend to limit the number of 3D Printer systems as equipment for lease to customers and because we completed the construction of our manufacturing facility in 2022.

#### Financing Activities

Net cash provided by financing activities during the nine months ended September 30, 2023 was \$76.2 million, consisting of proceeds of \$65.7 million from the issuance of the Initial Notes, net of issuance costs, proceeds of \$18.4 million from the issuance of common stock, net of issuance costs, pursuant to the ATM Offering, proceeds of \$14.0 million drawn from the revolver facility, proceeds of \$1.6 million drawn from the secured equipment loan facility and proceeds of \$0.4 million from the issuance of common stock upon exercise of stock options, partially offset by \$17.0 million in repayment of the revolver facility and \$7.0 million in repayment loans.

Net cash used in financing activities during the nine months ended September 30, 2022 was \$0.5 million, consisting of repayments of \$8.1 million for loan refinance, partially offset by \$6.7 million of proceeds from the loan refinance and \$1.2 million of proceeds from the issuance of common stock upon exercise of stock options.

We expect cash provided by financing activities to increase by issuing new equity or incurring new debt to continue operations, subject to our compliance with the covenants in the Notes. Our future cash requirements and the adequacy of available funds will depend on many factors, including our operating performance, competitive and industry developments, and financial market conditions.

#### **Off-Balance Sheet Arrangements**

As of September 30, 2023 and December 31, 2022, we did not have any off-balance sheet arrangements.

#### **Recent Accounting Pronouncements**

For a description of recent accounting pronouncements, including the expected dates of adoption and estimated effects, if any, on Velo3D's condensed consolidated financial statements, see Note 2, Summary of Significant Accounting Policies, in the notes to the condensed consolidated financial statements in this Quarterly Report.

#### Implications of Being an Emerging Growth Company

Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies are required to comply with the new or revised financial accounting standards. The JOBS Act provides that a company can choose not to take advantage of the extended transition period and comply with the requirements that apply to non-emerging growth companies, and any such election to not take advantage of the extended transition period is irrevocable. We are an "emerging growth company" as defined in Section 2(A) of the Securities Act and have elected to take advantage of the benefits of this extended transition period.

We will elect to use this extended transition period for complying with new or revised accounting standards that have different effective dates for public business entities and nonpublic business entities until the earlier of the date we (a) are no longer an emerging growth company or (b) affirmatively and irrevocably opt out of the extended transition period provided in the JOBS Act. This may make it difficult or impossible to compare our financial results with the financial results of another public company that is either not an emerging growth company or an emerging growth company that has chosen not to take advantage of the extended transition period exemptions because of the potential differences in accounting standards used. Please refer to Note 2. Summary of Significant Accounting Policies, of the condensed consolidated financial statements included elsewhere in this Quarterly Report for the recent accounting pronouncements adopted and the recent accounting pronouncements not yet adopted for the nine months ended September 30, 2023 and 2022.

We will remain an emerging growth company under the JOBS Act until the earliest of (a) December 31, 2025, (b) the last date of our fiscal year in which we have total annual gross revenue of at least \$1.235 billion, (c) the last date of our fiscal year in which we are deemed to be a "large accelerated filer" under the rules of the SEC or (d) the date on which we have issued more than \$1.0 billion in nonconvertible debt securities during the previous three years.

#### Implications of Being a Smaller Reporting Company

We are a "smaller reporting company" as defined in Item 10(f)(1) of Regulation S-K. Smaller reporting companies may take advantage of certain reduced disclosure obligations, including, among other things, providing only two years of audited consolidated financial statements.

We may continue to be a smaller reporting company even after we are no longer an emerging growth company. We will remain a smaller reporting company and may take advantage of certain scaled disclosures available to smaller reporting companies until the last day of the fiscal year in which (a) the market value of our voting and nonvoting common stock held by non-affiliates equals or exceeds \$250.0 million measured on the last business day of that year's second fiscal quarter and (b) our annual revenue equals or exceeds \$100.0 million during the most recently completed fiscal year or our voting and nonvoting common stock held by non-affiliates equals or exceeds \$700.0 million measured on the last business day of that year's second fiscal quarter.

# **Critical Accounting Policies and Significant Estimates**

Our discussion and analysis of our financial condition and results of operations are based upon our condensed consolidated financial statements, which have been prepared in accordance with U.S. GAAP. We evaluated the

development and selection of our critical accounting policies and estimates and believe that the following involve a higher degree of judgement or complexity and are most significant to reporting our results of operations and financial position and are therefore discussed as critical. The following critical accounting policies reflect the significant estimates and judgements used in the preparation of our condensed consolidated financial statements. Actual results could differ materially from those estimates and assumptions, and those differences could be material to our condensed consolidated financial statements. We re-evaluate our estimates on an ongoing basis. For more information, see Note 2, Summary of Significant Accounting Policies, included in the notes to the condensed consolidated financial statements in this Quarterly Report, and Critical Accounting Policies and Significant Estimates in Part II, Item 7 "Management's Discussion and Analysis of Financial Condition and Results of Operations" of our Annual Report on Form 10-K for the year ended December 31, 2022.

#### Item 3. Quantitative and Qualitative Disclosures about Market Risk

We are a smaller reporting company as defined in Rule 12b-2 under the Exchange Act. As a result, pursuant to Item 305(e) of Regulation S-K, we are not required to provide the information required by this Item.

#### Item 4. Controls and Procedures

## Evaluation of Disclosure Controls and Procedures

Disclosure controls and procedures are controls and other procedures that are designed to ensure that information required to be disclosed by us in reports filed or submitted under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed by an issuer in the reports filed or submitted under the Exchange Act is accumulated and communicated to management, including our Chief Executive Officer and Chief Financial Officer, to allow timely decisions regarding required disclosure.

Our management, with participation of our Chief Executive Officer and Chief Financial Officer, evaluated the effectiveness of our disclosure controls and procedures as defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act as of September 30, 2023. Based upon this evaluation our Chief Executive Officer and Chief Financial Officer concluded that, as of September 30, 2023, our disclosure controls and procedures were not effective due to the material weaknesses in internal control over financial reporting described below.

#### Material Weaknesses in Internal Control over Financial Reporting

As described Part II, Item 9A. "Controls and Procedures" of our Annual Report on Form 10-K for the year ended December 31, 2022, weidentified material weaknesses in our internal control over financial reporting. These material weaknesses have not been remediated as of September 30, 2023. A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting such that there is a reasonable possibility that a material misstatement of the annual or interim consolidated financial statements will not be prevented or detected on a timely basis. The material weaknesses are as follows:

- We did not design and maintain an effective control environment commensurate with our financial reporting requirements. Specifically, we did not maintain a sufficient complement of personnel with an appropriate degree of internal controls and accounting knowledge, experience, and training commensurate with our accounting and financial reporting requirements. Additionally, the lack of a sufficient complement of personnel resulted in an inability to consistently establish appropriate authorities and responsibilities in pursuit of our financial reporting objectives, as demonstrated by, among other things, insufficient segregation of duties in our finance and accounting functions. This material weakness contributed to the following additional material weaknesses.
- We did not design and maintain effective controls over the segregation of duties related to journal entries and account reconciliations. Specifically, certain personnel have the ability to both (i) create and post journal entries within our general ledger system and (ii) prepare and review account reconciliations.
- We did not design and maintain effective controls over inventory. Specifically, we did not design and maintain effective controls over the existence, accuracy and completeness of inventory, the completeness and presentation of accrued inventory purchases, the completeness and accuracy of related expenses.
- · We did not design and maintain effective controls over the presentation of contract assets and liabilities.

These material weaknesses resulted in adjustments to accounts receivable, inventory, other current assets, current and non-current contract liabilities, accrued expenses and other current liabilities, which were recorded prior to the issuance of the consolidated financial statements as of and for the years ended December 31, 2019, 2020 and

2021 and as of and for the interim periods ended September 30, 2021 and December 31, 2021; adjustments to inventory, cost of revenue, research and development expense, contract assets and contract liabilities as of and for the year ended December 31, 2022; and adjustments to current and non-current contract liabilities as of and for the interim period ended June 30, 2022. Additionally, these material weaknesses could result in a misstatement of substantially all of our accounts or disclosures that would result in a material misstatement to the annual or interim consolidated financial statements that would not be prevented or detected.

- We did not design and maintain effective controls over certain information technology ('T'') general controls for information systems that are relevant to the preparation of our consolidated financial statements. Specifically, we did not design and maintain effective:
  - user access controls to ensure appropriate segregation of duties and that adequately restrict user and privileged access to financial applications, programs, and data to appropriate company personnel; and
  - program change management controls to ensure that information technology program and data changes affecting certain financial IT applications and underlying accounting records are identified, tested, authorized and implemented appropriately.

These IT deficiencies did not result in a misstatement to the consolidated financial statements, however, the deficiencies, when aggregated, could impact maintaining effective segregation of duties, as well as the effectiveness of IT dependent controls (such as automated controls that address the risk of material misstatement to one or more assertions, along with the IT controls and underlying data that support the effectiveness of system-generated data and reports) that could result in misstatements potentially impacting all consolidated financial statement accounts and disclosures that would not be prevented or detected. Accordingly, management has determined these deficiencies in the aggregate constitute a material weakness.

#### Remediation Measures for Remaining Material Weaknesses in Internal Control over Financial Reporting

We have taken measures to remediate the material weaknesses remaining as of September 30, 2023, including the following: hired additional accounting and IT personnel to bolster our reporting, technical accounting and IT capabilities; provided ongoing training for our personnel on accounting, financial reporting and internal control over financial reporting; engaged a third-party to assist in designing and implementing controls, including controls related to segregation of duties and IT general controls; designing and implementing controls to formalize roles and review responsibilities to align with our team's skills and experience and designing and implementing controls over segregation of duties; designing and implementing controls over the preparation and review of journal entries and account reconciliations, and designing and implementing controls over the existence, accuracy and completeness of inventory, the completeness and presentation of accrued inventory purchases, the completeness and accuracy of related expenses, and controls over the presentation of contract assets and liabilities; and designing and implementing IT general controls, including controls over the review and update of user access rights and privileges and program change management controls.

We are making progress toward the effectiveness of our internal control over financial reporting and disclosure controls and procedures. The measures that we are taking are subject to continued testing, ongoing senior management review, as well as audit committee oversight. We will not be able to conclude whether the measures we are taking will fully remediate these material weaknesses in our internal control over financial reporting until we have completed our remediation efforts and subsequent evaluation of their effectiveness. We may also conclude that additional measures may be required to remediate the material weaknesses in our internal control over financial reporting, which may necessitate additional implementation and evaluation time. We will continue to assess the effectiveness of our internal control over financial reporting and take steps to remediate the known material weaknesses expeditiously.

Changes	in	Internal	Control	over	Financial	Reporting
Chunges	u	micinu	Common	UVEI	1 munciui	Reporting

There were no changes in internal control over financial reporting during the quarter ended September 30, 2023 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

# PART II. OTHER INFORMATION

# Item 1. Legal Proceedings

From time to time, we may be involved in various disputes and litigation matters that arise in the ordinary course of business. We are currently not a party to any material legal proceedings.

#### Item 1A. RISK FACTORS

Investing in our securities involves risks. You should consider carefully the risks and uncertainties described in our Annual Report on Form 10-K for the year ended December 31, 2022 in the section titled Part I, Item 1A. "Risk Factors, together with all of the other information in this Quarterly Report, including the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our condensed consolidated financial statements and related notes included elsewhere in this Quarterly Report, before deciding whether to purchase any of our securities. Our business, results of operations, financial condition, and prospects could also be harmed by risks and uncertainties that are not presently known to us or that we currently believe are not material. If any of these risks actually occur, our business, results of operations, financial condition, and prospects could be materially and adversely affected

Except for as set forth below, there have been no material changes in the risk factors described in Part I, Item 1A. "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2022.

#### Risks Related to Our Financial Position and Need for Additional Capital

# Our failure to satisfy certain financial covenants under the Notes raises substantial doubt about our ability to continue as a going concern.

As described Note 9, *Long-Term Debt*, in the unaudited condensed consolidated financial statements included elsewhere in this Quarterly Report, the Notes require us to maintain minimum levels of quarterly revenue through the quarter ended June 30, 2026. As of the filing date of this Quarterly Report, we were not in compliance with the minimum revenue covenant for the quarter ended September 30, 2023 and, as a result, the holders of the Notes have the right to declare the Notes due and payable in cash in an amount equal to the Event of Default Acceleration Amount (as defined in the Notes). In addition, we have experienced less revenue growth than expected due to the impact of delayed shipments during the nine months ended September 30, 2023. Further, due to the impact of fourth quarter customer order delays and our bookings to date, we expect additional contraction of revenue growth in the near term.

We have been negotiating a proposed amendment to the Notes with the holders thereof, which we anticipate completing in November 2023; however, we may not be able to obtain a waiver or an amendment on favorable terms or at all. If a waiver or amendment is not agreed to by the holders of the Notes, we do not have sufficient capital to satisfy the outstanding principal and interest due, and the holders may foreclose upon the assets securing the Notes. Accordingly, as the Company was not in compliance with the minimum quarterly revenue covenant and the Notes are now callable by the Investor, we believe that there is substantial doubt about our ability to continue as a going concern for the twelve-month period following the filing date of this Quarterly Report.

Our conclusion that there is substantial doubt about our ability to continue as a going concern may be viewed unfavorably by current and prospective investors, as well as by analysts and creditors. As a result, this conclusion may make it more difficult for us to raise the additional financing necessary to continue to operate our business. In addition, this conclusion may make it more difficult for us to sell our products and meet our sales forecasts, which may further impede our ability to raise additional financing. If we become unable to continue as a going concern, we may find it necessary to file a petition for reorganization under Title 11 of the U.S. Code in order to provide us additional time to identify an appropriate solution to our financial situation and implement a plan of reorganization aimed at improving our capital structure.

## We expect to require additional capital to fund our operations in the near-term, and this capital might not be available on acceptable terms, if at all.

Even if we are able to agree an amendment with the holders of the Notes, we expect that we will need to engage in additional financings to fund our operations in the near-term as well as to respond to business challenges and opportunities, including the need to repay the Notes, provide working capital, develop new features or enhance our products, expand our manufacturing capacity, improve our operating infrastructure or acquire complementary businesses and technologies. Accordingly, subject to our compliance with the covenants in the Notes, we expect we

will need to engage in equity or debt financings to secure additional funds, including seeking additional capital from public or private offerings of our equity or debt securities, electing to repay, restructure or refinance our existing indebtedness, or electing to borrow additional amounts under new credit lines or from other sources. We may also seek to raise additional capital, including from offerings of our equity or debt securities on an opportunistic basis when we believe there are suitable opportunities. For example, on November 14, 2022, we filed with the SEC a shelf registration statement (the "Shelf Registration Statement") that was subsequently declared effective on November 21, 2022 and permits us to sell from time to time additional shares of our common stock or other securities in one or more offerings in amounts, at prices and on the terms that we will determine at the time of offering for aggregate gross sale proceeds of up to \$300.0 million, of which we may offer and sell up to \$40.0 million shares of our common stock from time to time pursuant to an "at-the-market" offering sales agreement (the "ATM Sales Agreement") we entered into in February 2023 with Needham, as agent, subject to the terms and conditions described in the ATM Sales Agreement and SEC rules and regulations. Furthermore, the Notes require us, on or before December 31, 2023, to establish a new "at-the-market" offering program with aggregate available, accessible and unused capacity to generate gross proceeds to us of at least \$75,000,000 as of December 31, 2023. However, our recent and projected financial results, our existing event of default under the Notes and the related conditions that raise substantial doubt about our ability to continue as a going concern, and general concerns among potential investors and creditors about our financial well-being may make taking such actions on commercially reasonable terms especially difficult.

If we raise additional funds through future issuances of equity or convertible debt securities, our existing stockholders could suffer significant dilution, and any new equity securities we issue could have rights, preferences and privileges superior to those of holders of our common stock. Any debt financing that we may secure in the future could involve restrictive covenants relating to our capital-raising activities and other financial and operational matters, which may make it more difficult for us to obtain additional capital and to pursue business opportunities, including potential acquisitions. We may not be able to obtain additional financing on terms favorable to us, if at all. Our ability to raise additional capital when needed may be adversely affected by external factors beyond our control, including changes in the political climate, geopolitical actions, changes in market interest rates or foreign exchange rates, market volatility in the trading prices for our common stock and other technology companies, a recession, depression, high inflation or other sustained adverse market event, and the COVID-19 pandemic. If we are unable to obtain adequate financing or financing on terms satisfactory to us when we require it, our ability to continue to fund our operations in the near-term as well as to respond to business challenges and opportunities could be significantly impaired, and our business may be adversely affected.

## Risks Related to Our Business and Industry

# Market conditions, economic uncertainty or downturns could adversely affect our business and operating results.

In recent years, the United States and other significant markets have experienced cyclical downturns and worldwide economic conditions remain uncertain, including as a result of the COVID-19 pandemic, supply chain disruptions, the war in Ukraine and the war in Israel, instability in the U.S. and global banking systems, rising fuel prices, increasing interest rates or foreign exchange rates and high inflation and the possibility of a recession. Economic uncertainty and associated macroeconomic conditions make it extremely difficult for our customers and us to accurately forecast and plan future business activities, and could cause our customers to slow spending on our products, which could delay and lengthen sales cycles.

A significant downturn in economic activity, or general spending on additive manufacturing systems, may cause our current or potential customers to react by reducing their capital and operating expenditures in general or by specifically reducing their spending on additive manufacturing systems and related technologies. Moreover, competitors may respond to challenging market conditions by lowering prices and attempting to lure away our customers.

We cannot predict the timing, strength, or duration of any economic slowdown or any subsequent recovery generally, or in any industry. If the conditions in the general economy and the markets in which we operate worsen from present levels, our business, financial condition, and operating results could be adversely affected.

For example, instability in the U.S. banking system has resulted in the failure of several U.S. banks, including the closure of SVB by banking regulators in March 2023. If other banks and financial institutions enter receivership or become insolvent in the future in response to financial conditions affecting the banking system and financial markets, our ability to access our existing cash, cash equivalents and investments or to access financing may be threatened, which could have a material adverse effect on our business and financial condition.

#### Risks Related to Our Notes

The terms of the Notes restrict our current and future operations. Upon an event of default, we may not be able to make any accelerated payments under the Notes or our other permitted indebtedness.

On August 10, 2023, we entered into a securities purchase agreement with certain affiliated institutional investors for the purchase and sale of up to \$105 million in aggregate principal amount of our senior secured convertible notes. We issued \$70 million in aggregate principal amount of the Initial Notes on August 14, 2023 (the "Initial Closing"). In addition, we have granted the investors the right to purchase up to an additional \$35 million in aggregate principal amount of the Notes (the "Additional Notes") so long as the notice to exercise such option is provided no later than the first anniversary of the Initial Closing.

The Notes contain a number of restrictive covenants that impose significant operating and financial restrictions on us and may limit our ability to engage in acts that may be in our long-term best interest. In particular, the Notes contain customary affirmative and negative covenants (including covenants that limit our ability to incur debt, make investments, transfer assets, engage in certain transactions with affiliates and merge with other companies, in each case, other than those permitted by the Notes) and events of default. Furthermore, we are required to maintain a minimum of \$30 million of unrestricted cash and cash equivalents and to maintain minimum levels of quarterly revenue through the quarter ended June 30, 2026. Our ability to meet the financial tests under the Notes can be affected by events beyond our control, and we may be unable to meet them

A breach of the covenants or restrictions under the Notes or under the agreements governing any of our other permitted indebtedness could result in an event of default under the applicable indebtedness. Such a default may allow holders of the Notes or the holders or lenders of our other permitted indebtedness, as appropriate, to accelerate the related indebtedness, which may result in the acceleration of other indebtedness to which a cross-acceleration or cross-default provision applies. In addition, such lenders or holders could terminate commitments to lend money, if any. Furthermore, if we were unable to repay the Notes or other permitted indebtedness, then due and payable, secured lenders could proceed against the assets, if any, securing such indebtedness. In the event such lenders or holders accelerate the repayment of the Notes or our other permitted borrowings, we may not have sufficient assets to repay that indebtedness. A default would also significantly diminish the market price of our common stock.

In particular, as described in more detail above, we were not in compliance with the minimum revenue covenant under the Notes for the quarter ended September 30, 2023, and it is probable that we will not be able to maintain compliance with the minimum levels of quarterly revenue during the twelve-month period following the filing date of this Quarterly Report. As a result, the holders of the Notes have the right to declare the Notes due and payable in cash in an amount equal to the Event of Default Acceleration Amount (as defined in the Notes). We have been negotiating a proposed amendment to the Notes with the holders thereof, which we anticipate completing in November 2023; however, we may not be able to obtain a waiver or an amendment on favorable terms or at all. If a waiver or amendment is not agreed to by the holders of the Notes, we do not have sufficient capital to satisfy the outstanding principal and interest due, and the holders of the Notes may foreclose on our assets securing the Notes.

Furthermore, as a result of these restrictions, we may be limited in how we conduct and grow our business, or unable to compete effectively or to take advantage of new business opportunities. These restrictions may affect our ability to grow in accordance with our strategy.

Servicing the Notes requires a significant amount of cash, and we may not have sufficient cash flow from our business to pay our obligations under the Notes or our other permitted indebtedness.

Even if we are able to agree to an amendment with the holders of the Notes, our ability to make scheduled payments of principal or to pay interest on or to refinance the Notes or our other permitted indebtedness depends on our future performance, which is subject to economic, financial, competitive and other factors, some of which are beyond our control. As of September 30, 2023, we have outstanding \$70.0 million of Initial Notes, and the terms of the Initial Notes require us to pay approximately \$80.5 million to repay the full principal amount of the Initial Notes at maturity or any other time; however, as noted above, if the holders accelerate the Notes as a result of the existing event of default, we would be required to pay the Event of Default Acceleration Amount. Our business may not continue to generate cash flow from operations in the future sufficient to satisfy our obligations under the Notes or our other permitted indebtedness and, in particular, we expect that we will need to engage in additional financings to fund our operations in the near-term, the terms of which may be onerous or highly dilutive. If we are unable to generate such cash flow and obtain such additional financing, we may be required to adopt one or more alternatives, such as reducing or delaying investments or capital expenditures, selling assets, or refinancing or restructuring our indebtedness on terms that may be unfavorable. We may not prepay the Notes without the consent of the holders, and our ability to refinance the Notes or our other permitted indebtedness will also depend on the capital markets and our financial condition at such time. We may not be able to engage in any of these activities or engage in these activities on desirable terms, which could result in a default on the Notes or our other indebtedness.

## Conversion of the Notes may dilute the ownership interest of existing stockholders or may otherwise depress the price of our common stock.

The conversion of the Notes will dilute the ownership interests of existing stockholders to the extent we deliver shares upon conversion of the Notes. Any sales in the public market of the common stock issuable upon such conversion could adversely affect prevailing market prices of our common stock. In addition, the existence of the Notes may encourage short selling by market participants because the conversion of the Notes could be used to satisfy short positions, or anticipated conversion of the Notes into shares of our common stock could depress the price of our common stock.

None.
m 3. Defaults Upon Senior Securities
As described in more detail elsewhere in this Quarterly Report, we were not in compliance with the minimum revenue covenant under the Notes for the quarter ended stember 30, 2023. As a result, the holders of the Notes et the right to declare the Notes due and payable in cash in an amount equal to the Event of Default Acceleration Amount (as defined in the Notes). We have been negotiating roposed amendment to the Notes with the holders thereof, which we anticipate completing in November 2023; however, we may not be able to obtain a waiver or an endment on favorable terms or at all. If a waiver or amendment is not agreed to by the holders of the Notes, we do not have sufficient capital to satisfy the outstanding ancipal and interest due, and the holders of the Notes may foreclose on our assets securing the Notes.
m 4. Mine Safety Disclosures
Not applicable.
m 5. Other Information
None.
64

Item 2. Unregistered Sales of Equity Securities, Use of Proceeds and Issuer Purchases of Equity Securities

Sales of Unregistered Securities

**Issuer Purchases of Equity Securities** 

None.

## Item 6. Exhibits

#### **Exhibit Number** Indenture, dated as of August 14, 2023, by and between the Company and U.S. Bank Trust Company, National Association, as trustee. First Supplemental Indenture, dated as of August 14, 2023, by and between the Company and U.S. Bank Trust Company, National Association, as trustee. 4.3 Form of Note. Securities Purchase Agreement, dated as of August 10, 2023, by and among the Company and High Trail Investors ON LLC and HB SPV I Master Sub LLC, as buyers. 10.1† Placement Agent Agreement, dated as of August 10, 2023, by and between the Company and Credit 10.2 Suisse Securities (USA) LLC, as placement agent. Security Agreement, dated as of August 14, 2023, by and among the Company, Velo3D US, Inc. and 10.3† High Trail Investors ON LLC, as collateral agent. Form of Voting Agreement (included as Exhibit D to the Securities Purchase Agreement filed as Exhibit 10.1). 10.5 Form of Change in Control Agreement 31.1\*\* Certification of the Chief Executive Officer required by Rule 13a-14(a) or rule 15d-14(a) Certification of the Chief Financial Officer required by Rule 13a-14(a) or rule 15d-14(a) 31.2\*\* Certification of the Chief Executive Officer required by Rule 13a-14(b) or Rule 15d-14(b) and 18 U.S.C. 32.1\*\* Certification of the Chief Financial Officer required by Rule 13a-14(b) or Rule 15d-14(b) and 18 U.S.C. 32.2\*\* Inline XBRL Instance Document - the instance document does not appear in the Interactive Data file 101.INS because its XBRL tags are embedded within the Inline XBRL document. 101.SCH Inline XBRL Taxonomy Extension Schema Document 101 CAL Inline XBRL Taxonomy Extension Calculation Linkbase Document 101.DEF Inline XBRL Taxonomy Extension Definition Linkbase Document 101.LAB Inline XBRL Taxonomy Label Linkbase Document

104 Cover Page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101)

\*\* Filed herewith.

101.PRE Inline XBRL Taxonomy Extension Presentation Linkbase Document

<sup>\*</sup> This certification is not deemed "filed" for purposes of Section 18 of the Exchange Act or otherwise subject to the liability of that section, nor shall it be deemed incorporated by reference into any filing under the Securities Act or the Exchange Act.

<sup>†</sup> Portions of this exhibit (indicated with markouts) have been redacted in accordance with Item 601(b)(10)(iv).

# SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized, in the City of Campbell, State of California, on the 20th day of November, 2023.

# VELO3D, INC.

Date: November 20, 2023 By: \_\_\_/s/ Bernard Chung

Name: Bernard Chung

Title: Acting Chief Financial Officer and Principal Financial Officer

# VELO3D, INC.

And

# U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION, as Trustee

# INDENTURE

Dated as of August 14, 2023

# TABLE OF CONTENTS

ARTICLE 1 – DEFINITIONS AND INCORPORATION BY REFERENCE	1
1.1 DEFINITIONS	1
1.2. OTHER DEFINITIONS	3
1.3. INCORPORATION BY REFERENCE OF TRUST INDENTURE ACT	4
1.4. RULES OF CONSTRUCTION	4
ARTICLE 2 – THE SECURITIES	5
2.1. ISSUABLE IN SERIES	5
2.2. ESTABLISHMENT OF TERMS OF SERIES OF SECURITIES	5
2.3. EXECUTION AND AUTHENTICATION	6
2.4. REGISTRAR AND PAYING AGENT	7
2.5. PAYING AGENT TO HOLD ASSETS IN TRUST	7
2.6. SECURITYHOLDER LISTS	8
2.7. TRANSFER AND EXCHANGE	8
2.8. REPLACEMENT SECURITIES	8
2.9. OUTSTANDING SECURITIES	8
2.10. WHEN TREASURY SECURITIES DISREGARDED; DETERMINATION OF HOLDERS' ACTION	9
2.11. TEMPORARY SECURITIES	9
2.12. CANCELLATION	9
2.13. PAYMENT OF INTEREST; DEFAULTED INTEREST; COMPUTATION OF INTEREST	9
2.14. CUSIP NUMBER	10
2.15. PROVISIONS FOR GLOBAL SECURITIES	10
2.16. PERSONS DEEMED OWNERS	10
ARTICLE 3 – REDEMPTION	11
3.1. NOTICES TO TRUSTEE	11
3.2. SELECTION BY TRUSTEE OF SECURITIES TO BE REDEEMED	11
3.3. NOTICE OF REDEMPTION	11
3.4. EFFECT OF NOTICE OF REDEMPTION	12
3.5. DEPOSIT OF REDEMPTION PRICE	12
3.6. SECURITIES REDEEMED IN PART	12
ARTICLE 4 – COVENANTS	12
4.1. PAYMENT OF SECURITIES	12
4.2. SEC REPORTS	12
4.3. WAIVER OF STAY, EXTENSION OR USURY LAWS	13
4.4. COMPLIANCE CERTIFICATE	13
4.5. CORPORATE EXISTENCE	13
ARTICLE 5 – SUCCESSOR CORPORATION	13
5.1. LIMITATION ON CONSOLIDATION, MERGER AND SALE OF ASSETS	13
5.2. SUCCESSOR PERSON SUBSTITUTED	14
ARTICLE 6 – DEFAULTS AND REMEDIES	14
6.1. EVENTS OF DEFAULT	14
6.2. ACCELERATION	15
6.3. REMEDIES	15
6.4. WAIVER OF PAST DEFAULTS AND EVENTS OF DEFAULT	15
6.5. CONTROL BY MAJORITY	15
6.6. LIMITATION ON SUITS	15
6.7. RIGHTS OF HOLDERS TO RECEIVE PAYMENT	16
6.8. COLLECTION SUIT BY TRUSTEE	16
6.9. TRUSTEE MAY FILE PROOFS OF CLAIM	16
6.10. PRIORITIES	16
6.11. UNDERTAKING FOR COSTS	17
ARTICLE 7 – TRUSTEE	17
7.1. DUTIES OF TRUSTEE	17
7.2. RIGHTS OF TRUSTEE	17
7.3. INDIVIDUAL RIGHTS OF TRUSTEE	18

	7.4. TRUSTEE'S DISCLAIMER	18
	7.5. NOTICE OF DEFAULT	18
	7.6. REPORTS BY TRUSTEE TO HOLDERS	19
	7.7. COMPENSATION AND INDEMNITY	19
	7.8. REPLACEMENT OF TRUSTEE	19
	7.9. SUCCESSOR TRUSTEE BY CONSOLIDATION, MERGER OR CONVERSION	20
	7.10. ELIGIBILITY; DISQUALIFICATION	20
	7.11. PREFERENTIAL COLLECTION OF CLAIMS AGAINST COMPANY	20
	7.12. PAYING AGENTS	20
AR	TICLE 8 – AMENDMENTS, SUPPLEMENTS AND WAIVERS	20
	8.1. WITHOUT CONSENT OF HOLDERS	20
	8.2. WITH CONSENT OF HOLDERS	21
	8.3. COMPLIANCE WITH TRUST INDENTURE ACT	22
	8.4. REVOCATION AND EFFECT OF CONSENTS	22
	8.5. NOTATION ON OR EXCHANGE OF SECURITIES	22
	8.6. TRUSTEE TO SIGN AMENDMENTS, ETC.	22
AR	TICLE 9 – DISCHARGE OF INDENTURE; DEFEASANCE	22
	9.1. DISCHARGE OF INDENTURE	22
	9.2. LEGAL DEFEASANCE	23
	9.3. COVENANT DEFEASANCE	23
	9.4. CONDITIONS TO LEGAL DEFEASANCE OR COVENANT DEFEASANCE	23
	9.5. DEPOSITED MONEY AND U.S. AND FOREIGN GOVERNMENT OBLIGATIONS TO BE HELD IN TRUST; OTHER MISCELLANEOUS PROVISIONS	24
	9.6. REINSTATEMENT	24
	9.7. MONEYS HELD BY PAYING AGENT	25
	9.8. MONEYS HELD BY TRUSTEE	25
AR	ΓICLE 10 – MISCELLANEOUS	25
	10.1. TRUST INDENTURE ACT CONTROLS	25
	10.2. NOTICES	25
	10.3. COMMUNICATIONS BY HOLDERS WITH OTHER HOLDERS	26
	10.4. CERTIFICATE AND OPINION AS TO CONDITIONS PRECEDENT	26
	10.5. STATEMENT REQUIRED IN CERTIFICATE AND OPINION	26
	10.6. RULES BY TRUSTEE AND AGENTS	26
	10.7. BUSINESS DAYS; LEGAL HOLIDAYS; PLACE OF PAYMENT	27
	10.8. GOVERNING LAW	27
	10.9. NO ADVERSE INTERPRETATION OF OTHER AGREEMENTS	27
	10.10. NO RECOURSE AGAINST OTHERS	27
	10.11. SUCCESSORS	27
	10.12. MULTIPLE COUNTERPARTS	27
	10.13. TABLE OF CONTENTS, HEADINGS, ETC.	27
	10.14. SEVERABILITY	27
	10.15. SECURITIES IN A FOREIGN CURRENCY OR IN EUROS	27
	10.16. JUDGMENT CURRENCY	28

INDENTURE, dated as of August 14, 2023, by and between Velo3D, Inc., a Delaware corporation, as Issuer (the "Company") and U.S. Bank Trust Company, National Association, a national banking association organized under the laws of United States, as Trustee (the "Trustee").

## RECITALS OF THE COMPANY

The Company has duly authorized the execution and delivery of this Indenture to provide for the issuance from time to time of its debentures, notes or other evidences of indebtedness to be issued in one or more series (the "Securities"), as herein provided, up to such principal amount as may from time to time be authorized in or pursuant to one or more resolutions of the Board of Directors or by supplemental indenture.

All things necessary to make this Indenture a valid agreement of the Company in accordance with its terms have been done, and the execution and delivery thereof have been in all respects duly authorized by the parties hereto.

## NOW, THEREFORE, THIS INDENTURE WITNESSETH:

For and in consideration of the premises and the purchase of the Securities by the Holders thereof, it is mutually agreed, for the equal and proportionate benefit of all Holders of the Securities of a Series thereof, as follows:

# ARTICLE 1 – DEFINITIONS AND INCORPORATION BY REFERENCE

## 1.1 DEFINITIONS.

"Affiliate" of any specified Person means any other Person which, directly or indirectly through one or more intermediaries, controls, or is controlled by or is under common control with, such specified Person. For the purposes of this definition, "control" (including, with correlative meanings, the terms "controlling," "controlled by" and "under common control with"), as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through the ownership of voting securities, by agreement or otherwise.

"Agent" means any Registrar, Paying Agent, co-registrar or agent for service of notices and demands.

"Board of Directors" means the Board of Directors of the Company or any committee duly authorized to act therefor.

"Board Resolution" means a copy of a resolution certified pursuant to an Officers' Certificate to have been duly adopted by the Board of Directors of the Company and to be in full force and effect on the date of such certification which has been delivered to the Trustee.

"Capital Stock" means, with respect to any Person, any and all shares or other equivalents (however designated) of capital stock, partnership interests or any other participation, right or other interest in the nature of an equity interest in such Person or any option, warrant or other security convertible into any of the foregoing.

"Company" means the party named as such in the first paragraph of this Indenture until a successor replaces such party pursuant to Article 5 of this Indenture, and thereafter means the successor and any other primary obligor on the Securities.

"Company Order" means a written order signed in the name of the Company by two Officers, one of whom must be its Chief Executive Officer or its Chief Financial Officer.

"Company Request" means any written request signed in the name of the Company by its Chief Executive Officer, its President, any Vice President, its Chief Financial Officer or its Treasurer and attested to by its Secretary or any Assistant Secretary.

"Corporate Trust Office" means the office of the Trustee at which at any particular time its corporate trust business relating to this Indenture shall be principally administered.

"Default" means any event that is, or that with the passing of time or giving of notice or both would be, an Event of Default.

"Depository" means, with respect to the Securities of any Series issuable or issued in whole or in part in the form of one or more Global Securities, the Person designated as Depository for such Series by the Company, which Depository shall be a clearing agency registered under the Exchange Act, until a successor Depository shall have become such pursuant to the applicable provisions of this Indenture, and thereafter "Depository" shall mean each Person who is then a Depository hereunder, and if at any time there is more than one such Person, such Persons.

"Dollars" means the currency of the United States of America.

"Euro" means the single currency of participating member states of the economic and monetary union as contemplated in the Treaty on European Union.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Foreign Currency" means any currency or currency unit issued by a government other than the government of the United States of America.

"Foreign Government Obligations" means, with respect to Securities that are denominated in a Foreign Currency, (i) direct obligations of the government that issued or caused to be issued such currency for the payment of which obligations its full faith and credit is pledged or (ii) obligations of a Person controlled or supervised by, or acting as an agency or instrumentality of, such government, the timely payment of which is unconditionally guaranteed as a full faith and credit obligation by such government, which, in either case under clauses (i) and (ii), are not callable or redeemable at the option of the issuer thereof.

"GAAP" means generally accepted accounting principles consistently applied as in effect in the United States of America from time to time.

"Global Security" or "Global Securities" means a Security or Securities, as the case may be, in the form established pursuant to Section 2.2, evidencing all or part of a Series of Securities issued to the Depository for such Series or its nominee, and registered in the name of such Depository or nominee, and bearing the legend set forth in Section 2.15(c) (or such other legend(s) as may be applied to such Securities in accordance with Section 2.2(24)).

"Holder" or "Securityholder" means the Person in whose name a Security is registered on the Registrar's books.

"Indebtedness" means (without duplication), with respect to any Person, any indebtedness at any time outstanding, secured or unsecured, contingent or otherwise, which is for borrowed money (whether or not the recourse of the lender is to the whole of the assets of such Person or only to a portion thereof), or evidenced by bonds, notes, debentures or similar instruments, or representing the balance, deferred and unpaid, of the purchase price of any property (excluding any balances that constitute accounts payable or trade payables, and other accrued liabilities arising in the ordinary course of business), if and to the extent any of the foregoing indebtedness would appear as a liability upon a balance sheet of such Person prepared in accordance with GAAP.

"Indenture" means this Indenture as amended, restated or supplemented from time to time.

"Interest Payment Date," when used with respect to any Security, means the Stated Maturity of an installment of interest on such Security.

"Lien" means, with respect to any property or assets of any Person, any mortgage or deed of trust, pledge, hypothecation, assignment, deposit arrangement, security interest, lien, charge, easement, encumbrance, preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever on or with respect to such property or assets (including, without limitation, any capitalized lease obligation, conditional sales or other title retention agreement having substantially the same economic effect as any of the foregoing).

"Maturity," when used with respect to any Security, means the date on which the principal of such Security, or an installment of principal, becomes due and payable as therein or herein provided, whether at the Stated Maturity or by declaration of acceleration, call for redemption, notice of option to elect payment or otherwise.

"Officer" means the Chief Executive Officer, the President, any Vice President, the Chief Financial Officer, the Treasurer or the Secretary of the Company, or any other officer designated by the Board of Directors, as the case may be.

"Officers' Certificate" means, with respect to any Person, a certificate signed by the Chairman, Chief Executive Officer, President, any Vice President or Secretary and the Chief Financial Officer or any Treasurer of such Person, that shall comply with applicable provisions of this Indenture.

"Opinion of Counsel" means a written opinion from legal counsel, which counsel is reasonably acceptable to the Trustee. The counsel may be an employee of or outside counsel to the Company.

"Person" means any individual, corporation, limited liability company, partnership, joint venture, association, joint-stock company, trust, unincorporated organization or government (including any agency or political subdivision thereof).

"Redemption Date," when used with respect to any Security to be redeemed, means the date fixed for such redemption pursuant to this Indenture.

"Responsible Officer," when used with respect to the Trustee, means any officer within the corporate trust department or division of the Trustee (or any successor group of the Trustee) or any other officer of the Trustee customarily performing functions

similar to those performed by any of the above designated officers, and also means, with respect to a particular corporate trust matter relating to this Indenture, any other officer to whom such matter is referred because of his knowledge of and familiarity with the particular subject, and who, in each case, shall have direct responsibility for the administration of this Indenture.

"SEC" means the United States Securities and Exchange Commission as constituted from time to time, or any successor performing substantially the same functions.

"Securities" means the securities that are issued under this Indenture, as amended, restated or supplemented from time to time pursuant to this Indenture.

"Securities Act" means the Securities Act of 1933, as amended.

"Series" or "Series of Securities" means each series of debentures, notes or other debt instruments of the Company created pursuant to Sections 2.1 and 2.2.

"Significant Subsidiary" means (i) any direct or indirect Subsidiary of the Company that would be a "significant subsidiary" as defined in Article 1, Rule 1-02 of Regulation S-X, promulgated pursuant to the Securities Act, as such regulation is in effect on the date hereof, or (ii) any group of direct or indirect Subsidiaries of the Company that, taken together as a group, would be a "significant subsidiary" as defined in Article 1, Rule 1-02 of Regulation S-X, promulgated pursuant to the Securities Act, as such regulation is in effect on the date hereof.

"Stated Maturity," when used with respect to any Security or any installment of principal thereof or interest thereon, means the date specified in such Security as the fixed date on which the principal of such Security and the interest thereon, or such installment of principal or interest, is due and payable, and when used with respect to any other Indebtedness, means the date specified in the instrument governing such Indebtedness as the fixed date on which the principal of such Indebtedness, or any installment of interest thereon, is due and payable.

"Subsidiary" of any specified Person means any corporation, limited liability company, partnership, joint venture, association or other business entity, whether now existing or hereafter organized or acquired, (i) in the case of a corporation, of which more than 50% of the total voting power of the Capital Stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors thereof is held, directly or indirectly, by such Person or any of its Subsidiaries; or (ii) in the case of a partnership, joint venture, association or other business entity, with respect to which such Person or any of the Subsidiaries has the power to direct or cause the direction of the management and policies of such entity by contract or otherwise, or if in accordance with GAAP such entity is consolidated with such Person for financial statement purposes.

"TIA" means the Trust Indenture Act of 1939 (15 U.S. Code Section 77aaa-77bbbb) as in effect on the date of this Indenture (except as provided in Section 8.3).

"Trustee" means the party named as such in this Indenture until a successor replaces it pursuant to this Indenture, and thereafter means the successor, and if at any time there is more than one such Person, "Trustee" as used with respect to the Securities of any Series shall mean the Trustee with respect to Securities of that Series.

"U.S. Government Obligations" means direct non-callable obligations of, or non-callable obligations guaranteed by, the United States of America for the payment of which obligation or guarantee the full faith and credit of the United States of America is pledged.

## 1.2. OTHER DEFINITIONS.

The definitions of the following terms may be found in the sections indicated as follows:

TERM	DEFINED IN SECTION
"Bankruptcy Law"	6.1
"Business Day"	10.7
"Covenant Defeasance"	9.3
"Custodian"	6.1
"Event of Default"	6.1
"Journal"	10.15
"Judgment Currency"	10.16
"Legal Defeasance"	9.2
"Legal Holiday"	10.7
"Market Exchange Rate"	10.15
"Paying Agent"	2.4
"Place of Payment"	10.7
"Registrar"	2.4
"Required Currency"	10.16
"Service Agent"	2.4

## 1.3. INCORPORATION BY REFERENCE OF TRUST INDENTURE ACT.

Whenever this Indenture refers to a provision of the TIA, the portion of such provision required to be incorporated herein in order for this Indenture to be qualified under the TIA is incorporated by reference in and made a part of this Indenture. The following TIA terms used in this Indenture have the following meanings:

"Commission" means the SEC.

"Indenture securities" means the Securities.

"Indenture securityholder" means a Holder or Securityholder.

"Indenture to be qualified" means this Indenture.

"Indenture trustee" or "institutional trustee" means the Trustee.

"Obligor on the indenture securities" means the Company or any other obligor on the Securities.

All other terms used in this Indenture that are defined by the TIA, defined in the TIA by reference to another statute or defined by SEC rule have the meanings therein assigned to them.

# 1.4. RULES OF CONSTRUCTION.

Unless the context otherwise requires:

(1) a term has the meaning assigned to it herein, whether defined expressly or by reference;

- (2) unless otherwise specified herein, all accounting terms used herein shall be interpreted, all accounting determinations hereunder shall be made, and all financial statements required to be delivered hereunder shall be prepared in accordance with GAAP;
  - (3) "or" is not exclusive;
  - (4) words in the singular include the plural, and in the plural include the singular;
  - (5) words used herein implying any gender shall apply to each gender;
- (6) the words "herein", "hereof" and "hereunder" and other words of similar import refer to this Indenture as a whole and not to any particular Article, Section or other subdivision; and
  - (7) "\$," refers to Dollars, or such other money of the United States that at the time of payment is legal tender for payment of public and private debts.

#### ARTICLE 2 – THE SECURITIES

## 2.1. ISSUABLE IN SERIES.

The aggregate principal amount of Securities that may be authenticated and delivered under this Indenture is \$105,000,000. The Securities may be issued in one or more Series. All Securities of a Series shall be identical except as may be set forth in a Board Resolution, a supplemental indenture or an Officers' Certificate detailing the adoption of the terms thereof pursuant to the authority granted under a Board Resolution. In the case of Securities of a Series to be issued from time to time, the Board Resolution, Officers' Certificate or supplemental indenture may provide for the method by which specified terms (such as interest rate, Stated Maturity, record date or date from which interest shall accrue) are to be determined. Securities may differ between Series in respect of any matters, PROVIDED, that all Series of Securities shall be equally and ratably entitled to the benefits of the Indenture.

## 2.2. ESTABLISHMENT OF TERMS OF SERIES OF SECURITIES.

At or prior to the issuance of any Securities within a Series, the following shall be established (as to the Series generally, in the case of Subsection 2.2(1) and either as to such Securities within the Series or as to the Series generally in the case of Subsections 2.2(2) through 2.2(24)) by a Board Resolution, a supplemental indenture or an Officers' Certificate, in each case, pursuant to authority granted under a Board Resolution:

- (1) the title of the Series (which shall distinguish the Securities of that particular Series from the Securities of any other Series);
- (2) any limit upon the aggregate principal amount of the Securities of the Series which may be authenticated and delivered under this Indenture (except for Securities authenticated and delivered upon registration of transfer of, or in exchange for, or in lieu of, other Securities of the Series pursuant to Section 2.7, 2.8, 2.11, 3.6 or 8.5):
  - (3) the price or prices (expressed as a percentage of the principal amount thereof) at which the Securities of the Series will be issued;
  - (4) the date or dates on which the principal of the Securities of the Series is payable;
- (5) the rate or rates (which may be fixed or variable) per annum or, if applicable, the method used to determine such rate or rates (including, but not limited to, any commodity, commodity index, stock exchange index or financial index) at which the Securities of the Series shall bear interest, if any, the date or dates from which such interest, if any, shall accrue, the date or dates on which such interest payable and any regular record date for the interest payable on any Interest Payment Date;
- (6) the place or places where the principal of, and interest and premium, if any, on, the Securities of the Series shall be payable, or the method of such payment, if by wire transfer, mail or other means;
- (7) if applicable, the period or periods within which, the price or prices at which and the terms and conditions upon which the Securities of the Series may be redeemed, in whole or in part, at the option of the Company;
- (8) the obligation, if any, of the Company to redeem or purchase the Securities of the Series pursuant to any sinking fund or analogous provisions or at the option of a Holder thereof, and the period or periods within which, the price or prices at which and the terms and conditions upon which Securities of the Series shall be redeemed or purchased, in whole or in part, pursuant to such obligation;

- (9) the dates, if any, on which and the price or prices at which the Securities of the Series will be repurchased by the Company at the option of the Holders thereof, and other detailed terms and provisions of such repurchase obligations;
  - (10) if other than denominations of \$1,000 and any integral multiple thereof, the denominations in which the Securities of the Series shall be issuable;
- (11) the forms of the Securities of the Series in bearer (if to be issued outside of the United States of America) or fully registered form (and, if in fully registered form, whether the Securities will be issuable as Global Securities);
- (12) if other than the principal amount thereof, the portion of the principal amount of the Securities of the Series that shall be payable upon declaration of acceleration of the Maturity thereof pursuant to Section 6.2;
- (13) the currency of denomination of the Securities of the Series, which may be Dollars or any Foreign Currency, including, but not limited to, the Euro, and, if such currency of denomination is a composite currency other than the Euro, the agency or organization, if any, responsible for overseeing such composite currency;
- (14) the designation of the currency, currencies or currency units in which payment of the principal of, and interest and premium, if any, on, the Securities of the Series will be made;
- (15) if payments of principal of, or interest or premium, if any, on, the Securities of the Series are to be made in one or more currencies or currency units other than that or those in which such Securities are denominated, the manner in which the exchange rate with respect to such payments will be determined;
- (16) the manner in which the amounts of payment of principal of, or interest and premium, if any, on, the Securities of the Series will be determined, if such amounts may be determined by reference to an index based on a currency or currencies or by reference to a commodity, commodity index, stock exchange index or financial index;
  - (17) the provisions, if any, relating to any collateral provided for the Securities of the Series;
  - (18) any addition to or change in the covenants set forth in Articles 4 or 5 that applies to Securities of the Series;
- (19) any addition to or change in the Events of Default which applies to any Securities of the Series, any provision for the payment of additional interest or liquidated damages in connection with any Event of Default, and any change in the right of the Trustee or the requisite Holders of such Securities to declare the principal amount thereof due and payable pursuant to Section 6.2;
- (20) the terms and conditions, if any, for conversion of the Securities into or exchange of the Securities for shares of common stock, preferred stock, other debt securities, or warrants, subscription rights or units for common stock, preferred stock or other securities of any kind of the Company that apply to Securities of the Series;
- (21) any Trustees, depositories, interest rate calculation agents, exchange rate calculation agents or other agents with respect to Securities of such Series if other than those appointed herein;
  - (22) the terms and conditions, if any, upon which the Securities shall be subordinated in right of payment to other Indebtedness of the Company;
  - (23) if applicable, that the Securities of the Series, in whole or any specified part, shall be defeasible pursuant to Article 9; and
- (24) any other terms of the Securities of the Series (which terms shall not be inconsistent with the provisions of this Indenture, except as permitted by Section 8.1, but which may modify or delete any provision of this Indenture insofar as it applies to such Series).

All Securities of any one Series need not be issued at the same time, and may be issued from time to time, consistent with the terms of this Indenture, if so provided by or pursuant to the Board Resolution, supplemental indenture or Officers' Certificate referred to above, however, the authorized principal amount of any Series may not be increased to provide for issuances of additional Securities of such Series, unless otherwise provided in such Board Resolution, supplemental indenture or Officers' Certificate.

## 2.3. EXECUTION AND AUTHENTICATION.

The Securities shall be executed on behalf of the Company by two Officers of the Company or an Officer and an Assistant Secretary of the Company. Each such signature may be either manual, electronic or facsimile. The Company's seal may be impressed, affixed, imprinted or reproduced on the Securities and may be in facsimile

If an Officer whose signature is on a Security no longer holds that office at the time the Security is authenticated, the Security shall nevertheless be valid.

A Security shall not be valid until authenticated by the manual signature of the Trustee or an authenticating agent. The signature shall be conclusive evidence that the Security has been authenticated under this Indenture. The Trustee shall at any time, and from time to time, authenticate Securities for original issue in the principal amount provided in the Board Resolution, supplemental indenture hereto or Officers' Certificate, upon receipt by the Trustee of a Company Order. Such Company Order may authorize authentication and delivery pursuant to oral or electronic instructions from the Company or its duly authorized agent or agents, which oral instructions shall be promptly confirmed in writing. Each Security shall be dated the date of its authentication.

The aggregate principal amount of Securities of any Series outstanding at any time may not exceed any limit upon the maximum principal amount for such Series set forth in the Board Resolution, supplemental indenture hereto or Officers' Certificate delivered pursuant to Section 2.2, except as provided in Section 2.8.

Prior to the issuance of Securities of any Series, the Trustee shall have received and (subject to Section 7.1) shall be fully protected in relying on: (a) the Board Resolution, supplemental indenture hereto or Officers' Certificate establishing the form of the Securities of that Series or of Securities within that Series and the terms of the Securities of that Series or of Securities within that Series, (b) an Officers' Certificate complying with Section 10.4, and (c) an Opinion of Counsel complying with Section 10.4.

The Trustee shall have the right to decline to authenticate and deliver any Securities of any Series: (a) if the Trustee, being advised in writing by outside counsel, determines that such action may not lawfully be taken; or (b) if the Trustee in good faith shall reasonably determine that such action would expose the Trustee to personal liability, or cause it to have a conflict of interest with respect to Holders of any then outstanding Series of Securities.

The Trustee may appoint an authenticating agent acceptable to the Company to authenticate Securities. An authenticating agent may authenticate Securities whenever the Trustee may do so. Any appointment shall be evidenced by an instrument signed by an authorized officer of the Trustee, a copy of which shall be furnished to the Company. Each reference in this Indenture to authentication by the Trustee includes authentication by such agent. An authenticating agent has the same rights as an Agent to deal with the Company or an Affiliate of the Company.

#### 2.4. REGISTRAR AND PAYING AGENT.

The Company shall maintain in each Place of Payment for any Series of Securities (i) an office or agency where such Securities may be presented for registration of transfer or for exchange ("Registrar"), (ii) an office or agency where such Securities may be presented for payment ("Paying Agent"), and PROVIDED, that at the option of the Company payment of interest may be made by check mailed to the address of the Person entitled thereto as such address shall appear in the register for the Securities maintained by the Registrar), and (iii) an office or agency where notices and demands to or upon the Company in respect of the Securities and this Indenture may be served ("Service Agent"). The Registrar shall keep a register of the Securities and of their transfer and exchange. The Company may have one or more co-Registrars and Service Agents and one or more additional Paying Agents. The Company shall give prompt written notice to the Trustee of the location, and any change in the location, of such office or agency. If at any time the Company shall fail to maintain any such required office, or to furnish the Trustee with the address thereof, such presentations, surrenders, notices and demands may be made or served at the address of the Trustee as set forth in Section 10.2, except for the service of legal process on the Company. The Company or any of its Subsidiaries may also act as Paying Agent. If the Company acts as Paying Agent, it shall segregate the money held by it for the payment of principal of, and interest and premium, if any, on, the Securities and hold it as a separate trust fund. The Company may change any Paying Agent, Registrar, co-registrar, Service Agent or any other Agent without notice to any Securityholder.

The Company may also from time to time designate one or more other offices or agencies where the Securities may be presented or surrendered for any or all such purposes, and may from time to time rescind such designations; PROVIDED, HOWEVER, that no such designation or rescission shall in any manner relieve the Company of its obligation to maintain an office or agency in each Place of Payment for Securities of any Series for such purposes. The Company hereby initially designates the Corporate Trust Office of the Trustee as such office of the Company. The Company shall give prompt written notice to the Trustee of such designation or rescission, and of any change in the location of any such other office or agency.

The Company shall enter into an appropriate agency agreement with any Registrar, Paying Agent or Service Agent not a party to this Indenture. The agreement shall implement the provisions of this Indenture that relate to such Agent. The Company shall notify the Trustee of the name and address of any such Agent. If the Company fails to maintain a Registrar, Paying Agent, or Service Agent and demands, or fails to give the foregoing notice, the Trustee shall act as such. The Company hereby appoints the Trustee as the initial Registrar, Paying Agent and Service Agent for each Series unless another Registrar, Paying Agent or Service Agent, as the case may be, is appointed prior to the time Securities of that Series are first issued.

## 2.5. PAYING AGENT TO HOLD ASSETS IN TRUST.

The Trustee as Paying Agent shall, and the Company shall require each Paying Agent other than the Trustee to agree in writing that each Paying Agent shall, hold in trust for the benefit of the Holders of any Series of Securities or the Trustee all assets

held by the Paying Agent for the payment of principal of, or interest or premium, if any, on, such Series of Securities (whether such assets have been distributed to it by the Company or any other obligor on such Series of Securities), and the Company and the Paying Agent shall notify the Trustee in writing of any Default by the Company (or any other obligor on such Series of Securities) in making any such payment. The Company at any time may require a Paying Agent to distribute all assets held by it to the Trustee and account for any assets disbursed, and the Trustee may, at any time during the continuance of any payment default with respect to any Series of Securities, upon written request to a Paying Agent, require such Paying Agent to distribute all assets held by it to the Trustee and to account for any assets distributed. Upon distribution to the Trustee of all assets that shall have been delivered by the Company to the Paying Agent, the Paying Agent shall have no further liability for such assets.

#### 2.6. SECURITYHOLDER LISTS.

The Trustee shall preserve in as current a form as is reasonably practicable the most recent list available to it of the names and addresses of Securityholders of each Series of Securities. If the Trustee is not the Registrar, the Company shall furnish to the Trustee as of each regular record date for the payment of interest on the Securities of a Series and before each related Interest Payment Date, and at such other times as the Trustee may request in writing, a list in such form and as of such date as the Trustee may reasonably require of the names and addresses of Securityholders of each Series of Securities.

## 2.7. TRANSFER AND EXCHANGE.

When Securities of a Series are presented by the Holder to the Registrar with a request to register the transfer thereof, the Registrar shall register the transfer as requested if the requirements of applicable law are met, and when such Securities of a Series are presented to the Registrar with a request to exchange them for an equal principal amount of other authorized denominations of Securities of the same Series, the Registrar shall make the exchange as requested. To permit transfers and exchanges, upon surrender of any Security for registration of transfer at the office or agency maintained pursuant to Section 2.4, the Company shall execute and the Trustee shall authenticate Securities at the Registrar's request.

If Securities are issued as Global Securities, the provisions of Section 2.15 shall apply.

All Securities issued upon any registration of transfer or exchange of Securities shall be the valid obligations of the Company, evidencing the same debt, and entitled to the same benefits under this Indenture, as the Securities surrendered upon such registration of transfer or exchange.

Every Security presented or surrendered for registration of transfer or for exchange shall (if so required by the Company or the Registrar or a co-registrar) be duly endorsed, or be accompanied by a written instrument of transfer in form satisfactory to the Company and the Registrar or a co-registrar, duly executed by the Holder thereof or his attorney duly authorized in writing.

Any exchange or transfer shall be without charge, except that the Company may require payment by the Holder of a sum sufficient to cover any tax or other governmental charge that may be imposed in relation to a transfer or exchange, but this provision shall not apply to any exchange pursuant to Section 2.11, 3.6 or 8.5. The Trustee shall not be required to register transfers of Securities of any Series, or to exchange Securities of any Series, for a period of 15 days before the record date for selection for redemption of such Securities. The Trustee shall not be required to exchange or register transfers of Securities of any Series called or being called for redemption in whole or in part, except the unredeemed portion of such Security being redeemed in part.

## 2.8. REPLACEMENT SECURITIES.

If a mutilated Security is surrendered to the Trustee, or if the Holder of a Security presents evidence to the satisfaction of the Company and the Trustee that the Security has been lost, destroyed or wrongfully taken, the Company shall issue and the Trustee shall authenticate a replacement Security of the same Series and of like tenor and principal amount and bearing a number not contemporaneously outstanding. An indemnity bond may be required by the Company or the Trustee that is sufficient in the reasonable judgment of the Company or the Trustee, as the case may be, to protect the Company, the Trustee or any Agent from any loss which any of them may suffer if a Security is replaced. The Company may charge such Holder for the Company's out-of-pocket expenses in replacing a Security, including the fees and expenses of the Trustee. Every replacement Security shall constitute an original additional obligation of the Company, whether or not the destroyed, lost or stolen Security shall be at any time enforceable by anyone, and shall be entitled to all the benefits of this Indenture equally and proportionately with any and all other Securities of that Series duly issued hereunder.

## 2.9. OUTSTANDING SECURITIES.

Securities outstanding at any time are all Securities authenticated by the Trustee, except for those canceled by it, those delivered to it for cancellation and those described in this Section 2.9 as not outstanding.

If a Security is replaced pursuant to Section 2.8 (other than a mutilated Security surrendered for replacement), it ceases to be outstanding until the Company and the Trustee receive proof satisfactory to each of them that the replaced Security is held by a bona fide purchaser. A mutilated Security ceases to be outstanding upon surrender of such Security and replacement thereof pursuant to Section 2.8.

If a Paying Agent holds on a Redemption Date or the Stated Maturity money sufficient to pay the principal of, premium, if any, and accrued interest on, Securities payable on that date, and is not prohibited from paying such money to the Holders thereof pursuant to the terms of this Indenture (PROVIDED, that if such Securities are to be redeemed, notice of such redemption has been duly given pursuant to this Indenture or provision therefor satisfactory to the Trustee has been made), then on and after that date and upon payment of such money to the Holder, such Securities will be cancelled and cease to be outstanding and interest on them ceases to accrue.

A Security does not cease to be outstanding solely because the Company or an Affiliate holds the Security.

## 2.10. WHEN TREASURY SECURITIES DISREGARDED; DETERMINATION OF HOLDERS' ACTION.

In determining whether the Holders of the required aggregate principal amount of the Securities of any Series have concurred in any direction, waiver or consent, the Securities of any Series owned by the Company or any other obligor on such Securities, or by any Affiliate of any of them, shall be disregarded, except that for the purposes of determining whether the Trustee shall be protected in relying on any such direction, waiver or consent, only Securities of such Series which a Responsible Officer of the Trustee actually knows are so owned shall be so disregarded. Securities of such Series so owned which have been pledged in good faith shall not be disregarded if the pledgee establishes to the satisfaction of the Trustee the pledgee's right so to act with respect to the Securities of such Series and that the pledgee is not the Company or any other obligor on the Securities of such Series, or an Affiliate of any of them.

#### 2.11. TEMPORARY SECURITIES.

Until definitive Securities are ready for delivery, the Company may prepare and execute, and upon the receipt of a Company Order, the Trustee shall authenticate, temporary Securities. Temporary Securities shall be substantially in the form, and shall carry all rights, of definitive Securities, but may have variations that the Company considers appropriate for temporary Securities. Without unreasonable delay, the Company shall prepare and execute, and upon the receipt of a Company Order, the Trustee shall authenticate, definitive Securities in exchange for temporary Securities without charge to the Holder.

## 2.12. CANCELLATION.

All Securities surrendered for payment, redemption or registration of transfer or exchange, or for credit against any sinking fund payment, shall, if surrendered to any Person other than the Trustee, be delivered to the Trustee for cancellation. The Company may at any time deliver to the Trustee for cancellation any Securities previously authenticated and delivered hereunder which the Company may have acquired in any manner whatsoever, and may deliver to the Trustee (or to any other Person for delivery to the Trustee) for cancellation any Securities previously authenticated hereunder which the Company has not issued and sold. The Registrar and the Paying Agent shall forward to the Trustee any Securities surrendered to them for transfer, exchange or payment. The Trustee, at the written direction of the Company, or, at the direction of the Company, the Registrar or the Paying Agent, and no one else, shall cancel, and at the written request of the Company shall dispose of, all Securities surrendered for transfer, exchange, payment or cancellation. If the Company shall acquire any of the Securities, such acquisition shall not operate as a redemption or satisfaction of the Indebtedness represented by such Securities unless and until the same are surrendered to the Trustee for cancellation pursuant to this Section 2.12. No Securities shall be authenticated in lieu of or in exchange for any Securities cancelled as provided in this Section 2.12, except as expressly permitted by this Indenture.

## 2.13. PAYMENT OF INTEREST; DEFAULTED INTEREST; COMPUTATION OF INTEREST.

Except as otherwise provided as contemplated by Section 2.2 with respect to any Series of Securities, interest on any Security which is payable, and is punctually paid or duly provided for, on any Interest Payment Date shall be paid to the Person in whose name that Security is registered at the close of business on the regular record date for such interest, as provided in the Board Resolution, supplemental indenture hereto or Officers' Certificate establishing the terms of such Series.

If the Company defaults in a payment of interest on the Securities, it shall pay the defaulted amounts, plus any interest payable on defaulted amounts pursuant to Section 4.1, to the Persons who are Securityholders on a subsequent special record date, which date shall be the 15th day next preceding the date fixed by the Company for the payment of defaulted interest, or the next succeeding Business Day if such date is not a Business Day. At least 15 days before the special record date, the Company shall deliver or cause to be delivered to each Securityholder, with a copy to the Trustee, a notice that states the special record date, the payment date and the amount of defaulted interest, and interest payable on such defaulted interest, if any, to be paid.

Except as otherwise specified as contemplated by Section 2.2 for Securities of any Series, interest on the Securities of each Series shall be computed on the basis of a 360-day year of twelve 30-day months.

## 2.14. CUSIP NUMBER.

The Company in issuing the Securities may use one or more "CUSIP" numbers, and, if the Company does so, the Trustee shall use the CUSIP number(s) in notices of redemption or exchange as a convenience to Holders, PROVIDED, that any such notice may state that no representation is made as to the correctness or accuracy of the CUSIP number(s) printed in the notice or on the

Securities, and that reliance may be placed only on the other identification numbers printed on the Securities, and that any such redemption or exchange shall not be affected by any defect in or omission of any such numbers. The Company shall promptly notify the Trustee in writing of any change in the "CUSIP" numbers assigned to the Securities.

## 2.15. PROVISIONS FOR GLOBAL SECURITIES.

- (a) A Board Resolution, a supplemental indenture hereto or an Officers' Certificate shall establish whether the Securities of a Series shall be issued in whole or in part in the form of one or more Global Securities, and the Depository for such Global Securities or Securities.
- (b) Notwithstanding any provisions to the contrary contained in Section 2.7 and in addition thereto, if, and only if the Depository (i) at any time is unwilling or unable to continue as Depository for such Global Security or ceases to be a clearing agency registered under the Exchange Act and (ii) a successor Depository is not appointed by the Company within 90 days after the date the Company is so informed in writing or becomes aware of the same, the Company promptly will execute and deliver to the Trustee definitive Securities, and the Trustee, upon receipt of a Company Request for the authentication and delivery of such definitive Securities (which the Company will promptly execute and deliver to the Trustee) and an Officers' Certificate to the effect that such Global Security shall be so exchangeable, will authenticate and deliver definitive Securities, without charge, registered in such names and in such authorized denominations as the Depository shall direct in writing (pursuant to instructions from its direct and indirect participants or otherwise) in an aggregate principal amount equal to the principal amount of the Global Security with like tenor and terms. Upon the exchange of a Global Security for definitive Securities, such Global Security shall be canceled by the Trustee. Unless and until it is exchanged in whole or in part for definitive Securities, as provided in this Section 2.15(b), a Global Security may not be transferred except as a whole by the Depository with respect to such Global Security to a nominee of such Depository to such Depository or another nominee of such Depository or any such nominee to a successor Depository or a nominee of such a successor Depository.
  - (c) Any Global Security issued hereunder shall bear a legend in substantially the following form:
    - "This Security is a Global Security within the meaning of the Indenture hereinafter referred to, and is registered in the name of the Depository or a nominee of the Depository. This Security is exchangeable for Securities registered in the name of a Person other than the Depository or its nominee only in the limited circumstances described in the Indenture, and may not be transferred except as a whole by the Depository to a nominee of the Depository, by a nominee of the Depository to the Depository or another nominee of the Depository or any such nominee to a successor Depository or a nominee of such a successor Depository."
- (d) The Depository, as a Holder, may appoint agents and otherwise authorize participants to give or take any request, demand, authorization, direction, notice, consent, waiver or other action which a Holder is entitled to give or take under the Indenture.
- (e) Notwithstanding the other provisions of this Indenture, unless otherwise specified as contemplated by Section 2.2, payment of the principal of, and interest and premium, if any, on, any Global Security shall be made to the Depository or its nominee in its capacity as the Holder thereof.
- (f) Except as provided in Section 2.15(e) above, the Company, the Trustee and any Agent shall treat a Person as the Holder of such principal amount of outstanding Securities of any Series represented by a Global Security as shall be specified in a written statement of the Depository (which may be in the form of a participants' list for such Series) with respect to such Global Security, for purposes of obtaining any consents, declarations, waivers or directions required to be given by the Holders pursuant to this Indenture, PROVIDED, that until the Trustee is so provided with a written statement, it may treat the Depository or any other Person in whose name a Global Security is registered as the owner of such Global Security for the purpose of receiving payment of the principal of, and any premium and (subject to Section 2.13) any interest on, such Global Security and for all other purposes whatsoever, and none of the Company, the Trustee or any agent of the Company or the Trustee shall be affected by notice to the contrary.

## 2.16. PERSONS DEEMED OWNERS.

Prior to due presentment of a Security for registration of transfer, the Company, the Trustee, the Registrar and any agent of the Company, the Registrar or the Trustee may treat the Person in whose name such Security is registered as the owner of such Security for the purpose of receiving payment of the principal of, and any premium and (subject to Section 2.13) any interest on, such Security and for all other purposes whatsoever, and none of the Company, the Trustee, the Registrar or any agent of the Company, the Trustee or the Registrar shall be affected by notice to the contrary.

# ARTICLE 3 – REDEMPTION

## 3.1. NOTICES TO TRUSTEE.

The Company may, with respect to any Series of Securities, reserve the right to redeem and pay the Series of Securities, or may covenant to redeem and pay the Series of Securities or any part thereof, prior to the Stated Maturity thereof at such time and on

such terms as provided for in such Securities or the related Board Resolution, supplemental indenture or Officers' Certificate. If a Series of Securities is redeemable and the Company elects to redeem all or part of such Series of Securities, it shall notify the Trustee in writing of the Redemption Date and the principal amount of Securities to be redeemed at least 45 days (unless a shorter notice shall be satisfactory to the Trustee) before the Redemption Date. Any such notice may be canceled at any time prior to notice of such redemption being delivered to any Holder, and shall thereby be void and of no effect.

## 3.2. SELECTION BY TRUSTEE OF SECURITIES TO BE REDEEMED.

Unless otherwise indicated for a particular Series of Securities by a Board Resolution, a supplemental indenture or an Officers' Certificate, if fewer than all of the Securities of a Series are to be redeemed, the Trustee shall select the Securities of a Series to be redeemed pro rata, by lot or by any other method that the Trustee considers fair and appropriate (unless the Company specifically directs the Trustee otherwise) subject, in the case of Global Securities, to the applicable rules and procedures of the Depository and, if such Securities are listed on any securities exchange, by a method that complies with the requirements of such exchange.

The Trustee shall make the selection from Securities of a Series outstanding and not previously called for redemption, and shall promptly notify the Company in writing of the Securities selected for redemption and, in the case of any Security selected for partial redemption, the principal amount thereof to be redeemed at least 35 but not more than 60 days before the Redemption Date. Securities of a Series in denominations of \$1,000 may be redeemed only in whole. The Trustee may select for redemption portions of the principal of Securities of a Series that have denominations larger than \$1,000. Securities of a Series and portions of them it selects shall be in amounts of \$1,000 or, with respect to Securities of any Series issuable in other denominations pursuant to Section 2.2(10), the minimum principal denomination for each Series and integral multiples thereof. Provisions of this Indenture that apply to Securities called for redemption also apply to portions of Securities called for redemption. Notwithstanding the above, the applicable procedures of the Depository regarding the selection of Securities issued as Global Securities shall govern.

Neither the Trustee nor any Agent will have any responsibility for any action taken or not taken by the Depository. The Trustee and the Paying Agent will have no responsibility or obligation to any beneficial owner of a Global Security or a depository participant or other Person with respect to the accuracy of the records of the Depository or its nominee or of any participant or member thereof with respect to any ownership interest in the Securities or with respect to the delivery to any participant, member, beneficial owner or other Person (other than the Depository) of any notice (including any Redemption Notice) or the payment of any amount, under or with respect to such Securities. The rights of beneficial owners in any Global Securities will be exercised only through the Depository subject to the Depository Procedures. The Trustee may rely and shall be fully protected in relying upon information furnished by the Depository with respect to its members, participants and any beneficial owners.

## 3.3. NOTICE OF REDEMPTION.

Unless otherwise indicated for a particular Series by Board Resolution, a supplemental indenture hereto or an Officers' Certificate, at least 30 days, and no more than 60 days, before a Redemption Date, the Company shall deliver, or cause to be delivered, a notice of redemption by first-class mail to each Holder of Securities to be redeemed at his or her last address as the same appears on the registry books maintained by the Registrar, or, with respect to Securities issued as Global Securities, such notice shall be delivered in accordance with the applicable procedures of the Depository. The notice shall identify the Securities to be redeemed and shall state:

- (1) the Redemption Date;
- (2) the redemption price, and that such redemption price shall become due and payable on the Redemption Date;
- (3) if any Security of a Series is being redeemed in part, the portion of the principal amount of such Security of a Series to be redeemed and that, after the Redemption Date and upon surrender of such Security of a Series, a new Security or Securities in principal amount equal to the unredeemed portion will be issued;
  - (4) the name and address of the Paying Agent;
- (5) that Securities of a Series called for redemption must be surrendered to the Paying Agent to collect the redemption price, and the place or places where each such Security is to be surrendered for such payment;
- (6) that, unless the Company defaults in making the redemption payment, interest on the Securities of a Series called for redemption ceases to accrue on the Redemption Date, and the only remaining right of the Holders of such Securities is to receive payment of the redemption price upon surrender to the Paying Agent of the Securities redeemed;
- (7) if fewer than all of the Securities of a Series are to be redeemed, the identification of the particular Securities of a Series (or portion thereof) to be redeemed, as well as the aggregate principal amount of Securities of a Series to be outstanding after such partial redemption.
  - (8) the CUSIP number, if any, printed on the Securities being redeemed; and

(9) that no representation is made as to the correctness or accuracy of the CUSIP number, if any, listed in such notice or printed on the Securities.

At the Company's request, the Trustee shall give the notice of redemption in the Company's name and at the Company's sole expense; provided, however, that the Company has delivered to the Trustee prior to the notice date, an Officers' Certificate requesting that the Trustee give such notice and setting forth the information to be stated in such notice and the form of such notice.

#### 3.4. EFFECT OF NOTICE OF REDEMPTION.

Once the notice of redemption described in Section 3.3 is delivered, Securities of a Series called for redemption become due and payable on the Redemption Date and at the redemption price, plus interest, if any, accrued to the Redemption Date. Upon surrender to the Trustee or Paying Agent, such Securities of a Series shall be paid at the redemption price, plus accrued interest, if any, to the Redemption Date; PROVIDED, that if the Redemption Date is after a regular interest payment record date and on or prior to the next Interest Payment Date, the accrued interest shall be payable to the Holder of the redeemed Securities registered on the relevant record date, as specified by the Company in the notice to the Trustee pursuant to Section 3.1.

## 3.5. DEPOSIT OF REDEMPTION PRICE.

On or prior to the Redemption Date (but no later than 11:00 A.M. Eastern Time on such date), the Company shall deposit with the Paying Agent money sufficient to pay the redemption price of and accrued interest, if any, on all Securities to be redeemed on that date other than Securities or portions thereof called for redemption on that date which have been delivered by the Company to the Trustee for cancellation.

On and after any Redemption Date, if money sufficient to pay the redemption price of, and accrued interest on, Securities called for redemption shall have been made available in accordance with the preceding paragraph and the Company and the Paying Agent are not prohibited from paying such moneys to Holders, the Securities called for redemption will cease to accrue interest and the only right of the Holders of such Securities will be to receive payment of the redemption price of and, subject to the proviso in Section 3.4, accrued and unpaid interest on such Securities to the Redemption Date. If any Security called for redemption shall not be so paid, interest will be paid, from the Redemption Date until such redemption payment is made, on the unpaid principal of the Security and any interest or premium, if any, not paid on such unpaid principal, in each case, at the rate and in the manner provided in the Securities.

## 3.6. SECURITIES REDEEMED IN PART.

Upon surrender of a Security of a Series that is redeemed in part, the Company shall execute, and the Trustee shall authenticate, for a Holder a new Security of the same Series equal in principal amount to the unredeemed portion of the Security surrendered.

# ARTICLE 4 – COVENANTS

## 4.1. PAYMENT OF SECURITIES.

The Company shall pay the principal of, and interest and premium, if any, on, each Series of Securities on the dates and in the manner provided in such Securities and this Indenture.

An installment of principal or interest shall be considered paid on the date it is due if the Trustee or Paying Agent holds on that date money designated for and sufficient to pay such installment and is not prohibited from paying such money to the Holders pursuant to the terms of this Indenture or otherwise.

The Company shall pay interest on overdue principal, and overdue interest, to the extent lawful, at the rate specified in the Series of Securities.

## 4.2. SEC REPORTS.

The Company will deliver to the Trustee within 15 days after the filing of the same with the SEC, copies of the quarterly and annual reports and of the information, documents and other reports, if any, which the Company is required to file with the SEC pursuant to Section 13 or 15(d) of the Exchange Act; PROVIDED, HOWEVER, that each such report or document will be deemed to be so delivered to the Trustee if the Company files such report or document with the SEC through the SEC's EDGAR database no later than the time such report or document is required to be filed with the SEC pursuant to the Exchange Act. Notwithstanding that the Company may not be subject to the reporting requirements of Section 13 or 15(d) of the Exchange Act, the Company will file with the SEC, to the extent permitted, and provide the Trustee with, such quarterly and annual reports and such information, documents and other reports specified in Sections 13 and 15(d) of the Exchange Act. The Company will also comply with the other provisions of TIA Section 314(a).

Delivery of such reports, information and documents to the Trustee is for informational purposes only and the Trustee's receipt of such shall not constitute actual or constructive notice of any information contained therein or determinable from information contained therein, including the Company's compliance with any of its covenants hereunder (as to which the Trustee is entitled to rely exclusively on Officers' Certificates). The Trustee will not be responsible and shall have no liability whatsoever to determine whether any financial information has been filed or posted on the EDGAR system (or any successor electronic delivery procedure) or have any duty to monitor or determine whether the Company has delivered the reports described under this Section 4.2 or otherwise complied with its obligation under this Section 4.2.

## 4.3. WAIVER OF STAY, EXTENSION OR USURY LAWS.

The Company covenants (to the extent that it may lawfully do so) that it will not at any time insist upon, or plead (as a defense or otherwise) or in any manner whatsoever claim or take the benefit or advantage of, any stay, extension, usury or other law which would prohibit or forgive the Company from paying all or any portion of the principal of, and/or interest and premium, if any, on, the Securities as contemplated herein, wherever enacted, now or at any time hereafter in force, or which may affect the covenants or the performance of this Indenture; and the Company hereby expressly waives (to the extent that they may lawfully do so) all benefit or advantage of any such law, and covenants that it will not hinder, delay or impede the execution of any power herein granted to the Trustee, but will suffer and permit the execution of every such power as though no such law had been enacted.

#### 4.4. COMPLIANCE CERTIFICATE.

(a) The Company shall deliver to the Trustee, within 120 days after the end of each fiscal year of the Company, an Officers' Certificate which complies with TIA Section 314(a)(4) stating that a review of the activities of the Company and its Subsidiaries during such fiscal year has been made under the supervision of the signing Officers with a view to determining whether the Company has kept, observed, performed and fulfilled its obligations under this Indenture, and further stating, as to each such Officer signing such certificate, that to the best of his or her knowledge the Company has kept, observed, performed and fulfilled each and every covenant contained in this Indenture and that there is no default in the performance or observance of any of the terms, provisions and conditions hereof (or, if a Default or Event of Default shall have occurred, describing all such Defaults or Events of Default of which he or she may have knowledge and what action the Company is taking or proposes to take with respect thereto) and that to the best of his or her knowledge no event has occurred and remains in existence by reason of which payments on account of the principal of, or interest or premium, if any, on, the Securities is prohibited, or if such event has occurred, a description of the event and what action the Company is taking or proposes to take with respect thereto.

(b)(i) If any Default or Event of Default has occurred and is continuing or (ii) if any Holder seeks to exercise any remedy hereunder with respect to a claimed Default under this Indenture or the Securities, within five Business Days after the Company becoming aware of such occurrence the Company shall deliver to the Trustee an Officers' Certificate specifying such event, notice or other action and what action the Company is taking or proposes to take with respect thereto.

## 4.5. CORPORATE EXISTENCE.

Subject to Article 5, the Company shall do or cause to be done all things necessary to preserve and keep in full force and effect its corporate existence, in accordance with the organizational documents (as the same may be amended from time to time) of the Company and the rights (charter and statutory), licenses and franchises of the Company; PROVIDED, HOWEVER, that the Company shall not be required to preserve any such right, license or franchise, or its corporate existence, if the Board of Directors shall determine that the preservation thereof is no longer desirable in the conduct of the business of the Company and that the loss thereof is not adverse in any material respect to the Holders.

# ARTICLE 5 – SUCCESSOR CORPORATION

## 5.1. LIMITATION ON CONSOLIDATION, MERGER AND SALE OF ASSETS.

(a) The Company will not, in any transaction or series of transactions, merge or consolidate with or into, or sell, assign, convey, transfer, lease or otherwise dispose of all or substantially all of its properties and assets (as an entirety or substantially as an entirety in one transaction or a series of related transactions), to any Person or Persons, unless at the time of and after giving effect thereto (i) either (A) if the transaction or series of transactions is a merger or consolidation, the Company shall be the surviving Person of such merger or consolidation, or (B) the Person formed by such consolidation or into which the Company is merged or to which the properties and assets of the Company are transferred (any such surviving Person or transferee Person being the "Surviving Entity") shall be a corporation organized and existing under the laws of the United States of America, any state thereof or the District of Columbia, or a corporation or comparable legal entity organized under the laws of a foreign jurisdiction and shall expressly assume by a supplemental indenture executed and delivered to the Trustee, in form reasonably satisfactory to the Trustee, all of the obligations of the Company (including, without limitation, the obligation to pay the principal of, and premium and interest, if any, on, the Securities and the performance of the other covenants) under the Securities of each Series and this Indenture, and in each case, this Indenture shall remain in full force and effect; and (ii) immediately before and immediately after giving effect to such transaction or series of transactions), no Default or Event of Default shall have occurred and be continuing.

(b) In connection with any consolidation, merger or transfer of assets contemplated by this Section 5.1, the Company shall deliver, or cause to be delivered, to the Trustee an Officers' Certificate and an Opinion of Counsel, each stating that such consolidation, merger or transfer, and the supplemental indenture in respect thereto, comply with this Section 5.1, and that all conditions precedent herein provided for relating to such transaction or transactions have been complied with.

## 5.2. SUCCESSOR PERSON SUBSTITUTED.

Upon any consolidation, merger or transfer of all or substantially all of the assets of the Company in accordance with Section 5.1 above, the successor corporation formed by such consolidation, or into which the Company is merged or to which such transfer is made, shall succeed to, and be substituted for, and may exercise every right and power of, the Company under this Indenture with the same effect as if such successor corporation had been named as the Company herein, and thereafter (except with respect to any such transfer which is a lease) the predecessor corporation shall be relieved of all obligations and covenants under this Indenture and the Securities.

# ARTICLE 6 – DEFAULTS AND REMEDIES

#### 6.1. EVENTS OF DEFAULT.

An "Event of Default," wherever used herein with respect to Securities of any Series, means any one of the following events, unless in the establishing Board Resolution, supplemental indenture or Officers' Certificate, it is provided that such Series shall not have the benefit of said Event of Default:

- (1) there is a default in the payment of any principal of, or premium, if any, on, the Securities when the same becomes due and payable at Maturity, upon acceleration, redemption or otherwise;
- (2) there is a default in the payment of any interest on any Security of a Series when the same becomes due and payable, and the Default continues for a period of 30 days;
- (3) the Company defaults in the observance or performance of any other covenant in the Securities of a Series or in this Indenture for 90 days after written notice from the Trustee or the Holders of not less than 25% in the aggregate principal amount of the Securities of such Series then outstanding, which notice must specify the Default, demand that it be remedied and state that the notice is a "Notice of Default";
  - (4) the Company or any Significant Subsidiary pursuant to or within the meaning of any Bankruptcy Law:
    - (A) commences a voluntary case,
    - (B) consents to the entry of an order for relief against it in an involuntary case,
    - (C) consents to the appointment of a Custodian of it or for all or substantially all of its property,
    - (D) makes a general assignment for the benefit of its creditors, or
    - (E) generally is not paying its debts as they become due;
  - (5) a court of competent jurisdiction enters an order or decree under any Bankruptcy Law that:
    - (A) is for relief against the Company or any Significant Subsidiary in an involuntary case;
- (B) appoints a Custodian of the Company or any Significant Subsidiary, or for all or substantially all of the property of the Company or any Significant Subsidiary; or
  - (C) orders the liquidation of the Company or any Significant Subsidiary, and the order or decree remains unstayed and in effect for 90 consecutive days; or
- (6) any other Event of Default provided with respect to Securities of that Series, which is specified in a Board Resolution, a supplemental indenture hereto or an Officers' Certificate, in accordance with Section 2.2(19).

The term "Bankruptcy Law" means Title 11, U.S. Code, or any similar federal or state law for the relief of debtors. The term "Custodian" means any receiver, trustee, assignee, liquidator or similar official under any Bankruptcy Law.

The Trustee may withhold notice of any Default (except in the payment of the principal of, or interest or premium, if any, on, the Securities) to the Holders of the Securities of any Series in accordance with Section 7.5. When a Default is cured, it ceases to exist.

The Trustee shall not be charged with knowledge of any Event of Default unless written notice thereof shall have been actually received by a Responsible Officer at the Corporate Trust Office of the Trustee from the Company, a Paying Agent, any Holder or any agent of any Holder.

## 6.2. ACCELERATION.

If an Event of Default with respect to Securities of any Series at the time outstanding (other than an Event of Default arising under Section 6.1(4) or (5)) occurs and is continuing, the Trustee by written notice to the Company, or the Holders of not less than 25% in aggregate principal amount of the Securities of that Series then outstanding by written notice to the Company and the Trustee, may declare that the entire principal amount of all the Securities of that Series then outstanding plus accrued and unpaid interest to the date of acceleration are immediately due and payable, in which case such amounts shall become immediately due and payable; PROVIDED, HOWEVER, that after such acceleration but before a judgment or decree based on such acceleration is obtained by the Trustee, the Holders of a majority in aggregate principal amount of the outstanding Securities of that Series may rescind and annul such acceleration and its consequences if (i) all existing Events of Default, other than the nonpayment of accelerated principal, interest or premium, if any, that has become due solely because of the acceleration, have been cured or waived, (ii) to the extent the payment of such interest is lawful, interest on overdue installments of interest and overdue principal, which has become due otherwise than by such declaration of acceleration, has been paid and (iii) the rescission would not conflict with any judgment or decree. No such rescission shall affect any subsequent Default or impair any right consequent therefor. In case an Event of Default specified in Section 6.1(4) or (5) with respect to the Company occurs, such principal, premium, if any, and interest amount with respect to all of the Securities of that Series shall be due and payable immediately without any declaration or other act on the part of the Trustee or the Holders of the Securities of that Series

## 6.3. REMEDIES.

If an Event of Default with respect to Securities of any Series at the time outstanding occurs and is continuing, the Trustee may, but shall not be obligated to, pursue any available remedy by proceeding at law or in equity to collect the payment of the principal of, or interest and premium, if any, on, the Securities of that Series, or to enforce the performance of any provision of the Securities of that Series or this Indenture.

The Trustee may maintain a proceeding even if it does not possess any of the Securities of that Series or does not produce any of them in the proceeding. A delay or omission by the Trustee or any Securityholder in exercising any right or remedy accruing upon an Event of Default shall not impair the right or remedy or constitute a waiver of or acquiescence in the Event of Default. No remedy is exclusive of any other remedy. All available remedies are cumulative to the extent permitted by law.

## 6.4. WAIVER OF PAST DEFAULTS AND EVENTS OF DEFAULT.

Subject to Sections 6.2, 6.7 and 8.2, the Holders of a majority in principal amount of the Securities of any Series then outstanding by written notice to the Trustee have the right to waive any existing Default or Event of Default with respect to such Series or compliance with any provision of this Indenture (with respect to such Series) or the Securities of such Series. Upon any such waiver, such Default with respect to such Series shall cease to exist, and any Event of Default with respect to such Series arising therefrom shall be deemed to have been cured for every purpose of this Indenture; but no such waiver shall extend to any subsequent or other Default or Event of Default or impair any right consequent thereto. This Section 6.4 shall be in lieu of TIA Section 316(a)(1)(B), and TIA Section 316(a)(1)(B) is hereby expressly excluded from this Indenture and Section as permitted by the TIA.

## 6.5. CONTROL BY MAJORITY.

Subject to Sections 6.2, 6.7 and 8.2, the Holders of a majority in principal amount of the Securities of any Series then outstanding may direct the time, method and place of conducting any proceeding for any remedy available to the Trustee or exercising any trust or power conferred on the Trustee by this Indenture with respect to such Series. The Trustee, however, may refuse to follow any direction that conflicts with law or this Indenture, or that the Trustee determines may be unduly prejudicial to the rights of another Securityholder or the Trustee, or that may involve the Trustee in personal liability; PROVIDED, that the Trustee may take any other action deemed proper by the Trustee which is not inconsistent with such direction. This Section 6.5 shall be in lieu of TIA Section 316(a)(1)(A), and TIA Section 316(a)(1)(A) is hereby expressly excluded from this Indenture and Section as permitted by the TIA.

# 6.6. LIMITATION ON SUITS.

Subject to Section 6.7, a Securityholder may not institute any proceeding or pursue any remedy with respect to this Indenture or the Securities of a Series unless:

- (1) the Holder gives to the Trustee written notice of a continuing Event of Default with respect to the Securities of that Series;
- (2) the Holders of at least 25% in aggregate principal amount of the Securities of such Series then outstanding make a written request to the Trustee to pursue the remedy;
- (3) such Holder or Holders offer to the Trustee indemnity reasonably satisfactory to the Trustee against any loss, liability or expense to be incurred in compliance with such request;
  - (4) the Trustee does not comply with the request within 60 days after receipt of the request and the offer of indemnity; and
- (5) no direction inconsistent with such written request has been given to the Trustee during such 60-day period by the Holders of a majority in aggregate principal amount of the Securities of such Series then outstanding.

A Securityholder may not use this Indenture to prejudice the rights of another Securityholder, or to obtain a preference or priority over another Securityholder.

#### 6.7. RIGHTS OF HOLDERS TO RECEIVE PAYMENT.

Notwithstanding any other provision of this Indenture, the right of any Holder of a Security of a Series to receive payment of the principal of, and interest and premium, if any, on, the Security of such Series on or after the respective due dates expressed in the Security of such Series, or to bring suit for the enforcement of any such payment on or after such respective dates, is absolute and unconditional, and shall not be impaired or affected without the consent of the Holder.

## 6.8. COLLECTION SUIT BY TRUSTEE.

If an Event of Default in payment of principal, interest or premium, if any, specified in Section 6.1(1) or (2) with respect to Securities of any Series at the time outstanding occurs and is continuing, the Trustee may recover judgment in its own name and as trustee of an express trust against the Company (or any other obligor on the Securities of that Series) for the whole amount of unpaid principal and premium, if any, and accrued interest remaining unpaid, together with interest on overdue principal and premium, if any, and, to the extent that payment of such interest is lawful, interest on overdue installments of interest, in each case at the rate then borne by the Securities of that Series, and such further amounts as shall be sufficient to cover the costs and expenses of collection, including the reasonable compensation, expenses, disbursements and advances of the Trustee, its agents and counsel, as set forth in Section 7.7.

## 6.9. TRUSTEE MAY FILE PROOFS OF CLAIM.

The Trustee may file such proofs of claim and other papers or documents, and take other actions (including sitting on a committee of creditors), as may be necessary or advisable in order to have the claims of the Trustee (including any claim for the reasonable compensation, expenses, disbursements and advances of the Trustee, its agents and counsel) and the Securityholders allowed in any judicial proceedings relative to the Company (or any other obligor on the Securities), any of their respective creditors or any of their respective property, and the Trustee shall be entitled and empowered to collect and receive any monies or other property payable or deliverable on any such claims, and to distribute the same after deduction of its charges and expenses to the extent that any such charges and expenses are not paid out of the estate in any such proceedings, and any custodian in any such judicial proceeding is hereby authorized by each Securityholder to make such payments to the Trustee, and in the event that the Trustee shall consent to the making of such payments directly to the Securityholders, to pay to the Trustee any amount due to it for the reasonable compensation, expenses, disbursements and advances of the Trustee, its agents and counsel, and any other amounts due the Trustee under Section 7.7, and to the extent that such payment of reasonable compensation, expenses, disbursements and advances in any such proceedings shall be denied for any reason, payment of the same shall be secured by a lien on, and shall be paid out of, any and all distributions, dividends, money, securities and other property which the Holders may be entitled to receive in such proceedings, whether in liquidation or under any plan of reorganization or arrangement or otherwise.

Nothing herein contained shall be deemed to authorize the Trustee to authorize or consent to, or accept or adopt on behalf of any Securityholder, any plan of reorganization, arrangement, adjustment or composition affecting the Securities of a Series or the rights of any Holder thereof, or to authorize the Trustee to vote in respect of the claim of any Securityholder in any such proceedings.

## 6.10. PRIORITIES.

If the Trustee collects any money pursuant to this Article 6, it shall pay out the money in the following order:

FIRST: to the Trustee for amounts due under Section 7.7;

SECOND: to Securityholders for amounts then due and unpaid for the principal of, and interest and premium, if any, on, the Securities in respect of which, or for the benefit of which, such money has been collected, ratably, without preference or priority of any kind, according to the amounts due and payable on such Securities; for principal and any premium and interest, respectively; and

THIRD: to the Company.

The Trustee may fix a record date and payment date for any payment to Securityholders pursuant to this Section 6.10. At least 15 days before such record date, the Trustee shall deliver to each Securityholder a notice that states the record date, the payment date and amount to be paid.

## 6.11. UNDERTAKING FOR COSTS.

In any suit for the enforcement of any right or remedy under this Indenture, or in any suit against the Trustee for any action taken or omitted by it as Trustee, a court in its discretion may require the filing by any party litigant in the suit of an undertaking to pay the costs of the suit, and the court in its discretion may assess reasonable costs, including reasonable attorneys' fees, against any party litigant in the suit, having due regard to the merits and good faith of the claims or defenses made by the party litigant. This Section 6.11 does not apply to a suit by the Trustee, a suit by a Holder pursuant to Section 6.7 or a suit by Holders of more than 10% in principal amount of the Securities of a Series then outstanding.

## ARTICLE 7 – TRUSTEE

## 7.1. DUTIES OF TRUSTEE.

- (a) If an Event of Default has occurred and is continuing, the Trustee shall exercise such of the rights and powers vested in it by this Indenture and use the same degree of care and skill in their exercise as a prudent Person would exercise or use under the same circumstances in the conduct of his own affairs.
  - (b) Except during the continuance of an Event of Default:
- (1) The Trustee need perform only those duties that are specifically set forth in this Indenture, and no covenants or obligations shall be implied in this Indenture against the Trustee.
- (2) In the absence of gross negligence or willful misconduct on its part, the Trustee may conclusively rely, as to the truth of the statements and the correctness of the opinions expressed therein, upon certificates or opinions furnished to the Trustee and conforming to the requirements of this Indenture, but, in the case of any such certificates or opinions which by any provision hereof are specifically required to be furnished to the Trustee, the Trustee shall be under a duty to examine the same to determine whether or not they conform to the requirements of this Indenture.
  - (c) The Trustee may not be relieved from liability for its own negligent action, its own negligent failure to act or its own willful misconduct, except that:
    - (1) This paragraph does not limit the effect of paragraph (b) of this Section 7.1.
- (2) The Trustee shall not be liable for any error of judgment made in good faith by a Responsible Officer, unless it is proved that the Trustee was negligent in ascertaining the pertinent facts.
- (3) The Trustee shall not be liable with respect to any action it takes or omits to take in good faith in accordance with a direction received by it pursuant to Sections 6.2 and 6.5.
- (d) No provision of this Indenture shall require the Trustee to expend or risk its own funds, or otherwise incur any liability, including financial liability, in the performance of any of its rights or powers if it shall have reasonable grounds for believing that repayment of such funds or adequate indemnity satisfactory to it against such risk or liability is not reasonably assured to it. The Trustee shall be under no obligation to exercise any of the rights or power vested in it by this Indenture at the request or direction of any of the Holder pursuant to this Indenture, unless such Holders shall have offered, and if requested provided, to the Trustee indemnity satisfactory to the Trustee against the costs, expenses and liabilities that might be incurred by it in compliance with such request or direction.
- (e) Whether or not therein expressly so provided, paragraphs (a), (b), (c) and (d) of this Section 7.1 shall govern every provision of this Indenture that in any way relates to the Trustee.
- (f) The Trustee and Paying Agent shall not be liable for interest on any money received by either of them, except as the Trustee and Paying Agent may agree in writing with the Company. Money held in trust by the Trustee need not be segregated from other funds except to the extent required by the law.

- (g) The Paying Agent, the Registrar and any authenticating agent shall be entitled to the protections, immunities and standard of care set forth in paragraphs (a), (b), (c), (d) and (f) of this Section 7.1 and in Section 7.2 with respect to the Trustee.
  - (h) The Trustee shall not be required to give any bond or surety in respect of the performance of its powers or duties hereunder.

## 7.2. RIGHTS OF TRUSTEE.

- (a) Subject to Section 7.1:
- (1) The Trustee may rely on, and shall be protected in acting or refraining from acting upon, any document reasonably believed by it to be genuine and to have been signed or presented by the proper Person. The Trustee need not investigate any fact or matter stated in the document.
- (2) Before the Trustee acts or refrains from acting, it may require and may conclusively rely on an Officers' Certificate or an Opinion of Counsel, or both, which shall conform to the provisions of Section 10.5. The Trustee shall be protected and shall not be liable for any action it takes or omits to take in good faith in reliance on such certificate or opinion.
  - (3) The Trustee may act through agents and attorneys, and shall not be responsible for the misconduct or negligence of any agent appointed by it with due care.
  - (4) The Trustee shall not be liable for any action it takes or omits to take in good faith which it reasonably believes to be authorized or within its rights or powers.
- (5) The Trustee may consult with counsel of its selection, which may be counsel to the Company, and the advice or opinion of such counsel as to matters of law shall be full and complete authorization and protection from liability in respect of any action taken, omitted or suffered by it hereunder in good faith and in accordance with the advice or opinion of such counsel.
- (6) The Trustee shall be under no obligation to exercise any of the rights or powers vested in it by this Indenture at the request, order or direction of any of the Holders pursuant to the provisions of this Indenture, unless such Holders shall have provided to the Trustee security or indemnity satisfactory to the Trustee against the costs, expenses and liabilities which may be incurred therein or thereby.
- (7) The Trustee shall not be deemed to have knowledge of any fact or matter (including, without limitation, a Default or Event of Default) unless such fact or matter is known to a Responsible Officer of the Trustee.
- (8) Unless otherwise expressly provided herein or in the Securities of a Series or the related Board Resolution, supplemental indenture or Officers' Certificate, the Trustee shall not have any responsibility with respect to reports, notices, certificates or other documents filed with it hereunder, except to make them available for inspection, at reasonable times, by Securityholders, it being understood that delivery of such reports, information and documents to the Trustee is for informational purposes only and the Trustee's receipt of such shall not constitute constructive notice of any information contained therein or determinable from information contained therein, including the Company's compliance with any of its covenants hereunder (except as set forth in Section 4.4).
- (9) The Trustee shall not be bound to make any investigation into the facts or matters stated in any resolution, certificate, statement, instrument, opinion, report, notice, request, direction, consent, order, bond, debenture, note, other evidence of indebtedness or other paper or document.
  - (10) The permissive right of the Trustee to take actions permitted by this Indenture will not be construed as an obligation or duty to do so.
- (11) The rights, privileges, protection, immunities and benefits given to the Trustee, including, without limitation, its right to be indemnified, are extended to, and shall be enforceable by, the Trustee in each of its capacities hereunder, and to each agent, custodian and other person employed to act hereunder.
- (12) The Trustee may request that the Company deliver an Officers' Certificate setting forth the names of individuals and/or titles of officers authorized at such time to take specified actions pursuant to this Indenture, which Officers' Certificate may be signed by any Person authorized to sign an Officers' Certificate, including any Person specified as so authorized in any such certificate previously delivered and not superseded.
- (13) In no event shall the Trustee be responsible or liable for special, punitive, indirect or consequential loss or damage of any kind whatsoever (including, but not limited to, loss of profit), irrespective of whether the Trustee has been advised of the likelihood of such loss or damage and regardless of the form of action.

#### 7.3. INDIVIDUAL RIGHTS OF TRUSTEE.

The Trustee in its individual or any other capacity may become the owner or pledgee of Securities, and may make loans to, accept deposits from, perform services for or otherwise deal with the Company, or any Affiliate thereof, with the same rights it would have if it were not Trustee. Any Agent may do the same with like rights. The Trustee, however, shall be subject to Sections 7.10 and 7.11.

#### 7.4. TRUSTEE'S DISCLAIMER.

The Trustee makes no representation as to the validity or adequacy of this Indenture or the Securities (except that the Trustee represents that it is duly authorized to execute and deliver this Indenture and authenticate the Securities and perform its obligations hereunder), and the Trustee shall not be accountable for the Company's use of the proceeds from the sale of Securities or any money paid to the Company or pursuant to the Company's direction pursuant to the terms of this Indenture, and the Trustee shall not be responsible for any statement or recital herein or any statement in the Securities or any other document in connection with the sale of the Securities or pursuant to this Indenture other than its certificates of authentication.

#### 7.5. NOTICE OF DEFAULT.

If a Default or an Event of Default occurs and is continuing with respect to the Securities of any Series, and if it is known to a Responsible Officer of the Trustee, the Trustee shall deliver to each Securityholder of the Securities of that Series notice of the Default or the Event of Default, as the case may be, within 90 days after it occurs or, if later, after a Responsible Officer of the Trustee has knowledge of such Default or Event of Default (except if such Default or Event of Default has been validly cured or waived before the giving of such notice). Except in the case of a Default or an Event of Default in payment of the principal of, or interest or premium, if any, on, any Security of any Series, the Trustee may withhold the notice if and so long as it determines in good faith determine(s) that withholding the notice is in the interests of the Securityholders of that Series

## 7.6. REPORTS BY TRUSTEE TO HOLDERS.

If and to the extent required by the TIA, within 60 days after May 15 of each year, commencing the May 15 following the date of this Indenture, the Trustee shall deliver to each Securityholder a brief report dated as of such May 15 that complies with TIA Section 313(a). The Trustee also shall comply with TIA Sections 313(b) and 313(c).

A copy of each report at the time of its delivering to Securityholders shall be filed with the SEC and any stock exchange on which the Securities of that Series are listed. The Company shall promptly notify the Trustee in writing when the Securities of any Series are listed on any stock exchange or any delisting thereof, and the Trustee shall comply with TIA Section 313(d).

## 7.7. COMPENSATION AND INDEMNITY.

The Company shall pay to the Trustee from time to time compensation for its services as the Company and the Trustee shall from time to time agree in writing. The Trustee's compensation shall not be limited by any provision of law on compensation of a trustee of an express trust. The Company shall reimburse the Trustee within 45 days after receipt of request for all reasonable disbursements, expenses and advances incurred or made by it in connection with its duties under this Indenture, including the reasonable compensation, disbursements and expenses of the Trustee's agents and counsel.

The Company shall indemnify the Trustee (in any of its capacities), its officers, direction, employees and agents for, and hold it harmless against, any and all loss, liability or expense incurred by it in connection with the acceptance or performance of its duties under this Indenture including the reasonable costs and expenses of enforcing this Indenture and defending itself against any claim or liability in connection with the exercise or performance of any of its powers or duties hereunder. The Trustee, upon receiving written notice thereof, shall notify the Company promptly of any claim asserted against the Trustee for which it may seek indemnity.

The failure by the Trustee to so notify the Company shall not however relieve the Company of its obligations. Notwithstanding the foregoing, the Company need not reimburse the Trustee for any expense or indemnify it against any loss or liability incurred by the Trustee through its gross negligence or willful misconduct, as established by a final, non-appealable order of a court of competent jurisdiction. To secure the payment obligations of the Company in this Section 7.7, the Trustee shall have a lien prior to the Securities of any Series on all money or property held or collected by the Trustee except such money or property held in trust to pay the principal of, interest and premium, if any, on particular Securities of that Series.

When the Trustee incurs expenses or renders services after an Event of Default specified in Section 6.1(4) or (5) occurs, the expenses and the compensation for the services are intended to constitute expenses of administration under any Bankruptcy Law.

For purposes of this Section 7.7, the term "Trustee" shall include any trustee appointed pursuant to this Article 7.

The obligations of the Company under this Section 7.7 shall survive the termination or satisfaction and discharge of this Indenture or the resignation or removal of the Trustee for any reason

#### 7.8. REPLACEMENT OF TRUSTEE.

The Trustee may resign with respect to the Securities of one or more Series by so notifying the Company in writing at least 90 days in advance of such resignation.

The Holders of a majority in principal amount of the outstanding Securities of any Series may remove the Trustee with respect to that Series by notifying the removed Trustee in writing and may appoint a successor Trustee with respect to that Series with the consent of the Company, which consent shall not be unreasonably withheld. The Company may remove the Trustee with respect to that Series at its election if:

- (1) the Trustee fails to comply with, or ceases to be eligible under, Section 7.10;
- (2) the Trustee is adjudged a bankrupt or an insolvent, or an order for relief is entered with respect to the Trustee, under any Bankruptcy Law;
- (3) a Custodian or other public officer takes charge of the Trustee or its property; or
- (4) the Trustee otherwise becomes incapable of acting.
- (5) If the Trustee resigns or is removed, or if a vacancy exists in the office of Trustee, with respect to any Series of Securities for any reason, the Company shall promptly appoint, by Board Resolution, a successor Trustee.

If a successor Trustee with respect to the Securities of one or more Series does not take office within 60 days after the retiring Trustee resigns or is removed, the retiring Trustee, the Company or the Holders of at least 10% in principal amount of the outstanding Securities of the applicable Series may petition any court of competent jurisdiction for the appointment of a successor Trustee.

If the Trustee with respect to the Securities of one or more Series fails to comply with Section 7.10, any Securityholder of the applicable Series may petition any court of competent jurisdiction for the removal of the Trustee and the appointment of a successor Trustee.

A successor Trustee shall deliver a written acceptance of its appointment to the retiring Trustee and to the Company. Immediately following such delivery, (i) the retiring Trustee with respect to one or more Series shall, subject to its rights under Section 7.7, transfer all property held by it as Trustee with respect to such Series to the successor Trustee, (ii) the resignation or removal of the retiring Trustee shall become effective and (iii) the successor Trustee with respect to such Series shall have all the rights, powers and duties of the Trustee under this Indenture. A successor Trustee with respect to the Securities of one or more Series shall deliver notice of its succession to each Securityholder of such Series.

A retiring Trustee shall not be liable for the acts or omissions of any successor Trustee after its succession.

Notwithstanding replacement of the Trustee pursuant to this section 7.8, the Company's obligations under Section 7.7 shall continue for the benefit of the retiring Trustee

## 7.9. SUCCESSOR TRUSTEE BY CONSOLIDATION, MERGER OR CONVERSION.

If the Trustee, or any Agent, consolidates with, merges or converts into, or transfers all or substantially all of its corporate trust assets to, another entity, subject to Section 7.10, the successor corporation without any further act shall be the successor Trustee or Agent, as the case may be.

## 7.10. ELIGIBILITY; DISQUALIFICATION.

This Indenture shall always have a Trustee who satisfies the requirements of TIA Sections 310(a)(1), (2) and (5) in every respect. The Trustee (or in the case of a Trustee that is a Person included in a bank holding company system, the related bank holding company) shall have a combined capital and surplus of at least \$100,000,000 as set forth in its most recent published annual report of condition. The Trustee shall comply with TIA Section 310(b), including the provision in Section 310(b)(1). In addition, if the Trustee is a Person included in a bank holding company system, the Trustee, independently of such bank holding company, shall meet the capital requirements of TIA Section 310(a)(2). If at any time the Trustee shall cease to be eligible in accordance with the provisions of this Section 7.10, it shall resign immediately in the manner and with the effect specified in this Article 7.

## 7.11. PREFERENTIAL COLLECTION OF CLAIMS AGAINST COMPANY.

The Trustee shall comply with TIA Section 311(a), excluding any creditor relationship listed in TIA Section 311(b). A Trustee who has resigned or been removed shall be subject to TIA Section 311(a) to the extent indicated therein.

## 7.12. PAYING AGENTS.

The Company shall cause each Paying Agent other than the Trustee to execute and deliver to it and the Trustee an instrument in which such agent shall agree with the Trustee, subject to the provisions of this Section 7.12:

- (1) that it will hold all sums held by it as agent for the payment of the principal of, or interest or premium, if any, on, the Securities (whether such sums have been paid to it by the Company or by any obligor on the Securities) in trust for the benefit of Holders of the Securities or the Trustee;
- (2) that it will at any time during the continuance of any Event of Default, upon written request from the Trustee, deliver to the Trustee all sums so held in trust by it together with a full accounting thereof; and
- (3) that it will give the Trustee written notice within three Business Days after any failure of the Company (or by any obligor on the Securities) in the payment of any installment of the principal of, or interest or premium, if any, on, the Securities when the same shall be due and payable.

# ARTICLE 8 – AMENDMENTS, SUPPLEMENTS AND WAIVERS

## 8.1. WITHOUT CONSENT OF HOLDERS.

The Company, when authorized by a Board Resolution, and the Trustee may amend, restate or supplement this Indenture or the Securities of one or more Series without notice to or consent of any Securityholder:

- (1) to comply with Section 5.1;
- (2) to provide for certificated Securities in addition to uncertificated Securities;
- (3) to comply with any requirements of the SEC under the TIA;
- (4) to cure any ambiguity, defect or inconsistency, or to make any other change herein or in the Securities that does not materially and adversely affect the rights of any Securityholder;
  - (5) to provide for the issuance of, and establish the form and terms and conditions of, Securities of any Series as permitted by this Indenture; or
- (6) to evidence and provide for the acceptance of appointment hereunder by a successor Trustee with respect to the Securities of one or more Series, and to add to or change any of the provisions of this Indenture as shall be necessary to provide for or facilitate the administration of the trusts hereunder by more than one Trustee.

The Trustee is hereby authorized to join with the Company in the execution of any supplemental indenture authorized or permitted by the terms of this Indenture, and to make any further appropriate agreements and stipulations which may be therein contained, but the Trustee shall not be obligated to enter into any such supplemental indenture which adversely affects its own rights, duties or immunities under this Indenture.

## 8.2. WITH CONSENT OF HOLDERS.

- (a) The Company, when authorized by a Board Resolution, and the Trustee may amend, restate or supplement this Indenture or the Securities of one or more Series with the written consent of the Holders of not less than a majority in aggregate principal amount of the outstanding Securities of such Series affected by such amendment, restatement or supplement without notice to any Securityholder. The Holders of not less than a majority in aggregate principal amount of the outstanding Securities of each such Series affected by such amendment, restatement or supplement may waive compliance by the Company in a particular instance with any provision of this Indenture or the Securities of such Series without notice to any Securityholder. Subject to Section 8.4, without the consent of each Securityholder affected, however, an amendment, supplement or waiver may not:
  - (1) reduce the amount of Securities whose Holders must consent to an amendment, supplement or waiver to this Indenture or the Securities;
  - (2) reduce the rate of, or change the time for payment of, interest on any Security;
- (3) reduce the principal, or change the Stated Maturity, of any Security, or reduce the amount of, or postpone the date fixed for, the payment of any sinking fund or analogous obligation;
  - (4) make any Security payable in a currency other than that stated in the Security;

- (5) change the amount or time of any payment required by the Securities, or reduce the premium payable upon any redemption of the Securities, or change the time before which no such redemption may be made;
- (6) waive a Default or Event of Default in the payment of the principal of, or interest or premium, if any, on, any Security (except a rescission of acceleration of the Securities of any Series by the Holders of at least a majority in principal amount of the outstanding Securities of such Series and a waiver of the payment default that resulted from such acceleration);
  - (7) waive a redemption payment with respect to any Security, or change any of the provisions with respect to the redemption of any Securities;
  - (8) make any changes in Section 6.6 or this Section 8.2, except to increase any percentage of Securities the Holders of which must consent to any matter; or
  - (9) take any other action otherwise prohibited by this Indenture to be taken without the consent of each Holder affected thereby.
- (b) Upon the request of the Company, accompanied by a Board Resolution authorizing the execution of any such supplemental indenture, and upon the receipt by the Trustee of evidence reasonably satisfactory to the Trustee of the consent of the Securityholders as aforesaid and of the documents described in Section 8.6, the Trustee shall join with the Company in the execution of such supplemental indenture, unless such supplemental indenture affects the Trustee's own rights, duties or immunities under this Indenture, in which case the Trustee may in its discretion, but shall not be obligated to, enter into such supplemental indenture.
- (c) It shall not be necessary for the consent of the Holders under this section to approve the particular form of any proposed amendment, restatement, supplement or waiver, but it shall be sufficient if such consent approves the substance thereof.

After an amendment, restatement or supplement under this Section becomes effective, the Company shall deliver to Securityholders a notice briefly describing the amendment, restatement or supplement. Any failure of the Company to deliver any such notice, or any defect therein, shall not, however, in any way impair or affect the validity of any supplemental indenture.

#### 8.3. COMPLIANCE WITH TRUST INDENTURE ACT.

Every amendment to, or supplement of, this Indenture or the Securities shall comply with the TIA as then in effect.

## 8.4. REVOCATION AND EFFECT OF CONSENTS.

Until an amendment, restatement, supplement, waiver or other action becomes effective, a consent to it by a Holder of a Security is a continuing consent conclusive and binding upon such Holder and every subsequent Holder of the same Security or portion thereof, and of any Security issued upon the transfer thereof or in exchange therefor or in place thereof, even if notation of the consent is not made on any such Security. Any such Holder or subsequent Holder, however, may revoke the consent as to his Security or portion of a Security, if the Trustee receives the notice of revocation before the date the amendment, restatement, supplement, waiver or other action becomes effective.

The Company may, but shall not be obligated to, fix a record date for the purpose of determining the Holders entitled to consent to any amendment, restatement, supplement or waiver, which record date shall be at least 30 days prior to the first solicitation of such consent. If a record date is fixed, then, notwithstanding the preceding paragraph, those Persons who were Holders at such record date (or their duly designated proxies), and only such Persons, shall be entitled to consent to such amendment, restatement, supplement or waiver, or to revoke any consent previously given, whether or not such Persons continue to be Holders after such record date.

After an amendment, restatement, supplement, waiver or other action becomes effective, it shall bind every Securityholder, unless it makes a change described in any of clauses (1) through (9) of Section 8.2. In that case, the amendment, restatement, supplement, waiver or other action shall bind each Holder of a Security who has consented to it and every subsequent Holder of a Security or portion of a Security that evidences the same debt as the consenting Holder's Security; PROVIDED, that any such waiver shall not impair or affect the right of any Holder to receive payment of the principal of, and interest and premium, if any, on, a Security, on or after the respective due dates expressed in such Security, or to bring suit for the enforcement of any such payment on or after such respective dates without the consent of such Holder.

## 8.5. NOTATION ON OR EXCHANGE OF SECURITIES.

If an amendment, restatement, supplement or waiver changes the terms of a Security of any Series, the Trustee may request the Holder of such Security to deliver it to the Trustee. In such case, the Trustee shall place an appropriate notation on such Security about the changed terms and return it to the Holder. Alternatively, the Company, in exchange for such Security, may issue, and the Trustee shall authenticate, a new security that reflects the changed terms. Failure to make the appropriate notation or issue a new Security shall not affect the validity and effect of such amendment, restatement, supplement or waiver.

#### 8.6. TRUSTEE TO SIGN AMENDMENTS, ETC.

The Trustee shall sign any amendment, restatement, supplement or waiver authorized pursuant to this Article 8 if the amendment, restatement, supplement or waiver does not adversely affect the rights, duties, liabilities or immunities of the Trustee. If it does, the Trustee may, but need not, sign it. In signing or refusing to sign such amendment, restatement, supplement or waiver the Trustee shall be entitled to receive and, subject to Section 7.1, shall be fully protected in relying upon an Officers' Certificate and an Opinion of Counsel stating that such amendment, restatement or waiver is authorized or permitted by this Indenture. The Company may not sign an amendment, restatement or supplement until the Board of Directors of the Company approves it.

#### ARTICLE 9 – DISCHARGE OF INDENTURE; DEFEASANCE

## 9.1. DISCHARGE OF INDENTURE.

The Company may terminate its obligations under the Securities of any Series and this Indenture with respect to such Series, except the obligations referred to in the last paragraph of this Section 9.1, if there shall have been canceled by the Trustee, or delivered to the Trustee for cancellation, all Securities of such Series theretofore authenticated and delivered (other than any Securities of such Series that are asserted to have been destroyed, lost or stolen and that shall have been replaced as provided in Section 2.8) and the Company has paid all sums payable by it hereunder or deposited all required sums with the Trustee.

After such delivery the Trustee upon request shall acknowledge in a writing prepared by or on behalf of the Company the discharge of the Company's obligations under the Securities of such Series and this Indenture, except for those surviving obligations specified below.

Notwithstanding the satisfaction and discharge of this Indenture, the obligations of the Company in Sections 7.7, 9.5 and 9.6 shall survive.

#### 9.2. LEGAL DEFEASANCE.

The Company may at its option, by Board Resolution, be discharged from its obligations with respect to the Securities of any Series on the date upon which the conditions set forth in Section 9.4 below are satisfied (hereinafter, "Legal Defeasance"). For this purpose, such Legal Defeasance means that the Company shall be deemed to have paid and discharged the entire Indebtedness represented by the Securities of such Series and to have satisfied all its other obligations under such Securities and this Indenture insofar as such Securities are concerned (and the Trustee, at the expense of the Company, shall, subject to Section 9.6, execute proper instruments acknowledging the same, as are delivered to it by the Company), except for the following, which shall survive until otherwise terminated or discharged hereunder: (A) the rights of Holders of outstanding Securities of such Series to receive solely from the trust funds described in Section 9.4 and as more fully set forth in such section, payments in respect of the principal of, and interest and premium, if any, on, the Securities of such Series when such payments are due, (B) the Company's obligations with respect to the Securities of such Series under Sections 2.4, 2.5, 2.6, 2.7, 2.8 and 2.9, (C) the rights, powers, trusts, duties and immunities of the Trustee hereunder (including claims of, or payments to, the Trustee under or pursuant to Section 7.7) and (D) this Article 9. Subject to compliance with this Article 9, the Company may exercise its option under this Section 9.2 with respect to the Securities of any Series notwithstanding the prior exercise of its option under Section 9.3 below with respect to the Securities of such Series.

## 9.3. COVENANT DEFEASANCE.

At the option of the Company, pursuant to a Board Resolution, the Company shall be released from its obligations with respect to the outstanding Securities of any Series under Sections 4.2 through 4.5, inclusive, and Section 5.1, with respect to the outstanding Securities of such Series, on and after the date the conditions set forth in Section 9.4 are satisfied (hereinafter, "Covenant Defeasance"). For this purpose, such Covenant Defeasance means that the Company may omit to comply with and shall have no liability in respect of any term, condition or limitation set forth in any such specified section or portion thereof, whether directly or indirectly by reason of any reference elsewhere herein to any such specified Section or portion thereof or by reason of any reference in any such specified section or portion thereof to any other provision herein or in any other document, but the remainder of this Indenture and the Securities of any Series shall be unaffected thereby.

## 9.4. CONDITIONS TO LEGAL DEFEASANCE OR COVENANT DEFEASANCE.

The following shall be the conditions to application of Section 9.2 or Section 9.3 to the outstanding Securities of a Series:

(1) the Company shall irrevocably have deposited or caused to be deposited with the Trustee (or another trustee satisfying the requirements of Section 7.10 who shall agree to comply with the provisions of this Article 9 applicable to it) as funds in trust for the purpose of making the following payments, specifically pledged as security for, and dedicated solely to, the benefit of the Holders of the Securities, (A) money in an amount, or (B) U.S. Government Obligations or Foreign Government Obligations which through the scheduled payment of principal and interest in respect thereof in accordance with their terms will provide, not later than the due date of any payment, money in an amount, or (C) a combination thereof, sufficient, in the opinion of a nationally recognized firm of independent public accountants expressed in a written certification thereof delivered to the Trustee, to pay and discharge, and

which shall be applied by the Trustee (or other qualifying trustee) to pay and discharge, the principal of, and accrued interest and premium, if any, on, the outstanding Securities of such Series at the Stated Maturity of such principal, interest or premium, if any, or on dates for payment and redemption of such principal, interest and premium, if any, selected in accordance with the terms of this Indenture and of the Securities of such Series;

- (2) no Event of Default or Default with respect to the Securities of such Series shall have occurred and be continuing on the date of such deposit, or shall have occurred and be continuing at any time during the period ending on the 91st day after the date of such deposit or, if longer, ending on the day following the expiration of the longest preference period under any Bankruptcy Law applicable to the Company in respect of such deposit as specified in the Opinion of Counsel identified in paragraph (8) below (it being understood that this condition shall not be deemed satisfied until the expiration of such period);
- (3) such Legal Defeasance or Covenant Defeasance shall not cause the Trustee to have a conflicting interest for purposes of the TIA with respect to any securities of the Company:
- (4) such Legal Defeasance or Covenant Defeasance shall not result in a breach or violation of, or constitute default under, any other agreement or instrument to which the Company is a party or by which it is bound;
- (5) the Company shall have delivered to the Trustee an Opinion of Counsel stating that, as a result of such Legal Defeasance or Covenant Defeasance, neither the trust nor the Trustee will be required to register as an investment company under the Investment Company Act of 1940, as amended;
- (6) in the case of an election under Section 9.2, the Company shall have delivered to the Trustee an Opinion of Counsel stating that (i) the Company has received from, or there has been published by, the Internal Revenue Service a ruling to the effect that or (ii) there has been a change in any applicable Federal income tax law with the effect that, and such opinion shall confirm that, the Holders of the outstanding Securities of such Series or Persons in their positions will not recognize income, gain or loss for Federal income tax purposes solely as a result of such Legal Defeasance and will be subject to Federal income tax on the same amounts, in the same manner, including as a result of prepayment, and at the same times as would have been the case if such Legal Defeasance had not occurred;
- (7) in the case of an election under Section 9.3, the Company shall have delivered to the Trustee an Opinion of Counsel to the effect that the Holders of the outstanding Securities of such Series will not recognize income, gain or loss for Federal income tax purposes as a result of such Covenant Defeasance, and will be subject to Federal income tax on the same amounts, in the same manner and at the same times as would have been the case if such Covenant Defeasance had not occurred;
- (8) the Company shall have delivered to the Trustee an Officers' Certificate and an Opinion of Counsel, each stating that all conditions precedent provided for in this Article 9 relating to either the Legal Defeasance under Section 9.2 or the Covenant Defeasance under Section 9.3 (as the case may be) have been complied with;
- (9) the Company shall have delivered to the Trustee an Officers' Certificate stating that the deposit under clause (1) was not made by the Company with the intent of defeating, hindering, delaying or defrauding any creditors of the Company or others; and
- (10) the Company shall have paid, or duly provided for payment under terms mutually satisfactory to the Company and the Trustee, all amounts then due to the Trustee pursuant to Section 7.7.

## 9.5. DEPOSITED MONEY AND U.S. AND FOREIGN GOVERNMENT OBLIGATIONS TO BE HELD IN TRUST; OTHER MISCELLANEOUS PROVISIONS.

All money, U.S. Government Obligations and Foreign Government Obligations (including the proceeds thereof) deposited with the Trustee pursuant to Section 9.4 in respect of the outstanding Securities shall be held in trust and applied by the Trustee, in accordance with the provisions of such Securities and this Indenture, to the payment, either directly or through any Paying Agent as the Trustee may determine, to the Holders of such Securities, of all sums due and to become due thereon in respect of principal, accrued interest and premium, if any, but such money need not be segregated from other funds except to the extent required by law.

The Company shall pay and indemnify the Trustee against any tax, fee or other charge imposed on or assessed against the U.S. Government Obligations and Foreign Government Obligations deposited pursuant to Section 9.4 or the principal, interest and premium, if any, received in respect thereof other than any such tax, fee or other charge which by law is for the account of the Holders of the outstanding Securities.

Anything in this Article 9 to the contrary notwithstanding, but subject to payment of any of its outstanding fees and expenses, the Trustee shall deliver or pay to the Company from time to time upon Company Request any money, U.S. Government Obligations or Foreign Government Obligations held by the Trustee as provided in Section 9.4 which, in the opinion of a nationally-recognized firm of independent public accountants expressed in a written certification thereof delivered to the Trustee, are in excess of the amount thereof which would then be required to be deposited to effect an equivalent Legal Defeasance or Covenant Defeasance.

## 9.6. REINSTATEMENT.

If the Trustee or Paying Agent is unable to apply any money, U.S. Government Obligations or Foreign Government Obligations in accordance with Section 9.1, 9.2, 9.3 or 9.4 by reason of any legal proceeding or by reason of any order or judgment of any court or governmental authority enjoining, restraining or otherwise prohibiting such application, the Company's obligations under this Indenture and the Securities shall be revived and reinstated as though no deposit had occurred pursuant to this Article 9 until such time as the Trustee or Paying Agent is permitted to apply all such money, U.S. Government Obligations or Foreign Government Obligations, as the case may be, in accordance with Section 9.1, 9.2, 9.3 or 9.4; PROVIDED, HOWEVER, that if the Company has made any payment of principal of, or accrued interest or premium, if any, on, any Securities because of the reinstatement of its obligations, the Company shall be subrogated to the rights of the Holders of such Securities to receive such payment from the money, U.S. Government Obligations or Foreign Government Obligations held by the Trustee or Paying Agent.

## 9.7. MONEYS HELD BY PAYING AGENT.

In connection with the satisfaction and discharge of this Indenture, all moneys then held by any Paying Agent under the provisions of this Indenture shall, upon demand of the Company, be paid to the Trustee, or, if sufficient moneys have been deposited pursuant to Section 9.1, be paid to the Company, and thereupon such Paying Agent shall be released from all further liability with respect to such moneys.

## 9.8. MONEYS HELD BY TRUSTEE.

Any moneys deposited with the Trustee or any Paying Agent or then held by the Company in trust for the payment of the principal of, or interest or premium, if any, on, any Security that are not applied but remain unclaimed by the Holder of such Security for two years after the date upon which the principal of, or interest or premium, if any, on, such Security shall have respectively become due and payable shall be repaid to the Company upon Company Request, or if such moneys are then held by the Company in trust, such moneys shall be released from such trust; and the Holder of such Security entitled to receive such payment shall thereafter, as an unsecured general creditor, look only to the Company for the payment thereof, and all liability of the Trustee or such Paying Agent with respect to such trust money shall thereupon cease. After payment to the Company or the release of any money held in trust by the Company, Securityholders entitled to the money must look only to the Company for payment as general creditors, unless applicable abandoned property law designates another Person.

#### ARTICLE 10 – MISCELLANEOUS

## 10.1. TRUST INDENTURE ACT CONTROLS.

If any provision of this Indenture limits, qualifies or conflicts with another provision which is required to be included in this Indenture by the TIA, the required TIA provision shall control. If any provision of this Indenture modifies or excludes any provision of the TIA which may be so modified or excluded, the latter provision shall be deemed to apply to this Indenture as so modified or to be excluded, as the case may be.

## 10.2. NOTICES.

Any notice or communication shall be given in writing and delivered in Person, sent by facsimile (and receipt confirmed by telephone or electronic transmission report), delivered by commercial courier service or mailed by first-class mail, postage prepaid, addressed as follows:

If to the Company:

Velo3D, Inc. 511 Division St. Campbell, CA 95008 Attention: Chief Financial Officer

Copy to:

Fenwick & West LLP 902 Broadway New York, NY 10010 Fax: [\*] Attention: Per Chilstrom

If to the Trustee:

U.S. Bank Trust Company, National Association 1 California Street, Suite 1000 San Francisco, CA 94111 Attn: D. Jason (Velo3D, Inc.) E-mail: [\*]

The Company or the Trustee by written notice to the other may designate additional or different addresses for subsequent notices or communications. Any notice or communication to the Company or the Trustee shall be deemed to have been given or made as of the date so delivered if personally delivered; when receipt is confirmed by telephone or electronic transmission report, if sent by facsimile; and three Business Days after mailing if sent by registered or certified mail, postage prepaid (except that a notice of change of address shall not be deemed to have been given until actually received by the addressee).

Any notice or communication mailed to a Securityholder shall be mailed to such Securityholder by first-class mail, postage prepaid, at such Securityholder's address shown on the register kept by the Registrar.

Failure to deliver, or any defect in, a notice or communication to a Securityholder shall not affect its sufficiency with respect to other Securityholders. If a notice or communication to a Securityholder is mailed in the manner provided above, it shall be deemed duly given, three Business Days after such mailing, whether or not the addressee receives it.

In case by reason of the suspension of regular mail service, or by reason of any other cause, it shall be impossible to mail any notice as required by this Indenture, then such method of notification as shall be made with the approval of the Trustee shall constitute a sufficient mailing of such notice.

If the Company delivers a notice or communication to Securityholders, it shall deliver a copy to the Trustee and each Agent at the same time.

In the case of Global Securities, notices or communications to be given to Securityholders shall be given to the Depository, in accordance with its applicable policies as in effect from time to time.

In addition to the manner provided for in the foregoing provisions, notices or communications to Securityholders may be given by the Company by release made to Reuters Economic Services and Bloomberg Business News.

## 10.3. COMMUNICATIONS BY HOLDERS WITH OTHER HOLDERS.

Securityholders of any Series may communicate pursuant to TIA Section 312(b) with other Securityholders of that Series or any other Series with respect to their rights under this Indenture or the Securities of that Series or any other Series. The Company, the Trustee, the Registrar and any other Person shall have the protection of TIA Section 312(c).

## 10.4. CERTIFICATE AND OPINION AS TO CONDITIONS PRECEDENT.

Upon any request or application by the Company to the Trustee to take any action under this Indenture, the Company shall furnish to the Trustee:

- (1) an Officers' Certificate (which shall include the statements set forth in Section 10.5 below) stating that, in the opinion of the signers, all conditions precedent, if any, provided for in this Indenture relating to the proposed action have been complied with; and
- (2) an Opinion of Counsel (which shall include the statements set forth in Section 10.5 below) stating that, in the opinion of such counsel, all such conditions precedent have been complied with.

## 10.5. STATEMENT REQUIRED IN CERTIFICATE AND OPINION.

Each certificate and opinion with respect to compliance with a condition or covenant provided for in this Indenture (other than pursuant to Section 4.4) shall include:

- (1) a statement that the Person making such certificate or opinion has read such covenant or condition;
- (2) a brief statement as to the nature and scope of the examination or investigation upon which the statements or opinions contained in such certificate or opinion are based;
- (3) a statement that, in the opinion of such Person, it or he has made such examination or investigation as is necessary to enable it or him to express an informed opinion as to whether or not such covenant or condition has been complied with: and
  - (4) a statement as to whether or not, in the opinion of such Person, such covenant or condition has been complied with.

## 10.6. RULES BY TRUSTEE AND AGENTS.

The Trustee may make reasonable rules for action by or at meetings of Securityholders. The Registrar, Paying Agent and Service Agent may make reasonable rules for their functions.

## 10.7. BUSINESS DAYS; LEGAL HOLIDAYS; PLACE OF PAYMENT.

A "Business Day" is a day that is not a Legal Holiday. A "Legal Holiday" is a Saturday, a Sunday, a federally-recognized holiday or a day on which banking institutions are not authorized or required by law, regulation or executive order to be open in the State of New York or a Place of Payment.

If a payment date is a Legal Holiday at a Place of Payment, payment may be made at that place on the next succeeding day that is not a Legal Holiday, and no interest shall accrue for the intervening period. "Place of Payment" means the place or places where the principal of, and interest and premium, if any, on, the Securities of a Series are payable as specified as contemplated by Section 2.2. If the regular record date is a Legal Holiday, the record date shall not be affected.

#### 10.8. GOVERNING LAW.

THIS INDENTURE AND THE SECURITIES SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, AS APPLIED TO CONTRACTS MADE AND PERFORMED WITHIN THE STATE OF NEW YORK WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAW.

## 10.9. NO ADVERSE INTERPRETATION OF OTHER AGREEMENTS.

This Indenture may not be used to interpret another indenture, loan, security or debt agreement of the Company or any Subsidiary thereof. No such indenture, loan, security or debt agreement may be used to interpret this Indenture.

#### 10.10. NO RECOURSE AGAINST OTHERS.

A director, officer, employee, stockholder or incorporator, as such, of the Company shall not have any liability for any obligations of the Company under the Securities or the Indenture. Each Securityholder by accepting a Security waives and releases all such liability. Such waiver and release are part of the consideration for the issuance of the Securities.

## 10.11. SUCCESSORS.

All covenants and agreements of the Company in this Indenture and the Securities shall bind the Company's successors and assigns, whether so expressed or not. All agreements of the Trustee, any additional trustee and any Paying Agents in this Indenture shall bind their respective successors and assigns.

#### 10.12. MULTIPLE COUNTERPARTS.

The parties may sign multiple counterparts (which may include counterparts delivered by any standard form of telecommunication) of this Indenture. Each signed counterpart shall be deemed an original, but all of them together represent one and the same agreement. Counterparts may be delivered via facsimile, electronic mail (including any electronic signature covered by the U.S. federal ESIGN Act of 2000, Uniform Electronic Transactions Act, the Electronics Signatures and Records Act or other applicable law, e.g., www.docusign.com) or other transmission method and any counterparty so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes. All notices, approvals, consents, requests and any communications hereunder must be in writing (provided that any such communication sent to the Trustee hereunder must be in the form of a document that is signed manually or by way of a digital signature provided by DocuSign (or such other digital signature provider as specified in writing to the Trustee by the authorized representative)), in English. The Company agrees to assume all risks arising out of the use of using digital signatures and electronic methods to submit communications to the Trustee, including without limitation the risk of the Trustee acting on unauthorized instructions, and the risk of interception and misuse by third parties.

## 10.13. TABLE OF CONTENTS, HEADINGS, ETC.

The table of contents, cross-reference sheet and headings of the Articles and Sections of this Indenture have been inserted for convenience of reference only, are not to be considered a part hereof, and shall in no way modify or restrict any of the terms or provisions hereof.

## 10.14. SEVERABILITY.

Each provision of this Indenture shall be considered separable, and if for any reason any provision which is not essential to the effectuation of the basic purpose of this Indenture or the Securities shall be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby, and a Holder shall have no claim therefor against any party hereto.

#### 10.15. SECURITIES IN A FOREIGN CURRENCY OR IN EUROS.

Unless otherwise specified in a Board Resolution, a supplemental indenture hereto or an Officers' Certificate delivered pursuant to Section 2.2 with respect to a particular Series of Securities, whenever for purposes of this Indenture any action may be taken by the Holders of a specified percentage in aggregate principal amount of Securities of all Series affected by a particular action at the time outstanding and, at such time, there are outstanding Securities of any Series which are denominated in a coin or currency other than Dollars (including Euros), then the principal amount of Securities of such Series which shall be deemed to be outstanding for the purpose of taking such action shall be that amount of Dollars that could be obtained for such amount at the Market Exchange Rate at such time. For purposes of this Section 10.15, "Market Exchange Rate at such time in New York City for cable transfers of that currency as published by the Federal Reserve Bank of New York; PROVIDED, HOWEVER, in the case of Euros, Market Exchange Rate shall mean the rate of exchange determined by the Commission of the European Union (or any successor thereto) as published in the Official Journal of the European Union (such publication or any successor publication, the "Journal"). If such Market Exchange Rate is not available for any reason with respect to such currency, the Trustee shall use, in its sole discretion and without liability on its part, such quotation of the Federal Reserve Bank of New York or, in the case of Euros, the rate of exchange as published in the Journal, as of the most recent available date, or quotations or, in the case of Euros, rates of exchange as the Trustee, upon consultation with the Company, shall deem appropriate. The provisions of this paragraph shall apply in determining the equivalent principal amount in respect of Securities of a Series denominated in currency other than Dollars in connection with any action taken by Holders of Securities pursuant to the terms of this Indenture.

All decisions and determinations of the Trustee regarding the Market Exchange Rate or any alternative determination provided for in the preceding paragraph shall be in the Trustee's sole discretion, and shall, in the absence of manifest error, be conclusive to the extent permitted by law for all purposes and irrevocably binding upon the Company and all Holders.

## 10.16. JUDGMENT CURRENCY.

The Company agrees, to the fullest extent that it may effectively do so under applicable law, that (a) if for the purpose of obtaining judgment in any court it is necessary to convert the sum due in respect of the principal of, or interest or premium, if any, or other amount on, the Securities of any Series (the "Required Currency") into a currency in which a judgment will be rendered (the "Judgment Currency"), the rate of exchange used shall be the rate at which, in accordance with normal banking procedures, the Trustee could purchase in The City of New York the Required Currency with the Judgment Currency on the day on which final unappealable judgment is entered, unless such day is not a Business Day, in which instance, the rate of exchange used shall be the rate at which, in accordance with normal banking procedures, the Trustee could purchase in The City of New York the Required Currency with the Judgment Currency on the Business Day preceding the day on which final unappealable judgment is entered and (b) its obligations under this Indenture to make payments in the Required Currency (i) shall not be discharged or satisfied by any tender or any recovery pursuant to any judgment (whether or not entered in accordance with subsection (a)) in any currency other than the Required Currency, except to the extent that such tender or recovery shall result in the actual receipt, by the payee, of the full amount of the Required Currency expressed to be payable in respect of such payments, (ii) shall be enforceable as an alternative or additional cause of action for the purpose of recovering in the Required Currency the amount, if any, by which such actual receipt shall fall short of the full amount of the Required Currency so expressed to be payable and (iii) shall not be affected by judgment being obtained for any other sum due under this Indenture.

# 10.16. FORCE MAJEURE.

In no event shall the Trustee be responsible or liable for any failure or delay in the performance of its obligations hereunder arising out of or caused by, directly or indirectly, forces beyond its control, including, without limitation, strikes, work stoppages, accidents, acts of war or terrorism, civil or military disturbances, epidemic, pandemic, nuclear or natural catastrophes or acts of God, and interruptions, loss or malfunctions of utilities, communications or computer (software and hardware) services; provided that the Trustee shall use reasonable efforts consistent with accepted practices in the banking industry to resume performance as soon as practicable under the circumstances.

## 10.17. WAIVER OF JURY TRIAL.

EACH OF THE COMPANY AND THE TRUSTEE HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY PPLICABLE LAW, ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS INDENTURE, THE SECURITIES OR THE TRANSACTIONS CONTEMPLATED HEREBY.

## 10.18 U.S.A PATRIOT ACT.

The parties hereto acknowledge that in accordance with Section 326 of the U.S.A. Patriot Act, the Trustee, like all financial institutions, in order to help fight the funding of terrorism and money laundering, is required to obtain, verify and record information that identifies each person or legal entity that establishes a Relationship or opens an account with the Trustee. The parties to this Indenture agree that they will provide the Trustee with such information as it may reasonably request in order for the Trustee to satisfy the requirements of the U.S.A. Patriot Act.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

all as o	IN WITNESS WHEREOF, the parties hereto have caused this Indenture to be duly executed, and their respective corporate seals to be hereunto affixed and attested, f the day and year first above written.
VELO	3D, INC.
By:	/s/ Benny Buller
	Name: Benny Buller
	Title: Chief Executive Officer

U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION, as Trustee

By: /s/ David Jason

Name: David Jason Title: Vice President

[Signature Page to Indenture]

VELO3D, INC.,
Company
TO
U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION,
Trustee
SUPPLEMENTAL INDENTURE
Dated as of August 14, 2023
Supplement to Indenture Dated
as of August 14, 2023
Senior Convertible Note

SUPPLEMENTAL INDENTURE, dated as of August 14, 2023 (this "Supplemental Indenture") to the Indenture dated as of August 14, 2023, (the "Indenture"), between Velo3D, Inc., a Delaware corporation (herein called the "Company"), having its registered office at 511 Division Street., Campbell, California 95008, and U.S. Bank Trust Company, National Association, a national banking association, as trustee (the "Trustee"). Capitalized terms used but not otherwise defined herein shall have the meanings given to them in the Indenture.

## RECITALS OF THE COMPANY

WHEREAS, the Company has heretofore made, executed and delivered to the Trustee the Indenture, providing for the issuance from time to time of the Company's debt securities to be issued in one or more series;

WHEREAS, pursuant to the terms of the Indenture and this Supplemental Indenture, the Company has duly authorized the creation and issuance of its Senior Convertible Note due 2026 (the "Note") in an aggregate principal amount not to exceed \$105,000,000;

WHEREAS, Section 8.1 of the Indenture provides that, without the consent of any Securityholders, the Company, when authorized by a Board Resolution, and the Trustee may amend, restate or supplement the Indenture or the Securities of one or more Series to provide for the issuance of, and establish the form and terms and conditions of, Securities of any Series as permitted by the Indenture;

WHEREAS, this Supplemental Indenture is authorized by the resolutions adopted by the Board of Directors of the Company at the meeting duly called and held on August 9, 2023 and all requirements necessary to make this Supplemental Indenture a valid instrument in accordance with its terms and to make the Note, when executed by the Company and authenticated and delivered by the Trustee or an authenticating agent, the valid obligation of the Company, have been done and performed, and the execution and delivery of this Supplemental Indenture has been duly authorized in all respects;

## NOW, THEREFORE, THIS SUPPLEMENTAL INDENTURE WITNESSETH

For and in consideration of the provisions set forth herein, it is mutually agreed, for the equal and proportionate benefit of all Securityholders, from time to time, as follows:

## 1. Establishment

There is hereby established a new series of Securities to be issued under the Indenture and this Supplemental Indenture, to be designated as the Note. The Note shall be in substantially the form set forth in  $\underline{\text{Annex A}}$  hereto.

Upon execution of this Supplemental Indenture, the Note in an initial aggregate principal amount of \$70,000,000 shall be executed by the Company and delivered to the Trustee, and the Trustee shall thereupon authenticate and deliver such Note in accordance with a written order of the Company. The Company shall also be entitled to issue additional Notes in an aggregate principal amount up to \$35,000,000 which shall have identical terms as the initial Note issued on the date hereof, other than with respect to the date of issuance.

#### 2. Covenants

In addition to the covenants set forth in Articles 4 and 5 of the Indenture, the covenants that apply to the Note are set forth in the form of Note attached annex A hereto. To the extent of any inconsistency between, on the one hand, the Indenture and this Supplemental Indenture and on the other hand, the attached form of Note, the attached form of Note will modify or amend any provision of this Supplemental Indenture and the Indenture in so far as it applies to this series of Securities.

## 3. Miscellaneous Provisions

Other Terms of Indenture. Except insofar as herein otherwise expressly provided, all provisions, terms and conditions of the Indenture are in all respects ratified and confirmed and shall remain in full force and effect

Terms Defined. All terms defined in the Indenture shall have the same meanings when used herein.

Separability Clause. In case any provision in this Supplemental Indenture shall be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby.

Governing Law. This Supplemental Indenture shall be governed by and construed in accordance with the law of the State of New York.

Further Instruments. The parties hereto will execute and deliver such further instruments and do such further acts and things as may reasonably be required to carry out the intent and purpose of this Supplemental Indenture.

Trust Indenture Act. If any provision of this Supplemental Indenture limits, qualifies or conflicts with another provision that is required to be included in this Supplemental Indenture by the Trust Indenture Act of 1939, as amended, the required provision shall control.

Counterparts. This Supplemental Indenture may be executed in any number of counterparts, each of which shall be an original; but such counterparts shall together constitute one and the same instrument. The exchange of copies of this Supplemental Indenture and of signature pages by facsimile or PDF transmission shall constitute effective execution and delivery of this Supplemental Indenture as to the parties hereto and may be used in lieu of the original Supplemental Indenture for all purposes. Signatures of the parties hereto transmitted by facsimile or PDF shall be deemed to be their original signatures for all purposes. The Trustee agrees to accept and act upon notice, instructions or directions pursuant to this Indenture sent by unsecured e-mail, facsimile transmission or other similar unsecured electronic methods. If the party elects to give the Trustee e-mail or facsimile instructions (or instructions by a similar electronic method) and the Trustee in its discretion elects to act upon such instructions, the Trustee's understanding of such instructions shall be deemed controlling. The Trustee shall not be liable for any losses, costs or expenses arising directly or indirectly from the Trustee's reliance upon and compliance with such instructions notwithstanding such instructions conflict or are inconsistent with a subsequent written instruction. The party providing electronic instructions agrees to assume all risks arising out of the use of such electronic methods to submit instructions to the Trustee, including without limitation the risk of the Trustee acting on unauthorized instructions, and the risk or interception and misuse by third parties.

The Trustee. The Trustee makes no representation or warranty as to the validity or sufficiency of this Supplemental Indenture or with respect to the recitals contained herein, all of which recitals are made solely by the other parties hereto.

[Signature Page Follows]

IN WITNESS WHEREOF the parties hereto have caused this Supplemental Indenture to be duly executed, as of the day and year first above written.

VELO3D, INC.

By: /s/ Benyamin Buller

Name: Benyamin Buller Title: Chief Executive Officer

U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION, as Trustee

By: /s/ David Jason

Name: David Jason Title: Vice President

[Signature Page to Supplemental Indenture]

# Velo3D, Inc.

Form of Senior Secured Convertible Note due 2026

## Velo3D, Inc.

## Senior Secured Convertible Note due 2026

# Certificate No. A-[ ]

Velo3D, Inc., a Delaware corporation (the "Company"), for value received, promises to pay to [] (the "Initial Holder"), or its registered assigns, one hundred fifteen percent (115%) of the principal sum of [] (\$[]) (such principal sum, the "Principal Amount," and one hundred fifteen percent (115%) of such Principal Amount, the "Maturity Principal Amount;" provided that each of the Principal Amount and the Maturity Principal Amount is subject to reduction as provided herein) on August 1, 2026, and to pay any outstanding interest thereon, as provided in this Note, in each case, as provided in and subject to the other provisions of this Note, including the earlier redemption, repurchase or conversion of this Note.

This Note has been issued pursuant to the Indenture, dated as of [ ], 2023 (the "Base Indenture"), by and between the Company and U.S. Bank Trust Company, National Association, a national banking association, as trustee (the "Trustee"), as amended and supplemented by the First Supplemental Indenture, dated as of [ ], 2023 (the "Supplemental Indenture" and, the Base Indenture as amended and supplemented by the Supplemental Indenture, the "Indenture"), by and between the Company and the Trustee. The terms of this Note include those stated in the Indenture.

Unless otherwise indicated, references herein to "dollars" or "\$" are to U.S. dollars.

Additional provisions of this Note are set forth on the other side of this Note.

[The Remainder of This Page Intentionally Left Blank; Signature Page Follows]

IN WITNESS WHEREOF	, Velo3D, I	nc. has caused	d this instrument to	o be duly	v executed as of the	date set forth below.
--------------------	-------------	----------------	----------------------	-----------	----------------------	-----------------------

Velo3D, Inc.

Date: [ ] By:	Name: [ ] Title: [ ]
TRU	USTEE'S CERTIFICATE OF AUTHENTICATION
Dated as of [ ]	
U.S. BANK TRUST COMPANY, NATIONAL named Indenture.	ASSOCIATION, as Trustee, certifies that this is one of the Notes described in the within-
Date: [ ] By:	Authorized Signatory
(Signature Page	to Senior Convertible Note due 2026, Certificate No. A-[ ])

## Velo3D, Inc.

#### Senior Secured Convertible Note due 2026

This Note (this "Note") is issued by Velo3D, Inc., a Delaware corporation (the "Company"), and is one of a series of Notes that may be authenticated and delivered for original issue under the Indenture referred to below and pursuant to the Securities Purchase Agreement referred to below in an aggregate principal amount not to exceed one hundred five million dollars \$105,000,000 and designated as the Company's "Senior Secured Convertible Notes due 2026" (collectively, the "Notes")

This Note is subject to the terms of the Indenture.

#### Section 1. Definitions.

Capitalized terms used but not defined herein shall have the meanings set forth in the Indenture to the extent defined therein.

- "Affiliate" has the meaning set forth in Rule 144 under the Securities Act.
- "ATM Issuance" means an Equity Issuance made pursuant to an ATM Program, other than an Equity Issuance in which a single identified investor or group of identified investors purchases in excess of three million dollars (\$3,000,000) in the aggregate of Common Stock or warrants exercisable for any class of Common Stock in connection with any such Equity Issuance.
  - "ATM Program" means an "at-the-market" offering within the meaning of Rule 415(a)(4) of the Securities Act.
- "Attribution Parties" means, collectively, the following Persons and entities: (i) any investment vehicle, including, any funds, feeder funds or managed accounts, currently, or from time to time after the Issue Date, directly or indirectly managed or advised by the Holder's investment manager or any of its Affiliates or principals, (ii) any direct or indirect Affiliates of the Holder or any of the foregoing, (iii) any Person acting or who could be deemed to be acting as a "group" (within the meaning of Section 13(d)(3) of the Exchange Act) together with the Holder or any of the foregoing and (iv) any other Persons whose beneficial ownership of the Common Stock would or could be aggregated with the Holder's and the other Attribution Parties for purposes of Section 13(d) of the Exchange Act. For clarity, the purpose of the foregoing is to subject collectively the Holder and all other Attribution Parties to the Maximum Percentage.
- "Authorized Denomination" means, with respect to the Notes, a principal amount thereof equal to \$1,000 or any integral multiple of \$1,000 in excess thereof, or, if such principal amount then-outstanding is less than \$1,000, then such outstanding principal amount.
  - "Bankruptcy Law" means Title 11, United States Code, or any similar U.S. federal or state or non-U.S. law for the relief of debtors.
- "Board of Directors" means the board of directors of the Company or a committee of such board duly authorized to act on behalf of such board.
  - "Business Combination Event" has the meaning set forth in Section 9.
- "Business Day" means any day other than a Saturday, a Sunday or any day on which commercial banks in The City of New York are authorized or required by law or executive order

 $\|$ 

to close or be closed, or, with respect to any payment on this Note, the Place of Payment; provided, however, for clarification, commercial banks in The City of New York shall not be deemed to be authorized or required by law or executive order to close or be closed due to "stay at home", "shelter-in-place", "non-essential employee" or any other similar orders or restrictions or the closure of any physical branch locations at the direction of any governmental authority so long as the electronic funds transfer systems (including for wire transfers) of commercial banks in The City of New York are open for use by customers on such day.

"Capital Lease" means, with respect to any Person, any leasing or similar arrangement conveying the right to use any property, whether real or personal property, or a combination thereof, by that Person as lessee that, in conformity with GAAP, is required to be accounted for as a capital lease on the balance sheet of such Person.

"Capital Lease Obligation" means, at the time any determination is to be made, the amount of the liability in respect of a Capital Lease that would at that time be required to be capitalized on a balance sheet prepared in accordance with GAAP, and the stated maturity thereof shall be the date of the last payment of rent or any other amount due under such lease prior to the first date upon which such lease may be prepaid by the lessee without payment of a penalty.

"Capital Stock" of any Person means any and all shares of, interests in, rights to purchase, warrants or options for, participations in, or other equivalents of, in each case however designated, the equity of such Person, but excluding any debt securities convertible into such equity.

"Cash" means all cash and liquid funds.

"Cash Equivalents" means, as of any date of determination, any of the following: (i) securities issued or directly and fully guaranteed or insured by the government of the United States, Canada, Japan, the United Kingdom, Switzerland or any country that is a member of the European Union or any agency or instrumentality thereof, in each case maturing not more than two years from the date of acquisition; (ii) certificates of deposit, time deposits and eurodollar time deposits with maturities of one year or less from the date of acquisition, bankers' acceptances with maturities not exceeding one year and overnight bank deposits, in each case with any commercial bank having capital and surplus in excess of \$250,000,000 and whose long-term debt is rated "A" or the equivalent thereof by Moody's or S&P (or reasonably equivalent ratings of another internationally recognized rating agency); (iii) repurchase obligations for underlying securities of the types described in clauses (i) and (ii) above entered into with any financial institution meeting the qualifications specified in clause (ii) above; (iv) commercial paper issued by a Person (other than an Affiliate of the Company) rated at least "A-1" or the equivalent thereof by Moody's or S&P (or reasonably equivalent ratings of another internationally recognized rating agency), and in each case maturing within one year after the date of acquisition; (v) readily marketable direct obligations issued by any state or commonwealth of the United States of America or the District of Columbia or any political subdivision thereof having one of the two highest rating categories obtainable from either Moody's or S&P (or reasonably equivalent ratings of another internationally recognized rating agency), in each case with maturities not exceeding two years from the date of acquisition; (vi) Indebtedness issued by Persons (other than an Affiliate of the Company) with a rating of "A" or higher from S&P or "A-2" or higher from Moody's (or reasonably equivalent ratings of another internationally recognized rating agency), in each case with maturities not exceeding two years from the date of acquisition; and (vii) investment funds investing at least 95% of their assets in Cash and securities of the types described in clauses (i) through (vi) above.

- "Cash Sweep Amount" means, with respect to any Cash Sweep Financing, fifteen percent (15.0%) of the gross proceeds from such financing; *provided*, *however*, that in no event shall the aggregate amount of the gross proceeds of any Cash Sweep Financing that the Company is required to redeem in respect of this Note and all Other Notes exceed fifteen percent (15.0%) of the gross proceeds of such Cash Sweep Financing.
  - "Cash Sweep Certification" has the meaning set forth in Section 4(D)(ii).
- "Cash Sweep Financing" means any Equity Issuance other than (i) issuances of solely Common Stock at a price per share above the Conversion Price and (ii) ATM Issuances; provided, however, that the issuance of shares of Common Stock pursuant to the Notes shall not constitute a Cash Sweep Financing.
  - "Cash Sweep Notice" has the meaning set forth in Section 4(D)(iii).
  - "Cash Sweep Payment" has the meaning set forth in Section 4(D)(i).
  - "Collateral" has the meaning set forth in Section 8(Y).
- "Collateral Agent" means High Trail Investments ON LLC in its capacity as collateral agent for the Trustee, the Holder and each Other Holder, together with any successor thereto in such capacity.
  - "Close of Business" means 5:00 p.m., New York City time.
  - "Commission" means the U.S. Securities and Exchange Commission.
  - "Common Stock" means the common stock, par value \$0.00001 per share, of the Company, subject to Section 7(1).
  - "Common Stock Change Event" has the meaning set forth in Section 7(I)(i)(4).
  - "Compliance Certificate" has the meaning set forth in Section 8(J)(ii).
- "Contingent Obligation" means, as applied to any Person, any direct or indirect liability, contingent or otherwise, of that Person with respect to (A) any Indebtedness or other obligations of another Person, including any such obligation directly or indirectly guaranteed, endorsed, co-made or discounted or sold with recourse by that Person, or in respect of which that Person is otherwise directly or indirectly liable; (B) any obligations with respect to undrawn letters of credit, corporate credit cards or merchant services issued for the account of that Person; and (C) all obligations arising under any interest rate, currency or commodity swap agreement, interest rate cap agreement, interest rate collar agreement, or other agreement or arrangement designated to protect a Person against fluctuation in interest rates, currency exchange rates or commodity prices; provided, however, that the term "Contingent Obligation" shall not include endorsements for collection or deposit in the ordinary course of business. The amount of any Contingent Obligation shall be deemed to be an amount equal to the stated or determined amount of the primary obligation in respect of which such Contingent Obligation is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by such Person in good faith; provided, however, that such amount shall not in any event exceed the maximum amount of the obligations under the guarantee or other support arrangement.
  - "Conversion Consideration" has the meaning set forth in Section 7(E)(i).

- "Conversion Date" means the first Business Day on which the requirements set forth in Section 7(C)(i) or Section 7(D)(i) (as applicable) to convert this Note are satisfied.
- "Conversion Price" means, as of any time, an amount equal to (A) one thousand dollars (\$1,000) divided by (B) the Conversion Rate in effect at such time.
- "Conversion Rate" initially means 475.1722 [NTD: with respect to any Subsequently Purchased Notes, such amount shall be after giving effect to the adjustment contained in the proviso following immediately after this bracketed language] shares of Common Stock per \$1,000 Principal Amount of Notes[; provided, however, that the Conversion Rate shall adjust to an amount (rounded to the nearest fourth decimal place) equal to a fraction (1) whose numerator is \$1,000; and (2) whose denominator is one hundred ten percent (110%) of the average of the three Daily VWAPs during the three Trading Day period beginning on August 15, 2023 and ending on and including August 17, 2023 in the event that such amount calculated with respect to such three (3) Trading Day average is higher than the Conversion Rate then in effect; provided, further, that the Conversion Rate will not be adjusted pursuant to the foregoing proviso to an amount that would result in the Conversion Price per share of Common Stock being less than the par value per share of Common Stock]; provided, [however/further], that the Conversion Rate is subject to adjustment pursuant to Section 7; provided, further, that whenever this Note refers to the Conversion Rate as of a particular date without setting forth a particular time on such date, such reference will be deemed to be to the Conversion Rate immediately after the Close of Business on such date.
  - "Conversion Settlement Date" has the meaning set forth in Section 7(E)(iii).
- "Copyright License" means any written agreement granting any right to use any Copyright or Copyright registration, now owned or hereafter acquired by the Company or in which the Company now holds or hereafter acquires any interest.
- "Copyrights" means all copyrights, whether registered or unregistered, held pursuant to the laws of the United States, any State thereof, or of any other country.
  - "Covering Price" has the meaning set forth in Section 7(E)(iv)(1).
- "Daily VWAP" means, for any VWAP Trading Day, the per share volume-weighted average price of the Common Stock on The New York Stock Exchange (or the principal, in terms of volume, Eligible Exchange on which the Common Stock is listed for trading) as displayed under the heading "Bloomberg VWAP" on Bloomberg page "VLD <EQUITY> VAP" (or, if such page is not available, its equivalent successor page) in respect of the period from the scheduled open of trading until the scheduled close of trading of the primary trading session on such VWAP Trading Day (or, if such volume-weighted average price is unavailable, the market value of one share of Common Stock on such VWAP Trading Day, determined, using a volume-weighted average price method, by a nationally recognized independent investment banking firm selected by the Company). The Daily VWAP will be determined without regard to after-hours trading or any other trading outside of the regular trading session.
  - "Default" means any event that is (or, after notice, passage of time or both, would be) an Event of Default.
  - "Default Interest" has the meaning set forth in Section 10(D).

 $<sup>^{1}</sup>$  NTD: include with respect to the Initial Purchased Note.

"Disqualified Stock" means, with respect to any Person, any Capital Stock that by its terms (or by the terms of any security into which it is convertible or for which it is exchangeable at the option of the holder) or upon the happening of any event:

- (A) matures or is mandatorily redeemable pursuant to a sinking fund obligation or otherwise;
- (B) is convertible or exchangeable for Indebtedness or Disqualified Stock (excluding Capital Stock convertible or exchangeable solely at the option of the Company or a Subsidiary of the Company; provided that any such conversion or exchange will be deemed an incurrence of Indebtedness or Disqualified Stock, as applicable); or
  - (C) is redeemable at the option of the holder thereof, in whole or in part,
- (D) in the case of each of clauses (A), (B) and (C), at any point prior to the one hundred eight first (181st) day after the Maturity Date.

"DTC" means The Depository Trust Company.

"Eligible Exchange" means any of The New York Stock Exchange, The NYSE American LLC, The Nasdaq Stock Market, The Nasdaq Capital Market, The Nasdaq Global Market or The Nasdaq Global Select Market (or any of their respective successors).

"Equity Conditions" will be deemed to be satisfied as of any date if all of the following conditions are satisfied as of such date and on each of the thirty (30) previous Trading Days: (A) the shares issuable pursuant to this Note are Freely Tradable; (B) the Holder is not in possession of any material non-public information provided by or on behalf of the Company; (C) the issuance of such shares will not be limited by Section 7(J); (D) the Company is in compliance with Section 7(E)(i) and such shares will satisfy Section 7(F)(ii); (E) no pending, proposed or intended Fundamental Change has occurred that has not been abandoned, terminated or consummated; (F) the daily dollar trading volume (as reported on Bloomberg) of the Common Stock on The New York Stock Exchange (or the principal, in terms of volume, Eligible Exchange on which the Common Stock is listed for trading) is not less than three million five hundred thousand dollars (\$3,500,000); and (G) no Default will have occurred and be continuing which has not been waived and no Event of Default will have occurred which has not been waived.

"Equity Interest" shall mean, with respect to any Person, any and all shares, interests, participations or other equivalents, including preferred stock or membership interests (however designated, whether voting or non-voting), of equity of such Person, including, if such Person is a partnership, partnership interests (whether general or limited) and including, without limitation, any "equity security" (as that term is defined under Rule 405 promulgated under the Securities Act), and any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, such partnership.

"Equity Issuance" shall mean (a) any issuance or sale by the Company or any of its Subsidiaries of any Equity Interests (including any Equity Interests issued upon exercise or conversion of any Equity Rights) or any Equity Rights, or (b) the receipt by the Company or any of its Subsidiaries of any capital contribution (whether or not evidenced by any Equity Interest issued by the recipient of such contribution), in each case for bona fide capital-raising purposes and other than (i) any issuance of Equity Interests upon the exercise of any Equity Rights outstanding as of the date hereof provided, that such issuance is made pursuant to the terms of such Equity Rights in effect on the date hereof and such Equity Rights are not amended to increase the number of such Equity Interests or to decrease the exercise price, exchange price or

conversion price of Equity Rights, (ii) Equity Interests issuable upon the exercise of any Equity Rights or upon the lapse of forfeiture restrictions on awards made pursuant to an Approved Stock Plan (as defined in the Securities Purchase Agreement) (including Equity Interests withheld by the Company for the purpose of paying on behalf of the holder thereof the exercise price of stock options or for paying taxes due as a result of such exercise, vesting, settlement or lapse of forfeiture restrictions) or (iii) Common Stock issuable upon the exercise of stock options or upon the lapse of forfeiture restrictions on awards made pursuant to, any stock option exchange program of the Company that is approved by the Board of Directors or the compensation committee thereof or the Company's stockholders, whether now in effect or hereafter implemented.

"Equity Rights" shall mean, with respect to any Person, any then-outstanding subscriptions, options, warrants, commitments, preemptive rights, convertible debt, or other equity-linked securities or agreements of any kind for the issuance or sale, of any additional Equity Interests of any class, or partnership or other ownership interests of any type in, such Person.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the regulations promulgated thereunder.

"Event of Default" has the meaning set forth in Section 10(A).

"Event of Default Acceleration Amount" means, with respect to the delivery of a notice pursuant to Section 10(B)(ii) declaring this Note to be due and payable immediately on account of an Event of Default, a cash amount equal to the greater of (A) the sum of (i) one hundred and ten percent (110%) of the Maturity Principal Amount of this Note excluding interest (or such lesser principal amount accelerated pursuant to such notice) and (ii) the accrued and unpaid interest on this Note and (B) the sum of (i) one hundred fifteen percent (115%) of the product of (a) the Conversion Rate in effect as of the Trading Day immediately preceding the date that the Holder delivers such notice pursuant to Section 10(B)(ii); (b) the total then outstanding Principal Amount (expressed in thousands) of this Note; and (c) the greater of (x) the highest Daily VWAP per share of Common Stock occurring during the five (5) consecutive VWAP Trading Days ending on, and including, the VWAP per share of Common Stock occurring during the five (5) consecutive VWAP Trading Days ending on, and including, the VWAP per share of Common Stock occurring during the five (5) consecutive VWAP Trading Days ending on, and including, the VWAP Trading Day immediately before the date the applicable Event of Default occurred and (ii) the accrued and unpaid interest on this Note.

"Event of Default Notice" has the meaning set forth in Section 10(C).

"Event of Default Stock Payment" has the meaning set forth in Section 5(C).

"Event of Default Stock Payment Date" means any date on which the Holder delivers an Event of Default Stock Payment Notice pursuant to Section 5(C) hereunder.

"Event of Default Stock Payment Delivery Date" has the meaning set forth in Section 5(C).

"Event of Default Stock Payment Notice" has the meaning set forth in Section 5(C).

"Ex-Dividend Date" means, with respect to an issuance, dividend or distribution on the Common Stock, the first date on which shares of Common Stock trade on the applicable exchange or in the applicable market, regular way, without the right to receive such issuance, dividend or distribution (including pursuant to due bills or similar arrangements required by the

relevant stock exchange). For the avoidance of doubt, any alternative trading convention on the applicable exchange or market in respect of the Common Stock under a separate ticker symbol or CUSIP number will not be considered "regular way" for this purpose.

- "Excess Shares" has the meaning set forth in Section 7(J)(i).
- "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended.
- "Expiration Date" has the meaning set forth in Section 7(G)(i)(5).
- "Expiration Time" has the meaning set forth in Section 7(G)(i)(5).
- "Forced Conversion" has the meaning set forth in Section 7(D)(i).
- "Forced Conversion Trigger" means (A) the Last Reported Sale Price exceeds one hundred and seventy five percent (175%) of the Conversion Price on at least twenty (20) VWAP Trading Days in any thirty (30) consecutive VWAP Trading Day period beginning on any VWAP Trading Day after the Issue Date and ending on the date upon which the Company Conversion Notice is delivered by the Company to the Holder and (B) the Equity Conditions are satisfied on the date the Company Conversion Notice is delivered by the Company to the Holder.
- "Freely Tradable" means, with respect to any shares of Common Stock issued or issuable pursuant to this Note, that (A) such shares are (or, when issued, will be) issued by the Company pursuant to an effective registration statement or Section 3(a)(9) under the Securities Act and would not constitute "restricted securities" within the meaning of Rule 144 under the Securities Act and without any requirement for registration under any state securities or "blue sky" laws; (B) such shares are (or, when issued, will be) (i) represented by book-entries at DTC and identified therein by an "unrestricted" CUSIP number; (ii) not represented by any certificate that bears a legend referring to transfer restrictions under the Securities Act or other securities laws; and (iii) listed and admitted for trading, without suspension or material limitation on trading, on an Eligible Exchange; and (C) no delisting or suspension by such Eligible Exchange has been threatened (with a reasonable prospect of delisting occurring after giving effect to all applicable notice, appeal, compliance and hearing periods) or reasonably likely to occur or pending as evidenced by (x) a writing by such Eligible Exchange or (y) the Company falling below the minimum listing maintenance requirements of such Eligible Exchange.

## "Fundamental Change" means any of the following events:

- (E) a "person" or "group" (within the meaning of Section 13(d)(3) of the Exchange Act), other than (x) the Company or its Wholly Owned Subsidiaries, (y) the employee benefit plans of the Company or its Wholly Owned Subsidiaries, or (z) the Holder or any of its Affiliates (including an "group" including the Holder or any of its Affiliates) files any report with the Commission indicating that such person or group has become the direct or indirect "beneficial owner" (as defined below) of shares of the Company's common equity representing more than fifty percent (50%) of the voting power of all of the Company's then-outstanding common equity;
- (F) the consummation of (i) any sale, lease or other transfer, in one transaction or a series of transactions, of all or substantially all of the assets of the Company and its Subsidiaries, taken as a whole, to any Person (other than solely to one or more of the Company's Wholly Owned Subsidiaries); or (ii) any transaction or series of related transactions in connection with which (whether by means of merger, consolidation, share exchange, combination, reclassification, recapitalization, acquisition, liquidation or otherwise) all of the Common Stock is exchanged for, converted into, acquired for, or constitutes solely the right to

receive, other securities, cash or other property (other than a subdivision or combination, or solely a change in par value, of the Common Stock); *provided*, *however*, that any merger, consolidation, share exchange or combination of the Company pursuant to which the Persons that directly or indirectly "beneficially owned" (as defined below) all classes of the Company's common equity immediately before such transaction directly or indirectly "beneficially own," immediately after such transaction, more than fifty percent (50%) of all classes of common equity of the surviving, continuing or acquiring company or other transferee, as applicable, or the parent thereof, in substantially the same proportions vis-à-vis each other as immediately before such transaction will be deemed not to be a Fundamental Change pursuant to this **clause (B)**; or

(G) the Company's stockholders approve any plan or proposal for the liquidation or dissolution of the Company.

For the purposes of this definition, (x) any transaction or event described in both clause (A) and in clause (B)(i) or (ii) above (without regard to the proviso in clause (B)) will be deemed to occur solely pursuant to clause (B) above (subject to such proviso); and (y) whether a Person is a "beneficial owner" and whether shares are "beneficially owned" will be determined in accordance with Rule 13d-3 under the Exchange Act.

"Fundamental Change Notice" has the meaning set forth in Section 6(C).

"Fundamental Change Repurchase Date" means the date as of which this Note must be repurchased for cash in connection with a Fundamental Change, as provided in Section 6(B).

"Fundamental Change Repurchase Price" means, with respect to this Note (or any portion of this Note to be repurchased) upon a Repurchase Upon Fundamental Change, a cash amount equal to the greater of (A) one hundred (100%) of the Maturity Principal Amount of this Note excluding interest (or such lesser principal amount accelerated pursuant to such notice), plus accrued and unpaid interest on this Note to be so repurchased and (B) the sum of (i) one hundred ten percent (110%) of the product of (a) the Conversion Rate in effect as of the Trading Day immediately preceding the effective date of such Fundamental Change; (b) the total then outstanding Principal Amount (expressed in thousands) of this Note; and (c) the average of the five (5) Daily VWAPs per share of Common Stock occurring on the later of (i) the five (5) VWAP Trading Day period immediately before the effective date of such Fundamental Change and (ii) the five (5) VWAP Trading Day period immediately after any public announcement of such Fundamental Change, which includes, but is not limited to, public disclosure on a Form 8-K or otherwise and (ii) the accrued and unpaid interest on this Note.

"GAAP" means generally accepted accounting principles in the United States of America, as in effect from time to time; provided the definitions set forth in this Note and any financial calculations required thereby shall be computed to exclude any change to lease accounting rules from those in effect pursuant to Financial Accounting Standards Board Accounting Standards Codification 840 (Leases) and other related lease accounting guidance as in effect on the date hereof.

"Holder" means the person in whose name this Note is registered on the books of the Registrar, which initially is the Initial Holder.

The term "including" means "including without limitation," unless the context provides otherwise.

"Holder Conversion Notice" has the meaning set forth in Section 7(C)(i).

"Indebtedness" means, indebtedness of any kind, including, without duplication (A) all indebtedness for borrowed money or the deferred purchase price of property or services, including reimbursement and other obligations with respect to surety bonds and letters of credit, (B) all obligations evidenced by notes, bonds, debentures or similar instruments, (C) all Capital Lease Obligations, (D) all Contingent Obligations, and (E) Disqualified Stock.

"Indenture" means that certain Indenture, dated as of [ ], 2023, by and between the Company and the Trustee, as amended and supplemented by the First Supplemental Indenture, dated as of [ ], 2023, by and between the Company and the Trustee.

"Independent Investigator" has the meaning set forth in Section 8(R).

"Initial Holder" has the meaning set forth in the cover page of this Note.

"Intellectual Property" means all of the Company's Copyrights; Trademarks; Patents; Licenses; trade secrets and inventions; mask works; the Company's applications therefor and reissues, extensions, or renewals thereof; and the Company's goodwill associated with any of the foregoing, together with the Company's rights to sue for past, present and future infringement of Intellectual Property and the goodwill associated therewith.

"Interest Payment Date" means (A) each January 1, April 1, July 1 and October 1 of each calendar year, beginning on [October 1, 2023]<sup>2</sup>; and (B) if not otherwise included in clause (A), the Maturity Date.

"Investment" means any beneficial ownership (including stock, partnership or limited liability company interests) of or in any Person, or any loan, advance or capital contribution to any Person or the acquisition of all, or substantially all, of the assets of another Person or the purchase of any assets of another Person for greater than the fair market value of such assets to solely the extent of the amount in excess of the fair market value.

"Investment Grade Securities" means: (1) securities issued or directly and fully guaranteed or insured by the U.S. government or any agency or instrumentality thereof (other than Cash Equivalents); (2) securities that have a rating equal to or higher than "Baa3" (or equivalent) by Moody's or "BBB-" (or equivalent) by S&P, or an equivalent rating by another internationally recognized rating agency, but excluding any debt securities or loans or advances between and among the Company and its Subsidiaries; (3) investments in any fund that invests exclusively in investments of the type described in clauses (1) and (2), which fund may also hold immaterial amounts of cash pending investment or distribution; and (4) corresponding instruments in countries other than the United States customarily utilized for high quality investments and in each case with maturities not exceeding two years from the date of acquisition.

"Issue Date" means [ ], 20[ ].

"Last Reported Sale Price" of the shares of Common Stock for any Trading Day means the closing sale price per share (or, if no closing sale price is reported, the average of the last bid price and the last ask price per share or, if more than one in either case, the average of the average last bid prices and the average last ask prices per share) of Common Stock on such Trading Day as reported in composite transactions for the principal U.S. national or regional securities exchange on which the shares of Common Stock are then listed. If the Common Stock is not listed on a U.S. national or regional securities exchange on such Trading Day, then the

<sup>&</sup>lt;sup>2</sup> NTD: include with respect to the Initial Purchased Note; Subsequently Purchased Notes to include the next occurring interest payment date after the issue date of such Notes

Last Reported Sale Price will be the last quoted bid price per share of Common Stock on such Trading Day in the over-the-counter market as reported by OTC Markets Group Inc. or a similar organization. If the Common Stock is not so quoted on such Trading Day, then the Last Reported Sale Price will be the average of the mid-point of the last bid price and the last ask price per share of Common Stock on such Trading Day from a nationally recognized independent investment banking firm selected by the Company.

"License" means any Copyright License, Patent License, Trademark License or other license of rights or interests.

"Lien" means any mortgage, deed of trust, pledge, hypothecation, assignment for security, security interest, encumbrance, levy, lien or charge of any kind, whether voluntarily incurred or arising by operation of law or otherwise, against any property, any conditional sale or other title retention agreement, and any lease in the nature of a security interest; provided, that for the avoidance of doubt, licenses, strain escrows and similar provisions in collaboration agreements or research and development agreements that do not create or purport to create a security interest, encumbrance, levy, lien or charge of any kind shall not be deemed to be Liens for purposes of this Note.

"Market Disruption Event" means, with respect to any date, the occurrence or existence, during the one-half hour period ending at the scheduled close of trading on such date on the principal, in terms of volume, Eligible Exchange (or, if the Common Stock is not then listed on an Eligible Exchange, on the principal, in terms of volume, U.S. national or regional securities exchange on which the Common Stock is listed for trading) on which the Common Stock is listed for trading or trades, of any material suspension or limitation imposed on trading (by reason of movements in price exceeding limits permitted by the relevant exchange or otherwise) in the Common Stock or in any options contracts or futures contracts relating to the Common Stock.

"Market Stock Payment Price" means, with respect to any Event of Default Stock Payment Date, an amount equal to one hundred percent (100%) of the lowest of the ten (10) Daily VWAPs immediately prior to such Event of Default Stock Payment Date.

"Maturity Date" means August 1, 2026.

"Maturity Principal Amount" has the meaning set forth in the cover page of this Note; *provided*, *however*, that the Maturity Principal Amount of this Note will be subject to reduction pursuant to Section 4, Section 5, Section 6, and Section 7.

"Maximum Percentage" has the meaning set forth in Section 7(J)(i).

"Moody's" means Moody's Investors Service, Inc. or any successor to the rating agency business thereof.

"Note Documents" means the Securities Purchase Agreement, this Note, the Security Agreements and the other Transaction Documents.

"Obligation" means all liabilities, indebtedness and obligations (including interest accrued at the rate provided in the applicable Note Document after the commencement of a bankruptcy proceeding, whether or not a claim for such interest is allowed) of the Company, any Security Agreements or any other Note Document, in each case howsoever created, arising or evidenced, whether direct or indirect, absolute or contingent, now or hereafter existing, or due or to become due.

"Open of Business" means 9:00 a.m., New York City time.

The term "or" is not exclusive, unless the context expressly provides otherwise.

"Other Holder" means any person in whose name any Other Note is registered on the books of the Registrar.

"Other Notes" means any Notes that are of the same series as this Note and that are represented by one or more certificates other than the certificate representing this Note.

"Partial Redemption Date" means, with respect to this Note, (A) each of [January 1, 2024, April 1, 2024, July 1, 2024,] October 1, 2024, January 1, 2025, April 1, 2025, April 1, 2025, July 1, 2025, October 1, 2025, January 1, 2026, April 1, 2026 and July 1, 2026 and (B) if not otherwise included in clause (A), the Maturity Date.

"Partial Redemption Notice" has the meaning set forth in Section 4(A).

"Partial Redemption Payment" means, for any date that is a Partial Redemption Date, an amount equal to one hundred fifteen percent (115%) of twelve and a half percent (12.5%) of the Principal Amount as of the Issue Date of this Note and all other Notes issued on the Issue Date, as determined by Holder in its sole discretion; *provided*, *however*, that the Company shall not be required to partially redeem more than such amount in aggregate in respect of this Note and all Other Notes issued on the same Issue Date that have been elected to be partially redeemed by the Holder and the Other Holders on any Partial Redemption Date; *provided*, *further*, that the Holder, all of the Other Holders and the Company may agree to increase the size of any Partial Redemption Payment by mutual written consent and notice to the Trustee.

"Patent License" means any written agreement granting any right with respect to any invention covered by a Patent that is in existence or a Patent application that is pending, in which agreement the Company now holds or hereafter acquires any interest.

"Patents" means all letters patent of, or rights corresponding thereto, in the United States or in any other country, all registrations and recordings thereof, and all applications for letters patent of, or rights corresponding thereto, in the United States or any other country.

"Permitted Indebtedness" means (A) Indebtedness evidenced by the Notes; (B) Indebtedness actually disclosed pursuant to the Securities Purchase Agreement as of the date of the Securities Purchase Agreement; (C) Indebtedness to trade creditors incurred in the ordinary course of business consistent with past practices; (D) Indebtedness that also constitutes a Permitted Investment; (E) Subordinated Indebtedness of the Company; (F) reimbursement obligations in connection with letters of credit or similar instruments that are secured by Cash or Cash Equivalents and issued on behalf of the Company or a Subsidiary thereof in an aggregate amount not to exceed two hundred fifty thousand dollars (\$250,000) at any time outstanding; (G) Indebtedness consisting of capital and operating lease obligations and Indebtedness secured by Liens permitted under clause (M) of the definition of "Permitted Liens" hereunder; provided, however, all of such Indebtedness under this clause (G), plus all Indebtedness secured by Liens permitted under clause (M) of the definition of "Permitted Liens" (without duplication) shall not exceed One Million Four Hundred Thousand Dollars (\$1,400,000.00) in the aggregate; and (H) Contingent Obligations that are guarantees of Indebtedness described in clauses (A) through (G).

<sup>&</sup>lt;sup>3</sup> NTD: include with respect to the Initial Purchased Note; Subsequently Purchased Notes to include these dates to the extent applicable date occurs after the issue date of such Notes.

"Permitted Intellectual Property Licenses" means (A) Intellectual Property licenses actually disclosed pursuant to the Securities Purchase Agreement as of the date of the Securities Purchase Agreement, and (B) non-perpetual Intellectual Property licenses granted in the ordinary course of business on arm's length terms consisting of the licensing of technology, the development of technology or the providing of technical support which may include licenses with unlimited renewal options solely to the extent such options require mutual consent for renewal or are subject to financial or other conditions as to the ability of licensee to perform under the license; provided such license was not entered into during an Event of Default or continuance of a Default.

"Permitted Investment" means: (A) Investments actually disclosed pursuant to the Securities Purchase Agreement, as in effect as of the Issue Date; (B) Cash and Cash Equivalents; (C) Investments accepted in connection with Permitted Transfers; (D) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of the Company's business; (E) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers in the ordinary course of business and consistent with past practice, provided that this clause (E) shall not apply to Investments of the Company in any Subsidiary thereof; (F) Investments consisting of (i) loans not involving the net transfer on a substantially contemporaneous basis of cash proceeds to employees, officers or directors relating to the purchase of capital stock of the Company pursuant to employee stock purchase plans or other similar agreements approved by the Company's Board of Directors and (ii) travel advances and employee relocation loans and other employee loans and advances in the ordinary course of business, provided that the aggregate of all such loans outstanding may not exceed two hundred fifty thousand dollars (\$250,000) at any time; (G) Investments in Wholly Owned Subsidiaries; (H) Permitted Intellectual Property Licenses; and (I) additional Investments that do not exceed one million dollars (\$1,000,000) in the aggregate in any twelve (12) month period.

"Permitted Liens" means any and all of the following: (A) Liens in favor of Holder or the Collateral Agent; (B) Liens actually or deemed to be disclosed pursuant to the Securities Purchase Agreement, as in effect as of the Issue Date; (C) Liens for taxes, fees, assessments or other governmental charges or levies, either not delinquent or being contested in good faith by appropriate proceedings; provided, that the Company maintains adequate reserves therefor in accordance with GAAP; (D) Liens securing claims or demands of materialmen, artisans, mechanics, carriers, warehousemen, landlords and other like Persons arising in the ordinary course of business; provided, that the payment thereof is not yet required; (E) Liens arising from judgments, decrees or attachments in circumstances which do not constitute a Default or an Event of Default hereunder; (F) the following deposits, to the extent made in the ordinary course of business: deposits under workers' compensation, unemployment insurance, social security and other similar laws, or to secure the performance of bids, tenders or contracts (other than for the repayment of borrowed money) or to secure indemnity, performance or other similar bonds for the performance of bids, tenders or contracts (other than for the repayment of borrowed money) or to secure statutory obligations (other than Liens arising under ERISA or environmental Liens) or surety or appeal bonds, or to secure indemnity, performance or other similar bonds; (G) leasehold interests in leases or subleases and licenses granted in the ordinary course of the Company's business and not interfering in any material respect with the business of the licensor; (H) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of custom duties that are promptly paid on or before the date they become due; (I) Liens on insurance proceeds securing the payment of financed insurance premiums that are promptly paid on or before the date they become due (provided that such Liens extend only to such insurance proceeds and not to any other property or assets); (J) statutory and common law rights of set-off and other similar rights as to deposits of cash and securities in favor of banks, other depository institutions and brokerage firms; (K) easements, zoning restrictions, rights-of-way

and similar encumbrances on real property imposed by law or arising in the ordinary course of business so long as they do not materially impair the value or marketability of the related property; (L) Liens on Cash or Cash Equivalents securing obligations permitted under clauses (C) and (F) of the definition of Permitted Indebtedness; (M) purchase money Liens, or Liens in respect of equipment leases, (i) on equipment acquired or held by the Company incurred for financing the acquisition or lease of the equipment securing no more than One Million Four Hundred Thousand Dollars (\$1,400,000.00) in the aggregate amount outstanding; provided, however, all of the Indebtedness securing the Liens permitted under this clause (M), plus all Indebtedness permitted under clause (G) of the definition of "Permitted Indebtedness" (without duplication) shall not exceed One Million Four Hundred Thousand Dollars (\$1,400,000.00) in the aggregate, or (ii) existing on equipment when acquired, if the Lien is confined to the property and improvements and the proceeds of the equipment; and (N) Liens incurred in connection with the extension, renewal or refinancing of the Indebtedness secured by Liens of the type described in clauses (B) through (J) and (N) above (other than any Indebtedness repaid with the proceeds of this Note); provided, that any extension, renewal or replacement Lien shall be limited to the property encumbered by the existing Lien and the principal amount of the Indebtedness being extended, renewed or refinanced (as may have been reduced by any payment thereon) does not increase.

"Permitted Transfers" means (A) dispositions of Cash, Cash Equivalents or Investment Grade Securities in the ordinary course of business and consistent with past practice for fair value consideration in the form of Cash, Cash Equivalents or Investment Grade Securities, (B) dispositions of inventory, goods and other assets, leases, assignments or subleases of any real or personal property, and Permitted Intellectual Property Licenses entered into, in each case, in the ordinary course of business and consistent with past practice, (C) dispositions of worn-out, obsolete or surplus property at fair market value in the ordinary course of business; (D) dispositions of accounts or payment intangibles (each as defined in the UCC) resulting from the compromise or settlement thereof in the ordinary course of business for less than the full amount thereof; (E) transfers consisting of Permitted Investments in Wholly Owned Subsidiaries under clause (G) of Permitted Investments; (F) other transfers of assets to any Person other than to a joint venture and which have a fair market value of not more than fifty thousand dollars (\$50,000) in the aggregate in any twelve (12) month period; (G) dispositions in connection with Permitted Liens; or (H) the disposition of all or substantially all of the assets of the Company and its Subsidiaries in a manner permitted pursuant to Section 9 and any disposition that constitutes a Fundamental Change.

"Person" or "person" means any individual, sole proprietorship, partnership, limited liability company, joint venture, company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, firm, joint stock company, estate, entity or government agency.

"Principal Amount" has the meaning set forth in the cover page of this Note; *provided*, *however*, that the Principal Amount of this Note will be subject to reduction (A) pursuant to Section 4, Section 5, Section 6, and Section 7.

"Reference Property" has the meaning set forth in Section 7(I)(i)(4).

"Reference Property Unit" has the meaning set forth in Section 7(I)(i)(4).

"Registrar" has the meaning set forth in the Indenture.

"Related Parties" has the meaning set forth in Section 22.

"Reported Outstanding Share Number" has the meaning set forth in Section 7(J)(i).

- "Repurchase Upon Fundamental Change" means the repurchase of any Note by the Company pursuant to Section 6.
- "Required Holders" means holders of a majority of the aggregate principal amount of the Notes, so long as any Notes remain outstanding; provided, that such majority must include High Trail Investments ON LLC or HB SPV I Master Sub LLC, so long as High Trail Investments ON LLC or HB SPV I Master Sub LLC or any of their affiliates hold any Notes or are otherwise entitled to receive Underlying Shares.
  - "Required Reserve Amount" has the meaning in Section 8(Q).
- "Requisite Stockholder Approval" means the stockholder approval contemplated by Section 312.03(b) of the NYSE Listed Company Manual with respect to the issuance of shares of Common Stock upon conversion of the Notes in excess of the limitations imposed by such rule; provided, however, that the Requisite Stockholder Approval will be deemed to be obtained if, due to any amendment or binding change in the interpretation of the applicable listing standards of The New York Stock Exchange, such stockholder approval is no longer required for the Company to settle all conversions of the Notes by delivering shares of Common Stock without limitation pursuant to Section 7(B).
  - "Rule 144" means Rule 144 promulgated under the Securities Act.
  - "S&P" means Standard & Poor's Ratings Services or any successor to the rating agency business thereof.
- "Scheduled Trading Day" means any day that is scheduled to be a Trading Day on the principal, in terms of volume, Eligible Exchange on which the Common Stock is then listed or, if the Common Stock is not then listed on an Eligible Exchange, on the principal, in terms of volume, stock exchange or market on which the Common Stock is then listed or traded. If the Common Stock is not so listed or traded, then "Scheduled Trading day" means a Business Day.
  - "Securities Act" means the U.S. Securities Act of 1933, as amended.
- "Securities Purchase Agreement" means that certain Securities Purchase Agreement, dated as of August [ ], 2023, between the Company, High Trail Investments ON LLC and HB SPV I Master Sub LLC providing for the issuance of the Notes. The Trustee shall not been deemed to have any knowledge of or have any obligation to monitor the terms of the Securities Purchase Agreement.
  - "Share Delivery Date" any (i) Event of Default Stock Payment Delivery Date or (ii) Conversion Settlement Date.
- "Significant Subsidiary" means, with respect to any Person, any Subsidiary of such Person that constitutes a "significant subsidiary" (as defined in Rule 1-02(w) of Regulation S-X under the Exchange Act) of such Person.
  - "Spin-Off" has the meaning set forth in Section 7(G)(i)(3)(b).
  - "Spin-Off Valuation Period" has the meaning set forth in Section 7(G)(i)(3)(b).
  - "Stated Interest" has the meaning set forth in Section 4(B).
  - "Stated Interest Rate" means, as of any date, a rate per annum equal to six percent (6.0%).

"Stock Payment Determination Date" means (i) with respect to an Event of Default Stock Payment, the date of delivery of the related Event of Default Stock Payment Notice, and (ii) with respect to the delivery of Conversion Consideration, the related Conversion Date.

"Subordinated Indebtedness" means Indebtedness subordinated to the Notes pursuant to a written agreement between the Holder and the applicable lender in amounts and on terms and conditions satisfactory to the Holder in its sole discretion.

"Subsidiary" means, with respect to any Person, (A) any corporation, association or other business entity (other than a partnership or limited liability company) of which more than fifty percent (50%) of the total voting power of the Capital Stock entitled (without regard to the occurrence of any contingency, but after giving effect to any voting agreement or stockholders' agreement that effectively transfers voting power) to vote in the election of directors, managers or trustees, as applicable, of such corporation, association or other business entity is owned or controlled, directly or indirectly, by such Person or one or more of the other Subsidiaries of such Person; and (B) any partnership or limited liability company where (i) more than fifty percent (50%) of the capital accounts, distribution rights, equity and voting interests, or of the general and limited partnership interests, as applicable, of such partnership or limited liability company are owned or controlled, directly or indirectly, by such Person or one or more of the other Subsidiaries of such Person, whether in the form of membership, general, special or limited partnership or limited liability company interests or otherwise; and (ii) such Person or any one or more of the other Subsidiaries of such Person is a controlling general partner of, or otherwise controls, such partnership or limited liability company.

"Successor Corporation" has the meaning set forth in Section 9(A).

"Successor Person" has the meaning set forth in Section 7(I)(i).

"Tender/Exchange Offer Valuation Period" has the meaning set forth in Section 7(G)(i)(5).

"Trademark License" means any written agreement granting any right to use any Trademark or Trademark registration, now owned or hereafter acquired by the Company or in which the Company now holds or hereafter acquires any interest.

"Trademarks" means all trademarks (registered, common law or otherwise) and any applications in connection therewith, including registrations, recordings and applications in the United States Patent and Trademark Office or in any similar office or agency of the United States, any State thereof or any other country or any political subdivision thereof.

"Trading Day" means any day on which (A) trading in the Common Stock generally occurs on the principal, in terms of volume, Eligible Exchange on which the Common Stock is then listed or, if the Common Stock is not then listed on an Eligible Exchange, on the principal, in terms of volume, U.S. national or regional securities exchange on which the Common Stock is listed for trading; and (B) there is no Market Disruption Event; provided that the Holder, by written notice to the Company, may waive any such Market Disruption Event. If the Common Stock is not so listed or traded, then "Trading Day" means a Business Day.

"Transaction Documents" means, collectively, the Securities Purchase Agreement, the Notes, the Perfection Certificate, the Security Agreements (as defined in the Securities Purchase Agreement), the Indenture, each Lock-Up and Voting Agreement (as defined in the Securities Purchase Agreement), and the Irrevocable Transfer Agent Instructions (as defined in the Securities Purchase Agreement) and each of the other agreements and instruments entered into or

delivered by any of the parties hereto in connection with the transactions contemplated by the Securities Purchase Agreement and thereby, as may be amended from time to time.

"Trust Indenture Act" means the U.S. Trust Indenture Act of 1939, as amended.

"Trustee" means U.S. Bank Trust Company National Association in its capacity as trustee under the Indenture.

"UCC" means the Uniform Commercial Code as the same is, from time to time, in effect in the State of New York.

"Undelivered Shares" has the meaning set forth in Section 7(E)(iv).

"VWAP Market Disruption Event" means, with respect to any date, (A) the failure by the principal U.S. national or regional securities exchange on which the Common Stock is then listed, or, if the Common Stock is not then listed on a U.S. national or regional securities exchange, the principal, in terms of volume, Eligible Exchange on which the Common Stock is then traded, to open for trading during its regular trading session on such date; or (B) the occurrence or existence, for more than one half hour period in the aggregate, of any suspension or limitation imposed on trading (by reason of movements in price exceeding limits permitted by the relevant exchange or otherwise) in the Common Stock or in any options contracts or futures contracts relating to the Common Stock, and such suspension or limitation occurs or exists at any time before 1:00 p.m., New York City time, on such date.

"VWAP Trading Day" means a day on which (A) there is no VWAP Market Disruption Event; provided that the Holder, by written notice to the Company, may waive any such VWAP Market Disruption Event; and (B) trading in the Common Stock generally occurs on the principal, in terms of volume, Eligible Exchange on which the Common Stock is then listed or, if the Common Stock is not then listed on an Eligible Exchange, on the principal, in terms of volume, U.S. national or regional securities exchange on which the Common Stock is then traded. If the Common Stock is not so listed or traded, then "VWAP Trading Day" means a Business Day.

"Wholly Owned Subsidiary" of a Person means any Subsidiary of such Person all of the outstanding Capital Stock or other ownership interests of which (other than directors' qualifying shares) are owned by such Person or one or more Wholly Owned Subsidiaries of such Person.

"Withheld Shares" has the meaning set forth in Section 7(J)(ii).

#### Section 2. Persons Deemed Owners.

The Holder of this Note will be treated as the owner of this Note for all purposes.

## Section 3. Registered Form.

This Note, and any Other Note issued in exchange therefor or in substitution thereof, will be in registered form, without coupons.

## Section 4. Partial Redemption Payments; Interest; Maturity Date Payment; Prepayment.

(A) Partial Redemption Payments. If the Holder wishes to elect to require the Company to redeem all or a portion of this Note (in an Authorized Denomination) for a Partial Redemption Payment, the Holder shall deliver to the Company (with a copy to the Trustee) a

written notice of any such election, including the applicable amount of the Partial Redemption Payment (a "Partial Redemption Notice"), at least ten (10) Trading Days prior to the applicable Partial Redemption Date in order to make an effective election. Such notice may be canceled by the Holder at any time prior to the receipt of the applicable Partial Redemption Payment from the Company and the Holder shall also have the right to convert such Partial Redemption Payment (or any applicable portion thereof) into Common Stock pursuant to Section 7 hereof at any time prior to the receipt of the applicable Partial Redemption Payment from the Company. If the Holder or any Other Holder elects to redeem all or a portion of any Other Notes for a Partial Redemption Payment on the same Partial Redemption Date, the Trustee shall the select the redemption portions of the principal of this Note and such Other Notes (in Authorized Denominations) on a pro rata basis to the extent practicable or such other method the Trustee deems fair and appropriate. The Company shall pay the Holder the Partial Redemption Payment (or applicable portion thereof) by wire transfer of immediately available funds on the applicable Partial Redemption Date. Any Partial Redemption Payment (or applicable portion thereof) paid pursuant to this Section 4(A) shall reduce the Principal Amount by such paid amount divided by one and fifteen hundredths (1.15). If this Note (or any portion of this Note) is to be redeemed pursuant to this Section 4(A), then, from and after the date the related Partial Redemption Payment (or applicable portion thereof) is paid in full, this Note (or such portion) will cease to be outstanding and interest will cease to accrue on this Note (or such portion).

- (B) Interest. Except as provided in Section 10(D), this Note will accrue interest (the "Stated Interest") at a rate per annum equal to the Stated Interest Rate. Stated Interest on this Note will (i) accrue on the Principal Amount of this Note; (ii) accrue from, and including, the most recent date to which Stated Interest has been paid or duly provided for (or, if no Stated Interest has theretofore been paid or duly provided for, the Issue Date) to, but excluding, the date of payment of such Stated Interest; (iii) be paid to Holder in cash on each Interest Payment Date in accordance with Section 5(A); (iv) with respect to the amount of interest then accrued on the portion of the Principal Amount being redeemed in connection with a Cash Sweep Payment, be paid to the Holder in arrears as of the date of such Cash Sweep Payment in accordance with this Section 4, as applicable; (v) with respect to the amount of interest then accrued on the portion of the Principal Amount being redeemed on a Partial Redemption Date, be paid to the Holder in arrears as of each such date in accordance with this Section 4, as applicable; (vi) with respect to the amount of interest then accrued on the Principal Amount being repurchased on a Fundamental Change Repurchase Date, be paid to the Holder in arrears as of each such date in accordance with Section 6, as applicable; and (vi) be computed on the basis of a 360-day year comprised of twelve 30-day months.
- (C) Maturity Date Payment. On the Maturity Date, the Company will pay the Holder an amount in cash equal to the Maturity Principal Amount for the then-outstanding Principal Amount of this Note plus any accrued and unpaid interest on this Note.
  - (D) Cash Sweep Payments.
  - (i) For purposes of this Note, any payment made to the Holder pursuant to **Section 4(D)** shall be referred to as a **"Cash Sweep Payment"**. All Cash Sweep Payments shall be accompanied by accrued and unpaid interest on the amount repaid.
  - (ii) Concurrently with the completion of any Cash Sweep Financing, the Company shall certify to Holder in writing (i) the amount of the applicable Cash Sweep Financing and (ii) the calculation of the potential Cash Sweep Amount with respect to such Cash Sweep Financing (including a certification that such Cash Sweep Amount was calculated in accordance with the terms hereof) (such certification a "Cash Sweep Certification"); provided, however, that, unless consented to by the Holder in writing, in

the event that the extent of such Cash Sweep Financing and Cash Sweep Amount is such that the information required in such certification would constitute material non-public information regarding the Company, then the Company shall also concurrently publicly disclose such material non-public information on a Current Report on Form 8-K or otherwise.

- (iii) The Holder shall have the right to require the Company, exercisable by delivery of written notice to the Company (with a copy to the Trustee) of exercise of such right (a "Cash Sweep Notice"), to redeem all or a portion of this Note in cash within three (3) Business Days following the delivery of such Cash Sweep Notice (regardless of whether the Company actually delivers a Cash Sweep Certification) for all or a portion of the Cash Sweep Amount with respect to such Cash Sweep Financing. If the Holder or any Other Holder elects to redeem all or a portion of any Other Notes for all or a portion of the Cash Sweep Amount with respect to the same Cash Sweep Financing, the Trustee shall then select the redemption portions of the principal of this Note and such Other Notes on a pro rata basis to the extent practicable or such other method the Trustee deems fair and appropriate. The Company shall pay the Holder the Cash Sweep Payment by wire transfer of immediately available funds. Any Cash Sweep Payment paid pursuant to this Section 4(D) shall reduce the Principal Amount by such paid amount divided by one and fifteen hundredths (1.15). If this Note (or any portion of this Note) is to be redeemed pursuant to this Section 4(D), then, from and after the date the related Cash Sweep Payment is paid in full, this Note (or such portion) will cease to be outstanding and interest will cease to accrue on this Note (or such portion).
  - (E) Prepayment. The Company may not prepay this Note without the written consent of the Required Holders.

# Section 5. Method of Payment; When Payment Date is Not a Business Day.

- (A) *Method of Payment.* The Company will pay all cash amounts due under this Note by wire transfer of immediately available funds to the account of the Holder as set forth in a written notice of an account of such Holder delivered by the Holder to the Company at least one (1) Business Day before the date such amount is due.
- (B) Delay of Payment when Payment Date is Not a Business Day. If the due date for a payment on this Note as provided in this Note is not a Business Day, then, notwithstanding anything to the contrary in this Note, such payment may be made on the immediately following Business Day and no interest will accrue on such payment as a result of the related delay.
- (C) Event of Default Stock Payments. If an Event of Default occurs and the Company fails to pay the Event of Default Acceleration Amount when due in accordance with this Note, then the Holder may elect to receive such unpaid portion of the Event of Default Acceleration Amount, entirely or partially, in shares of Common Stock (an "Event of Default Stock Payment"), and shall deliver to the Company (with a copy to the Trustee) a written notice of such election stating which portion thereof the Holder has elected to receive in shares of Common Stock (an "Event of Default Stock Payment Notice"). On or before the second (2nd) Business Day following the date of delivery of any Event of Default Stock Payment Notice hereunder (the "Event of Default Stock Payment Delivery Date"), the Company shall issue and deliver to the Holder, a number of validly issued, fully paid and Freely Tradable shares of Common Stock equal to the quotient (rounded up to the closest whole number) obtained by dividing the Event of Default Acceleration Amount (or applicable portion thereof) by the Market Stock Payment Price as of the date of delivery of the Event of Default Stock Payment Notice; provided, that, if the Company fails to timely issue and deliver to the Holder such shares of Common Stock, then the Holder may revoke its election to receive shares of Common Stock and

elect to receive such Event of Default Acceleration Amount (or any portion thereof) in cash at any time prior to delivery of such shares of Common Stock. Any portion of the Event of Default Acceleration Amount not paid in shares of Common Stock because the Holder did not elect, or effectively revoked its election, to receive shares of Common Stock for such Event of Default Acceleration Amount (or applicable portion thereof) will be paid in cash; provided, that the Holder may deliver multiple Event of Default Stock Payment Notices in accordance with this **Section 5(C)** to the extent that any portion of the Event of Default Acceleration Amount remains unpaid when due in accordance with this Note.

## Section 6. Required Repurchase of Note upon a Fundamental Change.

- (A) Repurchase Upon Fundamental Change. Subject to the other terms of this **Section 6**, if a Fundamental Change occurs, then the Holder will have the right to require the Company to repurchase this Note (or any portion of this Note in an Authorized Denomination) on the Fundamental Change Repurchase Date for such Fundamental Change for a cash purchase price equal to the Fundamental Change Repurchase Price.
- (B) Fundamental Change Repurchase Date. The Fundamental Change Repurchase Date for any Fundamental Change will be a Business Day of the Holder's choosing that is no more than twenty (20) Business Days after the later of (x) the date the Company delivers to the Holder the related Fundamental Change Notice pursuant to **Section 6(C)**; and (y) the effective date of such Fundamental Change.
- (C) Fundamental Change Notice. No later than the fifth (5th) Business Day before the occurrence of any Fundamental Change, the Company will send to the Holder (with a copy to the Trustee) a written notice (the "Fundamental Change Notice") thereof (provided, however, in no event shall such notice be required prior to the actual public announcement of such Fundamental Change), stating the expected date such Fundamental Change will occur. No later than the fifth (5<sup>th</sup>) Business Day after the date of delivery of the Fundamental Change Notice, the Holder shall notify the Company in writing whether it will require the Company to repurchase this Note and specify the Fundamental Change Repurchase Date.
- (D) Effect of Repurchase. If this Note (or any portion of this Note) is to be repurchased upon a Repurchase Upon Fundamental Change, then, from and after the date the related Fundamental Change Repurchase Price is paid in full, this Note (or such portion) will cease to be outstanding and interest will cease to accrue on this Note (or such portion).

#### Section 7. Conversion.

- (A) Right to Convert.
- (i) Generally. Subject to the provisions of this **Section 7**, the Holder may, at its option, convert this Note, including any portion constituting a Partial Redemption Payment, into Conversion Consideration.
- (ii) Conversions in Part. Subject to the terms of this **Section 7**, this Note may be converted in part, but only in an Authorized Denomination. Provisions of this **Section 7** applying to the conversion of this Note in whole will equally apply to conversions of any permitted portion of this Note.
- (B) When this Note May Be Converted.
- (i) Generally. The Holder may convert this Note immediately at any time until the Close of Business on the second (2nd) Scheduled Trading Day immediately

before the Maturity Date. For the avoidance of doubt, the ability of the Holder to convert this Note shall not be delayed to give effect to any time required by the Trustee in which to reflect the issuance of this Note.

(ii) Limitations and Closed Periods. Notwithstanding anything to the contrary in this Section 7, if this Note (or any portion of this Note) is to be repurchased upon a Repurchase Upon Fundamental Change, then in no event may this Note (or such portion) be converted after the Close of Business on the Scheduled Trading Day immediately before the related Fundamental Change Repurchase Date; provided, that the limitations contained in this Section 7(B)(ii) shall no longer apply to this Note (or such applicable portion) if the applicable Fundamental Change Repurchase Price is not delivered on the Fundamental Change Repurchase Date in accordance with Section 6.

#### (C) Conversion Procedures.

- (i) Generally. To convert this Note, the Holder must (1) complete, sign and deliver to the Company (with a copy to the Trustee) the conversion notice attached to this Note on **Exhibit A** or portable document format (.pdf) version of such conversion notice (at which time such conversion will become irrevocable) (a "**Holder Conversion Notice**"); and (2) pay an amounts due pursuant to **Section 7(C)(iii)**. For the avoidance of doubt, the Holder Conversion Notice may be delivered by e-mail in accordance with **Section 14**. If the Company fails to deliver, by the related Conversion Settlement Date, any shares of Common Stock forming part of the Conversion Consideration of the conversion of this Note, the Holder, by notice to the Company, may rescind all or any portion of the corresponding Holder Conversion Notice at any time until such Undelivered Shares are delivered.
- (ii) Holder of Record of Conversion Shares. The person in whose name any shares of Common Stock is issuable upon conversion of this Note will be deemed to become the holder of record of such shares as of the Close of Business on the Conversion Date for such conversion, conferring, as of such time, upon such person, without limitation, all voting and other rights appurtenant to such shares.
- (iii) Taxes and Duties. If the Holder converts this Note, the Company will pay any documentary, stamp or similar issue or transfer tax or duty due on the issue of any shares of Common Stock upon such conversion; provided, however, that if any tax or duty is due because such Holder requested such shares to be issued in a name other than that of such Holder, then such Holder will pay such tax or duty and, until having received a sum sufficient to pay such tax or duty, the Company may refuse to deliver any such shares to be issued in a name other than that of such Holder.

## (D) Right of Company to Convert the Note.

(i) Generally. If the Forced Conversion Trigger occurs, then, subject to Section 7(J), the Company may provide written notice to the Holder (with a copy to the Trustee) in substantially the form attached hereto as Exhibit B (a "Company Conversion Notice") electing to convert all (but, for the avoidance of doubt, not less than all) of the Principal Amount into Conversion Consideration (a "Forced Conversion") and certifying that the Equity Conditions are satisfied on the date the Company Conversion Notice was delivered to the Holder; provided that (x) no Forced Conversion will be effected unless (i) the Equity Conditions are satisfied on each VWAP Trading Day from the date of the Company Conversion Notice until the corresponding Conversion Consideration is delivered and (ii) the Last Reported Sale Price exceeded one hundred and seventy five percent (175%) of the Conversion Price, on twenty (20) of the

- thirty (30) consecutive VWAP Trading Days during the thirty (30) VWAP Trading Day period ending on and including the date the Company Conversion Notice was delivered to the Holder and during the thirty (30) VWAP Trading Day period ending on and including the date the Company delivers the Conversion Consideration pursuant to such Forced Conversion, and (y) if the Company receives a Holder Conversion Notice prior to the date the Company delivers the Company Conversion Notice and any Conversion Shares due thereunder remain undelivered by the Company, the Forced Conversion may not occur until after such Conversion Shares are delivered to the Holder.
- (ii) Effect of Forced Conversion. A Forced Conversion will have the same effect as a conversion of the applicable outstanding Principal Amount of this Note effected at a Holder's election pursuant to Section 7(A) with a Conversion Date occurring on the Business Day upon which the Company Conversion Notice is delivered to the Holder (provided, for the avoidance of doubt, that the conditions set forth in Section 7(D)(i) are satisfied on such date) and a Conversion Settlement Date occurring on the Business Day referred to in Section 7(E)(iii).
- (E) Settlement upon Conversion.
- (i) Generally. The consideration (the "Conversion Consideration") due in respect of each one thousand dollars (\$1,000) Principal Amount of this Note, including any portion constituting a Partial Redemption Payment required to be paid by the Company on the next Partial Redemption Date, to be converted will consist of the following:
- (1) subject to **Section 7(E)(ii)**, a number of shares of Common Stock equal to the Conversion Rate in effect on the Conversion Date for such conversion; and
- (2) cash in an amount equal to the aggregate accrued and unpaid interest on this Note to, but excluding, the Conversion Settlement Date for such conversion.
- (ii) Fractional Shares. The total number of shares of Common Stock due in respect of any conversion of this Note pursuant to this **Section 7**, including any portion constituting a Partial Redemption Payment required to be paid by the Company on the next Partial Redemption Date, will be determined on the basis of the total Principal Amount of this Note to be converted with the same Conversion Date; provided, however, that if such number of shares of Common Stock is not a whole number, then such number will be rounded up to the nearest whole number.
- (iii) Delivery of the Conversion Consideration. The Company will pay or deliver, as applicable, the Conversion Consideration due upon the conversion of this Note, including any portion constituting a Partial Redemption Payment required to be paid by the Company on the next Partial Redemption Date, to the Holder on or before the second (2nd) Business Day (or, if earlier, the standard settlement period for the primary Eligible Exchange (measured in terms of trading volume for its Common Stock) on which the Common Stock is traded) immediately after the Conversion Date for such conversion (the "Conversion Settlement Date").
- (iv) Company Failure to Timely Deliver Stock Payments. If (x) the Company shall fail for any reason or for no reason on or prior to the applicable Share Delivery Date to deliver shares of Common Stock in accordance with Section 5(C), Section 7(C) or Section 7(D) (such shares to which Holder is entitled referred to as the "Undelivered Shares"); and (y) the Holder (whether directly or indirectly, including by any broker

acting on the Holder's behalf or acting with respect to such Undelivered Shares) purchases any shares of Common Stock (whether in the open market or otherwise) to cover any such Undelivered Shares (whether to satisfy any settlement obligations with respect thereto of the Holder or otherwise), then, without limiting the Holder's right to pursue any other remedy available to it (whether hereunder, under applicable law or otherwise), the Holder will have the right, exercisable by notice to the Company, to cause the Company to either:

- (1) pay, on or before the second (2nd) Business Day after the date such notice is delivered, cash to the Holder in an amount equal to the aggregate purchase price (including any brokerage commissions and other out-of-pocket costs) incurred to purchase such shares (such aggregate purchase price, the "Covering Price"); or
- (2) promptly deliver, to the Holder, such Undelivered Shares in accordance with this Note, together with cash in an amount equal to the excess, if any, of the Covering Price over the product of (x) the number of such Undelivered Shares; and (y) the Daily VWAP per share of Common Stock on the applicable Stock Payment Determination Date relating to such conversion.

To exercise such right, the Holder must deliver notice of such exercise to the Company, specifying whether the Holder has elected clause (1) or (2) above to apply. If the Holder has elected clause (1) to apply, then the Company's obligation to deliver the Undelivered Shares in accordance with this Note will be deemed to have been satisfied and discharged to the extent the Company has paid the Covering Price in accordance with clause (1). Nothing herein shall limit the Holder's right to pursue any other remedies available to it hereunder, at law or in equity including, without limitation, a decree of specific performance and/or injunctive relief with respect to the Company's failure to timely deliver shares of Common Stock as required pursuant to the terms hereof. In addition to the foregoing, if the Company fails for any reason to deliver Common Stock to the Holder by the applicable Share Delivery Date, the Company shall pay to the Holder, in cash, as liquidated damages and not as a penalty, for each one thousand dollars (\$1,000) of Undelivered Shares (based on the Daily VWAP on the applicable Share Delivery Date), \$10 per Trading Day (increasing to \$20 per Trading Day on the fifth Trading Day after such liquidated damages begin to accrue) for each Trading Day after the Conversion Settlement Date until the cash amount set forth in Section 7(E)(iv)(1) is paid to the Holder or the shares of Common Stock are delivered to the Holder pursuant to Section 7(E)(iv)(2).

- (v) Effect of Conversion. If this Note is converted in full, then, from and after the date the Conversion Consideration therefor is issued or delivered in settlement of such conversion, this Note will cease to be outstanding and all interest will cease to accrue on this Note.
- (F) Common Stock Issued upon Conversion.
- (i) Status of Conversion Shares; Listing. Each share of Common Stock delivered pursuant to this Note will be a newly issued or treasury share and will be duly and validly issued, fully paid, non-assessable, free from preemptive rights and free of any Lien or adverse claim (except to the extent of any Lien or adverse claim created by the action or inaction of the Holder or the Person to whom such share will be delivered). If the Common Stock is then listed on any securities exchange, or quoted on any inter-dealer quotation system, then the Company will cause each share of Common Stock issued pursuant to this Note, when delivered, to be admitted for listing on such exchange or quotation on such system.

- (ii) Transferability of Conversion Shares. Any shares of Common Stock issued pursuant to this Note will be issued in the form of book-entries at the facilities of DTC, identified therein by an "unrestricted" CUSIP number.
- (G) Adjustments to the Conversion Rate.
- (i) Events Requiring an Adjustment to the Conversion Rate. The Conversion Rate will be adjusted from time to time as follows:
- (1) Stock Dividends, Splits and Combinations. If the Company issues solely shares of Common Stock as a dividend or distribution on all or substantially all shares of the Common Stock, or if the Company effects a stock split or a stock combination of the Common Stock (in each case excluding an issuance solely pursuant to a Common Stock Change Event, as to which Section 7(I) will apply), then the Conversion Rate will be adjusted based on the following formula:

$$CR_1 = CR_0 \square \frac{OS_1}{OS_1}$$

where:

- $CR_{\theta}$  = the Conversion Rate in effect immediately before the Open of Business on the Ex-Dividend Date for such dividend or distribution, or immediately before the Open of Business on the effective date of such stock split or stock combination, as applicable;
- $CR_I$  = the Conversion Rate in effect immediately after the Open of Business on such Ex-Dividend Date or the Open of Business on such effective date, as applicable;
- $OS_0$  = the number of shares of Common Stock outstanding immediately before the Open of Business on such Ex-Dividend Date or effective date, as applicable, without giving effect to such dividend, distribution, stock split or stock combination; and
- $OS_I$  = the number of shares of Common Stock outstanding immediately after giving effect to such dividend, distribution, stock split or stock combination.

If any dividend, distribution, stock split or stock combination of the type described in this **Section 7(G)(i)(1)** is declared or announced, but not so paid or made, then the Conversion Rate will be readjusted, effective as of the date the Board of Directors determines not to pay such dividend or distribution or to effect such stock split or stock combination, to the Conversion Rate that would then be in effect had such dividend, distribution, stock split or stock combination not been declared or announced.

(2) Rights, Options and Warrants. If the Company distributes, to all or substantially all holders of Common Stock, rights, options or warrants (other than rights issued or otherwise distributed pursuant to a stockholder rights plan, as to which the provisions set forth in Section 7(G)(v) will apply) entitling such holders, for a period

of not more than sixty (60) calendar days after the record date of such distribution, to subscribe for or purchase shares of Common Stock at a price per share that is less than the average of the Last Reported Sale Prices per share of Common Stock for the ten (10) consecutive Trading Days ending on, and including, the Trading Day immediately before the date such distribution is announced, then the Conversion Rate will be increased based on the following formula:

$$CR_1 = CR_0 \square \frac{OS + X}{OS + Y}$$

where:

 $CR_{\theta}$  = the Conversion Rate in effect immediately before the Open of Business on the Ex-Dividend Date for such distribution;

 $CR_I$  = the Conversion Rate in effect immediately after the Open of Business on such Ex-Dividend Date;

OS = the number of shares of Common Stock outstanding immediately before the Open of Business on such Ex-Dividend Date;

X = the total number of shares of Common Stock issuable pursuant to such rights, options or warrants; and

Y = a number of shares of Common Stock obtained by dividing (x) the aggregate price payable to exercise such rights, options or warrants by (y) the average of the Last Reported Sale Prices per share of Common Stock for the ten (10) consecutive Trading Days ending on, and including, the Trading Day immediately before the date such distribution is announced.

To the extent that shares of Common Stock are not delivered after the expiration of such rights, options or warrants (including as a result of such rights, options or warrants not being exercised), the Conversion Rate will be readjusted to the Conversion Rate that would then be in effect had the increase to the Conversion Rate for such distribution been made on the basis of delivery of only the number of shares of Common Stock actually delivered upon exercise of such rights, option or warrants. To the extent such rights, options or warrants are not so distributed, the Conversion Rate will be readjusted to the Conversion Rate that would then be in effect had the Ex-Dividend Date for the distribution of such rights, options or warrants not occurred.

For purposes of this **Section 7(G)(i)(2)**, in determining whether any rights, options or warrants entitle holders of Common Stock to subscribe for or purchase shares of Common Stock at a price per share that is less than the average of the Last Reported Sale Prices per share of Common Stock for the ten (10) consecutive Trading Days ending on, and including, the Trading Day immediately before the date the distribution of such rights, options or warrants is announced, and in determining the aggregate price payable to exercise such rights, options or warrants, there will be taken into account any consideration the Company receives for such rights, options or warrants and any amount payable on exercise thereof, with the value of such consideration, if not cash, to be determined by the Board of Directors in good faith.

- (3) Spin-Offs and Other Distributed Property.
- (a) Distributions Other than Spin-Offs. If the Company distributes shares of its Capital Stock, evidences of its indebtedness or other assets or property of the Company, or rights, options or warrants to acquire Capital Stock of the Company or other securities, to all or substantially all holders of the Common Stock, excluding:
  - (v) dividends, distributions, rights, options or warrants for which an adjustment to the Conversion Rate is required pursuant to Section 7(G)(i)(1) or Section 7(G)(i)(2);
  - (w) dividends or distributions paid exclusively in cash for which an adjustment to the Conversion Rate is required pursuant to **Section 7(G)(i)(4)**;
  - (x) rights issued or otherwise distributed pursuant to a stockholder rights plan, except to the extent provided in Section 7(G)(v);
  - (y) Spin-Offs for which an adjustment to the Conversion Rate is required pursuant to **Section 7(G)(i)(3)** (b); and
- (z) a distribution solely pursuant to a Common Stock Change Event, as to which **Section 7(I)** will apply, then the Conversion Rate will be increased based on the following formula:

$$CR_1 = CR_0 \square \frac{SP}{SP-FMV}$$

where:

- $CR_{\theta}$  = the Conversion Rate in effect immediately before the Open of Business on the Ex-Dividend Date for such distribution;
- $CR_I$  = the Conversion Rate in effect immediately after the Open of Business on such Ex-Dividend Date;
- SP = the average of the Last Reported Sale Prices per share of Common Stock for the ten (10) consecutive Trading Days ending on, and including, the Trading Day immediately before such Ex-Dividend Date; and
- FMV = the fair market value (as determined by the Board of Directors in good faith), as of such Ex-Dividend Date, of the shares of Capital Stock, evidences of indebtedness, assets, property, rights, options or warrants distributed per share of Common Stock pursuant to such distribution;

provided, however, that if FMV is equal to or greater than SP, then, in lieu of the foregoing adjustment to the Conversion Rate, the Holder will receive, for each \$1,000 Principal Amount of this Note held by this Holder on the record date for such distribution, at the same time and on the same terms as holders of Common Stock, the amount and kind of shares of Capital Stock, evidences of indebtedness, assets, property, rights, options or warrants that such Holder would have received if such Holder had owned, on such record date, a number of shares of Common Stock equal to the Conversion Rate in effect on such record date.

To the extent such distribution is not so paid or made, the Conversion Rate will be readjusted to the Conversion Rate that would then be in effect had the adjustment been made on the basis of only the distribution, if any, actually made or paid.

(b) Spin-Offs. If the Company distributes or dividends shares of Capital Stock of any class or series, or similar equity interest, of or relating to an Affiliate, a Subsidiary or other business unit of the Company to all or substantially all holders of the Common Stock (other than solely pursuant to a Common Stock Change Event, as to which Section 7(I) will apply), and such Capital Stock or equity interest is listed or quoted (or will be listed or quoted upon the consummation of the transaction) on a U.S. national securities exchange (a "Spin-Off"), then the Conversion Rate will be increased based on the following formula:

$$CR_1 = CR_0 \square \frac{FMV + SP}{SP}$$

where:

 $CR_{\theta}$  = the Conversion Rate in effect immediately before the Open of Business on the Ex-Dividend Date for such Spin-Off:

 $CR_I$  = the Conversion Rate in effect immediately after the Open of Business on such Ex-Dividend Date;

FMV = the product of (x) the average of the Last Reported Sale Prices per share or unit of the Capital Stock or equity interests distributed in such Spin-Off over the ten (10) consecutive Trading Day period (the "Spin-Off Valuation Period") beginning on, and including, such Ex-Dividend Date (such average to be determined as if references to Common Stock in the definitions of Last Reported Sale Price, Trading Day and Market Disruption Event were instead references to such Capital Stock or equity interests); and (y) the number of shares or units of such Capital Stock or equity interests distributed per share of Common Stock in such Spin-Off; and

SP = the average of the Last Reported Sale Prices per share of Common Stock for each Trading Day in the Spin-Off Valuation Period.

The adjustment to the Conversion Rate pursuant to this **Section 7(G)(i)(3)(b)** will be calculated as of the Close of Business on the last Trading Day of the Spin-Off Valuation Period but will be given effect immediately after the Open of Business on the Ex-Dividend Date for the Spin-Off, with retroactive effect. If this Note is converted and the Conversion Date occurs during the Spin-Off Valuation Period, then, notwithstanding anything to the contrary in this Note, the Company will, if necessary, delay the settlement of such conversion until the second (2nd) Business Day after the last day of the Spin-Off Valuation Period.

To the extent any dividend or distribution of the type set forth in this **Section 7(G)(i)(3)(b)** is declared but not made or paid, the Conversion Rate will be readjusted to the Conversion Rate that would then be in effect had the adjustment been made on the basis of only the dividend or distribution, if any, actually made or paid.

(4) Cash Dividends or Distributions. If any cash dividend or distribution is made to all or substantially all holders of Common Stock, then the Conversion Rate will be increased based on the following formula:

$$CR_1 = CR_0 \square \frac{SP}{SP-D}$$

where:

 $CR_{\theta}$  = the Conversion Rate in effect immediately before the Open of Business on the Ex-Dividend Date for such dividend or distribution;

 $CR_I$  = the Conversion Rate in effect immediately after the Open of Business on such Ex-Dividend Date;

SP = the Last Reported Sale Price per share of Common Stock on the Trading Day immediately before such Ex-Dividend Date; and

D = the cash amount distributed per share of Common Stock in such dividend or distribution;

provided, however, that if D is equal to or greater than SP, then, in lieu of the foregoing adjustment to the Conversion Rate, the Holder will receive, for each \$1,000 Principal Amount of this Note held by the Holder on the record date for such dividend or distribution, at the same time and on the same terms as holders of Common Stock, the amount of cash that such Holder would have received if such Holder had owned, on such record date, a number of shares of Common Stock equal to the Conversion Rate in effect on such record date.

To the extent such dividend or distribution is declared but not made or paid, the Conversion Rate will be readjusted to the Conversion Rate that would then be in effect had the adjustment been made on the basis of only the dividend or distribution, if any, actually made or paid.

(5) Tender Offers or Exchange Offers. If the Company or any of its Subsidiaries makes a payment in respect of a tender offer or exchange offer for shares of

Common Stock (other than solely pursuant to an odd-lot tender offer pursuant to Rule 13e-4(h)(5) under the Exchange Act), and the value (determined as of the Expiration Time by the Board of Directors in good faith) of the cash and other consideration paid per share of Common Stock in such tender or exchange offer exceeds the Last Reported Sale Price per share of Common Stock on the Trading Day immediately after the last date (the "Expiration Date") on which tenders or exchanges may be made pursuant to such tender or exchange offer (as it may be amended), then the Conversion Rate will be increased based on the following formula:

$$CR = CR_0 \square \frac{AC + (SP \square OS)}{SP \square OS}$$

where:

 $CR_0$  = the Conversion Rate in effect immediately before the time (the "**Expiration Time**") such tender or exchange offer expires;

 $CR_1$  = the Conversion Rate in effect immediately after the Expiration Time;

AC = the aggregate value (determined as of the Expiration Time by the Board of Directors in good faith) of all cash and other consideration paid for shares of Common Stock purchased or exchanged in such tender or exchange offer;

 $OS_0$  = the number of shares of Common Stock outstanding immediately before the Expiration Time (including all shares of Common Stock accepted for purchase or exchange in such tender or exchange offer);

OS<sub>1</sub> = the number of shares of Common Stock outstanding immediately after the Expiration Time (excluding all shares of Common Stock accepted for purchase or exchange in such tender or exchange offer); and

SP = the average of the Last Reported Sale Prices per share of Common Stock over the ten (10) consecutive Trading Day period (the "**Tender/Exchange Offer Valuation Period**") beginning on, and including, the Trading Day immediately after the Expiration Date;

provided, however, that the Conversion Rate will in no event be adjusted down pursuant to this **Section 7(G)(i)(5)**, except to the extent provided in the immediately following paragraph. The adjustment to the Conversion Rate pursuant to this **Section 7(G)(i)** (5) will be calculated as of the Close of Business on the last Trading Day of the Tender/Exchange Offer Valuation Period but will be given effect immediately after the Expiration Time, with retroactive effect. If this Note is converted and the Conversion Date occurs on the Expiration Date or during the Tender/Exchange Offer Valuation Period, then, notwithstanding anything to the contrary in this Note, the Company will, if necessary, delay the settlement of such conversion until the second (2nd) Business Day after the last day of the Tender/Exchange Offer Valuation Period.

To the extent such tender or exchange offer is announced but not consummated (including as a result of the Company being precluded from consummating such tender or exchange offer under applicable law), or any purchases or exchanges of shares of Common Stock in such tender or exchange offer are rescinded, the Conversion Rate will be readjusted to the Conversion Rate that would then be in effect had the adjustment been made on the basis of only the purchases or exchanges of shares of Common Stock, if any, actually made, and not rescinded, in such tender or exchange offer.

- (ii) No Adjustments in Certain Cases.
- (1) Where the Holder Participates in the Transaction or Event Without Conversion. Notwithstanding anything to the contrary in Section 7(G)(i), the Company will not be obligated to adjust the Conversion Rate on account of a transaction or other event otherwise requiring an adjustment pursuant to Section 7(G)(i) (other than a stock split or combination of the type set forth in Section 7(G)(i)(1) or a tender or exchange offer of the type set forth in Section 7(G)(i)(5)) if the Holder participates, at the same time and on the same terms as holders of Common Stock, and solely by virtue of being the Holder of this Note, in such transaction or event without having to convert this Note and as if the Holder held a number of shares of Common Stock equal to the product of (i) the Conversion Rate in effect on the related record date; and (ii) the aggregate Principal Amount (expressed in thousands) of this Note held by this Holder on such date.
- (2) Certain Events. The Company will not be required to adjust the Conversion Rate except as provided in **Section 7(G)** and **Section 7(I)**. Without limiting the foregoing, the Company will not be obligated to adjust the Conversion Rate on account of:
  - (a) except as otherwise provided in **Section 7(G)**, the sale of shares of Common Stock for a purchase price that is less than the market price per share of Common Stock or less than the Conversion Price;
  - (b) the issuance of any shares of Common Stock pursuant to any present or future plan providing for the reinvestment of dividends or interest payable on the Company's securities and the investment of additional optional amounts in shares of Common Stock under any such plan;
  - (c) the issuance of any shares of Common Stock, restricted stock or options or rights to purchase shares of Common Stock pursuant to any present or future employee, director or consultant benefit plan or program of, or assumed by, the Company or any of its Subsidiaries;
  - (d) the issuance of any shares of Common Stock pursuant to any option, warrant, right or convertible or exchangeable security of the Company outstanding as of the Issue Date (other than an adjustment pursuant to **Section 7(G)(i)(3)(a)** in connection with the separation of rights under the Company's stockholder rights plan existing, if any, as of the Issue Date);
  - (e) repurchases of shares of Common Stock, including structured or derivative transactions, that are not pursuant to a tender offer as contemplated by Section 7(G)(i)(5);

- (f) solely a change in the par value of the Common Stock; or
- (g) accrued and unpaid interest on this Note.
- (iii) Adjustments Not Yet Effective. Notwithstanding anything to the contrary in this Note, if:
  - (1) this Note is to be converted;
- (2) the record date, effective date or Expiration Time for any event that requires an adjustment to the Conversion Rate pursuant to **Section 7(G)(i)** has occurred on or before the Conversion Date for such conversion, but an adjustment to the Conversion Rate for such event has not yet become effective as of such Conversion Date;
  - (3) the Conversion Consideration due upon such conversion includes any whole shares of Common Stock; and
- (4) such shares are not entitled to participate in such event (because they were not held on the related record date or otherwise),

then, solely for purposes of such conversion, the Company will, without duplication, give effect to such adjustment on such Conversion Date. In such case, if the date on which the Company is otherwise required to deliver the consideration due upon such conversion is before the first date on which the amount of such adjustment can be determined, then the Company will delay the settlement of such conversion until the second (2nd) Business Day after such first date.

- (iv) Conversion Rate Adjustments where the Converting Holder Participates in the Relevant Transaction or Event. Notwithstanding anything to the contrary in this Note, if:
- (1) a Conversion Rate adjustment for any dividend or distribution becomes effective on any Ex-Dividend Date pursuant to **Section 7(G)(i)**;
  - (2) this Note is to be converted;
- (3) the Conversion Date for such conversion occurs on or after such Ex-Dividend Date and on or before the related record date;
- (4) the Conversion Consideration due upon such conversion includes any whole shares of Common Stock based on a Conversion Rate that is adjusted for such dividend or distribution; and
- (5) such shares would be entitled to participate in such dividend or distribution (including pursuant to **Section 7(C)** (ii)),

then (x) such Conversion Rate adjustment will not be given effect for such conversion; (y) the shares of Common Stock issuable upon such conversion based on such unadjusted Conversion Rate will not be entitled to participate in such dividend or distribution; and (z) there will be added, to the Conversion Consideration otherwise due upon such conversion, the same kind and amount of consideration that would have been delivered in such dividend or distribution with respect to such shares of Common Stock had such shares been entitled to participate in such dividend or distribution.

- (v) Stockholder Rights Plans. If any shares of Common Stock are to be issued upon conversion of this Note and, at the time of such conversion, the Company has in effect any stockholder rights plan, then the Holder of this Note will be entitled to receive, in addition to, and concurrently with the delivery of, the Conversion Consideration otherwise payable under this Note upon such conversion, the rights set forth in such stockholder rights plan, unless such rights have separated from the Common Stock at such time, in which case, and only in such case, the Conversion Rate will be adjusted pursuant to Section 7(G)(i)(3)(a) on account of such separation as if, at the time of such separation, the Company had made a distribution of the type referred to in such Section to all holders of the Common Stock, subject to readjustment in accordance with such Section if such rights expire, terminate or are redeemed.
- (vi) Limitation on Effecting Transactions Resulting in Certain Adjustments. The Company will not engage in or be a party to any transaction or event that would require the Conversion Rate to be adjusted pursuant to **Section 7(G)(i)** or **Section 7(I)** to an amount that would result in the Conversion Price per share of Common Stock being less than the par value per share of Common Stock.
- (vii) Equitable Adjustments to Prices. Whenever any provision of this Note requires the Company to calculate the average of the Last Reported Sale Prices, or any function thereof, over a period of multiple days (including to calculate an adjustment to the Conversion Rate), the Company will make proportionate adjustments, if any, to such calculations to account for any adjustment to the Conversion Rate pursuant to Section 7(G)(i) that becomes effective, or any event requiring such an adjustment to the Conversion Rate where the Ex-Dividend Date or effective date, as applicable, of such event occurs, at any time during such period.
- (viii) Calculation of Number of Outstanding Shares of Common Stock. For purposes of this **Section 7(G)**, the number of shares of Common Stock outstanding at any time will (i) include shares issuable in respect of scrip certificates issued in lieu of fractions of shares of Common Stock; and (ii) exclude shares of Common Stock held in the Company's treasury (unless the Company pays any dividend or makes any distribution on shares of Common Stock held in its treasury).
- (ix) *Calculations*. All calculations with respect to the Conversion Rate and adjustments thereto will be made to the nearest 1/10,000th of a share of Common Stock (with 5/100,000ths rounded upward).
- (x) Notice of Conversion Rate Adjustments. Upon the effectiveness of any adjustment to the Conversion Rate pursuant to **Section 7(G)(i)**, the Company will promptly send notice to the Holder containing (i) a brief description of the transaction or other event on account of which such adjustment was made; (ii) the Conversion Rate in effect immediately after such adjustment; and (iii) the effective time of such adjustment.

## (H) Voluntary Adjustments.

(i) Generally. To the extent permitted by law and applicable stock exchange rules, the Company, from time to time, may (but is not required to) increase the Conversion Rate on any portion of this Note for any period of time by any amount if (i) the Board of Directors determines in good faith that such increase is either (x) in the best interest of the Company; or (y) advisable to avoid or diminish any income tax imposed on holders of Common Stock or rights to purchase Common Stock as a result of any dividend or distribution of shares (or rights to acquire shares) of Common Stock or any similar event and (ii) such increase is irrevocable for the period of time specified by the

Board of Directors. The Company and the Holder agree that any such voluntary adjustment to the Conversion Rate and any conversion of any portion of this Note based upon any such voluntary adjustment shall not constitute material non-public information with respect to the Company.

- (ii) *Notice of Voluntary Increases*. If the Board of Directors determines to increase the Conversion Rate pursuant to **Section 7(H)(i)**, then, no later than the first Business Day following such determination, the Company will send notice to the Holder of such increase, the amount thereof and the period during which such increase will be in effect.
- (I) Effect of Certain Recapitalizations, Reclassifications, Consolidations, Mergers and Sales.
  - (i) Generally. If there occurs:
- (1) recapitalization, reclassification or change of the Common Stock (other than (x) changes solely resulting from a subdivision or combination of the Common Stock, (y) a change only in par value or from par value to no par value or no par value to par value and (z) stock splits and stock combinations that do not involve the issuance of any other series or class of securities);
  - (2) consolidation, merger, combination or binding or statutory share exchange involving the Company;
  - (3) sale, lease or other transfer of all or substantially all of the assets of the Company and its Subsidiaries, taken as a whole, to any Person; or
    - (4) other similar event,

and, in each case, as a result of such occurrence, the Common Stock is converted into, or is exchanged for, or represents solely the right to receive, other securities or other property (including cash or any combination of the foregoing) (such an event, a "Common Stock Change Event," and such other securities or other property, the "Reference Property," and the amount and kind of Reference Property that a holder of one (1) share of Common Stock would be entitled to receive on account of such Common Stock Change Event (without giving effect to any arrangement not to issue fractional shares of securities or other property), a "Reference Property Unit"), then, notwithstanding anything to the contrary in this Note, at the effective time of such Common Stock Change Event, (x) the Conversion Consideration due upon conversion of this Note will be determined in the same manner as if each reference to any number of shares of Common Stock in this Section 7 (or in any related definitions) were instead a reference to the same number of Reference Property Units; (y) for purposes of Section 7(A), each reference to any number of shares of Common Stock in such Section (or in any related definitions) will instead be deemed to be a reference to the same number of Reference Property Units; and (z) for purposes of the definition of "Fundamental Change," the term "Common Stock" and "common equity" will be deemed to mean the common equity, if any, forming part of such Reference Property. For these purposes, (I) the Daily VWAP of any Reference Property Unit or portion thereof that consists of a class of securities will be determined by reference to the definition of "Daily VWAP," substituting, if applicable, the Bloomberg page for such class of securities in such definition; and (II) the Daily VWAP of any Reference Property Unit or portion thereof that does not consist of a class of common equity securities, and the Last Reported Sale Price of any Reference Property Unit or portion thereof that does not

consist of a class of securities, will be the fair value of such Reference Property Unit or portion thereof, as applicable, determined in good faith by the Company (or, in the case of cash denominated in U.S. dollars, the face amount thereof).

If the Reference Property consists of more than a single type of consideration to be determined based in part upon any form of stockholder election, then the composition of the Reference Property Unit will be deemed to be the weighted average of the types and amounts of consideration actually received, per share of Common Stock, by the holders of Common Stock. The Company will notify the Holder of such weighted average as soon as practicable after such determination is made.

At or before the effective date of such Common Stock Change Event, the Company and the resulting, surviving or transferee Person (if not the Company) of such Common Stock Change Event (the "Successor Person") will execute and deliver such instruments or agreements that (x) provides for subsequent conversions of this Note in the manner set forth in this Section 7(I); (y) provides for subsequent adjustments to the Conversion Rate pursuant to Section 7(G) and Section 7(H) in a manner consistent with this Section 7(I); and (z) contains such other provisions as the Company reasonably determines are appropriate to preserve the economic interests of the Holder and to give effect to the provisions of this Section 7(I). If the Reference Property includes shares of stock or other securities or assets of a Person other than the Successor Person, then such other Person will also execute such instruments or agreements and such instruments or agreements will contain such additional provisions the Company reasonably determines are appropriate to preserve the economic interests of the Holder.

- (ii) Notice of Common Stock Change Events. As soon as practicable after learning the anticipated or actual effective date of any Common Stock Change Event, the Company will provide written notice to the Holder of such Common Stock Change Event, including a brief description of such Common Stock Change Event, its anticipated effective date and a brief description of the anticipated change in the conversion right of this Note.
- (iii) Compliance Covenant. The Company will not become a party to any Common Stock Change Event unless its terms are consistent with this **Section 7(I)**.

#### Limitations on Conversions.

(i) Beneficial Ownership Limitation. Notwithstanding anything to the contrary contained herein, the Company shall not effect the conversion of any portion of this Note, or otherwise issue shares pursuant to this Note, and the Holder shall not have the right to convert any portion of this Note, pursuant to the terms and conditions of this Note and any such conversion or issuance shall be null and void and treated as if never made, to the extent that after giving effect to such conversion or issuance, the Holder together with the other Attribution Parties collectively would beneficially own in the aggregate in excess of 4.99% (the "Maximum Percentage") of the number of shares of Common Stock outstanding immediately after giving effect to such conversion or issuance. For purposes of the foregoing sentence, the aggregate number of shares of Common Stock beneficially owned by the Holder and the other Attribution Parties shall include the number of shares of Common Stock held by the Holder and all other Attribution Parties plus the number of shares of Common Stock issuable upon conversion of, or otherwise pursuant to, this Note with respect to which the determination of such sentence is being made, but shall exclude the number of shares of Common Stock which would be issuable upon (A) conversion of the remaining, unconverted portion of this Note beneficially owned by the Holder or any of the other Attribution Parties and (B) exercise or conversion of the unexercised or

unconverted portion of any other securities of the Company (including, without limitation, any convertible notes or convertible preferred stock or warrants) beneficially owned by the Holder or any other Attribution Party subject to a limitation on conversion or exercise analogous to the limitation contained in this Section 7(J)(i). For purposes of this Section 7(J)(i), beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act. For purposes of this Note, in determining the number of outstanding shares of Common Stock the Holder may acquire in connection with this Note without exceeding the Maximum Percentage, the Holder may rely on the number of outstanding shares of Common Stock as reflected in (x) the Company's most recent Annual Report on Form 10-K, Quarterly Report on Form 10-Q, Current Report on Form 8-K or other public filing with the Commission, as the case may be, (y) a more recent public announcement by the Company or (z) any other written notice by the Company or the Transfer Agent (as defined in the Securities Purchase Agreement) setting forth the number of shares of Common Stock outstanding (the "Reported Outstanding Share Number"). If the Company receives a notice from the Holder related to the conversion of this Note or any issuance of shares of Common Stock in connection with this Note at a time when the actual number of outstanding shares of Common Stock is less than the Reported Outstanding Share Number, the Company shall promptly notify the Holder in writing of the number of shares of Common Stock then outstanding and, to the extent that such conversion or issuance of shares of Common Stock would otherwise cause the Holder's beneficial ownership, as determined pursuant to this Section 7(J)(i), to exceed the Maximum Percentage, the Holder must notify the Company of a reduced number of shares of Common Stock to be issued pursuant to such notice. For any reason at any time, upon the written or oral request of the Holder, the Company shall within one (1) Trading Day confirm in writing or by electronic mail to the Holder the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to the conversion or exercise of securities of the Company, including this Note, by the Holder and any other Attribution Party since the date as of which the Reported Outstanding Share Number was reported. In the event that the issuance of shares of Common Stock to the Holder upon conversion of, or otherwise pursuant to, this Note results in the Holder and the other Attribution Parties being deemed to beneficially own, in the aggregate, more than the Maximum Percentage of the number of outstanding shares of Common Stock (as determined under Section 13(d) of the Exchange Act), the number of shares so issued by which the Holder's and the other Attribution Parties' aggregate beneficial ownership exceeds the Maximum Percentage (the "Excess Shares") shall be deemed null and void and shall be cancelled ab initio, and the Holder shall not have the power to vote or to transfer the Excess Shares; provided, however that, the aggregate number of shares of Common Stock that would otherwise be required to be issued to the Holder pursuant to the terms of this Note shall be held in abeyance for the benefit of the Holder until such time or times as its right thereto would not result in the Holder and the other Attribution Parties exceeding the Maximum Percentage, at which time or times the Holder shall be issued such shares (and any such shares declared or made on such initial issuance or on any subsequent issuance held similarly in abeyance) to the same extent as if there had been no such limitation. Upon delivery of a written notice to the Company, the Holder may from time to time increase or decrease the Maximum Percentage to any other percentage not in excess of 9.99% as specified in such notice; provided that (i) any such increase in the Maximum Percentage will not be effective until the sixty-first (61st) day after such notice is delivered to the Company and (ii) any such increase or decrease will apply only to the Holder and the other Attribution Parties and not to any Other Holder of Notes that is not an Attribution Party of the Holder. For purposes of clarity, the shares of Common Stock issuable pursuant to the terms of this Note in excess of the Maximum Percentage shall not be deemed to be beneficially owned by the Holder for any purpose including for purposes of Section 13(d) or Rule 16a-1(a)(1) of the Exchange Act. No prior inability to convert this Note or receive shares pursuant to this Note pursuant to this paragraph shall have any effect on the

applicability of the provisions of this paragraph with respect to any subsequent determination of convertibility. The provisions of this paragraph shall be construed and implemented in a manner otherwise than in strict conformity with the terms of this Section 7(J)(i) to the extent necessary to correct this paragraph or any portion of this paragraph which may be defective or inconsistent with the intended beneficial ownership limitation contained in this Section 7(J)(i) or to make changes or supplements necessary or desirable to properly give effect to such limitation. The limitation contained in this paragraph may not be waived and shall apply to a successor holder of this Note. The Holder hereby acknowledges and agrees that the Company shall be entitled to rely on the representations and other information set forth in any notice from the Holder related to the conversion of this Note or any issuance of Shares of Common Stock in connection with this Note and shall not be required to independently verify whether any issuance of Common Stock pursuant to this Note would cause the Holder (together with the other Attribution Parties) to collectively beneficially own in excess of the Maximum Percentage of the number of shares of Common Stock outstanding after giving effect to such conversion or otherwise trigger the provisions of this Section 7(J)(i). The Trustee shall have no obligation to monitor any limitations on beneficial ownership in connection with this Section 7(J)(i).

(ii) Stock Exchange Limitations. Notwithstanding anything to the contrary in this Note or any Other Note, until the Requisite Stockholder Approval is obtained, in no event will the number of shares of Common Stock issuable upon conversion or otherwise pursuant to this Note and all Other Notes, together with all other shares, if any theretofore issued upon conversion or otherwise pursuant to this Note and all Other Notes, including (for the avoidance of doubt) Event of Default Stock Payments exceed **[insert # representing one share less share than 20% of the outstanding shares]** shares in the aggregate. If any one or more shares of Common Stock are not delivered as a result of the operation of the preceding sentence (such shares, the "Withheld Shares"), then (1) on the date such shares of Common Stock are issuable hereunder (without effect to any limitations imposed under Section 7(J)(i)), the Company will pay to the Holder, in addition to the Conversion Consideration otherwise due upon such conversion or shares otherwise due to the Holder hereunder, cash in an amount equal to the product of (x) the number of such Withheld Shares; and (y) the Daily VWAP per share of Common Stock on the applicable Stock Payment Determination Date; and (2) to the extent the Holder purchases (in an open market transaction or otherwise) Common Stock to deliver in settlement of a sale by the Holder of such Withheld Shares, the Company will reimburse the Holder for (x) any brokerage commissions and other out-of-pocket expenses, if any, of the Holder incurred in connection with such purchases and (y) the excess, if any, of (A) the aggregate purchase price of such purchases over (B) the product of (I) the number of such Withheld Shares purchased by the Holder; and (II) the Daily VWAP per share of Common Stock on the applicable Stock Payment Determination Date.

### **Section 8.** Affirmative and Negative Covenants.

- (A) Stay, Extension and Usury Laws. To the extent that it may lawfully do so, the Company (A) agrees that it will not at any time insist upon, plead, or in any manner whatsoever claim or take the benefit or advantage of, any stay, extension or usury law (wherever or whenever enacted or in force) that may affect the covenants or the performance of this Note; and (B) expressly waives all benefits or advantages of any such law and agrees that it will not, by resort to any such law, hinder, delay or impede the execution of any power granted to the Holder by this Note, but will suffer and permit the execution of every such power as though no such law has been enacted.
  - (B) Corporate Existence. Subject to Section 9, the Company will cause to preserve and keep in full force and effect:

- (i) its corporate existence and the corporate existence of its Subsidiaries in accordance with the organizational documents of the Company or its Subsidiaries, as applicable; and
  - (ii) the material rights (charter and statutory), licenses and franchises of the Company and its Subsidiaries;

provided, however, that the Company need not preserve or keep in full force and effect any such rights (charter and statutory), license or franchise or existence of any of its Subsidiaries if the Board of Directors determines in good faith that (x) the preservation thereof is no longer desirable in the conduct of the business of the Company and its Subsidiaries, taken as a whole; and (v) the loss thereof is not, individually or in the aggregate, materially adverse to the Holder; provided further, that any Subsidiary may merge into, amalgamate or consolidate with any other Subsidiary and any Subsidiary may liquidate or dissolve if all of its property passes to the Company or another Subsidiary.

- (C) Ranking. All payments due under this Note shall rank (i) pari passu with all Other Notes, (ii) effectively senior to all unsecured obligations of the Company to the extent of the value of the Collateral securing the Notes for so long as the Collateral so secures the Notes in accordance with the terms hereof and (iii) senior to any Subordinated Indebtedness.
- (D) Indebtedness; Amendments to Indebtedness. The Company shall not and shall not permit any Subsidiary to: (a) create, incur, assume, guarantee or be or remain liable with respect to any Indebtedness, other than Permitted Indebtedness; (b) prepay any Indebtedness except by the conversion of Indebtedness into equity securities (other than Disqualified Stock) and the payment of cash in lieu of fractional shares in connection with such conversion; or (c) amend or modify any documents or notes evidencing any Indebtedness in any manner which shortens the maturity date or any amortization, redemption or interest payment date thereof or otherwise imposes materially more burdensome terms upon the Company or its Subsidiaries than exist in such Indebtedness prior to such amendment or modification without the prior written consent of Holder. The Company shall not and shall not permit any Subsidiary to incur any Indebtedness that would cause a breach or Default under the Notes or prohibit or restrict the performance of any of the Company's or its Subsidiaries' obligations under the Notes, including without limitation, the payment of interest and principal thereon.
- (E) Liens. The Company will not, and will not permit any of its Subsidiaries to, directly or indirectly, create, incur, assume or permit to exist any Lien of any kind on any asset now owned or hereafter acquired, except Permitted Liens.
- (F) Investments. The Company shall not directly or indirectly acquire or own, or make any Investment in or to any Person, or permit any of its Subsidiaries so to do, other than Permitted Investments; provided that the Company may not make any Investment (including a Permitted Investment) or permit any of its Subsidiaries to make any Investment (including a Permitted Investment) if (i) any Event of Default has occurred hereunder or (ii) any event or circumstance has occurred and is continuing which, with the giving of notice or passage of time or both, could constitute an Event of Default with respect to Section 10(A)(ii), Section 10(A)(iv), Section 10(A)(vi), Section 10(A)(xiv).
- (G) Distributions. The Company shall not, and shall not allow any Subsidiary to, (a) repurchase or redeem any class of stock or other Equity Interest other than pursuant to employee, director or consultant equity incentive, stock purchase or repurchase plans or awards granted thereunder or other similar agreements provided under plans approved by the Board of Directors; provided, however, in each case the repurchase or redemption price does not exceed

the original consideration paid for such stock or Equity Interest, or (b) declare or pay any cash dividend or make a cash distribution on any class of stock or other Equity Interest, except that a Subsidiary of the Company may pay dividends or make distributions to the Company or a parent company that is a direct or indirect Wholly Owned Subsidiary of the Company, or (c) lend money to any employees, officers or directors (except as permitted under clause (F) of the definition of Permitted Investment), or guarantee the payment of any such loans granted by a third party in excess of two hundred fifty thousand dollars (\$250,000) in the aggregate or (d) waive, release or forgive any Indebtedness owed by any employees, officers or directors in excess of two hundred fifty thousand dollars (\$250,000) in the aggregate. If there are dividends or distributions made by the Company or any Subsidiary, within one (1) Business Day following the date on which the Company files an Annual Report on Form 10-K or Quarterly Report on Form 10-Q with the Commission, the Company will provide the Holder with a written notice setting forth the aggregate amount of dividends or distributions made by the Company or any Subsidiary pursuant to this Section 8(G) for the period covered by such Annual Report on Form 10-K or Quarterly Report on Form 10-Q, as applicable. Notwithstanding anything herein to the contrary, the Company shall not, and shall not allow any Subsidiary to, declare or pay any cash dividend or make a cash distribution on any class of stock or other Equity Interest if (A) any Event of Default has occurred hereunder and has not been waived by the Required Holders or (B) any event or circumstance has occurred and is continuing which, with the giving of notice or passage of time or both, could constitute an Event of Default with respect to Section 10(A)(ii), Section 10(A)(iv), Section 10(A)(ix), Section 10(A)(ix)

- (H) *Transfers*. The Company shall not, and shall not allow any Subsidiary to, voluntarily or involuntarily transfer, sell, lease, license, lend or in any other manner convey any equitable, beneficial or legal interest in any material portion of the assets of the Company and its Subsidiaries (taken as a whole), except for Permitted Transfers and Permitted Investments.
- (I) Taxes. The Company and its Subsidiaries shall pay when due all taxes, fees or other charges of any nature whatsoever (together with any related interest or penalties) now or hereafter imposed or assessed against the Company and its Subsidiaries or their respective assets or upon their ownership, possession, use, operation or disposition thereof or upon their rents, receipts or earnings arising therefrom (except where the failure to pay would not, individually or in the aggregate, have a material effect on the Company or any of its Subsidiaries). The Company and its Subsidiaries shall file on or before the due date therefor all personal property tax returns (except where the failure to pay would not, individually or in the aggregate, have a material effect on the Company or any of its Subsidiaries). Notwithstanding the foregoing, the Company and its Subsidiaries may contest, in good faith and by appropriate proceedings, taxes for which they maintain adequate reserves therefor in accordance with GAAP.

#### (J) Minimum Liquidity.

- (i) The Company and its Subsidiaries shall have at all times liquidity calculated as unrestricted, unencumbered Cash and Cash Equivalents in one or more deposit accounts located in the United States in a minimum amount equal to thirty million dollars (\$30,000,000) and, subject to **Section 8(Y)**, subject to one or more control agreements entered into in favor of the Collateral Agent (each a "Controlled Account").
- (ii) On the first (1st) Business Day of each month (or, if earlier, immediately in the event an Event of Default has occurred as a result of a breach of **Section 8(J)(i)**, **Section 8(G)**, **Section 8(V)**, **Section 8(W)** or **Section 8(Z)** hereof), the Company shall provide to the Holder and the Trustee a certification, in the form attached hereto as **Exhibit C**, executed on behalf of the Company by the Chief Financial Officer of the Company, certifying whether or not the Company has satisfied the requirements of **Section**

- **8(J)(i)**, Section **8(G)**, Section **8(W)** and Section **8(Z)** during the immediately preceding calendar month (a "Compliance Certificate"), and on the day the Company files its Annual Report on Form 10-K or Quarterly Report on Form 10-Q, the Company shall provide to the Holder a Compliance Certificate certifying whether or not the Company has satisfied the requirements of Section **8(V)** during the last calendar quarter reported in such Annual Report on Form 10-K or Quarterly Report on Form 10-Q. If the Company determines in its sole discretion that such information constitutes material non-public information, then the Company will so indicate in the certification provided pursuant to the preceding sentence and the Company will concurrently disclose such material non-public information on a Current Report on Form 8-K or otherwise.
- (K) Change in Nature of Business. The Company shall not, and the Company shall cause each of its Subsidiaries to not, directly or indirectly, engage in any material line of business substantially different from those lines of business conducted by or publicly contemplated to be conducted by the Company and each of its Subsidiaries on the Issue Date or any business substantially related or incidental thereto.
- (L) Maintenance of Properties, Etc. The Company shall maintain and preserve, and the Company shall cause each of its Subsidiaries to maintain and preserve, all of its properties which are necessary or useful (as determined by the Company in good faith) to the conduct of its business in good working order and condition, ordinary wear and tear excepted, and comply at all times in all material respects with the provisions of all leases to which it is a party as lessee or under which it occupies property, so as to prevent any loss or forfeiture thereof or thereunder.
- (M) Maintenance of Intellectual Property. The Company will take, and the Company shall cause each of its Subsidiaries to take, all actions necessary or advisable to maintain and preserve all of the Intellectual Property Rights (as defined in the Securities Purchase Agreement) of the Company or such Subsidiary that are necessary or material (as determined by the Company in good faith) to the conduct of its business in full force and effect.
- (N) Maintenance of Insurance. The Company shall maintain, and the Company shall cause each of its Subsidiaries to maintain, insurance with responsible and reputable insurance companies or associations (including, without limitation, comprehensive general liability, hazard, rent and business interruption insurance) with respect to its properties (including all real properties leased or owned by it) and business, in such amounts and covering such risks as is required by any governmental authority having jurisdiction with respect thereto or as is carried generally in accordance with customary business practice by companies in similar businesses similarly situated.
- (O) Transactions with Affiliates. Neither the Company, nor any of its Subsidiaries, shall enter into, renew, extend or be a party to, any transaction or series of related transactions (including, without limitation, the purchase, sale, lease, transfer or exchange of property or assets of any kind or the rendering of services of any kind) with any affiliate (other than the Company or any of its Wholly Owned Subsidiaries), except transactions for fair consideration and on terms no less favorable to it than would be obtainable in a comparable arm's length transaction with a Person that is not an affiliate thereof.
- (P) Restricted Issuances. The Company shall not, and shall cause its Subsidiaries not to, directly or indirectly, without the prior written consent of the Required Holders, (i) issue any Notes (other than as contemplated by the Securities Purchase Agreement and the Notes) or (ii) issue any other securities or incur any Indebtedness, in each case, that would cause a breach or Default under the Notes or that by its terms would prohibit or restrict the performance of any of the Company's or its Subsidiaries' obligations under the Notes, including, without limitation, the payment of principal thereon.

- (Q) Share Reserve. So long as this Note remains outstanding, the Company shall at all times have no less than a number of authorized but unissued shares of Common Stock equal to the sum of (i) fifty million (50,000,000) shares of Common Stock which shall not be exclusively reserved for issuance pursuant to the Notes and (ii) one hundred percent (100%) of a fraction, the numerator of which shall be the then outstanding principal amount of all Notes, and the denominator of which shall be the Conversion Price which shall be reserved for issuance pursuant to the terms of the Notes (the "Required Reserve Amount"); provided that at no time shall the number of shares of Common Stock reserved pursuant to this Section 8(Q) be reduced other than in connection with any stock combination, reverse stock split or other similar transaction. If at any time the number of shares of Common Stock authorized and reserved for issuance is not sufficient to meet the Required Reserve Amount, the Company will promptly take all corporate action necessary to authorize and reserve a sufficient number of shares, including, without limitation, calling a special meeting of stockholders to authorize additional shares to meet the Company's obligations pursuant to the Transaction Documents, in the case of an insufficient number of authorized shares, obtain stockholder approval (if required) of an increase in such authorized number of shares, and voting the management shares of the Company in favor of an increase in the authorized shares of the Company to ensure that the number of authorized shares is sufficient to meet the Required Reserve Amount.
- Independent Investigation. At the request of the Required Holders at any time the Required Holders have determined in good faith that (i) an Event of Default has occurred or (ii) any event or circumstance has occurred and is continuing which, with the giving of notice or passage of time or both, could constitute an Event of Default but the Company has not timely agreed to such determination in writing, the Company shall hire an independent, reputable investment bank selected by the Company and approved by the Required Holders to investigate as to whether such Event of Default or event or circumstance has occurred (the "Independent Investigator"). If the Independent Investigator determines that such Event of Default or event or circumstance has occurred, the Independent Investigator shall notify the Company of such Event of Default or occurrence of such event or circumstance and the Company shall promptly deliver written notice to the Holder of such Event of Default if such Event of Default has occurred. In connection with such investigation, the Independent Investigator may, during normal business hours and upon signing a confidentiality agreement in a form reasonably acceptable to the Company, inspect all contracts, books, records, personnel, offices and other facilities and properties of the Company and its Subsidiaries and, to the extent available to the Company after the Company uses reasonable efforts to obtain them, the records of its accountants (including the accountants' work papers) and any books of account, records, reports and other papers not contractually required of the Company to be confidential or secret, or subject to attorney-client or other evidentiary privilege, and the Independent Investigator may make such copies and inspections thereof as the Independent Investigator may reasonably request. The Company shall furnish the Independent Investigator with such financial and operating data and other information with respect to the business and properties of the Company as the Independent Investigator may reasonably request. The Company shall permit the Independent Investigator to discuss the affairs, finances and accounts of the Company with, and to make proposals and furnish advice with respect thereto to, any of the Company's officers, directors, key employees and independent public accountants (and by this provision the Company authorizes said accountants to discuss with such Independent Investigator the finances and affairs of the Company and any Subsidiaries), all at such reasonable times, upon reasonable notice, and as often as may be reasonably requested.
- (S) Upon delivery by the Company to the Holder (or receipt by the Company from the Holder) of any notice in accordance with the terms of this Note, unless the Company has in good faith determined that the matters relating to such notice do not constitute material, non-public information relating to the Company or any of its Subsidiaries, the Company shall on or prior to 9:00 am, New York City time on the Business Day immediately following such notice

delivery date, publicly disclose such material, non-public information on a Form 8-K or otherwise. In the event that the Company believes that a notice contains material, non-public information relating to the Company or any of its Subsidiaries, the Company so shall indicate to the Holder explicitly in writing in such notice (or immediately upon receipt of notice from the Holder, as applicable), and in the absence of any such written indication in such notice (or notification from the Company immediately upon receipt of notice from the Holder), the Holder shall be entitled to presume that information contained in the notice does not constitute material, non-public information relating to the Company or any of its Subsidiaries. Nothing contained in this **Section 8(S)** shall limit any obligations of the Company, or any rights of the Holder, under the Securities Purchase Agreement.

- (T) The Company acknowledges and agrees that the Holder is not a fiduciary or agent of the Company, the Holder will not have any obligations hereunder except those obligations expressly set forth herein (and in the Securities Purchase Agreement) and the Holder is acting solely in the capacity of an arm's length contractual counterparty to the Company with respect to the Note and not as a fiduciary or agent of the Company. The Company agrees that it will not assert any claim against the Holder based on an alleged breach of fiduciary duty by the Holder in connection with the Note. The Company acknowledges that the Holder shall have no obligation to (a) maintain the confidentiality of any information provided by the Company or (b) refrain from trading any securities while in possession of such information in the absence of a written non-disclosure agreement signed by an officer of the Holder that explicitly provides for such confidentiality and trading restrictions. In the absence of such an executed, written non-disclosure agreement, the Company acknowledges that the Holder may freely trade in any securities issued by the Company, may possess and use any information provided by the Company in connection with such trading activity, and may disclose any such information to any third party.
- (U) The Company shall cause this Note and any shares of Common Stock issuable pursuant to this Note to be eligible to be offered, sold or otherwise transferred by the Holder pursuant to Rule 144, without any requirements as to volume, manner of sale, availability of current public information (whether or not then satisfied) or notice under the Securities Act and without any requirement for registration under any state securities or "blue sky" law, on and after the date that is six (6) months following the Issue Date. If this Note is to be transferred, the Holder shall notify the Company and surrender this Note to the Company (or provide the Company an affidavit in a form reasonably acceptable to the Company that this Note was lost, stolen or destroyed), whereupon the Company will forthwith issue and deliver upon the order of the Holder a new Note, registered as the Holder may request.
- (V) The Company shall maintain at all times, to be tested as of the last day of each calendar quarter, revenue (calculated in accordance with GAAP) for such quarter, in an amount equal to at least the amount set forth on **Exhibit D**.
- (W) So long as this Note remains outstanding, the Company shall at all times have access to an ATM Program, the terms of which have been approved by the Required Holders in accordance with the Securities Purchase Agreement with aggregate available, accessible and unused capacity to generate gross proceeds to the Company of at least ten million dollars (\$10,000,000). On or before December 31, 2023, the Company shall establish a new ATM program with aggregate available, accessible and unused capacity to generate gross proceeds to the Company of at least seventy five million dollars (\$75,000,000) as of December 31, 2023, and such ATM Program shall not be subsequently canceled, or amended or otherwise modified to reduce such amount, without the written consent of the Required Holders.

- (X) The Company shall pay when due any and all fees and expenses owed by it under all deposit accounts located in the United States and subject to a control agreement entered into in favor of the Collateral Agent.
- (Y) As of the Issue Date (which period may be extended in the reasonable discretion of the Collateral Agent), the Company or relevant Subsidiary of the Company shall have delivered to the Collateral Agent (i) a perfection certificate, in the form attached to the Securities Purchase Agreement as <a href="Exhibit C">Exhibit C</a>, which described in detail reasonably acceptable to the Collateral Agent the Collateral (as defined below) to be delivered (a "Perfection Certificate"), (ii) a U.S. Security Agreement, dated as of the Issue Date, among the grantors named therein and the secured party named therein (the "Main Security Agreement"), (iii) a U.S. Intellectual Property Security Agreement, dated as of the Issue Date, among the grantor(s) named therein and the security party named therein (the "IP Security Agreement") and (iv) UCC financing statements ("UCC Financing Statements"), each of (i) through (iv), in form and substance satisfactory to the Collateral Agent, which created a first lien security interest in all assets of the Company including, but not limited to, its intellectual property (subject to prior Liens and other customary exclusions, in each case acceptable to the Collateral Agent in its sole discretion) (the "Collateral") and perfected a first lien security interest in all such assets of the Company other than the Company's non-U.S. assets and its bank accounts. As soon as reasonably practicable, but in any event before thirty (30) days after the Issue Date of the Initial Purchased Notes (as defined in the Securities Purchase Agreement) (which period may be extended in the reasonable discretion of the Collateral Agent), the Company or relevant Subsidiary of the Company shall deliver to the Collateral Agent (a) such additional security documents, including deposit account control agreements, in the Company shall deliver to the Collateral Agent (the "Ancillary Security Documents"), which perfect a first lien security interest in all remaining assets of the Company (subject to prior Liens and other
- (Z) At no time shall any Intellectual Property (regardless of the value of such Intellectual Property) or any other assets with a fair market value in excess of five million dollars (\$5,000,000) in the aggregate be held by Subsidiaries of the Company formed in jurisdictions outside of the United States, unless the Holder shall have been previously granted a perfected first lien security interest in such assets pursuant to security documents, in form and substance reasonably acceptable to the Collateral Agent. Notwithstanding the foregoing, no assets shall be held by Subsidiaries of the Company formed in jurisdictions outside of the United States in the event that holding such assets in such Subsidiaries would be in violation of the Main Security Agreement, the IP Security Agreement or the Ancillary Security Documents.

#### Section 9. Successors.

The Company will not consolidate with or merge with or into, or (directly, or indirectly through one or more of its Subsidiaries) sell, lease or otherwise transfer, in one transaction or a series of transactions, all or substantially all of the assets of the Company and its Subsidiaries, taken as a whole, to another Person, other than the Holder or any of its Affiliates (a "Business Combination Event"), unless:

(A) the resulting, surviving or transferee Person either (x) is the Company or (y) if not the Company, is a corporation (the "Successor Corporation") duly organized and existing under the laws of the United States of America, any State thereof or the District of Columbia that expressly assumes (by executing and delivering to the Holder and the Trustee, at or before the effective time of such Business Combination Event, a supplement to this instrument) all of the Company's obligations under this Note; and

(B) immediately after giving effect to such Business Combination Event, no Event of Default will have occurred that has not been waived and no Default will have occurred and be continuing which has not been waived.

At the effective time of any Business Combination Event, the Successor Corporation (if not the Company) will succeed to, and may exercise every right and power of, the Company under this Note with the same effect as if such Successor Corporation had been named as the Company in this Note, and, except in the case of a lease, the predecessor Company will be discharged from its obligations under this Note.

This **Section 9** will not apply to a sale, assignment, transfer, lease, conveyance or other disposition of property or assets between or among any of the Company and its Wholly Owned Subsidiaries.

Upon the occurrence of any Business Combination Event, the Company shall deliver the supplemental indenture, officers' certificate and opinion of counsel contemplated pursuant to Section 5.1(b) of the Base Indenture.

#### Section 10. Defaults and Remedies

- (A) Events of Default. "Event of Default" means the occurrence of any of the following (whose occurrence, for the avoidance of doubt, may be waived, but may not be cured):
  - (i) a default in the payment when due of a Partial Redemption Payment (or applicable portion thereof), the Principal Amount, the Maturity Principal Amount, a Cash Sweep Payment or the Fundamental Change Repurchase Price under this Note;
    - (ii) a default for two (2) Business Days in the payment when due of the interest on this Note;
    - (iii) a default in the Company's obligation to issue shares pursuant to this Note;
  - (iv) a default in the Company's obligation to timely deliver a Fundamental Change Notice pursuant to **Section 6(C)**, Cash Sweep Certification in accordance with the requirements of **Section 4(D)** or Compliance Certificate, and such default continues for three (3) Business Days, or the delivery of a materially false or inaccurate Fundamental Change Notice, Cash Sweep Certification, Company Conversion Notice or Compliance Certificate;
  - (v) any failure to timely deliver an Event of Default Notice or a materially false or inaccurate certification as to whether any Event of Default has occurred or as to whether the Equity Conditions have been satisfied;
  - (vi) a default in any of the Company's obligations or agreements under this Note or the Transaction Documents (in each case, other than a default set forth in clauses (i) (v) or (vii) (xviii) of this Section 10(A)), or a breach of any representation or warranty in any material respect (other than representations or warranties subject to material adverse effect or materiality qualifications, which may not be breached in any respect) of any Transaction Document as of the date when made (except for representations and warranties that speak as of a specific date, which shall be true and correct as of such specific date); provided, however, that if such default can be cured, then such default shall

not be an Event of Default unless the Company has failed to cure such default within ten (10) days after its occurrence;

- (vii) any provision of any Transaction Document at any time for any reason (other than pursuant to the express terms thereof) ceases to be valid and binding on or enforceable against the parties thereto, or the validity or enforceability thereof is contested, directly or indirectly, by the Company or any of its Subsidiaries, or a proceeding is commenced by the Company or any of its Subsidiaries or any governmental authority having jurisdiction over any of them, seeking to establish the invalidity or unenforceability thereof;
- (viii) the Company fails to comply with any covenant set forth in Section 8(D), Section 8(E), Section 8(F), Section 8 (G), Section 8 (H), Section 8 (J), Section 8 (Q), Section 8 (V), Section 8 (W) or Section 8 (Y) or Section 8 (Z) of this Note;
- (ix) the suspension from trading or failure of the Common Stock to be trading or listed on the Company's primary Eligible Exchange (measured in terms of trading volume for its Common Stock) on which the Common Stock is traded for a period of three (3) consecutive Trading Days;
- (x) (i) the failure of the Company or any of its Subsidiaries to pay when due or within any applicable grace period any Indebtedness having an individual principal amount in excess of at least two hundred fifty thousand dollars (\$250,000) (or its foreign currency equivalent) in the aggregate of the Company or any of its Subsidiaries, whether such Indebtedness exists as of the Issue Date or is thereafter created, and whether such default has been waived for any period of time or is subsequently cured; or (ii) the occurrence of any breach or default under any terms or provisions of any other Indebtedness of at least two hundred fifty thousand dollars (\$250,000) (or its foreign currency equivalent) in the aggregate of the Company or any of its Subsidiaries, if the effect of such failure or occurrence is to cause or to permit the holder or holders of any such indebtedness, to cause, Indebtedness having an individual principal amount in excess of two hundred fifty thousand dollars (\$250,000) to become or be declared due prior to its stated maturity;
- (xi) one or more final judgments, orders or awards (or any settlement of any litigation or other proceeding that, if breached, could result in a judgment, order or award) for the payment of at least two hundred fifty thousand dollars (\$250,000) (or its foreign currency equivalent) in the aggregate (excluding any amounts covered by insurance pursuant to which the insurer has been notified and has not denied coverage), is rendered against the Company or any of its Subsidiaries and remains unsatisfied and (i) enforcement proceedings shall have been commenced by any creditor upon any such judgment, order, award or settlement or (ii) there shall be a period of ten (10) consecutive Trading Days after entry thereof during which (A) a stay of enforcement thereof is not in effect or (B) the same is not vacated, discharged, stayed or bonded pending appeal;
- (xii) (A) the Company fails to timely file its quarterly reports on Form 10-Q or its annual reports on Form 10-K with the Commission in the manner and within the time periods required by the Exchange Act (reports filed in compliance with the time period specified in Rule 12b-25 promulgated under the Exchange Act shall be considered timely for this purpose), or (B) the Company withdraws or restates any such quarterly report or annual report previously filed with the Commission or (C) the Company at any time prior to the earlier to occur of (i) the issuance of all Notes pursuant to the Securities

Instruction	Agreement as to Form S	and (ii) [], 2024 <sup>4</sup> ceases to satisfy the eligibility requirements set forth under Section I.A of the General i-3;
Collateral reason cea any party	Agent in ac se to be val thereto, or	In violation of <b>Section 8(Y)</b> , any security document shall for any reason fail or cease to create a separate valid and to the extent permitted by the terms hereof or thereof, first priority Lien on the Collateral, in each case, in favor of the cordance with the terms thereof, or any material provision of any such security document shall at any time for any id and binding on or enforceable against the Company or the validity or enforceability thereof shall be contested by a proceeding shall be commenced by the Company or any governmental authority having jurisdiction over the establish the invalidity or unenforceability thereof;
(\$250,000) act of Goo curtailmen reasonably	or more shall or public t of revenue be expected	any material damage to, or loss, theft or destruction of, any material portion of the Collateral (provided that any destruction of the Collateral that reduces the value of such Collateral by two hundred and fifty thousand dollar hall be deemed to be material), whether or not insured, or any strike, lockout, labor dispute, embargo, condemnation enemy, or other casualty which causes, for more than fifteen (15) consecutive days, the cessation or substantial e producing activities at any facility of the Company or any Subsidiary, if any such event or circumstance could be to have a Material Adverse Effect (as defined in the Securities Purchase Agreement). For clarity, an Event of tion 10(A)(xiii) will not require any curtailment of revenue;
	(xv)	at any time the shares of Common Stock issuable pursuant to this Note are not Freely Tradeable.
either:	(xvi)	the Company or any of its Significant Subsidiaries, pursuant to or within the meaning of any Bankruptcy Law
	(1)	commences a voluntary case or proceeding;
	(1) (2)	commences a voluntary case or proceeding; consents to the entry of an order for relief against it in an involuntary case or proceeding;
	` /	
	(2)	consents to the entry of an order for relief against it in an involuntary case or proceeding;
	(2)	consents to the entry of an order for relief against it in an involuntary case or proceeding; consents to the appointment of a custodian of it or for any substantial part of its property;
	(2) (3) (4)	consents to the entry of an order for relief against it in an involuntary case or proceeding; consents to the appointment of a custodian of it or for any substantial part of its property; makes a general assignment for the benefit of its creditors;
	(2) (3) (4) (5)	consents to the entry of an order for relief against it in an involuntary case or proceeding; consents to the appointment of a custodian of it or for any substantial part of its property; makes a general assignment for the benefit of its creditors; takes any comparable action under any foreign Bankruptcy Law; or

- (2) appoints a custodian of the Company or any of its Significant Subsidiaries, or for any substantial part of the property of the Company or any of its Significant Subsidiaries;
  - (3) orders the winding up or liquidation of the Company or any of its Significant Subsidiaries; or
- (4) grants any similar relief with respect to the Company or any of its Significant Subsidiaries under any foreign Bankruptcy Law,

and, in each case under this Section 10(A)(xvi), such order or decree remains unstayed and in effect for at least thirty (30) days.

- (xviii) the Company's stockholders approve any plan for the liquidation or dissolution of the Company
- (B) Acceleration.
- (i) Automatic Acceleration in Certain Circumstances. If an Event of Default set forth in Section 10(A)(xvi) or Section 10(A)(xvii) occurs with respect to the Company (and not solely with respect to a Significant Subsidiary of the Company), then the then outstanding portion of the Maturity Principal Amount of, and all accrued and unpaid interest on, this Note will immediately become due and payable without any further action or notice by any Person.
- (ii) Optional Acceleration. If an Event of Default (other than an Event of Default set forth in Section 10(A)(xvi) or Section 10(A)(xvii) with respect to the Company and not solely with respect to a Subsidiary of the Company) occurs and has not been waived by the Holder, then the Holder, by notice to the Company, may declare this Note (or any portion thereof) to become due and payable on the Business Day immediately following the date of such notice for cash in an amount equal to the Event of Default Acceleration Amount.
- (C) Notice of Events of Default. Promptly, but in no event later than one (1) Business Day after an Event of Default, the Company will provide written notice of such Event of Default to the Holder (an "Event of Default Notice"), which Event of Default Notice shall include (i) a reasonable description of the applicable Event of Default, (ii) the date on which the Event of Default occurred and (iii) the date on which the Default underlying such Event of Default initially occurred, if different than the date on which the Event of Default occurred.
- (D) Default Interest. If a Default or an Event of Default occurs, then in each case, to the extent lawful, interest ("**Default Interest**") will automatically accrue on the Principal Amount outstanding as of the date of such Default or Event of Default at a rate per annum equal to twelve percent (12.0%), from, and including, the date of such Default or Event of Default, as applicable, to, but excluding, the date such Default is cured and all outstanding Default Interest under this Note has been paid. Default Interest hereunder will be computed on the basis of a 360-day year comprised of twelve 30-day months and will be payable in arrears on the earlier of (i) the first day of each calendar month, (ii) the date such Default is cured, (iii) any Fundamental Change Repurchase Date, Conversion Settlement Date or any date that an Event of Default Acceleration Amount is paid by the Company to the Holder, and (iv) the Maturity Date.

### Section 11. Registrar and Paying Agent.

Notwithstanding Section 2.4 of the Base Indenture, the Company hereby appoints itself as the Paying Agent for this Note. In connection with this Note, all references in the Base Indenture to the Trustee as the Paying Agent (including, without limitation, the obligation of the Paying Agent to hold assets in trust under Section 2.5 of the Base Indenture), shall instead be obligations of the Company in its capacity as Paying Agent

#### Section 12. Ranking.

All payments due under this Note shall rank (i) *pari passu* with all Other Notes, (ii) effectively senior to all unsecured obligations of the Company to the extent of the value of the Collateral securing the Notes for so long as the Collateral so secures the Notes in accordance with the terms hereof and (iii) senior to any Subordinated Indebtedness.

#### Section 13. Replacement Notes.

If the Holder of this Note claims that this Note has been mutilated, lost, destroyed or wrongfully taken, then the Company will issue, execute and deliver a replacement Note upon surrender to the Company of such mutilated Note, or upon delivery to the Company of evidence of such loss, destruction or wrongful taking reasonably satisfactory to the Company. In the case of a lost, destroyed or wrongfully taken Note, the Company may require the Holder to provide such security or an indemnity that is reasonably satisfactory to the Company to protect the Company from any loss that it may suffer if this Note is replaced.

#### Section 14. Notices.

Any notice or communication to the Company will be deemed to have been duly given if in writing and delivered in person or by first class mail (registered or certified, return receipt requested), electronic transmission (including e-mail) or other similar means of unsecured electronic communication or overnight air courier guaranteeing next day delivery, or to the other's address, which initially is as follows:

Velo3D, Inc.
511 Division Street
Campbell, CA 95008
Attention: William McCo

Attention: William McCombe

Email address: bill.mccombe@velo3d.com

With a copy (which shall not constitute notice) to:

Fenwick & West LLP 902 Broadway New York, NY 10010 Attention: Per Chilstrom Email address: pchilstrom@fenwick.com

Notice to the Trustee shall be delivered as set forth in Section 10.2 of the Base Indenture.

The Company, by notice to the Holder, may designate additional or different addresses for subsequent notices or communications.

Any notice or communication to the Holder will be by e-mail to its e-mail address, which initially are as set forth in the Securities Purchase Agreement. The Holder, by notice to the

Company, may designate additional or different addresses for subsequent notices or communications.

If a notice or communication is mailed in the manner provided above within the time prescribed, it will be deemed to have been duly given, whether or not the addressee receives it.

#### Section 15. Successors and Assigns.

All agreements of the Company in this Note will bind its successors and will inure to the benefit of the Holder's successors and assigns.

#### Section 16. Severability.

If any provision of this Note is invalid, illegal or unenforceable, then the validity, legality and enforceability of the remaining provisions of this Note will not in any way be affected or impaired thereby.

#### Section 17. Headings, Etc.

The headings of the Sections of this Note have been inserted for convenience of reference only, are not to be considered a part of this Note and will in no way modify or restrict any of the terms or provisions of this Note.

#### Section 18. Amendments

Except as set forth in Section 8.2(a) of the Base Indenture, this Note may not be amended or modified unless in writing by the Company, the Trustee and the Required Holders, and no condition herein (express or implied) may be waived unless waived in writing by each party whom the condition is meant to benefit. Notwithstanding the foregoing, the Company and the Trustee may amend or modify this Note, and may waive any provision hereof, without notice to or consent of the Holder or any Other Holder: (i) to provide for the conversion of this Note into Reference Property under **Section 7(I)**; or (ii) to provide for the assumption by a Successor Corporation of the obligations of the Company under this Note in accordance with **Section 9**. Section 8.1 of the Base Indenture shall not be applicable in connection with this Note.

# Section 19. Governing Law; Waiver of Jury Trial.

All questions concerning the construction, validity, enforcement and interpretation of this Note shall be governed by the internal laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of Delaware. The Company and each Holder hereby irrevocably submits to the exclusive jurisdiction of the Court of Chancery of the State of Delaware, for the adjudication of any dispute hereunder or in connection herewith or under any of the other Transaction Documents or with any transaction contemplated hereby or thereby, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of such court, that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper. Each party hereby irrevocably waives personal service of process and consents to process being served in any such suit, action or proceeding by mailing a copy thereof to such party at the address for such notices to it under this Note and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. Nothing contained herein shall be deemed

or operate to preclude any Holder from bringing suit or taking other legal action against the Company in any other jurisdiction to collect on the Company's obligations to such Holder or to enforce a judgment or other court ruling in favor of such Holder. EACH PARTY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE TO, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR UNDER ANY OTHER TRANSACTION DOCUMENT OR IN CONNECTION WITH OR ARISING OUT OF THIS NOTE OR ANY OTHER TRANSACTION DOCUMENT OR ANY TRANSACTION CONTEMPLATED HEREBY OR THEREBY.

#### Section 20. Submission to Jurisdiction.

The Company (A) agrees that any suit, action or proceeding against it arising out of or relating to this Note may be instituted in the Court of Chancery of the State of Delaware; (B) waives, to the fullest extent permitted by applicable law, (i) any objection that it may now or hereafter have to the laying of venue of any such suit, action or proceeding; and (ii) any claim that it may now or hereafter have that any such suit, action or proceeding in such a court has been brought in an inconvenient forum; and (C) submits to the nonexclusive jurisdiction of such court in any such suit, action or proceeding.

#### Section 21. Enforcement Fees.

The Company agrees to pay all costs and expenses of the Holder incurred as a result of enforcement of this Note and the collection of any amounts owed to the Holder hereunder (whether in cash, Common Stock or otherwise), including, without limitation, reasonable and documented attorneys' fees and expenses.

## Section 22. Collateral Agent.

- (A) Appointment; Authorization. The Holder and the Trustee, together with any successors or assigns thereof, hereby irrevocably appoint, designate and authorize High Trail Investments ON LLC as collateral agent to take such action on their behalf under the provisions of this Note, each Security Agreement and to exercise such powers and perform such duties as are expressly delegated to it by the terms of each Security Agreement, together with such powers as are reasonably incidental thereto. The provisions of this Section 22 are solely for the benefit of the Collateral Agent, and the Company shall not have rights as a third-party beneficiary of any of such provisions. It is understood and agreed that the use of the term "agent" herein or in any Security Agreement (or any other similar term) with reference to the Collateral Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead such term is used as a matter of market custom, and is intended to create or reflect only an administrative relationship between contracting parties. Notwithstanding any provision to the contrary contained elsewhere in this Note, any Security Agreement or any other agreement, instrument or document related hereto or thereto, the Collateral Agent shall not have any duty or responsibility except those expressly set forth herein, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Note, any Security Agreement or any other agreement, instrument or document related hereto or thereto or otherwise exist against the Collateral Agent. The Collateral Agent shall act hereunder and under the Security Documents in accordance with the provisions of this Note and the Security Documents pursuant to the direction of the Required Holders or the Trustee.
- (B) Delegation of Duties. The Collateral Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any Security Agreement by or through any one or more sub-agents appointed by the Collateral Agent. The Collateral Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through

its Affiliates, partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives, or the partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives of any of its Affiliates (collectively, the "Related Parties"). The exculpatory provisions of this Section 22 shall apply to any such sub-agent and to the Related Parties of the Collateral Agent and any such sub-agent. The Collateral Agent shall not be responsible for the negligence or misconduct of any sub-agents except to the extent that a court of competent jurisdiction determines in a final and non-appealable judgment that the Collateral Agent acted with gross negligence or willful misconduct in the selection of such sub-agents.

#### (C) Exculpatory Provisions.

- (i) The Collateral Agent shall not have any duties or obligations except those expressly set forth in the Security Agreements, and its duties shall be administrative in nature. Without limiting the generality of the foregoing, the Collateral Agent: (i) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing or an Event of Default has occurred; (ii) shall not have any duty to take any discretionary action or exercise any discretionary powers; and (iii) shall not, except as expressly set forth in the Security Agreements, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Company or any of its Affiliates that is communicated to or obtained by the Collateral Agent or any of its Affiliates in any capacity.
- (ii) The Collateral Agent shall not be liable for any action taken or not taken by it in the absence of its own gross negligence or willful misconduct as determined by a court of competent jurisdiction by final and non-appealable judgment. The Collateral Agent shall be deemed not to have knowledge of any Default or Event of Default unless and until notice describing such Default or Event of Default is given to the Collateral Agent in writing by the Company.
- (iii) The Collateral Agent shall not be responsible for or have any duty to ascertain or inquire into (a) any statement, warranty or representation made in or in connection with this Notes, any Security Agreement or any other agreement, instrument or document related hereto or thereto, (b) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (c) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default or Event of Default, (d) the validity, enforceability, effectiveness or genuineness of this Note, any Security Agreement or any other agreement, instrument or document related hereto or thereto, or (e) any failure of the Company or any other party to this Note, any Security Agreements or any other agreement, instrument or document related hereto or thereto to perform its obligations hereunder or thereunder. The Collateral Agent shall not be under any obligation to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Note, any Security Agreement or any other agreement, instrument or document related to the hereto or thereto, or to inspect the properties, books or records of the Company or any Affiliate of the Company.
- (D) Reliance by Collateral Agent. The Collateral Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Collateral Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. The Collateral Agent may

consult with legal counsel, independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

- (E) Successor Agent. The Collateral Agent may resign as the Collateral Agent at any time upon ten (10) days' prior notice to the Holder and each Other Holder and the Company. If the Collateral Agent resigns under this Note, the Required Holders shall appoint a successor agent. If no successor agent is appointed prior to the effective date of the resignation of the Collateral Agent, the Collateral Agent may appoint a successor Collateral Agent on behalf of the Holder and each Other Holder after consulting with the Holder and each Other Holder. Upon the acceptance of its appointment as successor agent hereunder, such successor agent shall succeed to all the rights, powers and duties of the retiring Collateral Agent and the term "the Collateral Agent" shall mean such successor agent, and the retiring Collateral Agent's appointment, powers and duties as the Collateral Agent shall be terminated. After the Collateral Agent's resignation hereunder as the Collateral Agent, the provisions of this Section 22 shall continue to inure to its benefit as to any actions taken or omitted to be taken by it while it was the Collateral Agent. If no successor agent has accepted appointment as the Collateral Agent by the date which is thirty (30) days following a retiring Collateral Agent's notice of resignation, a retiring Collateral Agent's resignation shall nevertheless thereupon become effective and the Holder, together with each Other Holder, shall perform all of the duties of the Collateral Agent hereunder until such time as the Required Holders shall appoint a successor agent as provided for above.
- (F) Non-Reliance on the Collateral Agent. The Holder acknowledges that it has, independently and without reliance upon the Collateral Agent or any of its Related Parties and based on such documents and information it has deemed appropriate, made its own credit analysis and decision to enter into this Note. The Holder also acknowledges that it will, independently and without reliance upon the Collateral Agent or any of its Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Note, any Security Agreement or any related agreement or any document furnished hereunder or thereunder.
- (G) Collateral Matters. The Holder irrevocably authorizes the Collateral Agent to release any Lien granted to or held by the Collateral Agent under any Security Agreement (i) when all Obligations have been paid in full; (ii) constituting property sold or to be sold or disposed of as part of or in connection with any sale or other disposition permitted under this Note and each other agreement, instrument or document related hereto (it being agreed and understood that the Collateral Agent may conclusively rely without further inquiry on a certificate of an officer of the Company as to the sale or other disposition of property being made in compliance with this Note and each other agreement, instrument or document related hereto); or (iii) if approved, authorized or ratified in writing by the Holder and each Other Holder. The Collateral Agent shall have the right, in accordance with the Security Agreements to sell, lease or otherwise dispose of any Collateral for cash, credit or any combination thereof, and the Collateral Agent may purchase any Collateral at public or, if permitted by law, private sale and, in lieu of actual payment of the purchase price, may credit bid and setoff the amount of such price against the Obligations.
- (H) Reimbursement by Holder and Other Holders. To the extent that the Company for any reason fails to indefeasibly pay any amount required under Sections 4(g) or 9(k) of the Securities Purchase Agreement to be paid by it to the Collateral Agent (or any sub-agent thereof) or any Related Party of the Collateral Agent (or any sub-agent thereof), the Holder hereby agrees, jointly and severally with each Other Holder, to pay to the Collateral Agent (or any sub-agent) or such Related Party of the Collateral Agent (or any sub-agent thereof), as the case may be, such unpaid amount.

- (I) Marshaling; Payments Set Aside. Neither the Collateral Agent nor the Holder shall be under any obligation to marshal any assets in favor of the Company or any other Person or against or in payment of any or all of the Obligations. To the extent that the Company makes a payment or payments to the Collateral Agent, or the Collateral Agent enforces its Liens or exercises its rights of set-off, and such payment or payments or the proceeds of such enforcement or set-off or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Collateral Agent in its discretion) to be repaid to a trustee, receiver or any other party in connection with any bankruptcy, insolvency or similar proceeding, or otherwise, then (i) to the extent of such recovery, the obligation hereunder or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or set-off had not occurred and (ii) the Holder agrees to pay to the Collateral Agent upon demand its share of the total amount so recovered from or repaid by the Collateral Agent to the extent paid to the Holder.
- (J) Payment Priorities. If the Collateral Agent receives any proceeds of Collateral, the Collateral Agent shall deliver such proceeds to the Trustee for application in accordance with Section 6.10 of the Base Indenture.

#### Section 23. Calculations.

Except as otherwise expressly provided herein, the Collateral Agent shall be responsible for making all calculations called for under this Note or the Indenture. These calculations include, but are not limited to, determinations of the Cash Sweep Amount, Conversion Price, Conversion Rate, Daily VWAPs, Last Reported Sale Price, Fundamental Change Repurchase Price, Event of Default Acceleration Amount, Market Stock Payment Price, Partial Redemption Payments, and accrued interest on the Notes. The Collateral Agent shall make all these calculations in good faith and, absent manifest error, the Collateral Agent's calculations shall be final and binding on the Company and the Trustee. The Collateral Agent shall provide a schedule of its calculations to each of the Trustee, the Company and the Paying Agent (if no longer the Company), and the Trustee is entitled to rely conclusively upon the accuracy of the Collateral Agent's calculations without independent verification. The Collateral Agent shall forward its calculations to the Holder upon the request of that Holder at the sole cost and expense of the Company. For the avoidance of doubt, the Trustee shall have no obligation to calculate or verify the calculation of the accrued and unpaid interest on the Notes. The Trustee shall have no responsibility for calculations or determinations, or verifying the Collateral Agent's calculations or determination, of amounts, determining, or verifying the Collateral Agent's determination of, whether events requiring or permitting conversions of Notes have occurred, determining, or verifying the Collateral Agent's determination of, whether any adjustment is required with respect to conversion rights and, if so, how much, or for the delivery of the shares of Common Stock. The Trustee shall not be responsible for the Company's failure to issue, transfer or deliver any shares of Common Stock or stock certificates or other securities or property or cash upon the surrender of any Note for the purpose of conversion or to comply with any of the Company's duties, responsibilities or covenants hereunder.

The following provisions of the Base Indenture are not applicable to this Note, unless so required pursuant to applicable law: Section 2.15, Article 3, Section 4.3, Section 4.5, Article 5, Section 6.1 through Section 6.4, Section 8.1 and Section 8.2.

### **Section 24.** Trust Indenture Act.

If any provision of this Note contravenes the mandatory provisions of the Trust Indenture Act, the provisions of the Trust Indenture Act shall govern and control. The provisions of

Section 316(a)(1) of the Trust Indenture Act shall be excluded for the purposes of the Note and shall instead be governed by the terms hereof.

### Section 25. Electronic Execution.

The words "execution," "signed," "signature," and words of similar import in the Note shall be deemed to include electronic or digital signatures or the keeping of records in electronic form, each of which shall be of the same effect, validity, and enforceability as manually executed signatures or a paper-based recordkeeping system, as the case may be, to the extent and as provided for under applicable law, including the Electronic Signatures in Global and National Commerce Act of 2000 (15 U.S.C. §§ 7001-7006), the Electronic Signatures and Records Act of 1999 (N.Y. State Tech. §§ 301-309), or any other similar state laws based on the Uniform Electronic Transactions Act. All notices, approvals, consents, requests and any communications hereunder must be in writing (provided that any such communication sent to the Trustee hereunder must be in the form of a document that is signed manually or by way of a digital signature provided by DocuSign (or such other digital signature provider as specified in writing to the Trustee by the authorized representative)), in English. The Company agrees to assume all risks arising out of the use of using digital signatures and electronic methods to submit communications to the Trustee, including without limitation the risk of the Trustee acting on unauthorized instructions, and the risk of interception and misuse by third parties.

\* \* \*

# Exhibit A

# Form of Conversion Notice

Velo3D, Inc.

Senior Secured Convertible Note due 2026

Subject to the terms of this Note, by executing and delivering to convert the following Principal Amount of this Note: \$Shares of Common Stock to be delivered:	his Conversion Notice, the undersigned Holder of this Note directs the Company
Accrued interest amount:	
DTC Participant Number:	
DTC Participant Name:	
Date: (Legal Name of Holder)	
By: Name: Title:	

# Exhibit B

# Form of Company Conversion Notice

Velo3D, Inc.

Senior Secured Convertible Notes due 2026

	Senior Secured Convertible Notes due 2026
	s Company Conversion Notice, the Company hereby notifies the Holder of the Note of the Company's election mount of the Note identified by Certificate No
By delivering this Company Co of the date hereof.	onversion Notice, the Company hereby represents and warrants that the Equity Conditions have been satisfied as
	Velo3D, Inc.
Date: By:	Name: Title:
I	

## Exhibit C

# Form of Covenant Compliance Certificate

The undersigned, the duly qualified and elected Chief Financial Officer of Velo3D, Inc., a Delaware corporation (the "Company"), does hereby certify in such capacity and on behalf of the Company, pursuant to the Senior Secured Convertible Note due 2026, issued [ ], 20[ ] (the "Note"), issued by the Company to [ ], that:

[(i) the Company satisfied the req	uirements of <b>Section 8(G)</b> of the Note during the calendar month ended [ ]; and
(ii) the Company satisfied the req	uirements of <b>Section 8(J)(i)</b> of the Note during the calendar month ended [ ].
(iii) the Company satisfied the rec	quirements of Section 8(W) of the Note during the calendar month ended [ ].
(iv) the Company satisfied the req	uirements of <b>Section 8(Z)</b> of the Note during the calendar month ended [ ].] <sup>5</sup>
[(v) the Company satisfied the rec	quirements of <b>Section 8(V)</b> of the Note during the calendar quarter ended [ ].] <sup>6</sup>
Capitalized terms used herein with	hout definition shall have the meanings given to such terms in the Note.
	VELO3D, INC.
	By: Name:
	Title:
Date:	
<sup>5</sup> Clauses (i-iv) shall only be included in Compl	iance Certificates delivered in connection with month ends.
<sup>6</sup> Clause (v) shall only be included in Complian	ce Certificates delivered on the day the Company files its Annual Report on Form 10-K or a Quarterly Report on Form 10-Q.

Exhibit D

Minimum Revenue Covenant

Calendar Quarter Ending	Minimum Revenue
6/30/2023	\$23,500,000
9/30/2023	\$24,000,000
12/31/2023	\$27,000,000
3/31/2024	\$25,000,000
6/30/2024	\$28,000,000
9/30/2024	\$30,000,000
12/31/2024	\$34,000,000
3/31/2025	\$34,000,000
6/30/2025	\$36,000,000
9/30/2025	\$37,000,000
12/31/2025	\$37,000,000
3/31/2026	\$37,000,000
6/30/2026	\$37,000,000

Certain information in this document indicated with "[\*]" has been omitted from this exhibit because it is both (i) not material and (ii) would be competitively harmful if publicly disclosed.

#### SECURITIES PURCHASE AGREEMENT

This **SECURITIES PURCHASE AGREEMENT** (the "**Agreement**"), dated as of August 10 2023, is by and among Velo3D, Inc., a Delaware corporation with offices located at 511 Division Street, Campbell, CA 95008 (the "**Company**"), and each of the investors listed on the Schedule of Buyers attached hereto (individually, a "**Buyer**" and collectively, the "**Buyers**").

#### RECITALS

- A. The Company has authorized a new series of Senior Secured Convertible Notes in the form attached hereto as **Exhibit A** (the "**Notes**"), that it wishes to issue and sell to the Buyers at the Initial Closing (as defined below) and, at the Buyers' election, any Subsequent Closing (as defined below), upon the terms and conditions stated in this Agreement, which such Notes shall under certain circumstances entitle the Buyers to receive shares of the Company's common stock, par value \$0.00001 per share (together with any capital stock into which such Common Stock shall have been changed or any share capital resulting from a reclassification of such Common Stock, the "**Common Stock**") (such underlying shares of Common Stock issuable pursuant to the terms of the Notes, the "**Underlying Shares**"). The Notes will be issued pursuant to that certain Base Indenture (the "**Base Indenture**"), to be dated as of the Initial Closing Date (as defined below), between U.S. Bank Trust Company, National Association, as trustee (in such capacity, together with its successors and permitted assigns, the "**Trustee**"), and a First Supplemental Indenture (together with the Base Indenture, the "**Indenture**") with respect to the Notes substantially in the forms attached hereto as **Exhibits B-1** and **B-2**, respectively, to be dated as of the Initial Closing Date, by and between the Company, the Trustee and the collateral agent named therein.
- B. Each Buyer wishes to purchase, and the Company wishes to sell, upon the terms and conditions stated in this Agreement, (i) the aggregate principal amount of Initial Purchased Notes (as defined below) set forth opposite such Buyer's name in column (3) on the Schedule of Buyers and (ii) the aggregate principal amount of Subsequently Purchased Notes (as defined below) set forth opposite such Buyer's name in column (5) on the Schedule of Buyers.
- C. At or before the Initial Closing, the parties set forth on the Schedule of Lock-Up Parties attached hereto shall execute and deliver a Lock-Up and Voting Agreement, in the form attached hereto as **Exhibit D** (each, a "**Lock-Up and Voting Agreement**") pursuant to which such parties agree (i) for a period of ninety (90) days from the date hereof to not, without the prior written consent of the Buyers and subject to certain exceptions contained therein, (x) directly or indirectly, offer, pledge, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant to purchase, lend, or otherwise transfer or dispose of any shares of Common Stock or any securities convertible into or exercisable or exchangeable for Common Stock or file any registration statement under the 1933 Act (as defined below) with respect to any of the foregoing or publicly disclose the intention to undertake any of the foregoing or (y) enter into any swap or any other agreement or any transaction that transfers, in whole or in part, directly or indirectly, the economic consequence of ownership of the Common Stock, whether any such swap or transaction described in clause (x) or (y) above is to be settled by delivery of Common Stock or other securities, in cash or otherwise; and (ii) to vote their shares of the Company's Common Stock in favor of providing the Requisite Stockholder Approval (as defined in the Notes).
  - D. The Notes and Underlying Shares are collectively referred to herein as the "Securities."

- E. The Company and each Buyer is executing and delivering this Agreement with respect to the Initial Purchased Notes (as defined below) and the Subsequently Purchased Notes (as defined below) in reliance upon the effective registration statement on Form S-3 (Commission File No. 333-268346) (as amended, the "Registration Statement") filed by the Company with the United States Securities and Exchange Commission (the "SEC") pursuant to the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder (collectively, the "1933 Act"), for the registration of the Initial Purchased Notes and the Subsequently Purchased Notes, as such Registration Statement may be amended and supplemented from time to time (including pursuant to Rule 462(b) of the 1933 Act), including all documents filed as part thereof or incorporated by reference therein, and including all information deemed to be a part thereof at the time of effectiveness pursuant to Rule 430B of the 1933 Act, and the prospectus supplement and accompanying base prospectus (together, the "Prospectus Supplement") complying with Rule 424(b) of the 1933 Act that is delivered by the Company to each Buyer in connection with the execution and delivery of this Agreement, including the documents incorporated by reference therein, and that is filed with the SEC.
  - F. The effective time of this Agreement shall be 3:59 p.m. New York time, on the date of this Agreement...

#### **AGREEMENT**

NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements contained herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Company and each Buyer hereby agree as follows:

## 1. PURCHASE AND SALE OF PURCHASED SECURITIES.

- (a) <u>Purchase of Initial Purchased Notes.</u> Subject to the satisfaction (or waiver) of the conditions set forth in Sections 6 and 7(a), as applicable, the Company shall issue and sell to each Buyer, and each Buyer severally, but not jointly, agrees to purchase from the Company on the Initial Closing Date the following Securities:
  - (i) the aggregate principal amount of Notes as is set forth opposite such Buyer's name in column (3) on the Schedule of Buyers (the "Initial Purchased Notes").
- (b) <u>Initial Closing</u>. The closing (the "**Initial Closing**") of the purchase of the Initial Purchased Notes by the Buyers shall occur by electronic transmission or other transmission as mutually acceptable to the parties. The date and time of the Initial Closing (the "**Initial Closing Date**") shall be 10:00 a.m., New York time, on the second (2nd) Business Day on which the conditions to the Initial Closing set forth in Sections 6 and 7(a) are satisfied or waived (or such other date as is mutually agreed to by the Company and each Buyer). As used herein "**Business Day**" means any day other than a Saturday, a Sunday or any day on which commercial banks in the City of New York are authorized or required by law or executive order to close or be closed; *provided*, *however*, for clarification, commercial banks in the City of New York shall not be deemed to be authorized or required by law or executive order to close or be closed due to "stay at home", "shelter-in-place", "non-essential employee" or any other similar orders or restrictions or the closure of any physical branch locations at the direction of any governmental authority so long as the electronic funds transfer systems (including for wire transfers) of commercial banks in the City of New York are open for use by customers on such day.
- (c) <u>Initial Securities Purchase Price</u>. The aggregate purchase price for the Initial Purchased Notes to be purchased by each Buyer at the Initial Closing (the "**Initial Securities Purchase Price**") shall be the sum of the amounts set forth opposite such Buyer's name in column (4) on the Schedule of Buyers.

- (d) Form of Payment for Initial Purchased Notes. On the Initial Closing Date, (i) each Buyer shall pay its respective Initial Securities Purchase Price to the Company for the Initial Purchased Notes to be issued and sold to such Buyer at the Initial Closing Date (net of expenses payable pursuant to Section 4(g)), by wire transfer of immediately available funds in accordance with a Flow of Funds Letter (as defined below) with respect to the Initial Purchased Notes and (ii) the Company shall:
  - (i) cause the Trustee to deliver to each Buyer the aggregate principal amount of the Initial Purchased Notes as is set forth opposite such Buyer's name in column (3) of the Schedule of Buyers, duly executed on behalf of the Company and the Trustee and registered on the books and records of the Trustee in the name of such Buyer or its designee.
- (e) Purchase of Subsequently Purchased Notes at Buyer's Election. For one or more Subsequent Closings, the Buyers may deliver to the Company a written notice (an "Election Notice") setting forth a principal amount of additional Notes (the "Subsequently Purchased Notes" and, together with the Initial Purchased Notes, the "Purchased Securities"), that the Buyers desire to purchase, and which the Company shall issue and sell, at such Subsequent Closing. Subject to the satisfaction (or waiver) of the conditions set forth in Sections 6 and 7(b), as applicable, and in reliance upon the effective Registration Statement the Company agrees to issue and sell to each Buyer, and each Buyer severally, but not jointly, agrees to purchase from the Company at such Subsequent Closing, the aggregate principal amount of Subsequently Purchased Notes as is set forth on such Election Notice, provided that the maximum aggregate principal amount of Subsequently Purchased Notes issued pursuant to this Agreement to any Buyer shall not exceed the aggregate principal amounts as is set forth opposite such Buyer's name in column (5) on the Schedule of Buyers.
- (f) <u>Subsequent Closings</u>. The closing of the purchase of Subsequently Purchased Notes (each, a "**Subsequent Closing**" and together with the Initial Closing, each a "**Closing**") by the Buyers pursuant to an Election Notice shall occur by electronic transmission or other transmission as mutually acceptable to the parties. The date and time of a Subsequent Closing (each, a "**Subsequent Closing Date**" and together with the Initial Closing Date, each a "**Closing Date**") shall be 10:00 a.m., New York time, on the date upon which the conditions to such Subsequent Closing set forth in Sections 6 and 7(b) are satisfied or waived (or such other date as is set forth by the Buyers in the Election Notice); *provided*, that in no event will any Election Notice be delivered after the date that is one (1) year from the Initial Closing Date.
- (g) <u>Securities Purchase Price</u>. The aggregate purchase price for any issuance of Subsequently Purchased Notes to be purchased by each Buyer at any Subsequent Closing (the "**Subsequent Securities Purchase Price**") shall be one hundred percent (100%) of the aggregate principal amount of Subsequently Purchased Notes set forth on the Election Notice and allocated among the Buyers based on such Buyer's pro rata portion of the aggregate principal amount of Notes purchased hereunder, but in no event will the aggregate purchase price for all issuances of Subsequently Purchased Notes exceed the amount set forth opposite such Buyer's name in column (6) on the Schedule of Buyers.
- (h) Form of Payment for Subsequently Purchased Notes. On each Subsequent Closing Date, (i) each Buyer shall pay its respective Subsequent Securities Purchase Price to the Company for the Subsequently Purchased Notes to be issued and sold to such Buyer at such Subsequent Closing Date (net of expenses payable pursuant to Section 4(g)) pursuant to the applicable Election Notice, by wire transfer of immediately available funds in accordance with the Flow of Funds Letter delivered at the Initial Closing (unless the Company shall have previously supplied the Buyers with updated wire transfer information) with respect to such Subsequently Purchased Notes and (ii) the Company shall cause the Trustee to deliver to each Buyer the aggregate principal amount of Subsequently Purchased Notes pursuant to the

applicable Election Notice, duly executed on behalf of the Company and the Trustee and registered on the books and records of the Trustee in the name of such Buyer or its designee.

### 2. BUYER'S REPRESENTATIONS AND WARRANTIES.

Each Buyer, severally and not jointly, represents and warrants to the Company with respect to only itself that, as of the date hereof and as of the Initial Closing Date and each Subsequent Closing Date, if any:

- (a) <u>Organization; Authority</u>. Such Buyer is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization with the requisite power and authority to conduct its business as currently conducted and enter into and to consummate the transactions contemplated by the Transaction Documents (as defined below) to which it is a party and otherwise to carry out its obligations hereunder and thereunder.
- Information. Such Buyer and its advisors, if any, have been furnished with all materials relating to the business, finances and operations of the Company and materials relating to the offer and sale of the Securities that have been requested by such Buyer. Such Buyer and its advisors, if any, have had (i) the opportunity to review the Registration Statement and the Prospectus Supplement, the Transaction Documents and the SEC Documents (as defined below) and has been afforded the opportunity to ask such questions of the Company as it has deemed necessary of, and to receive answers from, representatives of the Company concerning the terms and conditions of the offering of the Securities and the merits and risks of investing in the Securities; (ii) access to information about the Company and its financial condition, results of operations, business, properties, management and prospects sufficient to enable it to evaluate its investment; and (iii) the opportunity to obtain such additional information that the Company possesses or can acquire without unreasonable effort or expense that is necessary to make an informed investment decision with respect to the investment. Neither such inquiries nor any other due diligence investigations conducted by such Buyer or its advisors, if any, or its representatives shall modify, amend or affect such Buyer's right to rely on the Company's representations and warranties contained herein. Such Buyer understands that its investment in the Securities involves a high degree of risk. Such Buyer acknowledges that it can bear the economic risk and complete loss of its investment in the Securities and has such knowledge and experience in financial or business matters that it is capable of evaluating the merits and risks of the investment contemplated hereby. Such Buyer did not learn of the investment in the Securities as a result of any general solicitation or general advertising. Such Buyer has sought such accounting, legal and tax advice as it has considered necessary to make an informed investment decision with respect to its acquisition of the Securities. Such Buyer is not relying upon, and has not relied upon, any statement, representation or warranty made by any person, except for statements, representations and warranties contained in this Agreement, in making its investment or decision to invest in the Company.
- (c) <u>No Governmental Review</u>. Such Buyer understands that no United States federal or state agency or any other government or governmental agency has passed on or made any recommendation or endorsement of the Securities or the fairness or suitability of the investment in the Securities nor have such authorities passed upon or endorsed the merits of the offering of the Securities.
- (d) <u>Validity; Enforcement</u>. This Agreement and the Security Agreements (as defined below) (only with respect to each Subsequent Closing Date) have been duly and validly authorized, executed and delivered on behalf of such Buyer and shall constitute the legal, valid and binding obligations of such Buyer enforceable against such Buyer in accordance with their respective terms, except as such enforceability may be limited by general principles of equity or to applicable bankruptcy, insolvency, reorganization, moratorium, liquidation and other similar

laws relating to, or affecting generally, the enforcement of applicable creditors' rights and remedies.

(e) No Conflicts. The execution, delivery and performance by such Buyer of this Agreement and the Security Agreements (only with respect to each Subsequent Closing Date) and the consummation by such Buyer of the transactions contemplated hereby and thereby will not (i) result in a violation of the organizational documents of such Buyer, or (ii) conflict with, or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, indenture or instrument to which such Buyer is a party, or (iii) result in a violation of any law, rule, regulation, order, judgment or decree (including federal and state securities laws) applicable to such Buyer, except in the case of clauses (ii) and (iii) above, for such conflicts, defaults, rights or violations which could not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of such Buyer to perform its obligations hereunder.

## 3. REPRESENTATIONS AND WARRANTIES OF THE COMPANY.

The Company represents and warrants to each of the Buyers that, as of the date hereof and as of the Initial Closing Date and each Subsequent Closing Date, if any:

- (a) <u>Compliance with Registration Requirements</u>. The Registration Statement has become effective under the 1933 Act. The Company has complied, to the SEC's satisfaction with all requests of the SEC for additional or supplemental information, if any. No stop order suspending the effectiveness of the Registration Statement is in effect and no proceedings for such purpose have been instituted or, to the knowledge of the Company, are pending or contemplated or threatened by the SEC. At the time each of the Registration Statement and the Company's Annual Report on Form 10-K for the year ended December 31, 2022 (the "2022 Annual Report") was filed with the SEC, the Company met the then-applicable requirements for use of Form S-3 under the 1933 Act. The documents incorporated or deemed to be incorporated by reference in the Registration Statement, at the time they were or hereafter are filed with the SEC, or became effective under the Securities Exchange Act of 1934, as amended (the "1934 Act"), as the case may be, complied and will comply with in all material respects with the requirements of the 1934 Act.
- Statement and any post-effective amendment thereto, at the time it became or becomes effective, complied and will comply in all material respects with the 1933 Act and the Trust Indenture Act of 1939, as amended, and the rules and regulations of the Commission thereunder (collectively, the "Trust Indenture Act"), and did not, and at any Closing Date will not, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading; provided, that the Company makes no representation or warranty with respect to that part of the Registration Statement that constitutes the Statement of Eligibility and Qualification (Form T-1) of the Trustee under the Trust Indenture Act. The Indenture, when filed with the SEC, complied in all material respects with the requirements of the Trust Indenture Act and was duly qualified as an indenture under the Trust Indenture Act. The Prospectus Supplement (including any prospectus wrapper), as of its date, did not, and at any Closing Date, will not, contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. There are no contracts or other documents required to be described in the Prospectus Supplement or to be filed as an exhibit to the Registration Statement which have not been described or filed as required. No event, liability, development or circumstance has occurred or exists, or is reasonably expected to exist or occur with respect to the Company, any of its Subsidiaries (as defined below) or any of their respective businesses, properties, liabilities,

prospects, operations (including results thereof) or condition (financial or otherwise), that (i) would be required to be disclosed by the Company under applicable securities laws on a registration statement on Form S-3 filed with the SEC relating to an issuance and sale by the Company of any shares of Common Stock and which has not been publicly announced, (ii) could have a material adverse effect on any Buyer's investment hereunder or (iii) could have a Material Adverse Effect (as defined below).

- Organization and Qualification. Each of the Company and each of its Subsidiaries are entities duly organized and validly existing and in good standing (if a good standing concept exists in such jurisdiction) under the laws of the jurisdiction in which they are formed, and have the requisite power and authority to own their properties and to carry on their business as now being conducted. Each of the Company and each of its Subsidiaries is duly qualified as a foreign entity to do business and is in good standing (if a good standing concept exists in such jurisdiction) in every jurisdiction in which its ownership of property or the nature of the business conducted by it makes such qualification necessary, except to the extent that the failure to be so qualified or be in good standing would not reasonably be expected to have a Material Adverse Effect. As used in this Agreement, "Material Adverse Effect" means any material adverse effect on (i) the business, properties, assets, liabilities, operations (including results thereof), condition (financial or otherwise) or prospects of the Company or its Subsidiaries, taken as a whole, (ii) the transactions contemplated hereby or in any of the other Transaction Documents or any other agreements or instruments entered into in connection herewith or therewith or (iii) the authority or ability of the Company or any of its Subsidiaries to perform any of their respective obligations under any of the Transaction Documents. Other than the Persons (as defined below) set forth on Schedule 3(c), the Company has no significant Subsidiaries within the meaning of Rule 1-02(w) of Regulation S-X. The Company's Permitted Investments (as defined in the Notes) are also set forth on Schedule 3(c). "Subsidiaries" means any Person in which the Company, directly or indirectly, (I) owns any of the outstanding capital stock or holds any equity or similar interest of such Person or (II) controls or operates all or any part of the business, operations or administration of such Person, and each of the foregoing, is individually referred to herein as a "Subsidiary." For purposes of this Agreement, "Person" means an individual, a limited liability company, a partnership, a joint venture, a corporation, a trust, an unincorporated organization, any other entity and any Governmental Entity (as defined below) or any department or agency thereof.
- (d) Authorization; Enforcement; Validity. The Company has the requisite power and authority to enter into and perform its obligations under this Agreement and the other Transaction Documents and to issue the Securities in accordance with the terms hereof and thereof. Each Subsidiary has the requisite power and authority to enter into and perform its obligations under the Transaction Documents to which it is a party. The execution and delivery of this Agreement and the other Transaction Documents by the Company, and the consummation by the Company and its Subsidiaries of the transactions contemplated hereby and thereby (including, without limitation, the issuance of the Initial Purchased Notes and the Subsequently Purchased Notes, if any, and the reservation for issuance and the issuance of the Underlying Shares as of the Initial Closing and each Subsequent Closing, if any), have been duly authorized by the Company's board of directors (the "Board of Directors") or a duly authorized committee thereof, and (other than (i) such filings, consents or authorizations as have been obtained, taken, given or made, (ii) any filings as may be required by any state securities agencies, (iii) filings necessary to perfect the Liens (as defined below) granted under the Security Agreements, and (iv) a Supplemental Listing Application with NYSE (as defined below) and (v) filings as may be required in connection with obtaining the Requisite Stockholder Approval (clauses (i) through (v), collectively, the "Required Filings"), no further filing, consent or authorization is required by the Company, its Subsidiaries, their respective boards of directors or their stockholders (other than the Requisite Stockholder Approval) or other governing body in connection therewith. This Agreement has been, and the other Transaction Documents to which it is a party will be duly executed and delivered by the Company prior to the Initial Closing, and each constitutes a legal,

valid and binding obligation of the Company, enforceable against the Company in accordance with its respective terms, except as such enforceability may be limited by general principles of equity or applicable bankruptcy, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally, the enforcement of applicable creditors' rights and remedies and except as rights to indemnification and to contribution may be limited by federal or state securities law (the "Enforceability Exceptions"). "Transaction Documents" means, collectively, this Agreement, the Notes, the Perfection Certificate (as defined below), the Security Agreements, the Indenture, each Lock-Up and Voting Agreement, and the Irrevocable Transfer Agent Instructions (as defined below) and each of the other agreements and instruments entered into or delivered by any of the parties hereto in connection with the transactions contemplated hereby and thereby, as may be amended from time to time; *provided*, that solely for representations and warranties made as of the Initial Closing Date pursuant to Section 2 and Section 3 hereof, Transaction Documents shall not include the Ancillary Security Documents.

- (e) <u>Issuance of Securities</u>. The issuance of the Securities is duly authorized and when issued and delivered in accordance with the terms of the Transaction Documents, the Securities shall be validly issued, fully paid and non-assessable and free from all preemptive or similar rights, mortgages, defects, claims, liens, pledges, charges, taxes, rights of first refusal, encumbrances, security interests and other encumbrances (collectively "**Liens**") with respect to the issuance thereof. As of the Initial Closing Date and each Subsequent Closing Date, if any, the Company shall have not less than a number of authorized but unissued shares of Common Stock equal to the sum of (i) fifty million (50,000,000) shares of Common Stock, which shall not be exclusively reserved for issuance pursuant to the Notes and (ii) one hundred percent (100%) of a fraction the numerator of which shall be the then outstanding Principal Amount (as defined in the Notes) of all Notes issued pursuant to this Agreement, and the denominator of which shall be the Conversion Price (as defined in the Notes), which shall be reserved for issuance pursuant to the Notes (which such reservation shall be for the sole benefit of and exclusive availability for the Buyers). The Underlying Shares (upon issuance in accordance with the Notes), will be validly issued, fully paid and non-assessable and free from all preemptive or similar rights or Liens with respect to the issuance thereof, with the holders being entitled to all rights accorded to a holder of Common Stock.
- (f) No Conflicts. The execution, delivery and performance of the Transaction Documents by the Company and the consummation by the Company of the transactions contemplated hereby and thereby (including, without limitation, the issuance of the Notes and the Underlying Shares, and the reservation of Common Stock for issuance of the Underlying Shares) will not (i) result in a violation of the Certificate of Incorporation (as defined below), Bylaws (as defined below), certificate of formation, memorandum of association, articles of association, bylaws or other organizational documents of the Company or any of its Subsidiaries, or any capital stock or other securities of the Company or any of its Subsidiaries, (ii) conflict with, or constitute a default (or an event which with notice or lapse of time or both would become a default) in any respect under, or give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, indenture or instrument to which the Company or any of its Subsidiaries is a party, or (iii) assuming the accuracy of the representations and warranties in Section 2, result in a violation of any law, rule, regulation, order, judgment or decree (including, without limitation, foreign, federal and state securities laws and regulations, and the rules and regulations of The New York Stock Exchange ("NYSE") and including all applicable foreign, federal and state laws, rules and regulations) applicable to the Company or any of its Subsidiaries or by which any property or asset of the Company or any of its Subsidiaries is bound or affected, assuming, with respect to clauses (ii) and (iii) above, the making of the Required Filings and except in the case of clauses (ii) and (iii) above, for such breaches, violations or conflicts as would not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

- Consents. Neither the Company nor any Subsidiary is required to obtain any consent from, authorization or order of, or make any filing or registration with (other than the Required Filings, filings necessary to perfect the Liens granted under the Security Agreements and such consents, authorizations, filings or registrations the absence of which would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect), any Governmental Entity or any regulatory or self-regulatory agency or any other Person in order for it to execute, deliver or perform any of its respective obligations under or contemplated by the Transaction Documents, in each case, in accordance with the terms hereof or thereof. Other than the Required Filings, all consents, authorizations, orders, filings and registrations which the Company or any Subsidiary is required to obtain pursuant to the preceding sentence have been or will be obtained or effected on or prior to the Initial Closing Date, and neither the Company nor any of its Subsidiaries are aware of any facts or circumstances which might prevent the Company or any of its Subsidiaries from obtaining or effecting any of the registration, application or filings contemplated by the Transaction Documents. The Company is not in violation of the requirements of NYSE and has no knowledge of any facts or circumstances which could reasonably lead to delisting or suspension of the Common Stock. "Governmental Entity" means any nation, state, county, city, town, village, district, or other political jurisdiction of any nature, federal, state, local, municipal, foreign, or other government, governmental or quasi-governmental authority of any nature (including any governmental agency, branch, department, official, or entity and any court or other tribunal), multi-national organization or body; or body exercising, or entitled to exercise, any administrative, executive, judicial, legislative, police, regulatory, or taxing authority or power of any nature or instrumentality of any of the foregoing, including any entity or enterprise owned or controlled by a government or a public international organization or any of the foregoing.
- (h) Acknowledgment Regarding Buyer's Purchase of Securities. The Company acknowledges and agrees that each Buyer is acting solely in the capacity of an arm's length purchaser with respect to the Transaction Documents and the transactions contemplated hereby and thereby and that no Buyer is (i) an officer or director of the Company or any of its Subsidiaries, (ii) an "affiliate" (as defined in Rule 144 promulgated under the 1933 Act (or a successor rule thereto), collectively, "Rule 144") of the Company or any of its Subsidiaries or (iii) to its knowledge, a "beneficial owner" (as defined for purposes of Rule 13d-3 of the 1934 Act) of more than 4.99% of the shares of any voting class of the Company's Common Stock. The Company further acknowledges that no Buyer is acting as a financial advisor or fiduciary of the Company or any of its Subsidiaries (or in any similar capacity) with respect to the Transaction Documents and the transactions contemplated hereby and thereby and any advice given by a Buyer or any of its representatives or agents in connection with the Transaction Documents and the transactions contemplated hereby and thereby is merely incidental to such Buyer's purchase of the Securities. The Company further represents to each Buyer that the Company's and each Subsidiary's decision to enter into the Transaction Documents to which it is a party has been based solely on the independent evaluation by the Company, each Subsidiary and their respective representatives.
- (i) No Placement Agent. The Company shall be responsible for the payment of any placement agent's fees, financial advisory fees, or brokers' commissions (other than for Persons engaged by any Buyer or its investment advisor) relating to or arising out of the transactions contemplated hereby. Neither the Company nor any of its Subsidiaries has engaged any placement agent or other agent, other than Credit Suisse Securities (USA) LLC (the "Placement Agent"), in connection with the offer or sale of the Securities. The Company shall pay, and hold each Buyer harmless against, any liability, loss or expense (including, without limitation, attorney's fees and reasonable and documented out-of-pocket expenses) arising in connection with any claim for the payment of any placement agent's fees, financial advisory fees, or brokers' commissions (other than for Persons engaged by any Buyer) relating to or arising out of the transactions contemplated hereby.

- (j) No Integrated Offering. None of the Company, its Subsidiaries or any of their affiliates, nor any Person acting on their behalf has, directly or indirectly, made any offers or sales of any security or solicited any offers to buy any security, under circumstances that would require approval of stockholders of the Company in connection with the offering of the Securities for purposes of the 1933 Act or under any applicable stockholder approval provisions, including, without limitation, under the rules and regulations of any exchange or automated quotation system on which any of the securities of the Company are listed or designated for quotation. None of the Company, its Subsidiaries, their affiliates nor any Person acting on their behalf has taken or will take any action or steps that would cause the offering of any of the Securities to be integrated with other offerings of securities of the Company.
- (k) <u>Dilutive Effect</u>. The Company understands and acknowledges that the number of Underlying Shares will increase in certain circumstances. The Company further acknowledges that its obligation to issue the Underlying Shares pursuant to the terms of the Notes in accordance with the terms thereof and this Agreement is absolute and unconditional regardless of the dilutive effect that such issuance may have on the ownership interests of other stockholders of the Company.
- (l) Application of Takeover Protections. The Company and its Board of Directors or a duly authorized committee thereof have taken or will take prior to the Initial Closing Date all necessary action, if any, in order to render inapplicable any control share acquisition, interested stockholder, business combination, poison pill, stockholder rights plan or other similar anti-takeover provision under the Certificate of Incorporation, Bylaws or other organizational documents or the laws of the jurisdiction of its incorporation which is or could become applicable to any Buyer as a result of the transactions contemplated by this Agreement, including, without limitation, the Company's issuance of the Securities and any Buyer's ownership of the Securities.
- Financial Statements. During the one (1) year prior to the date hereof and each Closing Date with respect to which this representation is being made, the Company has timely filed all reports, schedules, forms, proxy statements, statements and other documents required to be filed by it with the SEC (other than Section 16 ownership filings) pursuant to the reporting requirements of the 1934 Act (reports filed in compliance with the time period specified in Rule 12b-25 promulgated under the 1934 Act shall be considered timely for this purpose) (all of the foregoing filed prior to the date hereof and all exhibits and appendices included therein and financial statements, notes and schedules thereto and documents incorporated by reference therein being hereinafter referred to as the "SEC Documents"). The Company has delivered or has made available to the Buyers or their respective representatives true, correct and complete copies of each of the SEC Documents not available on the EDGAR system. As of their respective dates, the SEC Documents complied in all material respects with the requirements of the 1934 Act and the rules and regulations of the SEC promulgated thereunder applicable to the SEC Documents, and none of the SEC Documents, at the time they were filed with the SEC, contained any untrue statement of a material fact or omitted to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. As of their respective dates, the financial statements of the Company included in the SEC Documents complied in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto as in effect as of the time of filing. Such financial statements have been prepared in accordance with generally accepted accounting principles ("GAAP"), consistently applied, during the periods involved (except (i) as may be otherwise indicated in such financial statements or the notes thereto, or (ii) in the case of unaudited interim statements, to the extent they may exclude footnotes or may be condensed or summary statements) and fairly present in all material respects the financial position of the Company as of the dates thereof and the results of its operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal year-end audit adjustments which will not be material, either individually or in the aggregate).

No other information provided by or on behalf of the Company to any of the Buyers which is not included in the SEC Documents (including, without limitation, information referred to in the disclosure schedules to this Agreement) contains any untrue statement of a material fact or omits to state any material fact necessary in order to make the statements therein not misleading, in the light of the circumstance under which they are or were made. The Company is not currently contemplating to amend or restate any of the financial statements (including, without limitation, any notes or any letter of the independent accountants of the Company with respect thereto) included in the SEC Documents (the "Financial Statements"), nor is the Company currently aware of facts or circumstances which would require the Company to amend or restate any of the Financial Statements, in each case, in order for any of the Financials Statements to be in material compliance with GAAP and the rules and regulations of the SEC. The Company has not been informed by its independent auditors that they recommend that the Company amend or restate any of the Financial Statements or that there is any need for the Company to amend or restate any of the Financial Statements.

- (n) Absence of Certain Changes. Since the date of the Company's audited financial statements contained in the 2022 Annual Report, there has been no Material Adverse Effect. Since the date of the Company's audited financial statements contained in the 2022 Annual Report, except as set forth on Schedule 3(n), neither the Company nor any of its Subsidiaries has (i) declared or paid any dividends, (ii) sold any assets, individually or in the aggregate, outside of the ordinary course of business, (iii) made any capital expenditures, individually or in the aggregate, outside of the ordinary course of business or (iv) made any revaluation of any of their respective assets, including, without limitation, writing down the value of capitalized inventory or writing off notes or accounts receivable or any sale of assets other than in the ordinary course of business.
- (o) Insolvency. Neither the Company nor any of its Subsidiaries has taken any steps to seek protection pursuant to any law or statute relating to bankruptcy, insolvency, reorganization, receivership, liquidation or winding up, nor does the Company or any Subsidiary have any knowledge or reason to believe that any of their respective creditors intend to initiate involuntary bankruptcy proceedings or any actual knowledge of any fact which would reasonably lead a creditor to do so. The Company and its Subsidiaries, individually and on a consolidated basis, are not as of the date hereof and as of the Initial Closing Date and each Subsequent Closing Date, if any, and after giving effect to the transactions contemplated hereby to occur on the Initial Closing Date and each Subsequent Closing Date, if any, will not be Insolvent (as defined below). For purposes of this Section 3(0), "Insolvent" means, (i) with respect to the Company and its Subsidiaries, on a consolidated basis, (A) the present fair saleable value of the Company's and its Subsidiaries' assets is less than the amount required to pay the Company's and its Subsidiaries' total Indebtedness (as defined below), (B) the Company and its Subsidiaries are unable to pay their debts and liabilities, subordinated, contingent or otherwise, as such debts and liabilities become absolute and matured or (C) the Company and its Subsidiaries' intend to incur or believe that they will incur debts that would be beyond their ability to pay as such debts mature; and (ii) with respect to the Company and each Subsidiary, individually, (A) the present fair saleable value of the Company or such Subsidiary's (as the case may be) assets is less than the amount required to pay its respective total Indebtedness, (B) the Company or such Subsidiary (as the case may be) is unable to pay its respective debts and liabilities, subordinated, contingent or otherwise, as such debts and liabilities become absolute and matured or (C) the Company or such Subsidiary (as the case may be) intends to
- (p) <u>Listing and Trading. Regulatory Permits</u>. Since September 30, 2021, (i) the Common Stock has been listed or designated for quotation on NYSE, (ii) trading in the Common Stock has not been suspended by the SEC or NYSE and (iii) the Company has received no communication, written or oral, from the SEC or Nasdaq regarding the suspension or delisting of

the Common Stock from NYSE. The Company and each of its Subsidiaries possess all certificates, authorizations and permits issued by the appropriate regulatory authorities necessary to conduct their respective businesses, except where the failure to possess such certificates, authorizations or permits would not reasonably be likely to have, individually or in the aggregate, a Material Adverse Effect, and neither the Company nor any such Subsidiary has received any notice of proceedings relating to the revocation or modification of any such certificate, authorization or permit.

- (q) Foreign Corrupt Practices. Neither the Company, any of the Company's Subsidiaries, nor any director or officer thereof, nor, to the Company's knowledge, any employee or agent or any other person acting for or on behalf of the foregoing (individually and collectively, a "Company Affiliate") have violated the U.S. Foreign Corrupt Practices Act or any other applicable anti-bribery or anti-corruption laws (individually and collectively, "Anti-Corruption Laws"), nor, to the Company's knowledge, has any Company Affiliate offered, paid, promised to pay, or authorized the payment of any money, or offered, given, promised to give, or authorized the giving of anything of value, to any officer, employee or any other person acting in an official capacity for any Governmental Entity to any political party or official thereof or to any candidate for political office (individually and collectively, a "Government Official") or to any person under circumstances where such Company Affiliate knew or was aware of a high probability that all or a portion of such money or thing of value would be offered, given or promised, directly or indirectly, to any Government Official, for the purpose of:
- (i) (A) influencing any act or decision of such Government Official in his/her official capacity, (B) inducing such Government Official to do or omit to do any act in violation of his/her lawful duty, (C) securing any improper advantage, or (D) inducing such Government Official to influence or affect any act or decision of any Governmental Entity, or
- (ii) assisting the Company or its Subsidiaries in obtaining or retaining business for or with, or directing business to, the Company or its Subsidiaries.

Neither of the Company nor any of its Subsidiaries will use, directly or indirectly, any part of the proceeds of the offering in any manner that would constitute a violation of Anti-Corruption Laws.

- (r) <u>Sarbanes-Oxley Act</u>. The Company and each of its Subsidiaries is in material compliance with any and all applicable requirements of the Sarbanes-Oxley Act of 2002, as amended, and any and all applicable rules and regulations promulgated by the SEC thereunder.
- (s) Transactions With Affiliates. Except as set forth on Schedule 3(s), no current or former employee, partner, director, officer or shareholder (direct or indirect) of the Company or its Subsidiaries, or any associate, or, to the knowledge of the Company, any affiliate of any thereof, or any relative with a relationship no more remote than first cousin of any of the foregoing, is presently or has been (i) a party to any transaction with the Company or its Subsidiaries (including any contract, agreement or other arrangement providing for the furnishing of services by, or rental of real or personal property from, or otherwise requiring payments to, any such director, officer or shareholder or such associate or affiliate or relative Subsidiaries (other than for ordinary course services as employees, officers or directors of the Company or any of its Subsidiaries)) or (ii) the direct or indirect owner of an interest in any corporation, firm, association or business organization which is a competitor, supplier or customer of the Company or its Subsidiaries (except for a passive investment (direct or indirect) in less than 5% of the common stock or ordinary shares, as applicable, of a company whose securities are traded on or quoted through an Eligible Market (as defined below)), nor does any such Person receive income from any source other than the Company or its Subsidiaries which relates to the business of the Company or its Subsidiaries or should properly accrue to the Company or its Subsidiaries. No employee, officer, shareholder or director of the Company or

any of its Subsidiaries or member of his or her immediate family is indebted to the Company or its Subsidiaries, as the case may be, nor is the Company or any of its Subsidiaries indebted (or committed to make loans or extend or guarantee credit) to any of them, other than (i) for payment of salary for services rendered, (ii) reimbursement for reasonable expenses incurred on behalf of the Company or its Subsidiaries, as the case may be, and (iii) for other standard employee benefits made generally available to all employees or executives (including share option agreements outstanding under any share option plan approved by the Board of Directors of the Company).

## (t) <u>Equity Capitalization</u>.

- (i) <u>Authorized and Outstanding Capital Stock</u>. As of the date of this Agreement and as of the Initial Closing, the authorized capital stock of the Company consists of (A) 500,000,000 shares of Common Stock, of which, 192,747,463 are issued and outstanding and 41,660,092 shares are reserved for issuance pursuant to Convertible Securities (as defined below) (other than the Notes) exercisable or exchangeable for, or convertible into, shares of Common Stock, and (B) 10,000,000 shares of preferred stock, par value \$0.00001 per share, of which no shares are issued and outstanding.
- (ii) Valid Issuance; Available Shares; Affiliates. All of the Company's outstanding shares are duly authorized and have been validly issued and are fully paid and non-assessable. Schedule 3(t)(ii) sets forth the number of shares of Common Stock that are (A) reserved for issuance pursuant to Convertible Securities (other than the Initial Purchased Notes and the Subsequently Purchased Notes, if any) as of the date hereof and as of the Initial Closing, owned by Persons who are "affiliates" (as defined in Rule 405 of the 1933 Act and calculated based on the assumption that only officers, directors and holders of at least 10% of the Company's issued and outstanding shares of Common Stock are "affiliates" without conceding that any such Persons are "affiliates" for purposes of federal securities laws) of the Company or any of its Subsidiaries. To the Company's knowledge, as of the date hereof and the Initial Closing Date, no Person owns 10% or more of the Company's issued and outstanding shares of Common Stock (calculated based on the assumption that all Convertible Securities, whether or not presently exercisable or convertible, have been fully exercised or converted (as the case may be) taking account of any limitations on exercise or conversion (including "blockers") contained therein without conceding that such identified Person is a 10% stockholder for purposes of federal securities laws). "Convertible Securities" means any capital stock or other security of the Company or any of its Subsidiaries that is at any time and under any circumstances directly or indirectly convertible into, exercisable or exchangeable for, or which otherwise entitles the holder thereof to acquire, any capital stock or other security of the Company (including, without limitation, Common Stock and any RSUs or any rights, warrants or options to subscribe for or purchase shares of Common Stock or Convertible Securities (collectively, "Options")) or any of its Subsidiaries.
- (iii) Existing Securities; Obligations. Except as set forth on Schedule 3(t)(iii): (A) none of the Company's or any Subsidiary's shares, interests or capital stock is subject to preemptive rights or any other similar rights or Liens suffered or permitted by the Company or any Subsidiary; (B) other than (i) pursuant to the Business Combination Agreement, (ii) the Company's outstanding warrants (the "Warrants"), (iii) stock options, restricted stock units ("RSUs"), performance stock units, deferred stock units, rights and other stock-based awards awarded to employees, directors and consultants of the Company under equity incentive and employee stock purchase plans adopted by the Board of Directors of the Company and described in the SEC Documents and (iv) pursuant to this Agreement and following the Initial Closing, the Notes, there are no outstanding options, warrants, scrip, rights to subscribe to, calls or commitments of any character whatsoever relating to, or securities or rights convertible into, or exercisable or exchangeable for, any shares, interests or capital stock of the Company or any of its Subsidiaries, or contracts, commitments, understandings or arrangements by which the

Company or any of its Subsidiaries is or may become bound to issue additional shares, interests or capital stock of the Company or any of its Subsidiaries or options, warrants, scrip, rights to subscribe to, calls or commitments of any character whatsoever relating to, or securities or rights convertible into, or exercisable or exchangeable for, any shares, interests or capital stock of the Company or any of its Subsidiaries; (C) other than pursuant to the Amended and Restated Registration Rights Agreement, dated September 29, 2021, by and among the Company and holders party thereto and the Warrants, there are no agreements or arrangements under which the Company or any of its Subsidiaries is obligated to register the sale of any of their securities under the 1933 Act; (D) other than the Warrants and, following the Initial Closing, the Notes, there are no outstanding securities or instruments of the Company or any of its Subsidiaries which contain any redemption or similar provisions, and there are no contracts, commitments, understandings or arrangements by which the Company or any of its Subsidiaries is or may become bound to redeem a security of the Company or any of its Subsidiaries; (E) there are no securities or instruments containing anti-dilution or similar provisions that will be triggered by the issuance of the Securities; and (F) neither the Company nor any Subsidiary has any stock appreciation rights or "phantom stock" plans or agreements or any similar plan or agreement.

- (u) <u>Organizational Documents</u>. The Company has furnished to the Buyers true, correct and complete copies of the Company's Certificate of Incorporation, as amended, and as in effect on the date hereof and each Closing Date (the "Certificate of Incorporation") and the Company's Amended and Restated Bylaws, each in effect on the date hereof and each Closing Date (collectively, the "Bylaws"), and the terms of all Convertible Securities and the material rights of the holders thereof in respect thereto.
- <u>Indebtedness and Other Contracts</u>. Except as set forth on <u>Schedule 3(v)</u>, neither the Company nor any of its Subsidiaries (i) has any outstanding debt securities, notes, credit agreements, credit facilities or other agreements, documents or instruments evidencing Indebtedness of the Company or any of its Subsidiaries or by which the Company or any of its Subsidiaries is or may become bound, (ii) has any financing statements securing obligations in any amounts filed against the Company or any of its Subsidiaries or with respect to any of their respective assets; (iii) is in violation of any term of, or in default under, any contract, agreement or instrument relating to any Indebtedness, except where such violations and defaults would not result, individually or in the aggregate, in a Material Adverse Effect, or (iv) is a party to any contract, agreement or instrument relating to any Indebtedness, the performance of which, in the judgment of the Company's officers, has or is expected to have a Material Adverse Effect. Neither the Company nor any of its Subsidiaries have any liabilities or obligations required to be disclosed in the SEC Documents which are not so disclosed in the SEC Documents, other than those incurred in the ordinary course of the Company's or its Subsidiaries' respective businesses consistent with past practices and which, individually or in the aggregate, do not or could not have a Material Adverse Effect. For purposes of this Agreement: (x) "Indebtedness" of any Person means, without duplication, (A) all indebtedness for borrowed money, (B) all obligations issued, undertaken or assumed as the deferred purchase price of property or services (including, without limitation, "capital leases" in accordance with GAAP) (other than trade payables entered into in the ordinary course of business consistent with past practice), (C) all reimbursement or payment obligations with respect to letters of credit, surety bonds and other similar instruments, (D) all obligations evidenced by notes, bonds, debentures or similar instruments, including obligations so evidenced incurred in connection with the acquisition of property, assets or businesses, (E) all indebtedness created or arising under any conditional sale or other title retention agreement, or incurred as financing, in either case with respect to any property or assets acquired with the proceeds of such indebtedness (even though the rights and remedies of the seller or bank under such agreement in the event of default are limited to repossession or sale of such property), (F) all monetary obligations under any leasing or similar arrangement which, in connection with GAAP, consistently applied for the periods covered thereby, is classified as a capital lease, (G) all indebtedness referred to in clauses (A) through (F) above secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be

secured by) any Lien upon or in any property or assets (including accounts and contract rights) owned by any Person, even though the Person which owns such assets or property has not assumed or become liable for the payment of such indebtedness, and (H) all Contingent Obligations in respect of indebtedness or obligations of others of the kinds referred to in clauses (A) through (G) above; and (y) "Contingent Obligation" means, as to any Person, any direct or indirect liability, contingent or otherwise, of that Person with respect to any Indebtedness, lease, dividend or other obligation of another Person if the primary purpose or intent of the Person incurring such liability, or the primary effect thereof, is to provide assurance to the obligee of such liability that such liability will be paid or discharged, or that any agreements relating thereto will be complied with, or that the holders of such liability will be protected (in whole or in part) against loss with respect thereto.

- (w) <u>Litigation</u>. There is no material action, suit, arbitration, proceeding, inquiry or investigation before or by NYSE, any court, public board, other Governmental Entity, self-regulatory organization or body pending or, to the knowledge of the Company, threatened against or affecting the Company or any of its Subsidiaries (or pending or threatened by the Company or any of its Subsidiaries), the Common Stock or any of the Company's or its Subsidiaries' officers or directors, whether of a civil or criminal nature or otherwise, in their capacities as such, except as set forth in <u>Schedule 3(w)</u>. To the knowledge of the Company, no director, officer or employee of the Company or any of its Subsidiaries has willfully violated 18 U.S.C. §1519 or engaged in spoliation in reasonable anticipation of litigation. Without limitation of the foregoing, there has not been, and to the knowledge of the Company, there is not pending, contemplated or anticipated, any inquiry or investigation by the SEC involving the Company, any of its Subsidiaries or any current or former director or officer of the Company or any of its Subsidiaries. The SEC has not issued any stop order or other order suspending the effectiveness of any registration statement filed by the Company under the 1934 Act or the 1934 Act. After reasonable inquiry of its officers (as defined in Rule 16a-1(f) promulgated under the 1934 Act) and members of its Board of Directors, the Company is not aware of any fact which might result in or form the basis for any such action, suit, arbitration, investigation, inquiry or other proceeding. Neither the Company nor any of its Subsidiaries is subject to any order, writ, judgment, injunction, decree, determination or award of any Governmental Entity.
- (x) <u>Insurance</u>. The Company and each of its Subsidiaries are insured by insurers of recognized financial responsibility against such losses and risks and in such amounts as management of the Company believes to be prudent and customary in the businesses in which the Company and its Subsidiaries are engaged. Neither the Company nor any of its Subsidiaries has been refused any insurance coverage sought or applied for, and neither the Company nor any of its Subsidiaries has any reason to believe that it will be unable to renew its existing insurance coverage as and when such coverage expires or to obtain similar coverage from similar insurers as may be necessary to continue its business at a cost that would not have a Material Adverse Effect.
- (y) Employee Relations. Neither the Company nor any of its Subsidiaries is a party to any collective bargaining agreement or employs any member of a union. The Company and its Subsidiaries believe that their relations with their employees are good. No executive officer (as defined in Rule 501(f) promulgated under the 1933 Act) or other key employee of the Company or any of its Subsidiaries has notified the Company or any such Subsidiary that such officer intends to leave the Company or any such Subsidiary or otherwise terminate such officer's employment with the Company or any such Subsidiary. To the knowledge of the Company, no executive officer or other key employee of the Company or any of its Subsidiaries is, or is now expected to be, in violation of any material term of any employment contract, confidentiality, disclosure or proprietary information agreement, non-competition agreement, or any other contract or agreement or any restrictive covenant, and the continued employment of each such executive officer or other key employee (as the case may be) does not subject the Company or any of its Subsidiaries to any liability with respect to any of the foregoing matters. The Company

and its Subsidiaries are in material compliance with all applicable federal, state, local and foreign laws and regulations respecting labor, employment and employment practices and benefits, terms and conditions of employment and wages and hours, except where failure to be in compliance would not, either individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.

- (z) <u>Title</u>. Each of the Company and its Subsidiaries holds good title to, or a valid leasehold interest in, all real property, leases in real property, facilities or other interests in real property owned or held by the Company or any of its Subsidiaries that is material to the business of the Company (the "**Real Property**"). Except as set forth on <u>Schedule 3(z)</u>, the Real Property is free and clear of all Liens and is not subject to any rights of way, building use restrictions, exceptions, variances, reservations, or limitations of any nature except for (a) Liens for current taxes not yet due and (b) zoning laws and other land use restrictions that do not impair the present or anticipated use of the property subject thereto. Any Real Property held under lease by the Company or any of its Subsidiaries are held by them under valid, subsisting and, subject to the Enforceability Exceptions, enforceable leases with such exceptions as are not material and do not interfere in any material respect with the use made and proposed to be made of such property and buildings by the Company or any of its Subsidiaries.
- (aa) <u>Fixtures and Equipment</u>. Each of the Company and its Subsidiaries (as applicable) has good title to, or a valid leasehold interest in, the tangible personal property, equipment, improvements, fixtures, and other personal property and appurtenances that are used by the Company and its Subsidiaries to conduct their respective businesses (the "**Fixtures and Equipment**"). The Fixtures and Equipment are structurally sound, are in good operating condition and repair (ordinary wear and tear excepted), are adequate for the uses to which they are being put, are not in need of maintenance or repairs except for ordinary, routine maintenance and repairs and are sufficient for the conduct of the Company's and/or its Subsidiaries' businesses (as applicable) in the manner as conducted prior to the date hereof and each Closing Date. Except as set forth on <u>Schedule 3(aa)</u>, each of the Company and its Subsidiaries owns all of its Fixtures and Equipment free and clear of all Liens except for (i) Liens for current taxes not yet due, (ii) zoning laws and other land use restrictions that do not impair the present or anticipated use of the property subject thereto and (iii) other Permitted Liens (as defined in the Notes).
- (ab) Intellectual Property Rights. The Company and each of its Subsidiaries owns or possesses adequate rights or licenses to use all material trademarks, trade names, service marks, service mark registrations, service names, patents, patent applications, other patent rights, copyrights, original works, inventions, licenses, approvals, governmental authorizations, trade secrets and other intellectual property rights and all applications and registrations therefor ("Intellectual Property Rights") necessary to conduct their respective businesses as now conducted and as presently proposed to be conducted. Except as set forth on Schedule 3(bb), none of the Company's or its Subsidiaries' Intellectual Property Rights, which are necessary to conduct their respective businesses, have expired, terminated or been abandoned, or are expected to expire, terminate or be abandoned, within three years from the date of this Agreement. To the knowledge of the Company, and except as set forth on Schedule 3(bb), neither the Company nor any of its Subsidiaries has, (i) infringed, misappropriated, diluted or violated the Intellectual Property Rights of others, (ii) violated any material term or provision of any contract concerning Intellectual Property Rights, (iii) violated any material right of any person (including any right to privacy or publicity), or (iv) conducted its business in a manner that would constitute unfair competition or unfair trade practices under the laws of any jurisdiction. There is no claim, action or proceeding being made or brought, or to the knowledge of the Company or any of its Subsidiaries, being threatened, against the Company or any of its Subsidiaries regarding Intellectual Property Rights of others that would reasonably be expected to have a Material Adverse Effect on the Company. The Company is not aware of any facts or circumstances which might give rise to any of the foregoing infringements or claims, actions or proceedings. The

Company and each of its Subsidiaries have taken reasonable security measures to protect the secrecy, confidentiality and value of all trade secrets within the Intellectual Property Rights of the Company that are materially necessary to conduct their respective businesses. To the knowledge of the Company, no third party is infringing, violating or misappropriating any Company owned Intellectual Property Rights, and there is no claim pending or proceeding regarding any such actual or alleged infringement, misappropriation or other violation of any Company owned Intellectual Property Rights. All former and current employees, contractors and consultants of the Company who have contributed to the creation or development of the Company owned Intellectual Property Rights have executed a valid and enforceable agreement containing an irrevocable assignment to the Company of all of their ownership and other rights therein, including to any invention, improvement or discovery. The Company has not distributed, incorporated or otherwise used any "Open Source Code" (also known as "free software" (as defined by the Free Software Foundation) or "open source software" (as defined by the Open Source Initiative) or has not otherwise distributed publicly software under terms that permit modification and redistribution of such software) in a manner that would require that any of the proprietary software owned by the Company or included in a Company product or service: (i) be made available or distributed in source code form; (ii) be licensed for the purpose of making derivative works; (iii) be licensed under terms that allow reverse engineering, reverse assembly or disassembly of any kind; or (iv) be redistributable at no charge. The Company is in compliance with the terms and conditions of all licenses for free or Open Source Code.

(ac) Environmental Laws. (i) The Company and its Subsidiaries (A) are in compliance with any and all Environmental Laws (as defined below), (B) have received all permits, licenses or other approvals required of them under applicable Environmental Laws to conduct their respective businesses and (C) are in compliance with all terms and conditions of any such permit, license or approval where, except in each of the foregoing clauses (A), (B) and (C), where the failure to so comply or having such permits, licenses or other approval would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. The term "Environmental Laws" means all federal, state, local or foreign laws or regulations relating to pollution or protection of human health or the environment (including, without limitation, ambient air, surface water, groundwater, land surface or subsurface strata), including, without limitation, laws relating to emissions, discharges, releases or threatened releases of chemicals, pollutants, contaminants, or toxic or hazardous materials, substances or wastes (collectively, "Hazardous Materials") into the environment, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of, or exposure to, Hazardous Materials, as well as all authorizations, codes, decrees, demands or demand letters, injunctions, judgments, licenses, notices or notice letters, orders, permits, plans or regulations issued, entered, promulgated or approved thereunder.

## (ad) No Hazardous Materials:

- (A) to the Company's knowledge, have been disposed of or otherwise released from any Real Property of the Company or any of its Subsidiaries in violation of any Environmental Laws; or
- (B) to the Company's knowledge, are present on, over, beneath, in or upon any Real Property or any portion thereof in quantities that would constitute a violation of any Environmental Laws or in quantities, a manner or location that would reasonably be expected to require remedial action pursuant to any Environmental Laws. No prior use by the Company or any of its Subsidiaries of any Real Property has occurred that violates any Environmental Laws, which violation would have a Material Adverse Effect.
- (i) To the Company's knowledge, neither the Company nor any of its Subsidiaries knows of any other Person that has stored, treated, recycled, disposed of or

otherwise located on any Real Property any Hazardous Materials, including, without limitation, such substances as asbestos and polychlorinated biphenyls.

- (ii) To the knowledge of the Company, none of the Real Property is on any federal or state "Superfund" list or Comprehensive Environmental Response, Compensation and Liability Information System ("CERCLIS") list or any state environmental agency list of sites under consideration for CERCLIS, nor subject to any environmental related Liens.
- (iii) Neither the Company nor its Subsidiaries is subject to any pending or, to the Company and its Subsidiaries' knowledge, threatened claim or proceeding to any Environmental Laws, except for any claims or proceeding that would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (ae) Tax Status. The Company and each of its Subsidiaries (i) has timely made or filed all foreign, federal and state income and all other tax returns, reports and declarations required by any jurisdiction to which it is subject through the date of this Agreement or have requested extensions thereof (except where the failure to file would not, individually or in the aggregate, have a Material Adverse Effect) and (ii) has timely paid all taxes and other governmental assessments and charges, shown or determined to be due on such returns, reports and declarations, except those being contested in good faith and for which reserves required by GAAP have been created in the financial statements of the Company or for cases in which the failure to pay would not have a Material Adverse Effect. There is no tax deficiency that has been determined adversely to the Company or any of its Subsidiaries which has had a Material Adverse Effect, nor does the Company or its Subsidiaries have any knowledge or notice of any tax deficiency which could reasonably be expected to be determined adversely to the Company or its Subsidiaries and which could reasonably be expected to have a Material Adverse Effect.
- <u>Internal Accounting and Disclosure Controls.</u> Except as set forth on <u>Schedule 3(ff)</u>, the Company and each of its Subsidiaries maintains internal control over financial reporting (as such term is defined in Rule 13a-15(f) under the 1934 Act) that is effective to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP, including that (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain asset and liability accountability, (iii) access to assets or incurrence of liabilities is permitted only in accordance with management's general or specific authorization and (iv) the recorded accountability for assets and liabilities is compared with the existing assets and liabilities at reasonable intervals and appropriate action is taken with respect to any difference. Except as set forth on Schedule 3(ff), the Company maintains disclosure controls and procedures (as such term is defined in Rule 13a-15(e) under the 1934 Act) that are effective in ensuring that information required to be disclosed by the Company in the reports that it files or submits under the 1934 Act is recorded, processed, summarized and reported, within the time periods specified in the rules and forms of the SEC, including, without limitation, controls and procedures designed to ensure that information required to be disclosed by the Company in the reports that it files or submits under the 1934 Act is accumulated and communicated to the Company's management, including its principal executive officer or officers and its principal financial officer or officers, as appropriate, to allow timely decisions regarding required disclosure. Except as set forth on Schedule 3(ff), since the filing of the Company's 2022 Annual Report, neither the Company nor any of its Subsidiaries has received any notice or correspondence from any accountant, Governmental Entity or other Person relating to any potential material weakness or significant deficiency in any part of the internal controls over financial reporting of the Company or any of its Subsidiaries.
- (ff) Off Balance Sheet Arrangements. There is no transaction, arrangement, or other relationship between the Company or any of its Subsidiaries and an unconsolidated or other off

balance sheet entity that is required to be disclosed by the Company in its 1934 Act filings and is not so disclosed or that otherwise could be reasonably likely to have a Material Adverse Effect.

- (ag) <u>Investment Company Status</u>. The Company is not, and upon consummation of the sale of the Securities and the application of the proceeds thereof, will not be, an "investment company," or a company controlled by an "investment company" as such term is defined in the Investment Company Act of 1940, as amended.
- (ah) Acknowledgement Regarding Buyers' Trading Activity. It is understood and acknowledged by the Company that (i) prior to August 11, 2023 and following August 17, 2023, none of the Buyers have been asked by the Company or any of its Subsidiaries to agree, nor has any Buyer agreed with the Company or any of its Subsidiaries, to desist from effecting any transactions in or with respect to (including, without limitation, purchasing or selling, long and/or short) any securities of the Company, or "derivative" securities based on securities issued by the Company or to hold any of the Securities for any specified term; (ii) any Buyer, and counterparties in "derivative" transactions to which any such Buyer is a party, directly or indirectly, presently may have a "short" position in the Common Stock which was established prior to such Buyer's knowledge of the transactions contemplated by the Transaction Documents; (iii) each Buyer shall not be deemed to have any affiliation with or control over any arm's length counterparty in any "derivative" transaction; and (iv) each Buyer may rely on the Company's obligation to timely deliver shares of Common Stock as and when required pursuant to the Transaction Documents for purposes of effecting trading in the Common Stock of the Company. The Company further understands and acknowledges that prior to August 11, 2023 and following August 17, 2023, one or more Buyers may have engaged and may after August 17, 2023 engage in Hedging Arrangements (as defined below) (including, without limitation, the location and/or reservation of borrowable shares of Common Stock) at various times prior to or during the period that the Securities are outstanding, including, without limitation, during the periods that the value and/or number of the Underlying Shares deliverable with respect to the Securities are being determined, and such Hedging Arrangements (including, without limitation, the location and/or reservation of borrowable shares of Common Stock), if any, can reduce the value of the existing stockholders' equity interest in the Company both at and after the time the Hedging Arrangements are being conducted. The Company acknowledges that such aforementioned Hedging Arrangements do not constitute a breach of this Agreement, the Notes or any other Transaction Document or any of the documents executed in connection herewith or therewith. Notwithstanding the foregoing or any other provision of this Agreement to the contrary, each Buyer hereby agrees not to engage in any such Hedging Arrangements beginning on, and including, August 11, 2023 and ending on, and including, August 17, 2023. "Hedging Arrangements" means any forward, futures, swap, collar, put, call, floor, cap, option or other contract or trading activity that are intended to benefit from, or reduce or eliminate the risk of, fluctuations in price, but shall not include the location and/or reservation of borrowable shares or the conversion of the Notes or trading in Underlying Shares received upon any such conversion.
- (ai) Manipulation of Price. Neither the Company nor any of its Subsidiaries has, and, to the knowledge of the Company, no Person acting on their behalf has, directly or indirectly, (i) taken any action designed to cause or to result in the stabilization or manipulation of the price of any security of the Company or any of its Subsidiaries to facilitate the sale or resale of any of the Securities, (ii) other than the fees to be paid to the Placement Agent, sold, bid for, purchased, or paid any compensation for soliciting purchases of, any of the Securities, (iii) other than pursuant to an ATM Sales Agreement, paid or agreed to pay to any Person any compensation for soliciting another to purchase any other securities of the Company or any of its Subsidiaries or (iv) paid or agreed to pay any Person for research services with respect to any securities of the Company or any of its Subsidiaries.
- (aj) <u>U.S. Real Property Holding Corporation</u>. Neither the Company nor any of its Subsidiaries is, or has ever been, and so long as any of the Securities are held by any of the

Buyers, shall become, a U.S. real property holding corporation within the meaning of Section 897 of the Internal Revenue Code of 1986, as amended (the "Code"), and the Company and each Subsidiary shall so certify upon any Buyer's request.

- (ak) <u>Transfer Taxes</u>. All stock transfer or other taxes (other than income or similar taxes) which are required to be paid in connection with the issuance, sale and transfer of the Securities to be sold to each Buyer hereunder will be, or will have been, fully paid or provided for by the Company, and all laws imposing such taxes will be or will have been complied with; *provided*, that the Company shall not be required to pay any tax that may be payable in respect of any transfer involved in the issuance and delivery of any Underlying Shares pursuant to the Notes in a name other than that of the Buyer of such Notes, and the Company shall not be required to issue or deliver such Underlying Shares unless or until the Person or Persons requesting the issuance thereof shall have paid to the Company the amount of such tax or shall have established to the satisfaction of the Company that such tax has been paid.
- (al) <u>Bank Holding Company Act</u>. Neither the Company nor any of its Subsidiaries is subject to the Bank Holding Company Act of 1956, as amended (the "**BHCA**") and to regulation by the Board of Governors of the Federal Reserve System (the "**Federal Reserve**"). Neither the Company nor any of its Subsidiaries owns or controls, directly or indirectly, five percent (5%) or more of the outstanding shares of any class of voting securities or twenty-five percent (25%) or more of the total equity of a bank or any entity that is subject to the BHCA and to regulation by the Federal Reserve. Neither the Company nor any of its Subsidiaries exercises a controlling influence over the management or policies of a bank or any entity that is subject to the BHCA and to regulation by the Federal Reserve.
- (am) Illegal or Unauthorized Payments; Political Contributions. Neither the Company nor any of its Subsidiaries nor, to the best of the Company's knowledge (after reasonable inquiry of its officers and directors), any of the officers, directors, employees, agents or other representatives of the Company or any of its Subsidiaries or affiliates, has, directly or indirectly, made or authorized any payment, contribution or gift of money, property, or services, whether or not in contravention of applicable law, (i) as a kickback or bribe to any Person or (ii) to any political organization, or the holder of or any aspirant to any elective or appointive public office to influence official action or secure an improper advantage, except for personal political contributions not involving the direct or indirect use of funds of the Company or any of its Subsidiaries.
- (an) Money Laundering. The operations of the Company and its Subsidiaries are and have been conducted at all times in material compliance with the USA Patriot Act of 2001 and all other applicable U.S. and non-U.S. anti-money laundering laws and regulations, including, without limitation, the laws, regulations and Executive Orders and sanctions programs administered by the U.S. Office of Foreign Assets Control, including, but not limited, to (i) Executive Order 13224 of September 23, 2001 entitled, "Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism" (66 Fed. Reg. 49079 (2001)); and (ii) any regulations contained in 31 CFR, Subtitle B, Chapter V. The operations of the Company and its Subsidiaries are and have been conducted at all times in material compliance with the USA Patriot Act of 2001 and all other applicable U.S. and non-U.S. anti-money laundering laws and regulations.
- (ao) <u>Sanctions</u>. None of the Company, any of its Subsidiaries or any director or officer or, to the knowledge of the Company and its Subsidiaries, employee, agent or other person acting for or on behalf of the foregoing is the subject or target of any economic or financial sanctions imposed, administered or enforced by the United States (including the U.S. Department of the Treasury Office of Foreign Assets Control and the U.S. Department of State) or other relevant sanctions authority (collectively, "Sanctions" and each such Person, a "Sanctioned Person"). The operations of the Company and its Subsidiaries are, and have been conducted within the past

five (5) years, in compliance with applicable Sanctions. Neither the Company nor any of its Subsidiaries will, directly or indirectly, use any part of the proceeds of this offering, or lend, contribute or otherwise make available such proceeds to any subsidiary, joint venture partner or other Person, to fund or facilitate any dealings or transactions with, involving or for the benefit of any Sanctioned Person, or otherwise in any manner that would constitute or give rise to a violation of any Sanctions by any Person (including any Person participating in the offering, whether as buyer, underwriter, advisor, investor or otherwise).

- (pp) <u>Management</u>. During the past five year period, no current or former officer or director of the Company, to the knowledge of the Company, has been the subject of:
- (i) a petition under bankruptcy laws or any other insolvency or moratorium law or the appointment by a court of a receiver, fiscal agent or similar officer for such Person, or any partnership in which such person was a general partner at or within two years before the filing of such petition or such appointment, or any corporation or business association of which such person was an executive officer at or within two years before the time of the filing of such petition or such appointment;
- (ii) a conviction in a criminal proceeding or a named subject of a pending criminal proceeding (excluding traffic violations that do not relate to driving while intoxicated or driving under the influence);
- (iii) any order, judgment or decree, not subsequently reversed, suspended or vacated, of any court of competent jurisdiction, permanently or temporarily enjoining any such person from, or otherwise limiting, the following activities:
- (1) Acting as a futures commission merchant, introducing broker, commodity trading advisor, commodity pool operator, floor broker, leverage transaction merchant, any other person regulated by the United States Commodity Futures Trading Commission or an associated person of any of the foregoing, or as an investment adviser, underwriter, broker or dealer in securities, or as an affiliated person, director or employee of any investment company, bank, savings and loan association or insurance company, or engaging in or continuing any conduct or practice in connection with such activity;
  - (2) Engaging in any particular type of business practice; or
- (3) Engaging in any activity in connection with the purchase or sale of any security or commodity or in connection with any violation of securities laws or commodities laws;
- (iv) any order, judgment or decree, not subsequently reversed, suspended or vacated, of any authority barring, suspending or otherwise limiting for more than sixty (60) days the right of any such person to engage in any activity described in the preceding sub paragraph, or to be associated with persons engaged in any such activity;
- (v) a finding by a court of competent jurisdiction in a civil action or by the SEC or other authority to have violated any securities law, regulation or decree and the judgment in such civil action or finding by the SEC or any other authority has not been subsequently reversed, suspended or vacated; or
- (vi) a finding by a court of competent jurisdiction in a civil action or by the Commodity Futures Trading Commission to have violated any federal commodities law, and the judgment in such civil action or finding has not been subsequently reversed, suspended or vacated.

- (ap) Stock Option Plans. Each stock option granted by the Company was granted (i) in accordance with the terms of the applicable equity incentive plan of the Company and (ii) with an exercise price at least equal to the fair market value of the Common Stock on the date such stock option would be considered granted under GAAP and applicable law. To the Company's knowledge, no stock option granted under the Company's stock option plan has been backdated. The Company has not knowingly granted, and there is no and has been no policy or practice of the Company to knowingly grant, stock options prior to, or otherwise knowingly coordinate the grant of stock options with, the release or other public announcement of material information regarding the Company or its Subsidiaries or their financial results or prospects.
- Cybersecurity. The information technology assets and equipment, computers, systems, networks, hardware, software, websites, applications, and databases used or owned by, or leased or licensed to, the Company or any of its Subsidiaries (collectively, "IT Systems") are adequate for, and operate and perform in all material respects as required in connection with the operation of the business of the Company and its Subsidiaries as currently conducted free and clear of all material bugs, errors, defects, Trojan horses, time bombs, malware and other corruptants. The Company and its Subsidiaries have implemented and maintained commercially reasonable physical, technical and administrative controls, policies, procedures, and safeguards to maintain and protect their material confidential information and the integrity, continuous operation, redundancy and security of all IT Systems and data, including "Personal Data," used in connection with their businesses. "Personal Data" means (i) a natural person's name, street address, telephone number, e-mail address, photograph, social security number or tax identification number, driver's license number, passport number, credit card number, bank information, or customer or account number; (ii) any information which would qualify as "personally identifying information" under the Federal Trade Commission Act, as amended; (iii) "personal data" as defined by the European Union General Data Protection Regulation ("GDPR") (EU 2016/679); (iv) any information which would qualify as "protected health information" under the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act (collectively, "HIPAA"); and (v) any other piece of information that allows the identification of such natural person, or his or her family, or permits the collection or analysis of any data related to an identified person's health or sexual orientation. There have been no breaches, violations, outages or unauthorized uses of or accesses to same, except for those that have been remedied without material cost or liability or the duty to notify any other person, nor any incidents under internal review or investigations relating to the same. The Company and its Subsidiaries are presently in material compliance with all applicable laws or statutes and all judgments, orders, rules and regulations of any court or arbitrator or governmental or regulatory authority, internal policies and contractual obligations relating to the privacy and security of IT Systems and Personal Data and to the protection of such IT Systems and Personal Data from unauthorized use, access, misappropriation or modification.
- (ar) Compliance with Data Privacy Laws. The Company and its Subsidiaries are, and at all prior times were, in material compliance with all applicable state and federal data privacy and security laws and regulations, including without limitation HIPAA, and the Company and its Subsidiaries have taken commercially reasonable actions to prepare to comply with, and since May 25, 2018, have been and currently are in compliance with, the GDPR (EU 2016/679) (collectively, the "Privacy Laws"). To ensure compliance with the Privacy Laws, the Company and its Subsidiaries have in place, comply with, and take appropriate steps reasonably designed to ensure compliance in all material respects with their policies and procedures relating to data privacy and security and the collection, storage, use, disclosure, handling, and analysis of Personal Data (the "Policies"). The Company and its Subsidiaries have at all times made all disclosures to users or customers required by applicable laws and regulatory rules or requirements, and none of such disclosures made or contained in any Policy have, to the knowledge of the Company, been inaccurate or in violation of any applicable laws and regulatory rules or requirements in any material respect. Neither the Company nor any Subsidiary: (i) has

received notice of any actual or potential liability under or relating to, or actual or potential violation of, any of the Privacy Laws, and has no knowledge of any event or condition that would reasonably be expected to result in any such notice; (ii) is currently conducting or paying for, in whole or in part, any investigation, remediation, or other corrective action pursuant to any Privacy Law; or (iii) is a party to any order, decree, or agreement that imposes any obligation or liability under any Privacy Law.

- (as) <u>Margin Stock</u>. The application of the proceeds received by the Company from the issuance, sale and delivery of the Notes as described in the Transaction Documents will not violate Regulation T, U or X of the Board of Governors of the Federal Reserve system or any other regulation of such Board of Governors.
- (at) <u>Disclosure</u>. The Company confirms that neither it nor any other Person acting on its behalf has provided any of the Buyers or their agents or counsel with any information that constitutes or could reasonably be expected to constitute material, non-public information concerning the Company or any of its Subsidiaries, other than the existence of the transactions contemplated by this Agreement and the other Transaction Documents. The Company understands and confirms that each of the Buyers has relied on and will rely on the foregoing representations in effecting transactions in securities of the Company. The Company acknowledges and agrees that no Buyer makes or has made any representations or warranties with respect to the transactions contemplated hereby other than those specifically set forth in Section 2.
- (au) <u>ATM Sales Program</u>. The Company has access to an ATM Sales Program (as defined below), the terms of which have been approved by the Required Holders, and, as of the date hereof and the Closing Date with respect to which this representation is being made, the ATM Sales Program has aggregate available, accessible and unused capacity to generate gross proceeds to the Company of at least twenty million dollars (\$20,000,000).
- (av) <u>No Additional Agreements</u>. The Company does not have any agreement or understanding with any Buyer with respect to the transactions contemplated by the Transaction Documents other than as specified in the Transaction Documents.

#### 4. COVENANTS.

- (a) <u>Best Efforts</u>. Each Buyer shall use its best efforts to timely satisfy each of the covenants hereunder and conditions to be satisfied by it as provided in Section 6 of this Agreement. The Company shall use its best efforts to timely satisfy each of the covenants hereunder and conditions to be satisfied by it as provided in Section 7 of this Agreement.
- (b) <u>Blue Sky</u>. The Company shall, on or before the Initial Closing Date, take such action as the Company shall reasonably determine is necessary in order to obtain an exemption for, or to, qualify the Securities for sale to the Buyers at the Initial Closing and each Subsequent Closing, if any, pursuant to this Agreement under applicable securities or "Blue Sky" laws of the states of the United States (or to obtain an exemption from such qualification), and shall provide evidence of any such action so taken to the Buyers on or prior to the Initial Closing Date. Without limiting any other obligation of the Company under this Agreement, the Company shall timely make all filings and reports relating to the offer and sale of the Securities required under all applicable securities laws (including, without limitation, all applicable federal securities laws and all applicable "Blue Sky" laws), and the Company shall comply with all applicable foreign, federal, state and local laws, statutes, rules, regulations and the like relating to the offering and sale of the Securities to the Buyers. Notwithstanding the foregoing, the Company will not be obligated to qualify as a foreign corporation or as a dealer in securities in any jurisdiction where it would not otherwise be required to so qualify, file any general consent to service of process in

any such jurisdiction or subject itself to taxation in any such jurisdiction if it is not otherwise so subject.

- (c) Reporting Status. Until the earlier of (i) the date upon which the Buyers shall have sold all of the Securities issuable pursuant to this Agreement and (ii) the date upon which no Notes remain outstanding and no further Notes may be issued pursuant to this Agreement (the "Reporting Period"), the Company shall timely file all reports required to be filed with the SEC pursuant to the 1934 Act (reports filed in compliance with the time period specified in Rule 12b-25 promulgated under the 1934 Act shall be considered timely for this purpose), and the Company shall not terminate its status as an issuer required to file reports under the 1934 Act even if the 1934 Act or the rules and regulations thereunder would no longer require or otherwise permit such termination.
- (d) <u>Use of Proceeds</u>. The Company will use the net proceeds from the sale of the Securities for working capital, general corporate purposes and the repayment of Indebtedness outstanding under the SVB Loan Agreement (as defined below), but not, directly or indirectly for (i) the redemption or repurchase of any securities or repayment of any Indebtedness of the Company or any of its Subsidiaries (other than Indebtedness incurred under the Transaction Documents) or (ii) the settlement of any outstanding litigation. "SVB Loan Agreement" means that that certain Third Amended and Restated Loan and Security Agreement dated as of May 14, 2021, between the Company and Silicon Valley Bank, a division of First-Citizens Bank & Trust Company (successor by purchase to the Federal Deposit Insurance Corporation as Receiver for Silicon Valley Bridge Bank, N.A. (as successor to Silicon Valley Bank), as amended by a certain First Loan Modification Agreement dated as of May 13, 2022, as further amended by a certain Second Loan Modification Agreement dated as of June 13, 2022, as further amended by a certain Third Loan Modification Agreement dated as of July 11, 2022, as further amended and affected by a certain Joinder and Fourth Loan Modification Agreement dated as of July 25, 2022, and as affected and amended by that certain Letter Agreement dated as of April 7, 2023, and as further amended by a certain Fifth Loan Modification Agreement dated as of May 5, 2023.
- (e) <u>Financial Information</u>. The Company agrees to send the following to each Buyer during the Reporting Period (i) unless the following are filed with the SEC through EDGAR and are available to the public through the EDGAR system, within one (1) Business Day after the filing thereof with the SEC, a copy of its Annual Reports on Form 10-K and Quarterly Reports on Form 10-Q, any interim reports or any consolidated balance sheets, income statements, stockholders' equity statements and/or cash flow statements for any period other than annual, any Current Reports on Form 8-K and any registration statements (other than on Form S-8 or Form S-4) or amendments filed pursuant to the 1933 Act, (ii) unless the following are either filed with the SEC through EDGAR or are otherwise widely disseminated via a recognized news release service (such as PR Newswire), on the same day as the release thereof, e-mail copies of all press releases issued by the Company or any of its Subsidiaries and (iii) unless the following are filed with the SEC through EDGAR, copies of any notices and other information made available or given to the stockholders of the Company generally, contemporaneously with the making available or giving thereof to the stockholders.
- (f) <u>Listing</u>. The Company shall promptly secure the listing or designation for quotation (as the case may be) of all of the Underlying Shares upon each national securities exchange and automated quotation system, if any, upon which the Common Stock is then listed or designated for quotation (as the case may be) (subject to official notice of issuance) and shall maintain such listing or designation for quotation (as the case may be) of all Underlying Shares from time to time issuable under the terms of the Transaction Documents on such national securities exchange or automated quotation system. The Company shall maintain the Common Stock's listing or authorization for quotation (as the case may be) on the NYSE, the NYSE American, the Nasdaq Capital Market, the Nasdaq Global Market or the Nasdaq Global Select Market (each, an "Eligible Market"). Neither the Company nor any of its Subsidiaries shall

take any action which could be reasonably expected to result in the delisting or suspension of the Common Stock on an Eligible Market. The Company shall pay all fees and expenses in connection with satisfying its obligations under this Section 4(f).

- (g) Fees. The Company shall pay for the reasonable and documented due diligence and legal fees and expenses incurred by the Buyers in connection with the structuring, documentation, negotiation, and closing of the transactions contemplated by the Transaction Documents (and the enforcement thereof by the Buyers), including, without limitation, all reasonable and documented consultant fees, all reasonable and documented legal fees and disbursements of Latham & Watkins LLP, counsel to the Buyers, and due diligence and regulatory filings in connection therewith (the "Transaction Expenses") and such Transaction Expenses, to the extent they have not already been paid to the Buyer, may be withheld by the Buyers from its Initial Securities Purchase Price and Subsequent Securities Purchase Price at the Initial Closing and each Subsequent Closing, respectively; *provided*, *however*, that the Transaction Expenses (whether paid directly or withheld) through the Initial Closing shall not exceed an aggregate amount of two hundred fifty thousand dollars (\$250,000) excluding costs related to any intercreditor agreements with the Company's existing lender as of the date hereof. The Company shall be responsible for the payment of any placement agent's fees, financial advisory fees, transfer agent fees, The Depository Trust Company ("DTC") fees or broker's commissions (other than for Persons engaged by any Buyer) relating to or arising out of the transactions contemplated hereby. The Company shall pay, and hold each Buyer harmless against, any liability, loss or expense (including, without limitation, reasonable attorneys' fees and reasonable and documented out-of-pocket expenses) arising in connection with any claim relating to any such payment. Except as otherwise set forth in the Transaction Documents, each party to this Agreement shall bear its own expenses in connection with the sale of the Securities to the Buyers.
- (h) Pledge of Securities. Notwithstanding anything to the contrary contained in this Agreement, the Company acknowledges and agrees that the Securities may be pledged by a Buyer in connection with a bona fide margin agreement or other loan or financing arrangement that is secured by the Securities. The pledge of Securities shall not be deemed to be a transfer, sale or assignment of the Securities hereunder, and no Buyer effecting a pledge of Securities shall be required to provide the Company with any notice thereof or otherwise make any delivery to the Company pursuant to this Agreement or any other Transaction Document. The Company hereby agrees to execute and deliver such documentation as a pledgee of the Securities may reasonably request in connection with a pledge of the Securities to such pledgee by a Buyer.

#### (i) <u>Disclosure of Transactions and Other Material Information.</u>

(i) <u>Disclosure of Transaction.</u> Promptly after 4:00 p.m., New York time, but no later than 5:00 p.m., New York time, on the date of this Agreement, the Company shall issue a press release (a "**Press Release**") reasonably acceptable to the Buyers disclosing all the material terms of the transactions contemplated by the Transaction Documents. The Company agrees that the issuance of a Press Release for an Election Notice shall not be required as the delivery of an Election Notice shall not constitute material, nonpublic information. No later than 5:30 p.m., New York time, on the fourth (4<sup>th</sup>) Business Day after the date of this Agreement, the Company shall file a Current Report on Form 8-K describing all the material terms of the transactions contemplated by the Transaction Documents in the form required by the 1934 Act and attaching all the material Transaction Documents (the "8-K Filing"). From and after the issuance of the Press Release, the Company shall have disclosed all material, non-public information (if any) provided to any of the Buyers by the Company or any of its Subsidiaries or any of their respective officers, directors, employees or agents. In addition, effective upon the issuance of the Press Release, the Company acknowledges and agrees that any and all confidentiality or similar obligations under any agreement, whether written or oral,

between the Company, any of its Subsidiaries or any of their respective officers, directors, affiliates, employees or agents, on the one hand, and any of the Buyers or any of their affiliates, on the other hand, shall have terminated and none of the Buyers have been subject to any such obligation since the issuance of the Press Release.

Limitations on Disclosure. Other than as required under the Transaction Documents (but subject to any other disclosure obligations of the Company with respect thereto), the Company shall not, and the Company shall cause each of its Subsidiaries and each of its and their respective officers, directors, employees and agents not to, provide any Buyer with any material, non-public information regarding the Company or any of its Subsidiaries from and after the date hereof unless prior thereto such Buyer shall have consented in writing to the receipt of such information and agreed with the Company to keep such information confidential. If any material, nonpublic information is required to be provided by the Company or any of its Subsidiaries to any Buyer pursuant to the Transaction Documents, the Company shall obtain each Buyer's prior written consent prior to providing such information to such Buyer, and if any Buyer fails to provide such written consent, the Company shall not be deemed to be in breach of any of the Transaction Documents as a result of the failure to provide such information. To the extent that the Company delivers any material, non-public information to a Buyer without such Buyer's prior written consent in breach of the foregoing sentence, the Company hereby covenants and agrees that such Buyer shall not have any duty of confidentiality with respect to, or a duty not to trade on the basis of, such material, non-public information, provided that the Buyer shall remain subject to applicable law. Subject to the foregoing, neither the Company, its Subsidiaries nor any Buyer shall issue any press releases or any other public statements with respect to the transactions contemplated hereby, except the Press Release and the 8-K filing; provided, however, the Company shall be entitled, without the prior approval of any Buyer, to make any press release or other public disclosure with respect to such transactions (i) in substantial conformity with the 8-K Filing and (ii) as is required by applicable law and regulations (provided that in the case of clause (i) above, each Buyer shall be consulted by the Company in connection with any such press release or other public disclosure prior to its release). Without the prior written consent of the applicable Buyer (which may be granted or withheld in such Buyer's sole discretion), the Company shall not (and shall cause each of its Subsidiaries and affiliates to not) disclose the name of such Buyer in any filing, announcement, release or otherwise, except in the 8-K Filing and as otherwise may be required by applicable law or regulations. Notwithstanding anything contained in this Agreement to the contrary and without implication that the contrary would otherwise be true, the Company expressly acknowledges and agrees that no Buyer shall have (unless expressly agreed to by a particular Buyer after the date hereof in a written definitive and binding agreement executed by the Company and such particular Buyer (it being understood and agreed that no Buyer may bind any other Buyer with respect thereto)), any duty of confidentiality with respect to, or a duty not to trade on the basis of, any material, non-public information regarding the Company or any of its Subsidiaries.

# (j) Additional Issuance of Securities.

(i) The Company agrees that for the period commencing on the date hereof and ending on the date immediately following the ninetieth (90th) calendar day after the Initial Closing Date (such period, the "Restricted Period"), neither the Company nor any of its Subsidiaries shall directly or indirectly issue, offer, sell, grant any option or right to purchase, or otherwise dispose of (or register or amend any outstanding registration statements or file any shelf registration statements or announce any issuance, offer, sale, grant of any option or right to purchase or other disposition of) any equity security or any equity-linked or related security (including, without limitation, any "equity security" (as that term is defined under Rule 405 promulgated under the 1933 Act)), any Convertible

Securities, any preferred stock or any purchase rights. Notwithstanding the foregoing, this Section 4(j)(i) shall not apply during the Restricted Period in respect of (A) the issuance of Options or Convertible Securities issued under any Approved Stock Plan, so long as (i) the aggregate number of shares issued and issuable pursuant thereto does not exceed 5% of the Common Stock issued and outstanding immediately prior to the date hereof and (ii) the exercise price of any such options is not lowered, none of such options are amended to increase the number of shares issuable thereunder and none of the terms or conditions of any such options are otherwise materially changed in any manner that adversely affects any of the Buyers, (B) any shares of Common Stock issued by the Company upon the conversion, exercise, or settlement of (or otherwise pursuant to the terms of) Convertible Securities (other than standard options to purchase Common Stock or other incentive equity awards issued pursuant to an Approved Stock Plan that are covered by clause (A) above) outstanding on the date hereof and referred to in the Registration Statement (other than standard options to purchase Common Stock or other incentive equity awards issued pursuant to an Approved Stock Plan that are covered by clause (A) above); provided, that the conversion, exercise or other method of issuance (as the case may be) of any such Convertible Security is made solely pursuant to the conversion, exercise or other method of issuance (as the case may be) provisions of such Convertible Security that were in effect on the date immediately prior to the date of this Agreement, the conversion, exercise or issuance price of any such Convertible Securities (other than standard options to purchase Common Stock or other incentive equity awards issued pursuant to an Approved Stock Plan that are covered by clause (A) above) is not lowered, none of such Convertible Securities (other than standard options to purchase Common Stock issued pursuant to an Approved Stock Plan that are covered by clause (A) above) are amended to increase the number of shares issuable thereunder and none of the terms or conditions of any such Convertible Securities (other than standard options to purchase Common Stock or other incentive equity awards issued pursuant to an Approved Stock Plan that are covered by clause (A) above) are otherwise materially changed in any manner that adversely affects any of the Buyers, (C) the issuances of the Underlying Shares, (D) any sale or issuance of securities under an agreement, the terms of which have been approved by the Required Holders, providing for an "atthe-market" offering within the meaning of Rule 415(a)(4) of the Securities Act (an "ATM Sales Agreement") other than a sale or issuance in which a single identified investor or group of identified investors purchases in excess of three million dollars (\$3,000,000) in the aggregate of Common Stock, which shall include that certain Sales Agreement, dated as of February 6, 2023, by and between the Company and Needham & Company, LLC and any replacement thereof, (E) any sale or issuance of securities pursuant to that certain Business Combination Agreement, dated as of March 22, 2021, by and among JAWS Spitfire Acquisition Corporation, Spitfire Merger Sub, Inc. and the Company, as amended by Amendment No. 1 to the Business Combination Agreement, dated as of July 20, 2021, by and among JAWS Spitfire Acquisition Corporation, Spitfire Merger Sub, Inc. and the Company, as in effect as of the date hereof and without giving effect to any amendment, modification, waiver or supplement thereto (the "Business Combination Agreement"); provided, that the sale or issuance of any such securities is made solely pursuant to the Business Combination Agreement that was in effect on the date immediately prior to the date of this Agreement, the issuance price of any such securities is not lowered, none of such securities are amended to increase the number of shares issuable thereunder and none of the terms or conditions of any such securities are otherwise materially changed in any manner that adversely affects any of the Buyers or (F) the issuance of securities of the Company or any Subsidiary of the Company as consideration in acquisitions, divestitures, partnerships, licenses, collaborations or strategic transactions approved by the board of directors or a majority of the members of a committee of directors of the Company or applicable Subsidiary established for such purpose, provided that any such issuance shall only be to a Person (or to the equity holders of a Person) which is, itself or through its subsidiaries or controlled affiliates, an operating company or an asset in a

business synergistic with the business of the Company and/or its Subsidiaries and shall provide to the Company and/or its Subsidiaries additional benefits in addition to the investment of funds, but shall not include a transaction in which a Subsidiary of the Company is issuing securities primarily for the purpose of raising capital or to an entity whose primary business is investing in securities (the "Strategic Investment"). An "Approved Stock Plan" means any security-based compensation plan which has been approved by the Board of Directors of the Company prior to the date hereof, pursuant to which Common Stock, options to purchase Common Stock and other incentive equity awards may be issued to any employee, officer, consultant or director for services provided to the Company in their capacity as such, and not for the purpose of raising capital, pursuant to any consulting agreement, advisory agreement or independent contractor agreement approved by the Board of Directors or the compensation committee thereof.

- (ii) So long as any Notes remain outstanding or during any period of time when a Subsequent Closing could still potentially occur, the Company and each Subsidiary shall be prohibited from effecting, or entering into an agreement directly or indirectly to effect a Variable Rate Transaction. "Variable Rate Transaction" means a transaction in which the Company or any Subsidiary (i) issues or sells any Convertible Securities either (A) at a conversion, exercise or exchange rate or other price that is based upon and/or varies with the trading prices of or quotations for the shares of Common Stock at any time after the initial issuance of such Convertible Securities (other than the Notes), or (B) with a conversion, exercise or exchange price that is subject to being reset at some future date after the initial issuance of such Convertible Securities or upon the occurrence of specified or contingent events directly or indirectly related to the business of the Company or the market for the Common Stock, other than pursuant to customary adjustments for stock splits, stock dividends, stock combinations, recapitalizations and similar events or (ii) enters into any agreement (including, without limitation, an equity line of credit) whereby the Company or any Subsidiary may sell securities at a future determined price (other than standard and customary "preemptive" or "participation" rights); provided, that for avoidance of doubt, neither the entry into an ATM Sales Agreement nor the Business Combination Agreement nor the issuance of shares of Common Stock pursuant thereto shall be considered a "Variable Rate Transaction."
- (iii) So long as any Notes remain outstanding, the Company will not, without the prior written consent of the Required Holders (as defined below), issue any Notes (other than to the Buyers as contemplated hereby) and the Company shall not issue any other securities that would cause a breach or default under the Notes.
- (iv) Each Buyer shall be entitled to obtain injunctive relief against the Company and its Subsidiaries to preclude any issuance prohibited by this Section 4(j), which remedy shall be in addition to any right to collect damages.
- (k) <u>Compliance with Laws</u>. None of the Company or any of its Subsidiaries shall violate any law, ordinance or regulation of any Governmental Entity, except where such violations would not reasonably be expected to result, either individually or in the aggregate, in a Material Adverse Effect.
- (l) <u>Passive Foreign Investment Company</u>. The Company shall conduct its business, and shall cause its Subsidiaries to conduct their respective businesses, in such a manner as will ensure that the Company will not be deemed to constitute a passive foreign investment company within the meaning of Section 1297 of the Code.
- (m) Restriction on Redemption and Cash Dividends. So long as any of the Notes are outstanding or during any period of time when a Subsequent Closing could still potentially

occur, except as otherwise permitted under the Notes, the Company shall not, directly or indirectly, redeem, or declare or pay any cash dividend or distribution on, any securities of the Company without the prior express written consent of the Required Holders (other than as required by the Notes or as required by the terms thereof as in effect on the date hereof); *provided*, *however*, that such written consent shall not be required for any repurchases, forfeitures, withholdings or transfers of securities pursuant to a net exercise of a Convertible Security to cover the payment of the exercise prices or the payment of withholding of taxes associated with the exercise, settlement or vesting of equity awards under any equity compensation plan of the Company or repurchases of securities upon an employee's, contractor's or consultant's termination of services.

- (n) <u>Corporate Existence</u>. So long as any Notes remain outstanding, the Company shall not be party to any Fundamental Change (as defined in the Notes) unless the Company is in compliance with the applicable provisions governing Fundamental Changes set forth in the Notes
- (o) <u>Issuance Procedures</u>. The terms of the Notes set forth the totality of the procedures required of the Buyers in order to receive shares of Common Stock pursuant to the Notes. No additional legal opinion, other information or instructions shall be required of the Buyers to receive shares of Common Stock pursuant to the Notes. The Company shall honor conversion of the Notes, and shall deliver the Underlying Shares in accordance with the terms, conditions and time periods set forth in the Notes. Except as explicitly set forth in the Notes, no legal opinion, information or instructions shall be required of the Buyers to receive Underlying Shares pursuant to the Notes. The Company shall deliver the Underlying Shares in accordance with the terms, conditions and time periods set forth in the Notes.
- (p) Regulation M. The Company will not take any action prohibited by Regulation M under the 1934 Act, in connection with the distribution of the Securities contemplated hereby.
- (q) <u>Integration</u>. None of the Company, any of its affiliates (as defined in Rule 501(b) under the 1933 Act), or any person acting on behalf of the Company or such affiliate will sell, offer for sale, or solicit offers to buy or otherwise negotiate in respect of any security (as defined in the 1933 Act) which will be integrated with the sale of the Securities in a manner which would require stockholder approval under the rules and regulations of NYSE and the Company will take all action that is appropriate or necessary to assure that its offerings of other securities will not be integrated for purposes of the 1933 Act or the rules and regulations of NYSE, with the issuance of Securities contemplated hereby.
- (r) Stockholder Approval. At the next annual meeting of the stockholders following the Initial Closing or, if earlier, via a special meeting of stockholders within seventy-five (75) days of a written request (of which an email request may be sufficient) from the Buyers (the "Stockholder Meeting"), the Company agrees to hold a stockholder vote for the purpose of obtaining the Requisite Stockholder Approval. The Company will prepare and file with the SEC a proxy statement to be sent to the Company's stockholders in connection with the Stockholder Meeting (the "Proxy Statement"). The Proxy Statement shall include the Board of Directors' recommendation that the holders of shares of the Company's Common Stock vote in favor of the Requisite Stockholder Approval. The Company shall use its best efforts to obtain approval of the Requisite Stockholder Approval is not obtained at or prior to the Stockholder Meeting, the Company will hold a special meeting of the stockholders of the Company for the purposes of obtaining such Requisite Stockholder Approval is obtained, and the Board of Directors will recommend that the holders of shares of the Company's Common Stock vote in favor of the Requisite Stockholder Approval at each such meeting.

- (s) Press Releases. The Company shall not, and shall not permit any of its Subsidiaries to, issue or disseminate to the public (by advertisement, press release or otherwise), submit for publication or otherwise cause or seek to publish any information naming any of the Buyers without the prior written consent of such Buyer; provided, that nothing in the foregoing shall be construed to prohibit the Company from making any disclosure, submission or filing (i) which it is required to make by applicable law or pursuant to judicial process, (ii) pursuant to Section 4(j) or as otherwise required by federal securities law in connection with the filing of final Transaction Documents with the SEC, or (iii) to the extent such disclosure is required by law or NYSE regulations; provided, further, that (i) such filing or submission shall contain only such information as is necessary to comply with applicable law or judicial process and (ii) unless specifically prohibited by applicable law or court order, the Company shall promptly notify the Buyers of the requirement to make such submission or filing and provide the Buyers with a copy thereof.
- Share Reserve. So long as any of the Notes remain outstanding, the Company shall at all times have not less than a number of authorized but unissued shares of Common Stock equal to the sum of (i) fifty million (50,000,000) shares of Common Stock, which shall not be exclusively reserved for issuance pursuant to the Notes and (ii) one hundred percent (100%) of a fraction the numerator of which shall be the then outstanding Principal Amount of all Notes issued pursuant to this Agreement, and the denominator of which shall be the Conversion Price, which shall be reserved for issuance pursuant to the Notes (which such reservation shall be for the sole benefit of and exclusive availability for the Buyers) (collectively, the "Required Reserve Amount"); provided, that at no time shall the number of shares of Common Stock reserved pursuant to this Section 4(t) be reduced other than in connection with any stock combination, reverse stock split or other similar transaction. The amounts set forth in the definition of Required Reserve Amount (including, without limitation, each increase in the number of shares so reserved) shall be allocated pro rata among the holders of the Notes based on the number of shares of Common Stock issuable pursuant to the Notes held by each holder thereof on the date of issuance of the Notes (without regards to any limitations on issuance of shares contained therein) (collectively, the "Authorized Share Allocation"). In the event that a holder shall sell or otherwise transfer any of such holder's Notes, each transferee shall be allocated a pro rata portion of such holder's Authorized Share Allocation. Any Common Stock reserved and allocated to any Person which ceases to hold any Notes shall be allocated to the remaining holders of the Notes, pro rata based on the number of shares of Common Stock issuable pursuant to the Notes then held by such holders thereof (without regard to any limitations on exercise). If at any time the number of shares of Common Stock authorized and reserved for issuance is not sufficient to meet the Required Reserve Amount, the Company will promptly take all corporate action necessary to authorize and reserve a sufficient number of shares, including, without limitation, calling a special meeting of stockholders to authorize additional shares to meet the Company's obligations pursuant to the Transaction Documents, in the case of an insufficient number of authorized shares, obtain stockholder approval (if required) of an increase in such authorized number of shares, and voting the management shares of the Company in favor of an increase in the authorized shares of the Company to ensure that the number of authorized shares is sufficient to meet the Required Reserve Amount.
- (u) <u>Legends.</u> Certificates and any other instruments evidencing the Purchased Securities or the Underlying Shares, shall not bear any restrictive or other legend.
- (v) <u>Lock-Up and Voting Agreements</u>. The Company may not amend, modify or waive the terms of the Lock-Up and Voting Agreements in any respect without the prior written consent of the Required Holders.
- (w) Not an Underwriter. Neither the Company nor any Subsidiary or affiliate thereof shall identify any Buyer as being an underwriter or potentially being an "underwriter" in any disclosure to, or filing with, the SEC, NYSE or any other Eligible Market. No Buyer shall be

required to agree or admit that it is, or may be, acting as an "underwriter" in connection with the transactions contemplated hereby or agree to be named as an underwriter or as potentially being an underwriter in any public disclosure or filing with the SEC, the NYSE or any other Eligible Market, nor shall any Buyer be required to make any representations to, or undertake any obligations to, the SEC in connection with any registration statement filed by the Company. Any Buyer being deemed an underwriter, or potentially to be an underwriter, by the SEC shall not relieve the Company of any obligations it has under this Agreement or any other Transaction Document.

- Security Agreements. At the Initial Closing (which period may be extended in the reasonable discretion of the Collateral Agent (as defined in the Notes)), the Company or relevant Subsidiary of the Company shall deliver to the Collateral Agent (i) a perfection certificate, in the form attached hereto as Exhibit C, which describes in detail reasonably acceptable to the Collateral Agent the Collateral (as defined below) to be delivered (a "Perfection Certificate"), (ii) a U.S. Security Agreement, to be dated the Initial Closing Date, among the grantors named therein and the secured party named therein (the "Main Security Agreement"), (iii) a U.S. Intellectual Property Security Agreement, to be dated the Initial Closing Date, among the grantor(s) named therein and the security party named therein (the "IP Security Agreement") and (iv) UCC financing statements ("UCC Financing Statements"), each of (i) through (iv), in form and substance satisfactory to the Collateral Agent, which create a first lien security interest in all assets of the Company including, but not limited to, its intellectual property (subject to prior Liens and other customary exclusions, in each case acceptable to the Collateral Agent in its sole discretion) (the "Collateral") and shall perfect a first lien security interest in all such assets of the Company other than the Company's non-U.S. assets and its bank accounts. As soon as reasonably practicable, but in any event before thirty (30) days after the Initial Closing (which period may be extended in the reasonable discretion of the Collateral Agent (as defined in the Notes)), the Company or relevant Subsidiary of the Company shall deliver to the Collateral Agent (a) such additional security documents, including deposit account control agreements, in form and substance reasonably acceptable to the Collateral Agent, which perfect a first lien security interest in all remaining assets of the Company (subject to prior Liens and other customary exclusions, in each case acceptable to the Collateral Agent in its sole discretion) (the "Ancillary Security Documents" and together with the Perfection Certificate, Main Security Agreement, the IP Security Agreement and UCC Financing Statements, the "Security Agreements") and (b) with respect to such Ancillary Security Documents delivered, customary legal opinions relating to such Ancillary Security Documents, in form and substance reasonably acceptable to the Collateral Agent.
- (y) ATM Sales Program. The Company shall at all times have access to an ATM Sales Program, the terms of which have been approved by the Required Holders with aggregate available, accessible and unused capacity to generate gross proceeds to the Company of at least ten million dollars (\$10,000,000). On or before December 31, 2023, the Company shall establish a new ATM Sales Program with aggregate available, accessible and unused capacity to generate gross proceeds to the Company of at least seventy five million dollars (\$75,000,000) as of December 31, 2023, and such ATM Program shall not be subsequently canceled, or amended or otherwise modified to reduce such amount, without the written consent of the Required Holders. Each of the Buyers hereby approves the terms of the Company's ATM Sales Program and the related ATM Sales Agreement as in effect at the date of this Agreement, and agrees that its approval of any ATM Sales Program and related ATM Sales Agreement after the date of this Agreement will not be unreasonably withheld, conditioned or delayed.
- (z) <u>Registration Statement</u>. Until the later of (i) the date that no additional Subsequently Purchased Notes may be issued pursuant to this Agreement and (ii) the date that is one (1) year from the Initial Closing Date, the Company shall maintain the effectiveness of the Registration Statement and shall maintain unused issuance capacity thereunder sufficient for delivery of all the Subsequently Purchased Notes issuable pursuant to this Agreement.

## 5. REGISTER; TRANSFER AGENT INSTRUCTIONS.

- (a) Register. The Company shall cause the Trustee to maintain at its principal executive offices (or such other office or agency of the Trustee as it may designate pursuant to the Indenture), a register for the registration of the Notes in which the Trustee shall record the name and address of the Person in whose name the Purchased Securities have been issued (including the name and address of each transferee), the aggregate amount of the Notes held by such Person and the number of Underlying Shares issuable pursuant to the Notes held by such Person. The Company shall cause the Trustee to keep the register open and available at reasonable times upon reasonable prior notice during business hours for inspection of any Buyer or its legal representatives. This provision shall be construed such that the Securities and the Notes are at all times maintained in "registered form" within the meanings of Sections 163(f), 871(h)(2) and 881(c)(2) of the Code and any Treasury Regulations promulgated thereunder.
- Transfer Agent Instructions. The Company shall issue irrevocable instructions to its transfer agent and any subsequent transfer agent (as applicable) (the "Transfer Agent") in a form acceptable to each of the Buyers (the 'Irrevocable Transfer Agent Instructions") to credit shares to each such Buyer's (or its designee's) account at DTC through its Deposit/Withdrawal At Custodian ("DWAC") System, provided that the Transfer Agent is participating in the DTC Fast Automated Securities Transfer Program ("FAST") and the shares are then eligible for transfer through the DWAC System, or, if the Transfer Agent is not participating in FAST or if the shares are not then eligible for transfer through the DWAC system, issue and dispatch by overnight courier to the address as specified in the notice that the Company is electing to issue Common Stock pursuant to the terms of the Notes or that the Buyers are electing to receive Common Stock pursuant to the Notes, a certificate, registered in the name of such Buyer or its designee, for the Underlying Shares to which the Buyer is entitled, for the applicable Underlying Shares in such amounts as specified from time to time by the Company or the Buyers, as the case may be, pursuant to the terms of the Notes. The Company represents and warrants that no instruction other than the Irrevocable Transfer Agent Instructions referred to in this Section 5(b) will be given by the Company to the Transfer Agent with respect to the Securities, and that the Securities shall otherwise be freely transferable on the books and records of the Company, as applicable, to the extent provided in this Agreement and the other Transaction Documents. If a Buyer effects a sale, assignment or transfer of the Purchased Securities, the Company shall permit the transfer and shall promptly instruct the Transfer Agent to issue one or more certificates or credit shares to the applicable balance accounts at DTC in such name and in such denominations as specified by such Buyer to effect such sale, transfer or assignment. The Company acknowledges that a breach by it of its obligations hereunder will cause irreparable harm to a Buyer. Accordingly, the Company acknowledges that the remedy at law for a breach of its obligations under this Section 5(b) will be inadequate and agrees, in the event of a breach or threatened breach by the Company of the provisions of this Section 5(b), that a Buyer shall be entitled, in addition to all other available remedies, to an order and/or injunction restraining any breach and requiring immediate issuance and transfer, without the necessity of showing economic loss and without any bond or other security being required. Any fees (with respect to the Transfer Agent, counsel to the Company or otherwise) associated with the removal of any legends on any of the Securities shall be borne by the Company.
- (c) <u>FAST Compliance</u>. While any Notes remain outstanding, the Company shall maintain a transfer agent that participates in FAST.

# 6. CONDITIONS TO THE COMPANY'S OBLIGATION TO SELL THE PURCHASED SECURITIES.

(a) The obligation of the Company hereunder to issue and sell the Purchased Securities to each Buyer at the Initial Closing and each Subsequent Closing, if any, is subject to the satisfaction, at or before the Initial Closing Date and each Subsequent Closing Date, if any, of

each of the following conditions, provided that these conditions are for the Company's sole benefit and may be waived by the Company at any time in its sole discretion by providing each Buyer with prior written notice thereof:

- (i) Such Buyer shall have executed each of the other Transaction Documents (other than, solely with respect to the Initial Closing, the Ancillary Security Documents) to which it is a party and delivered the same to the Company.
- (ii) Such Buyer and each other Buyer shall have delivered to the Company the purchase price for the Purchased Securities being purchased by such Buyer at such Closing by wire transfer of immediately available funds in accordance with a Flow of Funds Letter with respect to the Securities to be purchased at such Closing.
- (iii) The representations and warranties of such Buyer shall be true and correct in all material respects (except for such representations and warranties that are qualified by materiality or material adverse effect, which shall be true and correct in all respects) as of the date when made and as of the date of each such Closing as though originally made at that time (except for representations and warranties that speak as of a specific date, which shall be true and correct as of such specific date), and such Buyer shall have performed, satisfied and complied in all material respects with the covenants, agreements and conditions required by this Agreement to be performed, satisfied or complied with by such Buyer at or prior to the date of such Closing.

#### 7. CONDITIONS TO EACH BUYER'S OBLIGATION TO PURCHASE THE PURCHASED SECURITIES.

- (a) The obligation of each Buyer hereunder to purchase its Initial Purchased Notes at the Initial Closing is subject to the satisfaction, at or before the Initial Closing Date, of each of the following conditions, provided that these conditions are for each Buyer's sole benefit and may be waived by such Buyer at any time in its sole discretion by providing the Company with prior written notice thereof:
  - (i) The Company and each Subsidiary (as the case may be) shall have duly executed and delivered to such Buyer each of the Transaction Documents (other than the Ancillary Security Documents) to which it is a party and the Company and the Trustee shall have duly executed and delivered to such Buyer the Initial Purchased Notes set forth across from such Buyer's name on the Schedule of Buyers at the Initial Closing pursuant to this Agreement.
  - (ii) Such Buyer shall have received the opinion of Fenwick & West LLP, the Company's counsel, dated as of the Initial Closing Date, in the form acceptable to such Buyer.
  - (iii) The Company shall have delivered to such Buyer a copy of the Irrevocable Transfer Agent Instructions, dated as of the Initial Closing Date, in the form acceptable to such Buyer, which instructions shall have been delivered to and acknowledged in writing by the Transfer Agent.
  - (iv) The Company shall have delivered to such Buyer a certificate evidencing the formation and good standing of the Company and each of its Subsidiaries in each such entity's jurisdiction of formation issued by the Secretary of State (or comparable office) of such jurisdiction of formation as of a date within ten (10) days of the Initial Closing Date, along with a bring-down letter certifying the good standing of the Company and each of its Subsidiaries as of the Initial Closing Date, along with a bring-

down letter certifying the good standing of the Company and each of its Subsidiaries as of the Initial Closing Date.

- (v) The Company shall have delivered to such Buyer a certified copy of the Certificate of Incorporation of the Company as certified by the Delaware Secretary of State within ten (10) days of the Initial Closing Date.
- (vi) The Company shall have delivered to such Buyer a certificate, in the form acceptable to such Buyer, executed by the Secretary of the Company and dated as of the Initial Closing Date, as to (i) the resolutions consistent with Section 3(d) as adopted by the Company's Board of Directors or a duly authorized committee thereof in a form reasonably acceptable to such Buyer, (ii) the Certificate of Incorporation of the Company and (iii) the Bylaws of the Company, each as in effect at the Initial Closing.
- (vii) Each and every representation and warranty of the Company shall be true and correct in all material respects (except for such representations and warranties that are qualified by materiality or material adverse effect, which shall be true and correct in all respects) as of the date when made and as of the Initial Closing Date as though originally made at that time (except for representations and warranties that speak as of a specific date, which shall be true and correct as of such specific date) and the Company shall have performed, satisfied and complied in all respects with the covenants, agreements and conditions required to be performed, satisfied or complied with by the Company at or prior to the Initial Closing Date. Such Buyer shall have received a certificate, duly executed by the Chief Executive Officer or Chief Financial Officer of the Company, dated as of the Initial Closing Date, to the foregoing effect and as to such other matters as may be reasonably requested by such Buyer in the form acceptable to such Buyer.
- (viii) The Company shall have delivered to such Buyer a letter from the Transfer Agent certifying the number of shares of Common Stock outstanding on the Initial Closing Date immediately prior to the Initial Closing.
- (ix) The Common Stock (A) shall be designated for quotation or listed (as applicable) on NYSE and (B) shall not have been suspended, as of the Initial Closing Date, by the SEC or NYSE from trading on NYSE nor shall suspension by the SEC or NYSE have been threatened, as of the Initial Closing Date, either (1) in writing by the SEC or NYSE or (2) by falling below the minimum maintenance requirements of NYSE.
- (x) The Company shall have obtained all governmental, regulatory or third party consents and approvals, if any, necessary for the sale of the Initial Purchased Notes, including without limitation, those required by NYSE, if any.
- (xi) No statute, rule, regulation, executive order, decree, ruling or injunction shall have been enacted, entered, promulgated or endorsed by any court or Governmental Entity of competent jurisdiction that prohibits the consummation of any of the transactions contemplated by the Transaction Documents.
- (xii) Since the date of execution of this Agreement, no event or series of events shall have occurred that would have or result in a Material Adverse Effect.
- (xiii) The Company shall have submitted a Supplemental Listing Application with NYSE relating to the issuance of the Underlying Shares as contemplated hereby.
- (xiv) Such Buyer shall have received a letter on the letterhead of the Company, duly executed by the Chief Executive Officer or Chief Financial Officer of the Company,

setting forth the wire amounts of each Buyer and the wire transfer instructions of the Company (a "Flow of Funds Letter") with respect to the Initial Purchased Notes.

- (xv) The Company shall have delivered to such Buyer the results of a recent lien, bankruptcy and judgment search in each relevant jurisdiction with respect to the Company and its Subsidiaries and such search shall reveal no Liens on any of the Collateral or other assets of the Company and its Subsidiaries except, in the case of assets other than Collateral, for Permitted Liens.
- (xvi) The Company shall have delivered to Buyer a duly completed and executed a Perfection Certificate, dated no earlier than five (5) days prior to the Initial Closing Date.
- (xvii) The Company shall have established and in effect an ATM Sales Agreement in place pursuant to which the Company has the ability to issue and sell shares of Common Stock from time to time (an "ATM Sales Program"). The ATM Sales Program shall have aggregate available, accessible and unused capacity to generate gross proceeds to the Company of at least twenty million dollars (\$20,000,000) as of the Initial Closing Date.
- (xviii) All costs, fees, expenses (including, without limitation, legal fees and expenses) contemplated hereby to be payable to the Buyers shall have been paid to the extent due and, in the case of expenses of the Buyers that are reimbursable in accordance herewith, invoiced at least one day prior to the Initial Closing Date.
- (xix) Such Buyer shall have received Lock-Up and Voting Agreements executed by (i) each member of the Company's Board of Directors and (ii) each of the officers (as defined in Rule 16a-1(f) promulgated under the 1934 Act) of the Company (all of (i) through (ii) of which are identified on the Schedule of Lock-Up Parties attached hereto).
- (xx) Concurrently with the issuance the Initial Purchased Notes, the Company shall have repaid, or cause to be repaid, all of its outstanding indebtedness under the SVB Loan Agreement, and the Company shall have obtained and delivered payoff letters and releases relating to the SVB Loan Agreement to such Buyer in form and substance satisfactory to such Buyer.
- (xxi) The Company or relevant Subsidiary of the Company shall deliver to the Buyer the Security Agreements (other than the Ancillary Security Documents), in form and substance satisfactory to the Buyer, which create a first lien security interest in the Collateral and perfect a first lien security interest in all assets of the Company other than the Company's non-U.S. assets and its bank accounts (subject to prior Liens and other customary exclusions, in each case acceptable to the Collateral Agent in its sole discretion).
- (xxii) The Company shall have made a public announcement of its earnings for the three-month period ending on June 30, 2023 by no later than 5:00 p.m. New York City time on August 10, 2023.
- (xxiii) The Company and its Subsidiaries shall have delivered to such Buyer such other documents, instruments or certificates relating to the transactions contemplated by the Transaction Documents (other than the Ancillary Security Documents) as such Buyer or its counsel may reasonably request.

- (b) The obligation of each Buyer hereunder to purchase the Subsequently Purchased Notes at each Subsequent Closing is subject to the satisfaction, at or before each Subsequent Closing Date, of each of the following conditions, provided that these conditions are for each Buyer's sole benefit and may be waived by such Buyer at any time in its sole discretion by providing the Company with prior written notice thereof:
  - (i) The Registration Statement shall remain effective and the Underlying Shares shall be Freely Tradeable (as defined in the Notes).
  - (ii) The Company and each Subsidiary (as the case may be) shall have duly executed and delivered to such Buyer each of the Transaction Documents to which it is a party and the Company shall have duly executed and delivered to the Buyers the Subsequently Purchased Notes.
  - (iii) Such Buyer shall have received the opinion of Fenwick & West LLP, the Company's counsel, dated as of each Subsequent Closing Date, in the form acceptable to such Buyer.
  - (iv) The Company shall have delivered to such Buyer a copy of the Irrevocable Transfer Agent Instructions, dated as of each Subsequent Closing Date, in the form acceptable to such Buyer, which instructions shall have been delivered to and acknowledged in writing by the Transfer Agent.
  - (v) The Company shall have delivered to such Buyer a certificate evidencing the formation and good standing of the Company and each of its Subsidiaries in each such entity's jurisdiction of formation issued by the Secretary of State (or comparable office) of such jurisdiction of formation as of a date within ten (10) days of each Subsequent Closing Date.
  - (vi) The Company shall have delivered to such Buyer a certified copy of the Certificate of Incorporation of the Company as certified by the Delaware Secretary of State within ten (10) days of each Subsequent Closing Date.
  - (vii) The Company shall have delivered to such Buyer a certificate, in the form acceptable to such Buyer, executed by the Secretary of the Company and dated as of each Subsequent Closing Date, as to (i) the resolutions consistent with Section 3(d) as adopted by the Company's Board of Directors or a duly authorized committee thereof in a form reasonably acceptable to such Buyer, (ii) the Certificate of Incorporation of the Company and (iii) the Bylaws of the Company, each as in effect at each Subsequent Closing Date.
  - (viii) Each and every representation and warranty of the Company shall be true and correct in all material respects (except for such representations and warranties that are qualified by materiality or material adverse effect, which shall be true and correct in all respects) as of the date when made and as of each Subsequent Closing Date as though originally made at that time (except for representations and warranties that speak as of a specific date, which shall be true and correct as of such specific date) and the Company shall have performed, satisfied and complied in all respects with the covenants, agreements and conditions required to be performed, satisfied or complied with by the Company at or prior to each Subsequent Closing Date. Such Buyer shall have received a certificate, duly executed by the Chief Executive Officer or Chief Financial Officer of the Company, dated as of each Subsequent Closing Date, to the foregoing effect and as to such other matters as may be reasonably requested by such Buyer in the form acceptable to such Buyer.

- (ix) The Company shall have delivered to such Buyer a letter from the Transfer Agent certifying the number of shares of Common Stock outstanding on each Subsequent Closing Date immediately prior to each Subsequent Closing.
- (x) The Common Stock (A) shall be designated for quotation or listed (as applicable) on NYSE and (B) shall not have been suspended, as of each Subsequent Closing Date, by the SEC or NYSE from trading on NYSE nor shall suspension by the SEC or NYSE have been threatened, as of each Subsequent Closing Date, either (1) in writing by the SEC or NYSE or (2) by falling below the minimum maintenance requirements of NYSE.
- (xi) The Company shall have obtained all governmental, regulatory or third party consents and approvals, if any, necessary for the sale of the Subsequently Purchased Notes, including without limitation, those required by NYSE, if any.
- (xii) No statute, rule, regulation, executive order, decree, ruling or injunction shall have been enacted, entered, promulgated or endorsed by any court or Governmental Entity of competent jurisdiction that prohibits the consummation of any of the transactions contemplated by the Transaction Documents.
- (xiii) Since the date of execution of this Agreement, no event or series of events shall have occurred that would have or result in a Material Adverse Effect.
- (xiv) The Company shall have delivered to such Buyer the results of a recent lien, bankruptcy and judgment search in each relevant jurisdiction with respect to the Company and its Subsidiaries and such search shall reveal no Liens on any of the Collateral or other assets of the Company and its Subsidiaries except, in the case of assets other than Collateral, for Permitted Liens.
- (xv) The Company shall have delivered to Buyer a duly completed and executed Perfection Certificate, dated no earlier than five (5) days prior to each Subsequent Closing Date.
- (xvi) The Company shall have an ATM Sales Program in effect which shall have aggregate available, accessible and unused capacity to generate gross proceeds to the Company of at least ten million dollars (\$10,000,000) as of each Subsequent Closing Date.
- (xvii) All costs, fees, expenses (including, without limitation, legal fees and expenses) contemplated hereby to be payable to the Buyers shall have been paid to the extent due and, in the case of expenses of the Buyers that are reimbursable in accordance herewith, invoiced at least one day prior to each Subsequent Closing Date.
- (xviii) The Company shall have submitted a Supplemental Listing Application with NYSE relating to the issuance of the Subsequently Purchased Notes and the Underlying Shares as contemplated hereby.
  - (xix) Such Buyer shall have received a Flow of Funds Letter with respect to the Subsequently Purchased Notes.
  - (xx) The Company has been, and continues to remain, in full compliance with the terms and conditions of the Notes.

- (xxvii) After giving pro forma effect to the Subsequent Closing proposed to occur on such Subsequent Closing Date, no Default shall occur or be continuing and no Event of Default shall occur.
- (xxi) The Company and its Subsidiaries shall have delivered to such Buyer such other documents, instruments or certificates relating to the transactions contemplated by the Transaction Documents as such Buyer or its counsel may reasonably request.

#### 8. TERMINATION.

In the event that the Initial Closing shall not have occurred with respect to a Buyer within five (5) Business Days of the date hereof, then such Buyer shall have the right to terminate its obligations under this Agreement with respect to itself at any time on or after the close of business on such date without liability of such Buyer to any other party; provided, however, (i) the right to terminate this Agreement under this Section 8 shall not be available to such Buyer if the failure of the transactions contemplated by this Agreement to have been consummated by such date is the result of such Buyer's breach of this Agreement and (ii) the abandonment of the sale and purchase of the Initial Purchased Notes shall be applicable only to such Buyer providing such written notice; provided, further, that no such termination shall affect any obligation of the Company under this Agreement to reimburse such Buyer for the expenses described in Section 4(g) above. Nothing contained in this Section 8 shall be deemed to release any party from any liability for any breach by such party of the terms and provisions of this Agreement or the other Transaction Documents or to impair the right of any party to compel specific performance by any other party of its obligations under this Agreement or the other Transaction Documents.

### 9. MISCELLANEOUS.

Governing Law; Jurisdiction; Jury Trial. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by the internal laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of Delaware. The Company hereby irrevocably submits to the exclusive jurisdiction of the Court of Chancery of the State of Delaware, for the adjudication of any dispute hereunder or in connection herewith or under any of the other Transaction Documents or with any transaction contemplated hereby or thereby, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of such court, that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper. Each party hereby irrevocably waives personal service of process and consents to process being served in any such suit, action or proceeding by mailing a copy thereof to such party at the address for such notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. Nothing contained herein shall be deemed or operate to preclude any Buyer from bringing suit or taking other legal action against the Company in any other jurisdiction to collect on the Company's obligations to such Buyer or to enforce a judgment or other court ruling in favor of such Buyer. EACH PARTY HEREBY IRRÊVÓCABLY WAIVES ANÝ RIGHT IT MAÝ HAVE TO, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR UNDER ANY OTHER TRANSACTION DOCUMENT OR IN CONNECTION WITH OR ARISING OUT OF THIS AGREEMENT, ANY OTHER TRANSACTION DOCUMENT OR ANY TRANSACTION CONTEMPLATED HEREBY OR THEREBY.

- (b) Counterparts; Electronic Signatures. This Agreement may be executed in two or more identical counterparts, all of which shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party. In the event that any signature is delivered by facsimile transmission or by an e-mail which contains a portable document format (.pdf) file of an executed signature page, such signature page shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such signature page were an original thereof. A party's electronic signature (complying with the New York Electronic Signatures and Records Act (N.Y. State Tech. §§ 301-309), as amended from time to time, or other applicable law) of this Agreement shall have the same validity and effect as a signature affixed by the party's hand.
- (c) <u>Headings; Gender; Interpretation</u>. The headings of this Agreement are for convenience of reference and shall not form part of, or affect the interpretation of, this Agreement. Unless the context clearly indicates otherwise, each pronoun herein shall be deemed to include the masculine, feminine, neuter, singular and plural forms thereof. The terms "including," "includes," "include" and words of like import shall be construed broadly as if followed by the words "without limitation." The terms "herein," "hereunder," "hereof" and words of like import refer to this entire Agreement instead of just the provision in which they are found. Unless the context otherwise requires, references herein: (x) to Articles, Sections, Schedules and Exhibits mean the Articles and Sections of, and Schedules and Exhibits attached to, this Agreement; (y) to an agreement, instrument or other document means such agreement, instrument or other document as amended, supplemented and modified from time to time to the extent permitted by the provisions thereof and (z) to a statute means such statute as amended from time to time and includes any successor legislation thereto and any regulations promulgated thereunder.
- Severability; Maximum Payment Amounts. If any provision of this Agreement is prohibited by law or otherwise determined to be invalid or unenforceable by a court of competent jurisdiction, the provision that would otherwise be prohibited, invalid or unenforceable shall be deemed amended to apply to the broadest extent that it would be valid and enforceable, and the invalidity or unenforceability of such provision shall not affect the validity of the remaining provisions of this Agreement so long as this Agreement as so modified continues to express, without material change, the original intentions of the parties as to the subject matter hereof and the prohibited nature, invalidity or unenforceability of the provision(s) in question does not substantially impair the respective expectations or reciprocal obligations of the parties or the practical realization of the benefits that would otherwise be conferred upon the parties. The parties will endeavor in good faith negotiations to replace the prohibited, invalid or unenforceable provision(s) with a valid provision(s), the effect of which comes as close as possible to that of the prohibited, invalid or unenforceable provision(s). Notwithstanding anything to the contrary contained in this Agreement or any other Transaction Document (and without implication that the following is required or applicable), it is the intention of the parties that in no event shall amounts and value paid by the Company and/or any of its Subsidiaries (as the case may be), or payable to or received by any of the Buyers, under the Transaction Documents (including without limitation, any amounts that would be characterized as "interest" under applicable law) exceed amounts permitted under any applicable law. Accordingly, if any obligation to pay, payment made to any Buyer, or collection by any Buyer pursuant the Transaction Documents is finally judicially determined to be contrary to any such applicable law, such obligation to pay, payment or collection shall be deemed to have been made by mutual mistake of such Buyer, the Company and its Subsidiaries and such amount shall be deemed to have been adjusted with retroactive effect to the maximum amount or rate of interest, as the case may be, as would not be so prohibited by the applicable law. Such adjustment shall be effected, to the extent necessary, by reducing or refunding, at the option of such Buyer, the amount of interest or any other amounts which would constitute unlawful amounts required to be paid or actually paid to such Buyer under the Transaction Documents. For greater certainty, to the

extent that any interest, charges, fees, expenses or other amounts required to be paid to or received by such Buyer under any of the Transaction Documents or related thereto are held to be within the meaning of "interest" or another applicable term to otherwise be violative of applicable law, such amounts shall be pro-rated over the period of time to which they relate.

Entire Agreement; Amendments. This Agreement, the other Transaction Documents and the schedules and exhibits attached hereto and thereto and the instruments referenced herein and therein supersede all other prior oral or written agreements between the Buyers, the Company, its Subsidiaries, their affiliates and Persons acting on their behalf, including, without limitation, any transactions by any Buyer with respect to Common Stock or the Securities, and the other matters contained herein and therein, and this Agreement, the other Transaction Documents, the schedules and exhibits attached hereto and thereto and the instruments referenced herein and therein contain the entire understanding of the parties solely with respect to the matters covered herein and therein; provided, however, nothing contained in this Agreement or any other Transaction Document shall (or shall be deemed to) (i) have any effect on any agreements any Buyer has entered into with, or any instruments any Buyer has received from, the Company or any of its Subsidiaries prior to the date hereof with respect to any prior investment made by such Buyer in the Company or (ii) waive, alter, modify or amend in any respect any obligations of the Company or any of its Subsidiaries, or any rights of or benefits to any Buyer or any other Person, in any agreement entered into prior to the date hereof between or among the Company and/or any of its Subsidiaries and any Buyer, or any instruments any Buyer received from the Company and/or any of its Subsidiaries prior to the date hereof, and all such agreements and instruments shall continue in full force and effect. Except as specifically set forth herein or therein, neither the Company nor any Buyer makes any representation, warranty, covenant or undertaking. For clarification purposes, the Recitals are part of this Agreement. No provision of this Agreement may be amended other than by an instrument in writing signed by the Company and the Required Holders, and any amendment to any provision of this Agreement made in conformity with the provisions of this Section 9(e) shall be binding on all Buyers and holders of Securities, as applicable; provided, that no such amendment shall be effective to the extent that it (A) applies to less than all of the holders of the Securities then outstanding or (B) imposes any obligation or liability on any Buyer without such Buyer's prior written consent (which may be granted or withheld in such Buyer's sole discretion). No waiver shall be effective unless it is in writing and signed by an authorized representative of the waiving party, provided that the Required Holders may waive any provision of this Agreement, and any waiver of any provision of this Agreement made in conformity with the provisions of this Section 9(e) shall be binding on all Buyers and holders of Securities, as applicable, provided that no such waiver shall be effective to the extent that it (1) applies to less than all of the holders of the Securities then outstanding (unless a party gives a waiver as to itself only) or (2) imposes any obligation or liability on any Buyer without such Buyer's prior written consent (which may be granted or withheld in such Buyer's sole discretion). No consideration (other than reimbursement of legal fees) shall be offered or paid to any Person to amend or consent to a waiver or modification of any provision of any of the Transaction Documents unless the same consideration also is offered to all of the parties to the Transaction Documents and all holders of the Purchased Securities. From the date hereof and while any Purchased Securities are outstanding, the Company shall not be permitted to receive any consideration from a Buyer or a holder of Purchased Securities that is not otherwise contemplated by the Transaction Documents in order to, directly or indirectly, induce the Company or any Subsidiary (i) to treat such Buyer or holder of Purchased Securities in a manner that is more favorable than to other similarly situated Buyers or holders of Purchased Securities, or (ii) to treat any Buyer(s) or holder(s) of Purchased Securities in a manner that is less favorable than the Buyer or holder of Purchased Securities that is paying such consideration; provided, however, that the determination of whether a Buyer has been treated more or less favorably than another Buyer shall disregard any securities of the Company purchased or sold by any Buyer. The Company has not, directly or indirectly, made any agreements with any Buyers relating to the terms or conditions of the transactions contemplated by the Transaction Documents except as set forth in the Transaction Documents. Without limiting the foregoing,

the Company confirms that, except as set forth in this Agreement, no Buyer has made any commitment or promise or has any other obligation to provide any financing to the Company, any Subsidiary or otherwise. As a material inducement for each Buyer to enter into this Agreement, the Company expressly acknowledges and agrees that (x) no due diligence or other investigation or inquiry conducted by a Buyer, any of its advisors or any of its representatives shall affect such Buyer's right to rely on, or shall modify or qualify in any manner or be an exception to any of, the Company's representations and warranties contained in this Agreement or any other Transaction Document and (y) nothing contained in any of the SEC Documents shall affect such Buyer's right to rely on, or shall modify or qualify in any manner or be an exception to any of, the Company's representations and warranties contained in this Agreement or any other Transaction Document. "Required Holders" means (I) prior to the Initial Closing Date, each Buyer entitled to purchase Initial Purchased Notes at the Initial Closing, and (II) on or after the Initial Closing Date, holders of a majority of the aggregate principal amount of the Notes, so long as any Notes remain outstanding, and if no Notes remain outstanding, holders of a majority of the Underlying Shares in the aggregate as of such time issued and held by the Buyers or issuable hereunder; provided, that such majority must include High Trail Investments ON LLC or HB SPV I Master Sub LLC, so long as High Trail Investments ON LLC or HB SPV I Master Sub LLC or any of their affiliates hold any Notes or are otherwise entitled to receive Underlying Shares.

Notices. Any notices, consents, waivers or other communications required or permitted to be given under the terms of this Agreement must be in writing and will be deemed to have been delivered: (i) upon receipt, when delivered personally; (ii) upon receipt, when sent by electronic mail (provided that such sent e-mail is kept on file (whether electronically or otherwise) by the sending party and the sending party does not receive an automatically generated message from the recipient's e-mail server that such e-mail could not be delivered to such recipient); or (iii) one (1) Business Day after deposit with an overnight courier service with next day delivery specified, in each case, properly addressed to the party to receive the same. The addresses and e-mail addresses for such communications shall be:

# If to the Company:

Velo3D, Inc. 511 Division Street Campbell, California 95008 Telephone:

William McCombe Attention:

E-Mail:

With a copy (for informational purposes only) to:

Fenwick & West LLP 902 Broadway New York, NY 10010 Attention: Per Chilstrom E-mail: [\*]

If to the Transfer Agent:

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, NY 10004 Attention: Erika Harris

Email: [\*]

If to a Buyer, to (i) its e-mail address set forth on the Schedule of Buyers, with copies to such Buyer's representatives as set forth on the Schedule of Buyers and (ii) to Eric Helenek, High Trail Capital, 80 River Street, Suite 4C, Hoboken, NJ 07030 (telephone: [\*].

with a copy (for informational purposes only) to:

Latham & Watkins LLP 12670 High Bluff Drive San Diego, CA 92130 Telephone: [\*]

Attention: Michael E. Sullivan

E-mail: [\*]

or to such other address, e-mail address and/or to the attention of such other Person as the recipient party has specified by written notice given to each other party five (5) days prior to the effectiveness of such change. Written confirmation of receipt (A) given by the recipient of such notice, consent, waiver or other communication, (B) electronically generated by the sender's e-mail or (C) provided by an overnight courier service shall be rebuttable evidence of personal service, receipt by e-mail or receipt from an overnight courier service in accordance with clause (i), (ii) or (iii) above, respectively.

- (g) Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the parties and their respective successors and assigns, including any purchasers of any of the Purchased Securities. The Company shall not assign this Agreement or any rights or obligations hereunder without the prior written consent of the Required Holders, including, without limitation, by way of a Fundamental Change (unless the Company is in compliance with the applicable provisions governing Fundamental Changes set forth in the Notes). A Buyer may assign some or all of its rights hereunder in connection with any transfer of any of its Purchased Securities without the consent of the Company, provided such assignee agrees in writing to be bound by the provisions hereof that apply to Buyers in which event such assignee shall be deemed to be a Buyer hereunder with respect to such assigned rights.
- (h) <u>No Third Party Beneficiaries</u>. This Agreement is intended for the benefit of the parties hereto and their respective permitted successors and assigns, and is not for the benefit of, nor may any provision hereof be enforced by, any other Person, other than the Indemnitees referred to in Section 9(k).
- (i) <u>Survival</u>. The representations, warranties, agreements and covenants shall survive the Initial Closing and the Subsequent Closings. Each Buyer shall be responsible only for its own representations, warranties, agreements and covenants hereunder.
- (j) <u>Further Assurances</u>. Each party shall do and perform, or cause to be done and performed, all such further acts and things, and shall execute and deliver all such other agreements, certificates, instruments and documents, as any other party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.

## (k) <u>Indemnification</u>.

(i) In consideration of each Buyer's execution and delivery of the Transaction Documents and acquiring the Securities thereunder and in addition to all of the Company's other obligations under the Transaction Documents, the Company shall defend, protect, indemnify and hold harmless each Buyer and each holder of any

Securities and all of their stockholders, partners, members, officers, directors, employees and direct or indirect investors and any of the foregoing Persons' agents or other representatives (including, without limitation, those retained in connection with the transactions contemplated by this Agreement) (collectively, the "Indemnitees") from and against any and all actions, causes of action, suits, claims, losses, costs, penalties, fees, liabilities and damages, and expenses in connection therewith (irrespective of whether any such Indemnitee is a party to the action for which indemnification hereunder is sought), and including reasonable attorneys' fees and disbursements (the "Indemnified Liabilities"), incurred by any Indemnitee as a result of, or arising out of, or relating to (i) any misrepresentation or breach of any representation or warranty made by the Company or any Subsidiary in any of the Transaction Documents, (ii) any breach of any covenant, agreement or obligation of the Company or any Subsidiary contained in any of the Transaction Documents or (iii) any cause of action, suit, proceeding or claim brought or made against such Indemnitee by a third party (including for these purposes a derivative action brought on behalf of the Company or any Subsidiary) or which otherwise involves such Indemnitee that arises out of or results from (A) the execution, delivery, performance or enforcement of any of the Transaction Documents (including, without limitation, any hedging or similar activities in connection therewith), or (B) the status of such Buyer or holder of the Securities either as an investor in the Company pursuant to the transactions contemplated by the Transaction Documents or as a party to this Agreement (including, without limitation, any hedging or similar activities in connection therewith or as a party in interest or otherwise in any action or proceeding for injunctive or other equitable relief); provided, however, that the Company will not be liable in any such case to a Buyer or its related Indemnitees to the extent that any such claim, loss, damage, liability or expense arise primarily out of or is based primarily upon (i) the inaccuracy of any representations and warranties made by such Buyer herein or (ii) the fraud, willful misconduct, gross negligence or bad faith of such Buyer or its related Indemnitees. To the extent that the foregoing undertaking by the Company may be unenforceable for any reason, the Company shall make the maximum contribution to the payment and satisfaction of each of the Indemnified Liabilities which is permissible under applicable law.

Promptly after receipt by an Indemnitee under this Section 9(k) of notice of the commencement of any action or proceeding (including, without limitation, any governmental action or proceeding) involving an Indemnified Liability, such Indemnitee shall, if a claim in respect thereof is to be made against any indemnifying party under this Section 9(k), deliver to the indemnifying party a written notice of the commencement thereof, and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume control of the defense thereof with counsel mutually satisfactory to the indemnifying party and the Indemnitee; provided, however, that an Indemnitee shall have the right to retain its own counsel with the fees and expenses of such counsel to be paid by the indemnifying party if: (i) the indemnifying party has agreed in writing to pay such fees and expenses; (ii) the indemnifying party shall have failed promptly to assume the defense of such Indemnified Liability and to employ counsel reasonably satisfactory to such Indemnitee in any such Indemnified Liability; or (iii) the named parties to any such Indemnified Liability (including, without limitation, any impleaded parties) include both such Indemnitee and the indemnifying party, and such Indemnitee shall have been advised by counsel that a conflict of interest is likely to exist if the same counsel were to represent such Indemnitee and the indemnifying party (in which case, if such Indemnitee notifies the indemnifying party in writing that it elects to employ separate counsel at the expense of the indemnifying party, then the indemnifying party shall not have the right to assume the defense thereof and such counsel shall be at the expense of the indemnifying party), provided further that in the case of clause (iii) above the indemnifying party shall not be responsible for the reasonable fees and expenses of more than one (1) separate

legal counsel for such Indemnitee. The Indemnitee shall reasonably cooperate with the indemnifying party in connection with any negotiation or defense of any such action or claim by the indemnifying party and shall furnish to the indemnifying party all information reasonably available to the Indemnitee which relates to such Indemnified Liability. The indemnifying party shall keep the Indemnitee reasonably apprised at all times as to the status of the defense or any settlement negotiations with respect thereto. No indemnifying party shall be liable for any settlement of any action, claim or proceeding effected without its prior written consent; provided, however, the indemnifying party shall not unreasonably withhold, delay or condition its consent. No indemnifying party shall, without the prior written consent of the Indemnitee, consent to entry of any judgment or enter into any settlement or other compromise which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnitee of a release from all liability in respect to such Indemnified Liability, and such settlement shall not include any admission as to fault on the part of the Indemnitee. Following indemnification as provided for hereunder, the indemnifying party shall be subrogated to all rights of the Indemnitee with respect to all third parties, firms or corporations relating to the matter for which indemnification has been made. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action shall not relieve such indemnifying party of any liability to the Indemnitee under this Section 9(k), except to the extent that the indemnifying party is materially and adversely prejudiced in its ability to defend such action. The indemnification required by this Section 9(k) shall be made by periodic payments of the amount thereof during the course of the investigation or defense, as and when bills are received or Indemnified Liabilities are incurred. The indemnity and contribution agreements contained herein shall be in addition to (i) any cause of action or similar right of the Indemnitees against the indemnifying party or others, and (ii) any liabilities the indemnifying party may be subject to pursuant to the law.

- (l) Construction. The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent, and no rules of strict construction will be applied against any party. No specific representation or warranty shall limit the generality or applicability of a more general representation or warranty. Each and every reference to share prices, shares of Common Stock and any other numbers in this Agreement that relate to the Common Stock shall be automatically adjusted for any stock splits, stock dividends, stock combinations, recapitalizations or other similar transactions that occur with respect to the Common Stock after the date of this Agreement. Notwithstanding anything in this Agreement to the contrary, for the avoidance of doubt, nothing contained herein shall constitute a representation or warranty against, or a prohibition of, any actions with respect to the borrowing of, arrangement to borrow, identification of the availability of, and/or securing of, securities of the Company in order for such Buyer (or its broker or other financial representative) to effect short sales or similar transactions in the future.
- (m) Remedies. Each Buyer and in the event of assignment by Buyer of its rights and obligations hereunder, each holder of Securities, shall have all rights and remedies set forth in the Transaction Documents and all rights and remedies which such holders have been granted at any time under any other agreement or contract and all of the rights which such holders have under any law. Any Person having any rights under any provision of this Agreement shall be entitled to enforce such rights specifically (without posting a bond or other security), to recover damages by reason of any breach of any provision of this Agreement and to exercise all other rights granted by law. Furthermore, the Company recognizes that in the event that it or any Subsidiary fails to perform, observe, or discharge any or all of its or such Subsidiary's (as the case may be) obligations under the Transaction Documents, any remedy at law would be inadequate relief to the Buyers. The Company therefore agrees that the Buyers shall be entitled to specific performance and/or temporary, preliminary and permanent injunctive or other equitable relief from any court of competent jurisdiction in any such case without the necessity of proving actual

damages and without posting a bond or other security. The remedies provided in this Agreement and the other Transaction Documents shall be cumulative and in addition to all other remedies available under this Agreement and the other Transaction Documents, at law or in equity (including a decree of specific performance and/or other injunctive relief).

- (n) Withdrawal Right. Notwithstanding anything to the contrary contained in (and without limiting any similar provisions of) the Transaction Documents, whenever any Buyer exercises a right, election, demand or option under a Transaction Document and the Company or any Subsidiary does not timely perform its related obligations within the periods therein provided or if no period is prescribed, within a reasonable period of time, then such Buyer may rescind or withdraw, in its sole discretion from time to time upon written notice to the Company or such Subsidiary (as the case may be), any relevant notice, demand or election in whole or in part without prejudice to its future actions and rights.
- (o) Payment Set Aside; Currency. To the extent that the Company makes a payment or payments to any Buyer hereunder or pursuant to any of the other Transaction Documents or any of the Buyers enforce or exercise their rights hereunder or thereunder, and such payment or payments or the proceeds of such enforcement or exercise or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside, recovered from, disgorged by or are required to be refunded, repaid or otherwise restored to the Company, a trustee, receiver or any other Person under any law (including, without limitation, any bankruptcy law, foreign, state or federal law, common law or equitable cause of action), then to the extent of any such restoration the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or setoff had not occurred. Unless otherwise expressly indicated, all dollar amounts referred to in this Agreement and the other Transaction Documents are in United States Dollars ("U.S. Dollars"), and all amounts owing under this Agreement and all other Transaction Documents shall be paid in U.S. Dollars. All amounts denominated in other currencies (if any) shall be converted into the U.S. Dollar equivalent amount in accordance with the Exchange Rate on the date of calculation. "Exchange Rate" means, in relation to any amount of currency to be converted into U.S. Dollars pursuant to this Agreement, the U.S. Dollar exchange rate as published in the Wall Street Journal on the relevant date of calculation.

## (p) <u>Judgment Currency</u>.

- (i) If for the purpose of obtaining or enforcing judgment against the Company in connection with this Agreement or any other Transaction Document in any court in any jurisdiction it becomes necessary to convert into any other currency (such other currency being hereinafter in this Section 9(p) referred to as the "**Judgment Currency**") an amount due in U.S. Dollars under this Agreement, the conversion shall be made at the Exchange Rate prevailing on the Business Day immediately preceding:
  - (1) the date actual payment of the amount due, in the case of any proceeding in the Court of Chancery of the State of Delaware or in the courts of any other jurisdiction that will give effect to such conversion being made on such date; or
  - (2) the date on which the foreign court determines, in the case of any proceeding in the courts of any other jurisdiction (the date as of which such conversion is made pursuant to this Section 9(p)(i)(2) being hereinafter referred to as the "Judgment Conversion Date").
- (ii) If in the case of any proceeding in the court of any jurisdiction referred to in Section 9(p)(i)(2), there is a change in the Exchange Rate prevailing between the Judgment Conversion Date and the date of actual payment of the amount due, the

applicable party shall pay such adjusted amount as may be necessary to ensure that the amount paid in the Judgment Currency, when converted at the Exchange Rate prevailing on the date of payment, will produce the amount of U.S. Dollars which could have been purchased with the amount of Judgment Currency stipulated in the judgment or judicial order at the Exchange Rate prevailing on the Judgment Conversion Date.

- (iii) Any amount due from the Company under this provision shall be due as a separate debt and shall not be affected by judgment being obtained for any other amounts due under or in respect of this Agreement or any other Transaction Document.
- Independent Nature of Buyers' Obligations and Rights. The obligations of each Buyer under the Transaction Documents are several and not joint with the obligations of any other Buyer, and no Buyer shall be responsible in any way for the performance of the obligations of any other Buyer under any Transaction Document. Nothing contained herein or in any other Transaction Document, and no action taken by any Buyer pursuant hereto or thereto, shall be deemed to constitute the Buyers as, and the Company acknowledges that the Buyers do not so constitute, a partnership, an association, a joint venture or any other kind of group or entity, or create a presumption that the Buyers are in any way acting in concert or as a group or entity, and the Company shall not assert any such claim with respect to such obligations or the transactions contemplated by the Transaction Documents or any matters, and the Company acknowledges that the Buyers are not acting in concert or as a group, and the Company shall not assert any such claim, with respect to such obligations or the transactions contemplated by the Transaction Documents. The decision of each Buyer to purchase Securities pursuant to the Transaction Documents has been made by such Buyer independently of any other Buyer. Each Buyer acknowledges that no other Buyer has acted as agent for such Buyer in connection with such Buyer making its investment hereunder and that no other Buyer will be acting as agent of such Buyer in connection with monitoring such Buyer's investment in the Securities or enforcing its rights under the Transaction Documents. The Company and each Buyer confirms that each Buyer has independently participated with the Company and its Subsidiaries in the negotiation of the transaction contemplated hereby with the advice of its own counsel and advisors. Each Buyer shall be entitled to independently protect and enforce its rights, including, without limitation, the rights arising out of this Agreement or out of any other Transaction Documents, and it shall not be necessary for any other Buyer to be joined as an additional party in any proceeding for such purpose. The use of a single agreement to effectuate the purchase and sale of the Securities contemplated hereby was solely in the control of the Company, not the action or decision of any Buyer, and was done solely for the convenience of the Company and its Subsidiaries and not because it was required or requested to do so by any Buyer. It is expressly understood and agreed that each provision contained in this Agreement and in each other Transaction Document is between the Company, each Subsidiary and a Buyer, solely, and not between the Company, its Subsidiaries and the Buyers collectively and not between and among the Buyers.
- (r) <u>Performance Date</u>. If the date by which any obligation under any of the Transaction Documents must be performed occurs on a day other than a Business Day, then the date by which such performance is required shall be the next Business Day following such date.
- (s) <u>Enforcement Fees</u>. The Company agrees to pay all costs and expenses of the Buyers incurred as a result of enforcement of the Transaction Documents and the collection of any amounts owed to the Buyers hereunder (whether in cash, equity or otherwise), including, without limitation, reasonable attorneys' fees and expenses.

### (t) <u>Collateral Agent</u>.

(i) Appointment; Authorization. The Buyers, together with any successors or assigns thereof, hereby irrevocably appoint, designate and authorize High Trail Investments ON

LLC as collateral agent to take such action on their behalf under the provisions of the Notes, each Security Agreement and to exercise such powers and perform such duties as are expressly delegated to it by the terms of each Security Agreement, together with such powers as are reasonably incidental thereto. The provisions of this Section 9(t) are solely for the benefit of the Collateral Agent, and the Company shall not have rights as a third-party beneficiary of any of such provisions. It is understood and agreed that the use of the term "agent" herein or in any Security Agreement (or any other similar term) with reference to the Collateral Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead such term is used as a matter of market custom, and is intended to create or reflect only an administrative relationship between contracting parties. Notwithstanding any provision to the contrary contained elsewhere in the Notes, any Security Agreement or any other agreement, instrument or document related hereto or thereto, the Collateral Agent shall not have any duty or responsibility except those expressly set forth herein, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into the Notes, any Security Agreement or any other agreement, instrument or document related hereto or thereto or otherwise exist against the Collateral Agent.

(ii) Delegation of Duties. The Collateral Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any Security Agreement by or through any one or more sub-agents appointed by the Collateral Agent. The Collateral Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through its Affiliates (as defined in the Notes), partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives, or the partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives of any of its Affiliates (collectively, the "Related Parties"). The exculpatory provisions of this Section 9(t) shall apply to any such sub-agent and to the Related Parties of the Collateral Agent and any such sub-agent. The Collateral Agent shall not be responsible for the negligence or misconduct of any sub-agents except to the extent that a court of competent jurisdiction determines in a final and non-appealable judgment that the Collateral Agent acted with gross negligence or willful misconduct in the selection of such sub-agents.

### (iii) Exculpatory Provisions.

- (A) The Collateral Agent shall not have any duties or obligations except those expressly set forth in the Security Agreements, and its duties shall be administrative in nature. Without limiting the generality of the foregoing, the Collateral Agent: (i) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing or an Event of Default has occurred; (ii) shall not have any duty to take any discretionary action or exercise any discretionary powers; and (iii) shall not, except as expressly set forth in the Security Agreements, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Company or any of its Affiliates that is communicated to or obtained by the Collateral Agent or any of its Affiliates in any capacity.
- (B) The Collateral Agent shall not be liable for any action taken or not taken by it in the absence of its own gross negligence or willful misconduct as determined by a court of competent jurisdiction by final and non-appealable judgment. The Collateral Agent shall be deemed not to have knowledge of any Default or Event of Default unless and until notice describing such Default or Event of Default is given to the Collateral Agent in writing by the Company.
- (C) The Collateral Agent shall not be responsible for or have any duty to ascertain or inquire into (a) any statement, warranty or representation made in or in connection with the Notes, any Security Agreement or any other agreement, instrument or document related hereto or thereto, (b) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (c) the performance or

observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default or Event of Default, (d) the validity, enforceability, effectiveness or genuineness of the Notes, any Security Agreement or any other agreement, instrument or document related to the Notes or Security Agreements, or (e) any failure of the Company or any other party to the Notes, any Security Agreements or any other agreement, instrument or document related to the Notes or Security Agreements to perform its obligations thereunder. The Collateral Agent shall not be under any obligation to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, the Notes, any Security Agreement or any other agreement, instrument or document related to the Notes or Security Agreements, or to inspect the properties, books or records of the Company or any Affiliate of the Company.

- (iv) Reliance by Collateral Agent. The Collateral Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Collateral Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. The Collateral Agent may consult with legal counsel, independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.
- (v) Successor Agent. The Collateral Agent may resign as the Collateral Agent at any time upon ten (10) days' prior notice to the Buyers and the Company. If the Collateral Agent resigns under the Notes, the Required Holders shall appoint a successor agent. If no successor agent is appointed prior to the effective date of the resignation of the Collateral Agent, the Collateral Agent may appoint a successor Collateral Agent on behalf of the Buyers after consulting with the Buyers. Upon the acceptance of its appointment as successor agent hereunder, such successor agent shall succeed to all the rights, powers and duties of the retiring Collateral Agent and the term "the Collateral Agent" shall mean such successor agent, and the retiring Collateral Agent's appointment, powers and duties as the Collateral Agent shall be terminated. After the Collateral Agent's resignation hereunder as the Collateral Agent, the provisions of this Section 9(t) shall continue to inure to its benefit as to any actions taken or omitted to be taken by it while it was the Collateral Agent. If no successor agent has accepted appointment as the Collateral Agent by the date which is thirty (30) days following a retiring Collateral Agent's notice of resignation, a retiring Collateral Agent hereunder until such time as Required Holders shall appoint a successor agent as provided for above.
- (vi) Non-Reliance on the Collateral Agent. The Buyers acknowledges that they have, independently and without reliance upon the Collateral Agent or any of its Related Parties and based on such documents and information as they have deemed appropriate, made their own credit analysis and decision to enter invest in the Notes. The Buyers also acknowledges that they will, independently and without reliance upon the Collateral Agent or any of its Related Parties and based on such documents and information as they shall from time to time deem appropriate, continue to make their own decisions in taking or not taking action under or based upon the Notes, any Security Agreement or any related agreement or any document furnished hereunder or thereunder.
- (vii) Collateral Matters. The Buyers irrevocably authorize the Collateral Agent to release any Lien (as defined in the Notes) granted to or held by the Collateral Agent

under any Security Agreement (i) when all Obligations (as defined below) have been paid in full; (ii) constituting property sold or to be sold or disposed of as part of or in connection with any sale or other disposition permitted under the Notes and each other agreement, instrument or document related thereto (it being agreed and understood that the Collateral Agent may conclusively rely without further inquiry on a certificate of an officer of the Company as to the sale or other disposition of property being made in compliance with the Notes and each other agreement, instrument or document related thereto); or (iii) if approved, authorized or ratified in writing by the Buyers. The Collateral Agent shall have the right, in accordance with the Security Agreements to sell, lease or otherwise dispose of any Collateral for cash, credit or any combination thereof, and the Collateral Agent may purchase any Collateral at public or, if permitted by law, private sale and, in lieu of actual payment of the purchase price, may credit bid and setoff the amount of such price against the Obligations. "Obligation" means all liabilities, indebtedness and obligations (including interest accrued at the rate provided in the applicable Note Document (as defined below) after the commencement of a bankruptcy proceeding, whether or not a claim for such interest is allowed) of the Company, any Security Agreements or any other Note Document, in each case howsoever created, arising or evidenced, whether direct or indirect, absolute or contingent, now or hereafter existing, or due or to become due. "Note Documents" means this Agreement, the Notes, the Security Agreements and the other Transaction Documents.

- (viii) Reimbursement by Buyers. To the extent that the Company for any reason fails to indefeasibly pay any amount required under Sections 4(g) or 9(k) to be paid by it to the Collateral Agent (or any sub-agent thereof) or any Related Party of the Collateral Agent (or any sub-agent thereof), the Buyers hereby agree, jointly and severally, to pay to the Collateral Agent (or any sub-agent) or such Related Party of the Collateral Agent (or any sub-agent thereof), as the case may be, such unpaid amount.
- (ix) Marshaling; Payments Set Aside. Neither the Collateral Agent nor the Buyers shall be under any obligation to marshal any assets in favor of the Company or any other Person or against or in payment of any or all of the Obligations. To the extent that the Company makes a payment or payments to the Collateral Agent, or the Collateral Agent enforces its Liens (as defined in the Notes) or exercises its rights of set-off, and such payment or payments or the proceeds of such enforcement or set-off or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Collateral Agent in its discretion) to be repaid to a trustee, receiver or any other party in connection with any bankruptcy, insolvency or similar proceeding, or otherwise, then (i) to the extent of such recovery, the obligation under the Notes intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or set-off had not occurred and (ii) the Buyers agree to pay to the Collateral Agent upon demand its share of the total amount so recovered from or repaid by the Collateral Agent to the extent paid to the Buyers.
- (u) Placement Agent. The Buyers acknowledge and agree that (i) Placement Agent is acting solely as placement agent in connection with the issue and sale of the Notes and is not acting as an underwriter or in any other capacity and is not and shall not be construed as a fiduciary in connection with the issue and sale of the Notes, (ii) the Placement Agent has not made any representation or warranty, whether express or implied, and has not provided any advice or recommendation in connection with the issue and sale of the Notes, and the Buyers have not relied upon any information provided by the Placement Agent, and (iii) the Placement Agent shall have no liability or obligation hereunder in respect of the issue and sale of the Notes. The parties agree that the Placement Agent shall be a third party beneficiary of Section 2 hereof.

**IN WITNESS WHEREOF**, each Buyer and the Company have caused their respective signature page to this Agreement to be duly executed as of the date first written above.

**COMPANY:** 

VELO3D, INC.

By: /s/ Benyamin Buller

Name: Benyamin Buller
Title: Chief Executive Officer

**IN WITNESS WHEREOF,** each Buyer and the Company have caused their respective signature page to this Agreement to be duly executed as of the date first written above.

## **BUYER:**

# HIGH TRAIL INVESTMENTS ON LLC

By: /s/[\*]\*
Name: [\*]

Title: Authorized Signatory

\*Authorized Signatory [\*] Capital Management LP Not individually, but solely as Investment Advisor to High Trail Investments ON LLC. **IN WITNESS WHEREOF,** each Buyer and the Company have caused their respective signature page to this Agreement to be duly executed as of the date first written above.

# **BUYER:**

# HB SPV I MASTER SUB LLC

By: /s/[\*]\*
Name: [\*]
Title: Authorized Signatory

\*Authorized Signatory [\*] Capital Management LP Not individually, but solely as Investment Advisor to HB SPV I Master Sub LLC.

# SCHEDULE OF BUYERS

(1)	(2)	(3)	(4)	(5)	(6)	(8)
Buyer	Address	Aggregate Principal Amount of Initial Purchased Notes	Aggregate Purchase Price of Initial Purchased Notes	Maximum Aggregate Principal Amount of Subsequently Purchased Notes	Maximum Aggregate Purchase Price of Subsequently Purchased Notes	Legal Representative's Address
High Trail Investments ON LL	High Trail Capital C 80 River Street, Suite 4C Hoboken, NJ 07030 Attn: Eric Helenek E-Mail: [*]	\$28,000,000	\$28,000,000	\$14,000,000	\$14,000,000	Latham & Watkins LLP 12670 High Bluff Drive San Diego, CA 92130 Telephone: [*] Attention: Michael E. Sullivan
HB SPV I Master Sub LLC	High Trail Capital 80 River Street, Suite 4C Hoboken, NJ 07030 Attn: Eric Helenek E-Mail: [*]	\$42,000,000	\$42,000,000	\$21,000,000	\$21,000,000	Latham & Watkins LLP 12670 High Bluff Drive San Diego, CA 92130 Telephone: [*] Attention: Michael E. Sullivan
TOTAL		\$70,000,000	\$70,000,000	\$35,000,000	\$35,000,000	

# SCHEDULE OF LOCK-UP PARTIES

Carl Bass Benyamin Buller William McCombe Michael Idelchik Adrian Keppler Stefan Krause Ellen Smith Gabrielle Toledano Matthew Walters Renette Youssef Bernard Chung

# Exhibit A Form of Senior Secured Convertible Note

# Exhibit B-1

# Form of Base Indenture

# Exhibit B-2 Form of Supplemental Indenture

# Exhibit C Form of Perfection Certificate

# Exhibit D

# Form of Lock-Up and Voting Agreement

[See Attached]

#### **VOTING AGREEMENT**

This **VOTING AGREEMENT** (this "**Agreement**"), dated as of August 10, 2023, is entered into by and between the undersigned stockholder (the "**Stockholder**") of Velo3D, Inc., a Delaware corporation (the "**Company**"), and the Company. The Company and the Stockholder are each sometimes referred to herein individually as a "**Party**" and collectively as the "**Parties.**" Capitalized terms that are used but not defined herein shall have the meaning ascribed to them in the Purchase Agreement (as defined below).

#### RECITALS

Concurrently with or following the execution of this Agreement, the Company has entered, or will enter, into a Securities Purchase Agreement (as the same may be amended from time to time, the "Purchase Agreement"), providing for, among other things, the sale to the Buyers of the Purchased Securities pursuant to the terms and conditions of the Purchase Agreement and the Transaction Documents.

In order to induce the Buyers to enter into the Purchase Agreement and the Transaction Documents, and for good and valuable consideration, the sufficiency of which is hereby acknowledged, the Stockholder hereby makes certain representations, warranties, covenants, and agreements as set forth in this Agreement with respect to the shares of Common Stock Beneficially Owned (as defined below) by the Stockholder and set forth below the Stockholder's signature on the signature page hereto (the "Original Shares" and, together with any additional shares of Common Stock pursuant to Section 6 hereof, the "Shares").

#### **AGREEMENT**

NOW, THEREFORE, in consideration of the premises and the mutual covenants contained herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

#### 10. <u>Definitions</u>.

When used in this Agreement, the following terms in all of their tenses, cases, and correlative forms shall have the meanings assigned to them in this Section 1.

- (a) "Beneficially Own" or "Beneficial Ownership" has the meaning assigned to such term in Rule 13d-3 under the 1934 Act, and a Person's beneficial ownership of securities shall be calculated in accordance with the provisions of such rule (in each case, irrespective of whether or not such rule is actually applicable in such circumstance). For the avoidance of doubt, "Beneficially Own" and "Beneficial Ownership" shall also include record ownership of securities.
  - (b) "Beneficial Owner" shall mean the Person who Beneficially Owns the referenced securities.

#### 11. Representations of Stockholder.

The Stockholder represents and warrants to the Company that:

(a) Ownership of Shares. The Stockholder: (i) is the Beneficial Owner of all of the Original Shares set forth below the Stockholder's signature on the signature pages hereto free and clear of any proxy, voting restriction, adverse claim, or other Liens, other than those created by this Agreement or under applicable federal or state securities laws; and (ii) has the sole voting power over all such Original Shares. Except pursuant to this Agreement, there are no options, warrants, or other rights, agreements, arrangements, or commitments of any character to which the Stockholder is a party relating to the pledge,

disposition, or voting of any such Original Shares and there are no voting trusts or voting agreements with respect to such Original Shares.

- (b) **Disclosure of All Shares Owned**. The Stockholder does not Beneficially Own any shares of Common Stock other than: (i) the Original Shares set forth below the Stockholder's signature on the signature pages hereto; and (ii) any options, warrants, or other rights to acquire any additional shares of Common Stock or any security exercisable for or convertible into shares of Common Stock, set forth below the Stockholder's signature on the signature pages hereto (collectively, "**Options**").
- (c) **Power and Authority; Binding Agreement.** If the Stockholder is an individual, the Stockholder has full power and authority and legal capacity to enter into, execute, and deliver this Agreement and to perform fully the Stockholder's obligations hereunder (including the proxy described in Section 3(b) below). If the Stockholder is not an individual, the Stockholder has requisite organizational power and authority to enter into, execute, and deliver this Agreement and to perform fully the Stockholder's obligations hereunder (including the proxy described in Section 3(b) below). This Agreement has been duly and validly executed and delivered by the Stockholder and constitutes the legal, valid, and binding obligation of the Stockholder, enforceable against the Stockholder in accordance with its terms, except as such enforceability may be limited by general principles of equity or applicable bankruptcy, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally, the enforcement of applicable creditors' rights.
- (d) No Conflict. The execution and delivery of this Agreement by the Stockholder does not, and the consummation of the transactions contemplated hereby and the compliance with the provisions hereof will not, conflict with or violate any law applicable to the Stockholder or result in any breach of or violation of, or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration, or cancellation of, or result in the creation of any Lien on any of the Shares attributable to the Stockholder pursuant to, any agreement or other instrument or obligation binding upon the Stockholder or any of the Shares attributable to the Stockholder.
- (e) No Consents. No consent, approval, order, or authorization of, or registration, declaration, or filing with, any Governmental Entity or any other Person on the part of the Stockholder is required in connection with the valid execution and delivery of this Agreement. If the Stockholder is an individual, no consent of the Stockholder's spouse is necessary under any "community property" or other laws in order for the Stockholder to enter into and perform its obligations under this Agreement.
- (f) **No Litigation.** There is no action, suit, investigation, or proceeding (whether judicial, arbitral, administrative, or other) pending against, or, to the knowledge of the Stockholder, threatened against or affecting, the Stockholder that could reasonably be expected to materially impair or materially adversely affect the ability of the Stockholder to perform the Stockholder's obligations hereunder or to consummate the transactions contemplated by this Agreement on a timely basis.

### 12. Agreement to Vote Shares; Irrevocable Proxy.

(a) Agreement to Vote and Approve. The Stockholder irrevocably and unconditionally agrees during the term of this Agreement, at any annual or special meeting of the Company called with respect to the approval contemplated by the Requisite Stockholder Approval (as defined in the Convertible Notes), and at every adjournment or postponement thereof, and on every action or approval by written consent or consents of the Company stockholders with respect to such matter, to vote or cause the holder of record to vote the Shares in favor of providing the Requisite Stockholder Approval. "Convertible Notes" means the Senior Secured Convertible Notes issued pursuant to the Purchase Agreement, as amended from time to time.

(b) Irrevocable Proxy. The Stockholder hereby appoints the Company and any designee of the Company, and each of them individually, until the Expiration Time (as defined below) (at which time this proxy shall automatically be revoked), its proxies and attorneys-in-fact, with full power of substitution and resubstitution, to vote or act by written consent during the term of this Agreement with respect to the Shares in accordance with Section 3(a). This proxy and power of attorney is given to secure the performance of the duties of the Stockholder under this Agreement. The Stockholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy. This proxy and power of attorney granted by the Stockholder shall be irrevocable during the term of this Agreement, shall be deemed to be coupled with an interest sufficient in law to support an irrevocable proxy, and shall revoke any and all prior proxies granted by the Stockholder with respect to the Shares. The power of attorney granted by the Stockholder herein is a durable power of attorney and shall survive the bankruptcy, death, or incapacity of the Stockholder. The proxy and power of attorney granted hereunder shall terminate upon the termination of this Agreement.

#### 13. No Voting Trusts or Other Arrangement.

The Stockholder agrees that during the term of this Agreement the Stockholder will not, and will not permit any entity under the Stockholder's control to, deposit any of the Shares in a voting trust, grant any proxies with respect to the Shares, or subject any of the Shares to any arrangement with respect to the voting of the Shares other than agreements entered into with the Company.

#### 14. <u>Transfer and Encumbrance</u>.

- (a) Subject to Section 5(a), the Stockholder agrees that for a period commencing upon the execution of the Purchase Agreement and ending on the date that is ninety (90) days after the date of the Purchase Agreement (the "Lock-Up Period"), the Stockholder will not, (1) directly or indirectly, offer, pledge, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant to purchase or otherwise transfer or dispose of any shares of Common Stock or any securities convertible into or exercisable or exchangeable for Common Stock ("Lock-Up Securities") (including without limitation, Lock-Up Securities which may be deemed to be Beneficially Owned and securities which may be issued upon exercise of a stock option or warrant) or file any registration statement under the 1933 Act with respect to any of the foregoing or publicly disclose the intention to undertake any of the foregoing or (2) enter into any swap or other transaction that transfers, in whole or in part, directly or indirectly, any of the economic consequences of ownership of the Lock-Up Securities, whether any such swap or transaction described in clause (1) or (2) above is to be settled by delivery of Lock-Up Securities, in cash or otherwise (collectively a "Transfer"). Any attempted Transfer of Shares or any interest therein in violation of this Section 5 shall be null and void. The Stockholder acknowledges and agrees that the foregoing precludes the Stockholder from engaging in any hedging or other transactions designed or intended, or which could reasonably be expected to lead to or result in, a sale or disposition of any Lock-Up Securities, even if any such sale or disposition transaction or transactions would be made or executed by or on behalf of someone other than the Stockholder.
- (b) The Stockholder may transfer Lock-Up Securities in connection with the vesting, conversion, settlement, exchange or exercise of restricted stock units, options, warrants or other rights to purchase or receive securities (including, in each case, a transfer to the Company by way of "net" or "cashless" exercise or a sale in the market to cover the payment of tax withholdings or remittance payments due in connection with such vesting, conversion, settlement, exchange or exercise), or in connection with the exercise or redemption of warrants or the conversion or redemption of convertible securities, in all such cases pursuant to equity awards or rights granted under a stock incentive plan, other equity award plan or stock purchase plan, or pursuant to the terms of warrants or convertible securities, provided that any Lock-Up Securities received upon such vesting, conversion, settlement, exchange, exercise or redemption, and after giving effect to such transfers to the Company or sale to cover transactions, shall be subject to the terms of this Agreement. A Transfer of Lock-Up Securities may be made to (i) a family member or trust or, if the Stockholder is a trust, to a trustor or beneficiary of the trust or the estate of a beneficiary of such trust, (ii) as one or more bona fide gifts or charitable contributions, or for bona fide estate planning, (iii) upon death by will, testamentary document or intestate succession, (iv) to a partnership, limited liability company or other entity of which the Stockholder or the family

members of the Stockholder are the legal and beneficial owners of all of the outstanding equity securities or similar interests or, if the Stockholder is such a partnership, limited liability company or other entity, as part of a distribution by the Stockholder to its partners, members or other equity holders or to the estate of any such partners, members or other equity holders, (v) to a nominee or custodian of a person or entity to whom a disposition or transfer would be permissible under clauses (i) through (iv) above, (vi) by operation of law, such as pursuant to a qualified domestic order, divorce settlement, divorce decree or separation agreement or (vii) to the Company from an employee of the Company upon death, disability or termination of employment, in each case, of such employee; provided that (A) in the case of clauses (i), (ii), (iii), (iv) and (v) above, the transferee agrees to be bound in writing by the terms of this Agreement prior to such Transfer, (B) such Transfer shall not involve a disposition for value and (C) no filing or public announcement by any party (donor, donee, transferor or transferee) under the 1934 Act shall be voluntarily made in connection with such Transfer and, if a filing or public shall be legally required under the 1934 Act during the Lock-Up Period, such filing or public announcement shall clearly indicate in the footnotes thereto the circumstances of such Transfer. For purposes of this Agreement, a "family member" shall mean any relationship by blood, marriage, domestic partnership or adoption, not more remote than first cousin.

A Transfer of Lock-Up Securities may also be made pursuant to a written trading plan that complied with Rule 10b5-1 under the 1934 Act as in effect at the time such plan was adopted (a "10b5-1 Trading Plan") and that was existing as of August 1, 2023 and is existing as of the date of this Agreement; provided that to the extent a public announcement or filing under the 1934 Act is required of the Stockholder or the Company regarding the Transfer, such announcement or filing shall include a statement to the effect that the Transfer occurred pursuant to such 10b5-1 Trading Plan. A Transfer of Lock-Up Securities may also be made pursuant to a *bona fide* third-party tender offer, merger, consolidation or other similar transaction that is approved the by the Board of Directors of the Company and made to all holders of the Company capital stock involving a Change of Control of the Company (for purposes hereof, "Change of Control" shall mean the transfer (whether by tender offer, merger, consolidation or other similar transaction), in one transaction or a series of related transactions, to a person or group of affiliated persons, of shares of capital stock if, after such transfer, such person or group of affiliated persons would hold at least a majority of the outstanding voting securities of the Company (or the surviving entity).

Notwithstanding anything in this Section 5 to the contrary, the Stockholder may, during the Lock-Up Period, enter into a written 10b5-1 Trading Plan that complies with Rule 10b5-1 under the 1934 Act as in effect at the time such plan is adopted; *provided* that no direct or indirect offers, pledges, sales, contracts to sell, sales of any option or contract to purchase, purchases of any option or contract to sell, grants of any option, right or warrant to purchase, loans, or other transfers or disposals of any Lock-Up Securities may be effected pursuant to such plan during the Lock-Up Period; and *provided* that no public announcement or filing under the 1934 Act regarding the establishment of such plan shall be voluntarily made by or on behalf of the Stockholder or the Company during the Lock-Up Period.

#### 15. Additional Shares.

The Stockholder agrees that all shares of Common Stock that the Stockholder purchases, acquires the right to vote, or otherwise acquires Beneficial Ownership of, after the execution of this Agreement and prior to the Expiration Time shall be subject to the terms and conditions of this Agreement and shall constitute Shares for all purposes of this Agreement. In the event of any stock split, stock dividend, merger, reorganization, recapitalization, reclassification, combination, exchange of shares, or the like of the capital stock of the Company affecting the Shares, the terms of this Agreement shall apply to the resulting securities and such resulting securities shall be deemed to be "Shares" for all purposes of this Agreement.

#### 16. Termination.

This Agreement shall terminate upon the earliest to occur of (the "Expiration Time"): (a) the date on which the Purchase Agreement is terminated in accordance with its terms; (b) the termination of this Agreement by mutual written consent of the Parties; and (c) the date on which the Requisite

Stockholder Approval is obtained. Nothing in this <u>Section 7</u> shall relieve or otherwise limit the liability of any Party for any intentional breach of this Agreement prior to such termination.

### 17. No Agreement as Director or Officer.

The Stockholder makes no agreement or understanding in this Agreement in its capacity as a director or officer of the Company or any of its subsidiaries (if the Stockholder holds such office), and nothing in this Agreement: (a) will limit or affect any actions or omissions taken by the Stockholder in its capacity as such a director or officer, and no such actions or omissions shall be deemed a breach of this Agreement; or (b) will be construed to prohibit, limit, or restrict the Stockholder from exercising its fiduciary duties as an officer or director to the Company or its stockholders.

#### 18. <u>Further Assurances</u>.

The Stockholder agrees, from time to time, and without additional consideration, to execute and deliver such additional proxies, documents, and other instruments and to take all such further action as the Company may reasonably request to consummate and make effective the transactions contemplated by this Agreement.

### 19. Stop Transfer Instructions.

At all times commencing with the execution and delivery of this Agreement and continuing until the Expiration Time, in furtherance of this Agreement, the Stockholder hereby authorizes the Company or its counsel to notify the Company's transfer agent that there is a stop transfer order with respect to all of the Shares (and that this Agreement places limits on the voting and transfer of the Shares), subject to the provisions hereof and provided that any such stop transfer order and notice will immediately be withdrawn and terminated by the Company following the Expiration Time.

#### 20. Specific Performance.

Each Party hereto acknowledges that it will be impossible to measure in money the damage to the other Party if a Party hereto fails to comply with any of the obligations imposed by this Agreement, that every such obligation is material and that, in the event of any such failure, the other Party will not have an adequate remedy at law or damages. Accordingly, each Party hereto agrees that injunctive relief or other equitable remedy, in addition to remedies at law or damages, is the appropriate remedy for any such failure and will not oppose the seeking of such relief on the basis that the other Party has an adequate remedy at law. Each Party hereto agrees that it will not seek, and agrees to waive any requirement for, the securing or posting of a bond in connection with the other Party's seeking or obtaining such equitable relief.

#### 21. Amendment; Assignment.

No provision of this Agreement may be amended other than by an instrument in writing signed by the Company and the Stockholder. No Party to this Agreement may assign any of its rights or obligations under this Agreement without the prior written consent of the other Party hereto, except that the Company may assign, in its sole discretion, all or any of its rights, interests and obligations hereunder to any of its "affiliates" (as defined in Rule 144 promulgated under the 1933 Act (or a successor rule thereto)) or to any party that acquires all of substantially all of the assets of the Company (whether by merger, sale of stock, sale of assets or otherwise). Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of and be enforceable by the Parties and their respective permitted successors and assigns. Any assignment contrary to the provisions of this Section 12 shall be null and void.

## 22. Notices.

Any notices, consents, waivers or other communications required or permitted to be given under the terms of this Agreement must be in writing and will be deemed to have been delivered: (i) upon receipt, when delivered personally; (ii) upon receipt, when sent by electronic mail (provided that such

sent email is kept on file (whether electronically or otherwise) by the sending party and the sending party does not receive an automatically generated message from the recipient's email server that such e-mail could not be delivered to such recipient); or (iii) one (1) Business Day after deposit with an overnight courier service with next day delivery specified, in each case, properly addressed to the party to receive the same. The addresses and e-mail addresses for such communications shall be:

If to the Company:

Velo3D, Inc. 511 Division St. Campbell, CA 95008 Telephone: [\*] Attention: Bill McCombe E-mail: [\*]

With a copy (for informational purposes only) to:

Fenwick & West LLP 902 Broadway New York, NY 10010 Telephone: [\*] Attention: Per Chilstrom Email: [\*]

If to the Stockholder, to the address, email address, or facsimile number set forth for the Stockholder on the signature pages hereof.

#### 23. Miscellaneous.

The provisions of Sections 9(a), (b), (c), (d), (e), (g), (h), (l), (n), (o) and (p) of the Purchase Agreement shall apply to this Agreement, mutatis mutandis.

[Remainder of Page Intentionally Left Blank]

VELO3D, INC.
By: Name: Title:
[STOCKHOLDER]
By: Name: Number of Shares of Common Stock Beneficially Owned as of the date of this Agreement Number of Options Beneficially Owned as of the date of this Agreement: Street Address: City/State/Zip Code: Email:

IN WITNESS WHEREOF, the Parties hereto have executed and delivered this Agreement as of the date first written above.

Up to \$105,000,000

VELO3D, INC.

Senior Secured Convertible Notes

#### PLACEMENT AGENT AGREEMENT

August 10, 2023

Credit Suisse Securities (USA) LLC As Placement Agent c/o Credit Suisse Securities (USA) LLC, Eleven Madison Avenue, New York, N.Y. 10010-3629

Dear Sirs and Madams:

1. Introductory. Velo3D, Inc., a Delaware corporation ("Company"), proposes, pursuant to the terms of this Placement Agent Agreement (this "Agreement") and the securities purchase agreement (the "Securities Purchase Agreement") entered into with the purchasers named therein (collectively, the "Purchaser"), to sell to the Purchaser \$70,000,000 aggregate principal amount of Senior Secured Convertible Notes ("Firm Securities"). The Company also agrees to sell to the Purchaser, at the option of the Purchaser (the "Option"), an aggregate of not more than \$35,000,000 aggregate principal amount of Senior Secured Convertible Notes ("Optional Securities"). The Firm Securities and the Optional Securities are herein collectively called the "Offered Securities". The Offered Securities shall be convertible into shares of the Company's common stock, par value \$0.00001 per share ("Common Stock", and any shares of Common Stock issuable upon the conversion of the Offered Securities, the 'Underlying Shares"). Except as provided in the Offered Securities, the Offered Securities will be secured by a first priority lien on all assets of the Company including, but not limited to, its intellectual property (subject to prior liens and other customary exclusions, in each case acceptable to the Collateral Agent (as defined in the Offered Securities) in its sole discretion) (the "Collateral") in accordance with the Indenture (as defined below), the Offered Securities and the Security Documents (as defined below).

2. Representations and Warranties of the Company. The Company represents and warrants to, and agrees with, the Placement Agent that:

(a) Filing and Effectiveness of Registration Statement; Certain Defined Terms. The Company has filed with the Commission a registration statement on Form S-3 (No. 333-268346) covering the registration of the Offered Securities under the Act, including a related preliminary prospectus or prospectuses. At any particular time, this initial registration statement, in the form then on file with the Commission, including all material then incorporated by reference therein, all information contained in the registration statement (if any) pursuant to Rule 462(b) and then deemed to be a part of the initial registration statement, and all 430B Information and 430C Information, that in any case has not then been superseded or modified, shall be referred to as the "Initial Registration Statement". The Company may also have filed, or may file with the Commission, a Rule 462(b) registration statement, in the form then on file with the Commission, including the contents of the Initial Registration Statement incorporated by reference therein and including all 430B Information and all 430C Information, that in any case has not then been superseded or modified, shall be referred to as the "Additional Registration Statement".

As of the time of execution and delivery of this Agreement, the Initial Registration Statement has been declared effective under the Act and is not proposed to be amended. Any Additional Registration Statement has or will become effective upon filing with the Commission pursuant to Rule 462(b) and is not proposed to be amended. The Offered Securities all have been or will be duly registered under the Act pursuant to the Initial Registration Statement and, if applicable, the Additional Registration Statement.

For purposes of this Agreement:

- "430B Information", with respect to any registration statement, means information included in a prospectus then deemed to be a part of such registration statement pursuant to Rule 430B(e) or retroactively deemed to be a part of such registration statement pursuant to Rule 430B(f).
- **"430C Information"**, with respect to any registration statement, means information included in a prospectus then deemed to be a part of such registration statement pursuant to Rule 430C.
  - "Act" means the Securities Act of 1933, as amended.
  - "Applicable Time" means 4:30 pm (Eastern time) on the date of the Securities Purchase Agreement.
  - "Closing Date" means any of the Initial Closing Date or each Optional Closing Date, if any.
  - "Commission" means the Securities and Exchange Commission.
- "Effective Time" with respect to the Initial Registration Statement, means the time of the first contract of sale for the Offered Securities, or, if filed prior to the execution and delivery of the Securities Purchase Agreement, with respect to the Additional Registration Statement means the date and time as of which such Registration Statement was declared effective by the Commission or has become effective upon filing pursuant to Rule 462(c). If an Additional Registration Statement has not been filed prior to the execution and delivery of the Securities Purchase Agreement but the Company has advised the Placement Agent that it proposes to file one, "Effective Time" with respect to such Additional Registration Statement means the date and time as of which such Additional Registration Statement is filed and becomes effective pursuant to Rule 462(b).
  - "Exchange Act" means the Securities Exchange Act of 1934, as amended.
  - "Exchange Act Rules and Regulations" means the rules and regulations of the Commission under the Exchange Act.
  - "Final Closing Date" means the Closing Date on which all of the Offered Securities have been sold.
- "Final Prospectus" means the Statutory Prospectus that discloses the public offering price, other 430B Information and other final terms of the Offered Securities and otherwise satisfies Section 10(a) of the Act.

The Initial Registration Statement and any Additional Registration Statement, after the filing thereof, are referred to collectively as the "Registration Statements" and each is individually referred to as a "Registration Statement". A "Registration Statement" with reference to a particular time means the Initial Registration Statement and any Additional Registration Statement as of such time. A "Registration Statement" without reference to a time means such Registration Statement as of its Effective Time. For purposes of the foregoing definitions, 430B Information with respect to a Registration Statement shall be considered to be included in such Registration Statement as of the time specified in Rule 430B.

- "Indenture" means that certain Indenture, to be dated as of the Initial Closing Date (as defined below), between the Company and the Trustee, as amended and supplemented by the First Supplemental Indenture thereto, to be dated as of the Initial Closing Date, between the Company and the Trustee.
- "Initial Closing Date" means 9:00 a.m., New York City time, on August 14, 2023 or such other time thereafter as the Purchaser and the Company determine for the settlement and payment of the Firm Securities pursuant to the Securities Purchase Agreement.
  - "Optional Closing Date" means each time for the delivery of and payment for the Optional Securities, which may be the Initial Closing Date.
  - "Placement Agent" means Credit Suisse Securities (USA) LLC.
  - "Rules and Regulations" means the rules and regulations of the Commission under the Act.
- "Securities Laws" means, collectively, the Sarbanes-Oxley Act of 2002, as amended and all rules and regulations promulgated thereunder or implementing the provisions thereof ("Sarbanes-Oxley"), the Act, the Exchange Act, the Rules and Regulations, the auditing principles, rules, standards and practices applicable to auditors of "issuers" (as defined in Sarbanes-Oxley) promulgated or approved by the Public Company Accounting Oversight Board and, as applicable, the rules of The New York Stock Exchange ("Exchange Rules").

"Security Documents" means (i) as of the Initial Closing Date, except as provided in the Offered Securities, the Perfection Certificate, Main Security Agreement, the IP Security Agreement (each, as defined in the Securities Purchase Agreement) and UCC-1 financing statements and similar documents to protect and perfect a first priority security interest (subject to no liens except as provided in the Indenture and the Offered Securities) in the applicable Collateral and (ii) as of the thirtieth (30<sup>th</sup>) day following the Initial Closing Date, except as provided in the Offered Securities, the documents referred to in clause (i) and the Ancillary Security Agreements (as defined in the Securities Purchase Agreement).

"Statutory Prospectus" with reference to a particular time means the prospectus included in a Registration Statement immediately prior to that time, including any document incorporated by reference therein and any 430B Information or 430C Information with respect to such Registration Statement. For purposes of the foregoing definition, 430B Information shall be considered to be included in the Statutory Prospectus as of the actual time that form of prospectus (including a prospectus supplement) is filed with the Commission pursuant to Rule 424(b) or Rule 462(c) and not retroactively.

"Testing-the-Waters Communication" means any oral or written communication with potential investors undertaken in reliance on Section 5(d) of the Act or Rule 163B under the Act.

"Trust Indenture Act" means the Trust Indenture Act of 1939.

"Trustee" means U.S. Bank Trust Company, National Association.

"Written Testing-the-Waters Communication" means any Testing-the-Waters Communication that is a written communication within the meaning of Rule 405 under the Act.

Unless otherwise specified, a reference to a "rule" is to the indicated rule under the Act.

- (b) Compliance with Securities Act Requirements. No order preventing or suspending the use of any Statutory Prospectus has been issued by the Commission, and no stop order suspending the effectiveness of the Registration Statement or any post-effective amendment thereto has been issued, and no proceeding for that purpose has been initiated or threatened by the Commission. At the time the Registration Statement initially became effective, at the time of each amendment thereto for the purposes of complying with Section 10(a)(3) of the Act (whether by post effective amendment, incorporated report or form of prospectus), at the Effective Time relating to the Offered Securities and on each Closing Date, the Registration Statement, as amended or as supplemented, including the financial statements, if any, included or incorporated by reference therein, did and will comply in all material respects with all applicable requirements of the Act, the Exchange Act, the Exchange Act Rules and Regulations and the Rules and Regulations and did not and will not include any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading. The preceding sentence does not apply to statements in or omissions from any such document based upon written information furnished to the Company by the Placement Agent specifically for use therein, it being understood and agreed that the only such information is that described as such in Section 8(c) hereof. The Final Prospectus and any amendment and supplement thereto, including the financial statements, if any, included or incorporated by reference therein, as of its date, at the time of the filing of the Final Prospectus pursuant to Rule 424(b) and on each Closing Date, will conform in all material respects to the requirements of the Act and the Rules and Regulations and did not, does not and will not include an untrue statement of a material fact or omit to state a material fact necessary in orde
- (c) *Ineligible Issuer Status*. As of the date hereof, the Company is an "ineligible issuer" in connection with the offering of the Offered Securities pursuant to Rules 164, 405 and 433 under the Act. Neither the Company nor any affiliate of the Company has prepared or had prepared on its behalf or used or referred to any "free writing prospectus" as defined in Rule 405 under the Act and has not distributed any written materials in connection with the offer or sale of the Offered Securities (other than through the Placement Agent as contemplated by this Agreement).
- (d) Emerging Growth Company Status. From the time of the filing of the Initial Registration Statement with the Commission (or, if earlier, the first date on which the Company engaged

directly or through any person authorized to act on its behalf in any Testing-the-Waters Communication through the date hereof, the Company has been and is an "emerging growth company," as defined in Section 2(a) of the Act (an "Emerging Growth Company").

- (e) General Disclosure Package. As of the Applicable Time, none of (i) the prospectus supplement, dated August 10, 2023, including the base prospectus, dated November 21, 2022 (which is the most recent Statutory Prospectus distributed to investors generally) and the other information, if any, stated in Schedule A to this Agreement to be included in the General Disclosure Package, all considered together (collectively, the "General Disclosure Package"), or (ii) any individual Written Testing-the-Waters Communication, when considered together with the General Disclosure Package, included any untrue statement of a material fact or omitted to state any material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. The preceding sentence does not apply to statements in or omissions from any Statutory Prospectus or any Written Testing-the-Waters Communication in reliance upon and in conformity with written information furnished to the Company by the Placement Agent specifically for use therein, it being understood and agreed that the only such information furnished by the Placement Agent consists of the information described as such in Section 8(c) hereof.
- (f) Incorporated Documents. The documents that are or are deemed to be incorporated by reference in the Registration Statement or Final Prospectus or from which information is so incorporated by reference (the "Incorporated Documents"), when they became or become effective or were or are filed with the Commission, as the case may be, complied or will comply in all material respects with the requirements of the Act and the Exchange Act, as applicable, and the Rules and Regulations and the Exchange Act Rules and Regulations, as applicable; and any further documents filed and incorporated by reference subsequent thereto shall, when they are filed with the Commission, comply in all material respects with the requirements of the Act and the Exchange Act, as applicable, and the Rules and Regulations and the Exchange Act Rules and Regulations, as applicable. Each Incorporated Document did not, and any further documents filed and incorporated by reference therein will not, when filed with the Commission, contain an untrue statement of a material fact or omit to state a material fact = necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.
- (g) Testing-the-Waters Communication. The Company (i) has not alone engaged in any Testing-the-Waters Communication and (ii) has not authorized anyone other than the Placement Agent to engage in Testing-the-Waters Communications. The Company has not presented to any potential investors or otherwise distributed any Written Testing-the-Waters Communication.
- (h) Good Standing of the Company and its Subsidiaries. The Company and each of its subsidiaries listed in Exhibit 21.1 to its most recent Annual Report on Form 10-K (collectively, the "Subsidiaries") are duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, except to the extent that the failure to be so qualified or be in good standing would not reasonably be expected to have a Material Adverse Effect (as defined below). The Company and each of its Subsidiaries has full power and authority to conduct all the activities conducted by it, to own or lease all the assets owned or leased by it, except as otherwise disclosed in the Registration Statement and the Final Prospectus, and to conduct its business as described in the Registration Statement and the Final Prospectus. The Company and each of its Subsidiaries is duly licensed or qualified to do business and in good standing as a foreign corporation or such other entity in all jurisdictions in which the nature of the activities conducted by it or the character of the assets owned or leased by it makes such license or qualification necessary. All of the outstanding shares of capital stock of each Subsidiary have been duly authorized and validly issued, are fully paid and nonassessable and free of any preemptive or similar rights, and are wholly owned by the Company free and clear of all claims, liens, charges, security interests, rights of first refusal and encumbrances; there are no securities outstanding that are convertible into or exercisable or exchangeable for capital stock of any Subsidiary. The Subsidiaries are not engaged in any discussions or a party to any agreement or understanding, written or oral, regarding the acquisition of an interest in any corporation, firm, partnership, joint venture, association or other entity where such discussions, agreements or understandings would require disclosure in, or amendment to, the Registration Statement. Complete and correct copies of the Certificate of Incorpo

- (i) Capital Stock. The Company has authorized, issued and outstanding capital stock as set forth or incorporated by reference in the Registration Statement as of the respective dates set forth therein (other than (i) the grant of additional equity awards or rights under the Company's existing equity incentive and stock purchase plans, or changes in the number of outstanding shares of Common Stock of the Company due to the issuance of shares upon the exercise, conversion or redemption of securities exercisable for, convertible into or redeemable for, Common Stock outstanding on the date hereof or (ii) pursuant to the Business Combination Agreement, dated as of March 22, 2021, by and among JAWS Spitfire Acquisition Corporation, Spitfire Merger Sub, Inc., and Velo3D, Inc., as amended by Amendment #1 to Business Combination Agreement dated as of July 20, 2021 (the "Business Combination Agreement")) and such authorized capital stock conforms in all material respects to the description thereof set forth in the Registration Statement and the Final Prospectus. All of the outstanding shares of capital stock of the Company have been duly authorized, validly issued, are fully paid and nonassessable, were issued in compliance in all material respects with all applicable state and federal securities laws and are not subject to any preemptive rights, rights of first refusal or similar rights. The description of the capital stock of the Company included or incorporated by reference in the Registration Statement and the Final Prospectus is complete and accurate in all material respects. The Underlying Shares, when issued upon conversion of the Offered Securities, will conform to the description thereof set forth in or incorporated into the Final Prospectus. Except as set forth in Registration Statement and the Final Prospectus, and except pursuant to the Securities Purchase Agreement, the Offered Securities and any ATM Program (as defined in the Securities Purchase Agreement) contemplated by the Securities Purchase Agreement,
- (j) No Finder's Fee. Except as disclosed in the Registration Statement, the Final Prospectus and as contemplated by this Agreement, there are no contracts, agreements or understandings between the Company and any person that would give rise to a valid claim against the Company or the Placement Agent for a brokerage commission, finder's fee or other like payment in connection sale of the Offered Securities.
- (k) Registration Rights. Other than pursuant to the Securities Purchase Agreement, no holder of securities of the Company has rights, contractual or otherwise, to require the Company to register any securities pursuant to the Registration Statement, or to include any securities in the Registration Statement or the offering contemplated thereby, whether as a result of the filing or effectiveness of the Registration Statement, the sale of the Offered Securities as contemplated hereby or otherwise, which rights have not been duly waived in a writing furnished to the Placement Agent by the holder thereof as of the date hereof.
- (1) Listing. The Underlying Shares have been approved for listing on The New York Stock Exchange (the "Exchange"), subject to notice of issuance.
- (m) Absence of Further Requirements. No consent, approval, authorization or order of, or any filing or declaration with, any court, arbitrator or governmental or regulatory agency or body is required for the consummation of the transactions contemplated by any Security Document, the Indenture, the Offered Securities, the Securities Purchase Agreement and hereby, except such as have been obtained under the Act or the Rules and Regulations, such as may be required under state securities or Blue Sky laws, the by-laws and rules of the Financial Industry Regulatory Authority, Inc. ("FINRA"), the Exchange in connection with the offering, issuance and sale of the Offered Securities, such as otherwise have already been obtained or made as of the date of this Agreement, such consents, authorization, filings or registrations the absence of which would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (n) Title to Property. Except as disclosed in the Registration Statement and the Final Prospectus, the Company and its Subsidiaries have good and marketable title in fee simple to all

properties and assets described in the Registration Statement and the Final Prospectus as owned by them, free and clear of all liens, charges, encumbrances, claims or restrictions, except such as are not material to the business of the Company or its Subsidiaries. The Company and its Subsidiaries have valid, subsisting and, subject to (i) bankruptcy, insolvency, reorganization, receivership, moratorium, fraudulent conveyance or other similar laws relating to creditor's rights generally and (ii) general principles of equity and the discretion of the court before which any proceeding therefor may be brought (collectively, the "Enforceability Exceptions"), enforceable leases for the properties described in the Registration Statement and the Final Prospectus as leased by them. The Company and its Subsidiaries own or lease all such properties as are necessary to their operations as now conducted or as proposed to be conducted, except where the failure to so own or lease could not have a material adverse effect or could reasonably be expected to have a material adverse effect on the Company and its Subsidiaries, taken as a whole, or their respective assets, businesses, operations, earnings, properties, prospects, conditions (financial or other), stockholders' equity or results of operations, or prevent or materially interfere with consummation of the transactions contemplated hereby (such effect is referred to herein as a "Material Adverse Effect"). Each of the properties of the Company and its Subsidiaries complies in all material respects with all applicable codes, laws and regulations (including, without limitation, building and zoning codes, laws and regulations and laws relating to access to such properties), except for such failures to comply that could not, individually or in the aggregate, reasonably be expected to interfere in any material respect with the use made of such property by the Company and its Subsidiaries or otherwise have a Material Adverse Effect. None of the Company or its Subsidiaries has rec

- (o) Absence of Defaults and Conflicts Resulting from Transaction. The execution and performance of this Agreement, the Securities Purchase Agreement, the Indenture and the Security Documents and the issuance and sale of the Offered Securities and the Underlying Shares and compliance with the terms and provisions thereof will not result in the creation or imposition of any lien, charge or encumbrance upon any of the assets of the Company or any of its Subsidiaries (other than liens pursuant to the Indenture and the Security Documents) pursuant to the terms or provisions of, or result in a breach or violation of any of the terms or provisions of, or conflict with or constitute a default under, or give any party a right to terminate any of its obligations under, or result in the acceleration of any obligation under, (i) the certificate or articles of incorporation or by-laws or other organizational documents of the Company or any of its Subsidiaries, (ii) any indenture, mortgage, deed of trust, voting trust agreement, loan agreement, bond, debenture, note agreement or other evidence of indebtedness, lease, contract or other agreement or instrument to which the Company or any of its Subsidiaries is a party or by which the Company, any of its Subsidiaries or any of its or their properties is bound or affected, or (iii) violate or conflict with any judgment, ruling, decree, order, statute, rule or regulation of any court or other governmental agency or body applicable to the business or properties of the Company or any of its Subsidiaries, which lien, charge, encumbrance, breach, violation, conflict, default, termination or acceleration, in the cases of clauses (ii) or (iii), could reasonably be expected to have a Material Adverse Effect.
- (p) Absence of Existing Defaults and Conflicts. The Company and each Subsidiary has and will have performed all the obligations required to be performed by it, and is not and will not be, (i) in default, and no event has occurred that, with notice or lapse of time or both, would constitute such a default, in the due performance or observance of any term, covenant or condition contained in any indenture, mortgage, deed of trust, loan agreement or any other contract, agreement or instrument to which the Company or any of its Subsidiaries is a party or by which the Company or any of its Subsidiaries are subject or (ii) in violation of any law or statute or any judgment, order, rule or regulation of any court or arbitrator or governmental or regulatory authority, which default or violation, in the cases of clauses (i) or (ii), could reasonably be expected to have a Material Adverse Effect. To the knowledge of the Company, no other party under any contract or other instrument to which it or any of its Subsidiaries is a party is in default in any respect thereunder, which default could reasonably be expected to have a Material Adverse Effect. Neither the Company nor any of its Subsidiaries is or will be in violation of any provision of its certificate or articles of incorporation or by-laws or similar organizational documents. Neither the Company nor any of its Subsidiaries has (i) failed to pay any dividend or sinking fund installment on preferred

- stock or (ii) defaulted on any installment or other payment on indebtedness for borrowed money or on any rental on one or more long-term leases, which defaults, individually or in the aggregate, could have a Material Adverse Effect.
- (q) Authorization of Agreement. The Company has full corporate power and authority to enter into this Agreement and perform the transactions contemplated hereby. This Agreement has been duly authorized, executed and delivered by the Company and constitutes a legal, valid and binding agreement of the Company, enforceable against the Company in accordance with the terms hereof, except that the enforcement (A) of this Agreement may be subject to the Enforceability Exceptions and (B) of the indemnification and contribution obligations and provisions of this Agreement may be limited by applicable law or considerations of public policy.
- (r) Securities Purchase Agreement. The Company has full corporate power and authority to enter into the Securities Purchase Agreement and perform the transactions contemplated thereby. The Securities Purchase Agreement has been duly authorized, executed and delivered by the Company and constitutes a legal, valid and binding agreement of the Company, enforceable against the Company in accordance with the terms thereof, except that the enforcement (A) of the Securities Purchase Agreement may be subject to the Enforceability Exceptions and (B) of the indemnification and contribution obligations and provisions of the Securities Purchase Agreement may be limited by applicable law or considerations of public policy.
- (s) Execution and Delivery of Indenture. The Indenture has been duly authorized and, when executed and delivered, will be duly qualified under the Trust Indenture Act; the Offered Securities have been duly authorized and, when the Offered Securities are duly executed and authenticated by the Trustee and delivered and paid for pursuant to the Securities Purchase Agreement on each Closing Date, the Indenture will have been duly executed and delivered by the Company, such Offered Securities will have been duly executed, authenticated, issued and delivered, will conform in all material respects to the description of such Offered Securities contained in the Final Prospectus and the Indenture and such Offered Securities will constitute valid and legally binding obligations of the Company, enforceable in accordance with their terms, subject to the Enforceability Exceptions, and be entitled to the benefits and security provided by the Indenture and the Security Documents.
- (t) Offered Securities and Underlying Shares. When the Offered Securities are delivered and paid for pursuant to the Securities Purchase Agreement on each Closing Date, such Offered Securities will be convertible into the Underlying Shares of the Company in accordance with the terms of the Indenture and the Offered Securities; the Underlying Shares initially issuable upon conversion of such Offered Securities have been duly authorized and reserved for issuance upon such conversion, and conform to the description of such Underlying Shares contained or incorporation by reference in the Final Prospectus; when issued upon conversion of the Offered Securities, the Underlying Shares will be validly issued, fully paid and nonassessable; the stockholders of the Company have no preemptive rights with respect to the Offered Securities or the Underlying Shares.
- (u) Possession of Licenses and Permits. The Company and its Subsidiaries own or possess all authorizations, approvals, orders, licenses, registrations, other certificates and permits of and from all governmental regulatory officials and bodies (collectively, "Licenses"), necessary to conduct their respective businesses as contemplated in the Registration Statement and the Final Prospectus, except where the failure to own or possess all such authorizations, approvals, orders, licenses, registrations, other certificates and permits could not have a Material Adverse Effect. (i) There is no proceeding pending or threatened (or any basis therefor known to the Company) that may cause any such authorization, approval, order, license, registration, other certificate or permit to be revoked, withdrawn, cancelled, suspended or not renewed; (ii) the Company and each of its Subsidiaries is conducting its business in compliance in all material respects with all laws, rules and regulations applicable thereto (including, without limitation, all applicable federal, state and local environmental laws and regulations); and (iii) the Company has not received a notice of non-compliance, nor knows of, nor has reasonable grounds to know of, any facts that could give rise to a notice of non-compliance with any such laws, rules and regulations, and is not aware of any pending change or contemplated change to any applicable laws, rules and regulations or governmental positions; in the case of each of clauses (i), (ii) and (iii), that would materially adversely affect the business of the Company or the business or legal environment under which the Company operates.

- (v) Absence of Labor Dispute. No material labor dispute with the employees of the Company or any of its Subsidiaries exists or, to the knowledge of the Company, is imminent; and the Company is not aware of any existing, threatened or imminent labor disturbance by the employees of any of its principal suppliers, manufacturers or contractors that could have a Material Adverse Effect.
- (w) Possession of Intellectual Property. (i) The Company and each of its Subsidiaries owns or has adequate rights to use all trademarks, trade names, domain names, patents, patent rights, mask works, copyrights, technology, know-how (including trade secrets and other unpatented or unpatentable proprietary or confidential information, systems or procedures), service marks, trade dress rights and other intellectual property and registrations and applications for registration for any of the foregoing (collectively, "Intellectual Property") and has such other licenses, approvals and governmental authorizations, in each case, sufficient to conduct its business as now conducted and as now proposed to be conducted, and, to the Company's and its Subsidiaries' knowledge, there are no rights of third parties to any such Intellectual Property owned by the Company and its Subsidiaries and none of the foregoing Intellectual Property rights owned or, subject to the Enforceability Exceptions, licensed by the Company or any of its Subsidiaries is invalid or unenforceable, (ii) the Company has no knowledge of any infringement by it or any of its Subsidiaries of Intellectual Property rights of others, and there is no pending or, to the Company's knowledge, threatened action, suit, proceeding or claim by others that the Company and its Subsidiaries infringe or otherwise violate any Intellectual Property rights of others, where such infringement or violation could have a Material Adverse Effect, (iii) the Company is not aware of any infringement, misappropriation or violation by others of, or conflict by others with rights of the Company or any of its Subsidiaries with respect to, any Intellectual Property that could have a Material Adverse Effect, (iv) there is no suit, proceeding or claim being made against the Company or any of its Subsidiaries or, to the knowledge of the Company and its Subsidiaries, any employee of the Company or any of its Subsidiaries, regarding Intellectual Property, challenging the Company's and its Subsidiaries' rights in or to any such Intellectual Property or alleging other infringement that could have a Material Adverse Effect, and the Company is unaware of any facts which could form a reasonable basis for any such action, suit, proceeding or claim, (v) to the Company's knowledge, there is no third-party U.S. patent or published U.S. patent application that contains claims for which an "interference proceeding" (as defined in 35 U.S.C. § 135) has been commenced against any patent or patent application described in the Registration Statement and the Final Prospectus as being owned by or licensed to the Company and (vi) the Company and its Subsidiaries have not received any notice of infringement with respect to any patent or any notice challenging the validity, scope or enforceability of any Intellectual Property owned by or licensed to the Company or any of its Subsidiaries, in each case the loss of which patent or Intellectual Property (or loss of rights thereto) could have a Material Adverse Effect. The Company and its Subsidiaries have taken all reasonable steps necessary to secure their interests in such Intellectual Property from their employees and contractors (including, but not limited to, assignments of such Intellectual Property from such employees and contractors) and to protect the confidentiality of all of their confidential information and trade secrets and that of third parties in their possession to the extent contractually required to do so. None of the Intellectual Property or technology (including information technology and outsourced arrangements) employed by the Company or the Subsidiaries has been obtained or is being used by the Company or the Subsidiaries in violation of any contractual obligation binding on the Company or any of the Subsidiaries or, to the knowledge of the Company and its Subsidiaries, any of their respective officers, directors or employees, where such violation or violation could have a Material Adverse Effect. The Company and the Subsidiaries own or have a valid right to access and use all computer systems, networks, hardware, software, databases, websites and equipment used to process, store, maintain and operate data, information and functions used in connection with the business of the Company and the Subsidiaries (the "Company IT Systems"). The Company IT Systems are adequate for, and operate and perform in all material respects as required in connection with, the operation of the business of the Company and the Subsidiaries as currently conducted, except as could not have a Material Adverse
- (x) Privacy and Data Protection. To the knowledge of the Company, the Company and its Subsidiaries have operated its business in a manner materially compliant with all privacy and data protection laws and regulations applicable to the Company's and its Subsidiaries' collection, handling, and storage of its customers' data. The Company and its Subsidiaries have policies and procedures in place reasonably designed to ensure the integrity and security of the data collected, handled or stored in connection with the delivery of its product offerings. The Company and its Subsidiaries materially comply with, have policies and procedures in place designed to ensure privacy and data protection laws are complied with and takes appropriate steps which are

reasonably designed to assure compliance in all material respects with such policies and procedures.

- (y) Environmental Laws. Except as to those which could not, individually or in the aggregate, have a Material Adverse Effect, (i) each of the Company and each of its Subsidiaries (A) is in compliance with all applicable rules, laws and regulation relating to pollution, the protection of health or the environment, and the use, transportation, treatment, storage and disposal of, or exposure to, hazardous or toxic substances or wastes ("Environmental Law") and (B) has received and is in compliance with all permits, licenses or other approvals required of them under applicable Environmental Law to conduct their respective businesses as described in the Registration Statement and the Final Prospectus, (ii) none of the Company nor any of its Subsidiaries has received any notice from any governmental authority or third the Company or any of its Subsidiaries to liability under Environmental Laws, including any liability for remediation of any releases or threatened releases of hazardous or toxic substances.
- (z) Accurate Disclosure. The statements in the Registration Statement and the Final Prospectus under the heading "Description of Capital Stock" and in the Company's Annual Report on Form 10-K incorporated by reference in the Registration Statement and the Final Prospectus under the headings "Business—Intellectual Property" and "Risks Related to Intellectual Property", insofar as such statements summarize legal matters, agreements, documents or proceedings discussed therein, are accurate and fair summaries of such legal matters, agreements, documents or proceedings and present the information required to be shown.
- (aa) Absence of Manipulation. None of the Company, any of its Subsidiaries or any of their respective directors, officers or controlling persons has taken, or will take, directly or indirectly, any action designed, or that might reasonably be expected to cause or result, under the Act or otherwise, in, or that has constituted, stabilization or manipulation of the price of any security of the Company to facilitate the sale or resale of the Offered Securities.
- (bb) Statistical and Market-Related Data. The statistical, industry-related and market-related data included or incorporated by reference in the Registration Statement and the Final Prospectus are based on or derived from sources the Company reasonably believes are reliable and accurate, and such data agrees with the sources from which they are derived, and the Company has obtained the written consent to the use of such data from such sources to the extent required.
  - (cc) Forward-Looking Statements. No forward-looking statement (within the meaning of Section 27A of the Act and Section 21E of the Exchange Act) contained or incorporated by reference in the Registration Statement and the Final Prospectus has been made or reaffirmed without a reasonable basis or has been disclosed other than in good faith.
- (dd) Compliance with the Sarbanes-Oxley Act. The Company is in compliance in all material respects with, and there has been no failure on the part of the Company to comply in all material respects with, all applicable provisions of the Sarbanes-Oxley Act and the rules and regulations of the Commission thereunder. Since September 29, 2021, each of the principal executive officer and the principal financial officer of the Company (or each former principal executive officer of the Company and each former principal financial officer of the Company, as applicable) has made all certifications required by Sections 302 and 906 of the Sarbanes-Oxley Act with respect to all reports, schedules, forms, statements and other documents required to be filed or furnished to the Commission. For purposes of the preceding sentence, "principal executive officer" and "principal financial officer" shall have the meanings given to such terms in the Sarbanes-Oxley Act.
- (ee) Internal Controls. The books, records and accounts of the Company and its Subsidiaries accurately and fairly reflect, in reasonable detail, the transactions in, and dispositions of, the assets of, and the results of operations of, the Company and its Subsidiaries. Except as disclosed in the Registration Statement and the Final Prospectus, the Company and each of its Subsidiaries maintains a system of internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management's general or specific authorization, (ii) transactions are recorded as necessary to permit preparation of the Company's consolidated financial statements in accordance with generally accepted accounting principles and to maintain asset accountability, (iii) access to assets is permitted only in accordance with management's general or specific authorization and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any

differences. Except as set forth in the Registration Statement and the Final Prospectus, the Company's internal control over financial reporting is effective and the Company is not aware of any material weaknesses in its internal control over financial reporting. Except as set forth in the Registration Statement and the Final Prospectus, since the date of the latest audited financial statements included or incorporated by reference in the Final Prospectus, there has been no change in the Company's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

- (ff) Disclosure Controls. The Company has established and maintains disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)); such disclosure controls and procedures have been designed to ensure that information required to be disclosed relating to the Company and its Subsidiaries is made known to the Company's principal executive officer and principal financial officer by others within those entities, particularly during the period in which the Company's Annual Report on Form 10-K or Quarterly Report on Form 10-Q, as the case may be, is being prepared. The Company's certifying officers have evaluated the effectiveness of the Company's disclosure controls and procedures as of a date within 90 days prior to the filing date of the Form 10-K for the most recently ended fiscal year (such date, the "Evaluation Date"). The Company presented in its Form 10-K for the most recently ended fiscal year the conclusions of the certifying officers about the effectiveness of the disclosure controls and procedures based on their evaluations as of the Evaluation Date and, except as set forth in the Registration Statement and the Final Prospectus, the disclosure controls and procedures are effective. Except as set forth in the Registration Statement and the Final Prospectus, since the Evaluation Date, there have been no significant changes in the Company's disclosure controls or, to the Company's knowledge, in other factors that could significantly affect the Company's disclosure controls.
- (gg) Litigation. Except as set forth in the Registration Statement and the Final Prospectus, there are no material actions, suits or proceedings pending or, to the knowledge of the Company, threatened against or affecting the Company, any of its Subsidiaries or any of its or their officers in their capacity as such, nor any basis therefor, before or by any federal or state court, commission, regulatory body, administrative agency or other governmental body, domestic or foreign, wherein an unfavorable ruling, decision or finding could have a Material Adverse Effect. There are no material current or pending legal, governmental or regulatory audits or investigations, suits or proceedings that are required under the Act to be described in the Registration Statement or the Final Prospectus that are not so described.
- (hh) Financial Statements. The financial statements, together with the related notes and schedules, included or incorporated by reference in the Registration Statement or the Final Prospectus present fairly the financial condition of the Company and its consolidated Subsidiaries as of the respective dates thereof and the results of operations, cash flows and changes in stockholders' equity of the Company and its consolidated Subsidiaries for the respective periods covered thereby, and have been prepared in all material respects in conformity with generally accepted accounting principles applied on a consistent basis throughout the entire period involved. No other financial statements or schedules (historical or pro forma) are required by the Act, the Exchange Act, the Exchange Act Rules and Regulations or the Rules and Regulations to be included or incorporated by reference in the Registration Statement or the Final Prospectus. To the extent applicable, any pro forma financial statements, information or data included or incorporated by reference in the Registration Statement and the Final Prospectus comply in all material respects with the requirements of Regulation Sc X of the Act, including, without limitation, Article 11 thereof, fairly present the information set forth therein, and the assumptions used in the preparation of such proforma adjustments and data are reasonable, the pro forma adjustments used therein are appropriate to give effect to the circumstances referred to therein and the proforma adjustments have been properly applied to the historical amounts in the compilation of those statements and data. PricewaterhouseCoopers LLP (the "Accountants"), who have reported on the consolidated financial statements and schedules of the Company, are and, during the periods covered by their report were, an independent registered public accounting firm with respect to the Company within the meaning of, and as required by, the Act, the Rules and Regulations and the Public Company Accounting Oversight Board (Un

Regulation S-K under the Act, to the extent applicable. The interactive data in eXtensible Business Reporting Language included or incorporated by reference in the Registration Statement and the Final Prospectus fairly presents the information called for in all material respects and has been prepared in all material respects in accordance with the Commission's rules and guidelines applicable thereto.

- (ii) No Material Adverse Change in Business. Except as otherwise disclosed in the Registration Statement and Final Prospectus, subsequent to the filing of the Registration Statement and the Final Prospectus, (i) there has not been any Material Adverse Effect, the occurrence of any development that the Company reasonably expects could result in a Material Adverse Effect or any material adverse change, or any development that could reasonably be expected to result in a material adverse change, in the general affairs, business, management, condition (financial or otherwise), earnings, results of operations, properties, operations, assets, liabilities or prospects of the Company and its Subsidiaries, taken as a whole, whether or not arising from transactions in the ordinary course of business (a "Material Adverse Change"), (ii) there has not been any material change in the capitalization or long-term indebtedness of the Company of the Company (other than in connection with the exercise or settlement of equity awards or rights granted pursuant to the Company's equity incentive and stock purchase plans from the shares reserved therefor as described in the Registration Statement and the Final Prospectus, the exercise or redemption of warrants described in the Registration Statement and the Final Prospectus, the issuance of shares of Common Stock pursuant to the Business Combination Agreement, the grant of equity awards or rights in the ordinary course of business and consistent with the past practice of the Company and the issuance of the Offered Securities pursuant to the Securities Purchase Agreement, the issuance of the Underlying Shares pursuant to the Offered Securities and the issuance of shares pursuant to any ATM Program contemplated by the Securities Purchase Agreement), (iii) neither the Company nor any of its Subsidiaries has incurred, except in the ordinary course of business as described in the Registration Statement or the Final Prospectus, any material liabilities or obligations, direct or contingent (i
  - (jj) Investment Company Act. The Company and its Subsidiaries are not and will not become as a result of or after giving effect to the offer and sale of the Offered Securities, and will not conduct their business in a manner that would cause any of them to be, required to register as an "investment company," an entity "controlled" by an "investment company" or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company," as each such terms are defined in the Investment Company Act of 1940 (the "Investment Company Act").
  - (kk) *Ratings*. No "nationally recognized statistical rating organization" as such term is defined in Section 3(a)(62) of the Exchange Act (i) has imposed (or has informed the Company that it is considering imposing) any condition (financial or otherwise) on the Company's retaining any rating assigned to the Company or any securities of the Company or (ii) has indicated to the Company that it is considering any of the actions described in Section 7(c)(ii) hereof.
  - (II) Taxes. The Company and each of its Subsidiaries have filed all federal, state, local and foreign tax returns that have been required to be filed (within any applicable time limit extensions permitted by the relevant tax authority) and has paid all taxes and assessments shown thereon to the extent that such taxes or assessments have become due, other than those being contested in good faith and for which adequate reserves have been provided. Neither the Company nor any of its Subsidiaries has any tax deficiency, penalty or assessment that has been or, to the knowledge of the Company, might be asserted or threatened against it that could have a Material Adverse Effect. On each Closing Date, all stock transfer or other taxes (other than income taxes) that are required to be paid in connection with the sale and transfer of the Offered Securities to be sold pursuant to the Securities Purchase Agreement will be, or will have been, fully paid or provided for by the Company and all laws imposing such taxes will be or will have been fully complied with.

(mm) Insurance. The Company and each of its Subsidiaries maintains or is covered by insurance of the types and in the amounts reasonably deemed adequate for its business and customary for companies engaged in similar businesses in similar industries, including, but not limited to, insurance covering real and personal property owned or leased by the Company and its Subsidiaries against theft, damage, destruction, acts of vandalism and all other risks customarily insured against, all of which insurance is in full force and effect.

- (nn) No Unlawful Payments. Neither the Company nor any of its Subsidiaries nor, to the the Company's knowledge, any director, officer, agent, employee or other person associated with or acting on behalf of the Company or any of its Subsidiaries has, directly or indirectly, (i) used any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expense relating to political activity, (ii) made any unlawful payment from corporate funds to any foreign or domestic government official or employee or foreign or domestic political party or campaign, (iii) violated any provision of the Foreign Corrupt Practices Act of 1977 or any comparable applicable law in another jurisdiction, or (iv) made any bribe, illegal rebate, payoff, influence payment, kickback or other unlawful payment. The Company, its Subsidiaries and each of their respective controlled affiliates have instituted and maintained, and will continue to maintain, policies and procedures designed to ensure, and which are reasonably expected to continue to ensure, continued compliance with all applicable anti-bribery and anti-corruption laws and with the representation and warranty contained herein.
- (oo) Compliance with Anti-Money Laundering Laws. The operations of the Company and its Subsidiaries are and have been conducted at all times in compliance with all applicable financial recordkeeping and reporting requirements, including the Currency and Foreign Transactions Reporting Act of 1970, the Bank Secrecy Act, as amended by Title III of the Uniting and Strengthening America by Providing Appropriate Tools Requires to Intercept and Obstruct Terrorism Act of 2001, and the money laundering laws of all applicable jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines administered or enforced by any applicable governmental agency (collectively, the "Money Laundering Laws") and no action, suit or proceeding by or before any court or governmental agency, authority or body or any arbitrator involving the Company or any of its Subsidiaries with respect to the Money Laundering Laws is pending or, to the knowledge of the Company, threatened.
- (pp) Economic Sanctions. Neither the Company nor any of its Subsidiaries thereof, nor, to the knowledge of the Company or any of its Subsidiaries, any director, officer or employee, agent, affiliate or representative of the Company or any of its Subsidiaries is a government, individual or entity that is, or is owned or controlled by an individual or entity that is (i) the subject of any sanctions administered or enforced by the Office of Foreign Assets Control of the U.S. Treasury Department, the United Nations Security Council, the European Union, Her Majesty's Treasury or other relevant sanctions authority ("Sanctions"), nor (ii) located, organized or resident in a country or territory that is the subject of Sanctions (including, without limitation, Burma/Myanmar, Cuba, Iran, Libya, North Korea, Sudan, Syria, the Crimea region of Ukraine, the so-called Donetsk People's Republic, and the so-called Luhansk People's Republic). The Company and its Subsidiaries have not engaged in, and are not now engaged in, and will not engage in any dealings or transactions with any government, individual or entity, or in any country or territory, that at the time of the Sanctions. The Company and its Subsidiaries will not, directly or indirectly, use the proceeds of the issuance and sale of the Offered Securities, or lend, contribute or otherwise make available such proceeds to any subsidiary, joint venture partner or other person (A) to fund or facilitate any activities or business of or with any government, individual or entity or in any country or territory that, at the time of such funding or facilitation, is the subject of Sanctions; or (B) in any other manner that will result in a violation of Sanctions by any government, individual or entity participating in the offering, whether as underwriter, advisor, investor or otherwise).
- (qq) Other Offerings. The Company has not sold, issued or distributed any shares of Common Stock during the six-month period preceding the date hereof, including any sales pursuant to Rule 144A under, or Regulation D or Regulation S of, the Act, other than shares of Common Stock issued (i) pursuant to employee benefit plans, qualified stock option plans or other employee compensation plans, (ii) pursuant to outstanding options, restricted stock units ("RSUs"), rights or warrants or (iii) pursuant to the Sales Agreement, dated February 6, 2023, by and between the Company and Needham & Company, LLC (the "Sales Agreement").
- (rr) Accuracy of Exhibits. There are no contracts or documents which are required to be described in the Registration Statement or the Final Prospectus or to be filed as exhibits to the Registration Statement which have not been so described and filed as required.
- (ss) Lending Relationship. Except as disclosed in the Registration Statement and the Final Prospectus, the Company and its Subsidiaries (i) do not have any material lending or other

relationship with any bank or lending affiliate of the Placement Agent and (ii) does not intend to use any of the proceeds from the sale of the Offered Securities to repay any outstanding debt owed to any affiliate of the Placement Agent.

(tt) Margin Rules. Neither the issuance, sale and delivery of the Offered Securities nor the application of the proceeds thereof by the Company as described in the Registration Statement and the Final Prospectus will violate Regulation T, U or X of the Board of Governors of the Federal Reserve System or any other regulation of such Board of Governors.

(uu) ERISA. To the knowledge of the Company, the Company and each of its Subsidiaries is in compliance with all applicable provisions of the Employee Retirement Income Security Act of 1974, including the regulations and published interpretations thereunder ("ERISA"); no "reportable event" (as defined in ERISA) has occurred with respect to any "pension plan" (as defined in ERISA) for which the Company and each of its Subsidiaries would have any material liability; each of the Company and each of its Subsidiaries has not incurred and does not expect to incur material liability under (i) Title IV of ERISA with respect to termination of, or withdrawal from, any "pension plan" or (ii) Sections 412 or 4971 of the Internal Revenue Code of 1986, including the regulations and published interpretations thereunder (the "Code"); and each "pension plan" for which the Company or any Subsidiary would have any material liability that is intended to be qualified under Section 401(a) of the Code is so qualified in all material respects and nothing has occurred, whether by action or by failure to act, which would cause the loss of such qualification.

(vv) Subsidiaries. No Subsidiary is currently prohibited, directly or indirectly, from paying any dividends to the Company, from making any other distribution on such Subsidiary's capital stock, from repaying to the Company any loans or advances to such Subsidiary from the Company or from transferring any of such Subsidiary's property or assets to the Company or any other Subsidiary of the Company.

(ww) Security Interests. The Company and each relevant Subsidiary has all requisite power and authority to authorize, execute and deliver each Security Document to which it is or will be a party, perform its obligations thereunder and grant and reaffirm security interests pursuant to the applicable Security Documents. Except as provided in the Offered Securities, when the Offered Securities are delivered and paid for in accordance with the Securities Purchase Agreement on each Closing Date, the Security Documents will have been duly authorized, executed and delivered by the Company and each relevant Subsidiary and all other actions necessary or desirable to protect and perfect a first priority security interest (subject to no liens except as provided in the Indenture and the Offered Securities) in the applicable Collateral for the benefit of the holders of the Offered Securities will have been duly made or taken in each place in which such filing or recording is required to create, protect, preserve and perfect the security interest created by the Indenture, the Offered Securities and the Security Documents and will be in full force and effect, and all taxes and recording and filing fees required to be paid with respect to the execution, recording or filing of the Indenture, the the Offered Securities and the Security Documents and the issuance of the Offered Securities will have been paid, in each case, except as could have, individually or in the aggregate, a Material Adverse Effect; and, except as provided in the Offered Securities, when the Offered Securities are so delivered and paid for, and all required filings of the Indenture, the Offered Securities and the Security Documents are made and all other such actions taken, the Trustee and Collateral Agent will have a valid and perfected first priority security interest (subject to no liens except as provided in the Indenture and the Offered Securities) in the applicable Collateral enforceable against all creditors of the Company, subject only to the excepti

(xx) *Title to Collateral*. Except as provided in the Offered Securities, when the Offered Securities are delivered and paid for pursuant to the Securities Purchase Agreement on each Closing Date, the Company and each relevant Subsidiary will have good and marketable title in fee simple to all real and fixed properties and good and marketable title to all other applicable Collateral owned by the Company or such relevant Subsidiary, as the case may be, in each case free from liens, encumbrances and defects other than the lien of the Indenture and the Offered Securities, in each case, except as could have, individually or in the aggregate, a Material Adverse Effect; and the descriptions of all the applicable Collateral contained in the granting clauses of the the Securities Purchase Agreement, the Indenture and the Offered

Securities will be correct and adequate in all material respects for the purposes of the Indenture and the Offered Securities.

- 3. Agreement to Act as Placement Agent. On the basis of the representations, warranties and agreements and subject to the terms and conditions set forth herein:
  - (a) The Company hereby authorizes the Placement Agent to act as its exclusive agent to solicit offers for the purchase of all or part of the Offered Securities from the Company in connection with the proposed offering of the Offered Securities. Until the earlier of the Final Closing Date and the termination of this Agreement, the Company shall not, without the prior consent of the Placement Agent, solicit or accept offers to purchase Offered Securities otherwise than through the Placement Agent.
  - (b) The Placement Agent agrees, as agent of the Company, to use its best efforts to solicit offers to purchase the Offered Securities from the Company on the terms and subject to the conditions set forth in the Final Prospectus. The Placement Agent has no authority to bind the Company with respect to any prospective offer to purchase Offered Securities. The Placement Agent shall not have any liability to the Company in the event any such purchase is not consummated for any reason, except to the extent it is finally determined by a court of competent jurisdiction to have resulted primarily from the bad faith, willful misconduct, fraud or gross negligence of the Placement Agent. Under no circumstances will the Placement Agent be obligated to purchase any Offered Securities for its own accounts and, in soliciting purchases of Offered Securities, the Placement Agent shall act solely as the Company's agent and not as principal.
  - (c) Offers for the purchase of Offered Securities may be solicited by the Placement Agent as agent for the Company at such times and in such amounts as the Placement Agent deems advisable. The Placement Agent shall communicate to the Company, orally or in writing, each offer to purchase Offered Securities received by it as Placement Agent of the Company. The Company shall have the sole right to accept offers to purchase the Offered Securities and may reject any such offer, in whole or in part. The Placement Agent shall have the right, in its reasonable discretion, without notice to the Company, to reject any offer to purchase Offered Securities received by it, in whole or in part, and any such rejection shall not be deemed a breach of its agreement contained herein.
    - (d) The purchases of the Offered Securities by the Purchaser shall be evidenced by the execution of a Securities Purchase Agreement by each of the parties thereto.
  - (e) As compensation for services rendered, on the Initial Closing Date, the Company shall pay or cause to be paid to the Placement Agent by wire transfer of immediately available funds to an account or accounts designated by the Placement Agent, an aggregate amount equal to the greater of: (i) \$1,500,000 and (ii) five percent (5%) of the gross proceeds received by the Company from the sale of the Offered Securities. On any subsequent Optional Closing Date, the Company will pay to the Placement Agent five percent (5%) of the gross proceeds received by the Company from the sale of the Offered Securities issued and sold on such Optional Closing Date to the extent the total fees (including fees received at the Initial Closing Date) to the Placement Agent would exceed \$1,500,000.
  - (f) No Offered Securities which the Company has agreed to sell pursuant to this Agreement shall be deemed to have been purchased and paid for, or sold by the Company, until such Offered Securities shall have been delivered to the Purchaser thereof against payment by such Purchaser. If the Company shall default in its obligations to deliver Offered Securities to a Purchaser whose offer it has accepted, the Company shall indemnify and hold the Indemnified Parties (as defined below) harmless against any loss, claim, damage or liability directly or indirectly arising from or as a result of such default by the Company.
- 4. [Reserved].
- 5. Certain Agreements of the Company. The Company agrees with the Placement Agent that:
  - (a) Additional Filings. Unless filed pursuant to Rule 462(c) as part of the Additional Registration Statement in accordance with the next sentence, the Company will file each Statutory Prospectus (including the Final Prospectus), in a form approved by the Placement Agent (which approval shall not be unreasonably withheld, conditioned or delayed), with the Commission pursuant to and in accordance with subparagraph (5) of Rule 424(b) not later than the second business day following the earlier of the date it is first used or the execution and delivery of the

Securities Purchase Agreement. The Company will advise the Placement Agent promptly of any such filing pursuant to Rule 424(b) and provide satisfactory evidence to the Placement Agent of such timely filing. If an Additional Registration Statement is necessary to register a portion of the Offered Securities under the Act but the Effective Time thereof has not occurred as of the execution and delivery of the Securities Purchase Agreement, the Company will file the Additional Registration Statement or, if filed, will file a post-effective amendment thereto with the Commission pursuant to and in accordance with Rule 462(b) on or prior to 10:00 P.M., New York time, on the date of the Securities Purchase Agreement or, if earlier, on or prior to the time the Final Prospectus is finalized and distributed to the Placement Agent, or will make such filing at such later date as shall have been consented to by the Placement Agent.

- (b) Filing of Amendments: Response to Commission Requests. Prior to the earlier of the Final Closing Date or the expiration of the Option pursuant to the Securities Purchase Agreement, the Company will promptly advise the Placement Agent of any proposal to amend or supplement at any time the Initial Registration Statement, any Additional Registration Statement or any Statutory Prospectus and will not effect such amendment or supplementation without the Placement Agent's consent; and the Company will also advise the Placement Agent promptly of (i) the effectiveness of any Additional Registration Statement (if its Effective Time is subsequent to the execution and delivery of this Agreement), (ii) prior to the earlier of the Final Closing Date or the expiration of the Option pursuant to the Securities Purchase Agreement, any amendment or supplementation of a Registration Statement or any Statutory Prospectus, (iii) any request by the Commission or its staff for any amendment to any Registration Statement, for any supplement to any Statutory Prospectus or for any additional information, (iv) the institution by the Commission of any stop order proceedings in respect of a Registration Statement or the threatening of any proceeding for that purpose, and (v) the receipt by the Company of any notification with respect to the suspension of the Qualification of the Offered Securities in any jurisdiction or the institution or threatening of any proceedings for such purpose. The Company will use its best efforts to prevent the issuance of any such stop order or the suspension of any such qualification and, if issued, to obtain as soon as possible the withdrawal thereof.
- (c) Continued Compliance with Securities Laws. If, at any time when a prospectus relating to the Offered Securities is (or but for the exemption in Rule 172 would be) required to be delivered under the Act by the Placement Agent or dealer, any event occurs as a result of which the Final Prospectus as then amended or supplemented would include an untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading, or if it is necessary at any time to amend the Registration Statement or supplement the Final Prospectus to comply with the Act, the Company will promptly notify the Placement Agent of such event and will promptly prepare and file with the Commission and furnish, at its own expense, to the Placement Agent and the dealers and any other dealers upon request of the Placement Agent, an amendment or supplement which will correct such statement or omission or an amendment which will effect such compliance. Neither the Placement Agent's consent to, nor the Placement Agent's delivery of, any such amendment or supplement shall constitute a waiver of any of the conditions set forth in Section 7 hereof.
- (d) Testing-the-Waters Communication. If at any time following the distribution of any Written Testing-the-Waters Communication there occurred or occurs an event or development as a result of which such Written Testing-the-Waters Communication included or would include an untrue statement of a material fact or omitted or would omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances existing at that subsequent time, not misleading, the Company will promptly notify the Placement Agent and will promptly amend or supplement, at its own expense, such Written Testing-the-Waters Communication to eliminate or correct such statement or omission.
- (e) Rule 158. As soon as practicable, but not later than the Availability Date (as defined below), the Company will make generally available to its security holders an earnings statement covering a period of at least 12 months beginning after the Effective Time of the Initial Registration Statement (or, if later, the Effective Time of the Additional Registration Statement) which will satisfy the provisions of Section 11(a) of the Act and Rule 158 under the Act. For the purpose of the preceding sentence, "Availability Date" means the day after the end of the fourth fiscal quarter following the fiscal quarter that includes such Effective Time on which the Company is required to file its Form 10-Q for such fiscal quarter except that, if such fourth fiscal quarter is the last quarter of the Company's fiscal year, "Availability Date" means the day after the end of such fourth fiscal quarter on which the Company is required to file its Form 10-K.

- (f) Furnishing of Prospectuses. The Company will furnish to the Placement Agent electronic copies of each Registration Statement, each related Statutory Prospectus, and, so long as a prospectus relating to the Offered Securities is (or but for the exemption in Rule 172 would be) required to be delivered under the Act, the Final Prospectus and all amendments and supplements to such documents, in each case in such quantities as the Placement Agent requests. The Final Prospectus shall be so furnished on or prior to 3:00 P.M., New York time, on the business day following the execution and delivery of the Securities Purchase Agreement. All other such documents shall be so furnished as soon as available. The Company will pay the expenses of printing and distributing to the Placement Agent all such documents.
- (g) Blue Sky Qualifications. The Company will arrange for the qualification of the Offered Securities for sale under the laws of such jurisdictions as the Placement Agent designates and will continue such qualifications in effect so long as required for the distribution. Notwithstanding the foregoing, the Company will not be obligated to qualify as a foreign corporation or as a dealer in securities in any jurisdiction where it would not otherwise be required to so qualify, file any general consent to service of process in any such jurisdiction or subject itself to taxation in any such jurisdiction if it is not otherwise so subject.
- (h) Reporting Requirements. Prior to the completion of the distribution of the Offered Securities within the meaning of the Act, the Company will furnish to the Placement Agent as soon as practicable after the end of each fiscal year, an electronic copy of its annual report to stockholders for such year; and the Company will furnish to the Placement Agent within one (1) Business Day after the filing thereof with the Commission, an electronic copy of each report and any definitive proxy statement of the Company filed with the Commission under the Exchange Act or mailed to stockholders. However, so long as the Company is subject to the reporting requirements of either Section 13 or Section 15(d) of the Exchange Act and is timely filing reports with the Commission on EDGAR, it is not required to furnish such reports or statements to the Placement Agent.
  - (i) Payment of Expenses. The Company agrees with the Placement Agent that the Company will pay all reasonable, documented out-of-pocket expenses incident to the performance of the obligations of the Company, as the case may be, under this Agreement, the Securities Purchase Agreement, the Indenture and the Security Documents, including but not limited to any filing fees and other expenses (including reasonable, documented fees and disbursements of counsel to the Placement Agent) incurred in connection with qualification of the Offered Securities for sale under the laws of such jurisdictions as the Placement Agent designates and the preparation and printing of memoranda relating thereto, costs and expenses related to the review by FINRA of the Offered Securities (including filing fees and the reasonable, documented fees and expenses of counsel for the Placement Agent relating to such review), costs and expenses relating to investor presentations or any "road show" in connection with the offering and sale of the Offered Securities including, without limitation, any travel expenses of the Company's officers and employees and any other expenses of the Company including fees and expenses incident to listing or maintaining the listing of the Underlying Shares on the New York Stock Exchange, American Stock Exchange, The Nasdaq Stock Market LLC and other national and foreign exchanges, fees and expenses in connection with the registration of the Offered Securities under the Exchange Act, costs and expenses incurred in connection with the creation, perfection, protection and continuation of the first priority liens contemplated by the Indenture and each Security Document and fees and expenses incurred in connection with lien and title searches, title insurance, taxes, fees and other charges for recording mortgages and filing UCC financing statements, intellectual property filings and continuations), fees and expenses of the Trustee, fees and expenses of the Collateral Agent, any transfer taxes payable in connection with
  - (j) Use of Proceeds. The Company will use the net proceeds received by it in connection with this offering in the manner described in the "Use of Proceeds" section of the Final Prospectus, the Company does not intend to use any of the proceeds from the sale of the Offered Securities hereunder to repay any outstanding debt owed to the Placement Agent or affiliate of the Placement Agent.

- (k) Reservation of Shares. The Company will reserve and keep available at all times from and after the Initial Closing Date, free of preemptive rights, shares of Common Stock for the purpose of enabling the Company to satisfy any obligations to issue the maximum number of Underlying Shares issuable upon conversion of the Offered Securities.
- (l) Absence of Manipulation. The Company will not take, directly or indirectly, any action designed to or that would constitute or that might reasonably be expected to cause or result in, stabilization or manipulation of the price of any securities of the Company to facilitate the sale or resale of the Offered Securities.
- (m) Post-Closing Perfection Actions. Except as provided in the Offered Securities, on or prior to the thirtieth (30th) day following the Initial Closing Date, the Company and each relevant Subsidiary will authorize, execute and deliver the Ancillary Security Agreements to which it will be a party pursuant to the Securities Purchase Agreement and take any other actions necessary to perfect and protect a first priority security interest (subject to no liens except as provided in the Indenture and the Offered Securities) in the applicable Collateral covered by such Ancillary Security Agreements for the benefit of the holders of the Offered Securities.
- (n) Emerging Growth Company Status. The Company will promptly notify the Placement Agent if the Company ceases to be an Emerging Growth Company at any time prior to the completion of the distribution of the Offered Securities within the meaning of the Act.
- (o) If, immediately prior to the third anniversary of the initial effective date of the Registration Statement (the "Renewal Date"), any of the Offered Securities remain unsold, the Company will, prior to the Renewal Date, file a new shelf registration statement or, if applicable, an automatic shelf registration statement relating to the Offered Securities, in a form satisfactory to the Placement Agent and its counsel, and, if such registration statement is not an automatic shelf registration statement, will use its best efforts to cause such registration statement to be declared effective within 180 days after the Renewal Date. The Company will take all other reasonable actions necessary or appropriate to permit the public offer and sale of the Offered Securities to continue as contemplated in the expired registration statement related to the Offered Securities. From and after the effective date thereof, references herein to the "Registration Statement" shall include such new shelf registration statement or such new automatic shelf registration statement, as the case may be.
- 6. Free Writing Prospectuses. The Company represents and agrees that and the Placement Agent represents and agrees that it has not made and will not make any offer relating to the Offered Securities that would constitute a "free writing prospectus," as defined in Rule 405, required to be filed with the Commission.
- 7. Conditions of the Obligations of the Placement Agent The obligations of the Placement Agent hereunder will be subject to the accuracy of the representations and warranties of the Company herein (as though made on such Closing Date), to the accuracy of the statements of Company officers made pursuant to the provisions hereof, to the performance by the Company of its obligations hereunder and to the following additional conditions precedent:
  - (a) Accountants' Comfort Letter. (i) The Placement Agent shall have received letters, dated, respectively, the date hereof and the Initial Closing Date confirming that they are a registered public accounting firm and independent public accountants within the meaning of the Securities Laws and in the form and substance satisfactory to the Placement Agent (except that, in the letter dated the Initial Closing Date, the specified date referred to in the comfort letters shall be a date no more than three days prior to the Initial Closing Date) and (ii) The Placement Agent shall have received letters, dated each subsequent Optional Closing Date, confirming that they are a registered public accounting firm and independent public accountants within the meaning of the Securities Laws and in the form and substance satisfactory to the Placement Agent.
  - (b) Effectiveness of Registration Statement. If the Effective Time of the Additional Registration Statement (if any) is not prior to the execution and delivery of the Securities Purchase Agreement, such Effective Time shall have occurred not later than 10:00 P.M., New York time, on the date of the Securities Purchase Agreement or, if earlier, the time the Final Prospectus is finalized and distributed to the Placement Agent, or shall have occurred at such later time as shall have been consented to by the Placement Agent. The Final Prospectus shall have been filed with the Commission in accordance with the Rules and Regulations and Section 5(a) hereof. Prior to such Closing Date, no stop order suspending the effectiveness of a Registration Statement shall

have been issued and no proceedings for that purpose shall have been instituted or, to the knowledge of the Company or the Placement Agent, shall be contemplated by the Commission.

- (c) No Material Adverse Change. Subsequent to the execution and delivery of this Agreement, there shall not have occurred (i) any change, or any development or event involving a prospective change, in the condition (financial or otherwise), results of operations, business, properties or prospects of the Company and its subsidiaries taken as a whole which, in the judgment of the Placement Agent, is material and adverse and makes it impractical or inadvisable to place the Offered Securities; (ii) any downgrading in the rating of any debt securities of the Company by any "nationally recognized statistical rating organization" (as defined in Section 3(a)(62) of the Exchange Act), or any public announcement that any such organization has under surveillance or review its rating of any debt securities of the Company (other than an announcement with positive implications of a possible upgrading, and no implication of a possible downgrading, of such rating); (iii) any change in U.S. or international financial, political or economic conditions or currency exchange rates or exchange controls the effect of which is such as to make it, in the judgment of the Placement Agent, impractical to place or to enforce contracts for the sale of the Offered Securities, whether in the primary market or in respect of dealings in the secondary market; (iv) any suspension or material limitation of trading in securities generally on the New York Stock Exchange or the Nasdaq Stock Market, or any setting of minimum or maximum prices for trading on such exchange; (v) or any suspension of trading of any securities of the Company on any exchange or in the over-the-counter market; (vi) any banking moratorium declared by any U.S. federal or New York authorities; (vii) any attack on, outbreak or escalation of hostilities or act of terrorism involving the United States or any other country where such securities are listed or (viii) any attack on, outbreak or escalation of hostilities or act of terrorism involving the United States, any declaration
- (d) Opinion and 10b-5 Statement of Counsel for the Company. The Placement Agent shall have received an opinion and 10b-5 statement, dated such Closing Date, of Fenwick & West LLP, counsel for the Company, and an opinion, dated such Closing Date, of intellectual property counsel to the Company satisfactory to the Placement Agent as to intellectual property matters, each in form and substance satisfactory to the Placement Agent and its counsel.
- (e) Opinion and 10b-5 Statement of Counsel for the Placement Agent. The Placement Agent shall have received from Skadden, Arps, Slate, Meagher & Flom LLP, counsel for the Placement Agent, such opinion or opinions and 10b-5 statement, dated such Closing Date, with respect to such matters as the Placement Agent may require, and the Company shall have furnished to such counsel such documents as they request for the purpose of enabling them to pass upon such matters.
- (f) Officers' Certificate. The Placement Agent shall have received a certificate, dated such Closing Date, of an executive officer of the Company and a principal financial or accounting officer of the Company in which such officers shall state that: the representations and warranties of the Company in this Agreement are true and correct; the Company has complied with all agreements and satisfied all conditions on its part to be performed or satisfied hereunder at or prior to such Closing Date; no stop order suspending the effectiveness of any Registration Statement has been issued and no proceedings for that purpose have been instituted or, to the best of their knowledge and after reasonable investigation, are contemplated by the Commission; the Additional Registration Statement (if any) satisfying the requirements of subparagraphs (1) and (3) of Rule 462(b) was timely filed pursuant to Rule 462(b), including payment of the applicable filing fee in accordance with Rule 456(a) or (b); and, subsequent to the date of the most recent financial statements included or incorporated by reference in the Final Prospectus, there has been no material adverse change, nor any development or event involving a prospective material adverse change, in the condition (financial or otherwise), results of operations, business, properties or prospects of the Company and its subsidiaries taken as a whole except as set forth in the Final Prospectus or as described in such certificate.

- (g) CFO Certificate. At the time of the execution of the Securities Purchase Agreement, and on each Closing Date, the Placement Agent shall have received a certificate of the Chief Financial Officer of the Company in form and substance satisfactory to the Placement Agent.
- (h) Listing. The Underlying Shares issuable upon conversion of the Offered Securities shall have been approved for listing on the Exchange, subject to notice of issuance.

The Company will furnish the Placement Agent with such conformed copies of such opinions, certificates, letters and documents as the Placement Agent reasonably requests. The Placement Agent may in its sole discretion waive compliance with any conditions to the obligations of the Placement Agent hereunder, whether in respect of an Optional Closing Date or otherwise.

- 8. Indemnification and Contribution. (a) Indemnification of the Placement Agent by Company. The Company will indemnify and hold harmless the Placement Agent, its partners, members, directors, officers, employees, agents, affiliates and each person, if any, who controls the Placement Agent within the meaning of Section 15 of the Act or Section 20 of the Exchange Act (each an "Indemnified Party"), against any and all losses, claims, damages or liabilities, joint or several, to which such Indemnified Party may become subject, under the Act, the Exchange Act, other Federal or state statutory law or regulation or otherwise, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) arise out of or are based upon any untrue statement or alleged untrue statement of any material fact contained in any part of any Registration Statement at any time, any Statutory Prospectus as of any time, the Final Prospectus or any Written Testing-the-Waters Communication, or arise out of or are based upon the omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, and will reimburse each Indemnified Party for any reasonable and documented legal or other expenses reasonably incurred by such Indemnified Party in connection with investigating or defending against any loss, claim, damage, liability, action, litigation, investigation or proceeding whatsoever (whether or not such Indemnified Party is a party thereto), whether threatened or commenced, and in connection with the enforcement of this provision with respect to any of the above as such expenses are incurred; provided, however, that the Company will not be liable in any such case to the extent that any such loss, claim, damage or liability arises out of or is based upon an untrue statement or alleged untrue statement in or omission or alleged omission from any of such documents in reliance upon and in conformity with written information furnished to the Company by the Plac
- (b) Indemnification of Company. The Placement Agent will indemnify and hold harmless the Company, each of its directors and each of its officers who signs a Registration Statement and each person, if any, who controls the Company within the meaning of Section 15 of the Act or Section 20 of the Exchange Act (each, a "Placement Agent Indemnified Party") against any losses, claims, damages or liabilities to which such Placement Agent Indemnified Party may become subject, under the Act, the Exchange Act, or other Federal or state statutory law or regulation or otherwise, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) arise out of or are based upon any untrue statement or alleged untrue statement of any material fact contained in any part of any Registration Statement at any time, any Statutory Prospectus as of any time, the Final Prospectus or any Written Testing-the-Waters Communication or arise out of or are based upon the omission or the alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, in each case to the extent, but only to the extent, that such untrue statement or alleged untrue statement or omission or alleged omission was made in reliance upon and in conformity with written information furnished to the Company by the Placement Agent specifically for use therein, and will reimburse any legal or other expenses reasonably incurred by such Placement Agent Indemnified Party in connection with investigating or defending against any such loss, claim, damage, liability, action, litigation, investigation or proceeding whatsoever (whether or not such Placement Agent Indemnified Party is a party thereto), whether threatened or commenced, based upon any such untrue statement or omission, or any such alleged untrue statement or omission as such expenses are incurred, it being understood and agreed that the only such information furnished by the Placement Agent consists of the following information in the F

(d) Actions against Parties; Notification. Promptly after receipt by an indemnified party under this Section of notice of the commencement of any action, such indemnified party will, if a claim in respect thereof is to be made against an indemnifying party under subsection (a), (b) or (c) above, notify the indemnifying party of the commencement thereof; but the failure to notify the indemnifying party shall not relieve it from any liability that it may have under subsection (a), (b) or (c) above except to the extent that it has been materially prejudiced (through the forfeiture of substantive rights or defenses) by such failure; and

provided further that the failure to notify the indemnifying party shall not relieve it from any liability that it may have to an indemnified party otherwise than under subsection (a), (b) or (c) above. In case any such action is brought against any indemnified party and it notifies an indemnifying party of the commencement thereof, the indemnifying party will be entitled to participate therein and, to the extent that it may wish, jointly with any other indemnifying party similarly notified, to assume the defense thereof, with counsel satisfactory to such indemnified party (who shall not, except with the consent of the indemnified party, be counsel to the indemnifying party), and after notice from the indemnifying party to such indemnified party under this Section for any legal or other expenses subsequently incurred by such indemnified party in connection with the defense thereof other than reasonable costs of investigation. No indemnifying party shall, without the prior written consent of the indemnified party, effect any settlement of any pending or threatened action in respect of which any indemnified party is or could have been a party and indemnity could have been sought hereunder by such indemnified party unless such settlement (i) includes an unconditional release of such indemnified party from all liability on any claims that are the subject matter of such action and (ii) does not include a statement as to, or an admission of, fault, culpability or a failure to act by or on behalf of an indemnified party. No indemnifying party will be liable for any settlement of any action or claim effected without its written consent (which consent will not be unreasonably withheld or delayed).

(e) Contribution. If the indemnification provided for in this Section is unavailable or insufficient to hold harmless an indemnified party under subsection (a), (b) or (c) above, then each indemnifying party shall contribute to the amount paid or payable by such indemnified party as a result of the losses, claims, damages or liabilities referred to in subsection (a), (b) or (c) above (i) in such proportion as is appropriate to reflect the relative benefits received by the Company on the one hand and the Placement Agent on the other from the offering of the Offered Securities or (ii) if the allocation provided by clause (i) above is not permitted by applicable law, in such proportion as is appropriate to reflect not only the relative benefits referred to in clause (i) above but also the relative fault of the Company on the one hand and the Placement Agent on the other in connection with the statements or omissions which resulted in such losses, claims, damages or liabilities as well as any other relevant equitable considerations. The relative benefits received by the Company on the one hand and the Placement Agent on the other shall be deemed to be in the same proportion as the total net proceeds from the offering (before deducting expenses) received by the Company bear to the total fees received by the Placement Agent under this Agreement. The relative fault shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the Company or the Placement Agent and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such untrue statement or omission. The amount paid by an indemnified party as a result of the losses, claims, damages or liabilities referred to in the first sentence of this subsection (e) shall be deemed to include any legal or other expenses reasonably incurred by such indemnified party in connection w

#### 9. [Reserved]

- 10. Termination. If any condition specified in Section 7 is not satisfied when and as required to be satisfied, this Agreement may be terminated by the Placement Agent by notice to the Company at any time on or prior to the Initial Closing Date. The Company may terminate this Agreement at any time prior to execution of a Securities Purchase Agreement upon written notice to the Placement Agent. Any termination pursuant to this Section 10 shall be without liability on the part of any party to any other party, except that Section 5, Section 8, Section 9, Section 10, Section 11, Section 12 and Section 13 shall at all times be effective and shall survive such termination.
- 11. Survival of Certain Representations and Obligations. The respective indemnities, agreements, representations, warranties and other statements of the Company or its officers and of the Placement Agent set forth in or made pursuant to this Agreement will remain in full force and effect, regardless of any investigation, or statement as to the results thereof, made by or on behalf of the Placement Agent, the Company or any of their respective representatives, officers or directors or any controlling person, and will survive delivery of and payment for the Offered Securities. If this Agreement is terminated

for any reason or, if for any reason, the purchase of the Offered Securities by the Purchaser is not consummated because of any refusal, inability or failure on the part of the Company to perform any agreement herein or to comply with any provision hereof (including the provisions of Section 7 hereof), the Company will reimburse the Placement Agent for all reasonable, documented out-of-pocket expenses (including reasonable, documented fees and disbursements of counsel) reasonably incurred by it in connection with the offering of the Offered Securities, and the respective obligations of the Company and the Placement Agent pursuant to Section 8 hereof shall remain in effect. In addition, if any Offered Securities have been purchased pursuant to a Securities Purchase Agreement, the representations and warranties in Section 2 and all obligations under Section 5 shall also remain in effect.

- 12. Notices. All communications hereunder will be in writing and, if sent to the Placement Agent, will be mailed, delivered or telegraphed and confirmed to the Placement Agent, c/o Credit Suisse Securities (USA) LLC, Eleven Madison Avenue, New York, N.Y. 10010-3629, Attention: LCD-IBD, with a copy (which shall not constitute notice) to Skadden, Arps, Slate Meagher & Flom LLP, One Manhattan West, New York, New York 10001, Attention: Ryan Dzierniejko, email: [\*], or, if sent to the Company, will be mailed, delivered or telegraphed and confirmed to it at 511 Division St. Campbell, CA 95008, Attention: William McCombe, email: [\*], with a copy (which shall not constitute notice) to Fenwick & West LLP, 902 Broadway, New York, NY, Attention: Per Chilstrom, email: [\*]. An electronic communication shall be deemed written notice for purposes of this Section 12 if sent to the electronic mail address specified by the receiving party in this Section 12.
- 13. Successors. This Agreement will inure to the benefit of and be binding upon the parties hereto and their respective successors and the officers and directors and controlling persons referred to in Section 7, and no other person will have any right or obligation hereunder.
- 14. Entire Agreement. This Agreement constitutes the entire agreement between the parties and supersedes all prior agreements, both written and oral, with respect to the subject matter hereof, including, without limitation, that certain letter agreement, dated as of June 14, 2023, between the Company and Credit Suisse Securities (USA) LLC.
- 15. Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute one and the same Agreement. Counterparts may be delivered via facsimile, electronic mail (including any electronic signature covered by the U.S. federal ESIGN Act of 2000, Uniform Electronic Transactions Act, the Electronic Signatures and Records Act or other applicable law, e.g., www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.
  - 16. Absence of Fiduciary Relationship. The Company acknowledges and agrees that:
- (a) No Other Relationship. The Placement Agent has been retained solely to act as a placement agent in connection with the sale of the Offered Securities and that no fiduciary, advisory or agency relationship between the Company, on the one hand, and the Placement Agent, on the other, has been created in respect of any of the transactions contemplated by this Agreement or the Final Prospectus, irrespective of whether the Placement Agent has advised or is advising the Company on other matters;
- (b) Arms' Length Negotiations. The fees paid to the Placement Agent set forth in this Agreement were established by Company following discussions and arms-length negotiations with the Placement Agent and the Company and the Placement Agent are capable of evaluating and understanding and understand and accept the terms, risks and conditions of the transactions contemplated by this Agreement;
- (c) Absence of Obligation to Disclose. The Company has been advised that the Placement Agent and its affiliates are engaged in a broad range of transactions which may involve interests that differ from those of the Company and that the Placement Agent has no obligation to disclose such interests and transactions to the Company by virtue of any fiduciary, advisory or agency relationship; and
- (d) Waiver. The Company waives, to the fullest extent permitted by law, any claims they may have against the Placement Agent for breach of fiduciary duty or alleged breach of fiduciary duty and agree that the Placement Agent shall have no liability (whether direct or indirect) to the Company in respect of

such a fiduciary duty claim or to any person asserting a fiduciary duty claim on behalf of or in right of the Company, including stockholders, employees or creditors of the Company.

#### 17. Applicable Law. This Agreement shall be governed by, and construed in accordance with, the laws of the State of New York.

The Company hereby submits to the non-exclusive jurisdiction of the Federal and state courts in the Borough of Manhattan in The City of New York in any suit or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby. The Company irrevocably and unconditionally waives any objection to the laying of venue of any suit or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby in Federal and state courts in the Borough of Manhattan in the City of New York and irrevocably and unconditionally waives and agrees not to plead or claim in any such court that any such suit or proceeding in any such court has been brought in an inconvenient forum.

# 18. WAIVER OF JURY TRIAL. THE COMPANY HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

- 19. Recognition of the U.S. Special Resolution Regimes.
  - (i) In the event that the Placement that is a Covered Entity becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer from the Placement Agent of this Agreement, and any interest and obligation in or under this Agreement, will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if this Agreement, and any such interest and obligation, were governed by the laws of the United States or a state of the United States.
  - (ii) In the event that the Placement Agent that is a Covered Entity or a BHC Act Affiliate of the Placement Agent becomes subject to a proceeding under a U.S. Special Resolution Regime, Default Rights under this Agreement that may be exercised against the Placement Agent are permitted to be exercised to no greater extent than such Default Rights could be exercised under the U.S. Special Resolution Regime if this Agreement were governed by the laws of the United States or a state of the United States.

As used in this Section 16(g):

"BHC Act Affiliate" has the meaning assigned to the term "affiliate" in, and shall be interpreted in accordance with, 12 U.S.C. § 1841(k).

"Covered Entity" means any of the following:

- (i) a "covered entity" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 252.82(b);
- (ii) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or
- (iii) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b).

"Default Right" has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F.R. §§ 252.81, 47.2 or 382.1, as applicable.

"U.S. Special Resolution Regime" means each of (i) the Federal Deposit Insurance Act and the regulations promulgated thereunder and (ii) Title II of the Dodd-Frank Wall Street Reform and Consumer Protection Act and the regulations promulgated thereunder.

[Signature pages follow]

If the foregoing is in accordance with the Placement whereupon it will become a binding agreement between the	Agent's understanding of our agreement, kindly sign and return to the Company one of the counterparts hereof. Company and the Placement Agent in accordance with its terms.
Very truly yours,	
Velo3D, Inc.	
By /s/ Benny Buller	Name: Benny Buller Title: Chief Executive Officer
The foregoing Placement Agent Agreement is hereby confir	med and accepted as of the date first above written.
Credit Suisse Securities (USA) LLC	
By: <u>/s/ Christopher F. Siska</u> Name: Christopher F. Siska Title: Managing Director	
	24

## SCHEDULE A

None.

Certain information in this document indicated with "[\*]" has been omitted from this exhibit because it is both (i) not material and (ii) would be competitively harmful if publicly disclosed.

#### SECURITY AGREEMENT

THIS **SECURITY AGREEMENT** (as amended, restated, supplemented or otherwise modified from time to time, this "**Agreement**") dated as of August 14, 2023 among Velo3D, Inc., a Delaware corporation (the "**Pledgor**"), each of the Subsidiaries of the Pledgor from time to time party hereto (together with the Pledgor and all Additional Grantors (as defined below), the "**Grantors**") and High Trail Investments ON LLC, a Delaware limited liability company, in its capacity as collateral agent for the benefit of the Holders (as defined below) and the Trustee (as defined in the Notes) (together with its successors and assigns in such capacity, the "**Secured Party**").

#### WITNESSETH:

- **WHEREAS**, the Pledgor will enter into that certain Securities Purchase Agreement, dated as of the date hereof (as amended, restated, supplemented or otherwise modified from time to time, the "Securities Purchase Agreement"), with High Trail Investments ON LLC and HB SPV I Master Sub LLC (collectively, the "Initial Holder") and each other party thereto, pursuant to which, among other things, the Pledgor will issue, and the Initial Holder will purchase, subject to the terms set forth therein, the Notes (as defined in the Securities Purchase Agreement);
- **AND WHEREAS**, it is a condition precedent to the closing under the Securities Purchase Agreement that the Grantors shall have executed and delivered this Agreement to the Secured Party for its benefit and the benefit of the Holders.
- NOW, THEREFORE, in consideration of the foregoing and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:
- Section 1. <u>Definitions</u>. Capitalized terms used herein without definition and defined in the Notes are used herein as defined therein. In addition, as used herein:
  - "Account" means any "account", as such term is defined in the UCC.
  - "Additional Grantor" shall have the meaning given in Section 4.14.
  - "Agreement" has the meaning set forth in the preamble hereof.
- "Applicable Law" means, in relation to any subject, all provisions applicable to that subject of all (i) constitutions, treaties, statutes, laws, rules, regulations and ordinances of any Governmental Entity, (ii) authorizations, consents, approvals, permits or licenses issued by, or a registration or filing with, any Governmental Entity and (iii) orders, decisions, judgments, awards and decrees of any Governmental Entity (including common law and principles of public policy).
- "Chattel Paper" means all "chattel paper", as such term is defined in the UCC, including, without limitation, "electronic chattel paper" and "tangible chattel paper", as each term is defined in the UCC.
  - "Collateral" has the meaning ascribed thereto in Section 3 hereof.
- "Collateral Records" means all books, records, ledger cards, files, correspondence, customer lists, blueprints, technical specifications, manuals, computer software, computer printouts, tapes, disks and related data processing software and similar items that at any time evidence or contain information relating to any of the Collateral or are otherwise necessary or helpful in the collection thereof or realization thereupon.
- "Collateral Support" means all property (real or personal) assigned, hypothecated or otherwise securing any Collateral and shall include any security agreement or other agreement granting a lien or security interest in such real or personal property.

- "Commercial Tort Claims" means "commercial tort claims", as such term is defined in the UCC, including, without limitation, all commercial tort claims listed on <a href="Schedule VIII">Schedule VIII</a> hereto.
- "Contracts" means all contracts, undertakings, or other agreements (other than rights evidenced by Chattel Paper, Documents or Instruments) in or under which a Grantor may now or hereafter have any right, title or interest, including, without limitation, with respect to an Account, any agreement relating to the terms of payment or the terms of performance thereof.
  - "Control Agreement" has the meaning set forth in Section 4.5 hereof.
- "Copyrights" means all copyrights and rights, title and interests (and all related IP Ancillary Rights) in copyrights, works protectable by copyrights, mask works, database and design rights, copyright registrations and copyright applications, including, without limitation, the copyright registrations and copyright applications listed on Schedule III attached hereto (if any), all Copyrights (as defined in the Notes), and all renewals of any of the foregoing.
  - "Deposit Accounts" means all "deposit accounts", as such term is defined in the UCC, now or hereafter held in the name of a Grantor.
- "**Documents**" means all "documents", as such term is defined in the UCC, and shall include, without limitation, all documents of title (as defined in the UCC), bills of lading or other receipts evidencing or representing Inventory or Equipment.
- "Equipment" means (i) all "equipment", as such term is defined in the UCC and, in any event, shall include, Motor Vehicles, (ii) all machinery, manufacturing equipment, data processing equipment, computers, office equipment, furnishings, furniture, appliances, fixtures and tools (in each case, regardless of whether characterized as equipment under the UCC); and (iii) all accessions or additions thereto, all parts thereof, whether or not at any time of determination incorporated or installed therein or attached thereto, and all replacements therefor, wherever located, now or hereafter existing, including any fixtures.
- "Excluded Accounts" means (a) any accounts maintained by a Grantor exclusively used for payroll, payroll taxes and other employee wage and benefit payments to or for the benefit of such Grantor's employees; provided amounts therein are transferred into such accounts no earlier than five Business Days prior to the use of all amounts contained therein for making such payments; or (b)(i) any merchant account in the nature of accounts with payment service providers; (ii) any bank or deposit account used solely for purposes of cash deposits or pledges constituting Permitted Liens; or (iii) any escrow account, trust account or other fiduciary account maintained in the ordinary course of business; provided, that at no time shall the aggregate amount in the accounts described in this clause (b) exceed \$1,000,000.
  - "Excluded Collateral" has the meaning set forth in Section 3.1(a) hereof.
- "General Intangibles" means all "general intangibles", as such term is defined in the UCC, and, in any event, shall include, without limitation, payment intangibles, contract rights, rights to payment, rights arising under common law, statutes, or regulations, choses or things in action, goodwill (including the goodwill associated with any Trademark), Patents, Trademarks, Copyrights, URLs and domain names, industrial designs and other Intellectual Property or rights therein or applications therefor, whether under license or otherwise, programs, programming materials, blueprints, drawings, purchase orders, customer lists, monies due or recoverable from pension funds, rights to payment and other rights under any royalty or licensing agreements, including Intellectual Property Licenses, infringement claims, computer programs, information contained on computer disks or tapes, software, literature, reports, catalogs, pension plan refunds, pension plan refund claims, insurance premium rebates, tax refunds, and tax refund claims, interests in a partnership or limited liability company which do not constitute a security under Article 8 of the UCC.
- "Goods" means all "goods", as such term is defined in the UCC, including, without limitation, fixtures and embedded Software to the extent included in "goods" as defined in the UCC.
- "Governmental Entity" means any nation, state, county, city, town, village, district, or other political jurisdiction of any nature, federal, state, local, municipal, foreign, or other

government, governmental or quasi-governmental authority of any nature (including any governmental agency, branch, department, official, or entity and any court or other tribunal), multi-national organization or body; or body exercising, or entitled to exercise, any administrative, executive, judicial, legislative, police, regulatory, or taxing authority or power of any nature or instrumentality of any of the foregoing, including any entity or enterprise owned or controlled by a government or a public international organization or any of the foregoing.

- "Grantors" has the meaning set forth in the preamble hereof.
- "Holders" means the Initial Holder and each Holder under and as defined in any Note.
- "Initial Holder" has the meaning set forth in the preamble hereof.
- "Instruments" means all "instruments", as such term is defined in the UCC, and shall include, without limitation, promissory notes, drafts, bills of exchange and trade acceptances.
- "Insurance" means (i) all insurance policies covering any or all of the Collateral (regardless of whether the Secured Party is the loss payee thereof) and (ii) all key man life insurance policies (if any).
- "Intellectual Property" means all rights, title and interests in intellectual property arising under any Applicable Law and all IP Ancillary Rights relating thereto, including all Copyrights, Patents, Trademarks, Internet Domain Names, Trade Secrets, industrial designs, integrated circuit topographies, confidential proprietary information and rights under Intellectual Property Licenses.
- "Intellectual Property Licenses" means all rights under or interests in any written agreement, including all contractual obligations (and all related IP Ancillary Rights), granting any right, title and interest in any Intellectual Property, including software license agreements, whether a Grantor is a licensee or licensor under any such license agreement, and including, without limitation, the license agreements listed on Schedule IV attached hereto and all Copyright Licenses, Patent Licenses and Trademark Licenses (each as defined in the Notes).
- "Internet Domain Name" means all right, title and interest (and all related IP Ancillary Rights) arising under any Applicable Law in Internet domain names.
- "IP Ancillary Rights" means, with respect to an item of Intellectual Property all foreign counterparts to, and all divisionals, reversions, continuations, continuations-in-part, reissues, reexaminations, renewals and extensions of, such Intellectual Property and all income, royalties, proceeds and liabilities at any time due or payable or asserted under or with respect to any of the foregoing or otherwise with respect to such Intellectual Property, including all rights to sue or recover at law or in equity for any past, present or future infringement, misappropriation, dilution, violation or other impairment thereof, and, in each case, all rights to obtain any other IP Ancillary Right.
- "Inventory" means (i) any "inventory", as such term is defined in the UCC, and (ii) all goods held for sale or lease or to be furnished under contracts of service or so leased or furnished, all raw materials, work in process, finished goods, and materials used or consumed in the manufacture, packing, shipping, advertising, selling, leasing, furnishing or production of such inventory or otherwise used or consumed in a Grantor's business; all goods in which a Grantor has an interest in mass or a joint or other interest or right of any kind; and all goods which are returned to or repossessed by a Grantor, all computer programs embedded in any goods and all accessions thereto and products thereof (in each case, regardless of whether characterized as inventory under the UCC).
  - "Investment Property" means all "investment property", as such term is defined in the UCC.
  - "Joinder Agreement" shall mean a Joinder Agreement substantially in the form of Annex I.

- "Letter-of-Credit Right" means any "letter-of-credit right", as such term is defined in the UCC.
- "Mortgage" means any mortgage, deed of trust, deed to secure debt or other document, in form and substance reasonably satisfactory to the Secured Party, creating in favor of the Secured Party a Lien on real property owned by a Grantor.
- "Motor Vehicles" means motor vehicles, tractors, trailers and other like property, whether or not the title thereto is governed by a certificate of title or ownership.
- "Note Documents" means the Securities Purchase Agreement, the Notes, the Security Documents and the other Transaction Documents (as defined in the Securities Purchase Agreement).
  - "Notes" has the meaning set forth in the Securities Purchase Agreement.
- "Obligations" means all liabilities, indebtedness and obligations (including interest accrued at the rate provided in the applicable Note Document after the commencement of a bankruptcy proceeding, whether or not a claim for such interest is allowed) of the Pledgor and the other Grantors under the Notes, any Security Document or any other Note Document, in each case howsoever created, arising or evidenced, whether direct or indirect, absolute or contingent, now or hereafter existing, or due or to become due.
- "Patents" means any patents and patent applications, including, without limitation, the inventions and improvements described and claimed therein, all patentable inventions and those patents and patent applications listed on <u>Schedule V</u> attached hereto (if any), all Patents (as defined in the Notes), and all IP Ancillary Rights in respect of any of the foregoing.
- "Person" means an individual, a limited liability company, a partnership, a joint venture, a corporation, a trust, an unincorporated organization, any other entity and any Governmental Entity.
  - "Pledge Supplement" has the meaning set forth in Section 4.1(j) hereof.
- "Pledged Collateral" means (a) all of the Pledged Interests, (b) the certificates, if any, representing the Pledged Interests and any interest of a Grantor on the books and records of any Pledged Entity pertaining to such Pledged Interests and (c) all dividends, distributions, cash, warrants, rights, options, instruments, securities and other property or proceeds from time to time received, receivable or otherwise distributed in respect of, or in exchange for, any or all of the Pledged Interests.
- "Pledged Entities" means the corporations, limited liability companies and other entities set forth on Exhibit A and each other corporation, limited liability company or other entity, the stock or other equity interests and securities of which are owned or acquired by a Grantor and described on a Pledge Supplement.
- "Pledged Interests" means all of the capital stock, limited liability company interests and other equity interests and securities of the Pledged Entities or any other entity now owned or hereafter acquired by a Grantor.
  - "Pledgor" has the meaning set forth in the preamble hereof.
- "Proceeds" means "proceeds", as such term is defined in the UCC and, in any event, includes, without limitation, (a) any and all proceeds of any insurance, indemnity, warranty or guaranty payable with respect to any of the Collateral, (b) any and all payments (in any form whatsoever) made or due and payable from time to time in connection with any requisition, confiscation, condemnation, seizure or forfeiture of all or any part of the Collateral by any Governmental Entity (or any person acting under color of a Governmental Entity), and (c) any and all other amounts from time to time paid or payable under, in respect of or in connection with any of the Collateral.

"Receivables" means all rights to payment, whether or not earned by performance, for goods or other property sold, leased, licensed, assigned or otherwise disposed of, or services rendered or to be rendered, including, without limitation all such rights constituting or evidenced by any Account, Chattel Paper, Instrument, General Intangible or Investment Property, together with all of a Grantor's rights, if any, in any goods or other property giving rise to such right to payment and all Collateral Support and Supporting Obligations related thereto and all Receivables Records.

"Receivables Records" means (i) all originals or copies of all documents, instruments or other writings or electronic records or other Records evidencing the Receivables; (ii) all books, correspondence, credit or other files, Records, ledger sheets or cards, invoices, and other papers relating to Receivables, including, without limitation, all tapes, cards, computer tapes, computer discs, computer runs, record keeping systems and other papers and documents relating to the Receivables, whether in the possession or under the control of a Grantor or any computer bureau or agent from time to time acting for such Grantor or otherwise; (iii) all evidences of the filing of financing statements and the registration of other instruments in connection therewith, and amendments, supplements or other modifications thereto, notices to other creditors or secured parties, and certificates, acknowledgments, or other writings, including, without limitation, lien search reports, from filing or other registration officers; (iv) all credit information, reports and memoranda related thereto; and (v) all other written or non-written forms of information related in any way to the foregoing or any Receivable.

"Record" has the meaning specified in the UCC.

"Related Person" has the meaning specified in Section 6.12(b).

"Representative" means any Person acting as agent, trustee or representative on behalf of the Secured Party from time to time.

"Security Documents" means this Agreement, the Subsidiary Guaranty, the Control Agreements, the Mortgages, if any, and each other agreement or instrument pursuant to or in connection with which the Pledgor or any of its Subsidiaries grants a security interest in any Collateral to the Secured Party, for its benefit and the benefit of the Holders, or pursuant to which any such security interest in Collateral is perfected, each as amended, restated, supplemented or otherwise modified from time to time in accordance with the terms hereof and thereof.

"Secured Party" has the meaning set forth in the preamble hereof.

"Securities Purchase Agreement" has the meaning set forth in the recitals hereof.

"Software" means all "software", as such term is defined in the UCC, now owned or hereafter acquired by a Grantor, other than software embedded in any category of Goods, including, without limitation, all computer programs and all supporting information provided in connection with a transaction related to any program.

"Subsidiary Guaranty" means that certain Subsidiary Guaranty, dated as of the date hereof, by and among the Pledgor and the Grantors party thereto in favor of the Secured Party, as amended, restated or modified from time to time.

"Supporting Obligation" means any "supporting obligation", as such term is defined in the UCC.

"Trade Secrets" means all right, title and interest (and all related IP Ancillary Rights) arising under any Applicable Law in or relating to trade secrets.

"Trademarks" means all rights, title and interests (and all related IP Ancillary Rights) arising under any Applicable Law in any trademarks, trade names, internet domain names, URLs, all websites, corporate names, company names, business names, fictitious business names, trade styles, service marks, logos, other source or business identifiers, all goodwill associated therewith, all registrations and recordings thereof and all applications in connection therewith, including, without limitation, the trademarks, trademark applications, internet domain names and

URLs listed in Schedule VI attached hereto (if any) and renewals thereof, all Trademarks (as defined in the Notes).

- "UCC" shall mean the Uniform Commercial Code as in effect from time to time in the State of New York; provided, that to the extent that the Uniform Commercial Code is used to define any term herein and such term is defined differently in different Articles of the Uniform Commercial Code, the definition of such term contained in Article 9 shall govern.
- Section 2. <u>Representations, Warranties and Covenants of the Grantors</u>. Each Grantor represents and warrants to, and covenants with, the Secured Party, as of the date hereof, as follows:
- (a) Such Grantor has rights in and the power to transfer the Collateral in which it purports to grant a security interest pursuant to Section 3 hereof (subject, with respect to after acquired Collateral, to such Grantor acquiring the same) and no Lien other than (x) with respect to the Collateral other than the Pledged Collateral, Permitted Liens, and (y) with respect to the Pledged Collateral, the Permitted Liens described in clause (B) or (C) of the definition thereof, in either case, exists or will exist upon such Collateral at any time.
- (b) This Agreement is the legal, valid and binding obligation of such Grantor, enforceable against such Grantor in accordance with its terms except to the extent that such enforceability is subject to applicable bankruptcy, insolvency, reorganization, fraudulent conveyance and moratorium laws and other laws of general application affecting enforcement of creditors' rights generally, or the availability of equitable remedies, which are subject to the discretion of the court before which an action may be brought.
- (c) This Agreement is effective to create in favor of the Secured Party a valid security interest in and Lien upon all of such Grantor's right, title and interest in and to the Collateral, and upon (i) the filing of appropriate UCC financing statements in the jurisdictions listed on Schedule I attached hereto, (ii) each Deposit Account (other than any Excluded Accounts) being subject to a Control Agreement (as hereinafter defined) among the applicable Grantor, depository institution and the Secured Party on behalf of the Holders, (iii) filings in the United States Patent and Trademark Office or United States Copyright Office with respect to Collateral that is Patents, Trademarks or Copyrights, as the case may be, (iv) the filing of the Mortgages in the jurisdictions listed on Schedule I hereto, (v) the delivery to the Secured Party of the Pledged Collateral together with assignments in blank, (vi) the security interest created hereby being noted on each certificate of title evidencing the ownership of any Motor Vehicle (which shall only be required to be noted on a certificate of title if required by Section 4.1(d) hereof), (vii) delivery to the Secured Party or its Representative of Instruments duly endorsed by the applicable Grantor or accompanied by appropriate instruments of transfer duly executed by the applicable Grantor with respect to Instruments not constituting Chattel Paper and (viii) the consent of the issuer and any confirmer of any letter of credit to an assignment to the Secured Party of the proceeds of any drawing thereunder, such security interest will be a duly perfected first priority security interest (subject only to Permitted Liens) in all of the Collateral. No consent, approval or authorization of or designation or filing with any Governmental Entity on the part of any Grantor is required in connection with the pledge and security interest granted under this Agreement (other than (x) any consent or approval which has been obtained and is in full force and effect and (y) the filings descri
- (d) The execution, delivery and performance of this Agreement will not violate (i) any material provision of any Applicable Law, (ii) any order, judgment, writ, award or decree of any court, arbitrator or governmental authority, which are applicable to any Grantor, (iii) the articles or certificate of incorporation, certificate of formation, bylaws or any other similar organizational documents of any Grantor or any Pledged Entity or of any securities issued by any Grantor or any Pledged Entity, (iv) any mortgage, indenture, lease, contract, or other agreement, instrument or undertaking to which any Grantor or any Pledged Entity is a party or which is binding upon any Grantor or any Pledged Entity or upon any of the assets of any Grantor or any Pledged Entity, and will not result in the creation or imposition of any lien, charge or encumbrance on or security interest in any of the assets of any Grantor or any Pledged Entity, except as otherwise contemplated by this Agreement.
- (e) All of the Equipment, Inventory and Goods with a value in excess of \$50,000 individually or in the aggregate owned by any Grantor is located at the places as specified on

Schedule I attached hereto (other than locations where such Equipment, Inventory and Goods is temporarily located for maintenance or repair and locations in transit). Except as disclosed on Schedule I, none of the Collateral is in the possession of any bailee, warehousemen, processor or consignee. Schedule I discloses each Grantor's name as of the date hereof as it appears in official filings in the state or province, as applicable, of its incorporation, formation or organization, the type of entity of such Grantor (including corporation, partnership, limited partnership or limited liability company), organizational identification number issued by such Grantor's state of incorporation, formation or organization (or a statement that no such number has been issued), such Grantor's state or province, as applicable, of incorporation, formation or organization and the chief place of business, chief executive office and the office where such Grantor keeps its books and records and the states in which such Grantor conducts its business. Such Grantor has only one state or province, as applicable, of incorporation, formation or organization. Such Grantor does not do business and has not done business during the past five years under any trade name or fictitious business name, and has not changed its jurisdiction of incorporation, formation or organization or its corporate structure in any way, except as disclosed on Schedule II attached hereto.

- (f) To such Grantor's knowledge, no Copyrights, Patents, Intellectual Property Licenses or Trademarks listed on Schedules III, IV, V and VI attached hereto, respectively, if any, have been adjudged invalid or unenforceable or have been canceled, in whole or in part, or are not presently subsisting. To such Grantor's knowledge and as of the date hereof, each of such Copyrights, Patents, Intellectual Property Licenses and Trademarks (if any) is valid and enforceable. To such Grantor's knowledge, such Grantor is the sole and exclusive owner of the entire and unencumbered right, title and interest in and to each of such Copyrights, Patents, Intellectual Property Licenses and Trademarks, identified on Schedules III, IV, V and VI, as applicable, as being owned by such Grantor, free and clear of any liens, charges and encumbrances, including without limitation licenses, shop rights and covenants by such Grantor not to sue third persons, other than Permitted Liens and Permitted Intellectual Property Licenses. Such Grantor has adopted, used and is currently using, or has a current bona fide intention to use, all of the Trademarks and Copyrights listed on Schedules III and VI, respectively. Such Grantor has not received written notice of any suits or actions commenced or threatened with reference to the Copyrights, Patents or Trademarks owned by it. To such Grantor's knowledge, such Grantor's use of the Intellectual Property owned by such Grantor is not interfering with, infringing upon, misappropriating, or otherwise in conflict with the Intellectual Property rights of any third party.
- (g) Without duplication of any information required to be delivered by any Grantor to the Secured Party under and in accordance with the terms of the Notes, then subject to Section 2(q), each Grantor agrees to deliver to the Secured Party (x) an updated Schedule I, II, VII and/or VIII within 20 Business Days of any change thereto and (y) an updated Schedule III, IV, V and/or VI in the case of any change thereto within 30 days after the end of each fiscal quarter.
- (h) All depositary and other accounts including, without limitation, Deposit Accounts, securities accounts, brokerage accounts and other similar accounts, maintained by any Grantor (other than any Excluded Accounts) as are described on Schedule VII hereto, which description includes for each such account the name of such Grantor maintaining such account, the name of the financial institution at which such account is maintained and the account number of such account. No Grantor shall open any new Deposit Accounts, securities accounts, brokerage accounts or other accounts (other than any Excluded Accounts) unless such Grantor shall have given the Secured Party prior written notice of its intention to open any such new accounts. Subject to Section 2(r), each Grantor shall deliver to the Secured Party a revised version of Schedule VII showing any changes thereto promptly following, but in any event within 10 Business Days of, any such change. Each Grantor hereby authorizes the financial institutions at which such Grantor maintains an account to provide Secured Party with such information with respect to such account as the Secured Party from time to time reasonably may request, and each Grantor hereby consents to such information being provided to the Secured Party. In addition, all of such Grantor's depositary, security, brokerage and other accounts including, without limitation, Deposit Accounts (other than any Excluded Accounts) shall be subject to the provisions of Section 4.5 hereof.
- (i) No Grantor owns any Commercial Tort Claims having a value in excess of \$50,000 individually or in the aggregate except for those disclosed on <a href="Schedule VIII">Schedule VIII</a> hereto (if any).
- (j) No Grantor has any interest in real property except as disclosed on <u>Schedule IX</u> (if any). Subject to <u>Section 2(r)</u>, such Grantor shall deliver to the Secured Party a revised

version of <u>Schedule IX</u> showing any changes thereto within 30 days of any such change. Except as otherwise agreed to by the Secured Party, all such interests that are fee interests in real property are, or within 30 days after such Grantor acquires any such fee interest shall be, subject to a Mortgage in favor of the Secured Party.

- (i) Each Grantor shall duly and properly record each interest in real property held by such Grantor that is required to be subject to a Mortgage, except with respect to easements, rights of way, access agreements, surface damage agreements, surface use agreements or similar agreements that such Grantor, in good faith, using prudent customs and practices in the industry in which it operates, does not believe are of material value or material to the operation of such Grantor's business or, with respect to state and federal rights of way, are not capable of being recorded as a matter of state and federal law.
- (ii) Each Grantor shall cause a title insurance company reasonably satisfactory to the Secured Party to issue, in respect of each mortgaged real property interest (including any additional real property interest (whether fee, leasehold or otherwise) that is required to be subject to a Mortgage), a mortgagee's title insurance policy (or policies) or marked up unconditional binder for such insurance or unconditional commitment to issue a title policy for such insurance. Each such policy shall (1) be in an amount satisfactory to the Secured Party; (2) insure that the Mortgage insured thereby creates a valid first Lien on, and security interest in, such mortgaged real property interest free and clear of all defects and encumbrances, except for Permitted Liens; (3) name the Collateral Agent, for the benefit of the Holders, as the insured thereunder; (4) be in the form of ALTA Loan Policy acceptable to the Collateral Agent; and (5) contain such endorsements and affirmative coverage as the Collateral Agent may reasonably request, each in form and substance reasonably acceptable to the Collateral Agent.
- (k) All Equipment (including, without limitation, Motor Vehicles) owned by each Grantor and subject to a certificate of title or ownership statute is described on <u>Schedule X</u> hereto.
  - (l) None of the Collateral constitutes, or is the Proceeds of, "farm products" (as defined in the UCC).
  - (m) No Grantor owns any "as extracted collateral" (as defined in the UCC) or any timber to be cut.
- (n) All actions and consents, including all filings, notices, registrations and recordings necessary for the exercise by the Secured Party of the rights provided for in this Agreement or the exercise of remedies in respect of the Collateral have been made or obtained, other than those required under federal and state securities laws (in each case, with respect only to the exercise of remedies).
- (o) Exhibit A sets forth (i) the authorized capital stock and other equity interests of each Pledged Entity, (ii) the number of shares of capital stock and other equity interests of each Pledged Entity that are issued and outstanding as of the date hereof and (iii) the percentage of the issued and outstanding shares of capital stock and other equity interests of each Pledged Entity held by each Grantor. Each Grantor is the record and beneficial owner of, and has good and marketable title to, the Pledged Interests, and such shares are and will remain free and clear of all pledges, liens, security interests and other encumbrances and restrictions whatsoever, except the liens and security interests in favor of the Secured Party created by this Agreement and any Permitted Lien described in clause (B) or (C) of the definition thereof.
- (p) Except as set forth on Exhibit A, there are no outstanding options, warrants or other similar agreements with respect to the Pledged Interests or any of the other Collateral.
- (q) The Pledged Interests have been duly and validly authorized and issued, are fully paid and non-assessable, and the Pledged Interests listed on Exhibit A constitute all of the issued and outstanding capital stock or other equity interests of the Pledged Entities.
- (r) Upon delivery by any Grantor to the Secured Party of any updated schedule required to be delivered pursuant to this Section 2, unless such Grantor has in good faith determined

that the matters contained in such updated schedule do not constitute material, nonpublic information relating to such Grantor or any of its Subsidiaries, such Grantor shall on or prior to 9:00 am, New York city time on the Business Day immediately following such updated schedule delivery date, publicly disclose such material, non-public information on a Form 8-K or otherwise. In the event that such Grantor believes that any updated schedule required to be delivered pursuant to this Section 2 contains material, non-public information relating to such Grantor or any of its Subsidiaries, such Grantor so shall indicate to the Secured Party explicitly in writing concurrently with the delivery of such updated schedule, and in the absence of any such written indication, the Secured Party shall be entitled to presume that information contained in such updated schedule does not constitute material, non-public information relating to such Grantor or any of its Subsidiaries.

#### Section 3. <u>Collateral</u>.

(a) As collateral security for the prompt payment and performance in full when due (whether at stated maturity, by acceleration or otherwise) of the Obligations, each Grantor hereby pledges and grants to the Secured Party (for the ratable benefit of the Holders) a Lien on and security interest in all of such Grantor's right, title and interest in the following properties and assets of such Grantor, whether now owned by such Grantor or hereafter acquired and whether now existing or hereafter coming into existence and wherever located (all being collectively referred to herein as "Collateral"):

- i) all Instruments, together with all payments thereon or thereunder, and Letter-of-Credit Rights;
- ii) all Accounts;
- iii) all Inventory;
- iv) all General Intangibles (including Software);
- v) all Equipment;
- vi) all Documents;
- vii) all Contracts;
- viii) all Goods;
- ix) all Investment Property;
- x) all Deposit Accounts and the balance from time to time in all bank accounts maintained by such Grantor;
- xi) all Commercial Tort Claims specified on Schedule VIII;
- xii) all Intellectual Property;

xiii) all Chattel Paper, all amounts payable thereunder, all rights and remedies of such Grantor thereunder including but limited to the right to amend, grant waivers and declare defaults, any and all accounts evidenced thereby, any guarantee thereof, and all collections and monies due or to become due or received by any Person in payment of any of the foregoing;

- xiv) all Receivables and Receivable Records;
- xv) all Insurance;
- xvi) all Pledged Collateral;

xvii) to the extent not otherwise included above, all Collateral Records, Collateral Support and Supporting Obligations relating to any of the foregoing; and

xviii) all other tangible and intangible property of such Grantor, including, without limitation, all fee interests in real property, Proceeds, tort claims, products, accessions, rents, profits, income, benefits, substitutions, additions and replacements of and to any of the property of such Grantor described in the preceding clauses of this Section 3 (including, without limitation, any proceeds of insurance thereon, insurance claims and all rights, claims and benefits against any Person relating thereto), other rights to payments not otherwise included in the foregoing, and all books, correspondence, files, records, invoices and other papers, including without limitation all tapes, cards, computer runs, computer programs, computer files and other papers, documents and records in the possession or under the control of such Grantor, any computer bureau or service company from time to time acting for such Grantor.

Notwithstanding anything to the contrary in this Agreement, this Agreement shall not constitute a grant of a security interest in, and the term "Collateral" shall be deemed to exclude, all of the following property (the "Excluded Collateral"): (A) any intent-to-use trademark applications filed pursuant to Section 1(b) of the Lanham Act, 15 U.S.C. §1051, to the extent that, and solely during the period in which, the grant of a security interest therein would otherwise invalidate any Grantor's right, title or interest therein, (B) any property owned by a Grantor that is subject to a purchase money Lien or a "capital lease" in accordance with GAAP permitted hereunder or under the Note Documents if the contractual obligation pursuant to which such Lien is granted (or the document providing for such capital lease) prohibits the creation of a Lien thereon or expressly requires the consent of any person other than a Grantor, unless such consent has been obtained or such prohibitions otherwise cease to exist, in which case such Collateral shall automatically become subject to the security interest granted hereunder, (C) any General Intangibles or other rights, in each case arising under any contracts, instruments, licenses or other document or give any other party to such contract, instrument, license or other document or give any other party to such contract, instrument, license or other document the right to terminate its obligations thereunder, (D) any asset, the granting of a security interest in which would be void or illegal under any applicable governmental law, rule or regulation, or pursuant thereto would result in, or permit the termination of, such asset, provided, that the property described in clauses (C) and (D) above shall only be excluded from the term "Collateral" to the extent the conditions stated therein are not rendered ineffective pursuant to Sections 9-406, 9-407, 9-408 or 9-409 of the UCC or any other Applicable Law, and (E) the Excluded Accounts. Notwithstanding the foregoing, all Proceeds of t

- (b) The security interest granted under this Section does not constitute and is not intended to result in a creation or an assumption by the Secured Party of any obligation of any Grantor or any other Person in connection with any or all of the Collateral or under any agreement or instrument relating thereto. Anything herein to the contrary notwithstanding, (i) the exercise by the Secured Party of any of its rights in the Collateral shall not release any Grantor from any of its duties or obligations in respect of the Collateral other than any duties and obligations arising with respect to Collateral after such Grantor has been dispossessed of such Collateral by the Secured Party (or its assignee), which, by their nature, may not be satisfied without possession of such Collateral and (ii) the Secured Party shall not have any obligations or liability in respect of the Collateral by reason of this Agreement, nor shall the Secured Party be obligated to perform any of the obligations or duties of any Grantor thereunder or to take any action to collect or enforce any claim for payment assigned hereunder.
- Section 4. <u>Covenants; Remedies</u>. In furtherance of the grant of the pledge and security interest pursuant to <u>Section 3</u> hereof, each Grantor hereby agrees with the Secured Party as follows:
  - 1.1 Delivery and Other Perfection; Maintenance, etc.
- (a) <u>Delivery of Instruments, Documents, Etc.</u> Each Grantor shall deliver and pledge to the Secured Party or its Representative any and all Instruments, negotiable Documents and Chattel Paper evidencing amounts greater than \$50,000 individually or in the aggregate for any such Instruments, negotiable Documents and Chattel Paper of such Grantor and certificated securities accompanied by stock/membership interest powers executed in blank, which stock/membership interest powers may be filled in and completed at any time upon the occurrence and during the continuance of any Event of Default duly endorsed and/or accompanied by such instruments of assignment and transfer executed by such Grantor in such form and substance as the Secured Party or

its Representative may request; provided, that so long as no Event of Default shall have occurred and be continuing, each Grantor may retain for collection in the ordinary course of business any Instruments, negotiable Documents and Chattel Paper received by such Grantor in the ordinary course of business, and the Secured Party or its Representative shall, promptly upon request of a Grantor, make appropriate arrangements for making any other Instruments, negotiable Documents and Chattel Paper pledged by such Grantor available to such Grantor for purposes of presentation, collection or renewal (any such arrangement to be effected, to the extent deemed appropriate by the Secured Party or its Representative, against a trust receipt or like document). If a Grantor retains possession of any Chattel Paper, negotiable Documents or Instruments evidencing amounts greater than \$50,000 individually or in the aggregate for any such Instruments, negotiable Documents and Chattel Paper of such Grantor at any time pursuant to the terms hereof, such Chattel Paper, negotiable Documents and Instruments shall be marked with the following legend: "This writing and the obligations evidenced or secured hereby are subject to the security interest of High Trail Investments ON LLC, in its capacity as agent for one or more creditors, as Secured Party."

- (b) Other Documents and Actions. Each Grantor shall give, execute, deliver, file and/or record any financing statement, registration, notice, instrument, document, agreement, Mortgage or other papers that may be necessary (as determined in the reasonable judgment of the Secured Party or its Representative) to create, preserve, perfect or validate the security interest granted pursuant hereto (or any security interest or mortgage contemplated or required hereunder, including with respect to Section 2(j) of this Agreement) or to enable the Secured Party or its Representative to exercise and enforce the rights of the Secured Party hereunder with respect to such pledge and security interest; provided that notices to account debtors in respect of any Accounts or Instruments shall be subject to the provisions of clause (e) below. Notwithstanding the foregoing each Grantor hereby irrevocably authorizes the Secured Party at any time and from time to time to file in any filing office in any jurisdiction any initial financing statements (and other similar filings or registrations under any Applicable Laws and regulations pertaining to the creation, attachment, or perfection of security interests) and amendments thereto that (a) indicate the Collateral (i) as all assets of such Grantor or words of similar effect, regardless of whether any particular asset comprised in the Collateral falls within the scope of Article 9 of the UCC or (ii) as being of an equal or lesser scope or with greater detail, and (b) contain any other information required by part 5 of Article 9 of the UCC for the sufficiency or filing office acceptance of any financing statement or amendment, including (i) whether such Grantor is an organization, the type of organization and any organization identification number issued to such Grantor agrees to furnish any such information to the Secured Party promptly upon request.
- Books and Records; Inspections. Each Grantor shall maintain at its own cost and expense, in accordance with sound business practices, complete and accurate, in all material respects, books and records of the Collateral, including, without limitation, a record of all payments received and all credits granted with respect to the Collateral and all other dealings with the Collateral. Upon the occurrence and during the continuation of any Event of Default, such Grantor shall deliver and turn over any such books and records (or true and correct copies thereof) to the Secured Party or its Representative at any time following 5 days' prior written notice. Such Grantor shall permit, at reasonable times during business hours and following 5 day's prior written notice, the Secured Party or its Representative to: (i) inspect the properties and operations of such Grantor or any Pledged Entity; (ii) visit any or all of its offices, to discuss its financial matters with its directors or officers; (iii) examine (and, at the expense of the Grantors, photocopy extracts from) any of its books or other records; and (iv)(A) inspect the Collateral and other tangible assets of such Grantor or any Pledged Entity, (B) perform appraisals of the equipment of such Grantor or any Pledged Entity and (C) inspect, audit, check and make copies of and extracts from the books, records, computer data, computer programs, journals, orders, receipts, correspondence and other data relating to any Collateral, for purposes of or otherwise in connection with conducting a review, audit or appraisal of such books and records. Each Grantor will pay the Secured Party the reasonable out-of-pocket costs and expenses of any audit or inspection of the Collateral promptly after receiving the invoice; provided that such Grantor shall not be required to reimburse the Secured Party for the foregoing expenses relating to more than one such inspection or audit in any calendar year unless an Event of Default has occurred and is continuing, in which event such Grantor shall be required to reimburse the Secured Party for any and all of the foregoing expenses. Notwithstanding anything contained in this Section 4.1(c) to the contrary, if an Event of Default shall have occurred and be continuing, then the Secured Party or its Representative may take any of the actions specified in clauses (i) through (iv) of

this Section 4.1(c) without prior notice to such Grantor, but shall endeavor in good faith to provide such Grantor subsequent notice.

- (d) Motor Vehicles. Each Grantor shall, promptly upon acquiring same, cause the Secured Party to be listed as the lienholder on each certificate of title or ownership covering any items of Motor Vehicles having a value in excess of \$50,000 individually or in the aggregate for all such Motor Vehicles of the Pledgor or any of its Subsidiaries, or otherwise comply with the certificate of title or ownership laws of the relevant jurisdiction issuing such certificate of title or ownership in order to properly evidence and perfect the Secured Party's security interest in the assets represented by such certificate of title or ownership. The Secured Party will, promptly after receipt of written request therefor from such Grantor, return any such certificate of title as needed by such Grantor to maintain the registration and licensing of such Equipment and Motor Vehicles, and such Grantor shall promptly return such certificates of title to Secured Party upon completion of such registration and licensing requirements.
- (e) <u>Notice to Account Debtors; Verification</u>. (i) Upon the occurrence and during the continuance of any Event of Default, upon request of the Secured Party or its Representative, each Grantor shall promptly notify (and each Grantor hereby authorizes the Secured Party and its Representative so to notify) each account debtor in respect of any Accounts or Instruments or other Persons obligated on the Collateral that such Collateral has been assigned to the Secured Party hereunder, and that any payments due or to become due in respect of such Collateral are to be made directly to the Secured Party, and (ii) the Secured Party and its Representative shall have the right at any time or times to make direct verification with the account debtors or other Persons obligated on the Collateral of any and all of the Accounts or other such Collateral.
- Intellectual Property. Each Grantor represents and warrants that the Copyrights, Patents, and Trademarks listed on Schedules III, V and VI, respectively (if any), constitute all of the registered Copyrights and all of the issued or applied-for Patents, and registered or applied-for Trademarks owned by such Grantor as of the date hereof or licensed to Grantor under any material Intellectual Property License as of the date hereof. Each Grantor represents and warrants that (i) each material Intellectual Property License is in full force and effect and constitutes a legal, valid, and binding obligation of the applicable Grantor and each other party thereto, and is enforceable in accordance with its terms, and (ii) neither the applicable Grantor nor, to such Grantor's knowledge, any other party to any material Intellectual Property License is, and no person has provided written notice to the applicable Grantor alleging it to be, in default or breach under any such material Intellectual Property License, and no person has provided written notice to the applicable Grantor indicating an intention to terminate (including by non-renewal) any such material Intellectual Property License. If such Grantor shall obtain ownership of or other right, title or interest in or to, or otherwise become entitled to the benefit of, any registered Copyrights, issued or applied-for Patents, registered or applied-for Trademarks, Intellectual Property Licenses or any other Intellectual Property (including, for the avoidance of doubt, any improvements to any existing Intellectual Property), the provisions of this Agreement above shall automatically apply thereto and such Grantor shall promptly give to the Secured Party notice with respect to any such registered Copyrights, issued or applied-for Patents, registered or applied-for Trademarks, or Intellectual Property Licenses. Each Grantor hereby authorizes the Secured Party to modify this Agreement by amending Schedules III, IV, V and VI, as applicable, to include any such property in any such notice. Each Grantor shall (i) prosecute diligently any patent, trademark or service mark applications pending as of the date hereof or hereafter to the extent material to the operations of the business of such Grantor, (ii) preserve and maintain all rights in the Copyrights, Patents, Intellectual Property Licenses and Trademarks, to the extent material to the operations of the business of such Grantor and (iii) ensure that the Copyrights, Patents, Intellectual Property Licenses and Trademarks are and remain enforceable, to the extent material to the operations of the business of such Grantor. Any expenses incurred in connection with such Grantor's obligations under this Section 4.1(f) shall be borne by such Grantor. Except for any such items that such Grantor reasonably believes in good faith are no longer necessary for the on-going operations of its business, such Grantor shall not abandon any material registered Copyright, issued or applied-for Patent, or registered or applied-for Trademark, or permit any right to apply for the registration of any material Intellectual Property to lapse or expire, without the prior written consent of the Secured Party, which consent shall not be unreasonably withheld, conditioned or delayed. Each Grantor represents that all Intellectual Property license agreements pursuant to which such Grantor is a licensee or licensor are in writing.

- (g) <u>Further Identification of Collateral</u>. Each Grantor will, when and as often as requested by the Secured Party or its Representative (but, absent the occurrence and continuance of an Event of Default, in no event more frequently than quarterly), furnish to the Secured Party or such Representative, statements and schedules further identifying and describing the Collateral and such other reports in connection with the Collateral as the Secured Party or its Representative may reasonably request, all in reasonable detail.
- (h) Investment Property. Each Grantor will promptly take any and all actions required or requested by the Secured Party, from time to time, to cause the Secured Party to obtain exclusive control of any Investment Property owned by such Grantor. For purposes of this Section 4.1(h), the Secured Party shall have exclusive control of Investment Property if (i) such Investment Property consists of certificated securities and such Grantor delivers such certificated securities to the Secured Party (with assignments in blank or appropriate endorsements if such certificated securities are in registered form); (ii) such Investment Property consists of uncertificated securities and the issuer thereof agrees, pursuant to documentation in form and substance reasonably satisfactory to the Secured Party, that it will comply with instructions originated by the Secured Party without further consent by such Grantor, and (iii) such Investment Property consists of security entitlements and either (x) the Secured Party becomes the entitlement holder thereof or (y) the appropriate securities intermediary agrees, pursuant to the documentation in form and substance reasonably satisfactory to the Secured Party, that it will comply with entitlement orders originated by the Secured Party without further consent by such Grantor.
- (i) <u>Commercial Tort Claims</u>. Each Grantor shall promptly notify the Secured Party of any Commercial Tort Claims acquired by it that concerns claims in excess of \$50,000 individually or in the aggregate for any such Commercial Tort Claims of the Pledgor and any of its Subsidiaries and if requested by the Secured Party, such Grantor shall enter into a supplement to this Agreement granting to the Secured Party a Lien on and security interest in such Commercial Tort Claim.
- (j) <u>Pledge Supplement</u>. Within five (5) Business Days of the creation or acquisition of any new Pledged Interests, each Grantor shall execute a supplement to <u>Exhibit A</u> in the form of <u>Annex II (a</u> "**Pledge Supplement**") and deliver such Pledge Supplement to the Secured Party. Any Pledged Collateral described in a Pledge Supplement delivered by such Grantor shall thereafter be deemed to be listed on <u>Exhibit A</u> hereto.
- 1.2 <u>Preservation of Rights</u>. Whether or not an Event of Default has occurred or is continuing, the Secured Party and its Representative shall have the right to take any steps the Secured Party or its Representative reasonably deems necessary or appropriate to preserve any Collateral or any rights against third parties to any of the Collateral upon any Grantor's failure to do so, including obtaining insurance for the Collateral at any time when such Grantor has failed to do so, and such Grantor shall promptly pay, or reimburse the Secured Party for, all reasonable and customary out-of-pocket expenses incurred in connection therewith.

#### 1.3 Name Change; Location; Bailees.

- (a) No Grantor shall form or acquire any subsidiary other than in accordance with any express terms of the Note Documents.
- (b) Each Grantor shall provide the Secured Party at least 10 Business Days prior written notice of (i) any reincorporation or reorganization of itself under the laws of any jurisdiction other than the jurisdiction in which it is incorporated or organized as of the date hereof, and/or (ii) any change of its name, identity or corporate structure.
- (c) Except for the sale of Inventory in the ordinary course of business, other sales of assets expressly permitted by the terms of the Note Documents and except for Collateral temporarily located for maintenance or repair (so long as such Grantor shall promptly provide the Secured Party with written notice of such temporary location), such Grantor will keep Collateral with a value in excess of \$100,000 individually or \$250,000 in the aggregate at the locations specified in <a href="Schedule I">Schedule I</a> attached hereto. Such Grantor will give the Secured Party 10 Business Days prior written notice before any change in such Grantor's chief place of business or of any new location for any of the Collateral with a value in excess of \$100,000 individually or \$250,000 in the aggregate for any Collateral granted in favor of the Secured Party by such Grantor.

- 1.4 Other Liens. No Grantor will create, permit or suffer to exist, and each Grantor will defend the Collateral against and take such other action as is necessary to remove, any Lien on the Collateral except Permitted Liens, and each Grantor will defend the right, title and interest of the Secured Party in and to the Collateral and in and to all Proceeds thereof against the claims and demands of all Persons whatsoever, subject to Permitted Liens.
- (a) If any Collateral with a value in excess of \$250,000 in the aggregate for any Collateral granted in favor of the Secured Party by such Grantor is at any time in the possession or control of any warehousemen, bailee, consignee or processor, the applicable Grantor shall promptly notify the Secured Party of such fact and, upon the request of the Secured Party or its Representative, notify such warehousemen, bailee, consignee or processor of the Lien and security interest created hereby and shall instruct such Person to hold all such Collateral for the Secured Party's account subject to the Secured Party's instructions.
- (b) Each Grantor acknowledges that it is not authorized to file any financing statement or amendment or termination statement with respect to any financing statement naming such Grantor, as debtor, and the Secured Party, as secured party, without the prior written consent of the Secured Party and agrees that it will not do so without the prior written consent of the Secured Party, subject to such Grantor's rights under Section 9-509(d)(2) to the UCC.
- 1.5 <u>Bank Accounts and Securities Accounts</u>. On or prior to the date hereof, each Grantor, as applicable, shall enter into an account control agreement or securities account control agreement, as applicable, in form and substance reasonably satisfactory to the Secured Party (each a "Control Agreement"), with Secured Party and each financial institution with which each such Grantor maintains from time to time any Deposit Accounts (general or special), securities accounts, brokerage accounts or other similar accounts (other than Excluded Accounts), which financial institutions are set forth on Schedule VII attached hereto. Pursuant to this Agreement, each Grantor grants and shall grant to the Secured Party a continuing lien upon, and security interest in, all such accounts (other than Excluded Accounts) and all funds at any time paid, deposited, credited or held in such accounts (whether for collection, provisionally or otherwise) or otherwise in the possession of such financial institutions. Following the date hereof, no Grantor shall establish any Deposit Account, securities account, brokerage account or other similar account (other than Excluded Accounts) with any financial institution unless prior or concurrently thereto the Secured Party and such Grantor shall have entered into a Control Agreement with such financial institution which purports to cover such account. Each Grantor shall deposit and keep on deposit all of its funds in a Deposit Account (other than Excluded Accounts) or Securities Account which is subject to a Control Agreement with Secured Party.
  - 1.6 Events of Default, Etc. During the period during which an Event of Default shall have occurred and be continuing:
- (a) each Grantor shall, at the request of the Secured Party or its Representative, assemble the Collateral and make it available to the Secured Party or its Representative at a place or places designated by the Secured Party or its Representative which are reasonably convenient to the Secured Party or its Representative, as applicable, and such Grantor;
- (b) the Secured Party or its Representative may make any reasonable compromise or settlement deemed desirable with respect to any of the Collateral and may extend the time of payment, arrange for payment in installments, or otherwise modify the terms of, any of the Collateral;
- (c) the Secured Party shall have all of the rights and remedies with respect to the Collateral of a secured party under the UCC (whether or not said UCC is in effect in the jurisdiction where the rights and remedies are asserted) and such additional rights and remedies to which a secured party is entitled under the laws in effect in any jurisdiction where any rights and remedies hereunder may be asserted, including, without limitation, the right, to the maximum extent permitted by law, to exercise all voting, consensual and other powers of ownership pertaining to the Collateral in accordance with this Agreement and the other Note Documents as if the Secured Party were the sole and absolute owner thereof (and each Grantor agrees to take all such action as may be appropriate to give effect to such right);

- (d) the Secured Party or its Representative shall have the right, in the name of the Secured Party or in the name of a Grantor or otherwise, to demand, sue for, collect or receive any money or property at any time payable or receivable on account of or in exchange for any of the Collateral, but shall be under no obligation to do so;
- (e) the Secured Party or its Representative shall have the right to take immediate possession and occupancy of any premises owned, used or leased by a Grantor and exercise all other rights and remedies which may be available to the Secured Party;
- (f) the Secured Party shall have the right, upon reasonable written notice (such reasonable notice to be determined by the Secured Party in its sole and absolute discretion, which shall not be less than 10 days), with respect to the Collateral or any part thereof (whether or not the same shall then be or shall thereafter come into the possession, custody or control of the Secured Party or its Representative), to sell, lease, license, assign or otherwise dispose of all or any part of such Collateral, at such place or places as the Secured Party deems best, and for cash or for credit or for future delivery (without thereby assuming any credit risk), at public or private sale, without demand of performance or notice of intention to effect any such disposition or of the time or place thereof (except such notice as is required above or by applicable statute and cannot be waived), and the Secured Party or anyone else may be the purchaser, lessee, licensee, assignee or recipient of any or all of the Collateral so disposed of at any public sale (or, to the extent permitted by law, at any private sale) and thereafter hold the same absolutely, free from any claim or right of whatsoever kind, including any right or equity of redemption (statutory or otherwise), of the applicable Grantor, any such demand, notice and right or equity being hereby expressly waived and released. The Secured Party may, to the fullest extent permitted by Applicable Law, without notice or publication, adjourn any public or private sale or cause the same to be adjourned from time to time by announcement at the time and place fixed for the sale, and such sale may be made at any time or place to which the sale may be so adjourned.
- (g) the Secured Party may, prior to the disposition of the Collateral, store, process, repair or recondition the Collateral or otherwise prepare the Collateral for disposition in any manner to the extent the Secured Party deems appropriate;
- (h) the Secured Party may proceed to perform any and all of the obligations of any Grantor contained in any Contract and exercise any and all rights of such Grantor therein contained as such Grantor itself could;
- (i) the Secured Party shall have the right to use any Grantor's rights under any Collateral consisting of Intellectual Property Licenses in connection with the enforcement of the Secured Party's rights hereunder; and
- (j) the rights, remedies and powers conferred by this <u>Section 4.6</u> are in addition to, and not in substitution for, any other rights, remedies or powers that the Secured Party may have under any Note Document, at law, in equity or by or under the UCC or any other statute or agreement. The Secured Party may proceed by way of any action, suit or other proceeding at law or in equity and no right, remedy or power of the Secured Party will be exclusive of or dependent on any other. The Secured Party may exercise any of its rights, remedies or powers separately or in combination and at any time.

Without limiting the foregoing, the Secured Party may, without demand of performance or other demand, advertisement or notice of any kind (except the notice specified below of time and place of public or private sale) to or upon any Grantor or any other person or entity (all and each of which demands, advertisements and/or notices are hereby expressly waived), upon the occurrence and during the continuance of an Event of Default forthwith collect, receive, appropriate and realize upon the Collateral, or any part thereof, and/or may forthwith date and otherwise fill in the blanks on any assignments separate from certificates or stock power or otherwise sell, assign, give an option or options to purchase, contract to sell or otherwise dispose of and deliver said Collateral, or any part thereof, in one or more portions at one or more public or private sales or dispositions, at any exchange or broker's board or at any of the Secured Party's offices or elsewhere upon such terms and conditions as the Secured Party may deem advisable and at such prices as it may deem best, for any combination of cash and/or securities or other property or on credit or for future delivery without assumption of any credit risk, with the right of the Secured Party (or the designee of the Secured Party) upon any such sale, public or private, to purchase the whole or any part of said Collateral so sold, free of any

right or equity of redemption of the applicable Grantor, which right or equity is hereby expressly waived or released. Each Grantor agrees that, to the extent notice of sale shall be required by Applicable Law or this Agreement, at least ten (10) days' prior written notice to the applicable Grantor of the time and place of any public sale or the time after which any private sale is to be made shall constitute reasonable notification. Notwithstanding any provision in any operating agreement or shareholder agreement of any issuer of the Collateral or any other Applicable Law to the contrary, the undersigned, constituting a member and/or shareholder of each issuer hereby acknowledges that such member and/or shareholder, as applicable, may pledge to the Secured Party all of such member's and/or shareholder's right, title and interest in such issuer, and upon foreclosure the successful bidder (which may include the Secured Party or any Holder) will be deemed admitted as a member and/or shareholder, as applicable, of such issuer, and will automatically succeed to all of such pledged right, title and interest, including without limitation such members' and/or shareholder's limited liability company and equity interests, right to vote and participate in the management and business affairs of the issuer, right to a share of the profits and losses of the issuer and right to receive distributions from the issuer.

The proceeds of each collection, sale or other disposition under this Section 4.6 shall be applied in accordance with Section 4.9 hereof.

- 1.7 <u>Deficiency</u>. If the proceeds of sale, collection or other realization of or upon the Collateral are insufficient to cover the costs and expenses of such realization and the payment in full of the Obligations, the Grantors shall remain liable for any deficiency.
- Private Sale. Each Grantor recognizes that the Secured Party may be unable to effect a public sale of any or all of the Collateral consisting of securities by reason of certain prohibitions contained in the Securities Act of 1933, as amended (the "Act"), and applicable state securities laws, but may be compelled to resort to one or more private sales thereof to a restricted group of purchasers who will be obliged to agree, among other things, to acquire such Collateral for their own account for investment and not with a view to the distribution or resale thereof. Each Grantor acknowledges and agrees that any such private sale may result in prices and other terms less favorable to the seller than if such sale were a public sale and such Grantor agrees that it is not commercially unreasonable for the Secured Party to engage in any such private sales or dispositions under such circumstances. Each Grantor agrees that it would not be commercially unreasonable for the Secured Party to dispose of the Collateral or any portion thereof by using Internet sites that provide for the auction of assets of the types included in the Collateral or that have the reasonable capability of doing so, or that match buyers and sellers of assets. The Secured Party may sell the Collateral without giving any warranties as to the Collateral. The Secured Party may specifically disclaim or modify any warranties of title or the like. This procedure will not be considered to adversely affect the commercial reasonableness of any sale of the Collateral. The Secured Party shall be under no obligation to delay a sale of any of the Collateral to permit any Grantor to register such Collateral for public sale under the Act, or under applicable state securities laws, even if such Grantor would agree to do so. The Secured Party shall not incur any liability as a result of the sale of any such Collateral, or any part thereof, at any private sale provided for in this Agreement and each Grantor hereby waives any claims against the Secured Party arising by reason of the fact that the price at which the Collateral may have been sold at such a private sale was less than the price which might have been obtained at a public sale or was less than the aggregate amount of the Obligations, even if the Secured Party accepts the first offer received and does not offer the Collateral to more than one offeree. The Secured Party may sell the Collateral without giving any warranties as to the Collateral. The Secured Party may specifically disclaim or modify any warranties of title or the like. This procedure will not be considered to adversely affect the commercial reasonableness of any sale of the Collateral.

Each Grantor further agrees to do or cause to be done all such other acts and things as may be necessary to make such sale or sales of any portion or all of any such Collateral valid and binding and in compliance with any and all applicable laws, regulations, orders, writs, injunctions, decrees or awards of any and all courts, arbitrators or governmental instrumentalities, domestic or foreign, having jurisdiction over any such sale or sales, all at such Grantor's expense. Each Grantor further agrees that a breach of any of the covenants contained in this Section 4.8 will cause irreparable injury to the Secured Party, that the Secured Party has no adequate remedy at law in respect of such breach and, as a consequence, agrees that each and every covenant contained in this Section 4.8 shall be specifically enforceable against such Grantor, and each Grantor hereby waives and agrees not to assert any defenses against an action for specific performance of such covenants except for a defense that no Event of Default has occurred and is continuing.

Each Grantor further agrees not to exercise any and all rights of subrogation it may have against a Pledged Entity upon the sale or disposition of all or any portion of the Pledged Collateral by the Secured Party pursuant to the terms of this Agreement until the termination of this Agreement in accordance with Section 4.12.

- 4.9 <u>Application of Proceeds</u>. The proceeds of any collection, sale or other realization of all or any part of the Collateral following the occurrence and during the continuance of an Event of Default, and any other cash at the time held by the Secured Party under this Agreement, shall be applied, first to the payment of all fees and expenses or other liabilities of any kind payable to the Secured Party in connection with the Note Documents (solely in its capacity as Collateral Agent and not in its capacity as a Holder or any other capacity) and second, to the Trustee for payment in accordance with Section 6.10 of the Base Indenture.
- 1.9 Attorney-in-Fact. Each Grantor hereby irrevocably constitutes and appoints the Secured Party, with full power of substitution, as its true and lawful attorney-in-fact with full irrevocable power and authority in the place and stead of such Grantor and in the name of such Grantor or in its own name, from time to time upon the occurrence and during the continuance of an Event of Default in the discretion of the Secured Party, for the purpose of carrying out the terms of this Agreement, to take any and all appropriate action and to execute and deliver any and all documents and instruments which may be necessary to perfect or protect any security interest granted hereunder, to maintain the perfection or priority of any security interest granted hereunder, and, without limiting the generality of the foregoing, hereby gives the Secured Party the power and right, on behalf of such Grantor, without notice to or assent by such Grantor (to the extent permitted by Applicable Law), to do the following upon the occurrence and during the continuation of an Event of Default:
- (a) to take any and all appropriate action and to execute and deliver any and all documents and instruments which may be necessary to accomplish the purposes of this Agreement;
- (b) to ask, demand, collect, receive and give acquittance and receipts for any and all moneys due and to become due under any Collateral and, in the name of such Grantor or its own name or otherwise, to take possession of and endorse and collect any checks, drafts, notes, acceptances or other Instruments for the payment of moneys due under any Collateral and to file any claim or to take any other action or proceeding in any court of law or equity or otherwise deemed appropriate by the Secured Party for the purpose of collecting any and all such moneys due under any Collateral whenever payable and to file any claim or to take any other action or proceeding in any court of law or equity or otherwise deemed appropriate by the Secured Party for the purpose of collecting any and all such moneys due under any Collateral whenever payable;
- (c) to pay or discharge charges or liens levied or placed on or threatened in writing against the Collateral, to effect any insurance called for by the terms of this Agreement or the Note Documents and to pay all or any part of the premiums therefor;
- (d) to direct any party liable for any payment under any of the Collateral to make payment of any and all moneys due, and to become due thereunder, directly to the Secured Party or as the Secured Party shall direct, and to receive payment of and receipt for any and all moneys, claims and other amounts due, and to become due at any time, in respect of or arising out of any Collateral;
- (e) to sign and indorse any invoices, freight or express bills, bills of lading, storage or warehouse receipts, drafts against any Grantor, assignments, verifications and notices in connection with accounts and other Documents constituting or relating to the Collateral;
- (f) to commence and prosecute any suits, actions or proceedings at law or in equity in any court of competent jurisdiction to collect the Collateral or any part thereof and to enforce any other right in respect of any Collateral;
  - (g) to defend any suit, action or proceeding brought against the Grantors with respect to any Collateral;

- (h) to settle, compromise or adjust any suit, action or proceeding described above and, in connection therewith, to give such discharges or releases as the Secured Party may deem appropriate;
- (i) to the extent that a Grantor's authorization given in <u>Section 4.1(b)</u> of this Agreement is not sufficient to file such financing statements with respect to this Agreement, with or without such Grantor's signature;
- (j) generally to sell, transfer, pledge, make any agreement with respect to or otherwise deal with any of the Collateral as fully and completely as though the Secured Party were the absolute owner thereof for all purposes;
- (k) to use any Intellectual Property or Intellectual Property Licenses of such Grantor, including but not limited to any Copyrights, Patents or Trademarks, in preparing for sale, advertising for sale, or selling Inventory or other Collateral and to collect any amounts due under Accounts of such Grantor;
- (1) in connection with the exercise of the Secured Party's rights under <u>Section 4.6</u>, to prepare, sign, and file any document which may be required by the United States Patent and Trademark Office, the United States Copyright Office or similar registrar in order to effect an absolute assignment of all right, title and interest in all registered Intellectual Property and any application for all such registrations, and record the same;
- (m) to prepare, sign, and file for recordation in any intellectual property registry, appropriate evidence of the lien and security interest granted herein in the Intellectual Property in the name of the applicable Grantor as debtor; and
- (n) to do, at the Secured Party's option and at the Grantors' expense, at any time, or from time to time, all acts and things which the Secured Party reasonably deems necessary to protect or preserve or realize upon the Collateral and the Secured Party's lien therein, in order to effect the intent of this Agreement, all as fully and effectively as the Grantors might do.

Each Grantor hereby ratifies, to the extent permitted by law, all that such attorneys lawfully do or cause to be done by virtue hereof provided the same is performed in a commercially reasonable manner. The power of attorney granted hereunder is a power coupled with an interest and shall be irrevocable until the Obligations are paid in full and this Agreement is terminated in accordance with <u>Section 4.12</u> hereof.

Each Grantor also authorizes the Secured Party, at any time from and after the occurrence and during the continuation of any Event of Default, (x) to communicate in its own name with any party to any Contract constituting Collateral with regard to the assignment of the right, title and interest of such Grantor in and under the Contracts hereunder and other matters relating thereto and (y) to execute, in connection with any sale of Collateral provided for in Section 4.6 hereof, any endorsements, assignments or other instruments of conveyance or transfer with respect to the Collateral.

- 1.10 <u>Perfection</u>. Prior to or concurrently with the execution and delivery of this Agreement, each Grantor shall:
- (a) file such financing statements, assignments for security and other documents in such offices as may be necessary to perfect the security interests granted by Section 3 of this Agreement;
- (b) at the Secured Party's request, deliver to the Secured Party or its Representative the originals of all Instruments required to be so delivered hereunder together with, in the case of Instruments constituting promissory notes, allonges attached thereto showing such promissory notes to be payable to the order of a blank payee;
- (c) deliver to the Secured Party or its Representative all certificates representing the Pledged Interests now owned by such Grantor, together with undated assignments separate from

certificates or stock/membership interest powers duly executed in blank by such Grantor and irrevocable proxies;

- (d) deliver to the Secured Party or its Representative a Mortgage with respect to all real property held by such Grantor that is required to be subject to a Mortgage;
- (e) deliver to the Secured Party or its Representative a Control Agreement for each Deposit Account (other than any Excluded Accounts) owned by such Grantor, acceptable in all respects to the Secured Party, duly executed by such Grantor and the financial institution at which such Grantor maintains such Deposit Account; and
- (f) deliver to the Secured Party or its Representative the originals of all Motor Vehicle titles with respect to Motor Vehicles having a value in excess of \$100,000 in the aggregate, duly endorsed indicating the Secured Party's interest therein as a lienholder, together with such other documents as may be required consistent with Section 4.1(d) hereof to perfect the security interest granted by Section 3 in all such Motor Vehicles (if any).
- 1.11 Termination; Partial Release of Collateral. This Agreement and the Liens and security interests granted hereunder shall continue in effect until the Obligations (other than inchoate indemnity obligations) are paid in full. When the Obligations (other than inchoate indemnity obligations) are paid in full, the security interest granted hereby shall immediately and automatically terminate and all rights to the Collateral shall revert to the applicable Grantor, and the Secured Party will promptly following such termination deliver possession of all Collateral (including, without limitation, the Pledged Interests, the other Pledged Collateral and any other property then held as part of the Pledged Collateral) to the applicable Grantor and execute and deliver to such Grantor such documents as are necessary to evidence such termination, including UCC termination statements and such other documentation as shall be reasonably requested by such Grantor to effect the termination and release of the Liens and security interests in favor of the Secured Party affecting the Collateral. Upon any sale of property, permitted by the Note Documents, to a party who is not a Grantor or a Subsidiary of a Grantor, the Liens granted herein with respect to such property shall be deemed to be automatically released and such property shall automatically revert to the applicable Grantor with no further action on the part of any Person. The Secured Party shall, at the Grantors' expense, execute and deliver or otherwise authorize the filing of such documents as any Grantor shall reasonably request, in form and substance reasonably satisfactory to the Secured Party, including financing statement amendments to evidence such release.
- 1.12 Further Assurances. At any time and from time to time, upon the written request of the Secured Party or its Representative, and at the sole expense of the Grantors, each Grantor shall promptly and duly execute and deliver any and all such further instruments, documents and agreements and take such further actions as the Secured Party or its Representative may reasonably require in order for the Secured Party to obtain the full benefits of this Agreement and of the rights and powers herein granted in favor of the Secured Party, including, without limitation, using such Grantor's commercially reasonable efforts to secure all consents and approvals necessary or appropriate for the assignment to the Secured Party of any Collateral held by such Grantor or in which such Grantor has any rights not heretofore assigned, the filing of any financing or continuation statements under the UCC with respect to the liens and security interests granted hereby, transferring Collateral to the Secured Party's possession (if a security interest in such Collateral can be perfected by possession), placing the interest of the Secured Party as lienholder on the certificate of title of any Motor Vehicle, using commercially reasonable efforts to obtain waivers of liens from landlords and mortgagees, and delivering to the Secured Party all such Control Agreements as the Secured Party or its Representative shall require (other than with respect to Excluded Accounts) duly executed by such Grantor and the financial institution at which such Grantor maintains a Deposit Account covered by such Control Agreement. Each Grantor also hereby authorizes the Secured Party and its Representative to file any such financing or continuation statement without the signature of such Grantor to the extent permitted by Applicable Law.
- 1.13 Additional Grantors. From time to time subsequent to the date hereof, if any Grantor acquires a Subsidiary, such Grantor shall (i) have provided at least five (5) Business Days' prior to such Grantor's acquisition of such Subsidiary notice of such acquisition to the Secured Party, and (ii) within thirty (30) days following the date any Grantor acquires such Subsidiary, (a) cause each such person so acquired to become a party hereto as an additional Grantor (each, an "Additional Grantor"), by executing a Joinder Agreement, together with a Pledge Supplement executed by the

applicable Grantor and any other attachments, all in form and substance reasonably satisfactory to Secured Party, (b) cause each Additional Grantor to execute a joinder to the Subsidiary Guaranty, (c) deliver each of the following documents: (x) a secretary's certificate, (y) a Lien and judgment search and (z) any financing statements or amendments to financing statements requested by the Secured Party, in the case of the items described in clauses (x) and (y), substantially in the form of such document or other item delivered under the Securities Purchase Agreement in respect of the Grantors at Closing (with any such changes reasonably satisfactory to the Secured Party), (d) if such Additional Grantor maintains any Deposit Account, securities account, brokerage account or other similar account (other than any Excluded Accounts), such Additional Grantor shall execute and deliver a Control Agreement with the applicable financial institution which purports to cover such account, (e) if reasonably requested by the Secured Party, such Grantor shall deliver an opinion of legal counsel to the Additional Grantor, in form and substance reasonably satisfactory to the Secured Party, covering the documents executed and the security interests granted by the Secured Party, (iii) if an Additional Grantor owns an interest in any real property, such Grantor and such Additional Grantor shall comply with the terms of Section 2(i) as if such real property was acquired on the date Grantor acquired such Pledged Interest, (iv) take all other such actions and execute and deliver, or cause to be executed and delivered, all such documents, instruments, agreements, and certificates reasonably requested by Secured Party, and (v) promptly following demand thereof, pay all fees and expenses of the Secured Party, including attorney's fees, incurred in connection with actions taken under this Section. Upon delivery of any such Joinder Agreement to the Secured Party, notice of which is hereby waived by each other Grantor, each Additional Grantor shall be a "Grantor" hereunder with the same force and effect as if it were originally a party to this Agreement and named as a "Grantor" hereunder. Each Grantor expressly agrees that their obligations arising hereunder shall not be affected or diminished by the addition or release of any other Grantor hereunder, nor by any election of the Secured Party not to cause any other Person to become an Additional Grantor hereunder. This Agreement shall be fully effective as to each Grantor that is or becomes a party hereto regardless of whether any other Person becomes or fails to become or ceases to be a Grantor hereunder.

1.14 <u>Limitation on Duty of Secured Party</u>. The powers conferred on the Secured Party under this Agreement are solely to protect the Secured Party's interest on behalf of itself and the Holders in the Collateral and shall not impose any duty upon it to exercise any such powers. Without in any way limiting the exculpation and indemnification provisions of the Note Documents, the Secured Party shall be accountable only for amounts that it actually receives and retains for its own account as a result of the exercise of such powers and neither the Secured Party nor its Representative nor any of their respective officers, directors, employees or agents shall be responsible to any Grantor for any act or failure to act, except for bad faith, gross negligence or willful misconduct. Without limiting the foregoing, the Secured Party and any Representative shall each be deemed to have exercised reasonable care in the custody and preservation of the Collateral in its respective possession if such Collateral is accorded treatment substantially similar to that which the relevant Secured Party or any Representative, in its individual capacity, accords its own property consisting of the type of Collateral involved, it being understood and agreed that neither the Secured Party nor any Representative shall have any responsibility for taking any necessary steps (other than steps taken in accordance with the standard of care set forth above) to preserve rights against any Person with respect to any Collateral.

Without limiting the generality of the foregoing, neither the Secured Party nor any Representative shall have any obligation or liability under any Contract or license by reason of or arising out of this Agreement or the granting to the Secured Party of a security interest therein or assignment thereof or the receipt by the Secured Party or any Representative of any payment relating to any Contract or license pursuant hereto, nor shall the Secured Party or any Representative be required or obligated in any manner to perform or fulfill any of the obligations of any Grantor under or pursuant to any Contract or license, or to make any payment, or to make any inquiry as to the nature or the sufficiency of any payment received by it or the sufficiency of any performance by any party under any Contract or license, or to present or file any claim, or to take any action to collect or enforce any performance or the payment of any amounts which may have been assigned to it or to which it may be entitled at any time or times.

1.15 <u>Dividends, Distributions, Etc.</u> If, prior to the payment in full of the Obligations (other than inchoate indemnity obligations), any Grantor shall receive any certificate (including, without limitation, any certificate representing a dividend or a distribution in connection with any reclassification, increase or reduction of capital, or issued in connection with any reorganization,

merger or consolidation), or any options or rights, whether as an addition to, in substitution for, or in exchange for any of the Pledged Interests or otherwise, such Grantor agrees, in each case, to accept the same as the Secured Party's agent and to hold the same in trust for the Secured Party, and to deliver the same promptly (but in any event within five (5) Business Days of receipt) to Secured Party in the exact form received, with the endorsement of such Grantor when necessary and/or with appropriate undated assignments separate from certificates or stock powers duly executed in blank, to be held by the Secured Party subject to the terms hereof, as additional Pledged Collateral. Each Grantor shall promptly deliver to the Secured Party (i) a Pledge Supplement with respect to such additional certificates, and (ii) any financing statements or amendments to financing statements as requested by the Secured Party. Each Grantor hereby authorizes the Secured Party to attach each such Pledge Supplement to this Agreement. Except as provided in Section 4.17(b) below, all sums of money and property so paid or distributed in respect of the Pledged Interests which are received by such Grantor shall, until paid or delivered to the Secured Party, be held by such Grantor in trust as additional Pledged Collateral.

#### 1.16 <u>Voting Rights; Dividends; Certificates.</u>

- (a) So long as no Event of Default has occurred and is continuing, each Grantor shall be entitled (subject to the other provisions hereof, including, without limitation, Section 4.18 below) to exercise its voting and other consensual rights with respect to the Pledged Interests and otherwise exercise the incidents of ownership thereof in any manner not inconsistent with this Agreement and/or any of the other Note Documents. Each Grantor hereby grants to the Secured Party or its nominee, an irrevocable proxy to exercise all voting, corporate and limited liability company rights to the Pledged Interests in any instance, which proxy shall be effective, at the discretion of the Secured Party, upon the occurrence and during the continuance of an Event of Default so long as the Secured Party has notified such Grantor in writing of its intent to exercise its voting power under this clause prior to the exercise thereof. Upon the request of the Secured Party at any time, each Grantor agrees to deliver to the Secured Party such further evidence of such irrevocable proxy or such further irrevocable proxies to vote the Pledged Interests as the Secured Party may reasonably request.
- (b) So long as no Event of Default shall have occurred and be continuing, the Grantors shall be entitled to receive cash dividends or other distributions made in respect of the Pledged Interests, to the extent permitted to be made pursuant to the terms of the Note Documents. Upon the occurrence and during the continuance of an Event of Default, in the event that any Grantor, as record and beneficial owner of the Pledged Interests, shall have received, any cash dividends or other distributions in the ordinary course, such Grantor shall deliver to the Secured Party, and the Secured Party shall be entitled to receive and retain, for the benefit of the Secured Party and the Holders, all such cash or other distributions as additional security for the Obligations.
- (c) Each Grantor shall cause all Pledged Interests required to be delivered pursuant to Section 4.1(a) to be certificated within thirty (30) days of the date of this Agreement and at all times thereafter while this Agreement is in effect and shall deliver to the Secured Party or its Representative within thirty (30) days of the date of this Agreement and at all times thereafter all certificates representing the Pledged Interests, together with undated assignments separate from certificates or stock/membership interest powers duly executed in blank by such Grantor and irrevocable proxies.
- (d) Any or all of the Pledged Interests held by the Secured Party hereunder may, if an Event of Default has occurred and is continuing and so long as the Secured Party has notified the applicable Grantor in writing of its intent to exercise its power of registration under this sentence prior to the exercise thereof, be registered in the name of Secured Party or its nominee, and the Secured Party or its nominee may thereafter without notice exercise all voting and corporate rights at any meeting with respect to any Pledged Entity and exercise any and all rights of conversion, exchange, subscription or any other rights, privileges or options pertaining to any of the Pledged Interests as if it were the absolute owner thereof, including, without limitation, the right to vote in favor of, and to exchange at its discretion any and all of the Pledged Interests upon the merger, consolidation, reorganization, recapitalization or other readjustment with respect to any Pledged Entity or upon the exercise by any Pledged Entity, any Grantor or the Secured Party of any right, privilege or option pertaining to any of the Pledged Interests, and in connection therewith, to deposit and deliver any and all of the Pledged Interests with any committee, depository, transfer agent, registrar or other designated agency upon such terms and conditions as the Secured Party may reasonably determine, all without liability except to account for property actually received by the Secured Party, but the Secured

Party shall have no duty to exercise any of the aforesaid rights, privileges or options and shall not be responsible for any failure to do so or delay in so doing.

1.17 No Disposition, Etc. Until the irrevocable payment in full of the Obligations (other than inchoate indemnity obligations), each Grantor agrees that, except as permitted under the Note Documents, it will not sell, assign, transfer, exchange, or otherwise dispose of, or grant any option with respect to, the Pledged Interests or any other Pledged Collateral, nor will such Grantor create, incur or permit to exist any pledge, lien, mortgage, hypothecation, security interest, charge, option or any other encumbrance with respect to any of the Pledged Interests or any other Pledged Collateral, or any interest therein, or any proceeds thereof, except for the lien and security interest of the Secured Party provided for by this Agreement, the other Security Documents and the Permitted Liens described in clause (B) or (C) of the definition thereof.

#### 1.18 Payment of Expenses.

The Grantors shall, jointly and severally, pay all invoiced costs and expenses (including attorneys' fees) incurred by the Secured Party and its Affiliates and each Holder in connection with the administration and enforcement of this Agreement and the other Note Documents, and in connection with any amendments, waivers or consents under or in respect of this Agreement, the Notes or any other Note Document (whether or not such amendment, waiver or consent becomes effective), including: (a) the costs and expenses incurred in enforcing or defending (or determining whether or how to enforce or defend) any rights under this Agreement, the Notes or any other Note Document or in responding to any subpoena or other legal process or informal investigative demand issued in connection with this Agreement, the Notes or any other Note Document, or by reason of being a holder of any Note, and (b) the costs and expenses, including the fees of one financial advisor for all holders of the Notes incurred in connection with the insolvency or bankruptcy of any Grantor or any Subsidiary thereof or in connection with any work-out or restructuring of the transactions contemplated hereby and by the Notes or any other Note Document.

#### Section 5. <u>Lien Absolute; Waiver of Suretyship Defenses.</u>

#### 1.1 Lien Absolute, Waivers.

- (a) Until payment in full of the Obligations (other than inchoate indemnity obligations), all rights of Secured Party hereunder, and all obligations of Grantors hereunder, shall be absolute and unconditional irrespective of, shall not be affected by, and shall remain in full force and effect without regard to, and each Grantor hereby waives all, rights, claims or defenses that it might otherwise have (now or in the future) with respect to, in each case, each of the following (whether or not such Grantor has knowledge thereof):
- i) the validity or enforceability of the Note Documents, any of the Obligations or any guarantee or right of offset with respect thereto at any time or from time to time held by the Secured Party;
- ii) any renewal, extension or acceleration of, or any increase in the amount of the Obligations, or any amendment, supplement, modification or waiver of, or any consent to departure from, the Note Documents;
- iii) any failure or omission to assert or enforce or agreement or election not to assert or enforce, delay in enforcement, or the stay or enjoining, by order of court, by operation of law or otherwise, of the exercise or enforcement of, any claim or demand or any right, power or remedy (whether arising under any Note Documents, at law, in equity or otherwise) with respect to the Obligations or any agreement relating thereto, or with respect to any other guaranty of or security for the payment of the Obligations;
- iv) any change, reorganization or termination of the corporate structure or existence of any Grantor or any of their Subsidiaries and any corresponding restructuring of the Obligations;

- v) any settlement, compromise, release, or discharge of, or acceptance or refusal of any offer of payment or performance with respect to, or any substitutions for, the Obligations or any subordination of the Obligations to any other obligations;
- vi) the validity, perfection, non-perfection or lapse in perfection, priority or avoidance of any security interest or lien, the release of any or all collateral securing, or purporting to secure, the Obligations or any other impairment of such collateral;
- vii) any exercise of remedies with respect to any security for the Obligations (including, without limitation, any collateral, including the Collateral securing or purporting to secure any of the Obligations) at such time and in such order and in such manner as the Secured Party may decide and whether or not every aspect thereof is commercially reasonable and whether or not such action constitutes an election of remedies and even if such action operates to impair or extinguish any right of reimbursement or subrogation or other right or remedy that any Grantor would otherwise have and without limiting the generality of the foregoing or any other provisions hereof, each Grantor hereby expressly waives any and all benefits which might otherwise be available to such Grantor under applicable law;
- viii) any other circumstance whatsoever which may or might in any manner or to any extent vary the risk of any Grantor as an obligor in respect of the Obligations or which constitutes, or might be construed to constitute, an equitable or legal discharge of the Pledgor or any other Grantor for the Obligations, or of such Grantor under the guarantee contained in the Note Documents or of any security interest granted by any Grantor, whether in a Bankruptcy Proceeding or in any other instance;
- ix) any right to require Secured Party, as a condition of payment or performance by such Grantor, to (i) proceed against any other guarantor (including any other Grantor) of the Obligations or any other Person, (ii) proceed against or exhaust any security held from any such other guarantor or any other Person, (iii) proceed against or have resort to any balance of any deposit account or credit on the books of Secured Party in favor of any Grantor, any such other guarantor or any other Person, or (iv) pursue any other remedy in the power of Secured Party whatsoever;
- x) any defense based upon any statute or rule of law which provides that the obligation of a surety must be neither larger in amount nor in other respects more burdensome than that of the principal; and
- xi) any defenses (other than the defense of payment or release in accordance with the Note Documents) or benefits that may be derived from or afforded by law which limit the liability of or exonerate guarantors or sureties, or which may conflict with the terms of this Agreement.
- (b) In addition, each Grantor further waives any and all other defenses, set- offs or counterclaims (other than a defense of payment or performance in full hereunder) which may at any time be available to or be asserted by it, any Grantor or Person against the Secured Party, including, without limitation, failure of consideration, breach of warranty, statute of frauds, statute of limitations, accord and satisfaction and usury.
- (c) Each Grantor waives diligence, presentment, protest, marshaling, demand for payment, notice of dishonor, notice of default and notice of nonpayment to or upon any of the Grantors with respect to the Obligations. Except for notices provided for herein, each Grantor hereby waives notice (to the extent permitted by applicable law) of any kind in connection with this Agreement or any collateral securing the Obligations, including, without limitation, the Collateral. When making any demand hereunder or otherwise pursuing its rights and remedies hereunder against any Grantor, the Secured Party may, but shall be under no obligation to, make a similar demand on or otherwise pursue such rights and remedies as it may have against any Grantor or any other Person or against any collateral security or guarantee for the Obligations or any right of offset with respect thereto, and any failure by the Secured Party to make any such demand, to pursue such other rights or remedies or to collect any payments from any Grantor or any other Person or to realize upon any such collateral security or guarantee or to exercise any such right of offset, or any release of any Grantor or any other Person or any such collateral security, guarantee or right of offset, shall not relieve any Grantor of any obligation or liability hereunder, and shall not impair or affect the rights and remedies, whether express, implied or available as a matter of law, of the Secured Party against any Grantor.

For the purposes hereof "demand" shall include the commencement and continuance of any legal proceedings.

#### Section 6. Miscellaneous.

- 1.1 No Waiver. No failure on the part of the Secured Party or any of its Representatives to exercise, and no course of dealing with respect to, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise by the Secured Party or any of its Representatives of any right, power or remedy hereunder preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The rights and remedies hereunder provided are cumulative and may be exercised singly or concurrently, and are not exclusive of any rights and remedies provided by law.
- 1.2 <u>Governing Law.</u> All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by the internal laws of the State of New York, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of New York or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of New York.
- 1.3 Notices. All notices, approvals, requests, demands and other communications hereunder shall be delivered or made in the manner set forth in, and shall be effective in accordance with the terms of, the Notes; <u>provided</u>, that, to the extent any such communication is being made or sent to the Secured Party, such communication shall be made to the Secured Party at the address set forth below the Secured Party's signature hereto; provided, further, that any communication to a Grantor (other than the Pledgor) may be made to the address of the Pledgor. The Pledgor and the Secured Party may change their respective notice addresses by written notice given to the other parties hereto 10 days following the effectiveness of such change.
- 1.4 <u>Amendments, Etc.</u> The terms of this Agreement may be waived, altered or amended only by an instrument in writing duly executed by the Grantors and the Secured Party, except that the Schedules hereto shall be deemed amended and supplemented by any information set forth from time to time in writing delivered by any Grantor to the Secured Party. Any such amendment or waiver shall be binding upon the Secured Party and the Grantors and their respective successors and assigns.
- 1.5 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the respective successors and assigns of each of the parties hereto; <u>provided</u>, that no Grantor shall assign or transfer any of its rights or obligations hereunder without the prior written consent of the Secured Party. The Secured Party, in its capacity as the collateral agent, may assign its rights and obligations hereunder without the consent of any Grantor.
- 1.6 <u>Counterparts: Headings</u>. This Agreement may be authenticated in any number of counterparts, all of which taken together shall constitute one and the same instrument and any of the parties hereto may authenticate this Agreement by signing any such counterpart. This Agreement may be authenticated by manual signature or facsimile, .pdf or similar electronic signature, all of which shall be equally valid. The headings in this Agreement are for convenience of reference only and shall not alter or otherwise affect the meaning hereof.
- 1.7 <u>Severability</u>. If any provision hereof is invalid and unenforceable in any jurisdiction, then, to the fullest extent permitted by law, (a) the other provisions hereof shall remain in full force and effect in such jurisdiction and shall be liberally construed in favor of the Secured Party and its Representative in order to carry out the intentions of the parties hereto as nearly as may be possible and (b) the invalidity or unenforceability of any provision hereof in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction.
- 1.8 <u>SUBMISSION TO JURISDICTION; WAIVER OF VENUE; SERVICE OF PROCESS</u>. EACH GRANTOR (A) AGREES THAT ANY SUIT, ACTION OR PROCEEDING AGAINST IT ARISING OUT OF OR RELATING TO THIS AGREEMENT MAY BE INSTITUTED IN ANY U.S. FEDERAL COURT WITH APPLICABLE SUBJECT MATTER JURISDICTION SITTING IN THE CITY OF NEW YORK; (B) WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, (I) ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY SUCH SUIT, ACTION OR

PROCEEDING; AND (II) ANY CLAIM THAT IT MAY NOW OR HEREAFTER HAVE THAT ANY SUCH SUIT, ACTION OR PROCEEDING IN SUCH A COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM; AND (C) SUBMITS TO THE NONEXCLUSIVE JURISDICTION OF SUCH COURTS IN ANY SUCH SUIT, ACTION OR PROCEEDING; PROVIDED THAT ANY SUIT SEEKING ENFORCEMENT AGAINST ANY COLLATERAL OR OTHER PROPERTY MAY BE BROUGHT, AT THE SECURED PARTY'S OPTION, IN THE COURTS OF ANY JURISDICTION WHERE SUCH COLLATERAL OR OTHER PROPERTY MAY BE FOUND. EACH GRANTOR HEREBY EXPRESSLY AND IRREVOCABLY SUBMITS TO THE JURISDICTION OF THE FOREGOING COURTS FOR THE PURPOSE OF ANY SUCH LITIGATION AS SET FORTH ABOVE. EACH GRANTOR FURTHER IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS BY REGISTERED MAIL, POSTAGE PREPAID, OR BY PERSONAL SERVICE WITHIN OR WITHOUT THE STATE OF NEW YORK. EACH GRANTOR HEREBY EXPRESSLY AND IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY SUCH LITIGATION BROUGHT IN ANY SUCH COURT REFERRED TO ABOVE AND ANY CLAIM THAT ANY SUCH LITIGATION HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.

- 1.9 WAIVER OF RIGHT TO TRIAL BY JURY. EACH GRANTOR AND THE SECURED PARTY HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO A TRIAL BY JURY OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY, IN ANY ACTION, PROCEEDING OR OTHER LITIGATION OF ANY TYPE BROUGHT BY ANY OF THE PARTIES AGAINST ANY OTHER PARTY OR PARTIES, WHETHER WITH RESPECT TO CONTRACT CLAIMS, TORT CLAIMS, OR OTHERWISE. EACH GRANTOR AND THE SECURED PARTY HEREBY AGREE THAT ANY SUCH CLAIM OR CAUSE OF ACTION SHALL BE TRIED BY A COURT TRIAL WITHOUT A JURY. WITHOUT LIMITING THE FOREGOING, THE PARTIES FURTHER AGREE THAT THEIR RESPECTIVE RIGHT TO A TRIAL BY JURY IS WAIVED BY OPERATION OF THIS SECTION 6.9 AS TO ANY ACTION, COUNTERCLAIM OR OTHER PROCEEDING WHICH SEEKS, IN WHOLE OR IN PART, TO CHALLENGE THE VALIDITY OR ENFORCEABILITY OF THIS AGREEMENT OR ANY PROVISION HEREOF. THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AGREEMENT.
- 1.10 <u>Survival</u>. All representations, warranties, covenants and agreements of the Grantors and the Secured Party shall survive the execution and delivery of this Agreement.

# 1.11 <u>Collateral Agent for Holders.</u>

- (a) The Holders have, pursuant to Section 9(t) of the Securities Purchase Agreement, designated and appointed the Secured Party as the collateral agent of the Holders under this Agreement and the other Note Documents.
- (b) The Secured Party shall have the discretion to allocate proceeds received by the Secured Party pursuant to the exercise of remedies under the Note Documents or at law or in equity (including without limitation with respect to any secured creditor remedies exercised against the Collateral and any other security provided for under any Security Documents) to the then outstanding Obligations in such order as the Secured Party shall elect.

#### 1.12 Collateral Agent.

(a) The Secured Party is hereby designated as the collateral agent under this Agreement, the Security Documents and the Note Documents, and the Holders irrevocably authorize the Secured Party to take such action on their behalf under the provisions of this Agreement, the Security Documents and the other Note Documents, and to exercise such powers and perform such duties as are expressly delegated to the Secured Party by the terms of this Agreement and the other Note Documents, and consents and agrees to the terms of the each Note Document, as the same may be in effect or may be amended, restated, supplemented or otherwise modified from time to time in accordance with their respective terms. Each Holder, by accepting the benefits of this Agreement, agrees to the appointment of the Secured Party pursuant to this Section 6.12. The Secured Party agrees to act as such on the express conditions contained in this Section 6.12. The Holders agree that any action taken by the Secured Party in accordance with the provisions of this Agreement and the

other Note Documents, and the exercise by the Secured Party of any rights or remedies set forth herein and therein shall be authorized and binding upon the Holders. Notwithstanding any provision to the contrary contained elsewhere in this Agreement and the other Note Documents, the duties of the Secured Party shall be ministerial and administrative in nature, and the Secured Party shall not have any duties or responsibilities, except those expressly set forth herein and in the Security Documents, to which the Secured Party is a party, nor shall the Secured Party have or be deemed to have any trust or other fiduciary relationship with the Holders, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into the Security Documents and the other Note Documents, or otherwise exist against the Secured Party. Without limiting the generality of the foregoing sentence, the use of the term "agent" in this Agreement with reference to the Secured Party is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties.

- (b) The Secured Party may perform any of its duties under this Agreement or the other the Note Documents, by or through receivers, agents, employees, attorneys-in-fact or with respect to any specified Person, such Person's Affiliates, and the respective officers, directors, employees, agents, advisors and attorneys-in-fact of such Person and its Affiliates (a "Related Person"), and shall be entitled to advice of counsel concerning all matters pertaining to such duties, and shall be entitled to act upon, and shall be fully protected in taking action in reliance upon any advice or opinion given by legal counsel. The Secured Party shall not be responsible for the acts or omissions of any receiver, agent, employee, attorney-in-fact or Related Person that it selects as long as such selection was made in good faith and with due care.
- (c) The Secured Party shall be entitled to rely, and shall be fully protected in relying, upon any writing, resolution, notice, consent, certificate, affidavit, order, letter, telegram, facsimile, certification, telephone message, statement, or other communication, document or conversation (including those by telephone or e-mail) believed by it to be genuine and correct and to have been signed, sent, or made by the proper Person or Persons, and upon advice and statements of legal counsel (including, without limitation, counsel to the any Grantor), independent accountants and other experts and advisors selected by the Secured Party. The Secured Party shall not be bound to make any investigation into the facts or matters stated in any resolution, certificate, statement, instrument, opinion, report, notice, request, direction, consent, order, bond, debenture, or other paper or document. The Secured Party shall be fully justified in failing or refusing to take any action under this Agreement and the other Note Documents. The Secured Party shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement and the other Note Documents, in accordance with a request, direction, instruction or consent of the Holders and such request and any action taken or failure to act pursuant thereto shall be binding upon all of the Holders.
  - (d) The Secured Party shall not be deemed to have knowledge or notice of the occurrence of any default or Event of Default.
- (e) The Secured Party may resign at any time by 5 days' written notice to the Holders, such resignation to be effective upon the acceptance of a successor agent to its appointment as Secured Party. If the Secured Party resigns under this Agreement, the Pledgor shall appoint a successor collateral agent. If no successor collateral agent is appointed pursuant to the preceding sentence within ten (10) days after the intended effective date of resignation (as stated in the notice of resignation) the Secured Party shall be entitled to petition a court of competent jurisdiction to appoint a successor. Upon the acceptance of its appointment as successor collateral agent hereunder, such successor collateral agent shall succeed to all the rights, powers and duties of the retiring Secured Party, and the retiring Secured Party's appointment, powers and duties as the Secured Party shall be terminated. After the retiring Secured Party's resignation hereunder, the provisions of this Section 6.12(e) shall continue to inure to its benefit and the retiring Secured Party shall not by reason of such resignation be deemed to be released from liability as to any actions taken or omitted to be taken by it while it was the Secured Party under this Agreement.
- (f) High Trail Investments ON LLC shall initially act as collateral agent and shall be authorized to appoint co-collateral agents as necessary in its sole discretion. Except as otherwise explicitly provided herein or in the Security Documents, neither the Secured Party nor any of its respective officers, directors, employees or agents or other Related Persons shall be liable for failure to demand, collect or realize upon any of the Collateral or for any delay in doing so or shall be

under any obligation to sell or otherwise dispose of any Collateral upon the request of any other Person or to take any other action whatsoever with regard to the Collateral or any part thereof. The Secured Party shall be accountable only for amounts that it actually receives as a result of the exercise of such powers, and neither the Secured Party nor any of its officers, directors, employees or agents shall be responsible for any act or failure to act hereunder, except for its own gross negligence or willful misconduct.

- (g) The Secured Party is authorized and directed to (i) enter into the Security Documents to which it is party, whether executed on or after the Closing, (ii) bind the Holders on the terms as set forth in the Security Documents and the other Note Documents, and (iii) perform and observe its obligations under the Security Documents and the other Note Documents.
- (h) The Secured Party shall have no obligation whatsoever to assure that the Collateral exists or is owned by any Grantor or is cared for, protected, or insured or has been encumbered, or that the Secured Party's Liens have been properly or sufficiently or lawfully created, perfected, protected, maintained or enforced or are entitled to any particular priority, or to determine whether all or part of the Grantor's property constituting Collateral intended to be subject to the Lien and security interest of the Security Documents has been properly and completely listed or delivered, as the case may be, or the genuineness, validity, marketability or sufficiency thereof or title thereto, or to exercise at all or in any particular manner or under any duty of care, disclosure, or fidelity, or to continue exercising, any of the rights, authorities, and powers granted or available to the Secured Party pursuant to this Agreement, any Security Document or the other Note Documents.
- (i) No provision of this Agreement, any Security Document or the other Note Documents shall require the Secured Party to expend or risk its own funds or otherwise incur any financial liability in the performance of any of its duties hereunder or thereunder or to take or omit to take any action hereunder or thereunder unless it shall have received indemnity satisfactory to the Secured Party in its sole discretion against potential costs and liabilities incurred by the Secured Party relating thereto. Notwithstanding anything to the contrary contained in this Agreement, the Security Documents or the other Note Documents, in the event the Secured Party is entitled or required to commence an action to foreclose or otherwise exercise its remedies to acquire control or possession of the Collateral, the Secured Party shall not be required to commence any such action or exercise any remedy or to inspect or conduct any studies of any property under the Mortgages or take any such other action if the Secured Party has determined that the Secured Party may incur personal liability as a result of the presence at, or release on or from, the Collateral or such property, of any hazardous substances. The Secured Party shall at any time be entitled to cease taking any action described in this clause (i) if it no longer reasonably deems any indemnity, security or undertaking to be sufficient.
- (j) The Secured Party (i) shall not be liable for any action taken or omitted to be taken by it in connection with this Agreement, any Security Document, the other Note Documents, or any instrument referred to herein or therein, except to the extent that any of the foregoing are found by a final, non-appealable judgment of a court of competent jurisdiction to have resulted from its own gross negligence or willful misconduct, (ii) shall not be liable for interest on any money received by it and (iii) may consult with counsel of its selection and the advice or opinion of such counsel as to matters of law shall be full and complete authorization and protection from liability in respect of any action taken, omitted or suffered by it in good faith and in accordance with the advice or opinion of such counsel. The grant of permissive rights or powers to the Secured Party shall not be construed to impose duties to act.
- (k) The Secured Party shall not be liable for delays or failures in performance resulting from acts beyond its control. Such acts shall include but not be limited to acts of God, strikes, lockouts, riots, acts of war, epidemics, governmental regulations superimposed after the fact, fire, communication line failures, computer viruses, power failures, earthquakes or other disasters. The Secured Party shall not be liable for any indirect, special, punitive, incidental or consequential damages (included but not limited to lost profits) whatsoever, even if it has been informed of the likelihood thereof and regardless of the form of action.
- (l) The Secured Party does not assume any responsibility for any failure or delay in performance or any breach by the Pledgor or any Grantor under this Agreement, the Security Documents and the other Note Documents. The Secured Party shall not be responsible any Person for any recitals, statements, information, representations or warranties contained in this Agreement, the Security Documents, the other Note Documents, or in any certificate, report, statement, or other

document referred to or provided for in, or received by the Secured Party under or in connection with, this Agreement, the Security Documents or the other Note Documents; the execution, validity, genuineness, effectiveness or enforceability of the Security Documents and any other Note Document of any other party thereto; the genuineness, enforceability, collectability, value, sufficiency, location or existence of any Collateral, or the validity, effectiveness, enforceability, sufficiency, extent, perfection or priority of any Lien therein; the validity, enforceability or collectability of any obligations; the assets, liabilities, financial condition, results of operations, business, creditworthiness or legal status of any obligor; or for any failure of any obligor to perform its obligations under this Agreement, the Security Documents and the other Note Documents. The Secured Party shall have no obligation to any Person to ascertain or inquire into the existence of any default or Event of Default, the observance or performance by any obligor of any terms of this Agreement, the Security Documents or the other Note Documents, or the satisfaction of any conditions precedent contained in this Agreement, the Security Documents or the other Note Documents. The Secured Party shall not be required to initiate or conduct any litigation or collection or other proceeding under this Agreement, the Intercreditor Agreements, and the Security Documents unless expressly set forth hereunder or thereunder.

- The parties hereto hereby agree and acknowledge that the Secured Party shall not assume, be responsible for or otherwise be obligated for any liabilities, claims, causes of action, suits, losses, allegations, requests, demands, penalties, fines, settlements, damages (including foreseeable and unforeseeable), judgments, expenses and costs (including but not limited to, any remediation, corrective action, response, removal or remedial action, or investigation, operations and maintenance or monitoring costs, for personal injury or property damages, real or personal) of any kind whatsoever, pursuant to any environmental law as a result of this Agreement, the Security Documents or the other Note Documents or any actions taken pursuant hereto or thereto. Further, the parties hereto hereby agree and acknowledge that in the exercise of its rights under Agreement, the Security Documents or the other Note Documents, the Secured Party may hold or obtain indicia of ownership primarily to protect the security interest of the Secured Party in the Collateral and that any such actions taken by the Secured Party shall not be construed as or otherwise constitute any participation in the management of such Collateral. In the event that the Secured Party is required to acquire title to an asset for any reason, or take any managerial action of any kind in regard thereto, in order to carry out any fiduciary or trust obligation for the benefit of another, which in either of the Secured Party's sole discretion may cause the Secured Party to be considered an "owner or operator" under the provisions of the Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA"), 42 U.S.C. §9601, et seq., or otherwise cause the Secured Party to incur liability under CERCLA or any other federal, state or local law, each of the Secured Party and the Secured Party reserves the right, instead of taking such action, to either resign as the collateral agent or arrange for the transfer of the title or control of the asset to a court-appointed receiver. The Secured Party shall not be liable to the Pledgor, the Grantors, or any other Person for any environmental claims or contribution actions under any federal, state or local law, rule or regulation by reason of the Secured Party's actions and conduct as authorized, empowered and directed hereunder or relating to the discharge, release or threatened release of hazardous materials into the environment.
- (n) Subject to the provisions of the applicable Security Documents and the other Note Documents, the Secured Party shall execute and deliver this Agreement, the Security Documents and the other Note Documents to which it is a party and all agreements, documents and instruments incidental thereto, and act in accordance with the terms thereof. For the avoidance of doubt, the Secured Party shall have no discretion under this Agreement, the Security Documents or the other Note Documents and shall not be required to make or give any determination, consent, approval, request or direction without the written direction of the Holders.
- (o) The Secured Party is authorized to receive any funds for the benefit of itself, and the Holders distributed under the Security Documents and to make further distributions of such funds to itself and the Holders in accordance with the Security Documents and the other Note Documents.
- (p) In each case that the Secured Party may or is required hereunder or under any Security Document or the other Note Documents, to take any action (an "Action"), including without limitation to make any determination, to give consents, to exercise rights, powers or remedies, to release or sell Collateral or otherwise to act hereunder or under any Security Document or the other Note Documents, the Secured Party may seek direction from the Holders. The Secured Party shall not be liable with respect to any Action taken or omitted to be taken by it in accordance with the direction

from the Holders. If the Secured Party shall request direction from the Holders with respect to any Action, the Secured Party shall be entitled to refrain from such Action unless and until the Secured Party shall have received direction from the Holders, and the Secured Party shall not incur liability to any Person by reason of so refraining.

- (q) Notwithstanding anything to the contrary in this Agreement, the Security Documents or the other Note Documents, in no event shall the Secured Party be responsible for, or have any duty or obligation with respect to, the recording, filing, registering, perfection, protection or maintenance of the security interests or Liens intended to be created by this Agreement, the Security Documents or the other Note Documents (including without limitation the filing or continuation of any UCC financing or continuation statements or similar documents or instruments), nor shall the Secured Party be responsible for, and the Secured Party makes no representation regarding, the validity, effectiveness or priority of any of the Security Documents or the security interests or Liens intended to be created thereby.
- (r) Notwithstanding anything to the contrary contained herein, the Secured Party shall only act pursuant to the instructions of the Holders with respect to the Security Documents and the Collateral.
- 1.13 No Strict Construction. The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent, and no rules of strict construction will be applied against any party.
- 1.14 ENTIRE AGREEMENT; AMENDMENT. THIS AGREEMENT, TOGETHER WITH THE OTHER NOTE DOCUMENTS, SUPERSEDES ALL OTHER PRIOR ORAL OR WRITTEN AGREEMENTS BETWEEN THE SECURED PARTY, EACH GRANTOR, THEIR RESPECTIVE AFFILIATES AND PERSONS ACTING ON THEIR BEHALF WITH RESPECT TO THE MATTERS DISCUSSED HEREIN, AND THIS AGREEMENT, TOGETHER WITH THE OTHER NOTE DOCUMENTS AND THE OTHER INSTRUMENTS REFERENCED HEREIN AND THEREIN, CONTAINS THE ENTIRE UNDERSTANDING OF THE PARTIES WITH RESPECT TO THE MATTERS COVERED HEREIN AND THEREIN AND, EXCEPT AS SPECIFICALLY SET FORTH HEREIN OR THEREIN, NEITHER THE SECURED PARTY NOR ANY GRANTOR MAKES ANY REPRESENTATION, WARRANTY, COVENANT OR UNDERTAKING WITH RESPECT TO SUCH MATTERS. AS OF THE DATE OF THIS AGREEMENT, THERE ARE NO UNWRITTEN AGREEMENTS BETWEEN THE PARTIES WITH RESPECT TO THE MATTERS DISCUSSED HEREIN. NO PROVISION OF THIS AGREEMENT MAY BE AMENDED, MODIFIED OR SUPPLEMENTED OTHER THAN BY AN INSTRUMENT IN WRITING SIGNED BY THE GRANTORS AND THE SECURED PARTY.
- 1.15 <u>Grantor Acknowledgement</u>. Each Grantor acknowledges receipt of an executed copy of this Agreement. Each Grantor waives the right to receive any amount that it may now or hereafter be entitled to receive (whether by way of damages, fine, penalty, or otherwise) by reason of the failure of the Secured Party to deliver to such Grantor a copy of any financing statement or any statement issued by any registry that confirms registration of a financing statement relating to this Agreement.

[Signature Pages Follow]

IN WITNESS WHEREOF, the parties hereto have caused this Security Agreement to be duly executed and delivered as of the day and year first above written.

# PLEDGOR:

VELO3D, INC.

By: <u>/s/ Benyamin Buller</u>
Name: Benyamin Buller
Title: Chief Executive Officer

Notice Address:

511 Division St. Campbell, CA 95008 Attention: Chief Financial Officer Email: [\*]

**GRANTORS** 

VELO3D US, INC.

By: /s/ Benyamin Buller
Name: Benyamin Buller
Title: Chief Executive Officer

Notice Address:

511 Division St. Campbell, CA 95008 Attention: Chief Financial Officer Email: [\*]

[Signature Page to Security Agreement]

## SECURED PARTY:

# HIGH TRAIL INVESTMENTS ON LLC, as Secured Party

By: /s/[\*]\*

Name: [\*]

Title: Authorized Signatory

\*Authorized Signatory
[\*] Capital Management LP
Not individually, but solely as
Investment Advisor to High Trail
Investments ON LLC.

# Notice Address:

High Trail Investments ON LLC c/o High Trail Capital 80 River Street, Suite 4C Hoboken, NJ 07030 Attention: Eric Helenek Telephone: [\*] Email: [\*]

[Signature Page to Security Agreement]

## JOINDER AGREEMENT

This **JOINDER AGREEMENT**, dated as of [ ] (this "**Joinder Agreement**"), is delivered by [NAME OF ADDITIONAL GRANTOR], a [ ] (the "**Additional Grantor**") pursuant to the Security Agreement, dated as of [ ], 2023 (as amended, amended and restated, supplemented or otherwise modified from time to time, the "**Security Agreement**"), among Velo3D, Inc., as the Pledgor, the Grantors party thereto from time to time, and High Trail Investments ON LLC, as the Secured Party. Capitalized terms used herein but not otherwise defined herein shall have the meanings ascribed thereto in the Security Agreement.

#### RECITALS:

WHEREAS, the Pledgor and certain of its Subsidiaries (other than the Additional Grantor) have entered into the Security Agreement in favor of the Secured Party;

WHEREAS, the agreements, documents and instruments related to the Obligations secured by the Security Agreement require the Additional Grantor to become a party to the Security Agreement; and

WHEREAS, the Additional Grantor has agreed to execute and deliver this Joinder Agreement in order to become a party to the Security Agreement.

NOW, THEREFORE, IT IS AGREED:

### 1. Security Agreement.

By executing and delivering this Joinder Agreement, the Additional Grantor, as provided in Section 4.14 of the Security Agreement, hereby becomes a party to the Security Agreement as a Grantor thereunder with the same force and effect as if originally named therein as a Grantor and, without limiting the generality of the foregoing, hereby expressly assumes all obligations and liabilities of a Grantor thereunder. Attached hereto is a Pledge Supplement with the information with respect to the Additional Grantor required under Section 2 of the Security Agreement. The Additional Grantor hereby represents and warrants that each of the representations and warranties contained in Section 2 of the Security Agreement is true and correct on and as of the date hereof (after giving effect to this Joinder Agreement) as if made by such Additional Grantor on and as of such date.

## 2. Grant of Security Interest.

The Additional Grantor hereby assigns as collateral security to the Secured Party, and hereby grants to the Secured Party, a security interest in and continuing lien on, all of the Additional Grantor's right, title and interest in, to and under all of the property described in Section 3 of the Security Agreement, in each case, whether now owned or existing or hereafter acquired or arising and wherever located (collectively and together with the Collateral under the Security Agreement, the "Collateral"), for the prompt and complete payment and performance when due (whether at stated maturity, by required prepayment, declaration, acceleration, demand or otherwise) of the Obligations, subject to the exclusions, limitations and other provisions set forth in Section 3 of the Security Agreement.

3. Governing Law. All questions concerning the construction, validity, enforcement and interpretation of this Joinder Agreement shall be governed by the internal laws of the State of New York, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of New York or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of New York.

[SIGNATURE PAGE FOLLOWS]

_			
	y:		
	itle:		

## PLEDGE SUPPLEMENT

This PLEDGE SUPPLEMENT, dated as of [ ], is delivered by [NAME OF GRANTOR] a [ ] (the "Grantor") pursuant to the Security Agreement, dated as of [ ], 2023 (as amended, restated, supplemented or otherwise modified from time to time, the "Security Agreement"), among Velo3D, Inc., a Delaware corporation (the "Pledgor"), each of the Subsidiaries of the Pledgor from time to time party thereto (together with the Pledgor, the "Grantors"), High Trail Investments ON LLC, in its capacity as collateral agent for the benefit of the Holders (as defined in the Security Agreement) (together with its successors and assigns in such capacity, the "Secured Party"). Capitalized terms used herein but not otherwise defined herein shall have the meanings ascribed thereto in the Security Agreement.

The Grantor hereby confirms the grants of security interests to the Collateral Agent set forth in Section 3 of the Security Agreement. The Grantor represents and warrants that the attached supplements to Schedules accurately and completely set forth all additional information required pursuant to the Security Agreement and hereby agrees that such supplements to Schedules shall constitute part of the Schedules to the Security Agreement.

IN WITNESS WHER	EOF, the Grantor has caused this Pledge Suppler	ement to be duly executed and delivered	I by its duly authorized officer as of
the date first set forth above.		•	•

[NAME OF GRANTOR]
By: Name: Title:

[NOTE: Grantor to attach each of the Schedules to the Security Agreement necessary to reflect any changes to such Schedules.]

#### **CHANGE IN CONTROL AGREEMENT**

This Change in Control Agreement (the "Agreement") is entered into by and between [●] (the "Executive") and Velo3D, Inc., a Delaware corporation (the "Company"), on [●], 2023, and is effective as of [●] 2023 (the "Effective Date"). All capitalized terms are as defined in this Agreement.

- 1. **CIC Qualifying Termination**. If Executive is subject to a CIC Qualifying Termination, then, subject to Sections 2, 6, and 7 below, Executive will be entitled to the following benefits:
- (a) **Severance Payments**. The Company or its successor shall pay Executive a payment equal to (i) [*CEO*: twelve (12)] [*Other Key Executives*: nine (9)] months of his or her monthly base salary,(ii) Executive's then-current annual target bonus, and (iii) Executive's pro-rata annual target bonus for the year of the CIC Qualifying Termination based on the number of days of service in such year, in each case at the rate in effect immediately prior to the CIC Qualifying Termination. Such payment shall be made in a lump sum cash payment in accordance with the Company's standard payroll procedures, which payment will be made no later than the first regular payroll date occurring after the sixtieth (60th) day following the date of Executive's CIC Qualifying Termination provided that the Release Conditions have been satisfied.
- (b) **Equity**. The time-based vesting schedule of each of Executive's then outstanding Equity Awards shall fully accelerate and become vested as to 100% of the then-unvested shares subject to the Equity Award. Subject to Section 2, the accelerated vesting described above shall be effective as of the date of the CIC Qualifying Termination. For the avoidance of doubt, in order to give effect to the acceleration contemplated by this Section 1(b), each of Executive's outstanding Equity Awards shall remain outstanding and eligible to vest (solely pursuant to the terms of this Section 1(b)) for a period of three (3) months following a termination by the Company of Executive's employment for any reason other than Cause or Executive's resignation of his or her employment for Good Reason prior to a Change in Control and the acceleration will be effective upon the Change in Control, if such Change in Control closes within three (3) months following such termination or resignation.
- (c) Continued Employee Benefits. If Executive timely elects continued coverage under the Consolidated Omnibus Budget Reconciliation Act ("COBRA"), the Company shall pay the full amount of Executive's COBRA premiums on behalf of Executive for Executive's continued coverage under the Company's health, dental and vision plans, including coverage for Executive's eligible dependents, for [CEO: twelve (12)] [Other Key Executives: nine (9)] months following the last date on which Executive was eligible for such coverage as an active employee of the Company. Notwithstanding the foregoing, if the Company, in its sole discretion, determines that it cannot provide the foregoing subsidy of COBRA coverage without potentially violating or causing the Company to incur additional expense as a result of noncompliance with applicable law (including, without limitation, Section 2716 of the Public Health Service Act), the Company instead shall provide to Executive a taxable monthly payment in an amount equal to the monthly COBRA premium that Executive would be required to pay to continue the group health coverage in effect on the date of the CIC Qualifying Termination (which amount shall be based on the premium for the first month of COBRA coverage), which payments shall be made regardless of whether Executive elects COBRA continuation coverage and shall commence on the later of (i) the first day of the month following the date of Executive's CIC Qualifying Termination and (ii) the effective date of the Company's determination of violation of applicable law, and shall end on the date that is [CEO: twelve (12)] [Other Key Executives: nine (9)] months following the last date on which Executive was eligible for such coverage as an active employee of the Company, provided that, any taxable payments under Section 1(c) will not be paid before the first business day occurring after the sixtieth (60th) day following the date of Executive's CIC Qualifying Termination and, once they commence, will include any unpaid amounts accrued from the date of Executive's CIC Qualifying Termination (to the extent not otherwise satisfied with continuation coverage). Executive shall have no right to an additional gross-up payment to account for the fact that such COBRA premium amounts are paid on an after-tax basis.

- 2. **General Release**. Any other provision of this Agreement notwithstanding, the benefits under Section 1 shall not apply unless Executive (i) has executed a general release of all known and unknown claims that he or she may then have against the Company or persons affiliated with the Company and such release has become effective and (ii) has agreed not to prosecute any legal action or other proceeding based upon any of such claims. The release must be in the form prescribed by the Company, without alterations (this document effecting the foregoing, the "Release"). The Company will deliver the form of Release to Executive within thirty (30) days after the date of Executive's CIC Qualifying Termination or such other time limit as is expressly provided in the Release documents, provided however that in all cases the Release must be executed and have become irrevocable within sixty (60) days following the date of Executive's CIC Qualifying Termination.
- 3. Accrued Compensation and Benefits. Notwithstanding anything to the contrary in Section 1 above, in connection with any termination of employment (whether or not a CIC Qualifying Termination), the Company shall pay Executive's earned but unpaid base salary and other vested but unpaid cash entitlements for the period through and including the termination of employment, including unreimbursed documented business expenses incurred by Executive through and including the date of termination (collectively "Accrued Compensation and Expenses"), as required by law and the applicable Company plan or policy. In addition, Executive shall be entitled to any other vested benefits earned by Executive for the period through and including the termination date of Executive's employment under any other employee benefit plans and arrangements maintained by the Company, in accordance with the terms of such plans and arrangements, except as modified herein.

#### 4. Definitions.

- (a) "Board" means the Company's board of directors.
- (b) "Cause" shall mean, as reasonably determined by the Board, (a) Executive's material breach of the employment agreement between Executive and the Company, (b) Executive's commission of an act of fraud against the Company, (c) Executive's willful misconduct or failure to comply with the Company's policies or code of ethics, (d) Executive's conviction of, or plea of 'guilty" or 'no contest" to, a felony or other crime involving fraud, dishonesty or moral turpitude or (e) Executive's willful and continued failure to perform duties and responsibilities after receiving written notification of the failure from the [Company's Chief Executive Officer or other supervisor]<sup>1</sup> [Board]<sup>2</sup>. Notwithstanding the foregoing, in the case of clause (e), the Company will not terminate Executive's employment for Cause without first providing Executive with written notification of the acts or omissions constituting Cause and providing Executive with 30 days following such notice to cure such conduct (to the extent capable of cure)
  - (c) "Code" means the Internal Revenue Code of 1986, as amended.
- (d) "Change in Control." For all purposes under this Agreement, a Change in Control shall mean a Corporate Transaction, as such term is defined in the Plan, provided that the transaction (including any series of transactions) also qualifies as a change in control event under U.S. Treasury Regulation 1.409A-3(i)(5)(v) or(vii).
- (e) "CIC Qualifying Termination" means a Separation (A) upon or within twelve (12) months following a Change in Control, or (B) within three (3) months preceding a Change in Control resulting, in either case (A) or (B), from (i) the Company or its successor terminating Executive's employment for any reason other than Cause, or (ii) Executive voluntarily resigning his or her employment for Good Reason. A termination or resignation due to Executive's death or disability shall not constitute a CIC Qualifying Termination. In the case of a termination or resignation before a Change in Control, solely for purposes of benefits under this Agreement, the date of Executive's CIC Qualifying Termination will be deemed the date the Change in Control is consummated.

<sup>&</sup>lt;sup>1</sup> For the Key Executives (other than the CEO)s.

<sup>&</sup>lt;sup>2</sup> For the CEO.

- (f) "Equity Awards" means all options to purchase shares of Company common stock as well as all other stock-based awards granted to Executive, including but not limited to stock bonus awards, restricted stock, restricted stock units and stock appreciation rights, provided that those options and other stock-based awards vest exclusively with the passage of time and excluding any equity awards that are subject, in whole or in part, to any unsatisfied performance-based vesting conditions (it being understood that service-based vesting conditions alone shall not be deemed performance-based for this purpose).
- (g) "Good Reason" means, without Executive's prior written consent, (i) a material diminution of Executive's duties, authority, or responsibilities relative to Executive's duties, authority or responsibilities as an officer or employee in effect immediately prior to such reduction, provided that Executive no longer being an officer of a public company (e.g. the Company is taken private) or a change in reporting relationship will not constitute a material diminution if the Executive's duties and responsibilities otherwise remain substantially the same, (ii) a reduction by 10% or more of Executive's annual base salary (other than a reduction generally applicable to other senior executives of the Company and in generally the same proportion as for the Executive) or (iii) a requirement that Executive relocate Executive's principal place of work to a location which would result in an increase of more than twenty (20) miles of Executive's commuting distance over his or her commuting distance immediately prior to the Change in Control. For Executive to receive any benefits under this Agreement as a result of a resignation for Good Reason, all of the following requirements must be satisfied: (1) Executive must provide notice to the Company of his or her intent to assert Good Reason within sixty (60) days of the initial existence of one or more of the conditions set forth in subclauses (i) through (iii); (2) the Company will have thirty (30) days (the "Company Cure Period") from the date of such notice to remedy the condition and, if it does so, Executive may within ten (10) days of the earlier of expiration of the Company Cure Period or written notice from the Company that it will not undertake to cure the condition set forth in subclauses (i) through (iii). Should the Company remedy the condition as set forth above and then one or more of the conditions arises again, Executive may assert Good Reason again, subject to all of the conditions set forth herein.
  - (h) "Plan" means the Company's 2021 Equity Incentive Plan, as may be amended from time to time.
- (i) "Release Conditions" mean the following conditions: (i) Company has received Executive's executed Release and (ii) any rescission period applicable to Executive's executed Release has expired (without Executive having rescinded the executed Release).
  - (j) "Separation" means a "separation from service," as defined in the regulations under Section 409A of the Code.

## Successors.

- (a) **Company's Successors**. The Company shall require any successor (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets, by an agreement in substance and form satisfactory to Executive, to assume this Agreement and to agree expressly to perform this Agreement in the same manner and to the same extent as the Company would be required to perform it in the absence of a succession. For all purposes under this Agreement, the term "Company" shall include any successor to the Company's business and/or assets or which becomes bound by this Agreement by operation of law.
- (b) **Executive's Successors**. This Agreement and all rights of Executive hereunder shall inure to the benefit of, and be enforceable by, Executive's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees.

#### 6. Golden Parachute Taxes.

- (a) Best After-Tax Result. In the event that any payment or benefit received or to be received by Executive pursuant to this Agreement or otherwise ("Payments") would (i) constitute a "parachute payment" within the meaning of Section 280G of the Code and (ii) but for this subsection (a), be subject to the excise tax imposed by Section 4999 of the Code, any successor provisions, or any comparable federal, state, local or foreign excise tax ("Excise Tax"), then, subject to the provisions of Section 6, such Payments shall be either (A) provided in full pursuant to the terms of this Agreement or any other applicable agreement, or (B) provided as to such lesser extent which would result in no portion of such Payments being subject to the Excise Tax ("Reduced Amount"), whichever of the foregoing amounts, taking into account the applicable federal, state, local and foreign income, employment and other taxes and the Excise Tax (including, without limitation, any interest or penalties on such taxes), results in the receipt by Executive, on an after-tax basis, of the greatest amount of payments and benefits provided for hereunder or otherwise, notwithstanding that all or some portion of such Payments may be subject to the Excise Tax. Unless the Company and Executive otherwise agree in writing, any determination required under this Section shall be made by independent tax counsel designated by the Company and reasonably acceptable to Executive ("Independent Tax Counsel"), whose determination shall be conclusive and binding upon Executive and the Company for all purposes. For purposes of making the calculations required under this Section, Independent Tax Counsel may make reasonable assumptions and approximations concerning applicable taxes and may rely on reasonable, good faith interpretations concerning the application of Sections 280G and 4999 of the Code: provided that Independent Tax Counsel shall assume that Executive pays all taxes at the highest marginal rate. The Company and Executive shall furnish to Independent Tax Counsel such information and documents as Independent Tax Counsel may reasonably request in order to make a determination under this Section. The Company shall bear all costs that Independent Tax Counsel may reasonably incur in connection with any calculations contemplated by this Section. In the event that Section 6(a)(ii)(B) above applies, then based on the information provided to Executive and the Company by Independent Tax Counsel, Executive may, in Executive's sole discretion and within thirty (30) days of the date on which Executive is provided with the information prepared by Independent Tax Counsel, determine which and how much of the Payments (including the accelerated vesting of equity compensation awards) to be otherwise received by Executive shall be eliminated or reduced (as long as after such determination the value (as calculated by Independent Tax Counsel in accordance with the provisions of Sections 280G and 4999 of the Code) of the amounts payable or distributable to Executive equals the Reduced Amount). If the Internal Revenue Service (the "IRS") determines that any Payment is subject to the Excise Tax, then Section 6(b) hereof shall apply, and the enforcement of Section 6(b) shall be the exclusive remedy to the Company.
- (b) Adjustments. If, notwithstanding any reduction described in Section 6(a) hereof (or in the absence of any such reduction), the IRS determines that Executive is liable for the Excise Tax as a result of the receipt of one or more Payments, then Executive shall be obligated to surrender or pay back to the Company, within one-hundred twenty (120) days after a final IRS determination, an amount of such payments or benefits equal to the "Repayment Amount." The Repayment Amount with respect to such Payments shall be the smallest such amount, if any, as shall be required to be surrendered or paid to the Company so that Executive's net proceeds with respect to such Payments (after taking into account the payment of the Excise Tax imposed on such Payments) shall be maximized. Notwithstanding the foregoing, the Repayment Amount with respect to such Payments shall be zero (0) if a Repayment Amount of more than zero (0) would not eliminate the Excise Tax imposed on such Payments or if a Repayment Amount of more than zero would not maximize the net amount received by Executive from the Payments. If the Excise Tax is not eliminated pursuant to this Section 6(b), Executive shall pay the Excise Tax.

## 7. Miscellaneous Provisions.

- (a) Section 409A. To the extent (i) any payments to which Executive becomes entitled under this Agreement, or any agreement or plan referenced herein, in connection with Executive's termination of employment with the Company constitute deferred compensation subject to Section 409A of the Code and (ii) Executive is deemed at the time of such termination of employment to be a "specified" employee under Section 409A of the Code, then such payment or payments shall not be made or commence until the earlier of (i) the expiration of the six (6)-month period measured from the Executive's Separation; or (ii) the date of Executive's death following such Separation; provided, however, that such deferral shall only be effected to the extent required to avoid adverse tax treatment to Executive, including (without limitation) the additional twenty percent (20%) tax for which Executive would otherwise be liable under Section 409A(a)(1)(B) of the Code in the absence of such deferral. Upon the expiration of the applicable deferral period, any payments which would have otherwise been made during that period (whether in a single sum or in installments) in the absence of this paragraph shall be paid to Executive or Executive's beneficiary in one lump sum (without interest). Except as otherwise expressly provided herein, to the extent any expense reimbursement or the provision of any in-kind benefit under this Agreement (or otherwise referenced herein) is determined to be subject to (and not exempt from) Section 409A of the Code, the amount of any such expenses eligible for reimbursement, or the provision of any in-kind benefit, in one calendar year shall not affect the expenses eligible for reimbursement or in kind benefits to be provided in any other calendar year, in no event shall any expenses be reimbursed after the last day of the calendar year following the calendar year in which Executive incurred such expenses, and in no event shall any right to reimbursement or the provision of any in-kind benefit be subject to liquidation or exchange for another benefit. To the extent that any provision of this Agreement is ambiguous as to its exemption or compliance with Section 409A, the provision will be read in such a manner so that all payments hereunder are exempt from Section 409A to the maximum permissible extent, and for any payments where such construction is not tenable, that those payments comply with Section 409A to the maximum permissible extent. To the extent any payment under this Agreement may be classified as a "short-term deferral" within the meaning of Section 409A, such payment shall be deemed a short-term deferral, even if it may also qualify for an exemption from Section 409A under another provision of Section 409A. Payments pursuant to this Agreement (or referenced in this Agreement) are intended to constitute separate payments for purposes of Section 1.409A-2(b)(2) of the regulations under Section 409A. To the extent any nonqualified deferred compensation subject to Section 409A of the Code payable to Executive hereunder could be paid in one or more taxable years depending upon Executive completing certain employment-related actions (such as resigning after a failure to cure a Good Reason event and/or returning the Release), then any such payments will commence or occur in the later taxable year to the extent required by Section 409A of the Code.
- (b) Other Arrangements. This Agreement supersedes any and all cash severance arrangements and vesting acceleration arrangements under any agreement governing Equity Awards, severance and salary continuation arrangements, programs and plans which were previously offered by the Company to Executive, including employment agreement or offer letter, and Executive hereby waives Executive's rights to such other benefits. In no event shall any individual receive cash severance benefits under both this Agreement and any other vesting acceleration, severance pay or salary continuation program, plan or other arrangement with the Company.
- (d) **Dispute Resolution**. To ensure rapid and economical resolution of any and all disputes that might arise in connection with this Agreement, Executive and the Company agree that any and all disputes, claims, and causes of action, in law or equity, arising from or relating to this Agreement or its enforcement, performance, breach, or interpretation, will be resolved solely and exclusively by final, binding, and confidential arbitration, by a single arbitrator, in Santa Clara County, and conducted by Judicial Arbitration & Mediation Services, Inc. ("JAMS") under its then-existing employment rules and procedures. Nothing in this section, however, is intended to prevent either party from obtaining injunctive relief in court to prevent irreparable harm pending the conclusion of any such arbitration. Each party to an arbitration or litigation hereunder shall be responsible for the payment of its own attorneys' fees.

- (e) **Notice**. Notices and all other communications contemplated by this Agreement shall be in writing and shall be deemed to have been duly given when personally delivered or when mailed by U.S. registered or certified mail, return receipt requested and postage prepaid or deposited with Federal Express Corporation, with shipping charges prepaid. In the case of Executive, mailed notices shall be addressed to him or her at the home address which he or she most recently communicated to the Company in writing. In the case of the Company, mailed notices shall be addressed to its corporate headquarters, and all notices shall be directed to the attention of its Secretary.
- (f) **Waiver**. No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by Executive and by an authorized officer of the Company (other than Executive). No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.
- (g) Withholding Taxes. All payments made under this Agreement shall be subject to reduction to reflect taxes or other charges required to be withheld by law.
- (h) **Severability**. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision hereof, which shall remain in full force and effect.
- (i) **No Retention Rights**. Nothing in this Agreement shall confer upon Executive any right to continue in service for any period of specific duration or interfere with or otherwise restrict in any way the rights of the Company or any subsidiary of the Company or of Executive, which rights are hereby expressly reserved by each, to terminate his or her service at any time and for any reason, with or without Cause.
- ( j ) **Choice of Law**. The validity, interpretation, construction and performance of this Agreement shall be governed by the laws of the State of California (other than its choice-of-law provisions).

[Signature Page Follows]

IN WITNESS WHEREOF, each of the parties has executed this Agreement, in the case of the Company by its duly authorized officer, as of the day and year first above written.				
EXECUTIVE	VELO3D, INC.			
Print Name:	By: Title:			

# CERTIFICATION OF PERIODIC REPORT UNDER SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

## I, Benyamin Buller, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Velo3D, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 20, 2023 By: /s/ Benyamin Buller

Benyamin Buller Chief Executive Officer (Principal Executive Officer)

# CERTIFICATION OF PERIODIC REPORT UNDER SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

## I, Bernard Chung, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Velo3D, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 20, 2023 By: /s/ Bernard Chung

Bernard Chung Acting Chief Financial Officer (Principal Financial Officer)

## CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Benyamin Buller, Chief Executive Officer of Velo3D, Inc. (the "Company"), do hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that to the best of my knowledge:

- (1) the Quarterly Report on Form 10-Q of the Company for the quarter ended September 30, 2023 (the "Report"), as filed with the Securities and Exchange Commission, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company for the periods presented therein.

Date: November 20, 2023 By: /s/ Benyamin Buller

Benyamin Buller Chief Executive Officer (Principal Executive Officer)

## CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Bernard Chung, Acting Chief Financial Officer of Velo3D, Inc. (the "Company"), do hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that to the best of my knowledge:

- (1) the Quarterly Report on Form 10-Q of the Company for the quarter ended September 30, 2023 (the "Report"), as filed with the Securities and Exchange Commission, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company for the periods presented therein.

Date: November 20, 2023 By: /s/ Bernard Chung

Bernard Chung

Acting Chief Financial Officer (Principal Financial Officer)